November 23, 1982



SECY-82-465

FOR: The Commissioners

FROM: William J. Dircks Executive Director for Operations

SUBJECT: PRESSURIZED THERMAL SHOCK (PTS)

PURPOSE: To request Commission approval of recommended near-term actions related to protection against pressurized thermal shock events.

DISCUSSION: The issue of pressurized thermal shock (PTS) arises because in pressurized water reactors (PWRs) transients and accidents can occur that result in severe overcooling (thermal shock) of the reactor pressure vessel, concurrent with or followed by repressurization. In these PTS events, rapid cooling of the reactor vessel internal surface results in thermal stress with a maximum tensile stress at the inside surface of the vessel. The magnitude of the thermal stress depends on the temperature profile across the reactor vessel wall as a function of time. The effects of this thermal stress are compounded by pressure stresses if the vessel is pressurized.

> Severe reactor system overcooling events which could be accompanied by pressurization or repressurization of the reactor vessel (PTS events) can result from a variety of causes. These include system transients, some of which

Contact: Stephen H. Hanauer, NRR 492-7517

SECY NOTE: This subject is scheduled for consideration at an open meeting on December 1, 1982.



1

are initiated by instrumentation and control system malfunctions including stuck open valves in either the primary or secondary system, and postulated accidents such as small break loss-of-coolant accidents (LOCAs), main steam line breaks (MSLBs), and feedwater pipe breaks. Eight overcooling events have already occurred in which the primary coolant system water temperature rapidly decreased by 200°F or more.

As long as the fracture resistance of the reactor vessel material is relatively high, such events are not expected to cause vessel failure. However, the fracture resistance of reactor vessel materials decreases with exposure to fast neutrons during the life of a nuclear power plant. The rate of decrease is dependent on the metallurgical composition of the vessel walls and welds. If the fracture toughness of the vessel has been reduced sufficiently by neutron irradiation, severe PTS events could cause propagation of fairly small flaws that might exist near the inner surface. The assumed initial flaw might initiate and propagate into a crack through the vessel wall of sufficient extent to threaten vessel integrity and, therefore, core cooling capability.

The PTS issue is a concern only for operating PWRs. Boiling water reactors (BWRs) are not a significant PTS concern. BWRs operate with a large portion of water inventory inside the pressure vessel at saturated conditions. Any sudden cooling will condense steam and result in a pressure decrease, so simultaneous creation of high pressure and low temperature is improbable. Also contributing to the lack of PTS concerns for BWRs is the lower fast neutron fluence at the vessel inner wall, and the use of a thinner vessel wall which results in a lower stress intensity for a postulated crack.

Evaluations of Pressurized Thermal Shock by the NRC staff in the spring of 1981 concluded that no immediate licensing actions were required at that time, but that since the consequences of overcooling events increase as the vessels accumulate additional neutron irradiation, extensive further investigations were needed to determine whether and when corrective actions will be needed to provide assurance of vessel integrity throughout the intended service life of a reactor vessel.

On March 31, 1981, the NRC staff held the first of many meetings that were to occur over the following sixteen

The Commissioners

months with licensees, reactor manufacturers, and owners groups to discuss pressurized thermal shock concerns and exchange technical information.

The Commission was briefed regarding PTS on June 11, 1981 (See SECY 81-286, May 4, 1981).

In letters dated August 21, 1981, the staff requested the licensees of eight plants representative of older reactor. vessels to provide more detailed information on the present and projected pressure vessel materials properties, on the probability and possible severity of events that could cause failure of embrittled vessels, and on the efficacy and feasibility of several potential corrective actions.

In Commission briefings on September 15, 1981 (SECY 81-286A, September 8, 1981) and on November 24, 1981, the staff reported on the progress of these activities and indicated its plans to develop a near-term position on PTS by the summer of 1982.

On December 28, 1981, the Commission approved the staff's proposal to designate pressurized thermal shock as in Unresolved Safety Issue (SECY 81-687, December 8, 1981). The Task Action Plan for the PTS issue (A-49) comprises both the development of the near-term position and completion of longer-range studies by mid-1984.

On March 9, 1982, the staff briefed the Commission on the status of the staff reviews of the responses to the August 21, 1981 letters (SEC7 82-97, March 5, 1982). Many of the event-sequence analyses provided by licensees in response to the August 21, 1981 letter can be characterized as design-basis event analyses of the type generally submitted in Safety Analysis Reports (SARs) in support of license applications. Such analyses tend not to be of much help in evaluations of PTS. Many of the assumptions in such analyses were developed and accepted for licensing purposes without regard to PTS concerns. While SAR analyses appear to be appropriately conservative for calculations of reactor core thermal performance, PTS evaluations are most usefully performed using best-estimate calculations of pressure and temperature behavior. In addition, some potential event sequences that are not generally analyzed in detail in Safety Analysis Reports, because their consequences for core cooling are bounded by the design-basis event analyses, can be of greater significance for PTS

evaluations. Thus, it became clear to the staff that plant-specific PTS evaluations must include a systematic examination of many potential events, with particular attention to the probability and consequences of various possible operator actions and umissions, and equipment malfunctions. At the March 9 priefing, the staff indicated that in June 1982 is expected to provide a reassessment of the PTS issue directed at the question of whether corrective actions are required at any plant that must be initiated before the longer term PTS program provides generic resolution and acceptance criteria.

In early June 1982, the staft described to the ACRS, and to the PWR owners groups, the status of the staff's work and the approach it expected to take in its June reassessment. The owners' groups took issue with the bases for several of the staff's recommendations and indicated that recently developed information would be provided to the staff that would support different conclusions. The ACRS noted in its letter of June 7, 1982, that it had not been provided sufficient information to evaluate the adequacy of the staff's approach but expressed the belief that there is no need for any immediate plant modifications to permit continued operation.

As a result of these comments, and since the staff, the ACRS, and the industry had concluded that there was no need for immediate action, the staff decided that an additional effort should be made to review the forthcoming additional industry information, attempt to resolve technical differences, and modify the staff position, if appropriate, before moving forward to the CRGR and the Commission. This reevaluation raised a number of significant technical problems to be resolved as part of the re-formulation of the staff recommendations. As a result of intensive staff efforts and many discussions with the owners groups and consultants, the staff developed an initial draft evaluation report on September 13, 1982. Subsequent to additional staff and management reviews, review by the ACRS on October 8, and review by CRGR on October 6 and October 28, the draft staff report has been revised (Enclosure A) and forms the technical basis for the staff's recommendations to the Commission.

The staff PTS report (Enclosure A) describes the nature of the PTS concern, the staff's approach to assessment of the risk to the public from PTS events, and the staff's technical conclusions. This paper contains the staff's recommendations to the Commission for future actions that we believe are needed to maintain that risk at an acceptable level as further vessel irradiation is incurred.

Section 1 of Enclosure A gives a general background and introduction to the PTS issue.

Section 2 discusses the frequency and characterization of overcooling events that have actually been experienced at operating domestic PWRs. These events are used as the primary basis for determining the frequency of potential PTS related events that may reasonably be expected in the future.

Section 3 summarizes deterministic fracture mechanics calculations performed for these experienced events and parametric studies of crack growth potential as a function of the event characteristics and RT_{HDT} values. For each of the eight most significant experienced events, a fracture mechanics parametric study was performed to determine the critical material condition that would have resulted in vessel failure (due to the event) if a flaw were present of whatever size would be most likely to propagate. This single critical material condition (Critical Reference Temperature for the Nil-Ductility Transition, or RT_{NDTC}) was then used to characterize the severity of each of the experienced events.

In Section 4, the frequency of PTS events, and the severity of such events are combined to determine a value of RT_{NDT} below which the PTS risk is acceptable. That RT_{NDT} is recommended as a screening criterion.

Section 5 presents the staff's proposed conservative method for estimation of vessel-specific values of RT_{NDT} for comparison with the screening criterion.

Section 6 describes an evaluation of the frequency and character of potential lower probability overcooling events. These events have not occurred and so are evaluated to have a frequency lower than the eight events discussed in Section 2. The frequency and severity of .

several such groups of less frequent but potentially more severe events is estimated, using probabilistic methods.

Section 7 summarizes sensitivity studies performed using a probabilistic treatment of PTS fracture mechanics calculations. In this approach, significant PTS parameters are estimated in a more realistic, probabilistic way (for example, materials properties, and flaw size probability distributions). The result is an estimate of reactor vessel failure probability, given the occurrence of a specified overcooling transient.

In Section 8, the probabilistic fracture mechanics calculations are used in combination with the overcooling sequence results of Section 6 to estimate probabilities of vessel failure due to PTS.

Section 8 also considers how all of the above analyses affect our perception of PTS risk for a vessel with RT_{NDT} at the proposed screening criterion, and why those risks are believed to be acceptable.

Section 9 presents an outline of the plant specific safety evaluations proposed to be furnished well before a plant reaches the screening criterion. The analyses would be to determine risk due to PTS at the specific plant, and to define corrective actions needed if any, to reduce that risk to an acceptable value.

Section 10 of the staff PTS report (Enclosure h^{∞} mesents the staff conclusions.

Summary of Technical Conclusions and Recommendations As a result of evaluations performed thus far of the issue of pressurized thermal shock, the NRC staff has reached the following conclusions:

- (1) The risk from PTS events for reactor vessels with RT_{NDT} values less than the proposed screening criterion (270°F for axial welds, and 300°F for circumferential welds) is acceptable. On the basis of presently available information, no reactor vessel will exceed the screening criterion for the next few years, therefore there is no need to shutdown or anneal any operating PWR in the next few years.
- (2) Most plants can avoid reaching the screening criterion throughout their service life by timely implementation of flux reduction programs. Such

flux reduction programs should be implemented on a time schedule that will avoid foreclosure of this option.

- (3) Any plant for which the value of RT_{NDT} is projected to reach the screening criterion before the end of service life, using the conservative method of RT_{NDT} determination described in Section 5 and Appendix E of Enclosure A, should submit plant-specific evaluations (of the type described in Section 9 of Enclosure A) to determine what, if any, modifications to equipment, systems and procedures should be required to provide acceptable protection against vessel failure from PTS events for the remainder of plant life. These evaluations should be submitted three years before the vessel is projected to reach the screening criterion.
- (4) In the near future, the staff should develop more detailed guidance for these evaluations and acceptance criteria for determining whether plant modifications are needed based on the evaluations.
- (5) Some of the Commission's regulations (Appendix G to 10 CFR Part 50, 10 CFR 50.46, and possibly others) may not appropriately reflect current understanding of the state of reactor vessel embrittlement and the potential for vessel failure as a result of PTS. Timely consideration should be given to the need for amendments to the regulations (as discussed in Section 8.6 of Enclosure A).

Discussion of Proposed Near-Term Actions

(1) <u>Proposed Rulemaking</u>: The staff proposes to develop a notice of proposed rulemaking. The proposed rule would: establish an RT_{NDT} screening criterion of 270°F for axial welds, and 300°F for circumferential welds; require that licensees of all operating PWRs submit a determination of the present RT_{NDT} values for their reactor vessels, and the estimated date at which the RT_{NDT} value will exceed the screening criterion; require licensees to implement such flux reduction programs as are feasible and necessary to avoid reaching the screening criterion before expiration of the operating license; and require licensees of operating PWRs for which the RT_{NDT} value is projected to exceed the screening criterion before the expiration of the operating license to submit a plant-specific PTS safety analysis, of the type outlined in Section 9 of Enclosure A, three years before the screening criterion is exceeded, or one year after the effective date of the regulation, whichever is later. For purposes of comparison with the screening criterion, the rule would require calculation of RT_{NDT} values in the manner described in Section 5 of Enclosure A. During the public comment period on the proposed rulemaking, the staff would develop more detailed guidance on the plant-specific analyses to be required and on the acceptance criteria to be used in judging the acceptability of the results. Since most of the technical basis for such a rulemaking has already been assembled in Enclosure A and the documents referenced therein, the staff believes that the notice of proposed rulemaking could be prepared for Commission approval in about six months or less.

(2) Flux Reduction Programs

On the basis of information currently available to the staff, it appears that although some plants will require no remedial actions for their vessel RT_{NDT} to remain below the screening criterion throughout their service life, a substantial number of PWR vessels are now predicted to exceed the criterion well before the end of life. Flux reduction programs can reduce the rate of increase of RT_{NDT}, but the effectiveness of flux reduction depends on early implementation. The staff has developed information on the costs and benefits of near-term flux reduction measures in Appendix I of Enclosure A. Those plants for which the vessel RT_{NDT} is expected to exceed the screening criterion before the end of service life can generally be grouped into three categories.

For several plants, action within the next few years to reduce the flux at critical welds by a factor of two or less will ensure that they do not exceed the screening criterion throughout their service life.

The Commissioners

It appears that such a reduction can be achieved through installation of a low leakage core (installation of partially burned fuel assemblies in the periphery of the core in place of fresh fuel assemblies). This fuel management option is already being implemented by some licensees at reportedly little or no additional cost. For plants in this category, the staff believes that the rulemaking proposed above will provide a mechanism for ensuring prompt consideration of appropriate flux reduction measures.

There is a group of about eight plants for which near-term action to reduce the flux at critical welds by factors of two to five would ensure that they do not exceed the screening criterion throughout their service life. It appears to the staff that such flux reduction factors could be attained through the installation of a low leakage core and the replacement of a few (8 - 12) peripheral fuel assemblies by dummy assemblies. These measures seem practical and cost-effective to the staff, based on present knowledge, with some loss in margin to core overheating limits in certain postulated transients. We estimate that there would be an engineering cost (redesign and safety analyses) of about \$20 million per plant, and some small increase in fuel cycle or operating costs. For such plants, the staff proposes to meet with licensee management when the notice of proposed rulemaking is issued to discuss the importance of prompt detailed evaluation of near-term flux reductions that would avoid foreclosure of that option as an effective means of reducing PTS risk, and to determine the licensee's plans for such a program. Should these meetings not result in effective actions by the licensees, the staff would propose to issue letters pursuant to 10 CFR 50.54(f) requesting information "to enable the Commission to determine whether or not the license should be modified, suspended or revoked." The scope of the information request is planned to include the amount of flux reduction needed to stay within the screening criteria, the flux reduction attainable, safety evaluation for non-PTS aspects of flux reduction, other actions proposed by the licensee, and value/impact information.

One plant, H. B. Robinson Unit 2, is so close to reaching the screening criterion that the fuel management options described above could not reduce the flux at the critical welds sufficiently to prevent reaching the screening criteria before the end of service life. The staff proposes that Carolina Power and Light Company be ordered to submit a comprehensive plan showing what actions will be taken to resolve the PTS issue for the Robinson plant and to show cause why the license should not be modified to provide adequate protection for PTS for this plant.

Discussion of Longer-Term Actions

(1) The ongoing program to improve procedures and operator training regarding prevention and mitigation of PTS events should continue, as described in Appendix C of Enclosure A.

The staff has audited training and procedures at eight older PWRs. Emphasis during these audits was on procedural adequacy and the operators' understanding of PTS events and the potentially conflicting requirements of avoiding PTS situations while at the same time assuring adequate cooling to the core. The audit reports are summarized in Appendix C to Enclosure A. Generally, it was found that adequate procedures and training exist at the eight plants, although longer range improvements are desirable. The exceptions were noted, and were corrected promptly by the liconsees.

The industry is pursuing a major revision of emergency operating procedures as part of TMI Action Plan Item II.C.1.

Longer range improvements to the PTS related procedures and training will result from the integrated, long range reassessment of procedures in this program which is aimed at adopting "symptom oriented" in addition to "event oriented" procedures. That program is also discussed in Appendix C to Enclosure A.

The staff believes that it is important to avoid quick and/or frequent changes to procedures with consideration being focused on a particular concern.

Therefore, only the urgent and necessary PTS-related procedure changes are being implemented before the integrated program results.

(2) Industry and NRC programs are needed to provide additional confirmatory PTS information, to decrease the large uncertainty of current PTS analyses, to apply the analysis to B&W and CE plants, and to investigate more thoroughly the alternatives to reduce and mitigate PTS risks. In particular, the analytical and experimental studies underway as part of the NRC research program, as described in Appendix N of Enclosure A, should continue on a high-priority basis. These programs should improve the staff's capability for independent audits and assessments of licensee evaluations, confirm or improve calculations methods and assumptions, and aid in further assessments of safety margins.

As discussed in Section 2, of Enclosure A, it is particularly important to obtain B&W analyses to confirm the acceptability of the screening criteria for B&W plants, since B&W plant design is significantly different from <u>W</u> and CE. Also as noted in Section 2, operating experience shows a significantly different history of PTS precursors for B&W plants.

- (3) The best available methods should be used for periodic in-service inspection of high-RT_{NDT} vessels, to maximize the likelihood of detecting any flaws that may be present relevant to PTS. In addition to the necessary assurance of vessel integrity provided by such inspections, it would be useful to investigate how inspection results could be used to improve the flaw probability distributions to be used in plant-specific analyses.
- (4) The staff and the Commission should give timely consideration to the possible need to amend certain of the regulations to better reflect the potential for PTS.

Value-Impact

Enclosure B provides a preliminary value-impact assessment of the recommended near-term requirements for a screening criterion and analysis.

The Commissioners

The major costs involved in the presently proposed action are the cost to industry to perform the indicated analyses and the cost to NRC to develop guidelines and to review the results. The costs to implement actions found to be necessary have not been estimated and will have to be determined on a plant-specific basis.

The proposed requirements for plant-specific analyses are but the first of two stages in the resolution of pressurized thermal shock. Their costs are modest and appear well justified since the consequence of not performing the recommended analyses is that certain reactor vessels would become increasingly embrittled and the risk of pressure vessel failure would continue to increase. The second stage of resolution, implementation of any facility modifications needed to reduce the risk of vessel failure from pressurized thermal shock, could result in large costs, and result in significant risk reduction benefits. These costs can be minimized by the early implementation of flux reduction measures, as discussed above. Value-impact assessment for such implementation will have to be performed on a plant-specific basis.

There are no direct safety benefits from performing the plant specific analyses. The safety benefits will be derived from implementing any required corrective actions identified in these analyses. The plant specific analyses are a necessary step in this process. The operating experience record and probabilistic studies described in Enclosure A indicate that the potential safety benefits are large for initiating PTS corrective regulatory action for plants approaching the screening criterion.

<u>COORDINATION</u>: In a letter to the Chairman on June 7, 1982, (Enclosure C) the ACRS commented that it had not been given sufficient information to evaluate the adequacy of the NRC Staff's approach to develop regulatory requirements concerning pressurized thermal shock (PTS). In response to the comment, the Staff prepared and sent a draft report to the ACRS entitled, "Draft NRC Staff Evaluation of Pressurized Thermal Shock" dated September 13, 1982. This report, with supporting information in the Appendices, addressed the numerous technical aspects of the PTS issue.

During the 270th meeting of the ACRS, October 7-8, 1982, the ACRS reviewed this draft report and provided comments

- 12 -

The Commissioners

in a letter to the Chairman dated October 14, 1982, (Enclosure D). The staff responded to the ACRS in a letter dated November 12, 1982 (Enclosure E). We believe that the actions proposed in this paper, and the technical information in Enclosure A are consistent with the ACRS recommendations.

The ACRS "believes that if due consideration is given to the items mentioned above, and if regulatory actions are based on the proposed screening criteria, the pressurized thermal shock matter should not present an undue risk to the health and safety of the public."

The Committee recommended that an orderly, comprehensive research program is needed and suggested some program elements. The staff agrees that such a program is needed, and is re-evaluating the ongoing program to define any changes or additions that should be made.

The Committee recommended five items for special attention:

- Careful assessment of the uncertainties and special circumstances related to each data point used in the correlation of vessel properties with material composition and fast neutron irradiation, in the range of high embrittlement. The staff intends to do this.
- Improvements in PTS-related operator training and procedures. These have been accomplished for the seven plants already audited (see Appendix C of Enclosure A), and are planned as part of the comprehensive procedures improvement program for all plants.
- Better characterization of PTS initiating events and subsequent operator actions. This is planned to be a part of recommended Long Term Action (2).
- Consideration of heating the ECCS water. This is included in the plant-specific analysis program outlined in Section 9 of Enclosure A. At least two plants have already announced their intention of heating the ECCS water to alleviate potential PTS problems.
- Fast neutron fluence reduction. The staff proposes in this paper to pursue a vigorous program for all

- (3) Note that the staff will meet with licensees of plants for which near-term flux reductions of factors of two to five would ensure that the screening criterion would not be exceeded throughout service life, to determine the licensees' plans for such programs, and that the staff will propose issuance of 10 CFR 50.54(f) letters to such licensees if appropriate following the meetings;
- (4) Note that the staff will prepare additional guidance for the plant-specific PTS safety analyses referred to in the proposed rulemaking, during the public comment period on the proposed rule;
- (5) <u>Note</u> that the staff will consider the possible need to amend certain of the regulations to better reflect the potential for PTS, as discussed in Section 8.6 of Enclosure A.
- (6) Note that the staff will continue the ongoing program to improve procedures and operator training regarding prevention and mitigation of PTS events, as described in Appendix C of Enclosure A; and
- (7) Note that the analytical and experimental programs described in Appendix N of Enclosure A, and in the Task Action Plan for USI A-49, "Pressurized Thermal Shock," will continue on a high-priority basis.

inch hach

William J. Dircks Executive Director for Operations

Enclosures:

- *A. NRC Staff Evaluation of Pressurized Thermal Shock (November 17, 1982)
 - B. Value-Impact Assessment
 - C. Letter, P. Shewmon to Chairman Palladino, June 7, 1982
 - D. Letter, P. Shewmon to Chairman Palladino, October 14, 1982
 - E. Letter, W. J. Dircks to P. Shewmon, November 12, 1982
 - F. Minutes of CRGR Meeting No. 21, October 30, 1982
 - G. Minutes of CRGR Meeting No. 23, November 12, 1982

*Commissioners, SECY, OGC, OPE and EDO only. The Commissioners

Commissioners' comments should be provided directly to SECY by c.o.b. Friday, December 10, 1982.

Commission Staff Office comments, if any, should be submitted to the Commissioners <u>NLT December 3, 1982</u>, with an information copy to SECY. If the paper is of such a nature that it requires additional time for analytical review and comment, the Commissioners and the Secretariat should be apprised of when comments may be expected.

DISTRIBUTION: Commissioners SECY Regional Offices Congressional Affairs Inspector & ditor Public Affairs Office of Investigations EDO ELD ACRS ASLBP ASLAP ERRATA SHEET FOR ENCLOSURE A (PTS Report and Appendices)

(Enclosure A was distributed seperately from SECY-82-465)

- 1. Insert the attached page numbered 4-3 at the end of Section 4.
- 2. Insert the attached page numbered 5-7 at the end of Section 5.
- 3. Note that references to $\mathrm{RT}_{\mathrm{NDT}}$ on Figure 7.3 are to mean values of $\mathrm{RT}_{\mathrm{NDT}}$.
- 4. Insert the corrections shown on attached markup of pages 8-10 and 8-16 (Corrections have already been made by hand on most copies).
- 5. Replace page D-6 in Appendix D with the attached corrected page numbered D-6.
- 6. Replace all 5 pages of Table P-1 at the end of Appendix P with the attached 5 page "Rev. 1" Table P-1

plants. The data also show that the CE experience (zero events in 60 reactor years) does not statistically justify use of a different, less restrictive criterion for CE plants. Such justification, if presented in the near future, could not be based on statistics alone.

The use of an experience base of on 1 - 1 (th events to develop an expected frequency distribution 1 - 1 - 2 + 1) yields values with large uncertainties and does not take account of low frequency events that have not occurred. In addition, the temperature histories used in the fracture mechanics calculations were meaning or in the cold leg piping, whereas the temperature of integest (at highly irradiated welds) is in the reactor vessel downcomer which might have been colder.

The fracture mechanics calculations assume that flaws of critical size are present at the limiting welds (those with highest RT_{NDT}). This is clearly a conservatism in the analysis, which is considered in an approximate manner in the probabilistic fracture mechanics analysis (Section 7).

Because the intent is to select a screening criterion generically, covering a wide range of transient sequences, the analysis does not take credit for the warm prestress phenomenon which would be beneficial in many actual transient sequences. No account has been taken of the effects of weld residual stresses, which would generally add to the thermal and pressure stresses.

Perhaps the most significant uncertainty in the treatment described thus far is that there are potential low frequency overcooling events much more severe than those that have been observed. Because these events have not occurred, they have not been taken into account in the frequency distribution used.

Because of all of the nonquantified uncertainties noted above, the staff has also examined what insights can be gained from calculations of the characteristics of various postulated overcooling events and estimates of their expected frequency of occurrence; and from a probabilistic study of the fracture mechanics calculations. These considerations are described in Sections 6, 7 and 8.

4-3

5.4 "Evaluation of Irradiation Response of Reactor Pressure Vessel Materials" Prepared by Combustion Engineering, Inc. (J. D. Varsik principal investigator) for Electric Power Research Institute, Research Project RP 1553-1, Final Report, to be published. reactor accidents. For analyzing how PTS events contribute significantly to the risk to the public, the following logic applies:

1-

- 1. PTS event sequences leading to RPV failure have overall frequency F per reactor-year. Figures 8-2 and 8-3 provide a very approximate estimate of F. A plant evaluated (as described in Section 5 or 9 and Appendix E) to be at the 270°F screening criterion is likely to have a true RT_{NDT} of 150-270°F (two sigma 7= 60°F). For the mean of 210°F, F= 6 x 10-6 per reactor-year on the NRC curve (Figure 8-3), and much smaller on the WOG curve (Figure 8-2).
- 2. A fraction X<1 of RPV failure sequences leads to core melt, giving an expected value of XF core melts per reactor-year.
- 3. A fraction Y of failures leading to core melt leads to significantly early radioactive releases, so the expected value of the frequency of significant early releases due to PTS is XYF, which is therefore the expected value of the frequency of events involving non-zero early deaths due to PTS.
- 4. The risk of early deaths to the average individual within one mile of the site is given by XYFD, where the factor D includes the effects of dispersion and wind direction.
- 5. To show PTS risk to be lower than 10% of the safety goal guidelines would involve showing

XF< 10-5 per reactor-year

and

S x 10⁻¹ XYFD₂ XE-5 per reactor-year

The limiting value for XYFD is approximate, for an averge site.

PTS REPORT SEC 8

<u>Flaws and Cracks</u> - The deterministic calculations assume the presence of a long through-clad flaw of critical depth--a substantial conservatism. The probabilistic calculations use a through-clad crack probability that is highly uncertain and that some people believe is conservative. No account is taken of actual in-service inspection results in these generic calculations.

- 16 4 ...

The crack growth/arrest model used by the staff assumes long initial flaws that grow uniformly over their length. This initial flaw shape is conservative. The growth/arrest shape is discussed in Section 3; we believe that, once a crack initiates, the long crack is a more realistic description than less conservative shapes used in other models.

<u>Stresses</u> - the models include no residual stresses, which is non-conservative. The NRC model includes cladding effect, which is realistic for through-clad cracks. The probabilistic analysis of small-break LOCA used in Section 8 took credit for warm prestressing.

None of the moduls used for generic events currently includes a warm prestress (WPS), which is a conservatism for transients satisfying the WPS conditions. WPS is included in calculations specifically limited to the small-break LOCA, where it has been demospitrated that WPS is applicable.

<u>Material Properties</u> - The estimation of RT_{NDT} as described in Section 5.4 and 8.5, is a substantial conservatism.

<u>Fracture Mechanics</u> - The use of linear elastic fracture mechanics in the uppershelf temperature region is appropriate. Assuming vessel failure when the stress intensity factor for a crack reaches the upper-shelf toughness is believed by many people to be conservative because considerable ductility exists in the remaining ligament. Until we have validated applicable elastic-plastic models, however, the degree of conservatism cannot be determined.

<u>Uncertainties in Probabilistic Calculations</u> - Substantial uncertainties exist in probabilistic calculations as discussed in Section 8.3. The characterization of event sequences by T_f , β , and P is an oversimplification that may or may not be on the conservative side.

n.1.2 Stress Algorithms

The total peak stresses (thermal plus pressure plus residual plus any other stresses) are assumed to be less than, or at least not significantly larger than, the material yield strength so that components of stress can be added and that linear-elastic fracture mechanics procedures can be utilized. For rapid thermal transients, high stresses usually occur locally at the inner vessel wall and acceptable stress distributions (total stress below yield) over the remaining section can still be obtained if the overstressed region is relatively thin.

D.1.3 Postulated Initial Cracks

Long through-clad cracks, either axial or circumferential, are assumed to exist in the welds of limiting (highest) RT_{NDT}. In this case, the cladding effect is conservatively applied in that the stresses due to the different expansion coefficients of the clad and base metal are added to the nominal thermal stresses. For short through-clad cracks or underclad cracks it is conceivable that the cladding can have a beneficial effect if the cladding is sufficiently tough, that is, it is less affected by irradiation damage than the base material. In this case, it could deter crack elongation or could even prevent crack initiation depending on the specific transient. At present, there are differences of opinion as to clad toughness after irradiation, and further research is needed as to the behavior of short or underclad cracks in an overcooling event. Also, analyses to date omit consideration of weld residual stresses and in the case of circumferential cracks, the effect of dead weight stresses. (An uneven temperature distribution in the azimuthal f direction increases K, value for circumferential cracks. Therefore, the NRC concludes that the more conservative assumption of long through-clad cracks should be used at least for scoping calculations, until further information s developed to permit a relaxation of this assumption.

D.1.4 Fracture Mechanics Algorithms

Fracture mechanics analyses utilize the linear-elastic boundary integral methods of Heliot, Labbens and Pellissier-Tanon (References D.1 and D.2).

D-6

Plant NSS5/Vessel	EFPY as of	Fluence n/cm ²	Copper %	Nickel 3	Hean Initial	Mean ARI _{NDT}	2√08+02	RT _{NDT} , ^o F, of Drc 31,	as 1981(6)	Licensee's
Fabricators	12/31/81	×10 ^{1#}			RTNOT		(5)	Circum.	Axtal	1101
Robinson 2 W/CE	7.10	(14.1)(3)(8) 14.8 (3)(8)	(0.35) 0.27	(1.20) 0.20	(-56) -56	(303)(4) 151	34 (4) 59	281	154	290 220
Turkey Point 4 W/88W	5.67 No Axia	9.1 (9) 1 Welds	(0.32)	(0.57)	(0)	(200)	59	259		211
Turkey Point 3 W/B&W	5.67 No Axii	(9.1)(9) 1 Welds	(0.32)	(0.57)	(0)	(200)	59	259		
Fort Calhoun CE/CE	5.07	(7.04) 5.1 (10)	(0, 35) 0, 35	C. 99 0. 99	(-56) -56	(264)(4) 248 (4)	34(4) 34 (4)	242	226	209 (239)(7)
Maine Yankee CE/CE	5.90	(5.02) 4.14	(0.36) 0.36	(0.99) 0.99	(-56) -56	(248)(4) 238(4)	34(4) 34(4)	226	216	170 (198)(7)
Indian Point 3 ¥/CE	2.98 Plate Go	(1.67) verns	(0.24) 0.24	(0.52) 0.52	(+74) +74	(90) 90	48 48	212	212	
Yankee Rowe W/B&W/B&W	14.56 Plate Go	(11.35) verns	(9,20) 0,20	(0.63) 0.63	(+30) +30	(133) 133	48 48	211	211	
Rancho Seco B&W/B&W	3.54	(2.33) 2.05	(0.31) 0.35	(0,59) 0.59	(0) 0	(135) 148	59 59	194	207	•
Three Mile Island 1 B&W/B&W	3.52	(1.87) (1.87)	(0.31) 0.35	(0.68) 0.60	(0) 0	(133) 145	59 59	192	204	145
Oconee 2	4.7 <u>1</u> No Arial	(2.87) Welds	(0.35)	(0.71)	(0)	(172)	59	231		

Table P.1 RT_{NUT} Values for All Plants⁽¹⁾ Calculated Per the Recommendations of the Working Group on RT_{NUT}⁽²⁾ for the Vessel Inside Surface.

(1) See footnote(s), last page of table.

These values are subject to change when plant-specific analyses yield better information.

Table	P-1	(fontinued)	
10012	- I	ACONCINGED .	•

Plant NSSS /Voccol	EFPY	Fluence	Copper	Nickel	Mean Initial	Hean	2102+02	RT _{NDT} F.	as	Licensee's
Fabricators	12/31/81	x10 ^{1.8}	·	~	RT _{NDT}	DUINDT	(5)	of Dec 31, Circum.	1981(6) Axial	RINDT. CF
Point Beach 1 W/B&W	8.07	(10.01) 7,34	(0.24) 0.24	(0.57) 0.57	(0) 0	(151) 139	59 59	210	198	<u> </u>
Oconee 1 BAW/B&W	5.04	(2.32) 2.73	(0.26) 0.31	(0.61) 0.55	(0) 0	(113) 138	59 59	172	197	160
Zion 1	4,97	(3.13) 0.99	(0.35) 0.31	(0.59) 0.61	(0) 0	(166) 108	59 59	225	167	
Indian Point 2 W/CE	4.40	No Circum Data 2.2	0.34	1.2	-56	211 (4)	34 (4)		189	
Arkansas ANO-1 B&W/B&W	4.42	(2.70) 1.99	(0.31) 0.31	(0.59) 0.59	(0) 0	(140) 129	59 59	199	188	
Point Beach 2 W/B&W, CE	7.54	(9.35) No Axial Welds	(0.25)	(0.59)	(0)	(156)	59	215		
Ginna W/B&W	8.18	(9.49) No Axial Welds	(0.25)	(0.56)	(0)	(154)	59	213		
San Onofre W/CE	9.04	(33,45) 27,12	(0.27) 0.27	(0.20) 0.20	(-56) -56	(188) 178	59 59	191	181	203
Zion 2 B&W/B&W	4,43	(2.83) 0.90	(0.26) 0.35	(0.61) 0.59	(0) 0	(119) 118	59 59	178	177	
Palisades CE/CE	4.12	(4.78) 4.78	(0.25) 0.25	(1.2) 1.2	(-56) -56	(174) 174	59 59	177	177	
Crystal River 3 B&W/B&W	2.49	(1.44) 1.36	(0.35) 0.31	(0.59) 0.61	(0) 0	(134) 118	59 59	193	177	

Plant NSSS (Verce)	EFPY	Fluence	Copper	Nickel	Mean	Mean	2102+02	RT NDT, "F.	as	Licensee's
Fabricators	12/31/81	×10 ¹⁸	ć	*	RT _{NDT}	NDT	(5)	of Dec 31, Circum.	, 1981(6) Axial	RT _{NDT} °F
Surry 1 W/B&W	4.88	(7.61) 1.65	(0.25) 0.21	(0.51) 0.59	(0) 0 ·	(141) 81	59 59	200	140	
Chok 1 _/CE	4,55	(2.87) 1.55	(0.40) 0.13	(0.82) 0.99	(-56) -56	(222) (4) 58	34(4) 59	200	61	
North Anna 1 W/RD	2.41	(4.42) No Axial Welds	(0.14)	(0.80) Forging	(+38) Governs	(76) 48	48	162	162	
Beaver Valley W/CE	1.87	(3.16) 0.47	(0.37) 0.36	(0.62) 0.62	(-56) -56	(179) 104	59 59	182	107	
North Anna 2 W/RD	0.77	(1.38) No Axial Welds	(0.13)	(0.83) Forging	(+56) Governs	(52) 52	48 48	156	156	
Salem 1 W/CE	2.26	(1.49)	(0.24) 0.24	(0.51) 0.51	(+51) Plate Governs	(87) 87	48 48	150	150	
Oconee 3 B&W/B&W	4.78	(2.92) No Axial Welds	(0.24)	(0.63)	(0)	(112)	59	(171)		
Surry 2 W/B&W, RD	4.83	(7.54) 1.64	(0.19) 0.21	(0.56) 0.59	(0) 0	(108) 81	59 59	167	140	• •
Calvert_Cliffs 1 CE/CE	4.65	(6.84) 6.84	(0.30) 0.21(11)	(0.18) 0.85	(-56) -56	(135) 136	59 59	138	139	205(244)(7)
St. Lucie CE/CE	3.52	(2.22) 2.22	(0.31) 0.30	(0.11) 0.64	(~56) -56	(98) 132	59 59	101	135	

Table P-1 (Continued)

Table F-1 (Continued)
-------------	------------

Plant NSSS/Vessel	EFPY as of	Fluence n/cm ²	Copper %	Nickel X	Mean Initial	Mean ART _{NDT}	2102+02	RI _{NDT} , °F, as of Dec 31, 19	Licensee 81(6) RT _{NDT} . °	rs F
Fabricators	12/31/81	×10 ¹⁸			RTNDT		(5)	Circum. Ax	181	
Calvert Cliffs 2 CE/CE	3.63	(5.34)	(0,30) 0.30	(0.18) 0.18	(-56) -56	(127) 127	59 59	130	30	
Trojan W CBI	3.00	(2.07)	(0.16)	(0.62) Plate Go	(+10) overns	(65) 65	48 48	123	23	
Davis Besse 1 B&W/B&W	1.68	(1.11) No Axial Welds	(0.24)	(0.61)	(0) •	(85)	59	144		
Haddam Neck W/CE	10.92	(14.30) 11.90	(0.22) 0.22	(0.10) 0.10	(-56) -56	(111) 106	59 59	114	109	
Kewaunee W/CE	5.87	(7.86) No Axial Welds	(0.20)	(0.77)	(-56)	(129)	59	132		
Farley 1 W/CE	2.19	(3.70) 0.83	(0.24) 0.27	(0.60) 0.60	(-56) -56	(117) 89	59 59	120	92	
Millstone 2	3.91	(2.19) No Data for Axial	(0.37) Welds	(0.06)	(-56)	(114)	59	117		
Prairie Island 2 W/SFAC	5.62	(7.53) No Axial Welds	(0.19)	(0.13)	(-56)	(81)	59	84	•	
Prairie Island 1 W/SFAC	5.90	(7.90) No Axial Welds	(0.14)	(0.17)	(-56)	(60)	59	63		

Footnotes

- (1) Arranged in descending order of the RT_{NDT} as of December 31, 1981 considering circumferential to be 30°F less severe than axial orientations.
- (2) Memorandum, M. Vagins to S. Hanauer, August 30, 1982.
- (3) "alues shown in parentheses on top line are for circumferential welds, bottom line is for axial welds. When plate governs--both lines.
- (4) Determined by Reg. Guide 1.99, Rev. 1, Upper Limit Line, $\sigma_{\rm A}$ = 0.
- (5) σ_{Δ} (17°F) and σ_{Δ} (24°F) are the standard deviations of the initial RI_{NDT} and ΔRT_{NDT} , respectively. If plate or forging governed, actual initial RT_{NDT} was available and $\sigma_{\Lambda} = 0$
- (5) The sum of the Mean Initial RT_{NDT}, the mean ΔRT_{NDT} and $2\sqrt{\sigma_{0}^{2}+\sigma_{A}^{2}}$, as of Dec. 31, 1981.
- (7) Initial RT_{NDT} assumed by licensee to be -50°F and by CE to be -20°F. Values in parentheses are by CE.
- (8) Fluence is per letter from CP&L Co., Sept. 24, 1982.
- (9) Fluence reduced from 11.16 n/cm² per letter from FPL Aug. 31, 1982, on TP 4. TP 3 tentatively assumed to be the same as TP 4.
- (10) Fluence reduced to 0.73 x peak per letter from Omaha PPD, Sept. 1, 1982.
- (11) Cooper and Nickel values reduced per letter from Baltimore G&E, Oct. 28, 1982.

Enclosure B

VALUE-IMPACT ASSESSMENT

INTRODUCTION

The revised guidelines for preparing value-impact analyses to satisfy the requirements of Executive Order 12291 are stated in a memorandum dated May 7, 1982 (SECY-82-187) to the Commissioners from William J. Dircks. This Value-Impact Assessment is based on the staff's technical evaluation in Enclosure A and follows a simplified outline based on the revised guidelines.

Proposed Regulatory Actions

- Establish an RT_{NDT} screening criterion of 270°F for axial welds and 300°F for circumferential welds.
- 2. Three years before the reactor vessel is predicted to reach the screening criterion, plant specific analyses must be started. These analyses will be used by the licensee and NRC on a plant specific basis to determine the regulatory action required to maintain risks from potential PTS events at an acceptably low level during continued plant operation. The analyses should consider: plant specific system features and possible improvement; identification of events which

contribute significantly to PTS risk; reliable estimate of the reactor vessel material properties; deterministic and probabilistic fracture mechanics evaluations; plant procedures and operator training improvements; inservice inspection for cracks at the vessel inner walls; flux reduction methods to slow further embrittlement; and detailed studies of the feasibility of in-place annealing of the pressure vessel.

3. Obtain information from licensees of a few plants for which near-term flux reduction would ensure that the screening criteria are not exceeded throughout plant life, to provide for early evaluation of the need, safety benefit, and cost of such programs.

Consequences of Action

*0-

There are no direct safety benefits for performing the plant specific analyses. The safety benefits will be derived from implementing any required corrective actions identified in these analyses. The plant specific analyses are a necessary step in this process. The operating experience record and probabilistic studies, as described in the Enclosure A, indicate that the potential safety benefits are large for initiating PTS corrective regulatory action for plants approaching the screening criterion. A Value-Impact Assessment pursuant to SECY-82-187, will be prepared after receipt of the plant specific analysis and the determination of the particular currective regulatory action necessary and expedient for a given operating facility.

- 2 -

The major costs involved in the presently proposed action $(anal_{3,2}, \infty)$ are the cost to industry to perform the indicated analyses and the cost to MRC to develop guidelines and to review the results. The costs to implement actions found to be necessary have not been estimated and will have to be determined and evaluated on a plant-specify basis. They could be large if system modifications or annealing are found to be needed.

Appendix P of Enclosure A provides RT_{NDT} data for forty pressurized water reactors. Based on our preliminary evaluation of the data using the proposed screening criteria, four reactors are estimated to reach the screening criterion within ten years. Nineteen plants are estimated to reach the screening criterion before end-of-life.

An estimate of the cost for performing the plant specific analyses was made for the staff by engineering consultants at Battelle Pacific Northwest Laboratories. The items considered in the cost estimates are identific! below. The items were reviewed and collectively discussed with reactor specialists in each of the engineering areas. A judgment of cost was then made based upon an estimate of the time and facility requirement.

- 3 -

(1)	Identification and quantification of PTS events	\$ 500K
(2)	Better identification of vessel material properties	70K
(3)	Deterministic and probabilistic fracture mechanics	
	analyses	100K
(4)	Flux reduction program analyses	100K
(5)	Inservice inspections and nondestrue ive evaluation	
	study	75K
(6)	Plant modification study	200K
(7)	Operating procedures and training study	150K
(8)	In-situ annealing study	<u>50K</u>
	TOTAL	\$1250K

RES experience with an ORNL/INEL study of PTS events similar to what will be needed in item 1 of the PNL estimate indicates that contractor costs for that item are about \$800K plus significant support from the utility.

Incorporating uncertainties and contingency, the staff estimates a cost of approximately two million dollars per plant to perform the required plantspecific analyses. For the four PWRs which are expected to reach the screening criterion within ten years, this implies an industry cost of eight million dollars. The NRC guideline development and review evaluation costs are estimated to be about two million dollars. These costs are modest and appear well justified since the consequence of not performing the recommended analyses is that certain reactor vessels would become increasingly embrittled and the risk of pressure vessel failure would continue to increase.

ENCLOSURE C

UNITED STATES NUCLEAR REGULATORY COMMISSION ADVISORY COMMITTEE ON REACTOR SAFEGUARDS WASHINGTON, D. C. 20555

June 7, 1982

Honorable Nunzio J. Palladino Chairman U. S. Nuclear Regulatory Commission Washington, D. C. 20555

Dear Dr. Palladino:

Subject: ACRS REPORT ON PRESSURIZED THERMAL SHOCK

During its 266th meeting, June 3-5, 1982, the Advisory Committee on Reactor Safeguards completed its review of the current status of the pressurized thermal shock problem (PTS). The NRC Staff is developing a regulation based on a combination of deterministic and probabilistic analyses to establish regulatory requirements concerning pressurized thermal shock. The ACRS has not been provided sufficient information to evaluate the adequacy of this approach.

The ACRS does not believe there is a need for any immediate plant modifications to permit continued operation of the plants which have been identified up to now as having potential PTS problems.

The most beneficial actions for these plants in the short term would be to:

- Make certain that the metallurgical properties of the vessel beltlines are established adequately with respect to fracture toughness.
- 2. Determine which is the most effective in-service inspection capability for the beltline that current technology can provide. For those welds of principal concern, inspection should be accomplished, if practical, at the next refueling shutdown using such techniques, if such inspection has not previously been accomplished.
- Provide effective operator training to avoid pressurized thermal shock and provide capability to diagnose events that could cause it.
- 4. Examine the depressurization capability for these plants and train operators when and how to use it.
- 5. Provide a demonstration of pressure vessel annealing to recover fracture toughness.

There are many intricacies associated with evaluation of pressurized thermal shock consequences that deserve attention, but the above actions would be the most effective contributors to assuring that pressurized thermal shock does not create public safety problems. Honorable Nunzio J. Palladino - 2 - June 7, 1982

The ACRS plans to continue its review of pressurized thermal shock and the related NRC Staff program. The Committee will report further at an appropriate time.

Sincerely,

mon

P. Shewmon Chairman

-04

(1:12





UNITED STATES NUCLEAR REGULATORY COMMISSION ADVISORY COMMITTEE ON REACTOR SAFEGUARDS WASHINGTON, D. C. 20555

October 14, 1982

Honorable Nunzio J. Palladino Chairman U.S. Nuclear Regulatory Commission Washington, D.C. 20555

Dear Dr. Palladino:

SUBJECT: ACRS REPORT ON DRAFT NRC STAFF EVALUATION OF PRESSURIZED THERMAL SHOCK DATED SEPTEMBER 13, 1982

During its 270th meeting, October 7-8, 1982, the Advisory Committee on Reactor Safeguards reviewed the Draft NRC Staff Evaluation of Pressurized Thermal Shock dated September 13, 1982. This matter was also considered at a Subcommittee meeting on September 30, 1982 in Washington, D.C. In its review the Committee had the benefit of discussions with representatives of the NRC Staff, the Westinghouse, Combustion Engineering, and Babcock and Wilcox Owners Groups, and the Southwest Research Institute. The Committee reported previously to you regarding this matter on June 7, 1982.

The NRC Staff is developing a regulation based on a combination of deterministic and probabilistic analyses to establish requirements concerning pressurized thermal shock (PTS). The NRC Staff proposes to use RT_{NDT} screening criteria for reactor vessels as the basis for further action concerning PTS. A value of 270°F for longitudinal welds and base material and a value of 300°F for circumferential welds have been selected. These proposed criteria are reasonable on the basis of current knowledge and provide adequate time for licensees to demonstrate plant-specific capability or planned actions in order to avoid unacceptable public safety consequences from PTS. For reactor vessels that are expected to be the earliest to exceed the screening criteria, we wish to be kept informed about PTS control actions under consideration.

The NRC Staff report indicates that several years are available before the first plant will exceed the screening criteria limits. This provides adequate time to conduct an orderly, comprehensive research program concerning measures needed to protect against pressurized thermal shock if a diligent effort to implement the program is applied. The NRC Staff has not yet defined a suitable program. We believe that the following should be elements of such a program: improved nondestructive examination capability; a more complete study of in situ reactor vessel annealing; improved fracture mechanics analysis methods that will account for realistic crack geometry, Honorable Nunzio J. Palladino

cladding effects, and crack arrest phenomena; use of three-dimensional and elastic-plastic techniques where appropriate; and potential improvements in diagnostic capability to help the operator recognize and control thermal shock events.

In accord with your desire to obtain our views on short-term steps regarding the NRC Staff's program on PTS, the recommendations in our letter of June 7, 1982 still apply. The following items deserve special attention:

- . The reactor vessels with the greatest potential increase in RT_{NDT} are those having relatively high copper content. Only a small fraction of the available irradiation test data can be fully correlated with composition effects at high fluence levels. The correlations relating the rise in RT_{NDT} to metallurgical composition would benefit from a careful assessment of the uncertainties and special circumstances related to each data point used in the correlation.
- . Improvements in PTS-related operator training and procedures should be carried out by all licensees with special emphasis on those plants indicated to have high RT_{NDT} vessels. Operational problems that need to be dealt with include the conflicting need to maintain adequate pressure for core cooling purposes while avoiding PTS, the control of feedwater and auxiliary feedwater to provide adequate heat removal while avoiding overcooling, and the recovery after a transient event which has caused violation of the cooldown rate limits.
- . The range of initiating events and subsequent operator actions which are most likely to cause PTS need to be better characterized. A more extensive evaluation of operational events, including operator errors of commission as they apply to specific plant designs, will improve our understanding.
- The ACRS has considered the value of heating the ECCS water as a means of reducing public safety risk from PTS. Heating ECCS water may be helpful in the specific set of small break LOCAs that result in primary loop flow stagnation. If this case is an important contributor to the overall thermal shock risk and if heating the water does not unacceptably diminish-containment integrity margins, then it can be a useful measure.
- Fast neutron fluence reduction is being implemented in some plants by using revised fuel management techniques. Further fluence reduction can be achieved by changing the core design. The value of such measures and the penalties involved must be determined for each plant individually.

Honorable Nunzio J. Palladino

- 3 -

The Committee believes that if due consideration is given to the items mentioned above, and if regulatory actions are based on the proposed screening criteria, the pressurized thermal shock matter should not present an undue risk to the health and safety of the public.

Additional comments by ACRS Member David Okrent are presented below.

Sincerely,

P. Shewmon Chairman

Additional comments by ACRS Member David Okrent

I generally support the ACRS recommendations in this report and have no problem with use of the proposed screening criteria on an interim basis. The comments which follow are made in no small part because of the generic implications to the regulatory process of how an issue like pressurized thermal shock is handled.

I believe it has been useful for the NRC Staff to attempt to examine the PTS issue probabilistically. The preliminary probabilistic studies reported thus far should be made more comprehensive, reported in depth, and subjected to extensive independent review.

In Section 8.4 of the September 13, 1982 Draft NRC Staff Evaluation of PTS, the NRC Staff compares its proposed PTS requirements to the Commission's, proposed policy statement on "Safety Goals for Nuclear Power Plants: A Discussion Paper," NUREG-0880.

If, for purposes of discussion, I accept the NRC Staff PRA results in Figure 8-3, as well as its statement that at the 270° F screening criterion, RT_{NDT} is likely to have a mean value of 210° F (RT_{NDT} of 270° F representing a 2 σ upper confidence bound), I have trouble agreeing with some significant statements made by the Staff in Section 8.4.

On page 8-5, the NRC Staff says the following:

"The core melt Safety Goal guideline states, 'The likelihood of a nuclear reactor accident that results in a large-scale core melt should normally be less than one in 10,000 per year of reactor operation.' This suggests that the core melt frequency ascribable to one sequence, for example PTS, should not exceed approximately 10⁻⁵ per reactor-year.

Honorable Nunzio J. Palladino

0 - 4 -

"Because of the unusually large uncertainty in the risk estimation for PTS, compared to other sequences, a value of less than 10^{-9} might well be assigned for a safety goal for PTS. We have not done this in the discussion in this section, but have used 10^{-9} . The reader should keep in mind that the risk numbers for PTS given in the following discussion are highly uncertain.

"We have no technical analysis of the course and consequences of a PTS sequence that involves RPV failure. Determination of the RPV failure mode (better, estimation of the probabilities of the various failure modes) has not been done and is dependent on the details of the scenario. Moreover, the outcome would likely be dependent also on the plant design details. In particular, ice condenser containments would be expected to have different failure modes, with different probabilities, than large dry containments."

I disagree with the use of 10^{-5} per reactor-year for at least two reasons. First, there are many more than ten potential initiators of large-scale core melt. Allocating 10^{-5} per reactor-year to a single initiator is questionable. It is even more questionable in view of the large uncertainty. Most importantly, until one knows with considerable confidence that a PTS failure has only a small likelihood of leading to early containment failure or otherwise leading to one of the large radioactivity release categories, the assignment of 10^{-5} per reactor-year (best estimate, with very large uncertainties) is probably unacceptable.

The NRC Staff states it has no technical analysis of the course and consequences of a PTS sequence that involves reactor pressure vessel failure. However, on page 8-6 the NRC Staff defines a quantity Y as the fraction of PTS-caused failure leading to core melt which leads to significant radioactive release. In the September 13, 1982 draft report, the NRC Staff used values of Y between 5×10^{-2} and 5×10^{-3} in suggesting that its proposed screening criterion is compatible with a PTS risk less than 10% of the proposed safety goal guidelines. In its oral presentation during the 270th ACRS meeting, the NRC Staff modified its approach such that a value of Y = 8×10^{-2} is compatible with meeting he safety goal.

The NRC Staff provides no basis for these values of Y, which are much less than unity. In the absence of any reasonable justification for the suggested range for Y, this aspect of the log'c supporting the choice of the screening criterion becomes weak.

I recommend that, before the proposed screening criteria are adopted for the long term, the relevant information on containment behavior, given a PTS failure, be developed and included in an expanded probabilistic study which attempts to deal quantitatively with all contributions to uncertainty and
Honorable Nunzio J. Palladino

states which uncertainties cannot be addressed meaningfully and why. The sensitivity studies reported in Appendix H are useful but do not take the place of a critical examination and evaluation of uncertainties, which the NRC Staff currently estimates are a factor of 100 in either direction.

The NRC Staff should then state a recommended position and include reasons for whatever approach is recommended in light of the uncertainties.

I might note the incongruity illustrated by the NRC Staff's comment on page 8-7 that "For scenarios involving core melt without significant releases, the core melt guideline will govern and ALARA is not a consideration." This conclusion by the NRC Staff may be compatible with NUREG-0880. However, I find it hard to believe that the Commission would not credit an appropriate benefit to some measure which significantly reduced the likelihood of pressure vessel failure, even if such failure were estimated to lead to core melt without significant release of radioactivity.

Finally, I should like to observe that PTS is a real issue in which the NRC Staff, the industry, and others are using probabilistic considerations coupled with ad hoc safety criteria as one input into engineering judgment. PTS is a significant issue which is subject to such large uncertainties that a plausible set of confidence bounds may well encompass a risk band which separates the clearly acceptable and clearly unacceptable areas and that these confidence bounds may also extend into those areas. Quite aside from any Commission action on safety goals, it seems to me important that the Commission take steps to assure that the probabilistic aspects of this issue are done as well as practical and that the appropriate review processes are established and accomplished. I also recommend that the Commission participate actively in establishing the criteria to be used on this issue for decision making under uncertainty. This includes the basis for action if and when flaws or indications of flaws in the size range of interest are found during forthcoming inspections of reactor vessels.

References:

- Draft NRC Staff Evaluation of Pressurized Thermal Shock, dated September 13, 1982, including Appendices A-P, dated September 15, 1982.
- Letter from Demetrios L. Basdekas, NRC to P. G. Shewmon, ACRS concerning comments on the September 13, 1982 Draft NRC Staff Evaluation of Pressurized Thermal Shock, dated October 6, 1982.

NOV 1 2 1982

MEMORANDUM FOR: P. S. Shewmon, Chairman Advisory Committee on Reactor Safeguards

FROM: William J. Dircks Executive Director for Operations

SUBJECT: ACRS COMMENTS ON DRAFT NRC EVALUATION OF PRESSURIZED THERMAL SHOCK

Your letter of October 14, 1982, to Chairman Palladino provided the comments of the ACRS regarding the draft NRC Staff Evaluation of Pressurized Thermal Shock, dated September 13, 1982. The Committee indicated that the RT_{NDT} screening criteria proposed in the draft staff evaluation "are reasonable on the basis of current knowledge and provide adequate time for licensees to demonstrate plant-specific capability or planned actions in order to avoid unacceptable public safety consequences from PTS." As you know, the draft staff evaluation is currently under management review prior to the submittal of recommendations to the Commission. The ACRS comments will be addressed by the staff in that Commission paper.

You requested that the Committee be kept informed about PTS control actions under consideration for reactor vessels that are expected to be the earliest to exceed the screening criteria. At the ACRS meeting on November 5, 1982, the staff informed the Committee that further consideration is being given to the need for actions to assure the early implementation of flux reduction programs for those plants that are currently projected to exceed the screening criteria before the end of design life. The staff will continue to keep the ACRS informed of the progress of these deliberations.

The Committee indicated that there is adequate time to conduct an orderly, comprehensive research program concerning measures needed to protect against pressurized thermal shock and recommended some elements of such a program. The staff agrees and is developing a better defined program. We will arrange a subcommittee briefing on the research program early next year.

The Committee recommended that the staff's program on FTS give special attention to improvements in PTS-related operator training and procedures; better characterization of initiating events and subsequent operator actions; the value of heating the ECCS water; and plant-specific evaluations of the value and costs of fast neutron fluence reduction programs. The staff agrees and consideration of these items is part of the planned program.

The Committee also recommended an additional careful assessment of uncertainties in the irradiation test data used to develop correlations for

FFICE			 	,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,	****	
NAME 🌶		••••••	 · · · · · · · · · · · · · · · · · · ·			
DATE 🌶	•••••	• • • • • • • • • • • • • • • • • • • •	 •••••••••••••••••••••••••••••••••••••••			

P. S. Shewmon

prediction of RT_{NDT} increases with fluence for various materials compositions. The staff will include additional studies of RT_{NDT} shift correlations in the research program under development.

.

The staff is also considering the additional comments of ACRS member David Okrent in its preparation of the Commission paper on PTS.

(Signed) William J. Dircks

William J. Dircks Executive Director for Operations

- **10-10** --

.

	Note: See p Revi	revious concu isid to refl	cit 11/5/82	PPAS#			
AcRs meeting.			10/28/82				
Officel	GR 18	DST	RES	NRR	NRR	EDO	
JRNAME	FSchroeder7j	nSHanauer*	GArlotto*	ECase*	HDenton*	WDircks	
DATE	10/27/82	10/27/82	10/26/82	10/28 /82	10/29/82		****



• UNITED STATES • ENCLOSURE F NUCLEAR REGULATORY COMMISSION WASHINGTON, D. C. 20555

OCT 2 0 1932

MEMORANDUM FOR: William J. Dircks Executive Director for Operations

FROM: Victor Stello, Jr., Chairman Committee to Review Generic Requirements

SUBJECT: MINUTES OF CRGR MEETING NUMBER 21

The Committee to Review Generic Requirements met on Wednesday, October 6, 1982, from 1~6 p.m. A list of attendees is enclosed.

S. Hanauer (NRR) presented the technical background and the recommendations proposed by NRR to address the issue of Pressurized Thermal Shock (PTS). These recommendations were included in a draft report that is intended by NRR to form the basis for a Commission paper. The Committee noted that the staff did a thorough job in examining the various technical aspects of this complicated issue. The draft report reflects a good balance between deterministic engineering analyses and probabilistic risk assessment (PRA) techniques. While acknowledging the large uncertainties involved (the staff estimated as much as two orders of "magnitude uncertainty in the vessel failure probability estimate) the Committee believes the PRA analysis is a valuable supplement to the deterministic analysis in arriving at a balanced engineering judgment on this issue.

NRR proposed screening criteria for the vessel reference temperature RT_{NDT}, a parameter that charcilerizes the state of embrittlement of reactor vessels. The Committee agreed that the proposed screening criteria (270°F for longitudinal welds and 300°F for circumferential welds) seem appropriate. NRR proposes that, whenever the value of RT_{NDT} for a given vessel is projected to exceed either of the screening criteria within the next 3 calendar years, the licensee would be required to submit a plant specific analysis, the scope of which has yet to be specified. NRR also proposes that a number of long term actions be required to ameliorate the PTS problem.

Demetrios Basdekas attended the CRGR meeting and summarized the comments in his memorandum of September 24, 1982 to Carl Johnson in RES. In his memo, Mr. Basdekas noted the short time available for him to offer comments on the NRR draft staff report and the fact that he "had not participated in PTS related activities...for quite some time..." Mr. Basdekas agreed with the thrust of CRGR discussions on the need for prompt decisions on plant modifications, such as low leakage fuel loadings. Mr. Basdekas expressed dissatisfaction with the screening criterion recommended by NRR but was unable to make specific recommendations for its alteration. Mr. Basdekas reiterated what he called his long standing concern that there is insufficient information available to him or to the NRC generally to properly address the safety implications of reactor

control systems (unresolved safety issue A-47 which Mr. Basdekas has responsibility to support through his research tasks). He urged more attention to obtaining better control system data from representative attention. It is his belief that the findings of A-47 may eventually licensees. It is his belief that the findings on the PTS issue. He did have an important influence on the decisions on the PTS issue. He did not offer any specific examples of this influence nor did he hold out any hope that the A-47 input would be available in any reasonable time frame to support the short term PTS decisions. He also did not have any specific criticism concerning the proposition that the draft staff report takes into account the control system contributions to PTS event sequences that have occurred or were covered by the PRA.

- 2 -

The Committee finds Mr. Basdekas' arguments that PTS decisions should depend on the resolution of USI A-47 to be generally lacking in substance; i.e., we see no reason that A-47 cannot follow A-49. Furthermore, it would appear that requiring resolution of A-47 before deciding on PTS would be contrary to the desire for reaching an early resolution of PTS.

Mr. Basdekas brought to the Committee no new technical information or unique insight on PTS not otherwise available to the staff and already utilized in the development of the draft report on PTS. This is not to fault Mr. Basdekas - he was asked to comment on the staff report in a very short time period and he is not normally assigned to work in that area. The Committee feels it is counterproductive to efficient and effective staff work for NRC management to seek his reactions to staff proposals on PTS in this manner.

Mr. Sanford Israel also cautioned the CRGR that there could be more severe overcooling transients than considered in the PRA analysis. The Committee agreed with his observation and suggested that NRR continue to evaluate the probabilities and consequences of the full range of potential overcooling transients in their ongoing PTS work.

Based on the briefing by NRR and review of the extensive background material, the following recommendations are made:

- 1. The draft report should be modified to make clear that a rule change will be required to finally resolve the PTS issue.
- 2. It was noted that, because the pressure vessel embrittlement increases with irradiation exposure, the risk from PTS increases with time. In the absence of remedial actions, some PWR vessels are estimated to have RT, pT well in excess of 300°F at the end of their service life. Since this indicates that some remedial action will be required for those vessels, the Committee requested that NRR develop further information on the costs and benefits of requiring near term flux reduction measures such as replacing outer row fuel assemblies with dummy assemblies. The CRGR stressed that this action, if implemented, was needed not because the PTS risk is unacceptably high at this time, but because the passage of time forecloses the flux reduction option as an effective remedy.

William J. Dircks

- 3. The Committee agreed that improved operator training and emergency operating procedures are needed. However, it was emphasized that these improvements must be done in an integrated manner and must not deemphasize the importance of maintaining adequate core cooling in the event of a loss-of-coolant accident by an over emphasis of the PTS issue.
- 4. The Committee suggested that NRR add a short term task to investigate whether the frequency of overcooling transients for B&W plants may be higher than the average, based on operating experience to date.

The CRGR will continue its review of the PTS issue on October 28, 1982, at which time NRR will present information on the costs and benefits of requiring near term flux reduction measures.

Vactor Stello, Jr., Chairman Committee to Review Generic Requirements

Enclosure: List of Attendees

CRGR Members Commission (5) Office Directors Regional Administrators G. Cunningham S. Hanauer CRGR MEETING #21 LIST OF ATTENDEES October 6, 1982

1 :

1.

CRGR MEMBERS

Vic Stello R. Bernero Roger Mattson (for D. Eisenhut) Ed Jordan Joe Scinto

OTHERS

Tom Murley Walt Schwink Steve Hanauer Roy Noods Mat Taylor L. F. Litton Karl Kniel Carl Johnson Frank Schroeder Harry Boulden Richard Donovan James Popelarski Demetrois Basdekas E. D. Throm Alan Rubin Bill Shields Jack Strosnider Larry Shao Jesse Funches Steve Stern Sanford Israel Norm Lauber Reil Randail Ray Klecker Hugh Thompson Lambros Lois Jim Clifford Ed Abbutt Tom Dorian John Austin Milt Vagins



. . - •

:

UNITED STATES NUCLEAR REGULATORY COMMISSION WASHINGTON, D. C. 20555

October 26, 1982

MEMORANDUM FOR: Victor Stello, Jr.

FROM: Joseph F. Scinto

SUBJECT: COMMENTS ON MINUTES OF CRGR MEETING NUMBER 21

On page 1 of the Minutes, dated October 20, 1982, I would note that the way in which RTNDT for the screening cr. arion is to be computed (by including a 2 sigma value to the mean values, p.5-5 of the draft) makes the 270°F screening criterion more protective than figure 8.3 in the draft report might suggest. That figure provides longitudinal crack extension frequencies for mean surface RTNDT, rather than for RTNDT computed as mean plus 2 sigma.

soh F. Scintà

cc: R. Bernero

- Ed Jordan
- R. Cunningham
- C. Heltemes
- D. Eisenhut

CRGR MEETING #23 LIST OF ATTENDEES October 28, 1982

.

CRGR MEMBERS

Vic Stello Ed Jordan Joe Scinto Jack Heltemes Bob Bernero Bob Purple (For Darrell Eisenhut)

OTHERS

s'

Tom Murley Steve Hanauer Walt Schulnk Lambros Lois Guy Vissing Norm Lauben Les Rubenstein Ed Case Steve Stern Bill Shields Ed Throm Ray Klecker Neil Randall Rainer Rantala Jack Strosnider Frank Schroeder Roy Woods Jim Milhoan Tom Cox Mat Taylor Ed Abbott Felix Litton



UNITED STATES * NUCLEAR REGULATORY COMMISSION WASHINGTON, D. C. 20555

NOV 1 2 1982

MEMORANDUM FOR: William J. Dircks Executive Director for Operations

FROM: Victor Stello, Jr., Chairman Committee to Review Generic Requirements

SUBJECT: MINUTES OF CRGR MEETING NUMBER 23

The Committee to Review Generic Requirements met on Thursday, October 28, 1982, from 1-4 p.m. A list of attendees is enclosed.

S. Hanauer (NRR) presented further technical information on the issue of Pressurized Thermal Shock (PTS) in response to questions raised at CRGR Meeting No. 21.

Material was presented which disaggregated the overall PWR operating experience according to reactor manufacturer. This analysis of operating experience suggested that the frequency of overcooling transients for B&W plants may be higher than that for Westinghouse and Combustion Engineering plants. However, because the sparcity of data leads to large uncertainty bands, the Committee felt that there was not a sound basis for establishing a different value of the RINDT screening criterion for B&W plants. The Committee agreed with NRR that more detailed analysis of B&W plants will be required.

NRR presented further analyses of small break loss-of-coolant accidents (SBLOCA), which the staff's PRA results had shown to be the dominant risk sequence for Westinghouse plants. If, for example, the small break LOCA were to occur in a location where the break were isolatable by operator action to close a valve, then the threat to the vessel would be greater due to (a) repressurization to full system pressure and (b) no credit could be taken for the ameliorating effect of warm prestress. On the other hand, recent data from ECC mixing tests show that there is a better mixing of cold ECC water with hot water in the cold leg pipes and vessel downcomer and, as a result, the vessel would not cool down as fast during a SBLOCA than previous analyses had indicated. The net result of all these factors is that the conditional vessel failure probability for an isolatable SBLOCA would be increased by a factor of 10 over earlier estimates for an nonisolatable SBLOCA. NRR did not have an estimate for the relative probabilities of isolatable vs. nonisolatable SBLOCAs, but in right of the large uncertainties in the overall risk analyses (the staff estimated at least a factor of 100 uncertainty) the Committee did not believe this new information would significantly alter the engineering judgments on this issue. The Committee recommended that NRR continue to evaluate probabilities and consequences of the full range of potential overcooling transients for each reactor manufacturer in their ongoing PTS work.

At Meeting No. 21, the CRGR had noted that, because the pressure vessel embrittlement increases with irradiation exposure, the risk from PTS increases with time. In the absence of remedial actions, some PWR vessels were estimated to have RT_{NDT} well in excess of 300°F at the end of their service life. Since this indicates that some remedial action will be required for those vessels, and since the passage of time reduces the effectiveness of flux reduction as a remedy, the Committee had asked NRR to develop further information on the costs and benefits of near term flux reduction measures.

NRR presented information which showed that there are some plants which will need no remedial actions for their vessel RT_{NDT} to remain below the screening criteria throughout their service life, based on current information. The remaining plants can generally be grouped into three categories.

- (1) There are several plants for which near term action to reduce the flux at critical welds by a factor of 2 or less will ensure that they do not exceed the screening criteria throughout their service life. Information available to NRC from reactor manufacturers indicates that a flux reduction factor of 2 can be attained through installation of a low leakage core, which is simply the installation of partially burned fuel assemblies in the periphery of the core in place of fresh fuel assemblies. This fuel management option is already being implemented by some licensees at reportedly little or no additional cost.
- (2) There is a group of about nine plants for which near term action to reduce the flux at critical welds by factors of 2 to 4.5 will ensure that they do not exceed the screening criteria throughout their service life. NRR presented analyses which showed that these flux reduction factors can be attained through the installation of a low leakage core and the replacement of some peripheral fuel assemblies (estimated 4-12) by dummy assemblies. There would be an estimated engineering cost of \$20 million per plant, but no substantial increase in fuel cycle costs or operating costs.
- (3) There is one plant, H. B. Robinson 2, which is close to reaching one of the screening criteria and for which the fuel management options described above could not reduce the flux at the critical welds enough to prevent reaching the screening criteria. It was suggested at the meeting that there were no practical fuel management schemes or operating modes that would prevent Robinson 2 from reaching the screening criteria within the next few years, and this was confirmed by NRR. It is possible, however, to reduce the flux at the critical weld by a factor of about 9 by replacing the entire outer row fuel assemblies with dummy assemblies. This option would entail an engineering cost of \$20 million and probably additional operating costs due to the need to derate the power level of the plant.

William J. Dircks

÷

The Committee discussed the technical options open to NRC to resolve the pressurized thermal shock issue.

- (a) NRC could continue to refine the analyses, including plant specific risk analyses, in the expectation that they would show that the plants could meet their service life without remedial actions. The Committee felt that this option would be costly and not likely to produce convincing arguments for no remedial actions.
- (b) NRC could require near term actions to reduce the flux levels for those plants where such actions would ensure that the vessel RT_{NDT} would remain below the screening criteria throughout the service life.
- (c) NRC could establish a regulatory limit on vessel embrittlement and permit plant operation until that limit is reached, at which time the vessel would have to be thermally annealed or the plant shut down. The Committee noted that vessel annealing appears to be technically feasible, although unproven on a large scale, and costly in terms of engineering, plant down time and occupational exposure.

The initial NRR proposal would establish screening criteria for the vessel reference temperature, RT_{NDT}, for critical welds. Whenever the value of RT_{NDT} for a given vessel would be projected to exceed the screening criteria within the following 3 calendar years, the licensee would be required to submit a plant specific analysis. The staff would develop acceptance criteria for determining whether plant modifications would be required after the staff review of the plant specific analyses. The initial NRR proposal also included a number of long term actions intended to ameliorate the PTS problem.

Discussions with the Committee made clear that for those plants that the staff estimates are currently near the screening criteria, the two-step process above would result in delays which could foreclose options that currently appear to be feasible at little additional cost, particularly flux reduction options. Such delays could mean that, at the point of decision, there may be few if any alternatives to annealing the vessel.

For those reasons, NRR proposed and the Committee agreed that the staff should take steps to initiate flux reductions for those plants in categories (1) and (2) above to ensure that RT_{NDT} for critical welds does not reach the screening criteria before the end of service life. The Committee believes that the PTS risk is not unacceptably high at this time, but by taking these relatively low cost actions in the near term, the PTS risk can be maintained at acceptable levels for these plants and the need for requiring costly and unproven actions in the future would thereby be obvizted for these plants. NRR staff indicated agreement with this course of action at the meeting.

William J. Dircks

Although various methods for initiating flux reduction requirements were discussed, including rulemaking, generic letters and orders, the Committee concluded that selection of the appropriate procedural method should be left to the staff.

For the case of H. B. Robinson 2, the Committee recommends that the staff promptly have the licensee submit a plan showing what actions they intend to take to resolve the PTS issue for their plant. It is expected that this plan could include consideration of heating ECC water, safety systems to prevent repressurization, flux reduction methods, annealing, or some combination thereof, but the Committee judges that some remedial actions will be needed in the next few years to ensure that the PTS risk for Robinson 2 remains within acceptable levels throughout its service life.

The CRGR reemphasized that staff actions to improve operator training and emergency operating procedures must be done in an integrated manner and must not deemphasize the importance of maintaining adequate core cooling in the event of a loss-of-coolant accident by an over emphasis of the PTS issue in the training and emergency procedures.

In summary, the Committee recommends that NRR develop a program to inform licensees of the need to modify plant operations through flux reduction programs to ensure that RT_{NDT} values do not reach unacceptable levers: For plants in categories (1) and (2) above, flux reduction appears adequate. In one plant, H. B. Robinson 2, a more comprehensive plan is needed.

Victor Stello, Jr., Chairman Committee to Review Generic Requirements

Enclosure: List of Attendees

cc: Commission (5) Office Directors Regional Administrators CRGR Members G. Cunningham

Decy-82-465

ENCLOSURE A

NRC STAFF EVALUATION

0F

PRESSURIZED THERMAL SHOCK

November 1982

.

CONTENTS

• _

			Page
1.	INTR	ODUCTION	1
	1.1 1.2	Background Staff Reviews of PTS Information Provided by	1
	1.3 1.4	Licensees and Industry Procedure and Training Reviews Proposed Approach for Future Evaluation	ි. ව ද
	1.5	Structure of this Report	ŕ
2.	FREQ	UENCY AND CHARACTERIZATION OF EXPERIENCED OVERCOOLING EVENTS	10
	2.0 2.1 2.2	Preface Introduction Event Descriptions	10- 11 13
		2.2.1 H. B. Robinson Steam Line Break (04/28/70) 2.2.2 H. B. Robinson Stuck Steam Generator Relief Valve	: `,
		(11/05/72) 2.2.3 H. B. Robinson RCP Seal SBLOCA (05/01/75) 2.2.4 Rancho Seco NNI/ICS (03/20/78)	[4] 4
		(Excess Feedwater Transient) 2.2.5 Three Mile Island 2 (03/28/79) 2.2.6 R. E. Ginna SGTR and PORV (01/25/82) 2.2.7 Crystal River 3 NNI/ICS (02/26/80)	26
		(Small-Break LOCA Transient) 2.2.8 Prairie Island SGTR (10/02/74)	21 32
	2.3 2.4	Summary of Operating Experience Comparison with Westinghouse Characterization of Operating Experience	8 J
3.	DETE	RMINISTIC FRACTURE MECHANICS ANALYSES	
	3.1 3.2 3.3	Fracture Mechanics Discussion Description of FM Computer Analysis Programs Determination of K _{Ic} and K _{Ia}	
	3.4	Generic Deterministic Studies of Crack Initiation	(† 1
		 3.4.1 Stylized Transients 3.4.2 Crack Initiation for Stylized Transients 3.4.3 Sensitivity Studies 3.4.4 Crack Arrest 	目的
	3.5	Prediction of Critical RT _{NDT} (RT _c) for Actual Events	45
4.	SELE	CTION OF SCREENING CRITERION	

i

CONTENTS (Continued)

			Page
5.	DET	ERMINATION OF RTNDT	i. i
	5.1	Introduction	
	5.2	Estimation of Initial RT	
	5.3	Estimates of the Shift in RT_{NDT} Due to Radiation (ΔRT_{NDT})	• •
	5.4	Recommended Method for Calculation of RT _{NDT}	
6.	CONS	DIDERATION OF LOWER PROBABILITY EVENTS	L- 5
	6.1 6.2	Identification of Event Sequences with Significant PTS Risk Characterization of Specific Groups of Event Sequences Identified as Contributors to PTS Risk	له م. د-د
		 6.2.1 Secondary (Steam Side) Depressurization 6.2.2 Small-Break Loss-of-Coolant Accident (SBLOCA) 6.2.3 Steam Generator Tube Rupture 	1-1 1- 1 _
	6.3	Low Frequency Event Sequences Contributing to PTS Risk	i
7.	PROB	ABILISTIC TREATMENT OF FRACTURE MECHANICS	17
	7.1 7.2 7.3	Introduction Description of VISA Code Probabilistic Fracture Mechanics Sensitivity Studies	28 19 51
8.	PROB	ABILITY OF VESSEL FAILURE	: (
	8.1 8.2 8.3 8.4 8.5 8.6 8.7	Introduction Methodology Accuracy and Completeness Results Relationship to Safety Goal Relationship to Licensing Criteria Conservatisms and Non-Conservatisms	
9.	PLAN BEFOR	T-SPECIFIC ANALYSES AND EVALUATIONS TO BE PROVIDED RE THE SCREENING CRITERION IS EXCEEDED	
	9.1 9.2 9.3	Introduction Evaluation of Overcooling Event Sequences Vessel Materials Properties	
		9.3.1 Improve Basis for Initial RT _{NDT}	, L ²
		9.3.2 Refinement of Chemistry Information	* ,
		9.3.3 Vessel Fluence	19 - S
			1.

CONTENTS (Continued)

.

?

•

١

		Page
	9.4 Deterministic Fracture Mechanics Evaluation	10 P 10 P
	9.6 Inservice Inspection and Nondestructive Evaluation Program 9.7 Plant Modifications	14 11 -
	9.8 Operating Procedures and Training Program Improvements 9.9 In-Situ Annealing	1• 1 : : : :
	9.10 Basis for Continued Operation	11.
10.	CONCLUSIONS AND RECOMMENDATIONS	i •
	10.1 Conclusions	モン
	10. Comment on Overall Approach	1:1

a

LIST OF APPENDICES

£.

4

1 _{₽0} endix A	ł	Overview of Activities Concerning the Pressurized Thermal Shock Issue
Appendix B	3	Chronology of Events Concerning Pressurized Thermal Shock
Appendix C	2	Procedures and Training
Appendix D)	Reactor Vessel Fracture Analysis
Appendix E		Determination of $\mathrm{RT}_{\mathrm{NDT}}$ for Plants for Comparison with Screening Criteria
Appendix F		Pressure Vessel Fast Neutron Fluence Uncertainty
Appendix G	ì	Overcooling Event Sequences
Appendix H	I	Reactor Pressure Vessel Failure Probability Study
Appendix I	•	Fluence Rate Reduction to PWR Pressure Vessels
Appendix J)	Summary of NRC Staff Position on Review of Control Systems
Appendix K	ζ.	Effects of Heating ECCS Water on Pressurized Thermal Shock
Appendix L		Nondestrictive Evaluation Methods
Appendix M	Ì	In Situ Annealing 🗸
Appendix N		Future Confirmatory Studies
Appendix O	Ì	Summary of ORNL Fracture-Mechanics Analysis for Several PWR Recorded OCA Transients
Appendix P		Calculated RT _{NDT} Values for Plants

iv

TABLE OF CONTENTS OF APPENDICES

APPENDIX A - OVERVIEW OF ACTIVITIES CONCERNING THE PRESSURIZED THERMAL SHOCK ISSUE

- A.1 Introduction
- A.2 Summary of Industry Responses to Staff Requests
- A.3 Summary of Industry Responses to Staff Requests
 - A.3.1 Responses Related to Westinghouse Plants
 - A.3.2 Responses Related to Combustion Engineering Plants
 - A.3.3 Responses Related to B&W Plants
- A.4 NRC Staff Audit of Operating Procedures, Operator Qualifications, and A.4.1 Robinson 2 Audit
 - A.4.2 Oconee 1 Audit
 - A.4.3 Fort Calhoun Audit
 - A.4.4 San Onofre 1 Audit
 - A.4.5 Maine Yankee Audit
- References

APPENDIX B - CHRONOLOGY OF EVENTS CONCERNING PRESSURIZED THERMAL SHOCK

APPENDIX C - PROCEDURES AND TRAINING

C.1 Human Factors Considerations

Westinghouse Plants C.1.1 C. 1. 1. 1 H. B. Robinson C.1.1.1.1 Present Procedures C.1.1.1.2 Present Operator Training C.1.1.1.3 Plant Audit Turkey Point 3 C.1.1.2 C.1.1.2.1 Procedures C.1.1.2.2 Present Operator Training C.1.1.2.3 Plant Audit C.1.1.3 San Onofre 1 C.1.1.3.1 Present Procedures C.1.1.3.2 Present Operator Training C.1.1.3.3 Plant Audit C. 1. 2 **Combustion Engineering Plants** C.1.2.1 Fort Calhoun C.1.2.1.1 Present Procedures C.1.2.1.2 Present Operator Training C.1.2.1.3 Plant Audit C.1.2.2 **Calvert Cliffs** C.1.2.2.1 Present Procedures C.1.2.2.2 Present Operator Training C.1.2.2.3 Plant Audit C.1.2.3 Maine Yankee C.1.2.3.1 Present Procedures C.1.2.3.2 Present Operator Training

C.1.2.3.3 Plant Audit

Table of Contents of Appendices (continued)

- C.1.3 Babcock & Wilcox Plants
 - C.1.3.1 Oconee

C.1.3.1.1 Present Procedures

C.1.3.1.2 Present Operator Training

C.1.3.1.3 Plant Audit

- C.2 Conclusions
 - C.2.1 Procedures
 - C.2.2 Training
 - C.2.2.1 Improvements in Emergency Operating Procedures

C.3 Recommendations

APPENDIX D - REACTOR VESSEL FRACTURE MECHANICS ANALYSIS

- D.1 NRC Analytical Procedures
 - D.1.1 Assumptions
 - D.1.2 Stress Algorithms
 - D.1.3 Postulated Initial Cracks
 - D.1.3 Fracture Mechanics Algorithms
- D.2 ORNL Analytical Procedures, OCA-I, OCA-II
- D.3 Warm Prestressing
- D.4 Determination and Utilization of Material Toughness
 - D.4.1 ASME Code Section XI Curves
 - D.4.2 Attentuation of Fluence and RT_{NDT} through the Vessel Wall
- D.5 Stress/Fracture Mechanics Procedures Summary
- D.6 Discussion of Results
 - D.6.1 OCA-II Parametric Study
 - D.6.2 Fracture Mechanics Analysis for Several PWR Recorded Transients
 - D.6.3 Fracture Mechanics Example Analyses

References

- APPENDIX E DETERMINATION OF RT_{NDT} FOR PLANTS FOR COMPARISON WITH SCREENING CRITERIA
- E.1 Introduction
- E.2 Initial RT_{NDT}
 - E.2.1 Code Definition
 - E.2.2 Absence of Actual Measurements of RT_{NDT}
 - E.2.3 Generic Data
 - E.2.4 Comparison with Vendor's Values

E.3 Adjustment of RT_{NDT} Due to Radiation (RT_{NDT})

- E.3.1 Trend Curves Versus Surveillance
- E.3.2 Regulatory Guide 1.99 Revision 1, Bounding Curves
- E.3.3 Guthrie Trend Curves
- E.3.4 Guthrie Upper Bound Trend Curves
- E.3.5 Comparison with MPC Curves
- E.3.6 Comparison with Vendor's Curves
- E.4 Screening Value of RT_{NDT}
 - E.4.1 Uncertainties
- E.4.2 Alternative Calculation of RT_{NDT}

References

F - PRESSURE VESSEL FAST NEUTRON FLUENCE UNCERTAINTY APP/ 1...roduction F.1 Benchmarking Uncertainty F.2 F.3 Computation Uncertainty References APPENDIX G - OVERCOOLING EVENT SEQUENCES G. 1 Critical Parameters G.1.1 Temperature G.1.2 Rate of Cooldown G.1.3 Pressure G. 1.4 Frequency G.1.5 Uncertainty in T_f , B, and P Values for Characterization G.2 Events to be Considered G. 2.1 Main Steam Line Break (MSLB) G.2.2 Small Steam Line Break (SSLB) or Stuck Open Steam Generator Safety/Relief Valve (SOSGSRV) G.2.3 Small Break Loss-of-Coolant Accident (SLOCA) G.2.4 Steam Generator Tube Rupture (SGTR) G.2.5 Excess Feedwater Other Events G. 2.6 References APPENDIX H - REACTOR PRESSURE VESSEL FAILURE PROBABILITY STUDY Reactor Pressure Vessel Description H.1 Probabilistic Model H. 2 H.2.1 Fracture Mechanics Algorithms H.2.2 Simulation Model H. 2.3 Statistical Distribution of Random Variables H.3 Results H. 3.1 Reference Case H. 3.2 Reference Case Sensitivity Studies H. 3. 2. 1 Copper Content Initial RINDT H. 3. 2. 2 H.3.2.3 Fluence H.3.2.4 Fracture Toughness H. 3. 2. 5 Simultaneous Increase in the Variability of All **Random Variables** H. 3. 2. 6 Flaw Distribution Shift in RI_{NDT} H. 3. 2. 7 Upper Shelf H. 3. 2. 8 H. 3. 2. 9 Cladding H. 3. 3 Transient Sensitivity Studies H.3.3.1 Main Steamline Break and Turbine Trip with Stuck Open Bypass Value Transients H. 3. 3. 2 Hypothesized Transients with Exponential Cooldowns and **Constant Pressures** 3.3.3.2.1 T_f-RT_{NDT} Sensitivity Study h. 3. 3. 2. 2 Cooldown Rate Sensitivity Study H.3.3.2.3 Pressure Sensitivity Study H.3.3.2.4 Heat Transfer Coefficient Sensitivity Study

v11

- 25-

4

H.3.4 Inservice Inspection Sensitivity Study Warm Prestressing Sensitivity Study H.3.5 H.3.6 Flaw Orientation Sensitivity Study H.4 Application of Probabilistic Analyses in Establishing Regulatory Criteria H.4.1 Limitations of Probabilistic Fracture Mechanics Analyses H.4.2 Application of the Probabilistic Fracture Mechanics Results References APPENDIX I - FLUENCE RATE REDUCTION TO PWR PRESSURE VESSELS I.1 Introduction 1.2 Survey of Licensees, Owners' Groups, and Vendors I.3 Survey of Foreign Experience I.3.1 Finland I.3.3 France I.3.4 Other Countries I.4 Evaluation of Fast Fleunce Rate Reduction Schemes I.B Conclusions I.S. Cost Estimate for Fuel Assembly Substitution to lower Neutron Flux to Pressure Vessel References APPENDIX J - SUMMARY OF NRC STAFF POSITION ON REVIEW OF CONTROL SYSTEMS J.1 Introduction J.2 Philosophy of Separation of Protection and Control Systems J.3 NRC Staff Reviews of Control Systems J.4 Instrumentation and Control System Impact on Pressurized Thermal Shock J.5 Actions Completed or Underway to Determine Potential Consequences of **Control System Failures** J.6 Conclusions J.6.1 Standard Review Plant Guidance for Control System Review APPENDIX K - EFFECTS OF HEATING ECCS WATER ON PRESSURIZED THERMAL SHOCK K.1 Large- and Small-Break LOCAs and Secondary Side Effects K.2 Plants Which Have Raised Their ECCS Water Temperature K.3 Mixing of ECCS Water K.4 Maximum ECCS Water Temperature References APPENDIX L - NONDESTRICTIVE EVALUATION METHODS L.1 Detectability of Underclad Cracks L.2 Influence of Improved NDE Techniques L.2.1 Fracture Implications of Improved Inservice Inspection L.3 Flaw Detectability Measurements APPENDIX M - IN SITU ANNEALING M.1 Time-Temperature Effects on Recovery of Properties M.2 Reirradiation Effects

£

References

APPENDIX N - FUTURE CONFIRMATORY STUDIES

- N.1 Introduction
- N.2 Summary of Prior Studies
- N.3 Present Programs Addressing Issues
- N.4 Applicability of Research
- N.5 Confirmatory Studies on Fluence Trend Curves
 - N.5.1 Refinement of Chemistry and Fluence Factors
 - N.5.2 Long-Range Effort
 - N.5.3 ORNL Study
 - N.5.3.1 Objective and Scope
 - N.5.3.2 Study Plan
 - N.5.4 Status and Schedule
- APPENDIX O SUMMARY OF ORNL FRACTURE-MECHANICS ANALYSIS FOR SEVERAL PWR RECORDED OCA TRANSIENTS
- APPENDIX P " CALCULATED RT NDT VALUES FOR PLANTS
- P.1 RT_{NDT} Screening Values for All Plants
- P.2 Comparison of NRC and Licensee's Values

LIST OF FIGURES

.

.

Figure 2-1	H. B. Robinson SLB 04/28/70
Figure 2-2	H. B. Robinson SLB 04/28/70
Figure 2-3	H. B. Robinson Stuck S.G. Valve 11/05/72
Figure 2-4	H. B. Robinson Stuck S.G. Valve 11/05/72
Figure 2-5	H. B. Robinson RCP Seal SBLOCA 05/01/75
Figure 2-6	H. B. Robinson RCP Seal SBLOCA 05/01/75
Figure 2-7	Rancho Seco NNI/ICS 03/20/78
Figure 2-8	Rancho Seco NNI/ICS 03/20/78
Figure 2-9	Three-Mile Island 2 03/28/79
Figure 2-10	Three-Mile Island 2 03/28//9
Figure 2-11	R. E. Ginna SGTR & PORV 01/25/82
Figure 2-12	R. E. Ginna SGTR & PORV 01/25/82
Figure 2-13	PTS Precursor Events Final Temperatures - 32 Events Identified
Figure 2-14	Frequency Based on Operating History - Final Fluid Temperature
Figure 3-1	Development of PTS Transient in a Vessel Wall
Figure 3-2	NRC Crack Initiation and Arrest Model
Figure 3-3	Generic Calcualtion of Crack Initiation for Stylized Transients
Figure 3-4	Sensitivity Studies
Figure 4-1	Frequency Based on Operating History Critical RT _{NDT} Value (OCA Results)
Figure 4-2	Operating Experiences (90% Confidence Intervals)
Figure 5-1	Example of NRC Prescription for RT _{NDT} (for assumed) RT _{NDT} (O) of O°F)
Figure 6-1	Frequency Based on PRA Studies - Final Fluid Temperature
Figure 7-1	CFP As Function of T _f - RT _{NDT}

.

Figure 7-3CFP As Function of Cooldown Rate, Min-1Figure 7-4CFP As Function of Heat Transfer CoefficientFigure 8-1Frequency Based on PRA Studies - Final Fluid Temperature

CFP As Function of Pressure

Figure 7-2

Figure 8-2 Longitudinal Crack Extension No Arrest - Westinghouse Owners Group PRA Results 4

Figure 8-3 Longitudinal Crack Extension No Arrest - NRC Staff PRA Results

LIST OF TABLES

•

Table 2-1	Parameters for Stylized Representation of Experienced Events
Table 3-1	Critical RT _{NDT} Values for Experienced Events
Table 6-1	Initiating Events
Table 6-2	Event Parameters for the Main Steam Line Break (MSLB)
Table 6-3	Event Parameters for the Small Steam Line Break (SSLB)
Table 6-4	Event Parameters for the Small-Break Loss-of-Coolant Accidents (SBLOCA), Break Size 2-6 Inches Equivalent Diameter

1. INTRODUCTION

1.1 Background

Reactor pressure vessels (RPVs) in nuclear power plants have traditionally been considered extremely reliable structural components. Indeed, studies completed in the United States and Europe have concluded that the disruptive failure rate (loss of the pressure retaining boundary) for nuclear pressure vessels is less than 10-6 per year at a 99% confidence level for RPVs designed, fabricated, inspected, and operated in accordance with the Boiler and Pressure Vessel Code of the American Society of Mechanical Engineers. However, recent results from surveillance and research programs and operating experience suggest that the issue of RPV failure probability should be reassessed. The renewed interest in RPV failure probability is due to the observation that thermal hydraulic transients occurring in commercially operating nuclear nower plants are subjecting RPVs to unanticipated loadings which could contril le significantly to the failure probability of the RPV. In addition, operating experience and research programs over the past few years have provided additional information that more clearly defines both material property variations in RPVs and the effect of neutron irradiation on the material's resistance to fracture.

ŧ

As a result of operating experience, it is now recognized that transients can occur in pressurized water reactors (PWRs) characterized by severe overcooling causing thermal shock to the vessel, concurrent with or followed by repressurization (that is, pressurized thermal shock, PTS). In these PTS transients, rapid cooling of the reactor vessel internal surface causes a temperature distribution across the reactor vessel wall. This temperature distribution results in the mal stress with a maximum tensile stress at the inside surface of the vessel. This magnitude of the thermal stress depends on the temperature profile across the reactor vessel wall. This thermal stress are compounded by pressure stresses if the vessel is pressurized.

PTS REPORT SEC 1

Severe reactor system overcooling events which could be accompanied by pressurization or repressurization of the reactor vessel (PTS events) can result from a variety of causes. These include instrumentation and control system malfunctions including stuck open valves in either the primary or secondary system, and postulated accidents such as small break loss-of-coolant accidents (LOCAs), main steam line breaks (MSLBs), and feedwater pipe breaks. As long as the fracture resistance of the reactor vessel material remains relatively high, such events are not expected to cause failure. After the fracture toughness of the vessel is reduced by neutron irradiation (and this occurs at a faster rate in vessels fabricated of materials which are relatively sensitive to neutron irradiation damage), severe PTS events could cause propagation of fairly small flaws that are postulated to exist near the inner surface. The assumed initial flaw might initiate and propagate into a crack through the vessel wall of sufficient extent to threaten vessel integrity and, therefore, core cooling capability.

1

The PTS issue is a concern only for operating PWRs. Boiling water reactors (BWRs) are not a significant PTS concern. BWRs operate with a large portion of water inventory inside the pressure vessel at saturated conditions. Any sudden cooling will condense steam and result in a pressure decrease, so simultaneous creation of high pressure and low temperature is improbable. Also contributing to the lack of PTS concerns for BWRs is the lower fluence at the vessel inner wall, and the use of a thinner vessel wall which results in a lower stress intensity for a postulated crack.

The PTS issue is being handled separately from the "cold overpressurization" issue that was investigated (and for which corrective actions were required) several years ago. The "cold overpressurization" problem and corrective actions are described in Reference 1.1. "Cold overpressurization" refers to events such as isolation of letdown flow and/or inadvertent starting of a high pressure pump, while the reactor is in cold shutdown condition. No thermal stresses are involved, as in PTS, because the plant is uniformly cold. All of the stresses result from the pressure, which alone can cause vessel rupture (if a flaw is present) due to the much colder temperature present during shutdown (compared to PIS events where the vessel is warmer).

PTS REPORT SEC 1

Corrective actions were applied, for cold shutdown conditions, involving disabling of pumps that could cause overpressurization, changing the setpoint of the pressurizer pressure relief valve, and incorporating the shutdown heat removal system's relief valves into the system.

Since the condition leading to "cold overpressurization" events, and the corrective actions for such events, are different from the PTS concern, "cold overpressurization" was handled separately and is not discussed further in this report.

1.2 Staff Reviews of PTS Information Provided by Licensees and Industry

Evaluations of Pressurized Thermal Shock by the NRC staff in the spring of 1981 concluded that no immediate licensing actions were required at that time, but that since the consequences of overcooling events increase as the vessels accumulate additional neutron irradiation, extensive further investigations were needed to determine whether and when corrective actions will be needed to provide assurance of vessel integrity throughout the intended service life of a reactor vessel.

On March 31, 1981, the NRC staff held the first of many meetings that were to occur over the following 16 months with licensees, reactor manufacturers, and owners groups to discuss pressurized thermal shock concerns and exchange technical information.

Subsequently, the NRC, in letters dated August 21, 1981, requested the licensees of eight plants representative of older reactor vessels to provide more deta led information on the present and projected pressure vessel materials properties, on the probability and possible severity of events that could cause failure of embrittled vessels, and on the efficacy and feasibility of several potential corrective actions.

Many of the event-sequence analyses provided by licensees in response to the August 21, 1981 letter can be characterized as design-basis event analyses of the type generally submitted in Safety Analysis Reports (SARs) in support of license applications. Such analyses tend not to be of much help in evaluations

PTS REPORT SEC 1

6

1-3

1

of PTS. Many of the assumptions in such analyses were developed and accepted for licensing purposes without regard to PTS concerns. While SAR analyses appear to be appropriately conservative for calculations of reactor core thermal performance, PTS evaluations need best estimate calculations of pressure and temperature behavior. In addition, some potential event sequences that are not generally analyzed in detail in Safety Analysis Reports, because their consequences are bounded by the design-basis event analyses, can be of greater significance for PTS evaluations. Thus, it is clear that plant-specific PTS evaluations must include a systematic examination of many potential events, with particular attention to the probability and consequences of various possible operator actions and omissions, and equipment malfunctions.

. 4 -

Appendix A to this report summarizes the meetings that have been held with industry, licensee responses to the August 21 letters, and the NRC staff audits of operating procedures, operator qualifications, and training with respect to the PTS issue. Appendix B lists significant events and meetings concerning PTS. Appendix C is a more detailed discussion of the procedures and training audits.

As a result of the review of the extensive information provided by the industry, and of studies and analyses performed by the staff, assisted by contractors and consultants (see particularly the fracture mechanics calculations performed by the Oak Ridge National Laboratory described in Appendix O, and the report of a technical review of PTS issues performed by Pacific Northwest Laboratory, Reference 1.2), in the spring of 1982, the staff reaffirmed its previous assessment that no immediate plant modifications were needed to protect against PTS events (other than improvements in procedures and operator training already underway, and flux reduction programs to reduce the rate of vessel embrittlement). However, the staff concludes that some plants will require hardware and procedural modifications in the near future. The experience of the past 18 months in generic evaluations of the PTS concerns has made it clear that decisions on the need for, nature of, and timing of, such modifications will require plantspecific, rather than generic evaluations.

1.3 Procedure and Training Reviews

Operators have a significant impact on the course of events following an initiating transient or accident, as shown by operating experience throughout the nuclear power industry. In most cases, operator actions have led to the mitigation of severe consequences that could have resulted from the initial event. There have been cases, however, where the operator has taken an action that was improper for the condition he was facing, which has led in some of those cases to severe consequences to plant components. Improper actions by the operators could have been the result of operating procedures with incorrect technical information for the specific event in progress, cognitive errors, operative errors, a lack of understanding of the fundamental concepts of plant design, or a myriad of other reasons that affect the range of human capabilities. Regardless of the reason, it is important that the operator be given information that is technically correct and useable, and that this information and the operator's concept of the plant be consistent.

. . . .

Audits of procedures and training at seven of the plants used for the short-term evaluation of PTS revealed that the mitigative capabilities of the industry for dealing with PTS were highly inconsistent, and that the knowledge gained from studies of the thermal-hydraulic phenomena and materials properties must be more closely tied to its application in plant operations. The mitigative capabilities were evaluated generally on the basis of the actions specified in plant operating procedures, with the evaluation of operator understanding of both PTS and mitigative actions used to determine if the procedural actions were understood, as well as to determine if the operator's understood thermalhydraulic phenomena and material properties well enough to overcome the procedural deficiencies suspected by the audit teams. The findings of each of these audits are included in Appendix C.

A review of the event-specific Westinghouse Owners' Group generic technical guidelines by Westinghouse resulted in a recommendation that 11 modifications be made to the guidelines. These changes were recommended based on a reevaluation of the basis for each step in light of the recent knowledge gained for PTS. The recommendations included strengthening cautions to maintain pressuretemperature limits, and lowering the target pressure for stabilizing plant

PTS REPORT SEC 1

conditions during the recovery phases to a pressure consistent with the Technical Specification limits.

Ĺ

Upgrading of operator's ability to deal with PTS is contingent upon having sound technical knowledge of the phenomena that are involved, a clear understanding of how the phenomena relate to equipment operation or malfunction, a program to translate the technical information into operational information, and a program for presenting the information to the operators in a coherent manner. This information must be integrated with other technical information to ensure that all relevant technical concerns are considered in the actions that the operator is instructed to take.

The only effective way of ensuring this integrated approach to accident mitigation is to provide the operator with procedures that are developed using an integrated analytical base, and to provide in the procedures training program the background information necessary for the operator to have confidence in the procedures, and thus make them useable. The approach the staff is taking is to have the industry use technical guidelines being developed for TMI Action Plan Item I.C.1 as a basis for integrating procedural actions to be used for dealing with PTS. By using the procedures developed from those guidelines, the industry will gain a consistent and improved ability for dealing with PTS. An additional benefit to this approach is that by having the analytic base documented, knowledge of PTS gained from the long-term program can be effectively integrated into the existing analytic base.

1.4 Proposed Approach for Future Evaluations

For the reasons noted above, there is a need for a disciplined technical basis to select plants for which detailed evaluations are required and to determine the timing of such evaluations. The approach proposed by the staff is to select a screening criterion that characterizes the present or projected state of embrittlement of reactor vessels as a function of neutron fluence. Licensees of plants with vessels that are projected to reach the screening criterion within 3 calendar years would be required to submit detailed, plant-specific evaluations of: the vessel condition; the expected frequency, course, and consequences of

experienced and postulated overcooling events; plant procedures and operator training related to prevention or mitigation of PTS events; possible modifications of plant equipment, systems, and procedures that could reduce the probability and/or severity of overcooling events; possible improvements in in-service inspection methods that could provide increased assurance of the detection of existing flaws in critical regions of the pressure vessel; and possible modifications to decrease the rate of vessel embrittlement or actions to recover ductility.

. 1.

These licensees would also be required to provide a technical basis for the acceptability of continued operation of the plant for the remainder of its design life taking into account the risk of pressure vessel failure from pressurized thermal shock events, based on the above plant-specific evaluations and such remedial actions as are proposed.

The screening criterion proposed by the staff is based on a parameter that characterizes the toughness state of the reactor vessel beltline region. This parameter is the reference temperature for nil ductility transition, RT_{NDT} . At normal operating temperatures, vessel materials are quite tough and resistant to crack propagation. As the temperature decreases, however, the metal gradually loses toughness over a temperature range of about 100°F. RT_{NDT} is used to identify where this toughness transition occurs. Its value depends on the specific material in the vessel wall and it increases with neutron irradiation. These effects are determined by destructive tests of material specimens.

The RT_{NDT} used for screening purposes is the value at the inner surface of the vessel wall. Fracture mechanics analyses, however, take into account the tougher material within the wall due to attenuation of neutron fluence and generally higher temperatures during a cooldown transient. Equations, based on tests of irradiated specimens, have been developed to calculate the shift in RT_{NDT} as a function of neutron fluence for various material compositions. The value of RT_{NDT} at a given time in a vessel's life is used in fracture mechanics calculations to determine whether assumed pre-existing flaws would propagate as cracks when the vessel is subjected to overcooling events.

L

The staff's approach to selection of an RT_{NDT} screening criterion has been to consider the overcooling events that have occurred in U.S. PWRs and, using a deterministic fracture mechanics algorithm, calculate the value of RT_{NDT} for which assumed pre-existing flaws in the reactor vessel would be predicted to initiate (grow deeper into the vessel wall). These "critical" values of RT_{NDT} were related to the expected frequency of the initiating events, based on the limited data base (only eight events in 350 reactor-years). Specific values of RT_{NDT} are selected for use as the screening criterion for axial and circumferential flaws.

ć

In addition, the staff considered a wide spectrum of postulated overcooling events that have not occurred. These events were grouped into types, estimates were made of their expected frequency, and stylized characterizations of the temperature and pressure time-histories were developed for each event type. A probabilistic treatment of the fracture mechanics calculations was developed that permitted performance of studies to gain insights into the sensitivity of the fracture mechanics calculations to uncertainties in the various input parameters. By combining the calculated frequencies and characteristics of postulated events with the probabilistic fracture mechanics results, some very approximate estimates of the probability of vessel failure resulting from PTS eents were developed and used by the staff to provide some insight into the residual risks inherent in use of the screening criterion approach for further evaluations and resolution of the issue of pressurized thermal shock.

1.5 Structure of this Report

This report provides the NRC staff's technical basis for the selection of the screening criterion, and a brief description of the type of plant-specific analyses that would be required for plants with pressure vessels that are projected to exceed the criterion.

Section 2 of the report discusses the frequency and characterization of overcooling events that have actually been experienced. Section 3 summarizes deterministic fracture mechanics calculations performed for these experienced events and parametric studies of crack growth potential as a function of the

event characteristics and RT_{NDT} values. Section 4 combines the results of Sections 2 and 3 and proposes values of RT_{NDT} for use as a screening criterion.

i

Section 5 presents the staff's proposed method for estimation of vessel-specific values of RT_{NDT} for comparison with the screening criterion.

Section 6 describes an evaluation of the frequency and character of potential lower frequency overcooling events.

Section 7 summarizes sensitivity studies performed using a probabilistic treatment of the fracture mechanics calculations that can be used in combination with the results of Section 6 to estimate probabilities of vessel failure. Consideration of these results is presented in Section 8.

Section 9 indicates the nature and timing of the plant-specific evaluations that would be requested for plants approaching the screening criterion.

Section 10 presents the conclusions and recommendations of the NRC staff regarding near-term actions and future programs for resolution of the pressurized thermal shock issue.

References

- 1.1 "Reactor Vessel Pressure Transient Protection for Pressurized Water Reactors," NUREG-0224, September 1978.
- 1.2a) L. T. Pedersen, W. J. Apley, S. H. Bian, L. J. Defferding, M. H. Morgenstern, P. J. Pelto, E. P. Simonen, D. L. Stevens and T. T. Taylor, Pacific Northwest Laboratory "PNL Technical Review of Pressurized Thermal Shock Issues," USNRC Report, NUREG/CR-2837, Jul 1982
 - b) Supplement 1 to above, September 1982
 - c) L. T. Pedersen, PNL, letter to R. Woods, NRC, regarding the September 13, 1982 Draft NRC PTS report, October 7, 1982
 - d) L. T. Pedersen, PNL, letter to F. Litton, NRC, regarding issues reviewed for reference 1.2b), October 22, 1982

e) Later revised version of Supplement 1, November 5, 1982

PTS REPORT SEC 1

1-9

\$

2. FREQUENCY AND CHARACTERIZATION OF EXPERIENCED OVERCOOLING EVENTS

·· · •

2.0 Preface

L

The data provided in this section were compiled by the staff as an attempt to quantify overcooling events that occurred at commercial U.S. PWRs which appeared to be of concern for PTS.

In reviewing the available data, a conscious decision was made to use the data in its elemental or raw form, to the extent deemed reasonable. Temperature measurements, without consideration of instrumentation errors, taken in the reactor coolant system cold leg upstream of the safety injection location are, in general, used to characterize an event.

It is recognized that use of the data in this manner may be somewhat nonconservative as the temperature of concern is that at the coldest location in the vessel downcomer. The actual downcomer temperature is dependent on a number of parameters which are not available, at least to the degree required, to quantify the actual temperature. These time-dependent parameters are: (1) cold leg fluid temperature upstream of the safety injection location, (2) cold leg loop flow rate, (3) system pressure, (4) safety injection fluid temperature, (5) safety injection flow rate, (6) heat addition from metal walls, (7) vent valve flow rate for B&W plants, (8) vent valve fluid temperature for B&W plants, and (9) local flow rates in the vessel downcomer.

Thermal-hydraulic models are just now beginning to be developed to the depth necessary to obtain the information required to determine these parameters ar i obtain a downcomer temperature profile. Current efforts by both industry and the NRC are also, just now, obtaining the experimental data required to verify the new analytical models needed to fully understand these events.

The presentation of an event as a PIS event, in this section, is speculative. The data used address the question "What if?" this pressure-temperature history actually occurred at the most limiting location.

11/12/82
We do not, at this time, have the necessary analyses to adjust the cold leg measurements to the downcomer temperature for any given event. We recognize that use of cold leg temperature measurements in this way is subject to criticism and cannot be completely justified. However, we believe these data are representative of operational occurrences and define an appropriate data set for insight into the PTS issue.

. .1 -

2.1 Introduction

This section of the report describes the staff's review of eight actual overcooling transients of interest as potential PTS initiators. The event descriptions were reviewed and plots of pressure and temperature as functions of time were developed, based on plant data available. These actual pressure and time histories were used, as described in Section 3, in deterministic fracture mechanics calculations for each event to determine critical RT_{NDT} values.

In addition, the actual temperature versus time data for each event were fit to a simple stylized characterization of the temperature transients that could be used conveniently in parametric fracture mechanics studies. For this purpose, the fluid temperature at the reactor pressure vessel inner surface is assumed to decrease exponentially from the initial temperature. The equation used is:

$$T = T_0 - (T_0 - T_f) (1 - \exp(-\beta t))$$

Where T_0 = initial temperature, ⁹F T_f = final temperature, ⁹F β = cooldown parameter, min⁻¹ t = time, min

For each of the operating experience events, the actual event sequences were reviewed and values of T_f and β were selected to characterize the event. The selection of T_f and β required some engineering judgment. In general, the final temperature is selected to characterize the observed value when a temperature plateau is reached. The cooldown temperature profile is the best fit of the cooldown or is adjusted somewhat for cases where the temperature increases

11/12/82

following termination of the uncontrolled cooldown. The adjusted cooldown used is based on the Westinghouse approach, which considered the fracture mechanics response to the actual temperature transients and the fracture mechanics response to the stylized formulation with an adjusted β value. The adjusted β^* is obtained from

Ε.

β* 2/t*

where $t^* = the time of lowest temperature,$ and $<math>\beta^*$ is never less than the "natural" cooldown β .

A representative constant value of the pressure was also selected for each event. These stylized representations of the experienced events were then used for comparison with parametric fracture mechanics calculations as described in Section 3.

Finally, the eight events of interest were used to construct a cumulative frequency distribution of observed events as a function of T_f which is considered in Section 4 in selecting a screening criterion.

The eight events represent very limited statistical data, with five events on Westinghouse plants, three on Babcock and Wilcox plants, and none on Combustion Engineering plants. The data represent the best information available, however, and were therefore used to define generic screening criteria. The same screening criteria were recommended fo^rall three vendors' plants because of the limited statistical significance of indicated differences between the three groups and (in CE's case) because of their plants' similarity to Westinghouse plants, where data is available. Use of the same screening criteria for B&W plants needs further near-term justification, as discussed in Section 4. In addition, if there are real differences in PTS risk between the three vendors plants, those differences will be identified in the plant-specific analyses that are required when a plant "trips" the screening criteria, and account will be taken of those differences before corrective actions resulting from those analyses are required. In the immediate future, however, we recommend use of the same screening criteria for all PWRs. 2.2 Event Descriptions

2.2.1 H. B. Robinson Steam Line Break (04/28/70)

On April 28, 1970, during hot functional testing (no fuel loaded), one of the steam generator safety value connections failed due to overloading. A 360° circumferential break allowed the safety value to blow off the main steam line. The plant conditions were:

- 13

- 533°F, 2225 psi primary
- 900 psi secondary
- 3 reactor coolant pumps (RCPs) running
- 45 gpm charging/letdown
- no feedwater to the steam generator

As a result of the 6-in. schedule 80 pipe break, and with no decay heat, the plant cooled down 213°F in \cong_2^1 hour to a 320°F cold leg temperature. The operator immediately tripped the RCPs (30 seconds) and started the remaining two coolant charging pumps (70 seconds). The minimum primary system pressure was 1880 psi, with the safety injection (SI) setpoint at 1715 psi, no safety injection occurred. The plant was recovered to a normal no-load condition of 2050 psig and charging/letdown reestablished prior to shutdown.

A post-event review of the data indicated that the pressurizer surge line did not empty. An analysis was performed for the event. In addition, a sensitivity analysis was performed without RCP trip, with only one charging pump, and with a primary heat source. The analysis showed that the pressurizer would drain and the primary system pressure would fall below the SI setpoint in about 3 minutes. The cooldown was less and the pressures were lower than the base case analysis. It is expected that the operator actions, based on current procedures, would be similar to this sen ivity analysis. The safety value stand-off piping was redesigned to prevent any similar occurrences.

The transient data for this event are provided in Figure 2-1. For the stylized characterization of the event the staff selected $T_f = 295^{\circ}F$, $\beta = 0.08 \text{ min}^{-1}$ and

a pressure of 2000 psig. This exponential temperature curve is compared with the broken loop cold leg temperature data in Figure 2-2.

1 14 . .

2.2.2 H. B. Robinson Stuck Steam Generator Relief Valve (11/05/72)

While at nominal full power operation conditions, the operator was using steam generator relief values to provide reactor coolant system (RCS) temperature control. One value would not reclose, resulting in the equivalent of a small steam line break. The secondary side blowdown resulted in a reactor trip and safety injection. The overall cooldown rate was 200°F over a 3-hour period, to 340°F during the course of the event. Insufficient information is currently available to address operator actions taken during this event.

The transient data for this event are provided in Figure 2-3. For the stylized characterization of the event the staff selected $T_f = 340^{\circ}F$, $\beta = 0.015 \text{ min}^{-1}$ and a pressure of 1000 psig. The exponential temperature curve is compared with the cold leg temperature data in Figure 2-4.

2.2.3 H. B. Robinson RCP Seal SBLOCA (05/01/75)

During full power operation, RCP "C" seal number one leakage exceeded the technical specification limit of 6 gpm. A load reduction was commenced at a rate of 10% per minute to 36% power and pump "C" was deenergized. Reactor trip occurred due to a turbine trip resulting from the load reduction. The decision was made to restart pump "C" when seal injection could not be restored to pumps "A" and "B." Shortly after restarting the pump, while at 1700 psig and 480°F, seals number two and three failed on pump "C" and the pressurizer level begar to decrease.

Safety injection pumps were manually started, charging flow was diverted to the auxiliary pressurizer spray to reduce pressure and the SI accumulators partially injected when the pressure dropped to 500 psig.

The cooldown for this event was from 450°F to approximately 310°F in one-hali hour, with the pressure decreasing from 1700 psig to about 1150 psig over the



.

۶۴.







د. م<u>ن</u> . 4 -

note : 17 and 18 deleted



period of interest. The use of the auxiliary pressurizer spray rapidly reduced the pressure to 500 psig.

The operator used SI to stabilize pressurizer level and pressure while using the main condenser to cool down the plant for RHR entry.

There is no indication that SI was used to repressurize the plant.

The transient data for this event are provided in Figure 2-5. For the stylized characterization of this event, the staff selected $T_f = 250^{\circ}F$, $\beta = 0.02 \text{ min}^{-1}$ and a pressure of 500 psig. The exponential temperature curve is compared with the broken loop cold leg temperature data in Figure 2-6.

2.2.4 Rancho Seco NNI/ICS (03/20/78) (excess feedwater transient)

On March 20, 1978, the Rancho Seco plant RCS was cooled from 582°F to about 285°F in slightly more than one hour (approximately 300°F/hr), while RCS pressure was about 2000 psig. The transient was initiated by an inadvertent short in a DC power supply causing a loss of power to the plant's non-nuclear instrumentation (NNI). Loss of NNI power caused the loss of most control room instrumentation and the generation of erroneous signals to the plant's Integrated Control System (ICS). The ICS reduced main feedwater, causing the reactor to trip on high pressure. The cooldown was initiated when feedwater was readmitted to one steam generator by the ICS (auxiliary feedwater was restored). The cooldown caused system pressure to drop to the setpoint (1600 psig) for the safety feature: actuation system, which started the high pressure injection pumps and auxiliary feedwater to both steam generators. High pressure injection flow restored pressure to 2000 psig. With control room instrumentation either unavailable or suspect for one hour and ten minutes (until NNI power wis restored), operators continued auxiliary feedwater and main feedwater to the steam generators while maintaining RCS pressure with the high pressure injection pumps.

The transient data for this event are provided in Figure 2-7. For the stylized characterization of this event the staff selected $T_f = 285^{\circ}F$, $\beta = 0.10 \text{ min}^{-1}$ and



)



<u>ن</u>





Å

)



(j. 188).

٤

RANCHO SECO NNI/ICS 03/20/78

)

Ĵ

)

a pressure of 2300 psig. The exponential temperature curve is compared with the cold leg temperature data in Figure 2-8.

· . l

2.2.5 Three Mile Island 2 (03/28/79)

This accident was initiated by a loss of normal feedwater to the steam generators resulting in a turbine trip. As a result of the loss of heat sink, the RCS overpressurized and the power-operated relief valve (PORV) opened, which is a normal response and in accordance with the design. The PORV stuck open and remained open for about 2.4 hours, unnoticed by the operator. High pressure injection (HPI) was actuated on low pressure. However, at about 3 minutes into the event an operator bypassed the injection actuation signal. One HPI pump was turned off, and the remaining flow was reduced as a result of a highlevel indication in the pressurizer. HPI was automatically actuated again at about 3.3 hours into the event. For the first 73 minutes the RCPs were running. After this time the pumps were turned off due to excessive vibration.

The transient data for this event are provided in Figure 2-9. For the stylized characterization of this event, the staff selected $T_f = 225^{\circ}F$, $\beta = 0.04 \text{ min}^{-1}$ and a pressure of 2300 psig. The exponential temperature curve is compared with the cold leg temperature data in Figure 2-10.

2.2.6 R. E. Ginna SGTR & PORV (01/25/82)

The plant was operating at 100% power with normal pressure and temperature prior to the steam generator tube rupture (SGTR). The SGTR resulted in automatic reactor trip and automatic actuation of safety injection. On the SI signal, automatic containment isolation occurred and the charging pumps were tripped. Both RCPs were tripped by the operator in accordance with plant procedures. The operators attempted to equalize the primary and faulted SG pressure, in accordance with plant procedures, by opening the PORV. The PORV failed open, and the operator manually closed the block valve to stop the coolant loss.

The transient data for this event are provided in Figure 2-11. For the stylized characterization of this event, the staff selected $T_r = 325^{\circ}F$, $\beta = 0.12$ min⁻¹,



, 1 .

J.

1

1

2







į,

3





. 1

. *.*

and a pressure of 1400 psig. The exponential temperature curve is compared with the cold leg temperature data in Figure 2-12.

1.1

The sudden temperature dip at about 45 minutes has been shown not to be significant in the fracture mechanics analysis, and has been ignored in characterizing this event.

2.2.7 Crystal River 3 NNI/ICS (02/26/80) (small-break LOCA transient)

On February 26, 1980, the Crystal River 3 plant experienced a small-break LOCA transient when a power-operated relief valve (PORV) was opened inadvertently. The resulting transient caused a decrease in RCS temperature (whose magnitude is discussed below) with a system pressure of about 2400 psig. The transient was initiated when an electrical short in a DC power supply for the plant's NNI caused a pressurizer PORV to open, a loss of most control room instrumentation, and the generation of erroneous signals to the plant's ICS. The ICS caused a reduction in feedwater flow and a withdrawal of control rods. RCS pressure initially increased, tripping the reactor on high pressure, and then decreased as coolant discharged through the open PORV. The high pressure injection pumps started at 1500 psig and repressurized the RCS to about 2400 psig. The PORV block valve was closed, but flow out of the RCS continued through the pressurizer safety valves. After approximately 30 minutes, the high pressure injection pumps were throttled back, but RCS pressure was maintained at about 2300 psig for the next one and a half hours while shutdown to cold shutdown conditions by normal operating procedures was initiated.

Since temperatures in the downcomer are not measured, and since many of the temperature measurements normally available were lost when instrumentation power was lost, minimum temperatures were calculated.

For the purpose of this evaluation, the minimum downcomer temperature is base on calculated mixing in the downcomer of the HPI with the minimum vent valve flow (1 vent valve), using the TRAC code and Creare (Ref. 2.1) data for thermal mixing. The mean mixed value for T_f is approximately 250°F (the same value indicated by B&W). A cooldown β of 0.10 min⁻¹ is used, based on a preliminary review of the TRAC analysis, and an approximate time span of 20 minutes prior to the operator regaining control of the transient. For the stylized characterization of this event, the staff selected $T_f = 250^{\circ}F$, $\beta = 0.10 \text{ min}^{-1}$, and a pressure of 2300 psig.

2.2.8 Prairie Island SGTR (10/02/79)

This event was similar to the Ginna SGTR; however, the minimum temperature was 350° F with a β of 0.1 per minute. β is estimated from the adjusted β^* value for a cooldown period of approximately 20 minutes. A pressure of 1000 psig was selected. No plots of temperature and pressure data were available.

2.3 Summary of Operating Experience

In addition to the eight events described in Section 2.2, 24 other events which could have led to PTS concern have been identified. The data sources are the work performed by Phung (Ref. 2.2) and the various licensee submittals on PTS. The Phung study utilized over 16,000 licensee event reports (LERs) covering 329 reactor years of operating experience at 47 PWR units. The search was for events that had one or more of the characteristics of a severe PTS event, which are rapid cooling of the pressure vessel to a low temperature and maintenance of the low temperature and/or rapid cooling rate for several minutes (typically greater than 10 minutes), plus maintenance of a high vessel pressure or vessel repressurization. A list of significant PTS events was presented and the most significant PTS event from the list and from the NRC staff's own experience are presented in Figure 2-13, where final temperature is shown for 24 events. It is noted that the CE submittals did not identify the Millstone-2 and St. Lucie-1 events as PTS events of concern. It is also noted that 2 of the 3 San Onofre-1 events were not identified by Phung. By vendor there are 21 Westinghouse events, 4 CE events, and 7 B&W events. Only the eight events discussed above which resulted in final temperatures of 350°F and less with significant pressure, fast cooling and sustained low temperature are of interest for the detailed PTS analyses. These eight events selected for detailed analyses are underlined in Figure 2-13. The values of T_f , β and pressure that have been selected to characterize these events are summarized in Table 2-1.



FIGURE 2-13

3.1



-بر ح The eight events characterized in Table 2-1 above occurred during approximately 350 reactor-years of PWR operating experience. On that basis, a cumulative frequency distribution has been plotted as a function of the final temperature of the event, T_f , as shown in Figure 2-14.

2.4 Comparison with Westinghouse Characterization of Operating Experience

Westinghouse believes the operational events referred to in this section that occurred in Westinghouse-designed plants should be characterized somewhat differently. (Their most recent discussion is contained in Reference 2.3)

The comparison is as follows:

Event		<u>NRC</u> T _f	W	T _f
HBR '75		250	3	27
Ginna		325	3	00
HBR '70		295	2	95
HBR '72		340	4	00
Prairie	Island	350	3	90

The differences are due to three causes, according to Westinghouse.

First, they state that we plotted the cumulative distribution of events incorrectly. We agree, with respect to a much earlier curve we used. We now plot T_f correctly in Figure 2-14.

Second, they state that one of the events (HBR-'70) was a pre-fuel loading event that occurred during testing conducted to detect weaknesses exactly like the one that was found and, therefore, should not be included. We agree that inclusion of the event is somewhat on the conservative side, but note that deletion of the event would make no significant difference in our conclusion.

Third, they state that we should terminate an event for PTS consideration when the opertor gets the plant within Appendix G cooldown limits. We do not agree. Certainly it is true that a shutdown under normal conditions within Appendix G limits is not a PTS concern. However, a ccoldown (whether deliberate or

1. -

11/12/82

Event	T _f (⁰F)	β(min- ¹)	P(psig)
Robinson SLB (W) ('70)	295	0. ປ8	2000
Robinson Stuck SG Valve (W) ('72)	340	0.015	<1000
Robinson RCP Seal SBLOCA (W) ('75)	250	0.02	<500
Rancho Seco (B&W)	285	0.10	2300
TMI-2 (B&W)	225	0.04	2300
Ginna ŠGTR (W)	325	0.12	1400
Crystal River-3 (B&W)	250	0.10	2300
Prairie Island SGTR (W)	350	0.10	1000

Table 2-1 Parameters for stylized representation of experienced events

· · · ·

uncontrolled) within Appendix G limits immediately following a more rapid cooldown of PTS concern can very well exacerbate the PTS concern and must be considered.

References:

- 2.1 "Fluid and Thermal Mixing in a Model Cold Leg and Downcomer With Vent Valve Flow," Creare Incorporated, EPRI Report NP-2227, March 1982.
- 2.2 Phung, D. L., "Pressure Vessel Thermal Shock at U.S. Pressurized Water Reactors: Events and Precursors, 1963 to Mid-1981," ORNL, Interim Report, May 1982, and final report ORNL/NSIC-112, pre-print version dated August 15, 1982.
- 2.3 Letter from O. D. Kingsley, Jr. to Harold R. Denton dated Sept. 2, 1982 regarding Westinghouse Owners Group Activities related to Pressurized Thermal Shock.

3. DETERMINISTIC FRACTURE MECHANICS ANALYSES

3.1 Fracture Mechanics Discussion

The calculations reported in this section are used to analyze the response of a reactor pressure vessel (RPV) to an overcooling transient. The input information includes (1) pressure and temperature of the reactor coolant as a function of time, obtained from thermal-hydraulic calculations; (2) materials properties, including temperature and irradiation effects; and (3) an actual or assumed initial flaw. Vessel integrity analyses, the results of which are reported in this document, include a determination of the temperature distribution across the vessel wall versus time, the thermal stresses as a consequence of this temperature distribution, as well as fracture mechanics results. Thus, the term "fracture mechanics (FM) analysis" used in this section really means vessel integrity analysis because it includes heat transfer and stress analysis. The stresses considered are those as a result of pressure and differential thermal expansion between the clad and the base metal as well as thermal stresses.

21

Once the stress distribution is determined as a function of time and position, FM examines the behavior of preexisting cracks (postulated or real) in this stress field. For specific crack geometries, a stress intensity factor, K_I , is calculated and compared to a material toughness property, K_{IC} . When K_I exceeds K_{IC} for a specified crack, the crack will initiate, i.e., grow deeper into the metal. K_I for the crack then increases until it reaches a value equal to K_{Ia} which is another material property. The crack then arrests, i.e., stop growing larger. The material properties (K_{Ic} and K_{Ia}) vary with temperature and degree of irradiation damage and hence are a function of time and depth into the vessel wall.

FM algorithms consider these factors. For pressurized thermal shock (PTS) evaluations, linear elastic fracture mechanics (LEFM) is used because, at the temperatures involved, the metal is at less than its maximum or upper shelf

11/10/82

toughness. Illustrations of typical temperature, stress and stress intensity factor distributions within the vessel wall at different times during the transient are shown in Figure 3-1 (a), (b) and (c) respectively.

~ ` **`**

The quantities K_{Ic} , the vessel toughness that determines crack initiation, and K_{Ia} the toughness at crack arrest, also vary with position and time, since they are functions or irradiation and temperature. When K_{I} exceeds the value of K_{Ic} at the location of the tip of the flaw, crack initiation is expected, if warm prestressing is not effective (warm prestressing is discussed below and in Appendix D). The crack would then grow to a depth where K_{I} equals the value of K_{Ia} at the tip of the growing crack. For some transients, metals properties, and flaws, K_{I} will remain above K_{Ia} and the crack is assumed to go through the vessel wall without arrest if the system is pressurized.

Similar results would occur for a circumferentially oriented crack except that arrest will generally occur at shallower depths because the stress intensity factor, K_{I} , for long axial cracks is higher than for long circumferential cracks, especially for cracks that extend relatively deep into the vessel wall.

Equivalent calculations are made as a function of time in the transient, and the results cross-plotted on a critical crack depth diagram. From this diagram, the behavior of a crack versus time for a particular PTS scenario can be determined. Such a diagram is shown in Figure 3-2. See additional discussion in Section 3.4.4.

Warm prestressing (WPS) is a phenomenon that can inhibit or prevent crack initiation even though the calculated stress intensity factor, K_I , becomes greater than the material toughness parameter, K_{IC} , at the time and location of the flaw tip. For WPS to be effective, K_I must be at less than a previous maximum value at the time K_I becomes equal to or greater than K_{IC} . This can occur if K_I is monotonically decreasing as the metal cools causing K_{IC} to decrease. When the course of a PTS transient can be described with confidence, the time behavior of K_I can be evaluated for determining whether WPS occurs. For generic studies, however, wherein the pressure variation

11/10/82

versus time cannot be unambiguously defined, the NRC does not assume the benefit of WPS. In general, K_{I} will increase after its initial peak only due to an increase in pressure.

÷.,

3.2 Description of FM Computer Analysis Programs

.

The NRC staff utilized its own in-house FM program in performing heat transfer, stress and fracture mechanics analyses related to pressurized thermal shock (PTS) and has also relied on ORNL to supplement the staff analyses by use of the OCA program as reported later in this section. The NRC program is also utilized as the deterministic portion of the VISA program in performing the probabilistic fracture mechanics analyses discussed in Section 7. The NRC and ORNL programs are equivalent, as described in Appendix D. Analytical results utilizing these programs for specific PTS scenarios have been compared and found to be in close agreement. Similar comparisons have been made with results of industry analyses. We conclude that the analytical methods used by the NRC, ORNL and the vendors yield essentially the same results if all input assumptions are the same. Differing conclusions result primarily from various assumptions regarding input parameters.

When material properties and the transient are known, fracture mechanics procedures can predict crack behavior quite well as demonstrated by comparison with a wide variety of experiments. The Heavy Section Steel Technology research program has included hundreds of irradiated test samples, plus model vessels tested at low temperatures to include brittle and transition behavior. Tests have included thermal shock, and plans for the near future include combined pressure and thermal stresses.

When a crack initiates and grows deeper into a reactor vessel wall, the shap it becomes depends on its initial shape, the stress intensity factor along t e crack front and the relative toughness of the metal in which it is growing. Thermal stress analyses for typical PTS transients result in higher tensile stresses at the cooled surface where the metal is colder and hence less toug: than deeper into the wall. Based on analyses where cladding effects are neglected and on thermal shock experiments, cracks tend to grow in length

prior to growing deeper. In other words, the cracks become relatively long. For this reason, the NRC postulates long cracks at the time of arrest regardless of the original postulated crack geometry.

Discussions with Westinghouse personnel indicated that their analyses assumed a self-similar crack shape with a length-to-depth ratio of six during crack growth and at arrest. The staff does not accept the Westinghouse assumption for the reasons discussed above. Subsequently, the Westinghouse Owners' Group has verified that if they utilized the same assumptions as the staff then their results are essentially the same as those of the NRC. (Ref. 3.1) The original differences in the models resulted in significant differences in critical RT_{NDT} at crack arrest.

In view of the importance of this matter, the staff has consulted with recognized experts in this field who have agreed that, although the NRC model is somewhat conservative, it is more realistic than models that include the assumption of a six-to-one ellipse.

3.3 Determination of KIC and KIA

The fracture analyses performed by utilities, vendors and the NRC have all utilized the values of K_{Ic} and K_{Ia} given in Section XI of the ASME Code and reproduced in Appendix D. The Code values are bounds on the conservative (low) side of experimentally determined toughness values. They have been correlated using the relative temperature, T minus RT_{NDT}, which is the reference temperature, nil-ductility transition.

 RT_{NDT} is defined in Appendix D. It is a reference temperature that is used to characterize the transition in material properties, from ductile to brittle, that takes place as the temperature is decreased. Actually, the transition in properties is gradual, taking place over a temperature range of about 100°F. The use of the relative temperature, T-RT_{NDT} has been shown to allow correlation of experimental toughness data in RPV materials at various temperatures, ard irradiation states. The Heavy Section Steel Technology experimental data also show the T-RT_{NDT} correlation.

3-4

11/10/82

PTS REPORT SEC 3

The initial value of RT_{NDT} in a new, unirradiated vessel is quite low, but increases with irradiation. The NRC staff's method for estimating the initial RT_{NDT} and the change in RT_{NDT} caused by irradiation for a given vessel are given in Section 5 and Appendix E of this report. Estimates are given for RT_{NDT} at the inside surface of the vessel wall (at the clad-base metal interface) for the critical locations, which are almost always the welds, either a longitudinal weld or a cirumferential weld in the beltline. The attenuation of RT_{NDT} through the vessel wall is then calculated to get K_{IC} and K_{Ia} at the tips of postulated cracks (see Appendix D).

.

3.4 Generic Deterministic Studies of Crack Initiation

Using the models described in the preceding sections and in Appendix D, NRC and ORNL have performed a variety of deterministic FM analyses. The results are given in Appendix D and are summarized here.

3.4.1 Stylized Transients

.

The stylized transients used are described in Section 2.1, characterized by constant pressure, P, initial water temperature of $550^{\circ}F$, final water temperature, T_f, and exponential decay constant β , minutes⁻¹. The water temperature is assumed to be uniform over the inner surface of the vessel. A constant heat transfer coefficient, h, is used for the water-metal interface. An infinitely long through-clad flaw is assumed to exist on the inner surface of the vessel wall.

3.4.2 Crack Initiation for Stylized Transience

At the request of the NRC staff, ORNL performed a series of analyses with different assumed values of T_f , β , and P assuming that crack arrest and WPS were not effective. The results are plotted as a series of curves of pressure versus $T_f = RT_{NDT}$, an example of which is Figure 3-3. Other examples are provided in Appendix D. Note that from these diagrams, the thresholds of crack initiation can be determined. Thus, for a specific vessel RT_{NUT} and a given β and T_f , it is possible to determine the limiting pressure to avoid crack initiation. Utilizing Figure 3-3, it is possible to relate approximately the RT_{NDT}, that a vessel must possess to avoid crack initiation for a given transient, to the final temperature of the transient. For conservatism when considering a generalized event, it is assumed that a moderately fast cooldown has occurred ($\beta = 0.15 \text{ min}^{-1}$) and that full pressure (2300 psig) exists in the vessel since there is no assurance that it will be possible to take credit for automatic or manual pressure reduction. Thus, the upper right-hand portion of the figure is used, and it is seen that, for T_f of 250 to 300°F, and for longitudinal flaws, final temperatures approximately 5°F above RT_{NDT} are acceptable, but as one proceeds to more severe cooldown events (T_f = 150°F) the final temperature must stay as much as 20°F above RT_{NDT}.

3.4.3 Sensitivity Studies

In addition to the many uncertainties regarding PTS scenarios such as the temperature and pressure profiles versus time, the degree of mixing of cold with warm water, etc., parametric uncertainties in the stress and fracture mechanics analysis become significant when the cooldown temperature, T_f , is approximately equal to RT_{NDT} because small changes in assumptions can influence whether or not crack initiation is predicted. The staff performed analyses similar to those by ORNL with various assumptions as to crack shape and orientation with and without cladding-induced stresses and for different models for fluence attenuation through the wall in order to determine the effects of these assumptions. (Cladding stresses are induced because of the different coefficients of expansion of the stainless steel cladding and the carbon steel of the vessel wall.)

Sensitivity studies used a base case with $T_f = 250^{\circ}F$, $\beta = 0.15 \text{ min}^{-1}$, and considered various values of P. Some results are shown in Figure 3-4. The threshold value of RT_{NDT} for crack initiation is given.

The importance of the pressure (assumed constant in these stylized transients) is shown in Figure 3-4. For axial welds, the critical RT_{NDT} value of 245°F for a 2250 psig transient is increased to 290°F if the pressure can be limited to 500 psig during the time interval of high thermal stresses.

Cladding-no cladding comparisons (Figure 3-4) show a decrease of about 10° F in critical RT_{NDT} when the cladding effect is included.

13

For this reference transient, with $R_{NDT}^{T} = 294^{\circ}F$, the pressure has to be reduced to near saturation within about 30 minutes to avoid crack initiation. However, if the pressure remains constant after an initial drop or monotonically decreases with time for this stylized transient, WPS would be effective at about 18 minutes and crack initiation would not occur after that time. The measured temperatures and pressure experienced in actual overcooling transients (Section 2.2) show ups and downs, some of which would be predicted to negate WPS.

The orientation of postulated cracks affects their behavior during a PTS event. For a specified thermal transient and the same shape and depth of a pre-existing crack, the thermal stress intensity factor for a circumferential orientation is less than that for an axial orientation. The difference is minimal for shallow cracks but becomes significant for deep cracks. The reason for this difference is the relative stiffness of the vessel wall in the two directions which is accounted for in the fracture mechanics analytical model. For typical reactor vessels, the axial and circumferential thermal stresses are essentially equal in magnitude. Axial pressure stresses, on the other hand, are about a factor of two lower than tangential stresses; the axial stresses affecting circumferential cracks and tangential stresses affecting axial cracks. Thus, the total axial PTS stresses are equal to or less than the tangential stresses depending on the system pressure. For the above reasons, circumferential cracks are more tolerant of PTS events.

The difference between the two orientations in terms of critical RT_{NDT} depends on the specific PTS scenario. A limited number of examples described in Appendix D show that for relatively severe postulated transients, the RT_{NDT} difference is about 30°F for crack initiation and the order of 100°F for crack arrest situations. The higher RT_{NDT} s are for circumferential cracks.

Detailed comparison of Westinghouse and NRC calculations show the following sensitivities:

11/10/82

1

PTS REPORT SEC 3

Assumptions		Change in critical value of RT _{NDT} , ^o F	
(a)	Cladding vs. no clad stress	10	
(b)	Continuous flaw for initiation		
	vs. elliptical flaw $(a/c = 1/3)$	20	
(c)	$h = 300 BTU/hr-ft^2-{}^{\circ}F vs.$	15	
	Westinghouse free convection		
	correlation		

41 -

The above tabulated assumption differences account for a total variation of about 45° in critical RT_{NDT} between staff analyses and those of Westinghouse, with the Westinghouse assumptions giving higher values of critical RT_{NDT} than the NRC assumptions. The NRC staff is inclined to accept the Westinghouse assumptions (b) and (c) as more nearly realistic than the NRC staff assumptions, but believes that the cladding effect should be included in accordance with the NRC assumption.

Such "fine tuning" details are relevant to all calculations but are believed by the NRC staff to be within the error band of such calculations. Only for limiting transients like the small break LOCA with loss of loop flow (Section 6 and Appendix G) are these minor corrections important; they are taken into consideration there.

3.4.4 Crack Arrest

For much more severe thermal transients, crack initiation may occur due to high thermal stresses. In this case it is appropriate to consider the potential for crack arrest. Figure 3-2 is a schematic representation of a critical crack depth diagram to illustrate the analytical model used by the staff fcr determing acceptable arrest criteria. For a small crack, the path of the transient is shown by the dotted line in Figure 3-2. An initial flaw of critical depth is shown; smaller or larger flaws would initiate later. Af'er initiation, the crack runs until $K_{I} = K_{Ia}$ as shown, then arrests.

Although the K_{Ia} arrest value becomes quite high at larger times, the model in its simple form does not include ductile tearing. For this reason, a maximum allowable value of K_{Ia} is imposed, the "upper shelf" value. For NRC calculations, an upper shelf toughness of 200 ksi (in.)^{1/2} is assumed; however, higher or lower values may be more appropriate for a specific material.

. 45 .

The vessel remains intact if WPS prevent crack initiation or, if a crack initiates, and it arrests, for crack depths where K_I is lower than the upper shelf value.

Since the total stress intensity is the sum of pressure and thermal contributions, if the thermal value is known at the time of WPS, a diagram like Figure 3-2 gives the maximum pressure allowable for crack arrest. When the thermal stress intensity factor is known at the time of WPS, the maximum pressure is determined such that arrest will occur at or before the time of WPS and for crack depths such that K_I is below the upper shelf curve. The limiting case is shown as point "A" in the figure.

3.5 Prediction of Critical RT_{NDT} (RT_{c}) for Actual Events

For transients that have actually occurred, it is not necessary to make assumptions of the stylized transients of Section 2.1 and the preceding sections of this chapter. Rather it is possible to perform fracture mechanics calculations for the pressure and temperature history as it actually occurred. These calculations were performed assuming a range of RT_{NDT} values, for the eight overcooling transients experienced to date and described in Section 2.2. No credit was taken for "warm prestressing" in these calculations. Thus, it was possible to predict the limiting vessel material condition (critical RT_{NDT} or RT_c) necessary to prevent vessel failure for each of these experienced transients. The results are shown in Table 3.1, for longitudinal cracks, together with results from Section 2.2 of estimating T_f , β , and P for stylized transients to approximate the course of the events actually experienced.

It is seen that, with the TMI exception (where cooldown must stop $16^{\circ}F$ above RT_{NDT}) cooldowns that have actually occurred to estimated values of T_{f} shown

٤

in Table 3.1 would not be predicted to fail the vessel unless they $(T_{f}s)$ were from 10°F to 100°F below the RT_{NDT} of the vessel at the time of the cooldown.

When compared with the results of the stylized procedure presented in Section 3.4.2, which showed that relatively severe cooldown (β =0.15 min⁻¹) should stop 5°F to 20°F above RT_{NDT}, this result shows some of the conservatism generally present in the stylized procedure compared to direct calculations of critical RT_{NDT} for experienced events. Figure 3-3 presents results showing how T_f is related to RT_{NDT} (for deterministic prediction of vessel failure) at two cooldown rates and at any pressure.

Reference:

3.1 Letter (OG~79) form O. Kingsley, Jr., Chairman Westinghouse Owner's Group, to H. R. Denton dated September 16, 1982.

|--|

. +1

T _f (°F)	β(min ⁻¹)	P(psig)	RTc*
225	0.04	2300	209
250	0.02	500	354
325	0.12	1400	378**
285	0.10	2300	295
295	0.08	2000	321
340	0.015	1000	381
250(?)	0.10	2300	(250)
350	0.10	1000	(400)
	T _f (°F) 225 250 325 285 295 340 250(?) 350	$T_f(^{\circ}F)$ $\beta(min^{-1})$ 2250.042500.023250.122850.102950.083400.015250(?)0.103500.10	$T_f(^{\circ}F)$ $\beta(min^{-1})$ $P(psig)$ 2250.0423002500.025003250.1214002850.1023002950.0820003400.0151000250(?)0.1023003500.101000

Critical RT_{NDT} Values for Experienced Events

*RT_c is the RT_{NDT} that is necessry to prevent crack initiation based on actual pressure and temperature variations with time. Stylized values of T_f, β , and P are shown from Section 2 but were not used in these calculations to determine RT_c.

**Applies to circumferential weld. All others apply to axial welds.



FIGURE 3 -1 RESULT OF PTS ON VESSEL WALL

۸

• بر به


FIGURE 3-2, NRC CRACK INITIATION AND ARREST MODEL



•

.

1





1 B





4. SELECTION OF SCREENING CRITERION

The experienced events discussed in Section 3 were used as the basis for selecting a RT_{NDT} screening value, as described in this section.

• ; .. •

The events were listed on Table 3.1 in terms of the cool down temperature (T_f , first column) and in terms of the critical RT_{NDT} described in Section 3.5 (RT_c , last column). Based on about 350 total PWR reactor years operating experience in the United States, the T_f values for the eight events can be used to develop a plot of the cumulative frequency per reactor year of events with final temperatures lower than the temperature shown. This was done in Figure 2-14. Similarly, the RT_c results of Table 3.1 were used to develop a plot of events versus the RT_c for which the deterministic fracture mechanics calculations predict crack extension will occur (Figure 4-1).

The value selected, 270°F, for longitudinal cracks, was based on earlier curves similar to Figures 2-14 and 4-1, which gave values of approximately 260°F and 280°F, respectively, for a nominal event frequency of 10^{-2} per reactor-year. The justification for choosing 10^{-2} is only that this is comfortably lower than the range of "anticipated operating occurrences."

Since the 270°F screening criterion was chosen, the curves of Figures 2-14 and 4-1 have been corrected for earlier errors in the interpretation of the experienced events, so that the 10^{-2} frequency now corresponds to $T_f = 285^{\circ}F$ and RTc = $320^{\circ}F$. The original 270°F screening criterion is now believed to correspond to a frequency of 9 x 10^{-3} for T_f and 6 x 10^{-3} for RT_c. The staft has not readjusted the screening criterion for this change.

The RTc evaluation is, in many ways, a better way to characterize an event than using T_f alone. Calculating RT_c includes the actual time variation of temperature and pressure, and is preferable to the stylized constant pressure

and simple exponential temperature behavior approximation inherent in the T_f evaluation of Figure 2-14. Moreover, characterization of events by T_f alone requires neglect of the effect of different pressure and different time decay constants on PTS severity. Therefore, if selecting a frequency of 10^{-2} had any real justification, a screening criterion of $320^{\circ}F$ would be given by examination of Figure 4-1.

But in fact using 10^{-2} was only an intuitively attractive place to start. The probabilistic analyses reported in Section 8 of this report show a moderate factor of safety -- a moderate degree of conservatism -- for a screening criterion of 270° F, so the staff has decided not to increase this value as Figure 4-1 and a reference event frequency of 10^{-2} might otherwise suggest.

The staff therefore proposes that RT_{NDT} values of 270°F for axial welds, and 300°F for circumferential welds be used as screening criteria to determine when plant-specific evaluations should be performed for operating plants. It is recognized that the choice of a criterion for action on the basis of generic deterministic fracture mechanics analyses and the limited number of overcooling events that have occurred is subject to many uncertainties and assumptions, some of which are conservative, and some are nonconservative.

The experience base used for these evaluations comprises the eight overcooling events experienced so far in U.S. power plants (Section 2). Non-U.S. experience (and non-U.S. reactor-years) have not been included for lack of detailed knowledge.

U.S. experience has been aggregated for the three PWR reactor manufacturers and various balance-of-plant designers, thus neglecting any differences n experence that may exist among the different designs. The data, however, suggests that B&W plants have a history of more frequent and more severe PTS precursor events. Figure 4-2 shows that the difference appears to have statistical significance. On the other hand, the primary cause of two of the three B&W events discussed in Section 2 of the staff PTS report has been corrected. Nevertheless, the staff believes that although at present the same screening criterion should be used for all plants, near-term further justification is needed from the B&W owners for longer term use of this criterion for B&W

5. DETERMINATION OF RT_{NDT}

5.1 Introduction

4

If as is recommended in this report, a value of RT_{NDT} is selected to serve as a screening criterion to determine the timing of plant specific evaluations of possible needed modifications to provide protection against pressurized thermal shock events, then it is important for the staff to select a suitably conservative and uniform method for determining the plant-specific values of RT_{NDT} at a given time. During the service life of the reactor vessel the intitial value of RT_{NDT} (RT_{NDTo}) increases because of neutron irradiation by an amount ΔRT_{NDT} which depends on fluence and materials properties. The initial value, RT_{NDTo}, is determined from materials tests made at the time the vessel was fabricated. The change, ΔRT_{NDT} , is determined from fluence measurements and calculations and from trend curves, based on tests of irradiated specimens, that predict the effects of neutron irradiation. There are a number of uncertainties in the estimation of both RT_{NDTo} and ΔRT_{NDT} , and it is important to establish a prescribed method for calculation with a degree of conservatism appropriate for use in connection with the screening criterion. The methods described in this section were selected based on the recommendations of an NRC Working Group of staff members and consultants (Reference 5.1). The methods and the bases for them are presented in greater detail in Appendix E of this report. The uncertainties in estimates of fluence are discussed in Appendix F.

- -7-

5.2 Estimation of Initial RT_{NDT}

The summer 1972 Addenda to Section III of the ASME Boiler and Pressure Vessel Code contained the first requirements for measurements to be made at the time of fabrication of RT_{NDT} for the plates, forgings, and welds that make up the reactor vessel. Two types of tests are required--drop weight tests and Charp/ tests. However, most of the vessels of concern regarding PTS were fabricated in the 1960's when only Charpy tests were required. Typically, the data available comprise three Charpy tests at 10° F for each plate, forging and weld, complete Charpy curves for the surveillance weld and base materials, and in cases where the base material was controlling, some drop weight data on archive or surveillance material. In the past, the NRC has used the guidelines given in the Standard Review Plan Branch Technical Position MTEB 5-2, to obtain an estimate of initial RT_{NDT}. In summary, those guidelines were to use the temperature corresponding to a Charpy 30 ft-lb level, but not lower than 0° F. The Charpy curves from the surveillance tests were used to guide any extrapolation needed to get the 30 ft-lb temperature from the three test results at + 10° F. Such estimates are not very satisfactory, however. They are overly conservative for some cases.

_ I, .

For compilations of data obtained subsequent to the time the vessels in question were made, it is possible to divide the welds into two groups according to the weld flux used, and to develop a mean value and a standard deviation (sigma) for the generic data. One must then decide if it is prudent to use the mean generic value as the best estimate for the vessel welds in question. Except for some archive material, the welds that are represented in the data based were made at a later time than the vessel welds. There may have been some differences in weld chemistry or welding practice. Furthermore, even if there were actual RT_{NCT} values for the vessel weld in question, the samples would come from weld metal qualification welds, not from actual vessel weld prolongations and not from full thickness test pieces.

The staff has concluded that a suitably conservative method for estimating the initial value of RT_{NDT} for use in comparisons with the screening criteria proposed in Section 4 is to use the mean value as described above with an adjustment for the standard deviation as discussed in Section 5.4 below. Additional discussion and details regarding the estimation of the initial RT_{NDT} are presented in Appendix E and in Reference 5.1.

5.3 Estimates of the Shift in RT_{NDT} Due to Radiation (ΔRT_{NDT})

Two methods are generally used to estimate the shift in RT_{NDT} caused by neutron irradiation of the pressure vessel: (1) tests of metallurgical

PTS REPORT SEC 5

surveillance specimens irradiated in the reactor vessel, and (2) "trend curves" of ΔRT_{NDT} as a function of weld chemistry and neutron flux developed from analyses of a large number of irradiated specimens.

- Ni

Surveillance specimens have been withdrawn and tested for many older operating plants. However, there are problems associated with using individual surveillance results as the sole source of information about a plant. First, the surveillance weld often does not match the critical vessel weld exactly, i.e., the weld wire heat numbers are different. A broader problem is that caused by scatter in the ΔRT_{NDT} data. This results in part from the fact that ΔRT_{NDT} is the difference between the curves for irradiated and unirradiated material, both of which were fitted to data that typically show considerable scatter. Thus, there is a preference for the use of trend curves, instead of individual surveillance data.

Regulatory Guide 1.99 Rev. 1 published in April 1977 contains the procedure recommended at that time by the NRC to obtain ΔRT_{NDT} as a function of chemistry and reutron fluence. Copper was the dominant residual element in the chemistry term (the other was phosphorus).

Critics of Regulatory Guide 1.99 have asserted that (a) the curves are too conservative at high fluences, especially for low-nickel materials, and (b) the phosphorus term is not supported by recent studies such as that of the Metal Properties Council (Reference 5.2) and EPRI (Reference 5.4). Evidence has been accumulating for several years that low-nickel materials are less sensitive to neutron radiation. When the PWR surveillance data base was analyzed by the NRC in October 1981, the difference between high- and low-nickel content material was apparent. Westinghouse and CE reported similar findings and presented empirical equations for the low-nickel material. (B&W has no plants with lo *r*nickel materials in the reactor vessel.)

The PWR surveillance data have now been fitted by a multiple regression analysis technique. The work was done at HEDL by George Guthrie (Ref. 5.3). The Guthrie mean curve is as follows:

4

PTS REPORT SEC 5

$$\Delta RT_{NDT} = (-10 + 470 \text{ Cu} + 350 \text{ Cu Ni}) (f/10^{19})^{0.27}$$

1. .

where:

ΔRT_{NDT} = adjustment of reference temperature, degrees F Cu = weight percent copper Ni = weight per nickel f = fluence, n/cm² (E>1 MeV)

The standard deviation obtained from the analysis is 24°F.

As shown in Appendix E, the Guthrie mean curve has been compared with a mean curve developed by the Metal Properties Council (MPC) for ASTM Committee E-10 on Nuclear Technology and Applications (Ref. 5.2) and EPRI (Reference 5.4). The MPC data based contains all of the test reactor and surveillance data that were available in November 1977, and that fit the criteria for material type and irradiation temperature. There is reasonably good agreement between the MPC trend curves and the Guthrie curves, considering that the MPC curves were for a range of nickel content, but were without a nickel term in the equation.

The MPC trend curve did not contain a phosphorus term, because in the regression analysis the addition of a phosphorus term did not produce any significant decrease in the residual variance. In a further study of this finding, the MPC Task Group found a statistically significant relationship of phosphorus content to copper content, i.e., high phosphorus was found with high copper. Thus, their combined effects were represented in the MPC trend curve formulation by a copper term alone.

For high values of copper and nickel contents, the Guthrie mean curve described above gives values higher than those predicted by the part of the Upper Limit Curve of RG 1.99, given by the equation:

$$\Delta RT_{\rm NDT} = 283 \ (f/10^{19})^{0.194}$$

Experience has shown that the latter equation bounds the available data.

Therefore, in developing the method for estimating RT_{NDT} values to be compared with the screening criteria proposed in Section 4, the staff

11/12/82

5-4

٨

recommends that ΔRT_{NDT} be calculated using a combination of the Guthrie mean curve and the RG 1.99 upper bound curve, with adjustments for the standard deviation as discussed in Section 5.4.

11

5.4 Recommended Method for Calculation of RT_{NDT}

4

An NRC Working Group of staft members and consultants reviewed the available information regarding RT_{NDT} determinations and recommended that the following method for calculating RT_{NDT} values for specific reactor vessels be used for comparison with the screening criteria of Section 4 (Ref. 5-1).

The value of RT_{NDT} at the inside surface of the vessel should be taken as the lesser of:

$$RT_{NDT} = RT_{NDT}(0) + \Delta RT_{NDT} (mean) + (\sigma^{2} + \sigma^{2})^{\frac{1}{2}}$$

$$RT_{NDT} = RT_{NDT}(0) + \Delta RT_{NDT}(RG) + 2\sigma_{0}$$
where:

$$RT_{NDT}(0) = \text{ the mean value of the initial } RT_{NDT} \text{ determined as described in Section 5.2 above and in Appendix E.}$$

$$\Delta RT_{NDT}(mean) = \text{ the mean value of } RT_{NDT} \text{ based on the Guthrie trend curve}$$

$$= (-10 + 470 \text{ Cu} + 350 \text{ CuNi}) (f/10^{19})^{0.27}$$

$$\Delta RT_{NDT}(RG) = \text{ the upper bound curve of Regulatory Guide 1.99 for high values of copper fluence extended to low fluence if necessary$$

$$= 283 (f/10^{19})^{0.194}$$

$$\sigma_{0} = \text{ the standard deviation value from the } RT_{NDT}(0)$$

$$\sigma_{\Delta}$$
 = the standard deviation for the Guthrie mean curve
=' 24°F

11/13/82

Cu = weight percent

- Le-

Ni = weight percent nickel

and

f = fluence, n/cm² (E>1 MeV) (See discussion of fluence uncertainty in Appendix F.)

Note that the second of the two equations above does not include a standard deviation term for $\Delta RT_{NDT}(RG)$ since the Regulatory Guide term used is an upper bound equation.

This formulation is plotted in Figure 5-1 for three values of copper content and a nickel content of 1%.

5.5 Plant-Specific RT_{NDT} Valves

The results of applying the above methodology to all operating PWRs to calculate RT_{NDT} now, projected three effective-full-power-years into the future, are presented in Appendix P, Table P-1.

REFERENCES

- 5.1 Report of the Working Group on RT_{NDT}, Memo, M. Vagins to S. Hanauer, August 30, 1982.
- 5.2 Prediction of the Shift in the Brittle-Ductile Transition Temperature of LWR Pressure Vessel Materials, Edited by J. J. Koziol, Prepared by a Task Group of Subcommittee 6 of the Metal Properties Council, April 6, 1981 as a Report to ASTM Committee E10. To be published by ASTM.
- 5.3 LWR Pressure Vessel Irradiation Surveillance Dosimetry Quarterly Progress Report, Jan.-Mar. 1982, HEDL-TME 82-18. To be published.

4



• •

FIGURE 5-1: EXAMPLE OF NRC PRESCRIPTION FOR RT_{NDT} (FOR ASSUMED $RT_{NDT}(0) = 0^{\circ}F$)

P

a.

K

s

.

6. CONSIDERATION OF LOW FREQUENCY EVENTS

6.1 <u>Identification of Event Sequences with Significant PTS Risk</u>

15-

In order to determine the potential significance of challenges to reactor vessel integrity due to pressurized thermal shock from low frequency events, a systematic approach which identifies all relevant sequences of single and multiple failures from all pertinent initiating events is needed. Event tree techniques are an orderly approach for performing this quantification. Such a study using probabilistic methods and mostly existing PRA data bases was performed by Westinghouse for the Westinghouse Owners Group (WOG). The description and results of this study were submitted to the NRC by the WOG as Reference 6.1. The staff accepts the methodology used in the study to identify event sequences which contribute to risk from pressurized thermal shock and important portions of the discussion presented below have been adapted from Reference 6.1. Although there is agreement with WOG on the structure of the events that should be considered, the staff differs with the WOG in the resulting frequencies for many of the event sequences significant to PTS.

"The approach taken is to first identify the set of all the initiating transients or events which either by themselves or along with succeeding failures could lead to potential challenges to vessel integrity. The sequence of accompanying branching chains of events including component failures and their probabilities is logically traced out in the event trees. The output of the event tree is a set of end states and their frequencies. These end states can then be evaluated for potential challenges to the vessel from pressurized thermal shock. The sum of the frequencies of the end states which are potential challenges is the total frequency per reactor year of vessel integrity challenges summed over all types of initiating events" (Reference 6.1).

Initiating events used in this study, based on the WOG PRA, are presented in Table 6.1 and include those which either directly or through consequential

failures may lead to PTS. These events are the same as those used in recent risk studies. The first eight of these initiating events do not in themselves result in PTS, however, consequential events postulated as a result of these first eight initiators do result in transients with PTS. The events which could lead directly to a PTS challenge are small LOCAs, excessive feedwater, steamline rupture, and steam generator tube rupture. Consequential failures for these initiators can also enhance their seriousness as a cooldown challenge.

.11

٨

We believe that the WOG study has been sufficiently general and thorough to identify the event sequences of greatest significance to PTS risk. For the purposes of this study, we have adopted the significant event sequences that they have identified in modified form after staff review. All of the significant event sequences have been characterized by the staff with respect to frequency per reactor year, the temperature reciprocal time constant, β , and the final reactor coolant system temperature at the pressure vessel wall. The staff review and evaluation of these event sequences included important changes to the initiating and consequential event frequencies based on the staff's PRA studies including Reference 6.2. Some changes in the reciprocal time constant and the final reactor coolant system temperature at the pressure vessel wall were also based on what we believe to be better themal hydraulic analyses for some of the events considered. The bases for our selection of event frequencies, reciprocal time constants and final reactor coolant system temperatures and how they differ from the values presented by the WOG are further discussed in Appendix G. The event sequences determined by the WOG as reviewed and evaluated by the staff are further addressed in separate categories below.

We have chosen to apply the WOG PRA results to all three PWR vendors' plants for purposes of recommending the generic screening criteria. This is because: (1) the WOG results are the most complete available; (2) we believe that the PTS risk at all three vendors's plants is acceptably covered by meeting those recommended criteria; and (3) if there are significant differences in PTS risk at the different vendors' plants, these differences will be identified in the plant-specific analyses that are required when a plant "trips" the screening

criteria, and account will be taken of those differences before corrective actions resulting from those analyses are required.

· 6 (··

The design of the Combustion Engineering (CE) nuclear plants is sufficiently similar to Westinghouse plants to conclude that the event sequences identified below and based on the WOG study are applicable to CE plants. Babcock and Wilcox (B&W) plants which all include the Once Through Steam Generator (OTSG) and the Integrated Control System (ICS) are known to behave somewhat differently in responding to transients. One of the conclusions of a task force which studied and reported on the transient response of B&W designed reactors was that the B&W plants are more susceptible to overcooling transients (Reference 6.3). For the B&W OTSG the effective heat transfer area increase rapidly with overfeed since the water level which is normally at about two thirds the height of the tubes increases, and thereby substitutes boiling heat transfer surface for steam superheating surface causing more rapid cooling. Also, with overfeed in the OTSG the average secondary system water temperature is reduced more quickly because of the smaller OTSG water inventory causing more rapid cooling.

A compensating factor, with respect to severity of overcooling transients, is the presence of vent valves in the pressure vessel of B&W reactors. Although vent valves do not reduce the frequency of overcooling transients they do reduce minimum temperature obtained in many overcooling transients by allowing mixing of the warmer core exit water with the incoming cooler cold leg flow in the vessel downcomer.

B&W plants are now in the process of installing equipment to reduce sensitivity to overcooling. Modifications which should be completed in 1983 and 1984 include:

• Main feedwater overfill protection

٤

- Auxiliary feedwater (AFW) overfill protection
- Cavitating venturies in AFW lines (TMI-1)

Programmed AFW fill rate

Safety grade AFW system with automatic actuation

AFW flow indication in the control room

Evaluations by the Argonne National Laboratory and the NRC staff based on evaluations of B&W plant operating history and plant analysis with the proposed changes concluded that the likelihood of overcooling will be greatly reduced by the above changes.

. 15.

For the purposes of this report we are using the event sequences based on the WOG study as modified by the staff and discussed below as being applicable to the B&W plants. It would be desirable to verify this approach by a study of PTS event sequences for B&W plants specifically.

6.2 <u>Characterization of Specific Groups of Event Sequences Identified as</u> Contributors to PTS Risk

As a result of the above more general approach to the problem of identifying event sequences reported in Reference 6.1, we believe that the events of significance to the PTS issue have been identified as secondary (steam side) depressurization, small-break loss-of-coolant accidents, and steam generator tube ruptures (special case of small-break LOCA). In order to characterize individual event sequences within each of these groups, certain additional parameters have been identified which determine the significance of these sequences as a PTS challenge.

The level of decay heat present during an initiating event is an important parameter in the cooldown from a given transient. The level of decay heat is related principally to the operational history (full power operation, hot zero power, other) immediately preceding the transient. The frequencies of challenging event sequences are thus differentiated by the operational status of the plant. The time allowed for initiation of proper operator action is another parameter that is important in some sequences. This variable has been used as a parameter in the results which characterize certain sequences that are presented below.

6.2.1 Secondary (steam side) Depressurizations

4

This group of cooldown events which involves some type of opening of the steam system includes steamline rupture of all sizes, inadvertent safety relief valve open to atmosphere, inadvertent steam dump valve open to the condenser, reactor trip without turbine trip, or operator arror which results in any of these malfunctions. The transient is characterized by a rapid cooldown of the primary coolant system with shrinkage and consequential rapid depressurization until safety injection is actuated providing additional cooling and eventual primary repressurization. Natural circulation and, therefore, good mixing conditions are maintained in this transient for greater than 30 minutues unless low decay heat levels exist.

Parameters which are important with respect to severity of reactor coolant system boldowns are (1) plant operational status (decay heat level), (2) operator action time to isolate auxiliary feedwater flow, (3) break size, (4) reactor coolant pump operation, and (5) location of the depressurization opening with respect to the main steam isolation valves.

Main Steam Line Break (MSLB)

An MSLB with break area larger than 6 inches equivalent diameter results in a rapid cooldown of the primary system. The final temperature can be as low as around 200°F, depending on the plant operating status (decay heat level) and operator action to terminate auxiliary feedwater. The system will repressur ze as a result of safety injection and may reach a pressure in excess of 2000 prig depending on when operator action is taken to terminate safety injection. The MSLB results in a signal to close the main steam isolation valves (MSIVs) so that leaks downstream of the MSIV are bounded by leaks upstream of the MSIVs. The event frequency, reciprocal time constant, β , and final reactor coolant

- 61

system temperature for the parameters of initial power and time for operator action to isolate feedwater are presented in Table 6.2.

- 12

The staff results presented in Table 6.2 show that the frequency of this event for hot full-power conditions is significant to PTS risk whereas the frequency for hot full-power conditions determined by the WOG study is extremely low. The staff's final reactor coolant system temperature for this transient is, however, much higher than the WOG result based on what we believe to be more realistic thermal hydraulic analyses for this transient.

Small Steam Line Break (SSLB) or Stuck Open Steam Generator Safety/Relief Valve

The SSLB or stuck open SG safety/relief value can result in an overcooling transient similar to the MSLB but of longer duration due to the smaller break size. This event has a much higher frequency than the MSLB. The event frequency, reciprocal time constant, β , and final reactor coolant system temperature for the parameters of initial power and time for operator action to isolate feedwater are presented in Table 6.3. The staff's results indicate a somewhat higher frequency for this event than the WOG study.

6.2.2 Small-Break Loss-of-Coolant Accident (SBLOCA)

The cooldown transient from an SBLOCA of the reactor coolant system includes reactor coolant pump seals, primary power-operated relief valve or safety valve failure or leakage as well as actual piping breaks of various sizes in hot or cold legs. For breaks less than a critical break size of about two inches (equivalent diameter, i.e., equal in area to a circle of this diameter), natural circulation is maintained and mixing occurs and the resulting cooldown rates are not expected to exceed Appendix G limits of less than 100°F per hour. Both the reactor coolant pump seal leak (break equivalent to 0.5 inches) and stuck-oper power-operated relief valve (break equivalent to 1.4 inches) are included in that category.

Break sizes greater than two inches up to six inches are also included as SBLOCAs. For these breaks, the safety injection flow is less than the break flow, resulting in a net mass loss from the primary system. Loop flow (natural

circulation) can be lost for this range of breaks, resulting in a rapid cooldown due to the cold safety injection flcw. The exact break size where loss of flow ocurs is dependent on the safety injection flow rate (and makeup flowrate), the break location, the decay heat level, and the SG (heat sink) performance. Because of the stagnation of flow, mixing of the safety injection water is poor and rapid cooldown of the vessel could result.

- 1. -

4

.

We have reviewed the frequency of events that may result in stagnated loop conditions such as SBLOCAs in the 2 to 6 inch equivalent diameter range. There are several small diameter pipes in the range of 2 to 4 inches connected to bypass lines, pressurizer spray lines, power-operated relief valve lines, and the main primary system piping. These include charging and letdown lines, RTD safety injection lines. SBLOCA events in the 2- to 6-inch range are dominated by non-isolatable breaks and, therefore, operator action is not a major parameter. The event frequency, reciprocal time constant, β , and final reactor coolant system temperature are presented in Table 6.4. The final temperature selected is sensitive to the mixing model assumed, control volumes, etc. A discussion is presented in Appendix G to this report (G.2.3, NRC Staff Characterization of SBLOCA) and in Table G.5 and References G.8 through G.11.

As discussed above, these LOCAs can be differentiated by breaks smaller than about 2 inches where loop circulation is maintained (and good mixing of the cold safety injection water is therefore achieved) and breaks larger than about 2 inches where loop circulation is lost (and poor mixing of safety injection water results). The WOG judged LOCAs with effective diameters greater than 2 inches to have a negligible probability of causing vessel failure, so they only included LOCAs with breaks less than two inches. We agree that breaks in the size range less than two inches have small β 's and appear similar to slightly accelerated shutdown transients where the operator can be expected to control the pressure.

The significance of breaks in the 2- to 6-inch range to PTS risk has been separately analyzed. Fracture mechanics analyses performed with a more exact representation of this cooldown transient show that the PTS risk from this transient is less than could be anticipated and consistent with the screening criteria proposed (see Section 8).

We have also considered other events that could lead to extended HPSI operation with stagnated loop(s) conditions and subsequently repressurize. These conditions are similar to those for a SBLOCA in the 2- to 6-inch range except that they may repressurize to >2000 psi. The potential sequences discussed in Appendix G appear to be dominated by a stuck-open safety valve that subsequently recloses. The event frequency, reciprocal time constant, β , and final reactor coolant system temperature are presented in Table 6.4.

6.2.3 Steam Generator Tube Rupture

The response of the reactor coolant system to a variety of steam generator tube failure events, up to the complete severance of a single tube, has been analyzed in Reference 6-1. These analyses simulate automatic protection systems, such as reactor trip and emergency core cooling systems, as well as operator actions. A steam generator tube failure should not result in a rapid cooldown of the primary system or excessively high reactor coolant system pressures if current plant operating procedures are used. In general, natural circulation should develop in all primary loops and mix with incoming safety injection flow to preclude local temperature depressions if RCPs are stopped. However, the subsequent operator actions to terminate primary-to-secondary leakage may rapidly cool the reactor coolant system for short periods and may stagnate the faulted loop. In that case, local temperature depressions due to continued safety injection flow may occur. The period of this temperature depressions is expected to be short and should not represent a significant PTS challenge to vessel integrity. The event frequency, reciprocal time constant, β , and final reactor coolant system temperature are presented in Table 6.5. The frequency for this event estimated by the staff is significantly greater than that determined by the WOG.

6.3 Low Frequency Event Sequences Contributing to PTS Risk

Tables 6.2, 6.3, 6.4, and 6.5 summarize the groups of postulated event sequences that appear to be significant PTS initiators in terms of their estimated expected frequency, and the parameters for a stylized transient (T_f , β and P). Calculations also have been made of typical temperature and pressure behavior for each class of events. In principle, each of the sequences could be used to perform deterministic fracture mechanics calculations for a range of vessel RT_{NDT}

6-8

- 7: -

values to determine the limiting value, RT_c , necessary to prevent crack initiation for each type of event, as was done for actually experienced events in Section 3.5 above. However, such calculations have not been made. Alternatively, the data of Tables 6.2, 6.3, 6.4, and 6.5 can be used to construct a cumulative frequency versus T_f distribution similar to that done for experienced events in Figure 2-14. This distribution is shown in Figure 6-1. In Section 3.5 above, it is shown that based on the deterministic fracture mechanics parametric studies, for relatively fast cooldown event ($\beta = 0.15 \text{ min}^{-1}$) with final temperatures in the 250-300°F range and high system pressures (~2300 psig), crack initiation (in longitudinal welds) is not predicted if T_f is about 5-10°F higher than RT_{NDT} . Thus, the distribution curve of Figure 6-1 suggests that vessels with an RT_{NDT} of 270°F (the suggested screening criterion discussed in Section 4) would not be expected to experience longitudinal crack extension for events with frequencies greater than about 6 x 10-³ per reactor-year. This conclusion is similar to that obtained in Section 4 considering actual experienced events.

. 11

However, the frequency distribution of Figure 6-1 extends to low frequency events with low values of T_f . This low frequency "tail" on the distribution indicates that there are postulated events with estimated frequencies as high as 10^{-4} per reactor-year for which the final temperature is substantially below 270°F that must be considered. This is discussed in Section 8.

The discussion above is subject to the same large uncertainties as are described in Section 4 above. To gain additional insights into the conservatisms in the deterministic fracture mechanics treatment and to gain some notion of the risk of vessel failure considering low probability events, the data of Tables 6.2, 6.3, 6.4, and 6.5 are used in combination with a probabilistic treatment of fracture mechanics described in Section 7. The results are discussed in Section 8.

REFERENCES - SECTION 6

6.1 "Summary of Evaluations Related to Reactor Vessel Integrity Performed for the Westinghouse Owners Group," Westinghouse Electric Corporation, Nuclear Technology Division, May 1982.

6.2 "Reactor Safety Study, An Assessment of Accident Risks in U.S. Commercial Nuclear Power Plants," WASH-1400 (NUREG-75/014) October 1975.

- 7+ -

•

4

4

6.3 "Transient Response of Babcock and Wilcox Designed Reactors" NUREG-0667, May 1980.

Table 6.1 Initiating events

.

-15-

4.

EVEN	Ţ	FREQUENCY PER REACTOR-YEAR
1.	Loss of Main Feedwater (LOFW)	3.41
2.	Closure of One Main Steam Isolation	6.00×10^{-1}
	Valve (MSIV)	
3.	Loss of Primary Flow (LOPF)	3.21×10^{-1}
4.	Core Power Increase (POWIN)	4.77 x 10- ²
5.	Turbine Trip (TT)	4.00
6.	Spurious Safety Injection	
	Activation (SSI)	1.59 x 10-1
7.	Reactor Trip (RT)	4.11
8.	Turbine Trip Due to Loss of Offsite	1.01 x 10- ³
	Power (TT/Loop)	
9.	Steam Generator Tube Rupture (SGTR)	3.92×10^{-2}
10.	Small LOCA, <1.5-inch diameter (LOCA-1)	9.07 x 10- ³
11.	Small LOCA, >1.5-inch diameter (LOCA-2)	6.11 x 10-4
12.	Large LOCA, >6-inch diameter (LOCA-3)	3.88×10^{-4}
13.	Excessive Main Feedwater (EX FW)	2.50×10^{-1}
14.	Steamline Rupture Inside Containment	3.88 x 10-4
	(STM BRK In)	
15.	Steam Rupture Outside Containment (STM BRK OUT)	3.87 x 10- ²

.

•

6-11

.

Plant Status		Hot Full Power			Hot Zer	o Power		<u>ny y na katang na ka</u> ta
Operator isolates auxiliary feedwater min.	0-5	5-10	10-30	30-60	0-5	5-10	10-30	30-60
Event frequency per reactor-year	8x10- ⁵	6x10- ⁵	1.5x10- ⁵	3×10-7	8.5x10- ⁶	6.8x10- ⁶	1.6x10-6	3x10- ⁸
Reciprocal time constant, β min-1	0.4	0.2	0.09	0.09	0.4	0.2	0.2	0.2
Final reactor coolant system temperature at vessel wall, °F	450	390	300	250	212	212	210	190

Table 6.2 Event parameters for the main steam line break (MSLB)

•

₽

- 11

Event frequency (MSLB) = $\sim 1.7 \times 10^{-4}$ per reactor-year

Plant Status	Hot Full Power				Hot Zero Power					
Operator isolates auxiliary feedwater min.	0-5	5-10	10-20	20-30	30-60	0-5	5-10	10-20	20-30	30-60
Event frequency per reactor-year	4.5x10- ³	3.6×10- ³	8.10-4	6.3×10- ⁵	1.8x10-5	4.5×10-4	3.6x10-4	8x10- ⁵	6.3x10-	⁶ 1.8x10- ⁶
Reciprocal time constant, β min-1	0.4	0.2	0.1	0.06	0.06	0.4	0.2	0.1	0.06	0.06
Final reactor coolant system temperature at vessel wall, °F	385	320	250	220	200	375	310	235	200	175

Table 6.3 Event parameters for the small steam line break (SSLB)

Event frequency (SSLB) = $\sim 10^{-2}$ per reactor-year

77-

5

٠

1



-!

4

FIGURE 6-1

Table 6.4 Event parameters for the small-break loss-of-coolant accident (SBLOCA); and extended HPSI operation with a stagnated loop and high pressure extended HPSI operation

.7...

æ

4

	SBLOCA	Operation
Event frequency per reactor-year	1.5x10-3	10-4
Reciprocal time constant, β , min ⁻¹	0.05	0.05
Final reactor coolant system temperature at vessel wall, ^o F	125	125

Table 6.5 Event parameters for steam gener- or tube rupture

Without steam line break or stuck-open SRV	
Event frequency per reactor-year	5x10-3
Reciprocal time constant, β, min-1	0.04
Final reactor coolant system	
temperature at vessel wall, °F	200

With steam line break or stuck-open SRV

Event frequency per reactor-year	Stuck open SRV outside containment 2x10-4	SLB inside containment* 1×10- ⁵
Reciprocal time constant, β min ⁻¹	0.04	0.04
Final reactor coolant system	170	170
temperature at vessel wall, °F		

*Excluded from these evaluations

15

7. PROBABILISTIC TREATMENT OF FRACTURE MECHANICS

7.1 Introduction

The deterministic fracture mechanics analyses riscussed in Section 3 assume specific values for all the input parameters necessary to predict crack initiation, growth and/or arrest. However, many of these parameters are not known precisely. In order to quantitatively analyze the effect of a large number of uncertainties, a probabilistic approach can be taken to estimate the failure probability of a reactor pressure vessel. A Vessel Integrity Simulation Analysis (VISA) code was developed to gain insight into reactor pressure vessel failure probability due to pressurized thermal shock. Appendix H discussed in detail the probabilitistic fracture mechanics analysis and the VISA code. A brief description of the VISA code is present in Section 7.2.

~S .

7.2 Description of VISA Code

The VISA code consists basically of two parts. The first is a deterministic fracture mechanics analysis for a specified pressure/temperature transient. This analysis is similar to that discussed in detail in Appendix D (D.1) and includes heat transfer, thermal and pressure stress, and applied stress intensity value calculations for a range of crack depths. The second part uses Monte Carlo techniques to assess failure probability based on a very large number of deterministic calculations in which the input parameters are varied.

Certain parameters are treated as random variables, and their values are sampled from a statistical distribution defined as an input to the program. In each calculation, values of the random variables (crack depths, copper content, initial mean value of RT_{NDT} , fluence, critical stress intensity factor, K_{Ic} , and stress intensity factor for crack arrest, K_{Ia}) are selected from the specified probability distributions, and deterministic calculations are made using these values. Each calculation results in one of three outcomes: (1) no crack initiation, (2) crack initiation followed by arrest, or (3) pressure vessel failure.

For each iteration of the simulation, values of fluence, fiaw size, and copper content are selected from their respective distributions. The means value of RT_{NDT} at the inner wall is calculated as a function of fluence and copper content. With these values fixed for the iteration, the code steps through the time history of the transient. For each time step, the stress intensity at the crack depth is taken from the deterministic portion of the code. A value of K_{IC} is simulated to determine fracture initiation. If initiation does not occur, the simulation moves to the next time step. If initiation does occur, the crack is extended 1/4 in., and the crack arrest toughness (K_{Ia}) is simulated. If arrest occurs, the simulation moves to the next time step; if not, the crack is extended another 1/4 in. and a new value of K_{Ia} is simulated. This process is continued until either the vessel fails or the duration of the transient is reached. Each pass through the simulation loop represents a single computer calculation conducted to determine if RPV failure will occur. Up to a million passes through this loop can be made. The code keeps track of the number of crack initiations and RPV failures. The probabilities of crack initiation and RPV failure then are estimated by dividing these values by the total number of trails. Thus, the VISA code actually performs millions of deterministic calculations with each set of calculations based on a different set of values selected from the appropriate statistical distributions for the significant variables. This is the

operating reactor pressure vessels to the pressurized thermal shock transient of interest and then inferring the failure probability based on the number of observed failures.

7.3 Probabilistic Fracture Mechanics Sensitivity Studies

Section 3 and Appendix D of this report discuss the sensitivity of crack initiation and vessel failure to the various PTS parameters. This section

PTS REPORT SEC 7

discusses the same sensitivities based on probabilistic fracture mechanics. Results are portrayed in Figures 7-1,7-2, 7-3 and 7-4 for the stylized thermal transient discussed in Section 2 above:

$$T_w = T_f + (To - T_f)e^{-\beta \hat{v}}$$

Figure 7-1 illustrates the sensitivity of the conditional vessel failure probability to T_f minus mean value of RT_{NDT} , for three values of β and a pressure of 1000 psig. A total of 225 cases were run, using three values of T_f -- 150, 225 and 300°F. It is seen that the relative risk is low for mean value of R_{IDT}^{T} less than T_f , but if cooldown drops temperature below the vessel mean value of R_{IDT}^{T} , then the risk rises quite rapidly.

Figure 7-2 illustrates the sensitivity of conditional failure probability to pressure for several values of T_f minus mean value of RT_{NDT} and a β of 0.15 reciprocal minutes.

Figure 7-3 illustrates the sensitivity to the decay parameter, β , with several values of T_f minus mean value of a RT_{NDT} and the other parameters held constant.

Figure7-4 illustrates the sensitivity of two postulated transients to the heat transfer coefficient used. For relatively low heat transfer coefficients, i.e., at low flow conditions, the risk is quite sensitive to the value (or correlation) used.

Note that Figures 7-1 through 7-4 are given in terms of failure probability per weld. Since there are six axial welds in the beltline region of the core, these values times six will yield the vessel conditional failure probability.

Appendix H includes more information regarding the sensitivity of relative failure probability to parametric assumptions. Although these studies assumed somewhat different input, assumptions regarding the relation of mean value of RT_{NDT} to fluence and fluence attenuation through the wall than were used for the deterministic fracture mechanics studies (Section 3 and Appendix D), the same trends are found.

These probabilistic sensitivity studies do not include the effect of cladding stresses. Based on the conclusions stated in Section 3, it is estimated that inclusion of the cladding stresses would shift the curves of Figures 7-1 approximately 10°F to the right, thus increasing the risk about a factor of 2 or 3 for that assumed transient.

Because the probabilistic fracture mechanics studies were conducted for only a limited range of parameters, the results should not be extended beyond these ranges. For instance, if T_f were only a few tens of degrees below 550°F, the thermal shock to the vessel would be significantly less severe than say for a cooldown to 200°F or lower. Thus, in terms of probability verus T_f minus mean value of RT_{NDT} , the results are expected to be considerably different.

The technology regarding probabilitistic fracture mechanics are related to PTS scenarios has evolved only during the past few years and perhaps is still some way from reaching maturity. It is, however, believed to be a useful tool. The NRC plans to develop the technology further, and the industry is encouraged to do the same. Future work is expected to include consideration of warm prestressing effects for a variety of postulated transients, the effect of cladding and perhaps other crack shapes.

The discussion in Appendix H suggests that the results which have been presented are most appropriately used in a relative sense for identifying significant variables and variable interactions. Because of the uncertainties associated with the calculated failure probabilities, use of the results in an absolute sense to establish an RT_{NDT} screening limit would be inappropriate. Nonetheless, there does exist a tendency to view the results in an absolute sense when evaluating proposed regulatory requirements. Furthermore, there is a desire to view the results in an absolute sense when performing a probabilistic risk assessment. Utilization of the results in these manners is useful in evaluating a regulatory position, but the limitations of the analysis as discussed in Appendix H must be kept in mind.



- . .






FIGURE 7-4: CONDITIONAL FAILURE PROBABILITY AS FUNCTION OF HEAT TRANSFER COEFFICIENT, h*,

-88-

8 PROBABILITY OF VESSEL FAILURE

8.1 Introduction

This Section summarizes a probabilistic study of reactor pressure vessel (RPV) failure as a result of pressurized thermal shock (PTS). The calculational method combines the frequencies of overcooling transient sequences (Sections 2 and 6) with the probabilistic treatment of RPV failure (Section 7). The results are expressed in terms of the probability, per reactor-year, of RPV failure due to PTS. Some risk considerations are also discussed and also the relationship of PTS to the current regulations.

- - 1

8.2 Methodology

The basic approach is essentially a combination of (1) a probabilistic analysis of overcooling transients, plus (2) a probabilistic analysis of the consequences of such transients to the RPV, and the probability of RPV failure, given the transients.

In order for this procedure to be valid, the transient sequences must be separated, and analyzed in groups with similar properties. The course and severity of each transient group can then be used as the input transient for analysis of RPV behavior. In the work reported here, the transient sequences were obtained from calculations furnished by the Westinghouse Owners Group (Ref. 8.1), and also from transient analyses based on the WOG analysis but revised by the NR(' staff as described in Section 6 and Appendix G.

The transient groups were derived from consideration of the various possible sequences following each of the initiating events--excess feedwater, smallbreak LOCA-etc.--given in Appendix G. The analyses of the frequencies and courses of the different sequences are also reviewed in Appendix G. The VISA code, in its present state of development, can accept only stylized input transients characterized by T_f , β , and P or T and P described by simple polynomial functions of time. Therefore, each transient group was stylized. Cumulative frequency distributions of the T_f values used in this study are given in Fig. 8-1.

10

For each transient group, the values of T_f , β , and P were used, with the VISA code, to establish a probability of RPV failure, given the occurrence of a transient with the specified characteristics. Multiplication of this conditional RPV failure probability by the frequency of occurrence of the transients comprising the group then gives the frequency of RPV failure caused by this group of transients.

The sum of these failure frequencies gives the RPV failure frequency (per reactor-year) as caused by the ensemble of transients of all the groups considered.

8.3 Accuracy and Completeness

In order for the RPV failure frequency so calculated to be correct, the ensemble of transients must be complete. That is, all transaient sequences capable of inducing RPV failure due to PTS must be included.

The WOG analysis included consideration of several hundred candidate sequences, not all of which turned out to be PTS precursers. The array of initiating events and event sequences is given in Ref. 8.1 and summarized in Appendix G. Variations in reactor power level, break size (for LOCA and steamline break sequences), and operator actions were included. The staff review concentrated on the transient groups shown in the WOG analysis to be dominant, but also considered other candidates not significant in the WOG analysis; see again Appendix G. In all, some hundreds of possible sequences were reviewed by WOG, staff, or both.

Like all probabilistic analyses based on event sequences, the probabilistic PTS analysis presented here cannot be proved complete. The differences between the

11/13/82

PTS REPORT SEC 8

8-2



FIGURE 8-1

1



. 14

FIGURE 8-2

WOG and staff analyses show that more work can and should be done to investigate additional candidate sequences and to validate some of the approximate sequence analyses. However, the work done to date suggests that the principal contributors have probably been identified as well as can be done in approximate, generic analyses. Improved approaches to completeness should be sought in connection with the plant-specific analyses we recommended for plants soon to exceed the screening criterion (Sections 4 and 9). Some sequences not included are discussed in Section 2.6 of Appendix G.

Over the past few months the staff, with some consultation with the Westinghouse Owners Group, has continued to work on improving these calculations. Our experience has been that the calculated results - Figure 8-3 - have changed several times in a few months. The causes of such changes have included the following:

- New systems response information, such as downcomer mixing experimental results and models;
- New insights regarding additional sequences to be included, such as the high-pressure extended HPI group;
- Addition of phenomena to the model, such as warm prestressing.

The changes in results have been well within the quoted overall uncertainty band of the calculational methods at their present state of development. However, the estimated risk for a given vessel RT_{NDT} has changed up to a factor of 5; the RT_{NDT} for a given risk level has changed by up to 30°F.

The staff interprets these changes to show the immaturity of the probabilist c calculations at their present state of development.

Completeness aside, the accuracy from both the transient sequences and the vessel calculations, is subject to substantial uncertainties. In particular the probabilistic treatment of fracture mechanics (Section 7, Appendix H) is still under development. Both the methodology and the probability distribution functions used as input information are sources of variation in the results. Detailed study of these variations has not yet been accomplished. Moreover,

4

8-4



Figure 8-3

much more extensive sensitivity studies are planned. Therefore, the numerical results given in this Section must be taken for what they are worth. Rather than close estimates of absolute RPV failure probability, the calculated values should be used for insight into trends and sensitivities. The values of the calculated probabilities of failure for a given set of nominal conditions is believed by the researchers (Appendix H) to be uncertain by plus or minus at least two orders of magnitude. Also, the steepness of the curves (Appendix H) shows a high sensitivity of the result (calculated RPV failure probability) to variations in the values of T_f , β and P assigned to the transient group. The calculation of these quantities is approximate, even for a well defined event sequence. The lesson from transients don't look like exponentially decaying temperatures with constant pressures. Thus, another source of uncertainty is introduced by the stylized transients necessarily used in this calculation, at the present state of the art.

11

These analyses -- both WOG and NRC staff -- are basically for Westinghouse plants. Both probabilistics and severities were calculated for such plants. The use of these analyses to represent B&W and CE plants has not been investigated by the NRC or the industry. The beginnings of a Combustion Engineering Owners Group (CEOG) analysis were described in the June 23, 1982, meeting, but only frequencies have so far been reported, with on evaluation of severities. No comparable study for B&W plants has been reported.

The results of the probabilistic calculations are, therefore, applicable directly only to Westinghouse plants.

The large uncertainties in probabilistic PTS evaluations at the present time have led the staff to use them to estimate the level of safety rather than attempt to derive licensing requirements directly from the probabilistic results.

8.4 Results

With due consideration of the uncertainties discussed just above, we present the results of the probabilistic PTS calculations in Figs. 8-2 and 8-3. The details are given in Appendices G and H.

Figures 8-2 and 8-3 show, as a function of RPV mean value of RT_{NDT} , the expected frequency of RPV failure due to PTS. The abcissas are the reference temperatures, mean value of RT_{NDT} , at the inner surface of a RPV having the mean values of RT_{NDTo} , neutron fluence, and copper content of the probability distribution functions used for these parameters. The ordinate is the failure frequency of the RPV so characterized, per reactor-year, owing to the PTS transient subclasses (LOCA, SLB, etc.) as labelled, and the total RPV failure frequency due to PTS. New vessels start at the left side of these diagrams, with very low mean value of RT_{NDT} and negligibly small PTS probability. As the vessels are irradiated, their characteristics move to the right, and an increasing number of increasingly probable overcooling transients have increasingly higher probability of inducing RPV failure.

. j(-

Figure 8-2 gives the results for the WOG distribution of transients; Figure 8-3, the NRC staff distribution.

The steepness of the curves in Figures 8-2 and 8-3 shows a high sensitivity of RPV failure probability to the value of RT_{NDT} (as defined for these curves). A change in RT_{NDT} as small as 20-30°F changes the calculated probability by a factor of 10, on some of these curves. Yet we know neither the actual value of RT_{NDT} for a given RPV, nor the severity of a given transient, to within this order of accuracy. This is another way of restating the substantial uncertainties in the present state-of-the-art of making analyses of this kind. For this reason, the NRC staff recommends that the PTS criteria--screening or otherwise--should not be determined by where these curves cross some acceptable value of risk. Rather, the probabilistic curves are used to estimate the margin of safety for vessel approaching the screening criterion.

8.5 Relationship to Safety Goal

In February 1982, the Commission published for comment a "Proposed Policy Statement on Safety Goals for Nuclear Power Plants" (Ref. 8.2). Although the Sabety Goals guidelines have not been adopted (at least not yet), it is instructive to compare the proposed PTS requirements to the guidelines.

.

<u>Core Melt</u>. - The core melt Safety Goal guideline states, "The likelihood of a nuclear reactor accident that results in a large-scale core melt should normally be less than one in 10,000 per year of reactor operation." This suggests that the core-melt frequency ascribable to one sequence, for example PTS, should not exceed approximately 10^{-5} per reactor-year.

411.

Because of the unusually large uncertainty in the risk estimation for PTS, compared to other sequences, a value of less than 10^{-5} might well be assigned for a safety goal for PTS. We have not done this in the discussion in this section, but have used 10^{-5} . The reader should keep in mind that the risk numbers of PTS given in the following discussion are highly uncertain.

We have no technical analysis of the course and consequences of a PTS sequence that involves RPV failure. Determination of the RPV failure mode (better, estimation of the probabilities of the various failure modes) has not been done and is dependent on the details of the scenario. Moreover, the outcome would likely be dependent also on the plant design details. In particular, ice condenser containments would be expected to have different failure models, with different probabilities, than large dry containments.

The breach in the RPV would be a LOCA, which might or might not prevent ECCS effectiveness. A large through-wall crack peak the midplane level of the core would most probably lead to core melt. Axial cracks and most circumferential cracks would not likely lead to early containment failure; the massive concrete shielding would intercept missiles and the containment could stand the temperature and pressure. (Again, ice condensers have not been evaluated.) Whether a complete circumferential failure (which seems low in probability) would 'ead to large RPV (and core) motions is not well known.

The result of such approximate and intuitive analysis is that not all PTS failure events lead to core melt, but the fraction that do has not been analyzed quantitatively.

<u>Public Risk</u>. - the Draft Policy Statement includes quantitative guidelines for risk to individual members of the public, and for society at large, from

11/13/82

4

PTS REPORT SEC 8

reactor accidents. For analyzing how PTS events contribute significantly to the risk to the public, the following logic applies:

1 14

- 1. PTS event sequences lending to RPV failure have overall frequency F per reactor-year. Figures 8-2 and 8-3 provide a very approximate estimate of F. A plant evaluated (as described in Section 5 or 9 and Appendix E) to be at the 270°F screening criterion is likely to have a true RT_{NDT} of 150-270°F (two sigma 7 \cong 60°F). For the mean of 210°F, F \cong 6 x 10-6 per reactor-year on the NRC curve (Figure 8-3), and much smaller on the WOG curve (Figure 8-2).
- 2. A fraction X<1 of RPV failure sequences leads to core melt, giving an expected value of XF core melts per reactor-year.
- 3. A fraction Y of failures leading to core melt leads to significantly early radioactive releases, so the expected value of the frequency of significant early releases due to PTS is XYF, which is therefore the expected value of the frequency of events involving non-zero early deaths due to PTS.
- 4. The risk of early deaths to the average individual within one mile of the site is given by XYFD, where the factor D includes the effects of dispersion and wind direction.
- 5. To show PTS risk to be lower than 10% of the safety goal guidelines would involve showing

XF< 10⁻⁵ per reactor-year

and

 5×10^{-8} XYFD<u><</u> **m** per reactor-year

The limiting value for XYFD is approximate, for an averge site.

.

We have only approximate values for F, and no quantitative evaluation to give values for X and Y. It is, however, possible to calculate backwards how small these quantities must be to stay within the safety goal guidelines.

11

D is approximately equal to 5×10^{-2} .

4

If the core-melt guideline is just barely met, XF equals 10^{-5} . For such a plant, Y would have to be 10-1 or less to meet the risk guideline. For a lower value of XF, a proportionately higher value of Y would be low enough to meet the safety goal. For F = 6 x 10^{-6} (see Item 1, above) and X set arbitrarily at the unrealistically high value of 1, Y would have to be 2 x 10^{-1} or less to meet the risk guideline.

While we have no analysis of Y for PTS sequences, PTS involves overcooling of the reactor vessel and may well result in a high probability of cooling the containment, thus leading to a low value for Y.

Aside from the containment failure modes given in PRAs for non-PTS events, which apply to PTS events also with some probability, an additional prompt containment failure mode peculiar to PTS arises from a postulated circumferential crack. If the vessel breaks all the way around and separates, the top half will be accelerated upwards. Available analyses are not definitive whether the upward force will break the restraining structures and pipes and allow the loose half to become a jet-propelled missile, or whether such a missile would cause early containment failure. The answer may depend on the details of the plant design. If this mode could dominate the risk, then the plant-specific analyses described in Section 9 should include its evaluation.

<u>ALARA</u>. - The Draft Policy Statement gives a cost-benefit guideline for decis onmaking of \$1000 per man-rem averted.

For scenarios involving core melt without significant releases, the core-melt. guideline will govern and ALARA is not a consideration.

For early containment failure scenarios at an average site, as much as 50×10^6 man-rem might be involved, at a frequency of XYF. The expected value of the

exposure would, therefore, be 50 x 10^6 XYF. For XYF $\leq 1 \times 10^{-6}$, the expected value would be less than 50 man-rem, and the ALARA guideline would not be a consideration for these sequences, either. For a highly populated site, the values would be several times higher, but the conclusion would not be changed.

14 4 1

In summary, the results of the approximate probabilistic PTS analysis reported here suggests that the proposed screening criterion is in satisfactory conformance with the Draft Policy Statement on Safety Goals for values of X and Y that seem achievable, even considering the large uncertainties. Further analysis would be appropriate to improve the technical basis of this conclusion.

8.6 <u>Relationship to Licensing Criteria</u>

Several of the Commission's regulations have applicability to the issue of pressurized thermal shock.

Appendix G of 10 CFR Part 50 sets forth requirements for the fracture toughness of reactor pressure vessels "during system hydrostatic tests and any condition of normal operation, including anticipated operational occurrences." In the definitions Section of Appendix A to 10 CFR Part 50, "anticipated operational occurrences" are defined as "those conditions of normal operation which are expected to occur one or more times during the life of the nuclear power unit." Since the usual design life of nuclear plants is forty years, events with an expected frequency of greater than 2.5 x 10^{-2} per reactor-year would be considered "anticipated operational occurrences." On the basis of experienced overcooling events (see Section 4) and low probability event sequences (see Section 6), PTS events with a severity of concern for reactor vessels with a value of RT_{NDT} less than 270°F have an expected frequency of less than 10^{-2} per reactor-year. Thus, the applicability of Appendix G of 10 CFR Part 50 to severe PTS events is questionsable, for pressure vessels that do not exceed the proposed screening criteria. However, the uncertainties are large.

Section IV.A.2.a of Appendix G, 10 CFR Part 50 states that:

Calculated stress intensity factors shall be lower than the reference stress intensity factors by the margins specified in the ASME Code

......

Appendix G, "Protection Against Non-Ductile Failure." The calculational procedures shall comply with the procedures specified in the ASME Code Appendix G, but additional and alternative procedures may be used if the Commission determines that they provide equivalent margins of safety against fracture, making appropriate allowance for all uncertainties in the data and analyses.

. 111

The ASME Code Appendix G procedure used for determining "an adequate margin of protection" includes the postulation of a reference semi-elliptical surface flaw having a depth of 1/4 of the section thickness with a length six times its depth. The Appendix G procedure also requires the use of the ASME K_{IR} curve for crack initiation which is approximately 70°F more conservative in the range of interest than the K_{IC} curve used in analyzing PTS. In addition, the stress intensity factor due to pressure is increased by a factor of two. Because pressure stresses dominate for hydro testing and normal startup and shutdown situations, this factor provides additional margin for normal operation, beyond the conservatism provided by the 1/4 T assumed flaw and the use of the K_{IR} curve.

For severe cooldown transients of interest to PTS, however, thermal stresses near the inner vessel surface are dominant. The material toughness is also lower near the surface than deeper into wall because of the lower temperatures near the surface. Hence, consideration must be given to relatively shallow flaws. Thus, procedures different from those of Appendix G are necessary to provide an adequate margin of protection for PTS events. Such procedures will be developed as part of the resolution of the PTS issue. Consideration will need to be given as to whether Appendix G to 10 CFR 50 should be amended or supplemented by separate guidance.

Another potential regulation interface is 10 CFR 50.46 and 10 CFR 50, Appendix K. While the trust of these regulations is to cooling effectiveness, paragraph (b)(5) of 10 CFR 50.46 requires,

After any calculated successful initial operation of the ECCS, the calculated core temperature shall be maintained at an acceptably low value and decay heat shall be removed for the extended period of time required by the long-lived radioactivity remaining in the core.

If "successful initial operation" involves a PTS scenario, as can happen for 2 to 6 inch breaks (Sections 2 and 6, Appendix G), then "long-term cooling" can be jeopardized.

- 141

This scenario is discussed in Section 6. This sequence was the subject of detailed examination, as discussed in Section 6.2.2. For this sequence, detailed calculations of system response were used, rather than the stylized T_f , β , and P. The WOG calculations, which we accepted, include fluid mixing in the cold leg as predicted from experimental results, heat input from hot piping walls, benefit of warm pre-stressing is assumed, and an assumed temperature of 60°F for the injected coolant. These calculations used in the NRC assumptions for crack arrest, but should be corrected by -10° F to allow for cladding effect (see Section 3). The WOG result is a predicted allowable RT_{NDT} of approximately 270°F, consistent with the proposed screening criterion. Recently obtained information has led the NRC staff to accept a less rapid cooldown for this event. This would indicate a higher, less restricted RT_{NDT} would be acceptable, based on this event.

We conclude that a small break LOCA in a vessel within the proposed screening criterion would cause an acceptably low probability of vessel failure, so 10 CFR 50.46 is not infringed by the proposed requirements.

Another regulation related to PTS is General Design Criterion 31 Appendix A to 10 CFR 50, which states:

Criterion 31 - Fracture prevention of reactor coolant pressure boundary. The reactor coolant pressure boundary shall be designed with sufficient margin to assure that when stressed under operating, maintainance, testing, and postulated accident conditions (1) the boundary behaves in a nonbrittle manner and (2) the probability of rapidly propagating fracture is minimized. The design shall reflect consideration of service temperatures and other conditions of the boundary material under operating maintenance, testing, and postulated accident conditions and the uncertainties in determining (1) material properties, (2) the effects of irradiation on material properties, (3) residual, steady state and transient stresses, and (4) size of flaws.

11/13/82

.

PTS REPORT SEC 8

The proposed new requirements relating to PTS provide additional guidance in the implementation of Criterion 31, and it does not appear necessary to modify this regulation.

Section 100.11 of 10 CFR sets forth dose guidelines for assessing the acceptability of Exclusion Area and Low Population Zone boundaries, based upon fission product releases from hypothesized accidents considered credible. In the past, pressure vessel failures have not been considered in the assessments required by 10 CFR 100. The staff believes that the considerations of PTS risk presented above support continuation of this position, but further consideration of the relationship of PTS to 10 CFR 100 is warranted.

8.7 Conservatisms and Non-Conservatism

The calculations summarized in Sections 3, 5, 7 and 8 and described in detail in the appendices and references, contain uncertainties of various sorts. The following paragraphs briefly summarize the most significant sources of uncertainty.

<u>Operating Experience</u> - The three most severe events took place in B&W plants. We have neglected, for lack of sufficient data to do otherwise, plant design differences in evaluating the experience. We have also neglected all the action taken since TMI, Rancho Seco, and Crystal River to improve design and operations and thereby make these transient sequences less likely in the future. These are substantiated conservatisms in the inference from operating experience.

The temperatures used to characterize operating experience were measured in cold legs. The fluid in the downcomer could have been warmer (from mixing) (r colder (from stratification) than the measurements.

<u>Operation Actions</u> - The analyses include the probability of the operating staff failing or delaying performing a needed operation, but do not include either successful mitigating actions or wrong actions that could make the events more severe. <u>Flaws and Cracks</u> - The deterministic calculations assume the presence of a long through-clad flaw of critical depth--a substantial conservatism. The probabilistic calculations use a through-clad crack probability that is highly uncertain and that some people believe is conservative. No account is taken of actual in-service inspection results in these generic calculations.

- 14 M

The crack growth/arrest model used by the staff assumes long initial flaws that grow uniformly over their length. This initial flaw shape is conservative. The growth/arrest shape is discussed in Section 3; we believe that, once a crack initiates, the long crack is a more realistic description than less conservative shapes used in other models.

<u>Stresses</u> - the models include no residual stresses, which is non-conservative. The NRC model includes cladding effect, which is realistic for through-clad cracks. The probabilistic analysis of small-break LOCA used in Section 8 took credit for warm prestressing.

None of the models used for generic events currently includes a warm prestress (WPS), which is a conservatism for transients satisfying the WPS conditions. WPS is included in calculations specifically limited to the small-break LOCA, where it has been demosntrated that WPS is applicable.

<u>Material Properties</u> - The estimation of RT_{NDT} as described in Section 5.4 and 8.5, is a substantial conservatism.

<u>Fracture Mechanics</u> - The use of linear elastic fracture mechanics in the uppershelf temperature region is appropriate. Assuming vessel failure when the stress intensity factor for a crack reaches the upper-shelf toughness is believed by many people to be conservative because considerable ductility exists in the remaining ligament. Until we have validated applicable elastic-plastic mode's, however, the degree of conservatism cannot be determined.

<u>Uncertainties in Probabilistic Calculations</u> - Substantial uncertainties exis in probabilistic calculations as discussed in Section 8.3. The characterization of event sequences by T_f , β , and P is an oversimplification that may or may not be on the conservative side. The net result of the above considerations is that the PTS analyses have substantial uncertainty, and are on balance believed by the staff to be substantially conservative. Neither the uncertainty nor the conservatism has been quantified.

- 10 %

Reference 8.3 includes discussions of 38 conservatisms, nonconservatisms, and factors whose effect is not known.

Plants with higher RT_{NDT} would be predicted to have higher PTS risk, so the additional evaluations and requirements of Sections 9 and 10 are proposed.

References

4

- 8.1 Letters dated May 28, 1982 (OG-70) and July 15, 1982 (OG-73) from O. D. Kingsley, Chairman, Westinghouse Owners Group to H. R. Denton, NRC; and additional probabilistic PTS results transmitted informally on June 22, 1982.
- 8.2 "Safety Goals for Nuclear Power Plants: A Discussion Paper," NUREG-0880, February 1982. (The draft policy statement is quoted on pp. vii-xxi.)
- 8.3 Letter dated November 5, 1982, from L. T. Pedersen, PNL, to Felix Litton, NRC, enclosing draft Supplement 1 to NUREG/CR-2837.

9. PLANT-SPECIFIC ANALYSES AND EVALUATIONS TO BE PROVIDED BEFORE THE SCREENING CRITERION IS EXCEEDED

1.1.

9.1 Introduction

The study of pressurized thermal shock to determine if there exists a need for interim improvements while the long-term program continues has led the staff to recommend a two-step process. The first step is to establish a screening criterion based on RT_{NDT} to identify reactor vessels where radiation embrittlement has progressed to the point of potential concern. This criterion was selected using simplifications and generic treatment of certain design features, transients, fracture mechanics analyses and plant operating characteristics as described in Sections 2, 3, and 4. The second step, to be taken for plants with vessels with values of RT_{NDT} that exceed or are approaching the screening criterion, involves more detailed plant-specific analyses to determine what, if any, modifications are necessary to the plant design and/or operations to resolve the concern.

The purpose of this section is to outline the analyses and actions to be required of those licensees whose reactor vessels have exceeded the RT_{NDT} sreenning criterion or will exceed the screening criterion within three calendar years. More detailed requirements must be formulated by the staff in the near future so that it will be clearly understood what methods of analysis are acceptable to the staff and what level of detail is required. Further, and most important, acceptance criteria will be developed and promulgated regarding the required analyses and actions.

9.2 Evaluation of Overcooling Event Sequences

Assessment of pressurized thermal shock concerns on a plant-specific level requires a study of the uinque potential for and consequences of severe overcooling transients at the specific plant. The overcooling transients must be chosen for analysis based on a detailed plant-specific control and safety

9-1

system design, procedural, and a plant specific human factor study. The study must include a systematic search for, and identification of, potential overcooling event sequences to identify those sequences which are the dominant contributors to the risk of pressure vessel failure. The human factors studies to identify needed procedural changes must be based on an integrated evaluation of all plant procedures at the specific plant for PTS-related and non PTS-related events. The generic studies of potential event sequences done thus far by the staff and by the Westinghouse Owners Group, described in Section 6 of this report, have shown that consideration of only the design basis accident sequences conventionally presented in Safety Analysis Reports does not identify adequately those dominant sequences. The design study must include systems functions pertinent to cooldown transient sequences and must include such systems as the feedwater system, steam generator level control system, steam dump system, steam generator power operated relief valves, charging and letdown system, emergency core cooling system, monitoring instrumentation, and control and safety systems actuation instrumentation. The procedural and human factors study must include operating and emergency procedures, instrumentation available to the operators, operator training, and the ability of the operators to diagnose transients and accidents that occur or do result in a rapid cooldown of the primary system. Any thermal-hydraulic models used in these studies must be verified appropriate for use with PTS scenarios.

The purpose of this study is twofold. First, the results of this study will be used in event-tree analyses which would identify failures that could initiate cooldown transients and quantify the frequency of these events and end states. This information will then be used to select those events that should be subjected to detailed thermal-hydraulic analyses to determine the coldown rates and end states in characteristic pressures and temperatures, which will be used in fracture mechanics analyses whose results will help determine risk.

Second, the results of this study should identify systems, instrumentation, material, and procedural and training program improvements necessary to reduce the probability and consequences of pressurized thermal shock events.

9.3 Vessel Materials Properties (Refer to Appendix E for background and detail)

- 11

Available information on the vessel properties should be re-examined in detail to fill any gaps in the supporting data for making an estimate of RT_{NDT} and to support resolution of any disagreements about the validity of values used.

9.3.1 Improve Basis For Initial RT_{NDT}

٩.,

As noted in Section 5.2, and discussed in more detail in Appendix E, for many older reactor vessels, few data are currently readily available and validated to support the selection of a value for the initial RT_{NDT} . The confidence that can be placed in estimates of the initial RT_{NDT} depends not only on metal-lurgical tests, but also on the accurate documentation of welding technique, weld wire used, and weld flux used. The credibility of such estimates could be enhanced by performing more tests on archival material, by discovering previously unreported test results on weld specimens from the particular plant, or by evaluating properties of welds considered typical of the plant-specific weld.

9.3.2 Refinement of Chemistry Information for Critical Materials

If it were necessary to assume 0.35 percent copper, because there was no other information, attempts should be made to find archival material suitable for chemical analysis or data on the weld material from other vessels where it may have been used. If the surveillance material matches one of the critical welds, some check for analyses for copper and nickel contents of broken Charpy bars should be considered.

9.3.3 Vessel Fluence (See Appendix F)

Fluence calculations for the critical welds should be rechecked, using modern codes and information from surveillance dosimetry. Location of critical welds relative to the axial and azimuthal flux map should be taken into account, as well as changes in fuel loading during periods when dosimeters were expose.

9.4 Deterministic Fracture Mechanics Evaluations (See Appendix D)

For the limiting transients as determined in 9.2 and materials properties as determined in 9.3, licensees should provide sufficiently detailed fracture mechanics analyses to permit the NRC staff to interpret the results without its having to redo the calculations. The details should include a listing of the assumptions used, the bases for them and a discussion of the sensitivity of the results to variations in the assumptions. Items to be discussed are:

. . . . -

- Vessel wall thickness and clad thickness; vessel inner radius
- Location and orientation of the assumed initial crack
- Heat transfer coefficient used and material properties, k, $(\frac{E \alpha}{1-\gamma})$ vs. temperature
- Assumed crack shape at initiation and time(s) of initiation
- Crack shape at arrest
- Treatment of cladding-induced stresses
- Upper shelf toughness
- Bases for the determination of limiting RT_{NDT} (at the inner vessel radius)

The results of each transient analyzed should be protrayed as a plot of critical and arrest relative crack depths versus time into the transient. Superpose a line indicating when warm prestressing is deemed to be effective and a curve indicating the depth at which the upper shell toughness is reached. If crack arrest is predicted and accepted at or above the upper shelf, it must be justified.

9.5 Flux Reduction Programs (See Appendix I)

Fuel management programs should be instituted to reduce neutron flux at the reactor vessel wall and at critical weld locations. This would reduce the rate at which the reactor vessel experiences a decrease in ductility and fracture toughness properties. Particular areas of concern in the reactor vessel should be located from an analysis of the material properties of the reactor vessel plate and weld metals. Consideration should be given to replacing fuel assemblies in close proximity to these critical areas. To recuce flux levels, these fuel assemblies could be replaced by spent fuel, zircally or stainless steel spacers, or water. Another scheme to be investigated would be an in-out loading pattern where fresh fuel is loaded into the center of the

core and moved outward in later cycles. Implementation of revised fuel management techinques have demonstrated a reduction in the neutron flux at the positions of previous maxima by factors of approximately two without derating the reactor coolant power level.

- 111

Appendix I contains preliminary estimates of the flux reduction factors (FRP) that can be obtained by reshuffling fuel to place older fuel in the outer positions near the critical welds. It also contains estimates of the FRF that can be obtained from more radical actions such as removal of outer assemblies and replacement with stainless-steel (dummy) elements.

Appendix I also contains results of a study showing what FRF would be necessary to prevent exceeding the screening criteria before end-of-life for each domestic PWR. The factors range from about 10 less than 1 (the latter implying that plant will not reach the screening RT_{NDT} even with no flux reduction).

Finally, Appendix I contains first rough preliminary estimates of what the needed range of FRFs would cost. The costs will be quite dependent on plant specific information (power shape, margin, whether limited by thermal margin or ECCS limits etc.) and are greatly influenced by power derate necessary, if any.

9.6 Inservice Inspection and Nondestructive Evaluation Program (See Appendix L)

Current requirements specified in 10 CFR 50.55a endorse ASME Section XI as defining the examination requirements for reactor vessel welds. The volume of weld to be examined includes the near-surface area; however, currently employed techniques do not provide sufficient basis for assuming that all near-surface cracks can be detected.

The utilizaton of state-of-the-art nondestructive evaluation techniques provides an opportunity to decrease or eliminate a conservatism used in the generic assessment of pressurized thermal shock; that is, small cracks exis. at or near the surface of the reactor vessel. Existing inservice inspection programs should be reevaluated to consider incorporation of state-of-the-art examination techniques for inspecting the clad-base metal interface and the

8.₁

near-surface area. This would include plant-unique consideration of the clad surface conditions and may require grinding the clad metal smooth enough to utilize these techniques, and consideration of increased frequency of inspections. Other more detailed recommendations are covered in Appendix L to this report.

9.7 Plant Modifications

To adequately protect reactor vessels from the effects of pressurized thermal shock, the protection needs to be compatible with the plant design and commensurate with the vessel's fracture toughness properties and/or susceptibility to cooldown transients. Modifications to be considered should include the following:

(1) Instrumentation and Controls (See Appendix J)

- (a) reactor vessel downcomer water temperature monitor
- (b) instantaneous and integrated reactor coolant system cooldown rate monitors
- (c) steam dump interlock
- (d) feedwater isolation/flow control logic
- (e) reactor coolant system pressure and temperature monitors
- (f) NDT margin monitor
- (2) Automatic Depressurization Logic
- (3) Increased Emergency Core Cooling Water and Emergency Feedwater Temperatures (See Appendix K)

Because of design differences and transients response characteristics, plan specific consideration should be given to any system modification. Further for active system modifications such as an automatic depressurization system, a failure mode and effects analysis should be performed to verify that

9-6

بالمعارية المراجع

inadvertent operation of the system would not induce transients more severe than the mitigative capabilities of the plant's safety systems or that otherwise create an unacceptable risk.

· 11 c

9.8 Operating Procedures and Training Program Improvements (See Appendix C)

As a result of generic pressurized thermal shock event tree analysis and actual reactor operating experience it has been shown that operator actions and associated plant response play a key role in the initiation and mitigation of pressurized thermal shock events. The licensees of the seven plants currently being evaluated by the NRC for susceptibility to pressurized thermal shock have reviewed these current operating procedures for information relevant to the pressurized thermal shock issue. Based on the NRC's and the licensee's review of their own procedures, a number of revisions have been incorporated.

These are extensive, industry-NRC ongoing programs to develop a set of integrated procedures that will consider the balance of requirement necessary to ensure core cooling while preventing vessel overcooling. This program is described in Appendix C.

As part of the detailed analyses and actions to be taken when a plant is within 3 calendar years of the screening RT_{NNT}, the licensee should:

- (1) Ensure that the actions specified in the procedure guidelines are based on an integrated evaluation of relevant technical considerations, including PTS, core cooking, environmental releases and containment integrity.
- (2) Implement upgraded Emergency Operating Procedures which are based on technical guidelines developed or described in (1) above for NUREG-0737, I.C.1, "Guidance for the Evaluation and Development of Procedures for Transients and Accidents." All licensees will be required to implement these upgraded procedures on a schedule developed in accordance with SECY-82-111B.

9-7

11/13/82

*

48

The procedures developed and implemented by items (1) and (2) above should address the following types of concerns:

- (a) Instructions should include allowance for system response delay times
 (e.g., loop transport time, thermal transport time).
- (b) The need for cooldown rate limits for periods shorter than one hour should be evaluated.
- (c) Methods for controlling cooldown rats should be provided.
- (d) Guidance should be provided for the operator if cooldown rates or brittle fracture limits are exceeded.
- (e) The desired region of operation (e.g., subcooling band) on the pressuretemperature curve should be evaluated to determine if it can be revised to maximize the operator's ability to prevent brittle fracture.
- (f) Instructions for controlling pressure following depressurization transients should be provided.
- 9.9 <u>In-Situ Anrealing</u> (Ser Appendix M)

Annealing of the reactor vessel is a possible, although difficult and expensive, remedial measure for the radiation embrittlement problem. Research sponsored by both the regulated industry and the NRC has provided a basis for selecting the temperatures and duration of the annealing process with some data on reirradiation damage. Research is being funded by the Electric Power Research Institute on the feasibility of annealing. A draft report on annealing proposes the use of electric resistance heating elements supported by a frame that can be lowered into the reactor vessel. The draft report on this study finds no insurmountable difficulties; however, many engineering details remain to be resolved. These include the potential for vessel damage, and protecting the concrete and vessel support structures from the effects of high temperatures. For those plants where proposed remedial actions of the

ţ.

.......

types described in Sections 9.2 through 9.3 above do not result in acceptable risks of vessel failure for the whole design lifetime, a plant-specific engineering evaluation of in-situ annealing should be performed.

115

The feasibility of conducting a demonstration anneal on an older, irradiated vessel is being investigated and is reported in Appendix M. The investigation has not reached a final conclusion. It is believed that useful information could be obtained regarding both material property recovery and regarding practical difficulties of actually performing the operation, if a suitable vessel could be found. However, the costs would be great and a suitable vessel has not been identified.

9.10 Basis for Continued Operation

Finally, as part of the plant-specific analysis package, the licensee will provide a basis for concluding whether or not continued plant operation is justified, and if it is justified the licensee must demonstrate that interim compensatory measures are in place that will effectively protect the public health and safety and that the licensee is proceeding in good faith in a timely manner to achieve a final resolution.

This basis should include details regardi ~ frequency of PTS events, description of the dominant risk contributors, and assessment of the total risk from all such events. Vessel and containment failure modes should be discussed, and it should be shown quantitatively how such considerations are factored into the overall risk assessment. The total projected PTS risk for the interim period until acceptance criteria can be met by corrective action should then be compared to the NRC safety goal.

10.0 CONCLUSIONS

۶.

10.1 Conclusions

As a result of evaluations performed thus far of the issue of pressurized thermal shock, the NRC staff has reached the following conclusions:

- (1) The risk from PTS events for reactor vessels with RT_{NDT} values less than the proposed screening criterion (270°F for axial welds, and 300°F for circumferential welds) is acceptable. On the basis of presently available information, no reactor vessel will exceed the screening criterion for the next few years, therefore there is no need to shutdown or anneal any operating PWR in the next few years.
- (2) Most plants can avoid reaching the screening criterion throughout their service life by timely implementation of flux reduction programs. Such flux reduction programs should be implemented on a time schedule that will avoid foreclosure of this option.
- (3) Any plant for which the value of RT_{NDT} is projected to reach the screening criterion before the end of service life, using the conservative method of RT_{NDT} determination described in Section 5 and Appendix E, should submit plant-specific evaluations (of the type described in Section 9) to determine what, if any, modifications to equipment, systems and procedures should be required to provide acceptable protection against vessel failure from PTS events for the remainder of plant life. These evaluations should be submitted three years before the vessel is projected to reach the screening criterion.
- (4) In the near future, the staff should develop more detailed

PTS REPORT SEC 10

guidance for these evaluations and acceptance criteria for determining whether plant modifications are needed based on the evaluations.

nt

(5) Some of the Commission's regulations (Appendix G to 10 CFR Part 50, 10 CFR 50.46, and possibly others) may not appropriately reflect current understanding of the state of reactor vessel embrittlement and the potential for vessel failure as a result of PTS. Timely consideration should be given to the possible need for amendments to the regulations (as discussed in Section 8.6).

10.2 Comment on Overall Approach

It will be evident from this report that the staff is <u>not</u> proposing to resolve PTS issues by requiring a "design-basis pressurized thermal shock event" to be analyzed by a prescribed conservative evaluation model with the results to be compared to specified acceptance criteria. Rather than this traditional approach, the staff has used analyses of overcooling event sequences actually experienced, plus a wide spectrum of possible sequences that have not occurred, together with explicit consideration of the frequencies or probabilities of occurrence of the various events. Moreover, the staff has used analysis models as realisitic as the state of the art permits, with a few explicit conservatisms to provide the needed margin of safety. The overall level of safety thus provided has "Leen estimated (very approximately) using probabilistic analysis.

The staff believes that this approach, a departure from regulatory practice in the past, gives improved coverage of the spectrum of PTS event sequences, improved evaluation of the level of safety, and improved accounting for the various possible events that can mitigate or aggravate an ongoing PTS sequence. APPENDICES TO

٠

84

NRC STAFF EVALUATION

1 1

OF

PRESSURIZED THERMAL SHOCK

(November 1982)

APPENDIX A

11

OVERVIEW OF ACTIVITIES CONCERNING THE PRESSURIZED THERMAL SHOCK ISSUE

A.1 INTRODUCTION

The subject of thermal shock to reactor pressure vessels from overcooling transients is not a new concern; both industry and the NRC have held meetings and issued written reports on the subject for several years. The thermal shock concern after a Loss of Coolant Accident (LOCA) has been subject to considerable review in the past. Analyses and experiments indicate that the vessel will still hold water after a large LOCA. Therefore, for large LOCA, thermal shock to the reactor pressure vessel is not a new concern.

The TMI Action Plan (NUREG-0737, Item II.K.2.13 "Thermal Mechanical Report Effect of High-Pressure Injection on Vessel Integrity for Small-Break Loss of Coolant Accident with No Auxiliary Feedwater") identified one transient of concern which is characterizd by severe overcooling causing thermal shock to the vessel, concurrent with or followed by repressurization (that is, Pressurized Thermal Shock, PTS). The staff has recognized that there are many other scenarios which could result in PTS. On the basis of events which have occurred at operating PWRs, the staff recognized early in 1981 that some operating reactor pressure vessels of the older plants were approaching material property conditions which made the PTS issue a greater concern. Thus the NRC staff requested a meeting with industry representatives on March 31, 1981, to discuss the PTS problem. This initiated the current effort concerning the PTS issue.

The PTS issue is a concern only for operating PWRs. Boiling water reactors (BWRs) are not a significant PTS concern. BWRs operate with a large portion of water inventory inside the pressure vessel at saturated conditions. Any sidden cooling will condense steam and result in a pressure decrease, so simultaneous creation of high pressure and low temperature is improbable. Also contributing

to the lack of PTS concerns for BWRs is the lower fluence at the vessel inner wall and the use of thinner vessel wall which results in a lower stress intensity for a postulated crack.

- 11 -

The attached appendix provides a time table of events concerning the PTS issue.

A.2 Summary of Industry Meetings with the Staff

On March 31, 1981,¹ the NRC staff met with the PWR Owners Groups and representatives of NSSS vendors to discuss the effects of potential thermal shock to reactor pressure vessels by overcooling transients and the potential consequences of subsequent repressurization at relatively low temperatures. The staff requested the industry to make a plant-by-plant assessment of the problem and to scope and bound the problem. As a result of this meeting the industry representatives committed to a report by May 15, 1981, providing an account of what immediate problems exist. Subsequently, by letter dated April 20, 1981,² the NRC requested the Owners of PWR operating plants to respond by May 22, 1981, identifying the specific action which the plant Owners propose to take.

Meetings were held with Babcock & Wilcox (B&W), Westinghouse (W) and Combustion Engineering (CE) Owners Groups (OG) on July 28, 29, and 30, 1981,³ respectively, at the request of the staff in order to present the staff's analysis of the problem and the actions the staff intends to take and to hear from the PWR Owners the results of their analyses and their proposed actions concerning the problem. The staff concluded at this meeting that Owners of plants of each NSSS type which have the highest RT_{NDT} values would be requested to take action to resolve the problem for their plants. Subsequently, the NRC requested under 10 CFR 50.54f, by letters dated August 21, 1981,¹⁷ the licensees of Oconee 1, TMI-1, Robinson 2, Turkey Point 4, San Onofre 1, Calvert Cliffs 3, Fort Calhoun and Maine Yankee (1) a 60-day response for information related to RT_{NDT} and operator action to prevent PTS and ensure vessel integrity and (2) a 150-day response for information which would define actions and schedules for resolution of the PTS issue and analyses to support continued operation.

Follow-up meetings were held with the W, B&W, and CE OGs on September 18 and 22, and October 7, 1981,⁴⁺⁵⁺⁶ respectively, to review the progress and to discuss the technical issues concerning the systems analyses, operator responses, and the materials and fracture mechanics aspects of the PTS issue.

The WOG indicated their report for the TMI Action Plan Item II.K.2.13 due at the end of the year would address (1) the Small Break Loss of Coolant Accident with Loss of Feedwater (SBLOCA + LOFW) (TMI II.K.2.13) and other scenarios including steam line breaks, (2) fracture mechanics calculations for each operating plant, (3) the date and RT_{NDT} for each plant when acceptable conditions will not be met, and (4) evaluation of remedial actions. WOG indicated that the most limiting plant has at least 3 EFPY remaining before there is a concern.

The B&WOG indicated that their work would be concentrated on the Oconee 1 150-day response to the August 21, 1981 letter and plant-specific analyses thereafter.

The CEOG indicated that the report due at the end of the year would address the TMI Action Plan Item II.K.2.13 and other scenarios including the main steam line break event. The CEOG indicated that the most limiting plant has at least 5 EFPY remaining before there is a concern assuming the no-crack initiation criteria.

Meetings were held with the WOG and CEOG including the Owners of the six selected plants who received the August 21, 1981 letter on February 24 and March 3, 1982,^{7,8} at the request of the staff. A meeting was held with Dule Power Company on March 24, 1982.⁹ These meetings were to discuss the respective Owners groups' reports and the "150 day" responses concerning San Onoire 1, Robinson 2, Turkey Point 4, Fort Calhoun, Calvert Cliffs 1, Maine Yankee, and Oconee 1. TMI-1 was not included in these discussions since GPU elected to delay their "150 day" submittal until June 1982. These meetings were designed to respond to specific staff concerns which were identified with the published meeting notices and later were, in part, transmitted to the Owners of the selected plants.³⁴⁻⁴¹

Meetings were held with the Omaha Public Power District and the WOG on May 6, and 10, 1982,^{10,11} respectively at their request to update the staff on the progress of the respective programs and the responses to the staff's concerns identified in the previous meetings.

The WOG provided the results of a study involving a methodology leading to a probabilistic risk assessment (PRA) related to PTS. The conclusion of this study was that the likelihood of a cooldown transient can challenge the reactor vessel is less than 10^{-4} to 10^{-3} per reactor year for that lead plant at 5 EFPY from today. WOG maintains that the total risk to public health is in the area of 10^{-9} .

The CEOG provided responses to the staff concerns identified in the meeting of March 3, 1982. In particular the CEOG provided the results of their review of operating experience of CE operating plants and the results of a probability analysis related to the PTS issue. The review representing 49 reactor years of operating experience identified 16 events which met a screening criterion. Of those only two met the selection criteria. These actual overcooling events were much less severe than the event analyzed in the "150 day" response and there was no uncontrolled repressurization in either event. The probability study conc²⁰ Jed that the main steam line break (MSLB) is the most severe event and ranges between a probability of 10^{-6} to 10^{-4} .

A meeting was held on June 2, 1982,¹² with General Public Utilities (GPU) at their request to provide the staff a status report on the PTS program for TMI-1 and to present a summary of the "150 day" response for TMI-1. Significant in this study was the use of the COMMIX Code in the mixing analysis. The COMMIX Code shows warmer temperatures for the SBLOCA events than the BAW 1648 or Oconee 1 mixing models. The SBLOCA and turbine bypass valve failure were the only events analyzed. GPU determined that based on EOL RT_{NDT} of 335°F for the most critical weld, operation would be acceptable for 32 EFPY.

A meeting was held on June 9, 1982,¹³ with the PWR industry representatives at the request of the staff for the purpose of discussing the current NRC staff considerations of possible recommendations for PTS requirements. The staff

was considering a limit of $T_f RT_{NDT}$ of 230°F for longitudinal welds and 255°F for circumferential welds based on a transient which resulted in a final temperature/pressure of 250°F/2500 psi ($\beta = 0.15$) which would initiate a crack. The industry representatives did not agree with the conservatism of the staff considerations. They objected to the crack initiation criteria. They believed the final temperature was too low and the pressure was not possible. They objected to the data base which was used for the probabilistic determinations. The staff provided the industry two weeks to submit comments in order for staff to consider the industry news in the determination of the staff's position.

Meetings were held on June 22 and 23, 1982, 1^{4} , 1^{5} with the WOG and CEOG respectively at their request to respond to the staff's request for comments to the staff's proposed recommendations on PTS requirements. WOG proposed a screening criteria of a RT_{NDT} of 310° F and 335° F at longitudinal and circumferential welds respectively. This criteria was based on $T_{f} = 290^{\circ}$ at the surface of the reactor vessel weld. The WOG PRA and the NRC probabilistic fracture mechanics was coupled with the <u>W</u> probabilistic transient evaluation to yield safety goals somewhat lower than those reported by the staff. The WOG analysis indicated that the PTS issue would be of no concern to operating plants for the transient for the next five years of plant operation.

The CEOG recommended the use of CEN-189 best-estimate initial RT_{NDT} values. They recommended the current Regulatory Guide 1.99 but used to predict the upper bound shift for high-copper, high-nickel material at fluence greater than 10^{19} nvt and that Guthrie (HEDL) correlation be used to predict the upper bound shift for medium-low copper, high-low nickel material at fluence les: than 10^{19} nvt. The CEOG believes arrest will occur. The probabilistic analysis indicated that the MSLB bounds the PTS events.

A meeting was held with the WOG on July 30, 1982,¹⁰⁰ at the request of the staff to discuss the apparent discrepancies between the WOG and the staff concerning the limiting transients which produce the greatest overcooling, the frequencies of such transients, and the fracture mechanics analysis associated with the transients. In particular, the meeting discussed the small break

Å-5

LOCAs (SBLOCAs) which result in stagnation flow and the factors in the fracture mechanics analysis which account for the differences between the WOG and the staff. WOG indicates that SBLOCAs in the area of 2" to 3" were the sizes of concern which result in stagnation flow and the frequencies of such events were conservatively estimated to be 6×10^{-4} . Factors which account for the differences in the fracture mechanics analyses were heat transfer coefficient used, crack length assumptions, and effects of the clad. WOG assumed the heat transfer coefficient was not 300° continuous. It varies as explained in WCAP 10019. WOG assumed an elliptic crack versus the staff's assumptions of an infinite long crack. WOG assumed the clad has no effect.

A follow-up meeting¹⁰¹ was held with the WOG on August 11, 1982. WOG indicated that a lower limit for the SBLOCA of concern was 5×10^{-5} (a medium value). More realistic mixing assumptions concerning other factors such as metal heat resulted in approximately 60° increase to prior results of analysis of the SBLOCA.

For the longitudinal flaw the calculational differences between the staff and the WOG amount to 45° RT_{NDT} .

The staff proposed a screening criteria as follows:

 $T_F = 260^{\circ}F$ at 10^{-2} frequency $RT_{NDT} = 270^{\circ}$ for longitudinal welds $RT_{NDT} = 300^{\circ}F$ for circumferential welds.

The above is based on operating references.

A.3 Summary of Industry Responses to Staff Requests

At the meeting of March 31, 1981, with the PWR industry representatives, the PWR Owners Groups agreed to provide individual owners groups reports by May 15, 1981, which would provide an accounting of what immediate problems exist. By letters dated April 20, 1981,² the NRC requested the Owners of operating FWR plants to provide responses by May 22, 1981 relating to their participation in
the Owners groups programs and specific action which they intend to take. By letters dated August 21, 1981,¹⁷ the NRC requested 30-, 60- and 150-day responses from each of eight selected utilities owning plants which represented the three different vendor NSSS and reactor vessels with the highest irradiation damage of each group.

The NRC responded¹⁸⁻⁴¹ to each of the utilities responses to the August 21, 1981 letter. As a result of the 60- and 150-day responses the staff requested additional information from each utility which received the August 21, 1981 letter.

Each of the Owners groups provided responses by May 25, 1981⁴²⁻⁴³ with an accounting of the immediate concern and their plans for resolving the issue. Owners of all operating PWR plants indicated their participation in the Owners groups programs by letters in response to the NRC letter dated April 20, 1981.

Tables A-1, A-2 and A-3 provide the summaries of the 30-, 60- and 150-day responses, respectively⁴⁵⁻⁶⁶ of the selected utilities which received the August 21, 1981 letter. Table A-4 provides a summary of the W and CE generic reports⁶⁷⁻⁶⁸ concerning the PTS issue.

A.3.1 Responses Relating to Westinghouse Plants

4

The WOG response dated May 14, 1981 from Mr. Robert W. Jurgensen⁴⁴ indicated that all Westinghouse operating plants could sustain severe thermal shock transient, including repressurization to beyond January 1983. The WOG program would be completed by December, 1981. Each utility of a Westinghouse plan. would provide additional information including a schedule for remedial actron if requested on completion of the WOG program.

Tables A-2, A-3 and A-4 provide summaries of the "60 and 150 day" response: and the generic reports. These responses were supplemented by additional information received from the WOG and each of the three selected Westinghouse operating plants in May 1982.⁶⁹⁻⁷² The Westinghouse WOG report concludes that a number of reactor vessels will require more plant-specific evaluations and may require that remedial actions be implemented at some point in the vessel life to demonstrate vessel integrity to end-of-life. The licensees of the three Westinghouse reactors all concluded that vessel integrity will be maintained to or beyond end-of-life.

The supplemental information provided by the WOG at the meeting of May 10, 1982,¹¹ concludes that the probability of a transient of PTS concern for the "lead" plant at 5 EFPY from today is between 10^{-4} to 10^{-3} .

By letter dated May 28, 1982,⁶⁹ the WOG provided supplemental information on Reactor Vessel Integrity in the form of a report "Summary of Evaluation Related to Reactor Vessel Integrity." This report supported the conclusions provided by the WOG at the meeting of May 10, 1982.

By letter dated June 16 1982,⁷³ the WOG provided a discussion of benefits and penalties of fuel management schemes to reduce fluence in the form of a report "Fuel Management To Reduce Neutron Flux." This report provides methods of reducing the flux to the pressure vessel with no power derating or economic penalty.

By letter dated June 22, 1982,¹⁴ the WOG provided the "Review of the Emergency Response Guidelines Related to Pressurized Thermal Shock." This report explicitly identified those steps in the Emergency Response Guidelines (ERG) that have been written to provide operator direction to prevent a mitigated PTS event. The report also determined those areas of the ERGs that should be modified to more clearly identify appropriate operator responses to prevent or mitigate potential PTS events.

By letter dated July 15, 1982 [103], the WOG provided the material which wa; presented at the meeting of June 22, 1982, with the staff on the PTS issue. The letter provided discussions on the calculational differences between WO3 and the NRC staff and the use of the technology relating to PTS in future regulatory actions. The WOG utilizes a transient with a final temperature of 290°F to obtain a screening RT_{NDT} value of 310°F to obtain a screening RT_{NDT}

value of 310°F. The WOG recommends this value for a screening criteria for prioritizing plants for plant specific programs for the resolution of the PTS issue.

By letter dated September 2, 1982 [104] supplemented by letter dated September 16, 1982 [105] the WOG provided material which was presented at the August 11, 1982, meeting with the NRC staff. The WOG indicated substantial agreement with the staff's methods, techniques, assumptions and the screening values for RT_{NDT} (270° longitudinal welds and approximatley 300°F for circumferential).

A.3.2 Responses Related to Combustion Engineering Plants

The CEOG response dated May 15, 1982, from Mr. K. P. Baskin⁴³ indicated that the steam line break transient produces the largest magnitude and rate of heat removal for the CE-NSSS design. With this transient, approximately 5 EFPY of operation would have to elapse before vessel integrity would theoretically become a concern.

The CEOG response indicates that a program is planned to address all aspects of the PTS issue and a generic response to TMI Action Plan Item II.K.2.13 would be provided by January 1, 1982.

The Combustion Engineering CEOG Generic Report⁶⁸ concludes that all CE plants can withstand the postulated small break LOCA (SBLOCA) with extended Loss of Feedwater (LOFW) scenarios for the assumed life of the plant. The 150-day responses⁶⁰⁻⁶² from the three licensees of operating CE plants all indicato that vessel integrity will be maintained for the lifetime of the plant.

The supplemental information provided by each of the three selected CE operating plant owners⁷⁵⁻⁷⁷ indicated that the main steam line break event is the most limiting event and ranges between a probability of 10^{-6} to 10^{-4} . They also provided an identification of overcooling events from the operating history of CE plants. In addition the responses discussed the sensitivity of controlling overcooling transients to operator action.

By letter dated June 14, 1982, the CEOG provided a response to the NRC staff proposed position that was presented at the June 9, 1982 meeting. This letter reiterated the CEOG conclusion that the MSLB event is the most limiting and probable concerning the PTS issue. The CEOG concludes that an $R\Gamma_{\rm NDT}$ value of 320°F is more appropriate for crack initiation criteria for the NRC proposed transient ($T_{\rm F}$ = 250°, P = 2500 psi, β =0.50m⁻¹).

. . . .

The NRC staff proposed crack initiation criteria was considered unnecessarily conservative. The ability of the CE-NSSS to cool down as rapidly as the NRC proposed temperature transient while maintaining pressure at 2500 psi is considered physically impossible. The CEOG contends that the NRC calculated probability of the NRC proposed temperature transient is much too high. The CEOG disagrees with the approach taken by the NRC to resolve the PTS issue.

Omaha Public Power District provided comments concerning the staff's proposed position by letter⁷⁹ dated June 26, 1982. OPPD suggested that some type of screening criteria would be appropriate to focus on plants which might develop a potential PTS concern. The screening criteria should reflect the assessment of plant operating histories and major design differences. The use of best-estimate RT_{NDT} value is most appropriate.

A.3.3 Responses Related to B&W Plants

The B&WOG response dated May 12, 1981 from Mr. John J. Mattimoe⁴² indicated that the SBLOCA with no repressurization is the bounding accident. This assumes operator action would mitigate repressurization (by throttling HPI and utilizing atmospheric dump or turbine bypass valves). B&W contended that 'he analysis is conservative and there is no concern for thermal shock through 1982. The B&W Owners submitted, in December 1980, BAW 1648, which addressed TMI Action Plan Item II.K.2.13 "Thermal Mechanical Report - Effect of HPI on Vessel Integrity for SBLOCA with Additional Loss of Feedwater." The B&WOG plans with respect to PTS to submit plant-specific analyses to address the conservatisms in the generic analysis. No generic report was planned for B&W plants.

Oconee 1 and TMI-1 were the B&W selected plants for the August 21, 1981 letter.

-

The 150-day response from Duke Power Company⁶⁶ concerning the Oconee 1 vessel concludes that no changes to the plant or additional fuel management, or reactor vessel annealing is necessary to assure safe operation of Oconee 1 through the design life of the plant. The Oconee 1 report indicated that severe PTS events were in the probability range of 10⁻⁶ to 10⁻⁴. The Duke Power Company letter dated April 30, 1982⁸⁰ provided additional information concerning operator responses and sensitivity of transient analysis to operator action times.

By letter dated March 17, 1982,⁸¹ GPU Nuclear informed the staff that the "150 day" response concerning TMI-1 would be submitted as soon as the revised mixing analysis could be inputted into the B&WOG plant-specific analyses (estimated completion June 1982).

By letter dated June 1, 1982,⁸² GPU provided a response to the NRC letters of August 21, 1981, and December 18, 1981. This letter provides a summary of an analysis which GPU proposed to provide at the end of June, 1982. The summary concludes that the TMI-1 reactor pressure vessel integrity will not be compromised due to PTS events during the lifetime of the plant. Also the rate of embrittlement of the TMI-1 vessel may be reduced further if the plant switches to low leakage fuel scheme in the near term reloads. GPU indicated because of the concerns by the PTS issue, operator response will be significantly improved through increased awareness and additional training.

By letter dated July 7, 1982,¹⁰² GPUN provided the "150 day" response to the NRC August 21, 1981 letter.

By letter dated June 22, 1982,⁸³ the B&WOG provided a response to the NRC staff's request at the meeting of June 9, 1982 concerning the NRC staff proposed position on the PTS issue. The B&WOG indicated that the generic position is unsound, unrealistic, and inappropriate unless used solely as a screening basis. Also the NRC proposed crack initiation criteria was considered to be highly conservative and the proposed generic transient does not realistically represent an actual B&W plant response. The B&WOG recommended the use of

RT_{NDT} as a means of "flagging" plants with potential concerns. Each plant should be analyzed for a realistic probable transient.

Letters⁸⁴⁻⁸⁸ were received from Duke Power Company, Arkansas Power & Light Company, Florida Power Corporation, GPU Nuclear and SMUD in response to the staff's request concerning the proposed staff position concerning the PTS issue. Duke Power Company indicated that the staff proposed approach can be utilized as a screening method of identifying plants for detailed analyses with respect to the PTS issue. However, the staff's analysis of the frequency of the transient events is not applicable to any real plant. The Duke Power Company letter expressed the concern that the staff has failed to provide a feedback loop such that plant improvements made are directly included in the analyses. Also the screening criterion may need to be established on a group of plants or even on an individual plant basis rather than a generic PWR basis.

Arkansas Power & Light Company's comments⁸⁵ on the staff proposed position follows:

- (1) Indexing the operation to actual fracture toughness rather than on RT_{NDT} should be pursued.
- (2) RT_{NDT} could be used as a screening criteria.
- (3) The staff's generic approach does not consider basic design differences.
- (4) The transient selected by the staff is unrealistic.
- (5) The crack initiation criteria is not consistent with the ASME Code.
- (6) AP&L disagrees with some of the basic assumptions of the staff's proposal.

Florida Power Corporation⁸⁶ included the following comments on the staff's proposal:

- (1) System pressure does not remain constant as proposed by the staff.
- (2) Emphasis should be focused on actual fracture toughness rather than RT_{NDT}.

GPU⁸⁷ offered the following comments to the staff's proposal:

- (1) The staff's proposed failure criteria is too conservative.
- (2) The proposed governing transient is too severe and overly conservative.
- (3) Emphasis should be placed on the actual toughness of the vessel material rather than RT_{NDT}.

SMUD indicated that the PTS issue cannot be realistically evaluated by focusing on a single parameter such as RT_{NDT}.

A.4 <u>NRC Staff Audit of Operating Procedures, Operator Qualifications and</u> Training With Respect to the PTS Issue

On March 16, 1982, a NRC short-term task force on PTS was organized to make a detailed review and prepare a report on the efforts on PTS at the H. B. Robinson Nuclear Plant. Specifically, the task force was to provide a report characterizing the problems, methodology of resolution, bases for conclusions and recommendations regarding the adequacy of in-place training programs and operating procedures.

The report⁸⁹ "NRC Staff Audit of Robinson and Procedures and Training for Pressurized Thermal Shock" dated April 15, 1982, recommended that prior to restart the Robinson 2 operators and STAs should be retrained in areas related to the PTS issue and that SI termination criteria and procedures be changed to accommodate the PTS issue.

The task force also recommended similar audits be performed at the other seven plants which were identified with the August 21, 1981, letter.

By letter dated April 26, 1982, 90-95 the seven other utilities of plants of concern were requested to cooperate in this effort.

A.4.1 Robinson 2 Audit

4

A visit to the Robinson 2 site took place on April 5-7, 1982, to evaluate procedures and training. By letter dated April 20, 1982,⁹⁶ the staff confirmed the understanding that of general acceptance of the recommendations of the

task force report. This was confirmed in writing by CP&L by letter dated May 4, 1982.⁹⁷

A.4.2 Oconee 1 Audit

A review of Oconee's procedures and training for PTS was conducted May 11-13, 1982. In general, the review team found the operators adequately knowledgeable of the PTS issue, except that knowledge of past PTS events at other facilities was weak. The procedures provided mitigative actions to prevent PTS, but needed to be strengthened to provide actions if an unacceptable pressure/ temperature condition was reached. The audit team felt that a means should be provided for plotting cooldown rate and subcooling margin with the plant computer out of operation.

A.4.3 Fort Calhoun Audit

A review of the procedures and training for PTS at Omaha Public Power developments Fort Calhoun plant was completed on June 8-10, 1982 by PNL. General recommendations⁹⁹ regarding procedures and control instrumentation made by PNL included:

- (1) The values of parameters of interest in procedures should be consistent as appropriate.
- (2) Emergency procedures should note both minimum and maximum subcooling temperatures.
- (3) Emergency procedures should identify only one form of saturation curve.
- (4) The current NDT curve should be in every procedure which references i .
- (5) The subcooling margin indications should be available for all ranges of RCS temperatures.

A.4.4 San Onofre 1 Audit

An onsite audit was conducted of the San Onofre Unit 1 procedures and training for PTS June 3-4, 1982. Preliminary findings from the audit indicate that the procedures are based on plant-specific analyses of transients and that the operations personnel were familiar with PTS even though their training was not completed at the time of the audit. The findings indicate that the remainder of the training program should include instruction on past cooldown events. Findings on the San Onofre 1 procedures are included in the audit report. The procedures were generally found adequate for PTS considerations, and were based on Westinghouse analysis. The findings indicate that a method for plotting cooldown rate should be provided to the operators.

A.4.5 Maine Yankee Audit

A review of Maine Yankee's procedures and training for PTS was conducted on May 25~27, 1982. The review team found the plant operations personnel and STAs adequately knowledgeable of the PTS issue, and the procedures provided adequate guidance for preventing PTS. One significant operating philosophy already in place at Maine Yankee is the throttling of HPI flow to maintain as close to 50°F subcooling as possible during potential cooldown events. It was noted by the review team that no written exam was conducted after the lectures on PTS. Rather, a seminar method was used to determine the level of comprehension. Questions regarding PTS have been included in the written requalification examinations. The review team concluded that the operators were sufficiently knowledgeable of PTS. No changes to the operating procedures or training program were recommended to meet the objectives of the audit.

A.4.6 Calvert Cliffs 1 Audit

An onsite audit was conducted of the Calvert Cliffs procedures and training for PTS on July 6-8, 1982. Procedure recommendations were identified in the instructions for preferred methods of accident mitigation, priorities for mitigative actions, and procedure cross-reference. Training recommendations included additional training on accident mitigation methods.

A.4.7 Turkey Point 4 Audit

An onsite audit was conducted of the Turkey Point 4 procedures and training for PTS on July 13-15, 1982. No substantive recommendations were identified.

Table A-1 Summary of responses to NRC letters dated August 21, 1981 concerning thermal shock issue

October 7, 1981

.

. .

ċ

\$

.

	Licen	see's 30 day response o	concerning 5 items to b	e submitted in 60 days	
Plant Name (Date of licensee's 30 day response to ltr of 8/21/01)	(1) Request for present RT _{NDT} @ plates and welds	(2) Rate of RT _{NDT} increase	(3) RT _{NOT} limit for continued operation	(4) Basis for RT _{NDT} limit	(5) Questions concerning operator actions
Calvert Cliffs 1 (September 24, 1981)	Will be answered within 60 days	Will be answered within 60 days	Do not think a a simple RT _{MDT} value is an appropriate limit.	Will provide a more reasonable criteria for a limit for continued operation in 150 day response.	Will be answered in 60 days
Fort Calhoun (September 22, 1981)	Will provide value for plate material within 60 days. Will provide value for weld material in 150 day response.	Will provide value for plate material within 60 days. Will provide value for weld material in 150 day response.	Do not think a simple RT _{HDT} value is an appropriate limit.	Will provide a response within 60 days. Response will provide basis for response to (3).	Will be answered in 60 days.
Turkey Point 4 (September 23, 1981)	Will provide responses within 60 days.	Will provide responses within 60 days.	Response will be delayed until March 1, 1982.	Information has been provided in 5/14/81 ltr. Will provide additional informa- tion with response to (3)	Will be answered in 60 days.
Robinson 2 (September 21, 1981)	Will provide responses within 60 days.	Will provide responses within 60 days.	Response will be deferred until 150 day response.	Information has been provided, however additional informa- mation will be provided in 150 day response.	Will be answered in in 60 days.

......

.

. .

A-15

8

•

TABLE A-1 (Continued)

	Licen	see's 30 day response	concerning 5 items to	be submitted in 60 days	
<u>Plant Name</u> (Date of licensee's 30 day response to ltr of 8/21/01)	(1) Request for present RT _{NDT} plates and welds	(2) Rate of RT _{NDT} increase	(3) RT _{NDT} limit for continued operation	(4) Basis for RT _{NDT} limit	(5) Questions concerning operator actions
San Onofre <u>1</u> (October 5, 1981)	Will be provided by November 4, 1981 (15 day delay).	Will be provided by November 4, 1981 (15 day delay).	Will be provided as soon as made avail- able. Owners Group work will not be complete until end of year.	Will be provided by November 4, 1981 (15 day delay).	Will be provided by November 4, 1981.
Oconee 1 (No 30 day response expected - telephone conversation indi- cated no conflicts seen).	Will provide 60 day response.	Will provide 60 response.	Will provide 60 day response.	Will provide 60 day response.	Will provide 60 day response.
<u>TMI-1</u> (October 1, 1981)	Will provide response within 60 days.	Will be answered within 60 days.	Will be answered within 60 days.	Will be answered within 60 days.	Will be answered within 60 days.
<u>Maine Yankee)</u> (September 29, 1981)	Could answer within 60 days, however, would prefer to respond more fully within a short time after 60 day response is due.	Could answer within 60 days, however, would prefer to respond more fully within a short time after 60 day is due.	Could answer within 60 days, however, would prefer to respond more fully within a short time after 60 day resp. is due.	Could answer within 60 days, however, would prefer to respond more fully within a short time after 60 day resp. is due.	Could answer within 60 days, however, ould prefer to respond more fully within a short time after 60 day resp. is due.

Table A-1 (Continued)

.

•

٠

.

۱ :

. . .

\$

ţ

Licensee's 30 day response concerning the 150 day response:	Remarks
Will provide a plan within 150 days. Item 7 of request for additional information will not be complete. Will provide a schedule for complete response.	NRC letter dated 10/22/81 indicated staff would accept what licensee will provide. Staff will continue effort which may result in specifying conservatisms where more definitive info. is not available.
Will provide the 150 day response.	NRC ltr. dtd. 10/27/81 acknowledged receipt of ltr. dt. 10/20/81 and requested licensee to submit weld material data as soon possible as a supplement to ltr. dtd. 10/20/81.
Will provide response Jan. 1, 1982. Will provide schedules & additional analyses for remedial actions by 3/1/82.	NRC letter dated 10/22/81 indicated staff would accept what licensee will provide. Staff will continue effort which may result in specifying conservatisms where more definitive info. is not available.
Will provide 150 day response.	NRC letter dated 10/22/81 indicated staff would accept what licensee will provide.

. .

÷ •

18

•

Licensee's 30 day response concerning the 150 day response.

No conflict seen.

.

Will provide 150 day response.

Intend to submit response following plant-specific analysis established by B&W OG (Oconee 1-Dec. 31, 1981, Rancho Seco-Mar. 1, 1982. Others-later).

Will respond to II.K.2.13 by mid June 1982. More detailed fracture mechanics data available in 1982. Remarks

Staff will continue effort which may result in specifying conservatisms where more definitive info. is not available. 3

NRC letter dated 10/22/81 indicated staff would accept what licensee will provide. Staff will continue effort which may result in specifying conservatisms where more definitive info. is not available.

NRC ltr. dtd. 10/22/81 confirms that licensee anticipates no conflicts in providing 60 day and 150 day response.

NRC ltr. dtd. 10/22/81 indicated that the staff will use Oconee 1 data as it is applicable to TMI-1. Staff encourages licensee to submit info. as it becomes available. Remainder of ltr. similar to to response for Calvert Cliffs 1.

NRC ltr. dtd. 10/23/81 indicated staff would favor the response according to the licensee's licensee's proposed schedule. The lack of timely information on requested information could result in conservatisms in the staff conclusions.

. .

5

19

4.

lable A-2	Summary of	60 day	responses	to NRC	letters	dated	August	21,	1981.	concerning	thermal	shock t	o RPV
					November	r 10,	1981			•			

Plant Name (Date of 60 day response)	(1) Initial & Present RT for plates & welds	(2) Rate of RT _{NDT} increase for plates & welds	(3) RT _{NOT} limit for continued oper.	(4) Basis for RT _{NDT} limit	(5) Response concerning operator actions	Remarks
Calvert Cliffs 1 (10/20/81)	Limiting RT _{NDT} Values <u>Initial Values</u> <u>Plate</u> 20° <u>Welds</u> ° circ20° long. 10° <u>e 4.77 EFPY (12/31/81)</u> <u>Plate</u> 92° <u>Welds</u> circ. 146° long. 178° <u>Peak ID Fluence</u> 7.05x1018 n/cm2	Licensee provided RT _{NDT} values for 7.97 EFPY (12/31/85) in resp. <u>Plate</u> 115° <u>Welds</u> circ. 194° long. 235° <u>Peak ID Fluence</u> 1.18x1019 n/cm2	Licensee does not consider it appropriate to define an upper limit RT NDT value.	Adoption of $\Re T_{NDT}$ limit would not per- mit consideration of warm prestressing or other factors.	Operators can control feed rate and terminate HPSI to prevent overpressurization. Generic program will review procedures after detailed analyses of transient:	Licensee RT _{NDT} values are well within the staff's estimate.
			-1			

• . •

Ť,

.....

. .

. .

. • •:.

. . ..

3

20

. -

. . .

- --

.

- - -

Table A-2 (Continued)

2

Plant Name (Date of 60 day response)	(1) Initial & Present RT _{NDT} for plates & welds	(2) Rate of RT _{NDT} increase for plate & welds	(3) RT _{NOT} limit for continued oper.	(4) Basis for RT _{NDT} limit	(5) Response concerning operator	Remarks actions
Fort Calhoun (10/20/81)	Limiting RT _{NDT} Values	Licensee provided RT _{NDT}	Licensee does not	Adoption of RT _{NDT}	HPSI throttling and termina-	Licensees RT _{NDT}
	Plate 10° Welds 000 circ. -20° long. -20° @ 5.36 EFPY (12/31/81) Plate 112° Welds 000 circ. 245° long. 255° Peak 10 Fluence - 7.04x1018 n/cm2	(12/31/85) in response. <u>Flate</u> 142° <u>Welds</u> circ. 270° long. 268° <u>Peak ID fluence</u> - 1.12x1019 n/cm2	appropriate to define an upper limit RT NOT value.	mit considerations of warm prestressing.	throttling criteria are provided in emergency transient procedures to prevent repressurization. CE will review procedures & where warranted procedure revisions will be proposed and evaluated. Following this, procedures will be changed and operating staff retrained as necessary.	within the estimates. By letter dtd. 10/23/81 the staff requested properties for archive material as soon as it is available as a supplement to to the 60 day resp.
	RT _{NDT} values are based on generic material proper- ties. Properties for archive material will be provided in 150 day response.	•			-	

.

.

.

Table A-2 (Continued)

Plant Name (Date of 60 day response)	(1) Initial & Present RT _{HDT} for plates & welds	(2) Rate of RT _{NDT} increase for plates & welds	(3) ST _{NOT} limit for continued oper.	(4) Bails for RT _{NDT} limit	(5) Response concerning operator	Remarks actions
Maine Yankee	Limiting RT. KIT Values	Licensee provided RT _{NDT}	Licensee docs not	The program the	Operators are instructed	Licensees RT NOT
(11/2/81)	Initial Values Plate -10° Weld -30°	values for 26 more calendar years & end of life (35 total calendar years) for welds.	consider it appropriate to define an upper limit RT _{NDT} value.	Dicensee is working on considers the many variables involved in the vecsel capabilities.	with procedures to limit repressurization that results frm HPSI operation and removing RC pump operation during transients. Licen-	values are well within the staff's estimates.
	As of 9/30/81 - (43 x 106 MMH electric RTNDT @ ID = 180° Peak ID Fluence 5.4x1018 n/cm2	<u>26 more cal. yrs</u> 300° <u>End of life</u> 295°			see maintains these actions contribute to problem.	
Turkey Point 4	Limiting RT _{NDT} Values	70/EFPY for next 10 yrs	Licensee stated	Licensee stated in	Licensee indicated no	Licensees RT NDT
(10/21/81)	Initial Values Forging 50° Circ. Welds 3° Current Values 5.61 EFPY (9/30/81 Forgings 85° Circ. Welds 193° Peak ID Fluence - 1.1x1019 n/cm2 Fluence @ 1/4 T 6.6x1018 n/cm2	5°/EFPY for remaining life. For forgings, this represents 30° inc. for remaining design lif⊾ of vessel.	in letter dtd. 9/23/81 that response will be delayed until 3/1/82.	letter dtd. 9/23/81 information has been provided in 5/14/81 ltr. Will provide sdditional informar mation with response to (3).	operator action is required for LOCA. Operator action is required within 10 min. for large MSLB. This includes criteria in proce- dures for HPSI termination and throttling AFW. LOCA procedures have similar procedures. Operators are trained in procedures and on simulator.	values are well within the staff's estimates.

٠

.

۰.

· · ·

.

22

ŝ

.

• •

. •

\$

Table A-2 (Continued)

Plant Name (Date of 60 day response)	(1) Initial & Present RT _{NOT} for plates & welds	(2) Rate of RT _{MDT} increase for plate & weld	(3) RT I limit for collinued oper.	(4) Basis for RT _{NDT} limit	(5) Response concerning operator	Remarks actions	
Robinson 2	Limiting RT _{NDT} Values	7°/EFPY for next 10 yrs	Licensee stated	Licensee stated in	Operators are provided in	Licensees RT HIT	
(10/26/81)	Initial Values Plate 46° @ 1/4 T Welds 0° Current Values	5°/EFPY for remaining life. 45° total for plate.	in ltr. dtd. 9/21/81 that responses will be deferred until 150 day response.	<pre>ltr. dtd. 9/21/81 that information has been provided; however, additional informa- tion will be provided in 150 day reparse.</pre>	procedures HPSI termination criteria and FW throttling criteria. HPSI pumps have 1500 psi shutoff heads. Training programs are established.	values are within the estimates.	
	10 Plate 124° 1/4 T Plate 113° 10 Weld 242° 1/4 T Weld 210°						
	Fluence @ ID Plate 1.42x1019 n/cm2						
	Fluence @ ID Weld 1.30x1019 n/cm2						
San Onofre	Limiting RT _{NDT} Values	For Plate 4°/EFPY	Licensee stated	RT _{NDT} should not be	Existing procedures provide	Licensees RT	
<u>(11/4/81)</u>	Initial Values Plate 60° Weld (long.) 0°	For Welds 3°/EFPY	that response will be provided upon completion of W Owners Group work.	used as sole parameter to determine vessel integrity. Such a limit should be qual- ified to the specific method of calcula- tion. Refers to Owners Group report	HPSI termination criteria for LOCA and SLB. Providec no provisions for throttling HPSI. Provided instruc- tion to throttle feedwater. For SLB operator action not required before 10 min. Training programs are pro-	values are well within the staff's estimates.	
	Current Values @ 8.93 EFPY (10/31/81)						
	Plate 222° Weld (long.) 229°			of 5/14/81.	vided. HPSI shutoff head is 1160 psi.		

Table A-2 (Continued)

•

rlant Name (Date of 60 day response)	(1) Initial & Present RT _{NDT} for plates & welds	(2) Rate of RT _{NDT} increase for plates & welds	(3) RT _{NOT} limit for continued oper.	(4) Basis for RT _{NDT} limit	(5) Response concerning operator	Remarks actions
Oconee 1	Limiting RT _{NOT} Values	Licensee provided	Licensee does not	A RT _{NDT} limit would	Emergency procedures require	The licensee's
(10720781)	Initial Values Plate @ Nozzle 60° Weld circ. 20° long. 20° <u>Current Values</u> 5.13 EFPY (10/1/81)	fluence rate of increase for peak fluence and for critical weld location. <u>Peak Fluence Rate</u> - 0.37x1018 n/cm2/EFPY	consider it appropriate to establish an upper limit RT _{NDT} value.	not provide confi- dence to predict toughness of materials and assurance that material with the greatest index is the controlling material for a given analysis.	operator action for control- ling steam line break (over- cooling) and LOCA. These include throttling and termination criteria for HPSI. The operator can take manual control of feedwater systems to limit plant coolidorm	RT _{NDT} values are within the stafi's estimates.
	Plate 89° Weld circ. 145° long. 160° Plate Fluence 1.94x1018 n/cm2	Weld Fluence Rate - 0.33x1018 n/cm2/EFPY				
	Weld Fluence 2.27x1018 n/cm2					

1

Table A-2 (Continued)

2

Plant Name (Date of 60 day response)	(1) Initial & Present RT _{NDT} for plates & welds	(2) Rate of RT _{NDT} increase for plates & welds	(3) RT _{NDT} limit for continued oper.	(4) Basis for RT _{NDT} limit	(5) Response concerning operator	actions Remarks
Three Mile Island 1 (10/23/81)	Limiting RT _{NDT} Values <u>Initial Values</u> Plate @ 1/4 T 40° Welds @ 1/4 T 20° <u>Current Values</u> <u>Plate</u> 83° <u>Weld</u> circ. 177° long. 170° <u>Fluence for Plate</u> 2.3x1018 n/cm2 <u>Fluence for long. welds</u> 1.7x1018 n/cm2	For Plate 6.2° RT _{NDT} /EFPY For Circ Welds 22.8° RT _{NDT} /EFPY For Long. Welds 19.9 RT _{NDT} /EFPY These are the curren ⁺ rates. As plant life increases da/dt decreases.	Use of RT _{NDT} as a limiting parameter for continued opera- tion is not considered appro- priate by licensee.	The owners group will establish a set of parameters that are expected to be other than RT _{NDT} .	B&W Report BAW 1648 Guide- lines have been incorporated in TMI-1 Emergency Proce- dures. For SBLOCA procedures provide for HPSI throttling (termination) criteria and feedwater control criteria. Training on these procedures is a part of operator training and retraining program.	The licensee's RT _{NDT} values for longitudinal welds are slightly higher than the staff's estimates (10°).

X

.

Plant (NSSS vendors)	Response to letters and gen. contents	Conclusions	Limiting transients	Criteria of accept.	War n prestressing considered	Operator actions considered actions	Remedial Actions
Robinson 2 1/25/82 (<u>W</u>) Referenced WCAP 10019	 Irradiation data Weld material info. Basis for continued operation Operator actions Remedial actions 	31 cal. yrs. of vessel life remaining for all transients considered.	Rpt. provided a table transients considered which include following: 1. Large LOCA 2. SBLOCA 3. LSLB *. SSLB 5. Rancho Seco	Min. flaw depth for crack initiation is greater than 1.0 in. Crack arrest occurs within 75% of vessel	Yes, for all transients considered.	Refers to WCAP 10019. Credit is taken for LSLB. AFW terminated HPSI terminated in 10 min.	 Will have low leakage core Will keep abreast on annealing developments. Studying benerfits of heating RWST. Verification analysis by EPRI.
Turkey Pt. 4 1/21/82 (<u>W</u>) Referenced WCAP 10019	 Irradiation information Weld material info Transient fracture analysis show ing basis for continued operation 	Reactor vessel integrity will be maintained throughout design life.	Rpt. provided a table of transients consid- ered which include following: 1. Large LOCA 2. SBLOCA 3. LSLB 4. SSLB 5. Rancho Seco	Min. flaw depth for crack initiation is greater than 1.0". Crack arrest occurs within 75% of vessel wall thick.	Yes, all transients except SSLB	Cannot determine but WCAP 10019 provides following: Control AFW	Since integrity has been demon- strated, no need for action plan.

~

.

\$

• • •

Table A-3 Summary of "150 day" responses concerning PTS

26

•.

. · ·

•

Table A-3 (Continued)

Plant (NSSS vendors)	Res; to gen	ponse letters and . contents	Conclusions	Limiting transients	Criteria of accept.	Warm prestressing considered	Operator actions considered actions	Remedial Actions
San Onofre 1/25/82 (<u>W</u>) Referenced WCAP 10019	1. 2. 3. 4. 5.	Irradiation effects Material property info Basis for continued operation Operation actions Remedial Actions	Reactor vessel integrity will be maintained beyond design lifetime.	Rpt. provided a table of transients consid- ered which include following: 1. Large LOCA 2. SBLOCA 3. LSLB 4. SSLB 5. Rancho Seco	Min. flaw depth for crack initiation is greater than 1.0". Crack arrest occurs within 75% of vessel wall thick.	Yes, for large and small LOCAs only.	For LSLB Terminate HPSI Terminate AFW to faulted SG. For SSLB Isolate break (PORV) Terminate HPSI	Plan for remedial actions not warranted. Low leakage core is place.
Ft. Calhoun 1/18/82 (CE)	1. 2. 3. 4.	Thermal-Hydro Eval. (a) SLB (b) Overcool- ing (anti- pated occurences) Fracture Mech Analy. for SLB Response to Dec. 18 ltr. Fluence data	Integrity will be maintained for lifetime of plant.	MSLB most limiting. Overcooling A00- stuck open dump valve. (SBLOCA + LOFW analyzed in CEN 189)	For MSLB (low prob- ability) - crack arrest. For A00 + Single failure - crack arrest For A00 - no crack initiation	Benefit from W.P. not considered, however, it was not needed. It would have been credited if needed and criteria met.	Yes For MSLB - 30 min. to reduce HPSI flow. For MSLB - trip RC pumps in 30 seconds. For A00 trip RCP in 10 min. Reduce HPSI in 90 min.	 Will implement reduced radial leakage fuel scheme in Cycle B. Will study othe fuel arrange- ment schemes Do not plan increase in ECC water temp. Evaluating annealing. Program plan will evaluate

۰.

.

÷

will evaluate control systems, procedures & potential design mods.

2

27

مريا والمودر بريوا مواد

Table A-3 (Continued)

Plant (NSSS vendors)	Response to letters and gen. contents	Conclusions	Limiting transients C	Criteria of accept.	Warm prestressing considered	Operator actions considered actions	Remedial Actions
Maine Yankee 1/21/82 (CE)	APP A - response to 4-150 day questions APP B - response to RFI of 8/21/81 APP C - response to 12/18/81 ltr. May do further RETRAN analyses	Vessel will retain integ- rity throughout design life.	MSLB most limiting (cooldown below 300°)	No crack initiation. Response references CEN 189 Report. Prob. of MSLB is very low.	Benefit from W.P. not considered, however, it was not needed. It would have credited if needed and criteria met.	Yes <u>For MSLB</u> Trip RCP @ 30 sec. Terminate HPSI @ 30 min. <u>For A00</u> Trip RCP @ 10 min. Terminate HPSI @ 90 min.	 Low leakage fuel mgmt. for Cycle 7. Will operate RMS to main- tain higher temp. not to exceed 80^e Will keep informed on annealing. Will evaluate control stra- tegy after plant-specific evaluation is in place.
Calvert Cliffs 1/28/82 (Resp. to 12/18/81 ltr 1/21/82) (CE)	 Was responsive Fluence cal. Systems Analysis Fracture mechanics 	No crack initi- ation for assumed plant life for SBLOCA + LOFW. Same for stuck open dump valve (AOO) if needed For MSLB, satis- factory perform- ance for 21 add'1 EFPY	MSLB most limiting, AOO + single failure. SBLOCA + LOFW analysis in CE 189	No crack initiation for AOO Crack arrest for MSLB	Benefit from W.P. not considered, however, it was not needed. It would have been credited and criteria met.	Yes For MSLB Trip RCP @ 30 sec. Reduce HPSI flow @ 30 min. For A00 Trip RCP @ 10 min. Terminate AFW @ 10 min. Reduce HPSI @ 90 min.	 Scoping studie: on fuel mgmt. Do not plan to increase RMST temperature. No discussion on annealing. Control system changes may be considered.

. .

.

\$

Table A-3 (Continued)

3

Plant (NSSS vendors)	Res to gen	ponse letters and . contents	Conclusions	Limiting transients	Criteria of accept.	Warm prestressing considered	Operator actions considered actions	Remedial Actions
Oconee 1 1/15/82 (B&W)	1.	Overcooling transient analysis##	Vessel failure is not calcu- lated to result from postulated transient.	SBLOCA + LOFW Overcooling transient	Crack initiation with arrest within 1/4 T	Yes, for SBLOCA. No for over- cooling transient	<u>SBLOCA + LOFW</u> Trip RCP. Throttle HPIS @ 93 min.	1. 18 month fuel cycle provides decrease in leakage flux. 2. Current water
	2. 3.	SBLOCA analysis Mixing	With minimal downcomer mix- ing, no credit				Overcooling Transient Trip RCP. Isolate EFWS @ 20 min.	temperature sufficient. 3. In-place
	4.	analysis Vessel wall thermal analysis	for mixing in hot leg, no credit for W.P 16 EFPY. With				Except MSLB - isolate all feedwater within 5 min. Only assumed above actions where	annealing not required. 4. No control system changes
	5.	Material	credit for W.P., for SBLOCA - 32					are necessary.
	6.	Fluence	EFPY. For over-	t			necessary to mitigate consequences and	
	7.	Fracture mechanics analysis	25 EFPY (Design life - approx. 27 EFPY)	-			achieve acceptable EFPY.	
	8. 9.	Frequency determination SLB analysis	•					
	**T 5 0 t	urbine bypass ystem failures, verfill ransients						

 $^{\circ}$

. .

Table A-3 (Continued)

Max mar

1

Plant	Responses and General Contents	Conclusions	Limiting Transients	Criteria of Acceptance	Prestressing Considered
TNI-1 7/7/82 (BAH)	 Transient Analysis a) Experience b) Overcooling Transient c) SBLOCA Mixing analysis Vessel Wall Thermal Analysis Haterial Properties Fluence Determination Fracture Mechanics Analysis 	Vessel integrity for SBLOCA & stuck open TBV is maintained at 32 EFPY and longer. Assum ed 1/4 T flaw-no defect growth.	1. S8LOCA + LOFW 2. 6 Stuck Open	1. No crack initiation during 2. Crack initiation with arrest	No

•

. *•

Operator Action Considered	SBLOCA & LOFW	Overcooling Transient	Remedial Actions		
	1. Trip RCP within 2 min. of ESFAS 2. Throttle HPI @ 180 min.	 Trip RCP within 30 sec. of HPI Throttle HPI and isolate TBV within 12 min. 	 Currently investigating 18 month low leakage fuel cycl Initiating a program to reinforce operator awareness o PTS through review of procedures. 		

•

.

•

.

•

147

•

3

•

•

Tabl	le /	\-4	Summary	of	generic	reports	concerning	PTS
------	------	------------	---------	----	---------	---------	------------	-----

Winers Group	General Contents	Conclusions	Limiting Transients*	Criteria of Acceptance	Warm Prestressing Considered	Operator Action Consider	Potential Remedial Action
Mestinghouse MCAP 10019 December 1981 "Summary Report on Reactor Vessel Integrity for Mestinghouse Operating Plants"	 Limited transient development Fluence Calc. Stress & Fracture Mechanics for Transients Vessel Integrity Evaluations Potential Remedial Actions Conclusions (for each operating plant) Don't address iden- fication of events causing highest PTS risk 	All plants can continue operation a number of yrs (3 for the least) before acceptance criteria is violated. A table pro- vides no. of yrs. for each plant. Eight plants are 5 yrs or less	**1. Small Steam Line Break 2. Rancho Seco 3. Large Steam Line Break 4. Small LOCA 5. Large LOCA *In order of severity. **Most limiting.	 No initiation of flaws less than 1 inch deep. (Flaws J 1 in. deep not assumed to exist) or Crack Arrest occurs within 75% of wall thickness. 	Benefit of W.P. considered for S8LOCA and some large LOCA and large SL breaks. Benefit was not considered for other transients.	Yes-Control AFW Trip RCPs as examples - Rept. is not very definitive.	 Heating RWST to 80° - provide of 3 to 30 EFPY operation. Limit AFW Ccatrol Systems to mitigate transient a. RC Press. Relief System Safety Injec- tion Control AFW Control Core Modifica- tions Low leakage loading Annealing Vessel

,

-

•

a. Is feasible

1

.

¢

.

. •

Table A-4 (Continued)

2

.

Owners Group	General Contents	Conclusions	Limiting Transient.	Criteria of Acceptance	Warm Prestressing Considered	Operator Action Consider	Potential Remedial Action
Combustion Engineering CEN-189 "Evaluation of Pressurized Thermal Shock Effects Due to Small Break LOCAs with Loss of Feedwater for CE NSSS"* December 1981 *(This is the Post-TMI "feed & bleed" rept. It is not a Generic PTS report.)	 Only addresses SBLOCA with loss of all FW transient Thermal Hydro- analysis Discussions on mixing Additional studies are expected to per- mit removal of certain conservatisms Scoping studies indi- cate range of HPSI flows must be considered Fluence Calculations Material Properties Vessel Integrity Evaluation Plant-Specific Analysis 	Each plant's vessel can safely with- stand SBLOCA + LOFW for design life without crack initiation.	1. Only com- siders SBLOCA + LOFW **(Note that MSL break is most limiting but was only considered in the 150 day responses)	 No initiation of flaws of credible size, or if it does initiate Arrest after limited extension. 	Benefit of WP was considered	Yes: 1. PORVs opened in 10 min. 2. AFW reinstated after 30 min.	None considered.

REFERENCES

- NRC Memorandum dated April 7, 1981, from T. J. Walker to S. S. Pawlicki -Minutes of PWR Owners Groups Meeting with NRC on March 31, 1981.
- NRC Letters dated April 20, 1981 from D. G. Eisenhut to Licensees of Operating PWR Nuclear Power Plants - Thermal Shock to Reactor Pressure Vessels (Generic Letter 81-19).
- Summary of Meetings with the Babcock & Wilcox, Westinghouse, and Combustion Engineering Owners Groups on July 28, 29 and 30, 1981, Respectively, Concerning Pressurized Thermal Shock to Reactor Pressure Vessels - dated August 18, 1981.
- Summary of Meeting with the Westinghouse Owners Group of September 18, 1981 Concerning Pressurized Thermal Shock to Reactor Pessure Vessels dated October 1, 1981.
- Summary of Meeting with the B&W Owners Group on September 22, 1981 Concerning Pressurized Thermal Shock to Reactor Pressure Vessels - dated October 1, 1981.
- Summary of Meeting with the Combustion Engineering Owners Group on October 7, 1981 Concerning Pressurized Thermal Shock to Reactor Pressure Vessels - Dated October 21, 1981.
- Summary of Meeting with the Westinghouse Owners Group and the Three Selected Owners on February 24, 1982 Concerning the Pressurized Thermal Shock Issue - dated March 8, 1982.
- Summary of Meeting with Combustion Engineering Owners Group and the Three Selected Owners of CE Designed Plants on March 3, 1982 Concerning the Pressurized Thermal Shock Issue - dated March 12, 1982.

- Summary of Meeting with Duke Power Company on March 24, 1982 Concerning the Pressurized Thermal Shock Issue for Oconee Unit No. 1 - dated October 31, 1981.
- Summary of Meeting with Omaha Public Power District on May 6, 1982
 Concerning the Pressurized Thermal Shock Issue dated May 13, 1982.
- Summary of Meeting with the Westinghouse Owners Group on May 10, 1982
 Concerning the Pressurized Thermal Shock Issue dated May 21, 1982.
- 12. Summary of Meeting with GPU Nuclear on June 2, 1982 Concerning the PTS issue for TMI-1 dated June 16, 1982.
- Summary of Meeting with PWR Industry Representatives on June 9, 1982 Concerning the PTS issue.
- Summary of Meeting with WOG on Reactor Vessel Integrity on June 22, 1982 Concerning the PTS issue - dated June 30, 1982.
- 15. Summary of Meeting with CEOG on June 23, 1982 Concerning the PTS issue dated July 8, 1982.
- 16. Summary of Meeting with the WOG on July 30, 1982 Concerning the PTS issued.
- 17. NRC Letters dated August 21, 1981, from Darrell G. Eisenhut to eight selected utilities (Florida Power & Light Company, Carolina Power & Light Company, Southern California Edison Company, Baltimore Gas & Electric Company, Omaha Public Power District, Maine Yankee Atomic Power Company, Duke Power Company and GPU Nuclear Corporation) Concerning Pressurized Thermal Shock to Reactor Pressure Vessels.
- NRC Letter dated October 2, 1981 from Mr. T. M. Novak, NRC, to Mr. A. E. Lundvall, Jr., Baltimore Gas & Electric Company, Concerning Responses to the NRC August 21, 1981 Letter.

- 19. NRC Letter dated October 26, 1982, from Mr. T. M. Novak, NRC, to Mr. W. C. Jones, Omaha Public Power District, Concerning Responses to NRC Letter dated August 21, 1981.
- 20. NRC Letter dated October 23, 1981 from Mr. T. M. Novak, NRC, to Mr. Robert H. Groce, Maine Yankee Atomic Power Company, Concerning Responses to NRC Letter dated August 21, 1981.
- 21. NRC Letter dated October 26, 1981 from Mr. T. M. Novak, NRC, to Mr. J. A. Jones, Carolina Power & Light Company, Concerning Responses to NRC Letter dated August 21, 1981.
- 22. NRC letter dated October 26, 1981 from Mr. T. M. Novak, NRC, to Dr. Robert E. Uhrig, Florida Power & Light Company, Concerning Responses to NRC Letter dated August 21, 1981.
- 23. NRC Letter dated October 23, 1981 from Mr. Gus C. Lainas, NRC, to Mr. R. Dietch, Southern California Edison Company, Concerning Responses to NRC Letter dated August 21, 1981.
- 24. NRC Letter dated October 23, 1982 from Mr. T. M. Novak, NRC, to Mr. William O. Parker, Jr., Duke Power Company, Concerning Responses to NRC Letteer dated August 21, 1981.
- 25. NRC Letter dated October 23, 1981 from Mr. T. M. Novak, NRC, to Mr. Henry D. Hukill, Metropolitan Edison Company, Concerning Responses to the NRC Letter dated August 21, 1981.

26.-

33. NRC Letters dated December 18, 1981, to Maine Yankee Atomic Power Company, Baltimore Gas & Electric Company, Omaha Public Power District, Carolina Power & Light Company, Florida Power & Light Company, Southern California Edison Company, Duke Power Company, and Metropolitan Edison Company - Provided Evaluations of the "60 day" Responses to the NRC Letter dated August 21, 1981, and Requested Additional Information to be Provided in the "150 day" Responses.

- 34. NRC Letter dated March 15, 1982 to Southern California Edison Company Concerning Request for Information Related to their "150 day" Response.
- 35. NRC Letter dated March 16, 1982, to Carolina Power & Light Company Concerning Request for Information Related to their "150 day" Response.
- 36. NRC Letter dated March 16, 1982, to Florida Power & Light Company Concerning Request for Information Related to their "150 day" Response.
- 37. NRC Letter dated March 16, 1982 to Mr. Oliver Kinglsey, Chairman of WOG, Concerning Request for Information Related to W Generic Program on PTS.
- 38. -
- 40. NRC Letters dated March 18, 1982 to Maine Yankee Atomic Power Company, Omaha Public Power District and Baltimore Gas & Electric Company Concerning Requests for Information Related to their "150 day" Responses.
- 41. NRC Letter dated April 4, 1982 to Duke Power Company Concerning Request for Information Related to their "150 day" Responses.
- 42. B&WOG Letter dated May 12, 1981 from John J. Mattimoe, Chairman B&WOG, to Harold Denton, NRC, Concerning Report on Reactor Vessel Brittle Fracture Concerns in B&W Operating Plants.
- 43. CEOG Letter dated May 15, 1981 from Mr. K. P. Baskin, Chairman CEOG, 'o Mr. Darrell G. Eisenhut, NRC, Concerning Reactor Vessel Pressurized Thermal Shock.
- 44. WOG Letter dated May 14, 1981 from Mr. Robert W. Jurgensen, Chairman OG, to Mr. D. G. Eisenhut, NRC, Concerning Thermal Shock to Reactor Pressure Vessel.

- 45. Baltimore Gas & Electric Company Letter dated September 24, 1981 to NRC Concerning 30 day Response for Calvert Cliffs 1 to August 21, 1981 Letter.
- 46. Omaha Public Power District Letter dated September 22, 1981 to NRC Concerning 30 day Response to August 21, 1981 Letter for Fort Calhoun.
- 47. Maine Yankee Atomic Power Company Letter dated September 29, 1981 to NRC Concerning 30 day Response to August 21, 1981 Letter for Maine Yankee.
- 48. Florida Power & Light Company Letter dated September 23, 1981 to NRC Concerning 30 day Response to August 21, 1981 Letter for Turkey Point 4.
- 49. Carolina Power & Light Company Letter dated September 21, 1981 to NRC Concerning 30 day Response to August 21, 1981 Letter for Robinson 2.
- 50. Southern California Edison Company Letter dated October 5, 1981 to NRC Concerning 30 day Response to August 21, 1981 Letter for San Onofre 1.
- 51. Metropolitan Edison Letter dated October 1, 1981 to NRC Concerning 30 day Response to August 21, 1981 Letter for TMI-1.
- 52. Baltimore Gas & Electric Company Letter dated October 20, 1981 to NRC Concerning 60 day Response to August 21, 1981 Letter for Calvert Cliffs 1.
- 53. Omaha Public Power District Letters dated October 20 and November 12 and 13, 1981, to NRC Concerning 60 day Response to August 21, 1981 Letter for Fort Calhoun.
- 54. Maine Yankee Atomic Power Company Letter dated November 2, 1981 to NRC Concerning 60 day Response to August 21, 1981 Letter for Maine Yankee.
- 55. Florida Power & Light Company Letter dated October 21, 1981 to NRC Concerning 60 day Response to August 21, 1981 Letter for Turkey Point 4.

56. Carolina Power & Light Company Letter dated October 26, 1981 to NRC Concerning 60 day Response to August 21, 1981 Letter.

4

- 57. Southern California Edison Company Letter dated November 4, 1981 to NRC Concerning 60 day Response to August 21, 1981 Letter for San Onofre 1.
- 58. Duke Power Company Letter dated October 20, 1981 to NRC Concerning 60 day Response to August 21, 1981 Letter for Oconee 1.
- 59. Metropolitan Edison Company Letter dated October 23, 1981, to NRC Concerning 60 day Response to August 21, 1981 Letter for TMI-1.
- 60. Baltimore Gas & Electric Company Letters dated January 21 and 28, 1982 to NRC Concerning Request for Information dated December 18, 1981 and 150 day Response to August 21, 1981 Letter, Respectively, for Calvert Cliffs 1.
- 61. Omaha Public Power District Letter dated January 18, 1982 to NRC Concerning 150 day Response to August 21, 1981 Letter for Fort Calhoun.
- 62. Maine Yankee Atomic Power Company Letter dated January 21, 1982 to NRC Concerning 150 day Response to August 21, 1981 Letter for Maine Yankee.
- 63. Carolina Power & Light Company Letter dated January 25, 1982 to NRC Concerning 150 day Response to August 21, 1981 Letter for Robinson 2.
- 64. Florida Power & Light Company Letter dated January 21, 1982 to NRC Concerning 150 day Response to August 21, 1981 Letter for Turkey Point 4.
- 65. Southern California Edison Company Letter dated January 25, 1982 to NRC Concerning 150 day Response to August 21, 1981 Letter for San Onofre 1.
- 66. Duke Power Company Letter dated January 15, 1982 to NRC Concerning 150 day Response to August 21, 1981 Letter for Oconee 1.

- 67. WOG Letter dated December 30, 1981 to NRC from O. D. Kinglsey, Chairman WOG, Concerning WCAP-10019 "Summary Report on Reactor Vessel Integrity for Westinghouse Operating Plants."
- 68. CEOG Letter dated December 32, 1981 to NRC from K. P. Baskin, Chairman CEOG, Concerning CEN-189 "Evaluation of Pressurized Thermal Shock Effects Due to Small Break LOCAs with Loss of Feedwater for Combustion Engineering NSSS."
- 69. WOG Letter dated May 28, 1982 to NRC from Mr. O. D. Kinglsey, Chairman WOG, Concerning Response to Request of NRC Letter dated March 16, 1982.
- 70. Carolina Power & Light Company Letter dated May 4, 1982 Concerning the NRC Requests for Information dated March 16 and April 20, 1982.
- 71. Florida Power & Light Company Letter dated May 3, 1982 to NRC Concerning Response to NRC Request for Information dated March 16, 1982.
- 72. Southern California Edison Company Letter dated May 25, 1982 to NRC Concerning Response to NRC Request for Information dated March 16, 1982.
- 73. WOG LEtter dated June 16, 1982 from O. D. Kingsley, Chairman WOG, to
 H. R. Denton, NRC, Concerning Report "Fuel Management to Reduce Neutron Flux."
- 74. WOG Letter dated June 22, 1982, from O. D. Kinglsey, Chairman WOG, to
 H. R. Denton, NRC, transmitting report "Review of Emergency Response
 Guidelines Relative to PTS."
- 75. Baltimore Gas & Electric Company Letter dated May 4, 1982 to NRC in Response to NRC Request for Information dated March 18, 1982.
- 76. Omaha Public Power District Letter dated April 30, 1982 to NRC in Response to NRC Request for Information dated March 18, 1982.

77. Maine Yankee Atomic Power Company Letter dated May 11, 1982 to NRC in Response to NRC Request for Information dated March 18, 1982.

-

- 78. CEOG Letter dated June 14, 1982 from Ken Baskin, Chairman CEOG, to H. R. Denton, NRC, Concerning the Proposed NRC recommendations on the PTS issue.
- 79. Omaha Public Power District Letter dated June 28, 1982 to NRC as a result of the meeting of June 23, 1982 with the staff.
- 80. Duke Power Company Letter dated April 30, 1982 in Response to NRC Request for Information dated April 5, 1982.
- 81. GPU Nuclear Corporation Letter dated March 17, 1982 to NRC Concerning 150 day Response to August 21, 1981 Letter for TMI-1.
- 82. GPU Nuclear Letter dated June 1, 1982 to the NRC Concerning a summary of the "150 day" response to the August 21, 1981 letter and request for information letter dated December 18, 1981.
- 83. B&WOG Letter dated June 22, 1982 from A. P. Rochino, Chairman of B&WOG, to H. R. Denton, NRC, Concerning the staff proposed recommendations on the PTS issue.
- 84. Duke Power Company Letter dated June 21, 1982, Concerning the staff's proposed recommendations on the PTS issue.
- 85. Arkansas Power & Light Company Letter dated June 21, 1982 Concerning the staff's proposed recommendations on the PTS issue.
- 86. Florida Power Corporation Letter dated June 28, 1982 Concerning the staff's proposed recommendations on the PTS issue.
- 87. GPU Nuclear Letter dated July 7, 1982 Concerning the staff's proposed recommendations on the PTS issue.

- 88. SMUD Letter dated June 21, 1982 Concerning the staff's proposed recommendations on the PTS issue.
- 89. NRC Internal Memorandum from G. R. Maetis, Chairman of Robinson PTS Task Force, to H. L. Thompson, Acting Director of DHFS, dated April 15, 1982 concerning staff audit of Robinson 2 procedures and training for PTS.

90.-

- 95. NRC Letters dated April 26, 22, 22, 22, 22 and 23 to Florida Power & Light Company, Maine Yankee Atomic Power Company, Baltimore Gas & Electric Company, Omaha Public Power District, Southern California Edison Company, and Duke Power Company requesting cooperation in the audit and evaluation effort on plant procedures and operator training related to PTS.
- 96. NRC Letter dated April 20, 1982 confirming the Carolina Power & Light Company's commitment to the recommendations of the NRC Robinson 2 Task Force on PTS.
- 97. Carolina Power & Light Company Letter dated May 4, 1982 confirming the utility's commitment to the recommendations of the NRC Robinson 2 Task Force on PTS and providing additional information which was requested in the NRC March 16, 1982 letter.
- 98. Pacific Northwest Laboratory's Letter dated June 4, 1982 Concerning the review of the procedures and training for PTS at Oconee 1.
- 99. NRC Summary of June 8-10, 1982 meeting with OPPD regarding the procedures and operator training relative to the PTS issue - dated June 16, 1982
- 100. NRC summary of July 30, 1982, meeting with WOG regarding SBLOCA which result in stagnation flow, frequencies of such events and differences between staff and WOG concerning the fracture mechanics analyses dated August 9, 1982.
101. NRC Summary of August 11, 1982 meeting with WOG to resolve differences between the staff and WOG concerning the screening criteria dated August 20, 1982.

4

- 102. GPU Nuclear letter dated July 7, 1982, to NRC transmitting the TMI-1 report concerning PTS (the "150 day" response to the NRC August 21, 1981, letter).
- 103. WOG letter dated July 15, 1982, from O.A. Kingsley, Chairman of WOG to H.R. Denton, NRC, concerning the June 22, 1982 meeting with the staff on the PTS issue.
- 104. WOG letter dated September 2, 1982, from O.P. Kingsley, Chairman WOG to H.R. Denton, NRC concerning the July 30 are August 11, 1982, meetings with the staff on the PTS issue.
- 105. WOG letter dated September 16, 1982, from O.D. Kingsley, Chairman WOG to H.R. Denton as a follow-up to the letter dated September 2, 1982.

APPENDIX B

· ••, 6 - •

CHRONOLOGY OF EVENTS CONCERNING PRESSURIZED THERMAL SHOCK

December 31, 1980 B&W licensees submitted Thermal Mechanical Report (BAW-1648).

March 31, 1981 - NRC meeting with PWR Owners Groups concerning thermal shock with repressurization issue. Owners Groups committed to a report by May 15, 1981 to put thermal shock issue into perspective.

April 20, 1981 letter to all operating PWR licensees requesting Owners Groups Reports by May 15, 1981 and licensee's responses by May 22, 1981.

May 4, 1981 - Commission Information Paper (SECY-286 (2B)).

May 12, 1981 - Board Notification.

May 15, 1981 - Received May 15 reports from Owners Groups.

May 19, 1981 - ACRS subcommittee meeting to discuss Owners Groups responses.

May 28-June 4 - Received responses from all operating PWR licensees.

June 5, 1981 - ACRS Briefing.

June 11, 1981 - Commission Briefing.

July 28, 29, 30, 1981 - Meetings with Babcock & Wilcox, Westinghouse and Combustion Engineering Owners Groups.

September 15, 1981 - Commission Briefing.

August 27, 1981 - NRC letter to eight licensees of eight operating PWR plants (Fort Calhoun, Robinson 2, San Onofre, Maine Yankee, Turkey Point 4, Calvert Cliffs 1, TMI-1 and Oconee 1) requesting 60-day response and 150-day response concerning Thermal Shock.

12

September 21 through October 5, 1981 - Received letters from 7 licensees in response to the August 21, 1981 letter identifying conflicts with the request.

September 18, 1981 - Meeting with Westinghouse Owners Group.

September 22, 1981 - Meeting with Babcock & Wilcox Owners Group.

October 7, 1981 - Meeting with Combustion Engineering Owners Group.

October 23-28, 1981 - Letters to eight licensees regarding their exceptions to the August 21, 1981 letter.

October 20 through November 13, 1981 - "60 day" responses from the eight licensees who received the August 21, 1981 letter.

December 8, 1981 - Commission Paper SECY 81-687 dated December 8, 1981, Subject: Designation of PTS as an Unresolved Safety Issue.

December 18, 1981 - NRC evaluations and request for information concerning "60 day" response.

December 30, 1981 - Westinghouse Owners Group Report Concerning Pressure Vessel Integrity.

December 31, 1981 - Combustion Engineering Owners Group Report concerning (MI Action Item II.K.2.13.

January 15 through January 25, 1982 - "150 day" responses to August 21, 1931 letter from seven utilities. GPU did not submit a "150 day" response for TMI-1. February 24, 1982 - Meeting with WOG to discuss the WOG generic report and the "150 day" responses of three \underline{W} plants of PTS concern.

. . . .

March 3, 1982 - Meeting with CEOG to discuss the CEOG generic report and the "150 day" responses of the three CE plants of PTS concern.

March 5, 1982 - Commission Information Paper (SECY 82-97) Subject: Commission Briefing on PTS.

March 9, 1982 - Commission Briefing, Status Report on PTS.

13

•

March 16, 1982 - Appointment of Special Task Groups to (1) investigate the reducing of irradiation damage to vessels, and (2) audit the operator training and procedures for the PTS concern at Robinson 2.

March 15, 16, 18 and 24, April 5, 1982 - Letters to seven of the eight licensees of the PTS concerned plants and the WOG requesting additional information related to the "150 day" responses and the generic reports.

March 24, 1982 - Meeting with Duke Power Company to discuss the Oconee 1 "150 day" response.

March 26, 1982 - Transmittal of Task Action Plan for USI A-49, "Pressurized Thermal Shock" (PTS).

March 24, 1982 - Meeting with Duke Power Company concerning the PTS issue for Oconee 1.

April 15, 1982 - Report of special task force on PTS for Robinson 2.

May 20, 1982 - Preliminary Assessment of Techniques for Fluence Rate Reduction for PWR Pressure Vessels.

April 30-May 4, 1982 - Received responses from licensees of special plants concerning PTS to NRC request for information during March 1982.

÷.

.

May 6, 1982 - Meeting with OPPD concerning PTS issue. Discussed OPPD's response of April 30, 1982.

e ture

May 10, 1982 - Meeting with WOG concerning the PTS issue. Discussed response of WOG due at end of May.

April 26, 1982 - Licensees of other six special plants of PTS concern requested to cooperate in audits of operating procedures and training.

May 28, 1982 - Received WOG Supplemental Information on Reactor Vessel Integrity.

June 2, 1982 - Meeting with GPU Nuclear concerning PTS, Summary of "150 day" response.

June 3, 1982 - ACRS Meeting - Discussed staff's consideration of possible recommendations for PTS requirements.

June 9, 1982 - Meeting with PWR industry representatives concerning the staff's considerations of possible recommendations for PTS requirements.

June 22, 1982 - Meeting with WOG concerning PTS issue - Followup to June 9, 1982 meeting.

June 23, 1982 - Meeting with CEOG concerning PTS issue - Followup to June 9, 1982 meeting.

June 16, 1982 - WOG report on Fuel Management to Reduce Neutron Flux.

June 22, 1982 - WOG report on PTS review of ERGs.

June 21-July 7, 1982 - Responses from CEOG, OPPD, B&WOG, and licensees of all operating B&W plants concerning staff's consideration of proposed recommendations on PTS requirements.

8-4

July 9, 1982 - Meeting with OPPD concerning PTS issue.

4

July 30, 1982 - Meeting with WOG concerning PTS issue - d'scussed staff's position on PTS issue.

- - -

August 11, 1981 - Meeting with WOG concerning the PTS issue ~ Discussed the differences between the staff and WOG in the evaluation of the critical RT_{NDT} for the limiting transient and provided a discussion of the staff's proposed position regarding a screening criteria.

• بر

APPENDIX C

11 1

PROCEDURES AND TRAINING

C.1 Human Factors Considerations

It was recognized by the task force early in the review of Pressurized Thermal Shock (PTS) that plant operators played a key role in the evaluation and mitigation of PTS events. There are some key considerations that must be evaluated in determining the acceptability of operator action as a mitigative action. The sum of these considerations is the need to provide a well-balanced, integrated approach to accident mitigation that is based on technical analysis and considers PTS in the context of other related concerns, such as core cooling, environmental releases, and containment integrity.

The first consideration is that reactor vessels have been designed to withstand the worst design-basis accident. The consequences of a vessel failure are so significant that we have always required vessel and system design adequate to prevent it. The second consideration is a concern for the ability of the operators to "prevent" PTS from breaking a vessel. Operators in general are excellent throughout the industry. But any human can make errors, both cognitive and operative. The likelihood of error increases with an increase in stress, poor control room design, fatigue, instructions inadequate to deal with the particular sequence in progress, and other similar factors. Because of possible human errors and the potential severe consequences of PTS, the NRC does not consider operator action an acceptable long-term "solution" to the PTS issue.

However, the NRC staff recognizes that there is a genuine need to provide clear, concise, and integrated procedures and training to the operators, to ensure they know the technical issues involved not only for this issue, but for other vital considerations they must be concerned with in plant operations. After the TMI-2 accident, NRC-directed 'enhancements' to HPI termination criteria were developed by the industry. The results of these changes is, as

11/13/82

perceived by the NRC staff, a 'mindset' to maintain HPI flow after an accident at all costs. Corrent analysis of accidents with continuous HPI flow shows that the challenge to vessel integrity is more severe than previously considered. In subsequent evaluations, the staff and the industry have learned that real events, with multiple failures, have led to transient cooldowns more severe than previously analyzed. This led the staff to recognize that a balance of considerations must be used to control the operation of HPI and other safety-related plant equipment. The industry should have a clear understanding of those considerations, including understanding of PTS, in determining the best method of operating plant equipment.

· 167 ·

C.1.1 <u>Westinghouse Plants</u>

The utilities of the three Westinghouse-designed plants being evaluated for PTS provided a list of procedural steps dealing with HPI termination and control of feedwater. These steps were provided in the 60-day responses to D. G. Eisenhut's August 21, 1981 letter to the Eight plants being evaluated for PTS. In response to D. G. Eisenhut's December 18, 1981 letter to the three utilities with Westinghouse-designed plants, additional procedures information was provided at the same time the 150-day response to our August 21, 1981 letter was provided.

At a meeting in February 1982, in Bethesda, Md., Westinghouse presented to the NRC staff an evaluation of the PTS mitigative actions contained in the Westinghouse guidelines, which were developed in response to NUREG-0660 Item I.C.1. The guideline for steam line breaks includes modified HPI termination criteria, to account for vessel integrity considerations, as described in the following: The HPI termination criteria require a level in the steam generator, a level in the pressurizer, adequate subcooling margin, and a minimum pressure. For vessel integrity considerations, the minimum pressure for HPI termination tas been lowered in the steam line break guideline from 2000 psig to 700 psig when primary loop temperature is below 350°F.

A letter from S. Varga to the three Westinghouse licensees dated March 16, 1982 requested evaluations regarding the need and effectiveness of upgrading current procedures, and requesting a formal commitment to upgrade operator

11/13/82

understanding of PTS. Responses from Carolina Power and Light (H. B. Robinson), Florida Power and Light (Turkey Point 4) and Southern California Edison (San Onofre 1) dated May 4, 1982, May 3, 1982, and May 20, 1982, respectively, were received.

C.1.1.1 H. B. Robinson

C.1.1.1.1 Present Procedures

H. B. Robinson's emergency operating procedures were based on the Westingnouse guidelines developed in response to NUREG-0660 Item I.C.1. They include the modified HPI termination criteria for steam line breaks. CP&L stated in their 150-day response that they believe procedures governing operator action and programs governing operator training should provide a balanced approach to handling transients and accidents. Their heatup and cooldown curves are used to define acceptable operation to prevent PTS events. An additional training program on the recent PTS concerns was completed March 31, 1982. H. B. Robinson continues to tie their efforts into the Westinghouse procedures development effort. Modifications to the Robinson procedures are being made, as outlined in Section C.1.1.1.3.

C.1.1.1.2 Present Operator Training

As stated in the previous section, H. B. Robinson believes in a balanced approach to operator training. As described in their 150-day response, CP&L has committed to assuring that each of their operators has a complete understanding of the PTS issue. CP&L stated in a June 25, 1982 letter that operator training has been upgraded as outlined in the staff audit report.

C.1.1.1.3 Plant Audit

On April 5-7, 1982, the procedures and training related to PTS were audited at the site. Some specific changes were recommended to the operating procedules to lower the required minimum pressure for HPI termination and to provide explicit instruction for pressure control during cooldown. More specific training was recommended, to include instruction on previous overcooling

events, walk-throughs of procedures as a shift team, and CP&L evaluation of the shift's ability to cope with a PTS event. In a letter from E. Eury to T. Novak dated May 4, 1982, CP&L committed to address the staff's recommendations and identified other procedure modifications required as the result of Westinghouse's review of the guidelines on which the procedures are based.

14

C.1.1.2 Turkey Point 3

C.1.1.2.1 Present Procedures

Turkey Point Units 3 and 4 emergency operating procedures were developed based on the Westinghouse guidelines developed in response in NUREG-0660 Item I.C.1. As stated in the FP&L 150-day response, they include the modified HPI termination criteria for steam line breaks, and specific direction to terminate HPI when termination criteria are met. Operating pressure-temperature limit curves are included. Emergency operating procedures provide instructions to (1) minimize RCS cooldown rate and (2) prevent repressurization following overcooling. In a letter from R. Uhrig to S. Varga dated May 3, 1982, FP&L committed to modify their procedures based on the information provided in the staff's H. B. Robinson audit report.

Additionally, FP&L stated that other NRC concerns with existing procedures will be resolved in the guidelines (and subsequent procedures) developed in response to NUREG-0737 Item I.C.1. These concerns include both technical and human factors considerations.

C.1.1.2.2 Present Operator Training

As stated in the FP&L 150-day response, pressure-temperature limit curves are presented and discussed in the Licensed Operator Training Program. Simulator training includes handling overcooling transients. In a letter from R. Uhrig to S. Varga dated May 3, 1982, FP&L stated they will be augmenting operator training based on the staff's recommendations in the H. B. Robinson audit report.

C.1.1.2.3 Plant Audit

÷.

On July 13-15, 1982, the procedures and training related to PTS were audited at the site.

C.1.1.3 San Onofre 1

C.1.1.3.1 Present Procedures

San Onofre 1's emergency operating procedures were developed based on the Westinghouse guidelines developed in response to NUREG-0660 Item I.C.1. As stated in Southern California Edison's 60-day response, they include modified HPI termination criteria for steam line breaks, but do not provide specific direction to terminate HPI when termination criteria are met. In a recent procedure modification made for the Systematic Evaluation Program evaluation of steam line breaks, the operators were specifically directed to terminate HPI. Information obtained from the staff's H. B. Robinson audit report is also incorporated into the recently revised procedures.

C.1.1.3.2 Present Operator Training

As stated in their 150-day response, SCE provided formal operator training for PTS during the operator requalification training program conducted in February 1982. Recent format changes to procedures, modified HPI termination pressures, and upgraded knowledge of steam generator tube ruptures have recently been incorporated into the San Onofre 1 emergency operating procedures. These procedures changes will require additional training of the San Onofre 1 operators, prior to startup from their current outage.

C.1.1.3.3 Plant Audit

An onsite audit of the San Onofre Unit 1 procedures and training for PTS was conducted on June 2-4, 1982. Preliminary findings from the audit indicated that the procedures are based on plant-specific analyses of transients and that the operations personnel were familiar with PTS even though their training was not completed at the time of the audit. It was noted that the remainder of

the training program should include instruction on post-cooldown actions. Recommendations regarding the San Onofre 1 procedures are included in the audit report. The procedures were generally found adequate for PTS considerations, and were based on Westinghouse analysis. It was recommended that a method for plotting cooldown rate should be provided to the operators.

C.1.2 Combustion Engineering Plants

The licensees of the three Combustion Engineering (CE)-designed plants being evaluated for PTS provided a description of the procedural actions for dealing with HPI termination and control of feedwater.

These steps were provided in the 60-day response to D. G. Eisenhut's December 18, 1981 letter to the three licensees of CE-designed plants. Additional procedures information was provided at the same time the 150-day responses to the August 21, 1981 letter were provided. At a March 1982 meeting in Bethesda, Md., licensee representatives of CE plants presented to the NRC staff an evaluation of the mitigative actions in the specific plant procedures. A letter from R. Clark to A. Lundvall, Jr., dated March 18, 1982, requested additional in Armation regarding the basis and sensitivity of operator action assumed in the analyses performed for the 150-day responses. Responses from Omaha Public Power District (Fort Calhoun), Baltimore Gas and Electric (Calvert Cliffs) and Maine Yankee Atomic Power Company (Maine Yankee) were received by letters dated April 30, 1982, May 4, 1982, and May 11, 1982, respectively.

C.1.2.1 Fort Calhoun

C.1.2.1.1 Present Procedures

Fort Calhoun's emergency operating procedures include design-basis events to cover the requirements for "10 CFR Part 50 Appendix B, Quality Assurance Criteria for Nuclear Power Plants and Fuel Reprocessing Plants." In OPPD's 60-day response, a detailed list of procedural actions was provided, including explanations of their applicability to PTS. In their 150-day response, Omaha Public Power District, OPPD stated that based on their evaluation of their procedures, and on the analysis performed for the PTS issue, changes to the

Fort Calhoun procedures should be made. These changes included the need to provide specific criteria for HPI and charging termination, and improved cautions to assure operator compliance with cooldown curves. These changes, and any other additional modifications based on the plant's analysis, were to be completed by June 1, 1982. These changes were in place at the time of the audit of the Fort Calhoun procedures and training. Fort Calhoun's procedures also require HPI operation for at least 20 minutes, based on previous NRC requirements.

11

C.1.2.1.2 Present Operator Training

As described in OPPD's 60-day response, the operator training program is part of a two-phase effort. The first phase is performing analysis for PTS. The second phase is determining modifications, if any, that may be necessary, including procedure changes. Retraining of operators will be conducted on the procedure revisions. Instruction on the PTS issue have been conducted for the operators and for all levels of OPPD's management.

C.1.2.1.3 Plant Audit

10

An onsite audit was conducted at Fort Calhoun to determine the level of operator understanding of PTS concerns. This audit was conducted June 7, 1982. Preliminary findings from the audit indicate that the procedures were generally adequate for PTS, and are based on a plant-specific analysis performed by CE. The operators were generally knowledgeable of the PTS issue. Recommendations from the audit team included the upgrading of pressure-temperature curves, and the consolidation of NDT, saturation and subcooling curves onto one plot for more effective utilization of the curves by the operators.

C.1.2.2 Calvert Cliffs

C.1.2.2.1 Present Procedures

Development of Calvert Cliffs' procedures is part of a two-phase program to address PTS. The first phase is the development of analyses for PTS. The second phase involves changing plant operating procedures, if necessary. In BG&E's 60-day response to D. G. Eisenhut's August 21, 1981 request for information, specific procedural actions were provided, related to operation and termination of HPI and charging flow, and control of feedwater. An evaluation of the Calvert Cliffs' procedures has been conducted by plant personnel who feel that the procedures adequately address PTS, considering the risk involved. BG&E considers an integrated, analyzed approach to plant operations, of which PTS is one concern, to be the only reasonable approach to responsible plant operations. As stated in a letter dated May 4, 1982, from R. Bryant to D. Eisenhut, BG&E agrees with the staff that vessel integrity concerns should be properly addressed. Changes to Calvert Cliffs' procedures have been made to remind the operators to observe the vessel integrity-related pressure temperature limits. BG&E stated that they will continue to be involved in the CE Owners' Group efforts for emergency operating procedures upgrades for NUREG-0737 Item I.C.1.

C.1.2.2.2 Present Operator Training

As described in BG&E's 60-day response, operator training will be conducted on any operating changes resulting from the plant's analysis. Operator training based on the changes identified was completed by June 30, 1982.

C.1.2.2.3 Plant Audit

An onsite audit was conducted of the Calvert Cliffs' procedures and training for PTS on July 6-8, 1982. The following changes to the Calvert Cliffs procedures were recommended: (1) provide clearer instructions for preferred methods of accident mitigation, (2) predetermine priorities of mitigative actions, and (3) upgrade procedure cross-references. Recommendations for training improvements included the need for additional training on accident mitigation methods, to include pressure-temperature control in various abnormal conditions (e.g., with and without vessel upper-head bubbles, and with and without forced circulation).

C.1.2.3 Maine Yankee

۴.

C.1.2.3.1 Present Procedures

Maine Yankee Atomic Power Company (MYAPC) provided a summary of their operator actions in their 60-day response to D. G. Eisenhut's August 21, 1981 letter to the eight plants being evaluated for PTS. These actions include criteria for HPI and charging termination and feedwater operation. The report further stated that a maximum subcooling limit was already in the plant's procedures. The subcooling limit is 200°F, and was based on pressurizer overstress concerns.

111

C.1.2.3.2 Present Operator Training

The training program described in the 60-day response included a discussion of operator training on emergency operating procedures, emphasizing maintenance of 50°F subcooling. A more detailed training outline was provided in the 150-day response, and included technical as well as operational information. A schedule, included in the 150-day response, showed that training for operating crews was completed by June 1982, and that RO and SRO trainees would receive training in this area.

C.1.2.3.3 Plant Audit

A review of Maine Yankee's procedures and training for PTS was conducted on May 25-27, 1982. The review team found the plant operations personnel and Shift Technical Advisors adequately knowledgeable of the PTS issue, and the procedures provided adequate guidance for preventing PTS. One significant operating philosophy already in place at Maine Yankee is the throttling of PPI flow to maintain as close to 50°F subcooling as possible during potential cooldown events. It was noted by the review team that no written exam was conducted after the lectures on PTS. Rather, a seminar method was used to determine the level of comprehension. Questions regarding PTS have been included in the written requalification examinations. The review team concluded that the operators were sufficiently knowledgeable of PTS. No changes to the operating procedures or training program were recommended.

C.1.3 Babcock & Wilcox Plants

Oconee Unit 1 is the B&W-designed plant being evaluated for PTS. All analyses performed by B&W are specific to the Oconee plant.

.

C.1.3.1 Oconee

C.1.3.1.1 Present Procedures

Oconee 1's current emergency operating procedures include design-basis events to cover the requirements of 10 CFR Part 50 Appendix B and were developed based on design analysis. In the Duke Power Company 60-day response to D. G. Eisenhut's August 21, 1981 letter to the eight licensees whose plants were being evaluated for PTS, a discussion of the procedural actions related to PTS were provided. The actions discussed included feedwater operation, HPI operation, and instrumentation. Also included was a graph of pressure vs. temperature, with allowable operating regions indicated for conditions with and without reactor coolant pumps operating. In the graph's notes, the operators are instructed to maintain a 50°F to 100°F subcooling band with RCPs off. Duke Power Company stated in their 150-day response that based on the analysis presented in their letter, no major changes in existing plant procedures were considered necessary. The letter also stated that when implemented, the Abnormal Transient Operating Guidelines will include appropriate operator instructions for mitigation of overcooling transients.

C.1.3.1.2 Present Operator Training

As stated in Duke Power Company's 60-day response, the Oconee operators receive instruction on HPI termination and feedwater control during requalification training. Training on plant response and emergency operating procedures is also conducted on the B&W simulator. The 150-day response further stated that Duke Power recognizes the importance of ensuring operators have sufficient training and the procedures are adequate to prevent the occurrence of severe thermal shock events. Additional training to augment operator understanding of PTS is to be conducted, but Duke considers the current knowledge of in-place plant procedures to be acceptable for the short term. Duke also stated that their operators have been made aware of the PTS concern, although no formal training has been conducted.

۰ <u>د</u> ۱

C.1.3.1.3 Plant Audit

4

A review of Oconee's procedures and training for PTS was conducted May 11-13, 1982. In general, the review team found the operators adequately knowledgeable of the PTS issue, except that knowledge of past PTS events at other facilities was weak. The procedures provided mitigative actions to prevent PTS, but needed to be strengthened to provide actions if an unacceptable pressuretemperature condition was reached. The audit team felt that a means should be provided for plotting cooldown rate and subcooling margin with the plant computer out of service.

C.2.0 Conclusions

Technical guidelines for Emergency Operating Procedures (EOPs) are being developed generically by the NSSS vendor owners' groups in response to TM1 Action Plan Item I.C.1, "Short-Term Accident Analysis and Procedures Revision." These guidelines and their supporting analyses will address the actions required for mitigating a wide range of accidents and transients including multiple failures and operator errors. These guidelines will include the operator actions necessary to prevent or mitigate pressurized thermal shock. Incorporation of PTS concerns in the guidelines is beneficial and more effective than current procedures in that the analyses supporting the guidelines will verify that the mitigating actions for PTS do not result in inadequate core cooling or other problems. The Westinghouse Owners' Group has reviewed its existing Emergency Procedures Guidelines and is considering the PTS issue in developing the remainder of these guidelines. This effort was completed in July 1982. The B&W Owners' Group has incorporated desired operating regions in the Anticipated Transient Operating Guideline (ATOG) for Oconee, which take PTS concerns into account. The B&W approach is considered acceptable until the long-term PTS program has been implemented. The CE Owners' Group has submit ed draft Emergency Procedure Guidelines which provide a desired operating range for pressure and temperature. The CE guidelines are presently being reviewed by the staff. Another CE Owners' Group activity deals with verifying the

"correctness" of the actions specified in the guidelines with respect to the PTS issue. The revision of the guidelines to be submitted in August 1982 will include the results of this activity.

The NRC staff recognizes that the owners' groups' efforts on the emergency procedure guidelines would not be completed until late 1982 and staff review will not be completed until early 1983. The staff considers this schedule acceptable considering the low probability of occurrence of PTS events, the past operating history of PTS precursor events, and upgrades in instrumentation reliability resulting from the Rancho Seco and Crystal River events. Nevertheless, the staff has undertaken a program to audit the procedures and training covering pressurized thermal shock at the following plants: H. B. Robinson, Oconee 1, San Onofre 1, Maine Yankee, Fort Calhoun, Turkey Point 3, and Calvert Cliffs 1. The purpose of these audits is to assess the adequacy of current procedural steps and operator training necessary to mitigate PTS events and to determine if corrective actions are required before the longer term PTS program provides acceptance criteria and generic resolution of the issue. Additional plant-specific recommendations or generic recommendations may result from these audits.

Based on the audits conducted to date, the staff concludes that industry operators are generally knowledgeable of the PTS issue and of the mitigative actions for PTS included in their procedures. Further, the procedures reviewed, with some specific exceptions delineated in the reports, provide a scheme for mitigation of PTS events. The procedures are usable for PTS, and can be understood by the operators.

C.2.1 Audit Results

The general conclusion based on review of the seven plant's procedures and training programs was that operators are adequately knowledgeable of the technical issues involved in PTS, and were aware of procedures guidance currently existing for dealing with PTS.

C.2.2 Procedures

The seven plants currently being evaluated for PTS have reviewed their current emergency operating procedures for instructions relevant to the PTS issue.

114

H. B. Robinson's procedures, based on generic Westinghouse guidelines, included in its HPI termination criteria a minimum required pressure of 2000 psig, with HPI shut-off head at approximately 1500 psig. This could have resulted in extended HPI operation when not desired. Based on an NRC audit and the licensee's evaluation, H. B. Robinson has lowered the minimum pressure for HPI termination to 1560 psig, changed the temperature monitoring for operation from the RCS hot leg to the RCS cold leg, strengthened the emphasis on terminating HPI when its termination criteria are met, and provided more detailed instructions on RCS pressure and temperature control. The staff finds the H. B. Robinson procedural guidance adequate for the immediate PTS concern.

Turkey Point 3 procedures, based on generic Westinghouse guidelines, contain specific direction for HPI termination when the criteria are met. Personnel at Turkey Point 3 have reviewed the staff's H. B. Robinson audit report and made changes to Turkey Point's procedures based on the findings from the staff's H. B. Robinson audit report. Based on these commitments the staff finds the Turkey Point 3 procedural guidance adequate for the immediate PTS concern. Further verification was conducted during the onsite audit.

San Onofre 1 procedures, based on generic Westinghouse guidelines and modified for the SEP evaluation, contain directive actions for termination of HPI and the information learned from the H. B. Robinson evaluation. The staff finds the San Onofre 1 procedural guidance adequate for the immediate PTS concern. Findings from the onsite audit are included in the audit report.

Fort Calhoun procedures provide some specific guidance to the operators for operation of HPI, charging, and feedwater. Omaha Public Power District (OPPI) has identified changes necessary to provide criteria for HPI termination to reflect the PTS concern and improved precautions to assure operator compliance with cooldown-based pressure-temperature curves. The staff concurs with the need for these changes. A reevaluation of the requirement for running HPI for at least 20 minutes after initiation should be made by OPPD and the staff. We strongly recommend removing any minimum running time requirements for HPI. A more detailed evaluation of Fort Calhoun procedures was conducted during the onsite audit.

-

Calvert Cliffs' procedures provide some specific guidance to the operators for operation of HPI and feedwater. Cautions and a Technical Specification are intended to provide assurance that HPI or feedwater flow will be terminated prior to vessel challenge. The instructions, by themselves, do not provide the specific guidance the staff feels is necessary. They should include a directive action step for control of HPI. A determination was made during the plant audit that some procedures modifications were necessary for the operators to effectively deal with PTS.

Maine Yankee's procedures provide some specific guidance to the operators for operation of HPI, charging and feedwater, including a subcooling band $(50^{\circ}F)$ minimum, 200°F maximum). Maine Yankee has requested, in their discussions, that the staff reevaluate its position on requiring HPI flow for a minimum of 20 minutes, and on requiring immediate RCP trip after a safety injection. We concur that this needs to be done before PTS can be completely addressed in any plant's procedure. The staff finds the Maine Yankee procedural guidance adequate for the immediate PTS concern.

Oconee Unit 1's procedures provide some specific guidance to the operators for operation of HPI and feedwater. When below 500°F, the operators are instructed to maintain a subcooling band (50°F minimum, 100°F maximum). The operator is specifically directed to throttle HPI when 50°F subcooling is reached. The staff finds the Oconee procedural guidance adequate for the immediate PTS concern.

C.2.3 Training

The seven plants currently being evaluated for PTS have all stated that they are augmenting their operator training for PTS. The staff conclusions regarding individual plants are included in each audit report.

C.2.3.1 Improvements in Emergency Operating Procedures

•

Westinghouse performed an evaluation of procedural actions for PTS by reviewing, step by step, guidelines that have a realistic technical basis. In reviewing the technical basis for each step, a determination could be made of its applicability to the PTS concern. This program shows the importance and viability of an integrated approach to accident mitigation, where new technical problems can be evaluated in a manner that includes incorporation of concerns of other technical issues. Combustion Engineering and Babcock & Wilcox have done a significant amount of work on developing their own approach to generic quidelines. All three owners' groups are developing guidelines to be functionoriented, in accordance with NUREG-0737, Item I.C.1. This approach to accident mitigation will provide a means to significantly reduce operator error by providing mitigative actions that are not dependent on diagnosis of specific transients or accidents. This approach will increase the accuracy of operator response by reducing complex diagnostic problems to a prioritized, simplified, function-level response that will be used even if an event is incorrectly diagnosed.

.:

The staff concurs, and strongly encourages, the approach stated by the seven plants being evaluated to ensure that the guidelines and subsequent plant procedures developed in accordance with NUREG-0737 Item I.C.1 address PTS, as well as coordinate the PTS actions with actions to mitigate other serious transients or accidents. We believe this is the best method to provide an integrated set of emergency operating procedures to deal with a wide range of transients and accidents, and will provide the analytic base for evaluation of future technical problems.

In reviewing industry responses to comply with NUREG-0737 Item I.C.1, the staff will review the technical guidelines for emergency operating procedures (EOPs) and will review a description of how EOPs are developed from the guidelines for each operating plant. This will provide assurance that procedures at each plant will be based on analysis of PTS and other events. This review will be performed for all operating reactors and operating license applicants.

C.3.0 Recommendations

In view of the above described ongoing programs to develop and implement a set of integrated procedures, and considering the low risk due to PTS events from plants below the screening RT_{NDT} (which is where all plants are, currently), we conclude that current procedures and training are acceptable, for the present. However, since PTS risk increases with further irradiation, we make the following recommendations for the future:

1 . 1

- (1) The industry and the NRC staff will ensure that actions to mitigate PTS are included in the technical guidelines developed for NUREG-0737, Item I.C.1. The NRC staff will, in their review of the analyses that form the basis for the technical guidelines, ensure that the actions specified in the guidelines are based on an integrated evaluation of relevant technical considerations, including PTS, core cooling, environmental releases, and containment integrity.
- (2) When plants are determined to be within 3 years of exceeding the staff criteria for RT_{NDT}, those licensees should be required to implement upgraded Emergency Operating Procedures which are based on technical guidelines developed for NUREG-0737, I.C.1, "Guidance for the Evaluation and Development of Procedures for Transients and Accidents." All licensees will be required to implement these upgraded procedures on a schedule developed in accordance with SECY-82-111B.

The procedures developed and implemented by items (1) and (2) above should address the following types of concerns:

- (a) Instructions should include allowance for system response delay times (e.g., loop transport time, thermal transport time).
- (b) The need for cooldown rate limits for periods shorter than one hour should be evaluated.
- (c) Methods for controlling cooldown rates should be provided.

(d) Guidance should be provided for the operator if cooldown rates or brittle fracture limits are exceeded.

.

- (e) The desired region of operation (e.g., subcooling band) on the pressure-temperature curve should be evaluated to determine if it can be revised to maximize the operator's ability to prevent brittle fracture.
- (f) Instructions for controlling pressure following depressurization transients should be provided.

This item should be completed in the same time frame stated in item (1).

- (3) The staff recommends that the initial training on the procedures developed from the guidelines discussed in recommendation 1 and 2 above include a specific section on the technical concerns of PTS, and the specific manner in which the procedures provide the mitigative actions. This training should be integrated into each plant's overall training program.
- (4) The staff recommends that training programs for periodic operator regualification include the recommendations of item (3) above.

This item should be implemented at the first requalification training cycle following implementation of the upgraded procedures.

The staff feels that these recommendations provide balanced approach to ensure the adequacy of operator response to PTS events. This is accomplished by determining the adequacy of operator understanding at the plants of most concern, then providing for all plants the best available means to ensure the procedures used for plant operation cover a wide range of transients and accidents, while covering a wide variety of multiple failures.

APPENDIX D

REACTOR VESSEL FRACTURE MECHANICS ANALYSIS

The vessel integrity analyses, the results of which are reported in this document, include a determination of the temperature distribution across the vessel wall versus tire, the thermal stresses as a consequence of this temperature distribution, as well as fracture mechanics results. The analyses were performed either by the NRC staff using its in-house program or by ORNL using the OCA program. These programs are described in the following sections. Illustrations of typical temperature, stress and stress intensity factor distributions across the vessel wall at a certain time in the transient are shown in Figures D-1 (a), (b) and (c) respectively. It should be noted in Figure D-1 (c) that the stress intensity factor, $K_{\rm I}$, for long axial cracks is higher than for long circumferential cracks, especially for cracks that extend relatively deep into the vessel wall. K_T is due to contributions from thermal stresses, pressure stress and other stresses that may be present. Superposed in Figure D-1 (c) are K_{Ic}, the vessel toughness that determines crack initiation, and K_{Ia} , the toughness at crack arrest. When K_{I} exceeds K_{Ic} , crack initiation is expected (for axial cracks having depths between points C and C^{\perp} in the diagram), if warm prestressing is not effective (warm prestressing is discussed in D.3). The crack would then grow to a depth where K_{T} intercepts the arrest curve, K_{Ia} (point A in the diagram). Similar results would occur for a circumferentially oriented crack except that arrest will generally occur at the shallower depths.

Equivalent calculations are made at other times into the transient and the results cross-plotted on a critical crack depth diagram as shown on Figure D-2 (b). Also shown in Figure D-2 (b) is the depth at which the upper shelf toughness of the metal is reached (nominally 200 ksi \sqrt{in}). If the arrest point falls above the upper shelf, arrest is assumed not to occur.

Figure D-2 (a) illustrates the trend of K_{I} for a particular crack depth versus time for a hypothetical PTS transient. If pressure remains constant or

decreases with time, K_I will increase to a maximum and then decrease as the thermal stresses die out. The time at which K_I reaches its maximum determines the time of warm prestressing (WPS). When the entire initiation curve falls to the right of the WPS time as shoon in Figure D-2 (b), crack initiation is not expected to occur; however, care must be exercised in reaching this judgment because of analytical and material uncertainties. The dotted line in Figure D-2 (b) indicates that crack initiation might occur because of uncertainties and might reinitiate later in time because of an increase in K_I at that time. If this were to occur, arrest is not expected because then the arrest curve is above the upper shelf toughness.

1. 1

D.1 NRC Analytical Procedures

The NRC procedure to evaluate the effect of cooldown transients and postulated accident scenarios on the integrity of reactor vessels was developed in 1978 and subsequently updated to include technological data as they become available. It is designed primarily for investigations of thermal shock to the beltline region of vessels with a vessel radius to wall thickness ratio of about ten.

Heat transfer algorithms are based on classical closed form solutions which provide temperature distributions across a vessel wall versus time into a cooldown transient. These temperature profiles are used to calculate thermal stresses versus time and depth into the wall. The calculation of fracture mechanics stress intensity factors is based on the linear-elastic boundary integral equations method for cylinders and the superposition of stresses due to all causes particularly those due to temperature differences, pressure and residual stresses in welds. Although certain simplifying assumptions are used in the procedure, its results have been compared with those from more sophisticated analyses and found to be in good agreement.

D.1.1. Assumptions

Geometry

For heat transfer and thermal stress analyses, slab geometry is assumed. For typical reactor vessels with a vessel radius to wall thickness ratio of ten or more, the error introduced by this assumption is negligible compared to other

uncertainties inherent in the analyses. This assumption permits a more simple calculational procedure that is adaptable to programmable calculators or computers. Cylindrical geometry is used, however, in the fracture mechanics analyses.

Heat Flow

*

In a cooldown or heatup transient, heat flow is assumed to occur only in the wall thickness direction. Thus, the procedures are one-dimensional.

Heat Transfer Coefficient

The heat transfer coefficient, h, during a typical transient can vary over a considerable range depending on the hydraulic and thermal conditions. Its magnitude may even be difficult to determine versus time as the transient progresses because of hydraulic and other uncertainties. However, for values of heat transfer coefficients in the range of interest for most thermal transients (approximately 300 Btu/hr ft² °F), short perturbations to higher values do not cause significant increases in thermal stresses. Therefore, for typical transients of interest, metal temperature and stress distributions are obtained by utilizing a constant heat transfer coefficient. The value used is conservatively selected on the basis of experience and judgment. For maximum conservatism, a value of infinity can be used. The heat transfer coefficient is also assumed to be the same at all water cooled portions of the vessel wall.

Temperature Dependence of Metal Properties

The physical properties of materials are temperature dependent. When thermal transients result in a significant temperature range and difference through the vessel wall, accurate results require consideration of this phenomenon. Data for materials of interest are taken from recent ASME publications.

Analytical Model

Prior to the thermal transient, the water temperature is assumed to have remained constant for a sufficiently long time so that the vessel wall is at a uniform temperature equal to the water temperature. Prior to and during the transient, heat flow at the outer insulated surface of the vessel is assumed to be zero and the vessel cooled or heated only at the inner surface with no sources of heat within the metal. For typical transients of interest, these assumptions introduce minimum uncertainties in the end results.

Finite Number of Series Terms

Solutions for metal temperature distributions at various times during a transient are in the form of an infinite series. Because of obvious practical considerations, it is necessary to truncate the series to a finite number of terms. The error introduced by limiting the number of series terms is significant only at or very shortly after the start of the transient (time = zero) where an infinite number of terms is required to obtain correct temperatures. Shortly thereafter, however, higher terms in the series decay rapidly to insignificant values. Because, for transients of interest, the maximum thermal stresses generally occur relatively late in the transient, little or no error is introduced by utilizing a finite number of terms. Six series terms are used for deterministic analyses; however, the last two terms contribute very little. Therefore, for probabilistic analyses only four terms are used.

Effect of Cladding

Because the material and physical properties of the stainless steel cladding differ from those of the carbon steel wall, the cladding effect must be accounted for in reactor vessel integrity analyses. The presence of cladding affects the heat transfer and stress calculations as well as the fracture mechanics analyses. The heat transfer coefficient is readily adjusted to account for the higher thermal resistance of the stainless steel clad (Figure D-3). The stress effect of the clad, however, depends on the stress relief and operational history of the vessel. Once this is established, this effect is accommodated by superposition of cladding induced stresses with

those from other causes including those due to temperature variations across the wall of the vessel (Figure D-4). Fracture mechanics effects of cladding depend on the assumed shape and location of postulated cracks. Procedures for treating long through-clad axial cracks are used. The treatment of elliptical and circumferential cracks needs further development. In general, thermal stresses and stress intensity factors due to temperature differentials across the wall are calculated assuming only the thermal resistance of the clad, then calculating the clad induced stresses and stress intensity factors assuming a constant metal temperature and superpairing the results. Thus, the effect of cladding is accounted for in the heat transfer, thermal stress and fracture mechanics analyses when long axial through-clad cracks are assumed.

The NRC model for determining the clad effect for postulated long through-clad cracks is as follows:

- Assume that the clad is stress-free at reactor operating temperature. As the vessel wall cools down, tensile stresses in the cladding and lesser compressive stresses in the base metal develop and reach a clad stress of about 30 ksi at room temperature.
- The average clad temperature is assumed to be the cooled surface temperature during a transient; however, to determine the incremental effect due to the clad, the entire wall temperature is assumed to be constant (the effect of the actual temperature variation across the wall during a transient is superposed later). The lower thermal conductivity of the cladding is included in the determination of the surface temperature by a reduction in the heat transfer coefficient.
 - Knowing the clad incremental stress, the stress intensity incremental effect due to the clad is then calculated via the influence function technique described brief'y in Section D.1.4.

The results of an example calculation of the clad effect on the stress intensity factor as determined independently by the staff and ORNL are show in Figure D-5.

D.1.2 Stress Algorithms

The total peak stresses (thermal plus pressure plus residual plus any other stresses) are assumed to be less than, or at least not significantly larger than, the material yield strength so that components of stress can be added and that linear-elastic fracture mechanics procedures can be utilized. For rapid thermal transients, high stresses usually occur locally at the inner vessel wall and acceptable stress distributions (total stress below yield) over the remaining section can still be obtained if the overstressed region is relatively thin.

D.1.3 Postulated Initial Cracks

Long through-clad cracks, either axial or circumferential, are assumed to exist in the welds of limiting (highest) RT_{NDT}. In this case, the cladding effect is conservatively applied in that the stresses due to the different expansion coefficients of the clad and base metal are added to the nominal thermal stresses. For short through-clad cracks or underclad cracks it is conceivable that the cladding can have a beneficial effect if the cladding is sufficiently tough, that is, it is less affected by irradiation damage than the base material. In this case, it could deter crack elongation or could even prevent crack initiation depending on the specific transient. At present, there are differences of opinion as to clad toughness after irradiation, and further research is needed as to the behavior of short or underclad cracks in an overcooling event. Also, analyses to date omit consideration of weld residual stresses and in the case of circumferential cracks, the effect of dead weight stresses. An uneven temperature distribution in the azimuthal direction increases K, value for circumferential cracks. Therefore, the NRC concludes that the more conservative assumption of long through-clad cracks should be used at least for scoping calculations, until further information s developed to permit a relaxation of this assumption.

D.1.4 Fracture Mechanics Algorithms

Fracture mechanics analyses utilize the linear-elastic boundary integral methods of Heliot, Labbens and Pellissier-Tanon (References D.1 and D.2).

At each time step, the thermal and other stresses are expressed as polynomial functions of the relative depth into the wall of the vessel:

$$\sigma (\underset{L}{X}, t) = \underset{j=0}{\overset{4}{\Sigma}} \sigma_{j} (\underset{L}{X})^{j}$$

where σ_j 's are constants determined by curve fitting. The stress intensity factor for this stress distribution is then;

$$K_{I} = \sqrt{\pi a} \begin{array}{c} 4 \\ \Sigma \\ j=0 \end{array} \sigma_{j} \left(\frac{a}{L}\right)^{j} i_{j}$$

In the NRC procedure, the i_j 's are expressed as polynomial functions of the relative crack depth. Different expressions for the i_j 's are used for different crack geometries and directions.

The stress distribution due just to the cladding, however, cannot be expressed by a polynomial equation without resorting to a large number of terms. For this application, the staff used the basic equations in the references and adapted them to obtain an expression for long axial cracks in a cylinder (expressions for other crack geometries and directions need further development):

$$K_{I} = \sqrt{\pi a} \left(\frac{2}{\pi - 2}\right) \left\{ (i_{0} - \frac{2}{\pi}) \int_{0}^{\frac{\pi}{2}} \sigma(w) dw - (i_{0} - 1) \int_{0}^{\frac{\pi}{2}} \sin w \sigma(w) dw \right\}$$

where:

 $\sin \omega = \frac{\chi}{a}$, $o \leq \chi \leq a$

and i is the influence function for a uniform stress.

D.2 ORNL Analytical Procedures, OCA-I, OCA-II

In addition to performing its own PTS analyses, the NRC staff also utilized the services of ORNL. The ORNL analytical code differs from that of the NRC,

yet compatible results are obtained. The ORNL program is described in Reference D.3 from which some of the following is taken.

The OCA-I code is a computer program that performs a two-dimensional linear-elastic fracture mechanics analysis for long axial inner-surface flaws in a cylinder subjected to time-dependent thermal and pressure loadings. Six basic calculations are performed: (1) a one-dimensional thermal analysis to obtain temperature distributions through the wall of the cylinder as a function of time; (2) stress analysis, neglecting presence of flaw, using thermal and pressure loadings; (3) calculation of stress intensity factor (K_I) as a function of flaw depth and time; (4) calculation of static initiation and arrest toughness values (K_{Ic} and K_{Ia}) at tip of flaw as a function of flaw depth and time; (5) calculation of K_I/K_{Ic} and K_I/K_{Ia} as a function of flaw depth curves, which indicate the behavior of the flaw at all times during the transient.

Input to the thermal analys's includes the coolant temperature vs. time, the fluid-film heat transfer coefficient, and the initial temperature of the cylinder. All necessary material properties, with the exception of the reference temperature (RT_{NDT^o}) and the concentrations of specific impurities (copper and phosphorous), are included in OCA-I, but different values may be inputted. The calculation of K_{Ic} and K_{Ia} considers the temperature and fast-neutron-fluence distributions through the wall, RT_{NDT^o} and the copper and phosphorous concentrations, which influence the radiation damage effect.

The K_I calculation is based on a superposition technique that uses the uncracked-cylinder stresses and a set of unit-load K_I values (K*) that correspond to cylinder dimensions typical of a $1000^{-}MW(e)$ pressurized-water reactor pressure vessel (4.37-m ID x 4.80-m OD). The K* values were calculated using finite-element techniques and are included in OCA-I.

The development of OCA-I was prompted by a growing interest in the behavior of surface flaws in reactor pressure vessels during overcooling accidents. The OCA-I code was designed specifically for these accidents in an effort to minimize time and expense associated with the analysis. To this end, special provisions were made for parametric-type analyses. OCA-II, which was used for

0-8

later studies, includes plotting refinements plus the incorporation of the cladding effect in the stress intensity factor.

Ŀ

The OCA-II code (Reference D.4) which was developed by Oak Ridge National Laboratory utilizes:

- (a) the latest NRC calculation method for determining neutron fluence attenuation with depth into the vessel wall, which is described in Section D.4 of this report,
- (b) the latest NRC calculation method for determining shift in RT_{HDT} with neutron fluence, which is described in Section D.4 of this report,
- (c) a finite element, one-dimensional code with a constant heat transfer coefficient, h, in thermal analyses,
- (d) thermal, pressure and clad stresses and infinitely long axial through-clad crack in the finite element linear-elastic fracture mech (ics (LEFM) analyses,)
- (e) any prescribed water temperature during the transie t.

The OCA-II code and NRC LEFM analyses performed for this study do not include plate-to-plate weld residual tensile stresses. We believe that OCA-II and NRC stress, thermal and fracture mechanics analyses are sufficiently conservative to permit a parametric study of vessel fracture without including these stresses.

D.3 Warm Prestressing

Although warm prestressing (WPS) can theoretically prevent crack initiation during a pressurized thermal shock transient, the staff believes that the fluctuations of pressure and temperature during these transients are possible; therefore, our scoping calculations did not rely upon WPS to prevent crack initiation. The NRC staff believes that it would not be prudent for operators to rely primarily on warm prestressing to assure reactor vessel integrity

during pressurized thermal shock transients. The staff is aware of and accepts the theoretical basis for warm prestressing. One explanation for the WPS effect is:

"During temperature reduction, initiation of crack propagation from an arrested crack in the reactor vessel cannot occur while the K value is constant or decreasing." (Reference D.5)

Another explanation rests on a physical picture of blunting at the crack tip and development of favorable residual stresses caused by the warm prestress.

The theoretical basis for warm prestressing assumed that K_I is decreasing with time in <u>monotonic</u> fashion after it reaches its maximum value. In a real transient, the pressure component of K_I may rise and fall in an unpredictable fashion as the system is being brought to a stable condition. Some variation in the thermal component of K_I may also occur. Of particular concern to the staff is that emergency operating procedures at some facilities permit repressurization after a thermal transient to as high as 2000 psig. Thus, the potential benefit effects of WPS may be deliberately negated.

Experiments have shown that when there is an increase in K_I after cooldown to a temperature at which K_I exceeds K_{Ic} , there is an ever-increasing probability of fracture as K_I increases such that the probability is very nearly one for $K_I = K_I$ maximum. The probability of fracture decreases to acceptably low values for $K_I - K_{Ic} \leq 25$ percent of $(K_I \max - K_{Ic})$. (Reference D.6) The experimental information also shows clearly that the beneficial effects of warm prestress are nearly eliminated if K_I drops to a low value after reaching K_I -maximum and then increases, for example, repressurization late in transient after the vessel has cooled down.

During a typical transient scenario, the reactor coolant temperature and pressure both decrease initially from their normal operating values. Therea ter, the trend of both temperature and pressure depends on the nature of the even and the actions taken by operators and/or automatic systems. Because of the relatively rapid decrease of the reactor coolant temperature, thermal stresses are developed in the vessel wall which are superposed on the pressure stresses.

The net result is an increase of the total stress intensity factor, K_I versus time. The thermal component of X_I reaches a maximum and then decreases and the wall temperature tends toward a uniform value. Typically, the total K_I also has a maximum during this initial period. Thereafter, the change in K_I versus time depends on the assumed actions taken by operators and by automatic systems.

There are, of course, many possible variations in the cooldown scenario that will produce different degrees of departure from the ideal monotonic decrease of K_I after reaching K_I -maximum. Our knowledge is insufficient to draw the line between acceptable versus unacceptable transients with regard to the acceptance of warm prestressing, other than to say that we ought not to rely on it at this time. The exceptions are transients such as certain LOCAs where pressure is limited as described below.

Following a severe cooldown transient, the NRC staff believes that facility operators should limit reactor pressure by manual and/or automatic means to the extent practicable. Preferably, reactor pressure should be decreased monotonically consistent with 50°F subcooling and the pressurizer water level increased only to its normal operating range. In particular, water-solid conditions should be avoided especially if the reactor coolant reaches low temperatures. Repressurization should not be permitted until the transient has been evaluated, and for severe transients, the vessel should be inspected to assure its integrity.

Even if these procedures are followed, it still is conceivable that a small crack may initiate and grow deeper. However, in the absence of pressure, it will not penetrate the wall. With pressure stresses also present, it is possible that a crack would create an opening in the vessel, especially when the wall material has cooled down.

In conclusion, the staff believes that it would not be prudent to rely on warm prestressing to assure reactor vessel integrity during a pressurized thermal shock transient. The basis for this position rests on uncertainties regarding system considerations and on insufficient experimental information to confirm the benefits of warm prestressing under these circumstances at this time.

While the odds in favor of warm prestressing being a viable phenomenon to prevent initiation or reinitiation of a crack during a particular transient scenario may be relatively high, facility operators should also consider the relative risk.

For small break LOCAs of sufficient size such that the pressure is limited to some low value during the critical period of the transient or is monotonically decreasing because of the inability of the ECC and charging systems to maintain high values, then conditions are attained where warm prestressing can be effective and credit can be considered for it.

D.4 Determination and Utilization of Material Toughness

To make the fracture analyses of pressurized thermal shock, it is necessary to have values for the fracture toughness of the material at the tip of the postulated cracks in the reactor vessel wall. Toughness must be known as a function of time in the transient, and temperature and fluence must be known as a function of position in the wall.

D.4.1 ASME Code Section XI Curves

The fracture analyses performed by utilities, vendors and the NRC have all utilized the values of K_{IC} and K_{Ia} given in Section XI of the ASME Code and reproduced in Figure D-6. The toughness values are given as a function of the temperature, I minus RT_{NDT} , the reference temperature, nil-ductility transition. The quantity, RT_{NDT} is the sum of two quantities; the initial RT_{NDT} and the ΔRT_{NDT} caused by irradiation. Appendix E of this report describes the bases for estimating initial RT_{NDT} and ΔRT_{NDT} for the individual plants. Estimate: are given for the inside surface of the vessel wall (at the clad-base metal interface) for the critical locations, which are almost always the welds, either a longitudinal weld or a circumferential weld in the beltline. The second step is to determine the attenuation of ΔRT_{NDT} through the vessel walt.

D.4.2 Attenuation of Fluence and RT_{NDT} through the Vessel Wall

Some recent changes have been made in the way the attenuation calculations are performed. These are illustrated in Figure D-7. In the past, the attenuation of fluence has been calculated by an exponential equation fitted to the results of calculations given in surveillance reports, as follows.

f	=	f _o e ^{33x}
f	a	fluence at any point, n/cm ² (E> 1 MeV)
fo	=	fluence at inside wall
x	=	distance from inside wall, inches

However, changes in the neutron energy spectrum within the wall cause the use of the above formula to be unconservative. Therefore, the NRC has chosen to use displacements per atom (dpa) as the damage function, following a report received from HEDL (Reference D.7). They provided six plots of the ratio, dpa/fluence (E > 1 MeV), versus depth in the vessel wall. At 8.0 inches, the ratio averaged 2.06. To achieve this reduction in the attenuation at 8.0 inches, the equation for fluence attenuation becomes:

$$f = f_0 e^{-.24x}$$

ψ.

Thus, we use a "dpa equivalent" attenuation equation, while retaining the description of fluence in terms of n/cm^2 (E > 1 MeV).

As illustrated in the lower part of Figure D.7, the combination of the dpa-equivalent equation for attenuation of fluence and the Guthrie trend cur e formula gives an expression for the attenuation of RT_{NDT} that is much less steep than that previously used. We believe that the new expression is realistic and have incorporated it into our program and the OCA-II code described in Section D.2.

The staff recognizes that the technical community is not in total agreement that dpa is the best physical model for the correlation of neutron damage as a function of fluence where the neutron energy spectrum varies significantly.
However, it does seem important to account for spectral changes and dpa is the most generally accepted method.

D.5 Stress/Fracture Mechanics Procedures Summary

The analytical methods used by NRC, ORNL and vendors differ somewhat but yield essentially the same results if all input assumptions are the same. Differing conclusions result primarily from assumptions as to crack shape, clad effects, effect of warm prestressing, etc.

When materials properties and the transient are known, these procedures can predict crack behavior quite well as demonstrated by results from the ORNL thermal shock experiments.

For generic studies, the NRC uses an exponential decay of water temperature to envelope a variety of transients. The staff has also used the Rancho Seco event as an analytical model. Our objectives are:

Avoid crack initiation, if possible. Avoid vessel failure, in any event.

The staff has studied the PTS issue both deterministically (conservative assumptions) and probabilistically (mean values of parameters) to assess risk to a vessel.

D.6 Discussion of Results

The NRC has performed both deterministic and probabilistic fracture mechanicanalyses to generate a basis for judgment regarding the safety margins again t PTS transients especially for the more highly irradiated vessels. Although recognizing that the transients that occurred at Rancho Seco in 1978 and Ginra in 1982 are unique, and are very unlikely to happen in the same way again, the staff concludes that they provide measures of the severity of a PTS event. The NRC and ORNL have arbitrarily utilized an idealized Rancho Seco pressuretemperature transient as a benchmark model for other vessels. For generic investigations, however, a postulated exponential decay of water temperature has proved to be more appropriate in that it can be characterized by two parameters, β (min.-1) which is the reciprocal time constant and T_f (°F) the final postulated equilibrium temperature. The initial temperature, To, is the normal operating temperature. Thus, $\overline{i}_{w} = T_{f} + (T_{o} - T_{f}) - \frac{1}{e}\beta t$ Information obtained from transients that have actually occurred at nuclear facilities indicates that the above formulation adequately describes the water temperature at least for the initial critical portion of the transient. After T_f is approached, operator and/or systems action can, of course, affect the longer term variation of water temperature. Typical values of β have been found to be in the range of 0.05 to 0.15 min.-1 as a consequence of the physical limitations of a real facility. Higher values of β of about 0.5 min.⁻¹ or more have been estimated for hypothetical, low probability design basis transients. Typical values of T_f for the worst cooldown transients to date (including those at BWR facilities) are in the range of 250 to 300°F. The Rancho Seco event, for instance, resulted in β of about 0.01 min.-1 and a T_{f} of about 290°.

For the more likely transients, the times of crack initiation have been calculated to be 20 to 30 minutes or more after the onset of cooldown, the actual time varying up to one hour depending on the pressure and RT_{NDT}. Thus, operators have time to gain control of the event if properly instructed and trained.

The OCA II Code was utilized to determine the lowest RT_{NDT} for crack initiation as a function of constant pressure, final water temperature (T_f) and the reciprocal time constant (β). From these data, Figures D-8 and D-9 were plotted which indicate the effect of T_f , β and pressure on crack initiation.

The principal objective of the NRC (and the industry) is to prevent crack initiation, and for more probable PTS events, this may be possible. However for the less likely events such as a postulated small break LOCA, crack init ation is likely in vessels with a relatively high RT_{NDT}. For these cases, the objectives must be to prevent any crack from propagating through the wall. Early in the transient, a pre-existing crack can initiate and propagate to the order of half the wall thickness or somewhat less, and then arrest because the

metal at this location is still much warmer than at the cooled surface and because the metal at this depth has experienced less irradiation damage. As previously mentioned, linear elastic fracture mechanics (LEFM) methods are used for analysis of PTS transients. Typical values of K_I at the first crack initiation range from 60 to 100 ksi \sqrt{in} .

LEFM is not valid for tough materials such as are encountered on or above the upper shelf. Other techniques are necessary. These techniques have been developed for analysis of piping flaws and for pressure vessels under pressure loads only (Task Action Plan A-11). To date, they have not been adequately developed for treatment of more complex stress patterns such as occur in a PTS event.

Therefore, the NRC conservatively assumes that, if a crack is calculated to propagate above the upper shelf of the material (200 ksi \sqrt{in} is assumed), it is assumed to continue propagating through the wall. It is recognized that subsequent elastic-plastic or fully plastic analyses may show that this may not be the case. On the other hand, it must be recognized that if the pressure is high enough, crack propagation through the wall is possible, even in tough material, because the remaining ligament may not be sufficient to sustain the pressure and residual thermal stress loads. Pending further research in this area, the NRC concludes that a conservative approach must be taken.

D.6.1 OCA-II Parametric Study

The OCA-II code was used to make a parametric study of the effects of pressure P, final water temperature, T_f , and the reciprocal time constant, β , on the critical values of RT_{NDT} at the inside wall for crack initiation and crack penetration through the wall (no arrest). (Strictly speaking, initial RT_{NDT} should be mentioned as a variable, because it is only ΔRT_{NDT} that attenuates through the vessell wall, but the difference in critical values of RT_{NDT} for different initial RT_{NDT} values is negligible.)

Some of the results of the parametric study, plotted in Figure D-8, show that the $T_f = RT_{NDT}$ is a fairly reasonable normalizing parameter, although the curves for different T_f values are separated by as much as 10-20 degrees at

low pressure. Figure D-8 indicates that crack initiation will occur at lower material RT_{NDT} as pressure increases or final water temperature decreases. Figure D-9 indicates that crack initiation will occur at lower material RT_{NDT} as β increases, but, the effect is slight for values of β greater than 0.15. The "dogleg" in the curves of Figures D-8 and D-9 occurs because the critical crack size changes. At low pressure, K_{I} - thermal predominates in the fracture analysis and the critical crack sizes are a fraction of an inch, whereas at high pressure the critical size is near the arbitrary limit of 1.25 inches.

A cross plot of these Figures, shown in Figure D-10, illustrates the effect of pressure and T_f on the critical value of RT_{NDT} , for a given value of β (0.15 min.⁻¹). To use Figure D-9, the plant condition as characterized by RT_{NDT} is related to the transient severity as characterized by P, T_f and β to determine if the vessel is safe from crack initiation. This is, of course, a deterministic calculation, which assumes that the critical flaw depth given by the analysis is indeed present in the critical weld. Stated in another way, if the value of RT_{NDT} used is the true value, the probability of crack initiation is the probability that the critical flaw is indeed present.

Also shown in Figure D-10 is a set of "no arrest" lines, which merge with the solid lines for crack initiation at about 600 psig. This means that at very low pressure, cracks will arrest if T_f is between the solid line and the dashed line. At higher than 600 psig, the analysis shows that a crack, once it has initiated, will penetrate the vessel wall. The assumptions on which this analysis is based are thought to be conservative--they assume that the material will behave as indicated by linear-elastic fracture mechanics.

Finally, in Figure D-10 there is a steeply slanting dashed line marked "Circumferential cracks." It was drawn on the basis that at low pressure K_I - thermal is the same for cracks of any orientation (which is nearly true for shallow cracks) and on the basis that K_I - pressure for circumferential cracks is approximately one-half of that for axial cracks.

The fluid film heat transfer coefficient, "h", is another variable (in addition to T_f , P and β) that is part of the characterization of a transient. The parametric study described above was made using an "h" of 1000 Btu/hr ft² °F,

which is characteristic of a "pumps on" condition. To check the effect of a change in "h" to 300, for a "pumps off" condition, Eight cases were repeated, using OCA-II. The results, shown in the following table, are the differences in critical RT_{NDT} (in degrees F) for a calculation using h = 300 minus the result for h = 1000. As expected, a higher value of RT_{NDT} can be tolerated when "h" is lower, but the difference is only about 10°F or less at high pressures. The difference is seen to be greatest at low pressure, where K_{I} - thermal is the predominant part of K_{I} - total, and for a severe cooldown. This means that the near vertical lines of Figure D-10 would move to the left 5°F at P = 2500 psig and about 29°F at 500 psig. Figure D-11 is a repeat of Figure D-10 for h = 300 instead of 1000.

			$T_{f} = 150^{\circ}F$		$T_f = 300^{\circ}F$	
			$\beta = 0.015$	$\beta = 0.15$	β = 0.015	$\beta = 0.15$
P = P =	500 2500	psig psig	9 11	29 5	0* 7	25 5

*Both calculations stopped at RT_{NDT} = 400°F

D.6.2 Fracture Mechanics Analysis for Several PWR Recorded Transients

In the past, a number of events have occurred that can be categorized as PTS transients. Some of these have previously been analyzed by fitting the actual temporal temperature and pressure variations with smoothed and/or bounding curves in order to facilitate the analysis. These transients have recently been reanalyzed using the recorded temperature and pressure traces with all their respective fluctuations. The results are presented in an ORNL report, Appendix 0, and as is discussed elsewhere in this document, were used as part of the basis in arriving at RI_{NNT} screening criteria.

D.6.3 Fracture Mechanics Example Analyses

In addition to the many uncertainties regarding PTS scenarios such as the temperature and pressure profiles versus time, the degree of mixing of cold water with warm water, etc. there exists parametric uncertainties in the stress and fracture mechanics analyses. The treatment of these uncertainties becomes significant when the cooldown temperature is to approximately RT_{NDT} because small changes in assumptions can influence whether or not a crack will initiate.

Assuming infinitely long cracks, h = 330 Btu/hr ft² ^oF (including clad effect) and using, for example, an assumed downcomer water temperature transient of

 $T_w = 250 + 300 e^{-0.15t} {}_{o}F$, t = minutes

42

which is only slightly more conservative than transients that have actually occurred and RT_{NDT} at the cooled surface of 294°F which is only slightly greater than that which exists in some facilities, the NRC staff found that to prevent crack initiation, the pressure versus time would have to be less than as shown in Figure D-12. That is, the pressure should be reduced to near saturation conditions by about 30 minutes if warm prestressing (WPS) is assumed to be ineffective. If the pressure had been reduced approximately monotonically, then WPS, which occurs at about 18 minutes for this assumed transient, could also preclude crack initiation. From the results of this transient provided by ORNL, which were calculated using somewhat more conservative assumptions regarding input parameters, crack initiation was predicted at about 24 minutes even for zero pressure if WPS is not effective. The main contributor to this difference in conclusions is believed to be the effective heat transfer coefficient used in the respective analyses. Thus, for cases where the final temperature is in the range of RT_{NDT} , the sensitivity of results to the various input parameters needs to be investigated before final conclusions can be reached as to limiting pressures.

A factor for consideration regarding these transients is that, in general, larger pre-existing cracks are necessary before crack initiation would occur for the cases of higher RT_{NDT}'s. This factor is not illustrated in the figures in this appendix.

This same temperature transient was also analyzed for different values of RT_{NDT} at the vessel inner radius and for a circumferential crack. The results are shown in Figure D-13. Note that the effect of the clad is approximately

8°F and that a circumferential crack will tolerate about a 30°F higher RT_{NDT} (considering crack initiation only) for this transient. Similar variations would be expected for other transients. This example illustrates the benefits to be attained by monotonically decreasing pressure in the event of a moderately severe thermal transient in that it is possible to avoid crack initiation.

For much more severe thermal transients, crack initiation may occur due to high thermal stresses. In this case it is necessary to consider the potential for crack arrest. Figure D-14 is a schematic representation of a critical crack depth diagram to illustrate the analytical model used by the staff for determining acceptable arrest criteria. An upper shelf toughness of 200 ksi \sqrt{in} . is assumed; however, higher or lower values may be more appropriate for a specific reactor vessel. When the thermal stress intensity factor is known at the time of warm prestressing (WPS), the maximum pressure is determined such that arrest will occur at or before the time of WPS and for crack depths below the upper shelf curve. The limiting case is shown as point "A" in the figure. The thermal transient selected for this example is:

 $T_{w} = 60 + 480 e^{-\beta t}$

Figure D-15 illustrates the effect of the cooldown rate with a water to metal heat transfer coefficient of 300 Btu/hr ft² °F. Figure D-16 shows the equivalent results for a lower coefficient. Note that the sensitivity to the heat transfer coefficient is greatest for the more rapid cooldown. Figure D-17 shows the effect of various assumptions regarding the attenuation of RT_{NDT} in the metal as discussed in Section 3 and Appendix E of this document. The above figures are for long axial cracks. Figure D-18 shows the effect of assuming long circumferential instead of axial cracks. In terms of RT_{NDT} , instead of being about 30°F for crack initiation, the difference now is about 100°F for crack arrest, depending on the specific pressure. Also shown in Figure D-18 is the effect of crack shape at arrest. (An a/c value of 0.1 represents a crack which is 20 times as long as it is deep.)

Figure D-19 is for another transient. It illustrates the uncertainty in RT_{NDT} that can occur due to the selection of the time of warm prestressing because of the relative flatness of K_T versus time near its peak value. Again, the

difference between axial and circumferential cracks is shown when warm prestressing and arrest are considered.

As stated earlier, the NRC staff assumes an infinite flaw length in its analyses; that is, an ellipse with an aspect ratio of zero. For circumferential cracks that arrest at some depth, this assumption is believed to be reasonable if the vessel wall is uniformly cooled in that direction. On the other hand, growing axial cracks could be limited in length by reaching the ends of a critical weld and intercepting tougher plate material. Also, they could extend into less irradiated regions of the vessel wall and hence into tougher materials even within the weld. Thus, the assumption of an infinitely long axial crack is conservative.

Figure D-20 shows the effect of the assumed crack shape at arrest. If an aspect ratio a/c = 0.1 is assumed instead of zero, there is a gain in RT_{NDT} of about 60° for the case illustrated. This appears to be reasonable in that, if the crack arrested at the ends of an axial weld, it would be approximately half-wall thickness in depth. An assumed aspect ratio of 1/3 would lead to higher tolerable RT_{NDT} 's; however, analyses and experiments related to growing cracks during a severe thermal transient indicate that cracks arresting with this shape are very unlikely. Also wall penetrations might occur before the ends of the crack reached tough materials. Therefore, the staff does not accept this assumption. If other than infinitely large arresting cracks, say those with an aspect ratio up to 0.1, are to be accepted, then reasonable assumptions have to be made regarding all stresses especially weld residual stresses that can be present in addition to those due to pressure and temperature distribution.

These illustrations are intended to demonstrate the importance of limiting the reactor system pressure in the event of a severe cooldown transient as well as the necessity to allow for uncertainties both in analyses of transients and in material properties.

Based on the examples illustrated in Section D.6.3 and on the analyses of other organizations for similar PTS scenarios, it is seen that variations in

input assumptions can lead to differences of limiting RT_{NDT}'s for axial crack initiation. Specific differences will, of course, depend on specific scenarios. The following are typical results:

	Assumption	Effect on RT _{NDT} , °F	
(a)	Clad stress vs. no clad stress	10°	
(b)	Continuous flaw for initiation vs. elliptical flaw (a/c = 1/3)	20°	
(c)	h = 300 Btu/hr ft ^{2 o} F vs. Westinghouse free convection correlation	15°	

The above assumption differences account for a total RT_{NDT} variation of about 45° between staff analyses and those of Westinghouse. The Westinghouse model for fluence attenuation into the wall is equivalent to the dpa model of the staff. Other vendors, however, may still be using other models. The attenuation effect on limiting RT_{NDT} 's for crack initiation is not expected to be great but for crack arrest situations, the difference can be significant as illustrated in Figure D-17.

REFERENCES

- D.1. R. Labbens, A. Pellissier-Tanon, J. Heliot, "Practical Method for Calculating Stress-Intensity Factors Through Weight Functions," Mechanics of Crack Growth, ASTM STP-590, American Society for Testing and Materials, 1976, pp. 368-384.
- D.2. J. Heliot, R. C. Labbens, A. Pellissier-Tanon, "Semi-Elliptical Cracks in the Meridional Plane of A Cylinder Subjected to Stress Gradients, Calculation of Stress Intensity Factors by the Boundary Integral Equations Method," XIth National Symposium on Fracture Mechanics, Blacksburg, Virginia, June 1978.
- D.3. S. K. Iskander, R. D. Cheverton, D. G. Ball, NUREG/CR-2113, ORNL/NUREG-84, OCA-I, "A Code for Calculating the Behavior of Flaws on the Inner Surface of a Pressure Vessel Subjected to Temperature and Pressure Transients."
- D.4. A Supplement to ORNL-TM-7931 in Preparation, "PWR Pressure Vessel Integrity During Overcooling Accidents: A Parametric Analysis," Draft dated April 13, 1982.
- D.5. "Benefits of Warm Prestressing," by George Irwin, University of Maryland. Minutes of ACRS Briefing on Thermal Shock, December 5, 1981.
- D.6. J. J. McGowan, "An Assessment of the Beneficial Effects of Warm Prestressing Under Severe Thermal Shock," WCAP 9178, March 1975.
- D.7. G. L. Guthrie, W. N. McElroy and S. L. Anderson, "A Preliminary Study of the Use of Fuel Management Techniques for Slowing Pressure Vessel Embrittlement," Paper presented at 4th ASTM-Euratom Symposium, March 1982.



FIGURE D-1 RESULT OF PTS ON VESSEL WALL

7



1-1

TIME

FIGURE D-2

CLAD EFFECTS



š

(



.







٠.

FIG. A-4200-1 LOWER BOUND K₁₄ and K₁₀ TEST DATA FOR SA-533 GRADE B CLASS 1, SA-508 CLASS 2, AND SA-508 CLASS 3 STEELS FIGURE D-6, ASME CODE SECTION XI CURVES FOR RELATING FRACTURE TOUGHNESS TO T-RT SECTION XI - DIVISION I

4







Distance, x, inches



212



1.

 $\left(\begin{array}{c} \cdot \cdot \\ \cdot \end{array} \right)$









٩.

PRESSURE, PSIG

1



FIGURE D-11 EFFECT OF T AND RT NDT ON THE CRITICAL PRESSURE FOR CRACK INITIATION



t, min.

٩.



•



FIGURE D-14 NRC CRACK INITIATION AND ARREST MODEL



•

ÿ,

PRESSURE, PSIG

na a ga ang shiring shiring sa sa



-



•

2

PRESSURE, PSIG





EFFECT OF ASPECT RATIO

AT CRACK ARREST



4

- 114 -

APPENDIX E

. 125-

DETERMINATION OF RT_{NDT} FOR PLANTS FOR COMPARISON WITH SCREENING CRITERIA

E.1 Introduction

In Appendix D, RT_{NDT} was shown to be an important quantity in the fracture analysis of PTS, because the toughness values, K_{IC} and K_{Ia} are given in the ASME Code, Section XI as a function of $T - RT_{NDT}$. (In such analyses, the metal temperature, T, and the adjusted reference temperature, RT_{NDT}, are the values at the tip of the postulated crack.) Moreover, the results of the parametric studies described in Sections 3 and 7 and Appendixes D and H show that $T_f - RT_{NDT}$ is an important factor in the characterization of cooldown transient severity for a given plant. In this case, T_f is the asymptotic cooldown temperature of the water in the downcomer, and RT_{NDT} is estimated at the inside surface of the vessel. This finding led to consideration of RT_{NDT} as a screening criterion. Obviously, RT_{NDT} for a given plant is not related to the severity or probability of occurrence of a PTS in that plant and is therefore not necessarily the overall criterion for rating plants. Nevertheless, the value of RT_{NDT} at the inside surface of the vessel is a good screening criterion for the tendency of a reactor vessel to suffer damage from PTS.

 RT_{NDT} is the sum of two quantities: the initial RT_{NDT} from tests made at the time the vessel was fabricated and the ΔRT_{NDT} estimated from tests designed to measure the effects of neutron radiation. The purpose of this discussion is to describe the bases for estimated initial RT_{NDT} and ΔRT_{NDT} for the individual plants. Estimates will be given for the inside surface of the vessel wall (at the clad-base metal interface) for the critical locations, either a longitudinal weld or a circumferential weld in the beltline or occasionally a beltline plant or forging.

As described below, there are a number of uncertainties in the estimation of initial RT_{NDT} and ΔRT_{NDT} , and thus there is the difficult question of

establishing a proper, known, degree of conservatism in the estimate of RT_{NDT}. To resolve this question, a Working Group on RT_{NDT} was assembled for a two-day meeting (June 17 and 18, 1982) to review the NRC methods and recommend a method for use in the report. The work of that group is described in Referenced E.10. The method described below follows the recommendations of the Working Group.

E.2 Initial RT_{NDT}

E.2.1 Code Definition

The Summer 1972 Addenda to Section III of the ASME Boiler and Pressure Vessel Code contained the first requirements for measurement of RT_{NDT} for the plates, forgings, and welds that make up the reactor vessel, measurements to be made at the time of fabrication. Two types of tests are required--drop weight tests and Charpy tests. However, most of the vessels in question were fabricated in the 1960's when only Charpy tests were required.

E.2.2 Absence of Actual Measurements of RT_{NDT}

Typically, the data available to the NRC staff comprise 3 Charpy tests at 10° F for each plate, forging and weld, complete Charpy curves for the surveillance weld and base materials, and in cases where the base material was controlling, some drop weight data on archive or surveillance material. In the past, the NRC has used the guidelines of Branch Position MTEB 5-2 to obtain an estimate of initial RT_{NDT}. In summary, those guidelines were to use the Charpy 30 ft. lb. level, but not lower than 0°F. The Charpy curves from the surveillance tests were used to guide any extrapolation needed to get the 30 ft. lb. temperature from the 3 tests results at +10°F.

In summary, values of initial RT_{NDT} measured according to ASME Code rules are not generally available for the welds in question. Estimates based on the 3 Charpy test results and MTEB 5-2 are not very satisfactory, because they are overconservative for some cases.

E.2.3 Generic Data

From compilations of data obtained subsequent to the time the vessels in question were made, it is possible to divide the welds into two groups according to the weld flux used, and to develop a mean value and a standard deviation for the generic data. One must then decide if it is purdent to use the mean generic value as the best estimate for the vessel welds in question. Except for some archive material, the welds that are represented in the data base were made at a later time than the vessel welds. There may have been some differences in weld chemistry or welding practice. Furthermore, even if there were actual RT_{NDT} values for the vessel weld in question, the samples would come from weld metal qualification welds, not from actual vessel weld prolongations and not from full thickness test pieces. Thus, a mean plus 2 sigma value appears to be the best engineering estimate the initial RT_{NDT} for use in the screening criterion.

· · · · · · ·

In the Combustion Engineering Report, CEN-189 (Ref. E.2) there is a table of values of initial RT_{NDT} which contains 49 values for Linde 0091, 20 values for Linde 124, and 13 values for unidentified weld fluxes, some of which we have identified as Linde 1092. By inspection, the three groups appear to be in the same population, and the total has been treated as such to yield a mean value of -56°F and a standard deviation of 17°F. It was pointed out by PNL (Ref. E.3) that these data are not normally distributed, but were skewed to the high side. However, the resulting error is swamped by the uncertainty in the application of these data to the actual vessels. An earlier weld flux, ARCOS B-5, used on one or two vessels, was deemed to be in the same population based on comparison of available Charpy energy values.

For Linde 80 weld flux, a set of 10 values provided by Babcock and Wilcox (Ref. E.4) had a mean value of 0° F and the range was from -40° to $+20^{\circ}$ F. Because the sample size for the Linde 80 welds was small, the standard deviation was taken to be the same as for Linde 0091 welds, 17° F.

11/13/82

PTS RPT APP E

E-3

E.2.4 Comparison with Vendor's Values

Westinghouse (WCAP 10019) (Ref. E.1) used MTEB 5-2 to estimate RT_{NDT} values. Combustion Engineering (CE) (Ref. E.2) proposed two bases: (1) 60°F below the Charpy 50 ft. lb. level, and (2) an upper-2-sigma value from generic data for the weld fluxes in their vessels, Linde 0091, 1092, and 124. Their utilities used method 1, but the CE report made for each plant used method 2. Babcock and Wilcox (B&W) used upper bound values from generic data for Linde 80 weld flux, which was used in their vessels. An exception is Three Mile Island 1, for which a lower value of initial RT_{NDT} was used, basis not specified.

The following table compares vendors' values with NRC values. The latter are mean plus two sigma values. As described as paragraph E.4, in combining initial and ΔRT_{NDT} , the full two sigma value is reduced about 10 degrees by the use of the quantity $2\sqrt{\sigma_0^2 + \sigma_\Lambda^2}$.

	Linde 80 flux	Linde 0091 etc. flux
NRC	0°F mean plus 34 = 34°F	-56°F mean plus 34 = -22°F
W	0 to +10°F	0 to +10°F
CD		-20°
CE Utilities		-50°F
B&₩	+20F	

E.3 Adjustment of RT_{NDT} Due to Radiation (ΔRT_{NDT})

E.3.1 Trend Curves versus Surveillance

Most of the plants in question in the thermal shock issue have withdrawn at least one surveillance capsule and tested the irradiated specimens therein. The fluence is generally not exactly the value of interest, but the result can be extrapolated to the fluence of interest by using one of the trend curves to be described. However, there are problems associated with using individual surveillance results as the sole source of information about a plant. First, the surveillance weld often does not match the critical vessel weld exactly, i.e., the weld wire heat numbers are different. A broader problem is that caused by scatter in the ΔRT_{NDT} data. This results in part from the fact that ΔRT_{NDT} is the difference between the curves for irradiated and unirradiated material, both of which were fitted to data that typically shows considerable scatter.

Thus, there is a preference for the use of trend curves instead of individual surveillance data. To use any of the trend curves, the chemistry of the material must be known, in particular, the copper content. This is obtained from analysis of the weld metal qualification weld for the weld wire heat number and weld flux number that were used for the critical weld. If not available, data were sought for that weld wire heat number as used in other vessels. Failing that, best estimates were made from the surveillance weld (even through the heat numbers did not match) and from generic data for welds made in that time period. As a last resort, a value of 0.35% copper was used, that being the value which gave the upper limit or bounding line for all data in Regulatory Guide 1.99 Revision 1 (Ref. E.5) as described below.

E.3.2 Regulatory Guide 1.99 Revision 1, Bounding Curves

Regulatory Guide 1.99 Rev. 1 published in April 1977, contains the procedure recommended at that time by the NRC to obtain ΔRT_{NDT} , the "adjustment of reference temperature" as a function of chemistry and neutron fluence. Copper was the dominant residual element in the chemistry term (the other was phosphorus) as can be seen at the top of Figure E-1. The exponent on the fluence term is 0.5, but there is a cut-off or upper limit line for which the exponent is 0.194 for high copper content and fluence exceeding 6 x 10^{18} n/cm² (E>1 MeV).

Criticism leveled at Regulatory Guide 1.99 became more insistent when the PTS issue made it necessary to look hard at all sources of conservatism. It was said that (a) the curves were too conservative at high fluences, especially for low-nickel materials, and (b) the phosphorus term was not supported by

E-5

recent studies such as the MPC report (Ref. E.6) described below and an EPRI report (Ref. E.11), and should be dropped. Nevertheless, Regulatory Guide 1.99 was used for high-nickel materials by all 3 vendors in the reports that were concurrent with the utilities' 150 day reports. The high-nickel materials are ASTM A 533 plates, A 508 forgings, and welds of comparable chemistry, for which the nickel content is generally between 0.5 and 1.0 percent. The low-nickel materials are ASTM A 302 plates and welds of comparable chemistry, which generally have less than 0.25 percent nickel as a residual element. A relatively small number of older vessels have low-nickel material.

. .

E.3.3 Gutherie Trend Curves

Evidence has been accumulating for several years that the low-nickel materials are less sensitive to neutron radiation. When the PWR surveillance data base was analyzed by the NRC in October 1981, the difference between high and low nickel content material was apparent. Westinghouse and CE reported similar findings and presented empirical equations for the low-nickel material. (B&W have no plants with low-nickel materials in the reactor vessel.) The PWR surveillance data have now been fitted by a multiple regression analysis technique. The work was done at HEDL by George Guthrie, whose name is attached to the new trend curves (Ref. E.7). The Guthrie mean curve is as follows:

 $\Delta RT_{NDT} = [-10 + 470 \text{ Cu} + 350 \text{ Cu} \text{ Ni}] [f/10^{19}]^{0.27}$

ΔRT_{NDT} = adjustment of reference temperature, degrees F Cu = weight percent copper Ni = weight percent nickel f = fluence, n/cm² (E > 1 MeV)

The use of a copper-nickel product term reflects the advice of J. R. Hawth rne (Ref. E.8) of the Naval Research Laboratory to the effect that nickel seem: to enhance the effect of copper, but nickel does not cause increased embrittler ment in the absence of copper. The product term is also consistent with work reported by Varsik and Byrne (Ref. E.9) in which their "chemistry factor" was the product of copper and a quantity, nickel plus other elements.

11/13/82

E-6
Figures E-2, E-3, and E-4 show how the Guthrie formula fits the PWR surveillance data. The residual value (predicted minus measured) for each line of data is plotted against fluence, copper content, and nickel content to give a graphical check on the effectiveness of the multiple regression analysis.

121

E.3.4 Guthrie Upper Bound Trend Curves

• _

44

The standard deviation for the data analysis described in paragraph E.3.3 was 24 degrees F. From inspection of Figure E-2, it appears that a constant 2-sigma upper bound is satisfactory over the fluence range of interest.

E.3.5 Comparison with MPC Curves

As further support for the Guthrie mean curve, Figure E-5 gives a comparison of the Guthrie mean curve for representative copper and nickel contents with a mean curve developed by the Metal Properties Council for ASTM Committee E-10 on Nuclear Technology and Applications (Ref. E.6). The latter is being ballotted as an ASTM Standard. The MPC data base contains all of the test reactor and surveillance data that fit the criteria for material form and irradiation temperature that were available in November, 1977. There is reasonably good agreement between the MPC trend curves and the Guthrie curves, considering that the MPC curves were for a range of nickel content, but were without a nickel term in the equation.

The MPC trend curve did not contain a phosphorus term, because in the regression analysis the addition of a phosphorus term did not produce any significant decrease in the residual variance. In a further study of this finding, the MPC Task Group found a statistically significant relationship of phosphorus content to copper content, i.e., high phosphorus was found with high copper. Thus, their combined effects were represented in the trend curve formulation by a copper term alone.

E.3.6 Comparison with Vendor's Curves

Westinghouse and Combustion Engineering drew bounding curves for low-nickel material. Figure E-6 gives a comparison of the Gutherie mean plus 2-sigma curves for 0.15% nickel material with the low-nickel trend curves presented by Westinghouse and CE. The latter lie below the Guthrie curves over most of the range of fluence.

E.4 Screening Value of RT_{NDT}

The Working Group on RT_{NDT} (Ref. E.10) agreed that the value of RT_{NDT} to be used in screening plants should be calculated as the sum of 3 quantities: the mean value of initial RT_{NDT} (RT_{NDTo}), plus the mean value of ΔRT_{NDT} at the inside surface of the vessel, plus twice the square root of the sum of the squares of the standard deviation on each, i.e., $2\sqrt{\sigma_o^2 + \sigma_\Lambda^2}$.

E.4.1 Uncertainties

Uncertainties in the screening value of RT_{NDT} arise from several sources. Those associated with the estimate of initial RT_{NDT} were discussed in paragraph E.2. For ΔRT_{NDT} , there is the scatter about the trend curve (shown in Figures E-2, E-3 and E-4) which is made up of the uncertainty in response of material to radiation, plus errors in the copper and fluence values in the data base and errors in the Charpy shift measurement itself. In addition, there is uncertainty in the copper content of the critical weld in the vessel. Because copper was introduced as a plating on the weld wire, and plating thickness was not controlled, variation in copper content through the vessel wall and along the length of the weld is expected to be considerable From a number of measurements for certain weld wire heat numbers, one stancard deviation is expected to be about 0.03 percent copper, typically. This is equivalent to 15 degrees F in the plants with higher fluences.

Nevertheless, the copper contents used in calculating RT_{NDT} for plants were best-estimate values. They were not mean plus 2 sigma values. This is one reason why the Working Group on RT_{NDT} felt that the screening values should have the 2 sigma measure of error added to the mean.

E.4.2 Alternative Calculation of RT_{NDT}

For high values of copper and nickel contents, the method described above gives values higher than those predicted by that part of the Upper Limit of R.G. 1.99, given by the equation:

$$\Delta RT_{NDT} = 283 (f/10^{19})^{0.194}$$

Experience has shown that the latter bounds the available data. Therefore, the screening value of RT_{NDT} is taken to be the lower of two quantities:

$$RT_{NDT} = RT_{NDT} \circ + Guthrie Mean \Delta RT_{NDT} + 2 \sqrt{\sigma_0^2 + \sigma_\Delta^2}$$

or

٩.

$$RT_{NDT} = RI_{NDTo} + 283 (f/10^{19})^{0.194} + 2 \sigma_{o}$$

as illustrated schematically in Figure E-7.

The 2-sigma term in the second equation does not include the error in ΔRT_{NDT} because the term for ΔRT_{NDT} is an upper-bound equation.

The Upper Limit line of R.G. 1.99 actually consists of two branches, the one described above, for fluences above 6 x 10^{18} , and a lower branch that has an exponent of 0.5. The latter was not used, because it does not bound all of the observed data in that fluence range. Thus, for the purpose of this screening criterion, the alterative equation,

$$\Delta RT_{NDT} = 283 (f/10^{19})^{0.194}$$

is used at fluences below 6 x 10^{18} as well as for higher fluences.

11/13/82

REFERENCES

E.1 Summary Report on Reactor Vessel Integrity for Westinghouse Operating Plants by T. A. Meyer, December, 1981, WCAP 10019.

1 3 .1

- E.2 Evaluation of Pressurized Thermal Shock Effects Due to Small Break Loca's With Loss of Feedwater for the Combustion Engineering NSS, CEN-189 Prepared for the C-E Owner's Group by Combustion Engineering, Inc., December 1981.
- E.3 PNL Technical Review of Pressurized Thermal Shock Issues Draft Report, June 1982.
- E.4 Methods of Compliance with Fracture Toughness and Operational Requirements of 10 CFR 50, Appendix G, B&W-10046A, Rev. 1, July 1977, by
 H. S. Palme, H. W. Behnke, and W. S. Keysworth, Babcock and Wilcox.
- E.5 Effects of Residual Elements on Predicted Radiation Damage to Reactor Vessel Materials, Regulatory Guide 1.99, Revision 1, April 1977.
- E.6 Prediction of the Shift in the Brittle-Ductile Transition Temperature of LWR Pressure Vessel Materials, Edited by J. J. Koziol, Prepared by a Task Group of Subcommittee 6 of the Metal Properties Council, April 6, 1981 as a Report to ASTM Committee E10.02. To be published by ASTM.
- E.7 LWR Pressure Vessel Irradiation Surveillance Dosimetry Quarterly Progress Report, Jan.-Mar. 1982, HEDL-TME 82-18. To be published.
- E.8 J. R. Hawthorne, "Status of Knowledge of Radiation Embrittlement in USN Reactor Pressure Vessel Steels," NUREG/CR-2511, NRL Memo Report 4737, Feb. 1982.
- E.9 J. D. Varsik and S. T. Byrne, "An Empirical Evaluation of the Irradiation Sensitivity of Reactor Pressure Vessel Materials," Effects of Radiation on Structural Material, ASTM STP 683, 1979, pp 252-266.

11/13/82

E-10

PTS RPT APP E

- E.10 Findings of the Ad Hoc Working Group on Selection of RT_{NDT}, Memorandum M. Vagins to S. Hanauer, Aug. 30, 1982.
- E.11 "Evaluation of Irradiation Response of Reactor Pressure Vessel Materials Prepared by Combustion Engineering, Inc., (J. D. Varsik principal investigator) for Electric Power Research Institute, Research Project RP 1553-1, Final Report, to be published.

•



\$

FIGURE E-1: COMPARISON OF REG. GUIDE 1.99 REV. 1 AND GUTHRIE MEAN PLUS 2-SIGMA TREND CURVES



Ker



KOE 10 X 10 TO THE INCH+ / X HU INCHES KEUFFEL & ESSER CO MAN IN \$3.4



) †

.



FIGURE E-5: COMPARISON OF GUTHRIE AND MPC FORMULAS FOR THE MEAN VALUES OF ΔRT_{NDT} FOR REPRESENTATIVE COPPER AND NICKEL CONTENTS









۰.

FIGURE E-7: SCHEMATIC DIAGRAM OF TREND CURVES

٠.

43

,"

APPENDIX F

PRESSURE VESSEL FAST NEUTRON FLUENCE UNCERTAINTY

F.1 Introduction

-

The following discussion deals with the components of the staff's fast fluence (>1 MeV) predictive calculational uncertainty.

There are two major sources of uncertainty in fast fluence computations, i.e., (a) the uncertainty which results from the measured values of the fluence used in benchmarking the computer codes, and (b) computational, which originates from uncertainties of input quantities to the code.

F.2 Benchmarking Uncertainty

The prediction of the calculation is benchmarked to measured values of carefully performed experiments. The benchmarking process has been instrumental in recent improvement of the uncertainty as shown in Fugure F-1. It can be seen that in the early years of commercial nuclear power the predictive uncertainty was very large. Figure F-1 represents the FSAR predicted values of the fluence and their comparison to a posterior measured value with the surveillance capsule. Measured values from the surveilance capsules and the Pool Critical Assembly improved the predictive capability in the 1970s and is shown in 1980-81 when surveillance capsules were removed. The staff has a techniccal assistance program to BNL to benchmark the neutron transport code DOT 3.5 and verify the fluence values in the eight pressure vessels which have been thought to have marginal toughness. At this time the benchmarking is nearly complete.

The benchmarking includes data from the following:

(i) The Pool Critical Assembly pressure vessel dosimetry benchmark experiment (Ref. F.2).

F-1

(ii) The ANO-1 surveillance capsule and reactor cavity flux measurements (Refs. F.3, F.4, F.5, and F.6).

EPRI-sponsored measurements in the reactor cavity provide flux values to an estimated accuracy of ± 15 percent (undesignated distribution). Surveillance capsule measurements are being used to adjust the fluence calculated on the inside of the pressure vessel.

- (iii) Fort Calhoun surveillance capsule.
- (iv) Main Yankee surveillance capsule.

Figure F-2 shows a typical configuration of a surveillance capsule. The overall length corresponds to that of a fuel assembly and contains an upper, middle, and lower tensile monitor compartments. Tensile specimens are housed in this section along with radiation monitors (Figure F-4). Charpy impact specimens are housed in separate compartments (Figure F-3). Typical locations of surveil-lance capsules are shown in Figure F-5.

The causes of uncertainty in dosimetry measurements are related to reaction rate cross-sections, the photofission correction, counting calibration, flux-time history, etc. The overall benchmarking uncertainty is ± 15 percent (1 σ).

F.3 Computational Uncertainty

Computational uncertainty results from uncertainties in cross-section data (inelastic scattering of iron is a particular source of error), modeling, numerical methods, source representation, geometry, etc., which are inputs to the DOT 3.5 code. The DOT series of codes are two-dimensional neutron transport codes based on finite differencing with anisotropic scattering in (x,y), $(1,\Theta)$, or (R,z) geometries. The DOT 3.5 version is operational at BNL (Ref. F.7)

In order to evaluate calculational uncertainties and provide an additional independent assessment of the uncertainty, a direct parametric analysis is

F-2

being performed. In this analysis major uncertainty components (e.g., source representation, geometry, cross-section, etc.) have been identified and are being quantified. DOT sensitivity calculations are being performed to propagate these uncertainties and determine their effects on vessel fluence and ΔRT_{NDT} (Ref. F.8). The expected uncertainty is ±15 percent (1 σ).

We estimate the overall predictive uncertainty to be ± 20 percent (1 σ) comparable to ± 15 to ± 20 percent recently claimed by the vendors (Refs. F.9 and F.10).

The above is illustrated in diagrammatic form in Figure F-6 which illustrates the overall uncertainty, its components, and the sources of the experimental uncertainty.

•*

REFERENCES

- F.1 W. N. McElroy, et al., "Surveillance Dosimetry of Operating Power Plants," HEDL-SA-2546, NRC Ninth Water Reactor Safety Information Meeting, NBS, Washington, D.C., October 1981.
- F.2 W. N. McElroy, Editor, "LWR Pressure Vessel Surveillance Dosimetry Improvement Program," NUREG/CR-1861, July 1981.
- F.3 N. Tsoulfanidis, et al., "Neutron Fluxes in the Core and the Cavity of a PWR," EXPRI RP-827-4, draft, February 1981.
- F.4. A. L. Lowe, Jr., et al., "Analysis of Capsule ANI-E from Arkansas Power and Light Company, Arkansas Nuclear One Unit 1, "BAW-1440, Lynchburg, Virginia 24505, April 1977.
- F.5. R. Mearker, et al., "development and Demonstration of an Advanced Methodology for LWR Dosimetry Applications," EPRI (RP-1399-1) confidentiai, draft.
- F.6 J. Grundl, et al., "NRC-EPRI Studies of Pressure-Vessel-Cavity Neutron Fields," NRC Ninth Water Reactor Safety Information Meeting, NBS, Washington, D.C., October 1981.
- F.7. DOT-3.5, "Two Dimensional Discrete Ordinates Radiation Transport Code," RSIC Computer Code Collection CCC-276, 1976.
- F.8 J. Carew, BNL, Personal Communication, December 7, 1981.
- F.9 "Letter Report on Reactor Vessel Brittle Fracture Concerns in B&W Operating Plants," Document ID 77-1125756, May 15, 1981.

F.10. "Thermal Shock to Reactor Pressure Vessel," letter from R. W. Jurgenson (Chairman, Westinghouse Users Group) to D. G. Eisenhut, May 14, 1981.

÷

••

.



FIGURE F-1 Estimated Exposure Parameter Uncertainties Obtained from FSAR and Surveillance Capsule Reports.

S.



њ., -

FIGURE F-2





Ć





ч Ч

 \bigcirc

:



FAST FLUENCE UNCERTAINTY



ŝ

- 253 -

•

-

APPENDIX G OVERCOOLING EVENT SEQUENCES LEADING TO POTENTIAL PIS CONDITONS

154 -

G.1 Critical Parameters

G.1.1 <u>Temperature</u>

The temperature of concern is the fluid temperature at the reactor pressure vessel (RPV) inner surface (at the point of interest, typically a weld). For PTS analyses, the initial steady-state temperature is taken to be 550°F. Final temperatures of 350°F and less are expected to be potential PTS initiating events.

G.1.2 Rate of Cooldown

The rate of cooldown is also important. Events which can be controlled within the limits of Appendix G heatup/cooldown limits are not considered to be PTS initiating events. As a guide, transients which do not exceed 100°F per hour cooldown may be excluded from PTS consideration.

Present fracture mechanics analysis methods input the temporal coolant temperature behavior in a stylized manner approximated with an equation of the form:

```
T = To - (To - T_f) (1 - exp ( - \beta t))
Where To - initial temperature, <sup>o</sup>F
T_f - final temperature, <sup>o</sup>F
\beta - cooldown parameter, per minute
t - time, minutes
```

Although using this formulation to characterize the coolant temperature temporal behavior is advantageous from a fracture mechanics standpoint, it does not consider variations in the cooldown rate which can occur as a result of operator action, and control or protection systems operations. Also, the predicted final

temperature is uncertain, as stored heat in the primary system metal and the decay heat output from the reactor core will increase the coolant temperature once action is taken to terminate the cooldown.

When approximating the coolant temperature temporal behavior as a result of specific initiating event with this form of stylized equation, the selection of T_f and β (the final temperature and the cooldown parameter) which best approximates the actual transient behavior, requires some engineering judgment. In general, the final temperature is selected as the lowest calculated value and the cooldown parameter is either the natural (for example best fit) cooldown parameter or an adjusted value for cases where the temperature increases following termination of the cooldown. The cooldown parameter used for these analyses has been adjusted to account for this temperature increase and is based on the Westinghouse approach. This approach considers the fracture mechanics response to the actual temperature transient and the fracture mechanics response to the stylized formulation with the adjusted β value, designated β^* . The designated β^* value is obtained from:

$$\beta^{\star} = 2/t^{\star}$$

where t* is the time of lowest temperature, minutes and β^* is never less than the natural cooldown rate.

G.1.3 Pressure

The predominant stress associated with PTS is the thermal stress associated with the rapid cooldown of the RPV wall. It is noted that the thermal stress alone is not sufficient to cause loss of RPV integrity. The other important stress is due to pressure. Maintaining system pressure relatively high or repressurizing following a cooldown is of primary concern. Control of the primary system pressure is achieved by operator action to terminate chargin; and HPI flow to the primary. Other systems, such as letdown, pressurizer heater and pressurizer sprays are not considered here for pressure control.

A review of current Westinghouse guidelines for HPI termination indicates that primary system pressures of 2250 psig could be obtained prior to HPI termination

G-2

11/12/82

PTS APPENDIX G

for certain coldown events, such as steam line breaks. This pressure can also be reached as a result of delayed operator action in terminating charging or safety injection flow. Therefore, 2250 psig was considered appropriate for this evaluation unless a lower pressure can be justified when a specific transient is considered.

G.1.4 Frequency

The expected frequency of occurrence of an event which could result in PTS conditions is an important parameter because it defines the relative probability of a PTS initiator. When combined with other probabilities and consequences associated with specific events in a sequence, the risk due to PTS from that sequence can be estimated. Theoretically, by adding the risk from all significant event sequences, the total risk can be obtained. The practical difficulty is obtaining reasonable assurance that all significant event sequences are in fact included.

G.1.5 Uncertainty in T_f , β , and P Values for Characterization

It is not nossible, at this time, to determine the uncertainty in either T_f or β as used to characterize a PTS transient. The computer programs and models used to determine the thermal-hydraulic characteristics of the PTS transients are still under review by the staff. The input data and assumptions used in these computer programs and models range from "best estimate" to "conservative for PTS," and in some cases inconsistent data are used to enhance the cooldown or the cooldown rate. While the staff has endorsed the use of best estimate analyses to characterize the PTS transient behavior, we relied very heavily on the analyses presented by Westinghouse to support our evaluation. (Reference G.1)

G.2 Events to be Considered

The selection of events to be considered is based on the event sequences identified by Westinghouse. The staff has reviewed the sequences, the associated frequencies of occurrences, the β 's and T_f 's. Based on a number of

11/12/82

technical meetings with Westinghouse, we have established a set of events of significance to the PTS issue which have been identified as Main Steam Line Break (MSLB), Small Steam Line Break (SSLB), Small Break Loss-of-Coolant Accidents (SBLOCA), and Steam Generator Tube Rupture (SGTR).

In order to characterize individual event sequences within each of these groups, certain additional parameters have been identified which determine the significance of these sequences as a PTS challenge.

The level of decay heat present during an initiating event is an important parameter in the cooldown from a given transient. The level of decay heat is related principally to the operational status (full power operation, hot zero power, other) immediately preceding the transient. The frequency of challenging event sequences are thus differentiated by the operational status of the plant.

The time allowed prior to initiation of proper operator action is another parameter that is important in some sequences. This variable has been used as a parameter in the results which characterize certain sequences that are presented below.

G.2.1 Main Steam Line Break (MSLB)

The MSLB, with a break area larger than 6 inches equivalent diameter, results in a rapid cooldown of the primary system. The final downcomer coolant temperatures can be on the order of 200°F, dependent on the decay heat level and operator action time to terminate auxiliary feedwater. The system will repressurize as a result of continued safety injection flow, and may repressurize in excess of 2000 psig dependent on the time operator action is taken to terminate safety injection. Stored metal heat and decay heat also aid in the repressurization.

Frequency of Main Steam Line Break

There have been three large breaks in steam-bearing lines in commercial nuclear plants. Two (at H. B. Robinson in 1970 and Turkey Point in 1971) occurred during hot functional tests prior to commercial operation and caused failures in

G-4

PTS APPENDIX G

branch lines or headers associated with the steam generator safety valves. These two failures were attributed to underdesign of the piping at the failed locations. A recent failure occurred at Oconee in a line off of the high pressure turbine exhaust. The preliminary evaluation attributed the failure to wall thinning by steam erosion.

The estimated median frequency of a pipe break (>6 inches) in the primary coolant system was 10-4/RY in WASH-1400 based on evaluations of several data sources including nonnuclear experience. The steam lines including bypass lines (>6 inches) between the steam generator and the main steam isolation valves in 3-loop Westinghouse plants are about 1.7 times longer than comparable primary system piping which would yield an equivalent steamline break frequency of 1.7×10^{-4} /RY. Median estimates of the frequency of steamline breaks (up to MSIV) in the Zion and Indian Point Probabilistic studies were 4 x $10^{-4}/RY$ and $10^{-4}/RY$, respectively, based on no major failures in these lines during commercial nuclear operations. There are about 600 reactor-years of commercial operation at nuclear power plants, so the recent Oconee failure represents a point estimate of 1.6 x $10^{-3}/RY$ for steamlines downstream of the MSIV which have a lower level of quality control than the steamlines upstream of the MSIV. The length of steam piping (>6 inches) downstream of the MSIV may vary from plant to plant but we assumed it to be three times the length of piping upstream of the MSIV, so that an equivalent failure frequency estimate for the piping upstream of the MSIV would be about $5 \times 10^{-4}/RY$.

The two events at Robinson and Turkey Point occurred prior to commercial operation during shakedown tests expected to identify plant design and operating deficiencies so that these data should not be given equal weight with the Oconee event. Bush (Reference G.2) estimated that about 70% of the piping failures occur in the first two years of operation. Thus if one assumes that the Robinson and Turkey Point events are part of the first two years' operation at all plants (140 RY), the effective steamline pipe break frequency for mature plants would be estimated to be about 4 x $10^{-3}/RY$.

G-5

."

n

PTS APPENDIX G

Based on an evaluation of several sources of data, Bush (Reference G.2) estimated the frequency of large disrupture failures to be between 10^{-5} and 10^{-6} /RY and also pointed out that "virtually any type of piping damage may be considered to result from some kind of human error or lack of knowledge in design, fabrication, or operation."

A probabilistic fracture mechanics analysis of pipe fracture in the primary coolant loop of a PWR plant was presented in Reference G.3. This study estimated very low pipe break frequencies $<10^{-6}$ /RY based on cyclic stresses and seismic events imposed in service. These results are limited by the scope of the analysis. Thus, these results are given little weight compared to experience.

The estimates of steamline break frequency appear to cluster around 10^{-4} /RY. A frequency of 1.7 x 10^{-4} /RY is used for evaluating steamline breaks for screening purposes. It is estimated that the upside uncertainty is a factor of 10 and the downside a factor of 100 based on the preceding discussion. The Westinghouse Owners Group contends that the frequency of major steamline breaks (>6 inches) is much smaller based on Reference G.3.

We considered other failures in addition to the initiating event that might result in extended HPSI operation such as failure of reactor coolant pump seals, unisolated letdown lines, and a stuck-open PORV. We have assumed that the RCP seal cooling will be maintained during this event; however, this assumption should be confirmed for the Westinghouse plants. Therefore, the conditional probability of RCP seal failures is considered to be low. Similarly, we have assumed that the letdown line will be automatically isolated and will not automatically open on reseting SIS or containment isolation signals. This assumption should be confirmed for the Westinghous? plants. In some plants with high shut-off head ECCS pumps, the PORV may be challenged. The conditional probability of a PORV sticking open is estimated to be $10^{-2}/D$ in WASH-1400. The conditional probability of an operator failing to isolate the PORV in ten minutes is about 2 x $10^{-2}/D$ from Reference G.4. The combined event frequency is not significant $(10^{-6}/RY)$. Thus, only simple

G-6

PTS APPENDIX G

2.58

large steamline breaks without any additional failures will be considered for this sequence category.

26

Operator Actions

The operator is trained to isolate the faulted steam generator in a steamline break event based on low pressure in one steam generator compared to the others. Operator action time to isolate the feedwater to the faulted steam generator is presented in Figure II-4 of Reference G.5 based on simulator experience. Using human error estimates from Reference G.6 for high stress situations and assuming three operators in the control room, human error probabilities (HEPs) for failing to recognize that the AFW has to be isolated are estimated to be 0.7 at 5 minutes and 0.04 at 10 minutes compared to 0.3 and 0.9 from Figure II-4 of Reference G.5. We will assume HEPs of 0.5 at 5 minutes and 0.1 at 10 minutes. 3 x 10^{-3} at 30 minutes (based on WASH-1400 estimate for switchover from injection ro recirculation) and 10^{-3} at 60 minutes for purposes of this screening analysis. These human error split fractions are presented in an event tree format in Figure G.1. The conditional probabilities for successful termination of auxiliary feedwater in specific time intervals are indicated in Figure G.1. The estimated uncertainty in the probability of successful operator action is a factor of 2 to 10.

The operator could potentially control the primary system pressure by terminating high pressure injection and venting through the PORV. This is a highly dynamic situation with the pressure recovering rapidly after water has reentered the pressurizer. The generic procedure guidelines have an allowable HPSI termination pressure of 2000 psi. For these screening analyses we have assumed that the pressure will be at 2250 psi.

Operational Status of the Plant at the Time of the Initiating Event

The operational status or power split data developed by Westinghouse considers the decay heat level following each outage and startup (the assumed point of the PTS initiating event) based on operating data. The staff has reviewed the operating history over a period of four years for one Westinghouse plant. The data indicates that the time at power (greater than 50% rated) is on the order of 90% to 95%, as shown in Figure G.2.

For the purpose of this evaluation, the power split selected is 0.9 for hot full power (HFP) and 0.1 for hot zero power (HZP). HFP is equivalent to decay heat levels of 1% or greater, HZP is equivalent to decay heat levels of 0.25% or less. It is noted that Westinghouse uses three levels: 0.75 for HFP, 0.1 for HZP, and 0.15 for intermediate power levels (0.25% to 1% decay heat), based on Westinghouse operating experience. HFP cases, unless otherwise stated, are cases run by Westinghouse using the conservative assumption of being at hot shutdown or hot zero power with decay heat levels of 1% or greater. The larger secondary side mass inventory enhances the cooldown, resulting in a conservatively low temperature calculation.

NRC Staff Characterization of the MSLB

Two cases are used to characterize the MSLB for this evaluation. The first case is for initiation from hot full power and nominal operating conditions. The data was obtained from the Westinghouse Owners Group (WOG) Emergency Response Guidelines E-2, Loss of Secondary Coolant (September 1, 1981). The case selected is for a 0.6 sq ft break. The data have been extrapolated for times greater than 10 minutes. This event is summarized in Table G.1 as a function of operator time to terminate auxiliary feedwater to the faulted steam generator. Also presented are the parameters for initiation from hot zero power. These data are taken from the Westinghouse data for the 0.77 sq ft break case.

WOG Characterization of the MSLB

The data used in the Westinghouse evaluation for the HFP case is conservatively based on the use of initial operating conditions at hot zero power. The data are presented in Table G.2.

G-8

シレ

G.2.2 <u>Small Steam Line Break (SSLB) or Stuck Open Steam Generator Safety/</u> Relief Valve (SOSGRV)

200

The SSLB, or stuck open SG safety/relief valve, can result in an overcooling transient similar to the MSLB but of a longer duration due to the smaller break size. This event is expected to occur at a much higher frequency than the MSLB.

Frequency of Small Steamline Break

**

The steamline contains several pressure relief devices to accommodate rapid pressure increases caused by plant trips or other perturbations. There are turbine bypass lines that are connected to the steamline downstream of the MSIV and discharge into the condenser. Valves in these lines are designed to open at pressures above operating conditions and provide normal decay heat removal following plant trip. Since these lines can be isolated by the MSIV, we believe that the frequency of a secondary depressurization caused by malfunctioning of these valves is bounded by potential relief valve failures upstream of the MSIVs.

For very rapid steam pressure increases associated with turbine trip or inadequate MSIV closure, the atmospheric dump valves or safety valves upstream of MSIV are likely to be challenged and have a finite probability of sticking open. In Reference G.1, Westinghouse estimated the frequency of a single stuck-open relief valve upstream of the MSIV to be $3.4 \times 10^{-3}/\text{RY}$ and CE estimated the frequency to be $1.5 \times 10^{-2}/\text{RY}$ in Reference G.7. A survey of events relevant to PTS has indicated only two events in about 350 RYs having a small secondary depressurization that resulted in relatively high cooldowns. For purposes of this screening analysis, we will assume a frequency of $10^{-2}/\text{RY}$ for small steamline breaks (equivalent to a stuck-open steam generator safety/ relief valve) with an uncertainty spread of a factor of 3.

We considered additional failures besides the initiating event that might result in extended HPSI flow. Specifically, we looked at the same failures explored for large steamline breaks. The potential impact of RCP seal failures and an unisolated letdown line are judged to be small as discussed

PTS APPENDIX G

under large steamline breaks. The conditional probability of an unisolable open PORV is less than 5×10^{-4} which would yield a combined small steamline break-small LOCA of less than 5×10^{-6} /RY. We do not have a thermal-hydraulic analysis of this event for extended time periods and, therefore, cannot evaluate its contribution. This type of event should be explored as part of the long-term program. Its estimated frequency is sufficiently low that it does not represent an immediate concern.

Operation Actions

The severity of this event can be reduced by terminating auxiliary feedwater to the faulted system generator. For this analysis we will assume the same operator action probabilities as were used in the section addressing main steamline breaks for similar actions.

NRC Staff Characterization of the SSLB

Two cases are used to characterize the small steamline break. The data are taken from the Westinghouse conservative bounding analysis performed at HZP for a 0.22 sq ft break. A decay heat level of 1% is used to represent HFP and a decay heat level of 0% for HZP. The actual cases analyzed were from HZP conditions, therefore, the HFP case is a conservative representation. The transient summary is presented in Table G.3 as a function of operator time to terminate auxiliary feedwater to the faulted steam generator for the initial conditions of HFP and HZP.

WOG Characterization of the SSLB

The data used in the Westinghouse evaluation for these cases appear to be based on a break size greater than 0.22 sq ft. The data are presented in Table G.4.

G.2.3 Small Break Loss-of-Coolant Accident (SBLOCA)

The SBLOCA was considered in two categories. The first category is for break sizes less than 2 inches equivalent diameter. For this category, the safety

injection (or makeup) flow exceeds the break flow. For breaks in this range, the break cannot remove all the decay heat generated in the core, and natural circulation, using the steam generators to remove the decay heat not removed by the break, will occur, maintaining loop flow. Within this first category are RCP seals, μ rimary PORV or safety valve leakage or failure as well as actual primary piping leaks. The cooldown rates for these SBLOCAs are not expected to violate Appendix G limits (less than 100°F per hour cooldown). For these reasons this category of small breaks was excluded as not being significant for this evaluation.

164

Typically a SBLOCA in this range will depressurize the primary system to the pressure corresponding to the saturation pressure of the secondary (steam side). The primary coolant temperature remains just above the secondary coolant temperature in order to maintain a decay heat removal path until sensible heat addition to the incoming SI water is capable of removing this energy. At that time RCS pressure and temperature should decrease slowly. If SI flow at this equilibrium pressure meets or exceeds the break flow, the RCS should remain in natural circulation. Operator actions which reduce the secondary pressure may have a favorable impact on natural circulation, but would increase the cooldown rate.

The second category includes the break size range from 2 to 6 inches equivalent diameter. For these breaks, the safety injection flow is less than the break flow, resulting in a net mass loss of primary coolant. In addition, for breaks greater than 2 inches, the break can remove all of the core decay heat. The combined effect of mass loss and energy removal by the break will result in an extended loss of natural circulation flow in the coolant loops. The continued injection of cold safety injection water into the stagnant loops has the potential to allow relatively unmixed safety injection water to rapidly reach the downcomer and contact the vessel wall. The exact break size where loss of flow occurs is dependent on the safety injection flow rate (and makeup flowrate), the break location, the decay heat level, and the SG (heat sink) performance.

The factors which affect the rate of cooldown following a SBLOCA are:

4

44

Break Location Break Size SI Flow Decay Heat Secondary Pressure (steam dump and feedwater)

Frequency of Small SBLOCA 2 to 6 inch range

The staff has reviewed the Westinghouse thermal hydraulic calculations and has determined that small LOCAs which have equivalent break sizes of 2 to 6 inches will result in stagnated loop conditions and primary system pressures less than 1000 psi. We have examined events that might result in stagnated loop conditions such as LOCAs in the 2 to 6-inch range. There are several small diameter pipes (in the range of 2 to 4 inches) connected to the main primary system piping. These include charging and letdown lines, RTD bypass lines, pressurizer spray and scoopline, PORV line, and safety injection lines. We have also considered the following events.

- (a) pipe breaks less than 2 inches with loss of one train of HPSI,
- (b) unisolated open PORVs with the loss of one train of HPSI,
- (c) failure of all RCP seals due to loss of cooling.

A median frequency of 3 x $10^{-4}/RY$ for small LOCAs in the range of 2 to 6 inches is based on WASH-1400, and the uncertainty band is ± a factor of 10. Assuming a lognormal distribution, the estimated mean frequency would be $1.5 \times 10^{-3}/RY$. The Westinghouse Owners Group estimated the mean frequency of LOCAs in this range to be 6 x $10^{-4}/RY$.

In evaluating small LOCAs, we have focused on events which stagnate the flow in the pressure vessel and involve a monotonic decrease in primary system pressure. Other stagnated loop flows type events may result in repressuri; ing

G-12

PTS APPENDIX G

the primary system following an extended period of HPSI operation with stagnated flow and low pressure. There are several events that have been considered in this category. These include:

シビル

- (a) Stuck open safety value that subsequently closes after the vessel is chilled.
- (b) Considerable RCS inventory is lost during the initial surge in an ATWS event. If the reactor coolant pumps are tripped, no primary system leaks occur, and the HPSI operates for an extended period of time, low vessel temperatures may be obtained with high system pressures.
- (c) A small-small LOCA initially with only one train of ECCS operating may result in stagnated loop conditions. If the other ECCS trains are subsequently recovered, high pressures may be achieved after the vessel is chilled.
- (d) A small-small LOCA with no cooling in one steam generator may lead to high pressure low temperature conditions.
- (e) A small steamline break with a subsequently stuck open PORV may lead to extended HPSI flow with a stagnated loop (caused by isolating the faulted steam generator).
- (f) Loss of all feedwater that results in losing primary system inventory with a subsequent recovery of feedwater and HPSI operation without loop flow may also lead to these conditions.

Engineering analyses are not available for all of the events postulated above. The most significant is thought to be the stuck open safety valve that subsequently recloses. In Ref. G.4, it is indicated that the Westinghouse Owners Group estimated the frequency of a stuck open safety valve to be $\sim 10^{-5}/\text{RY}$ assuming the PORV's are blocked 90% of the time. This estimate is based on a conditional failure to close probability of $10^{-3}/\text{D}$. The review presented in Ref. G.4 reestimated the conditional failure probability to be $\sim 10^{-2}/\text{D}$ so

4

٩,
that the frequency of a stuck open safety value in Westinghouse plants would be about $10^{-4}/RY$. This frequency estimate may be somewhat high; however, considering the other potential events noted above, it is prudent for screening purposes to use a frequency of $10^{-4}/RY$ for extended HPSI operation with loop stagnation and high pressure.

761

NRC Staff Characterization of SBLOCA

Three breaks in the hot leg for a typical 3 loop PWR were reviewed. These breaks were of equivalent diameters of 1, 2 and 3 inches. For the 2-inch break, loop natural circulation is lost at about 12 minutes, and for the 3inch break at about 7 minutes. In addition, three cold leg breaks, 2, 3, and 4 inches were also reviewed. The hot leg breaks are bounding for those cases where circulation is lost due to the higher energy removal rates. Loss of circulation occurs later for an equivalent cold leg break. The loss of circulation is dependent on the break location, the break size, the decay heat level, and operation of the RC pumps (pumps off). For breaks 2 inches and larger, circulation is assumed to be lost. For breaks less than 2 inches, the cooldown rates are less than 100°F per hour and therefore were not included in this evaluation since such slow cooldown does not present a PTS concern.

The most limiting break size for PTS concerns is the 2-inch break. The primary system pressure remains relatively high (1000 psig) for an extended period of time (30 minutes) following loss of loop natural circulation flow. For larger breaks, the pressure will be lower as a result of the faster time to uncover the break and depressurize the system.

The transient characterization based on recent work performed by SLI (Reference G.8) was used to determine the appropriate mixing control volume. This volume is the cold leg piping from the loop seal to the vessel plus one-half of the vessel downcomer volume. The wall heat from the piping, the vessel, and the core support barrel was included in the analysis. The results of this analysis show a leveling of the coolant temperature at about $125^{\circ}F$ (assuming a safety injection temperature of $60^{\circ}F$) for the time range of interest, which is 2000 to 3000 seconds after loss of flow is assumed.

On August 11, 1982, the staff solicited advice from leading thermal-hydraulic experts to determine how mixing during no-loop flow conditions could best be treated for this evaluation. In particular, we were interested in determining what was the most realistic acceptable method for predicting mixing under no-flow conditions. There was a consensus at the meeting that the method presented by Leviy in Reference G.8 was appropriate; however, stored energy in the vessel metal, the lower plenum volume, and the pipe volume upstream of the safety injection location to the pump should also be accounted for. In addition these volumes should be included in the mixing volume.

Westinghouse presented its evaluation at this meeting. There was general agreement on the methodology used by Westinghouse. The Office of Nuclear Regulatory Research (RES) requested three of their experts to perform independent analyses of this event, to confirm the Westinghouse results.

The RES analyses have been reviewed by the staff and the results of these analyses (Reference G.9 and G.10) have been used to characterize the SBLOCA event. The models used have been compared to recent Creare experiments, for a 1/5 scale geometry (Reference G.11). These comparisons indicate that the models are a reasonable representation of the mixing phenomena under no-loop flow condition. The characterization used for this evaluation for this event sequence is presented in Table G.5.

WOG Characterization of SBLOCA

For breaks in the cold leg, the Westinghouse evaluation uses a 1-inch break, which bounds the RC pump seal failure case and assumes no operator action. For decay heat levels of 1%, the T_f is 315°F and β is 0.007 per minute. At 0% decay heat, T_f is 105°F and β is 0.007 per minute. For breaks in the hot leg, Westinghouse uses a 0.75-inch break, slightly less than the area of a typical PORV (1.4 inches). For decay heat levels of 1%, T_f is 528°F and β is 0.004 per minute. At 0% decay heat, T_f is 166°F, and β is 0.004 per minute. Westinghouse excluded larger breaks from the PRA evaluation. For these breaks, crack extension is accepted. However, as a result of the decreasing pressure, the Westinghouse fracture mechanics analyses, which includes warm prestress, has shown that crack arrest occurs for these cases.

G-15

11/12/82

PTS APPENDIX G

269

G.2.4 Steam Generator Tube Rupture (SGTR)

The SGTR transient is governed by operator actions prescribed in current plant emergency procedures. The tripping of the RC pumps, the use of the PORV during a SGTR event, and the subsequent cooldown and depressurization to isolate the break all impact on the transient. When the RC pumps are tripped, natural circulation in the faulted loop can be lost for a short period of time, since the recommended action is to isolate the secondary side of the faulted steam generator, thus eliminating its heat removal capability, prior to the recovery procedures. Delayed safety injection termination can enhance the cooldown and increase the consequences of the event. In general, the SGTR event will not result in loss of vessel integrity when current plant procedures are observed, as the downcomer temperature is expected to remain greater than 325°F. The SGTR event does become a potential initiator for PTS if additional failures are included in the event sequence. Two of these are of definite interest. They are failure by the operator to terminate safety injection in accordance with the prescribed procedures and a stuck-open steam generator safety valve.

Frequency of SGTR

The estimated frequency of a steam generator tube rupture event is $2 \times 10^{-2}/\text{RY}$ (4 events in about 250 RY for Westinghouse plants) which by itself is not a particular PTS concern if the HPSI is terminated in a timely fashion (~30 minutes). The Ginna-type event (1982), which resulted in extended HPSI operation, may lead to PTS conditions of concern. Since this is one event out of four, the estimated frequency of SGTR events that lead to extended HPSI operation (1 hour) is 5 x $10^{-3}/\text{RY}$ unless subsequent guidelines are effective in reducing the frequency of this event.

We have considered other failures in addition to the SGTR which may result in extended HPSI operation. In the Ginna event, the operator isolated the atmospheric dump valve on the faulted steam generator and challenged the steam generator safety valve. Assuming a conditional probability of 10^{-2} /D for failure of the safety valve to close (WASH-1400), there is a potential event of an SGTR with a stuck-open SG safety with an estimated frequency of

G-16

169

 $2 \times 10^{-4}/RY$. The frequency of this scenario can be reduced if the HPSI is terminated in a timely fashion (~30 minutes) and/or the atmospheric dump values are not isolated.

210

Another potential compound event is extended HPSI operation which results in flooding the steamline and it subsequently fails. The staff has estimated the probability of failure of water-filled steamlines to be negligible. If we assume a conditional failure probability of 10^{-3} for water-filled steamline failure, then the estimated frequency for a combined event of SGTR followed by a steamline break is 10^{-5} /RY (there have been two SGTR events with water in the steamlines).

The PORV may be used to depressurize the primary system. The conditional probability of having an unisolated open PORV is 5×10^{-4} as discussed previously. The frequency of this combined event is estimated to be $10^{-5}/RY$. We do not have thermal-hydraulic calculations to describe this event for an extended time period. We would judge that it may be bounded by a small LOCA event which has an estimated frequency of $3 \times 10^{-4}/RY$.

We also considered the potential for developing excessive steam generator tube leaks following steamline breaks. We had insufficient information to evaluate this event in the short term; however, they should be pursued in the longer term study of pressurized thermal shock.

In summary, we will use the following frequencies for SGTR events for this screening process.

(a)	simple SGTR with HPSI termination in 1 hour	$5 \times 10^{-3}/RY$
(b)	SGTR with stuck-open secondary safety valve	$2 \times 10^{-4}/RY$
(c)	SGTR with subsequent steamline break	$1 \times 10^{-5}/RY$

There are considerable uncertainties in these event frequencies which may be as much as a factor of 10.

÷.

PTS APPENDIX G

NRC Staff Characterization of SGTR

The base case SGTR is similar to the one used by westinghouse $T_f = 325^{\circ}F$, $\beta = 0.12$ per minute, at 1000 psig. This case assumed operator action as prescribed in current plant procedures. Delayed SI termination will result in lower final temperatures. Based on the Westinghouse data, the characterization is as follows, as a function of operation time to terminate safety injection flow.

The data for the SGTR with a stuck-open safety valve is taken from the Westinghouse study case for the SGTR with a small break inside containment (SLB). The data for the case with a steamline break is taken from the Westinghouse study case for an SGTR with a small break outside containment (SOSV). The transient characteristics are presented in Table G.6.

WOG Characterization of SGTR

The Westinghouse data is used with the exception of the base case β value used by Westinghouse of 0.09 per minute. Based on a review of operating experiences, a value of 0.12 per minute appears to be more appropriate and is used for this evaluation.

G.2.5 Excess Feedwater

The contribution to PTS from an excess feedwater transient is considered to be negligible due to the low frequency of occurrence for Westinghouse and Combustion Engineering plants and is not considered in this evaluation. Because of the difference in steam generator water level and inventory and main feedwater design for Babcock and Wilcox plants, the excess feedwater transient should be considered as a potential PTS initiating event for B and W plants.

G.2.6 Other Events

There are numerous other events, or variations (multiple failures) of the above described events, which may also warrant consideration in a PIS evaluation of this nature. However, due to the scoping nature of this

PTS APPENDIX G

evaluation, combined with the time constraints to produce this preliminary evaluation, all of these additional events could not be systematically identified and evaluated. These other events include situations where loop flow is lost, as a result of loss of the heat sink (SG). For example, Westinghouse has indicated that secondary side depressurization (MSLB and SSLB) might result in loss of natural circulation some time after 30 minutes. dependent upon the transient specifics such as break size, delayed operator action, and decay heat level. Once loop flow is lost, the cooldown will be accelerated due to the nonmixing of the safety injection flow. In addition to cases where loop flow can be lost, there are many other events which should be considered as part of future, detailed evaluations. These are generally scenarios which include multiple failures (for example, MSLB and subsequent SGTR; multiple RC pump seal failure due to loss of component cooling water; and feed and bleed operations). A detailed PTS event tree analysis would be required to determine those events which could result in PTS conditions. Such an evaluation would have to consider plant-specific information, such as safety injection pump shut-off head, safety injection and auxiliary feedwater flow rates and temperatures, and specific control and protection system setpoints. This evaluation would have to be performed as part of a long-term PTS effort.

The above evaluations gave no credit for the operator reducing primary system pressure to minimize the potential for PTS and also assumed that the operator would not deliberately put the plant into a PTS condition after it has been stabilized following some event.

There have been events such as the TMI-2 accident in which the plant was repressurized even though low fluid temperatures were present. We cannot adequately evaluate the potential for the operator repressurizing the primary system because of some unforseen situation or dilemma. However, the potential impact of such events can be considered by combining operational experience with postulated events.

G.2.7 Limitations

The selection of events noted above to be analyzed for establishing a screening criterion were based on the dominant sequences identified by the Westinghouse Owners Groups in the June 22, 1982 meeting. In addition, the staff has considered some limited variations of these events in an uttempt to identify additional failure that may lead to significant PTS conditions. Transients with significant reliance on ad-hoc operator action will be based on operating experience. These events were based on a generic 3-loop Westinghouse plant, so plant-to-plant variations are not considered. The set of sequences considered above is not complete, but was meant to provide some short-term perspective for a screening criterion for Westinghouse plants in anticipation of more detailed long-term studies that are being pursued by NRC and the industry.

212

In developing the sequence frequencies, we have assumed that the reactor coolant pump seals are cooled following all the above events, the letdown line is isolated, the feedwater system has three isolation devices in series that are activated on the different signals, the allowable primary to secondary leakage and the operator will depressurize for small LOCAs (<2"). These assumptions should be confirmed as part of the longer term program. In addition, we have omitted some compound sequences because of insufficient information to perform an adequate evaluation in the short-term. This evaluation does not include low temperature overpressurization for which pressure relief devices are available.

REFERENCES

G.1 "Summary of Evaluations Related to Reactor Vessel Integrity Performed for the Westinghouse Owners Group." Westinghouse Electric Corporation, Nuclear Technology Division, May 1982.

214

- G.2 S. Bush, "Reliability of Piping in Lightwater Reactors," Nuclear Safety, Volume 17, No. 5, September-October 1976.
- G.3 O. Harris, et a¹., "Probability of Pipe Fracture in the Primary Coolant Loop of the PWR Plant," NUREG/CR-2189 Volume 5, August 1981.
- G.4 Technical Evaluation Report, Operating Reactor PORV and ECCS Reports (F-37) TMI Action Plan Requirements (draft), Franklin Research Center, October 29, 1982.
- G.5 Summary of Evaluations Related to Reactor Vessel Integrity, Letter fromO. Kinsley (WOG) to H. Denton (NRC), dated May 28, 1982.
- G.6 Handbook of Human Reliability Analysis with Emphasis on Nuclear Power Plant Applications, NUREG/CR-1278, October 1980.
- G.7 "An Assessment of the Probability of Overcooling Transients at Nuclear Plants with Combustion Engineering Nuclear Steam Supply System," CEN-212, June 1981.
- G.8 "An Approximate Prediction of Heat Transfer During Pressurized Thermal Shock," SLI-3213, June 1982.
- G.9 "An Approximate Prediction of Heat Transfer During Pressurized Thermal Shock With No Loop Flow But With Metal Heat Transfer," by S. Levy and J. M. Healzer, Reports SLI-8220, Revision 1, August 1982.

 G.10 "PWR Downcomer Fluid Temperature Transients Due To High Pressure Injection at Stagnate Loop Flow," T. G. Theofanous and H. P. Nourbakhsh, September 1982, present at Joint NRC/ANS Meeting on Basic Thermal Hydraulic Mechanism in LWR Analysis, September 14-15, 1982 at Bethesda, MD 20014.

A17

G.11 "Workshop on Pressurized Thermal Shock Fluid and Thermal Mixing," EPRI, October 7-8, 1982, Hanover, NH. Υ.



•

FIGURE G-1: HUMAN ERROR SPLIT FRACTIONS FOR AUXILIARY FEEDWATER TERMINATION

.

• .



Plant Status	Hot Ful	Hot Full Power					Power	<u>, , , , , , , , , , , , , , , , , , , </u>	<u></u>	
Operator Isolates Auxiliary Feedwater min.	5	10	20	30	60	5	10	20	30	60
Event Frequency per Reactor-Year	8×10-5	6×10- ⁵		1.5×10-5	3.x10-7	8.5x10- ⁶	6.8x10	_6	1.6x10- ⁸	3.×10-*
β∎in-1	0.4	0.2	3 .1	0.09	0.09	0.4	0.2	0.2	0.2	0.2
final Reactor Coolant System Temperature at Vessel Wall, ^o f	450	390	350	300	250	212	212	212	210	190
Final Pressure (psig)	2250	2250	2250	2250	2250	2250	2250	2250	2250	2250

Table G.1 NRC staff event parameters for the main steam line break (MSLB)

•

•

· \$

9-25

Plant Status	Но	t Zero Pow	er		
Operator Isolates Suxilirry Feed- water min.	5	10	20	30	60
Event Frequency per Reactor-Year	2×10-8	7×10-8		7x10-10	4. x10-11
β min-1	0.4	0.2	0.2	0.2	0.2
Final Reactor Coolant System Temperature at Vessel Wall, °F	220	220	212	210	200

Table G.2	Westinghouse Owners Group (WOG) event parameters for the main steam line break (MSLB)
	TOT the main sceam time break (MSLB)

.

•

w

.

Table G.3 NRC staff event parameters for the small steam line break (SSLB)

Plant Statu	5	Hot Full Power Hot Zero Power									
Operator Is Auxiliary water, Hi	olates Feed- N.	5	10	20	30	60	5	10	20	30	60
Event Frequ per React	ency or-Year	4.5×10-3	3.6x10-3	8×10-4	6.3x10-5	1.8×10-5	4.5x10-4	3.6x10-4	8×10- ⁵	6.3x10- ⁶	1.8x10-6
β min=1	0.4	0.2	0.1	0.06	0.06	0.4	0.2	0.1	0.06	0.06	
Final React System Te at Vessel	or Coolant mperature Wall, ^o f	385	320	250	220	200	375	310	235	200	175

.

5-27

23.

٠.

1

Plant Status	•	Hot Full Power . Hot Zero Power									
Operator Isc Auxiliary water, Mir	olates Feed-	5	10	20	30	60	5	10	20	30	60
Event Freque per Reacto	ency or-Year	2. x10-3	9.×10-4	9x10-5		4×10-6	5×10-5	2×10-5	2x10- ⁶	2×10-6	1×10-7
β min-1	0.4	0.2	0.1	0.06	0.06	0.4	0.2	0.1	0.06	0.05	
Final Reacto System Ten at Vessel	pr Coolant perature Wall, °F	320	290	260	250	220	230	230	230	220	200

Table G.4 Westinghouse Owner's Group (WOG) event parameters for the small steam line break (SSLB)

لا الصالات المعينة للالتحديد الأرار ال

.

.

- -

۹.

....

.

	SBLOCA	Extended HPSI Opera- tion
Event Frequency Per Reactor-Year	1.5×10-3	10-4
β, min-1	0.05	0.05
Final Reactor Coolant System Temperature at Vessel Wall, °F	125	125
Final Pressure, psig	1000	2250

Table G.5 NRC staff event parameters for the small-break loss-of-coolant accident (SBLOCA), and extended HPSI operation with a stagnant loop

252

-

.*

19.5

6**F**

•

Table G.6 N	NRC staff	event	parameters	for	steam	generator	tube	ruptur
-------------	-----------	-------	------------	-----	-------	-----------	------	--------

With delayed safety injection termination by the operator delay in terminating safety injection,	20	20	60
Event Frequency per Reactor-Year	5x10-3	(total 1 delays)	for all
β, min-1	0.1	0.07	0.04
Final Reactor Coolant System Temperature At Vessel Wall, °F	290	260	200
Final Pressure, psig	1000	1000	1000
With Steam Line Break or Stuck-Open SRV			
	Stuck o SRV Out Contain	pen side ment	SLB Inside Containment
Event Frequency Per Reactor-Year	2x10-4		1x10-5
β min1	0.04		0.04
Final Reactor Coolant System Temperature at Vessel Wall, °F	170		170

.

APPENDIX H

134

44

REACTOR PRESSURE VESSEL FAILURE PROBABILITY STUDY

Reactor pressure vessels (RPV) in nuclear power plants have traditionally been considered extremely reliable structural components. Indeed, studies completed in the United States and Europe have concluded that the disruptive failure rate (loss of the pressure retaining boundary) for nuclear pressure vessels is less than 10-6 at a 99% confidence level for RPVs designed, fabricated, inspected, and operated in accordance with the Boiler and Pressure Vessel Code of the American Society of Mechanical Engineers. However, recent results from surveillance and research programs and operating experience suggest that the issue of RPV failure probability should be reassessed. The renewed interest in RPV failure probability is due to the observation that thermal hydraulic transients occurring in commercially operating nuclear power plants are subjecting RPVs to unanticipated loadings which could contribute significantly to the failure probability of RPVs. In addition, operating experience and research programs over the past few years have provided additional information that more clearly defines both material property variations in RPVs and the effect of neutron irradiation on the material's resistance to fracture. The objective of this study is to assess the contribution to RPV failure probability of recently observed thermal hyraulic transients using recent material property data. (Note: The material properties formula and the model for ΔRT_{NDT} attenuation in the vessel wall differ from those specified in Appendix E. Future work will use consistent models.)

Generally, RPV reliability studies have used either one of two methods to calculate the probability of RPV failure. These methods are (1) the analysis of statistical data from observed non-nuclear pressure vessel failures to infer failure rates for both nuclear and non-nuclear pressure vessels and (2) the use of mathematical models that predict failure rates by analytically generating pressure vessel failures. Mathematical models used in the later technique have been primarily closed form analyses. In this effort, Monte Carlo simulation techniques have been used because of the ability to consider a greater number

of significant random variables and to perform a ...ide range of sensitivity studies. The results of extensive sensitivity studies which have been conducted are extremely important because they quantify the affect of uncertainties in the input parameters, thereby providing an estimate of the .ccuracy of the calculated failure probabilities, and they identify the significant variables and variable interactions. The results are best applied in a relative sense for use in decision making, and extreme caution must be exercised in applying the results in an absolute sense.

Section H.1 of this report describes the reactor pressure vessel considered in this study, Section H.2 describes the fracture mechanics techniques and simulation model used to calculate RPV failure probabilities; Section H.3 presents results of a reference case and sensitivity analysis performed using the simulation code; and Section H.4 presents a discussion and conclusions of the study.

H.1 Reactor Pressure Vessel Description

The reactor vessel geometry in this study has a 9-inch wall thickness and a 90-inch mean radius. Figure H-1 presents a schematic of how the RPV is fabricated. The failure probability is calculated for one vertical weld in the two beltline shell courses, which have lengths of approximately 72 inches. These dimensions are typical of most operating PWR vessels. Only the welds are considered because they have the greatest propensity for flaws, are most sensitive to radiation damage, and hence, should dominate the failure probability. The reactor vessel is fabricated of carbon steel with stainless steel cladding on the internal surfaces that are in contact with the primary coolant.

H.2 Probabilistic Model

H.2.1 Fracture Mechanics Algorithms

Pressurized thermal shock transients can subject the reactor pressure vessel to an unusual combination of high thermal and pressure stresses that create the potential for fracture of the reactor pressure vessel. Given well defined pressure and temperature-time histories for a pressurized thermal shock transient,

235

57

*

heat transfer and stress analyses can be conducted using either closed form or numerical analysis techniques.

1.56

In this study closed form solutions have been utilized for the heat transfer and stress analyses. The closed form solutions allow the primary coolant temperature time history to be expressed as either a fourth order polynomial or an exponential function of the form:

$$T = T_f + (T_o - T_f)e^{-\beta t}$$
(H-1)

where T is the temperature of the primary coolant as a function of time; T_o and T_f are the initial and final primary coolant temperatures, respectively; β is the decay constant that determines the rate of cooldown; and t is time. The pressure time history is represented by a fourth order polynomial. The heat transfer analysis is performed using an effective heat transfer coefficient which takes into account the fluid film heat transfer coefficient and the thermal resistance of the stainless steel cladding. However, the stresses due to the difference in thermal expansion between the stainless steel cladding and the base metal have not yet been included in the probabilistic code. A sensitivity study in Section H.3.2.9 provides an indication of how these stresses might affect the calculated failure probabilities.

The temperature and stress intensity values calculated using the above techniques were found to be in excellent agreement with the temperatures and stress intensity values calculated by the OCA-I code developed at ORNL.

Once the transient temperature and stress states have been calculated for the pressurized thermal shock event, linear-elastic fracture mechanics analysis is used to evaluate RPV integrity. Linear-elastic fracture mechanics (LEFM) is used to determine if a pre-existing flaw will propagate unstably through a material under certain loading and material conditions. The LEFM criteria for unstable fracture is:

64

(H-2)

where K_I is the applied stress intensity factor and K_{IC} is the critical stress intensity factor. Warm prestressing which can effectively inhibit crack extension even when K_I exceeds K_{IC} (see Section D.3) was not considered in the analyses with the exception of the sensitivity studies presented in Section H.3.5. Although for many of the transients analyzed, warm prestressing would be effective, these transients were only assumed for convenience in conducting parametric studies. System considerations and operator actions do not ensure that warm prestressing will be effective in every case.

The applied stress intensity factor, K_{I} , is a function of the stress state; crack depth, a; and flaw and component geometry. The stress state at any time in a pressurized thermal shock transient is defined by the pressure and temperature-time histories. The component geometry of interest in this study is the RPV beltline with an assumed longitudinally oriented flaw. The assumed longitudinal orientation is that expected in longitudinally oriented welds and is the flaw orientation that experiences the maximum stress and K_{I} in the reactor vessel beltline. Deterministic analyses assume that a flaw of a specific depth exists with certainty. In the probabilistic model developed in this study, the crack depth is treated as a random variable.

The critical stress intensity factor, K_{Ic} , is the material's resistance to unstable fracture. K_{Ic} is a function of the temperature at the crack tip; the material's initial nil-ductility reference temperature, RT_{NDTo} ; and the shift in RT_{NDT} , ΔRT_{NDT} . The temperature at any depth in the vessel wall is defined by the heat transfer analysis of the pressurized thermal shock transient.

 RT_{NDTo} is a material property determined by a destructive material testing procedure and is a measure of the temperature at which the material begins a trinsition from a "brittle" to ductile fracture mode. Determination of RT_{NDTo} is subject to material variability and measurement errors. Furthermore, estima es of the RT_{NDTo} for a specific plant often must be made from a generic data ball e not totally representative of the specific material of interest. Therefore RT_{NDTo} is treated as a random variable in the probabilistic model.

· , · , · (

The shift in RT_{NDT} is a result of neutron irradiation. As the vessel beltline fluence increases, the RT_{NDT} of the material becomes higher. This means that in order to exhibit the same resistance to fracture, K_{Ic} , the material must be at a higher temperature. The attenuation of fluence through the RPV wall for the results presented in this study was represented by the following relation

$$F(a) = F_{ID} e^{-.33a}$$
 (H-3)

where a is the depth in inches into the vessel wall and F_{ID} is the fluence (> 1 MEV) in neutrons/cm² at the surface of the RPV wall. More recent studies based on the concept of displacement per atom, dpa, consider a wider spectrum of neutron energies and suggest that the exponential decay constant should be smaller to more accurately predict radiation damage through the RPV wall. Fluence on the inside surface of the RPV wall varies with location in the RPV beltline due to the core design and power profile. In addition, there are relatively large uncertainties in calculating fluences. Thus, fluence has been considered a random variable in this study.

In the probabilistic analyses, the mean shift in RT_{NDT} has been represented by the following function:

$$\Delta RT_{NDT} = [-4.83 + 476 \cdot Cu + 267 \cdot Cu \cdot Ni] [F/10^{19}]^{0.218}$$
(H-4)

. . . .

where ΔRT_{NDT} is the mean shift in RT_{NDT} , Cu is the copper content in weight percent, Ni is the nickel content in weight percent, and F is the fluence in neutrons (> 1 MEV)/cm². This equation was developed at HEDL through regression analysis of surveillance and research program results. Copper and nickel contents vary throughout the RPV material, and uncertainties exist with the values specified for plant specific welds. Hence copper and nickel contents should be treated as random variables. Copper content was treated as a random variable in this study. However, the effect of nickel has just recently been recognized; and hence, nickel was not considered as a random variable in the original development of the code. Future versions of the code will include nickel as a random variable. The results presented here were generated assuming a constant nickel content of 0.65%. The equation used in Appendix E and in Section 5 of the main report for calculating the mean shift in RT_{NDT} is based on a more recent regression

251

٩à

analysis performed by HEDL. In order to maintain consistency between sensitivity studies, the earlier form of the equation (equ. H-4) was used throughout the probabilistic analyses. This equation predicts a lower shift in RT_{NDT} than the equation used in Section 5.

~ 15

The surveillance and research program data on ΔRT_{NDT} as a function of fluence exhibit significant variability as illustrated in Figure H-2. However, it is believed that much of the variability is due to variability and uncertainty in the measured fluences and copper contents in the data base. Therefore, it seems inappropriate to consider this variability twice and for the results presented in this study the mean trendline for ΔRT_{NDT} versus fluence specified in equation (H-4) was used. A proposed sensitivity study to be conducted in the future is to compare the results of this study with results generated by using mean copper contents and fluences and treating ΔRT_{NDT} as a random variable. However, for this study it was desirable to be able to conduct sensitivity studies on copper content and fluence; hence, these parameters were treated as random variables.

Once the initial RT_{NDT} and shift in RT_{NDT} have been specified either deterministically or probabilistically, the critical stress intensity factor, K_{Ic} , can be calculated. Figure H-3 shows a plot of K_{Ic} data versus $T-RT_{NDT}$, where T is the temperature of the material and RT_{NDT} is the sum of the initial RT_{NDT} and the shift in RT_{NDT} . Because K_{Ic} is a material property, it exhibits some variability and is treated as a random variable. A mean curve for K_{Ic} versus $T-RT_{NDT}$ was developed through regression analysis. The equation for this mean curve is:

$$\overline{K}_{1c} = 36.2 + 49.4exp(0.0104(T-RT_{NDT}))$$
 for T-RT_{NDT} $\leq -50^{\circ}F$ (H.5a)

 $\tilde{K}_{IC} = 55.1 + 28.0 exp(0.0214(T-RT_{NDT}))$ for T-RT_{NDT} > -50°F (H-5b)

If crack initiation is predicted, the crack may arrest as it runs deeper into the wall encountering hotter, less irradiated, and hence, tougher material. Arrest of the crack is predicted if

$$K_{I} < K_{Ia}$$
 (H-6)

where K_{Ia} is the stress intensity factor for crack arrest. Figure 5-4 shows the data for K_{Ia} versus T-RT_{NDT} and a mean curve fit using regression analysis. The equation for the mean curve is:

254

 $K_{Ia} = 19.9 + 43.9 \exp(.00993(T-RT_{NDT}))$ for $T-RT_{NDT} < 50^{\circ}F$ (H-7a)

Both the mean crack initiation and crack interest toughness were truncated at an upper shelf value of 200 KSI \sqrt{in} . Thus if crack arrest is not predicted before K_I reaches a value of 200 KSI \sqrt{in} . vessel failure is predicted.

H.2.2 Simulation Model

4

Figure H-5 illustrates the simulation model developed for RPV failure probability. The left hand column in the figure is the deterministic analysis which includes the heat transfer, thermal and pressure stress, and applied stress intensity value calculations for a range of crack depths at ten time steps in the transient. Matrices of temperature and K_{I} values are stored for use later in the simulation analysis.

The variables designated "simulate" in the diagram are treated as random variables, and their values are sampled from a statistical distribution defined by input parameters. As discussed in the previous section, crack depth, a; fluence; RT_{NDTo} ; copper content; K_{Ic} ; and K_{Ia} were treated as random variables in this study. A value for each of these random variables is sampled from the appropriate statistical distribution. Once the flaw size is simulated, the corresponding K_{I} value is retrieved from the K_{I} matrix developed earlier in the code. The mean K_{Ic} value is calculated according to the equation (H-5) using the temperature corresponding to the time step and simulated crack depth and an RT_{NDT} based on the values of copper content, fluence, and RT_{NDTo} sampled from their corresponding statistical distributions. Since the K_{Ic} data exhibits significant variability, the K_{Ic} value is simulated by sampling from a distribution about the mean K_{Ic} value.

If crack initiation is predicted, the crack is allowed to advance through the RPV wall in discrete steps of 0.25 inches, and a check for crack arrest is made at each crack advance. K_{Ia} is treated in a similar fashion to K_{Ic} as mentioned above. If crack arrest is predicted, the code continues to analyze successive time steps in the transient using the arrested crack depth. Since the applied K values and material temperature at the crack tip are a function of time in the transient, reinitiation of the crack may occur.

Each pass through the simulation loop depicted in Figure H-5 represents a single computer experiment conducted to determine if RPV failure will occur. Up to a million passes through this loop can be made. The code keeps track of the number of crack initiations and RPV failures and the probabilities of crack initiation and RPV failure are estimated by dividing these values by the total number of trials. Thus the code actually performs millions of deterministic calculations with each set of calculations based on a different set of values selected from the appropriate statistical distributions for the significant variables. This is equivalent to subjecting a population of up to a million operating reactor pressure vessels to the pressurized thermal shock transient of interest and then inferring the failure probability based on the number of observed failures.

H.2.3 Statistical Distributions of Random Variables

The simulation model described above suffers from the same problem as all analytic models, its output is only as good as its input. Unfortunately, very little information exists in the literature regarding the required statistical inputs, and the time frame of this initial study was not sufficient to allow the necessary research and analysis to develop rigorous statistical inputs. Therefore, mony of the statistical distributions associated with the random variables in the model are based on expert opinion and have somewhat ill-defined "levels of non-fidence." It is appropriate to interject at this point that, because of the uncertainties associated with the input parameters, the best use of the results of this study is in a relative sense to assist in the decision-making proce s.

The number and size of cracks in the weld material of the RPV is probably the random variable with the greatest uncertainty. Several crack size distributions exist in the literature. These distributions are based on the experience

H-8

١.

192.

٠.

of RPV fabricators and nondestructive examinations. The flaw distribution is of course difficult to quantify since the flaws of interest are not the flaws that have been detected, but those of unknown size and number that remain in the RPV because they were not detected. Figure H-6 shows the probability of having a flaw of depth a in a reactor pressure vessel longitudinal beltline weld as estimated in the OCTAVIA computer code. The weld volume associated with the OCTAVIA flaw distribution was defined as the volume of longitudinal weld material in the beltline region of a PWR. To obtain the flaw distribution, for a single beltline weld, as considered in this study, the OCTAVIA flaw distribution was adjusted assuming that the flaws were equally distributed among six longitudinal beltline welds. For illustration, the crack depth, a, in Figure H-6 is represented as a continuous random variable. However, in this study, the crack depth was used as a discrete random variable. For the curve in Figure H-6, approximately nine distinct crack depths ranging from 0.125 to 3.5 in. were used and the probabilities indicated at these crack depths were reduced by a factor of 1/6 to represent the probability of a flaw in one weld and were used to construct a stepwise cumulative probability distribution. The Monte Carlo simulation in the computer code used the stepwise cumulative distributions to generate a crack depth for each simulation cycle.

113

٩,

The distribution of RT_{NDTo} is dependent on the variability in the material and measurement error. In discussions with the metallurgists at materials testing laboratories, they indicated that they believed their accuracy in determining RT_{NDTo} was $\pm 20^{\circ}$ F. No data exist from which to infer the shape of the distribution. Therefore, for a reference case, a normal distribution with a standard deviation of 15°F was assumed. Sensitivity studies were conducted assuming that the standard deviation was 30°F.

The variance in fluence is due to the power distribution in the reactor core and inaccuracies in calculation. Experts at Hanford Engineering and Development Laboratory in Richland, Washington, have estimated the uncertainty in fluence estimates to be on the order of $\pm 30\%$ (1 σ) using common practice techniques. For the reference case, a normal distribution with a standard deviation of 30% was assumed. Sensitivity studies were conducted assuming standard deviations of 50% and 15%.

A study was conducted to evaluate the sensitivity of the calculated failure probabilities to the tails associated with the normal distributions assumed for RT_{NDTo} and fluence. In this study the distributions were truncated at the mean plus and minus three standard deviations. The results indicated no appreciable difference, and it was concluded that the tails of the assumed normal distributions do not dominate in the calculations.

-74

Copper was introduced into the welds of the RPV from welding rods that were copper coated to improve the welding process. Chemical composition analyses of welds from RPV prolongations have recently provided extensive data for welds representative of those in operating plants. Rigorous statistical analysis of these data is not yet complete. However, the distribution does appear to be symmetrical with a standard deviation in the range of .02% to 0.5%. For the reference case, a normal distribution with a standard deviation of 0.025% was assumed. In the sensitivity studies a 0.07% standard deviation was considered. In all analyses the range of simulated copper content values was limited to 0.08% to 0.40% copper.

As described in Section H.2.1, K_{IC} and K_{Ia} and also treated as random variables with a normal distribution and a 10% standard deviation about their respective means curves. Due to lack of sufficient data, the distribution of K_{IC} and K_{Ia} about their mean is difficult to rigorously determine. However, several papers have suggested using a normal distribution about the mean with a standard deviation of 10%, and this distribution was assumed in generating the results presented here. The normal distribution about the mean was applied to both the transition and upper shelf toughness regions. Sensitivity studies were conducted to evaluate the sensitivity of the calculated failure probabilities to the assumed variability in K_{IC} and K_{Ia} .

H.3 <u>Results</u>

This section presents results of a reference case and of certain sensitivit studies performed using the simulation model described in Section H.2. As stated earlier, due to uncertainties in the input data, it is suggested that the results be considered in a relative rather than an absolute sense. The sensitivity studies performed identify important parameters and their interaction

and suggest how sensitive the reference case failure probabilities are to uncertainties in the input data. The results presented are conditional probabilities; that is, the probability of failure of a RPV weld given that the pressurized thermal shock transient under consideration occurs. To convert the results presented here into failure rates, the frequency of occurrence of the transient considered must be defined. Since the results presented are for an individual weld in the RPV beltline, the total conditional failure probability of the RPV beltline welds is the appropriate summation of the failure probabilities for each weld. If these values are sufficiently low and independence is assumed, the failure probabilities for the six welds can simply be summed. If the failure probabilities become high, the intersection of the weld failure probabilities must be subtracted.

115

H.3.1 Reference Case

٩.

2

The reference case analysis is defined as follows:

- The Rancho Seco transient (Figure H-7)
- The OCTAVIA flaw distribution,
- Copper ~ $N(\mu, 0.025\%)$,
- $RT_{NDTo} \sim N(\mu, 15^{\circ}F)$,
- FLUENCE ~ N(μ , 30%),
- \cdot ΔRT_{NDT} HEDL mean curve, and
- K_{1c} and K_{1a} treated as random variables.

Figures H-8 through H-12 present the conditional failure probabilities calculated for the reference case condition. Each figure presents the failure probability versus the mean fluence for a specified mean copper content and three mean values of RT_{NDTo} . Also, plotted across the top of each figure, is the ΔRT_{NDT} calculated using the mean HEDL curve. These shifts are based on the

mean copper content and fluence value in each figure. These curves make it possible to estimate the failure probability for the beltline region of a PWR for which the mean values of the random variables can be estimated.

Several important observations can be made regarding Figures H-8 through H-12. The first observation is that no failure probabilities less than 10^{-5} are calculated for any combination of mean fluence copper content, or RT_{NDTO} . This result occurs because the Rancho Seco transient will result in an applied K_I value greater than the assumed mean upper shelf toughness of 200 KSI \sqrt{in} . for flaws of 3.0 inches or greater depth, and the probability of such flaws existing is nearly 10^{-5} in the flaw distribution assumed. Therefore, the lower limit on calculated failure probabilities would change for different transients, flaw distributions, or assumptions about the upper shelf toughness.

The second important observation is that any specified value of failure probability corresponds within a few degrees to a specific mean value of RT_{NDT} , independent of the copper content and fluence by which the RT_{NDT} value was achieved. For example in Figure H-8, based on a copper content of 0.34%, a failure probability of 2 x 10⁻⁵ corresponds to a mean RT_{NDT} value of approximately 255°F to 260°F for the three values of RT_{NDTo} . Similarly, in Figure H-11, based on a mean copper content of 0.28%, a failure probability of 2 x 10⁻⁵ corresponds to a mean RT_NDT of approximately 255°F for the two values of RT_{NDTo} . These results demonstrate that RT_{NDT} is in fact an excellent criterion for evaluating reactor pressure vessel integrity under specified thermal shock conditions. The mean RT_{NDT} value corresponding to a specific failure probability will, of course, be different for different pressurized thermal shock transients.

H.3.2 Reference Case Sensitivity Studies

Sensitivity studies were conducted on the distribution for copper content, initial RT_{NDT}, fluence, and fracture toughness. In addition, conditional failure pr - babilities were calculated assuming that specific flaw sizes exist with a probability of 1.0. Finally, a sensitivity study was conducted for a set of hype hetimical transients with assumed expontial temperature decays and constant pressures. These cases are intended to provide insight into how sensitive RPV failure

calculations are to thermal hydraulic parameters such as temperature, pressure, rate of cooldown, and heat transfer coefficient.

H.3.2.1 Copper Content

Figure H-13 illustrates the results of the sensitivity study on copper content. When the standard deviation for the copper distribution was increased from 0.025% to 0.07%, the calculated failure probabilities increased by approximately a factor of 5.

H.3.2.2 Initial RT_{NDT}

Figure H-14 illustrates the results of the sensitivity study on RT_{NDTo}. When the standard deviation for the RT_{NDTo} distribution was increased from 15°F to 25°F, the calculated failure probabilities were increased by a factor of approximately 3.

H.3.2.2 Fluence

Figure H-15 illustrates the results of the sensitivity study on fluence. The standard deviation for the fluence distribution was increased from 30% to 50% and decreased to 15%. The increased standard deviation resulted in approximately a factor of three increase in calculated failure probabilities, while the decrease in the standard deviation had little effect on the calculated failure probabilities.

H.3.2.4 Fracture Toughness

Figure H-16 illustrates the results of the sensitivity study on fracture toughness. Three different representations of the fracture toughness distribution were considered. In the first two cases the normal distribution about the mean fracture toughness values for K_{IC} and K_{Ia} was maintained but the standard deviation was increased to 15% and then 20% of the mean value. In the third case K_{IC} and K_{Ia} were treated deterministically using the lower bound fracture toughness curves from Section XI of the American Society of Mechanical Boiler and Pressure Vessel Code (see Figures H-3 and h. C).

1471

copper content of 0.34% and a mean initial RT_{NDT} of 0°F. Assuming the larger standard deviations resulted in less than a factor of three difference from the reference case failure probabilities for a mean RT_{NDT} of 236°F or less. At higher values of RT_{NDT} the calculated failure probabilities for the assumed standard deviation of 15% and 20% were a factor of 50 and over an order of magnitude greater than the reference case, respectively. When the lower bound fracture toughness curves from Section XI of the Code were used, the calculated failure probabilities were one order of magnitude to almost two orders of magnitude higher than the reference case.

715

Results of intermediate scale tests conducted at Oak Ridge National Laboratory suggest that long cracks in large reactor vessels may exhibit "lower bound" fracture toughness. Several points should be made regarding this hypothesis. First, the "lower bound" performance was relative to fracture toughness data generated from small specimens not all of sufficient size to qualify as valid in accordance with ASTM-E-399 criteria. Second, cracks that exhibited "lower bound" performance in the ORNL tests were long flaws (~38 inches), and shorter more realistic flaws are expected to exhibit toughness more closely represented by the toughness distribution assumed in the reference case. Finally, the intermediate scale tests performed have exhibited statistical variability in fracture toughness, but none of them have demonstrated fracture toughness as low as the ASME Code Section XI toughness curves.

The results of this sensitivity study show that the failure probabilities are sensitive to the distribution in fracture toughness, especially for mean values of RT_{NDT} greater than approximately 240°F. Thus, an effort should be made to better define this distribution. Experience to date suggests that fracture toughness may be a function of crack length as well as other parameters, and that in analyses assuming a bivariate flaw distribution of depth and length, it may also be appropriate to consider a relation between crack length and fracture toughness.

H.3.2.5 Simultaneous Increase in the Variability of All Random Variables

Figure H-17 presents the failure probabilities calculated when all the random variables were assumed to show the increased variances used in sensitivity studies,

including one case where K_{Ic} and K_{Ia} were treated as random variables and one case where they were modelled using the lower bound curves. For the first case the calculated failure probabilities were approximately an order of magnitude greater than the reference case, while for the second case (lower bound K_{Ic} and K_{Ia}) the calculated failure probabilities were almost three orders of magnitude higher.

49

H.3.2.6 Flaw Distribution

84

Figure H-18 presents the conditional failure probabilities calculated assuming that flaw sizes ranging from 0.125 inches to 2.0 inches exist with a probability of 1.0 and for several different mean fluence values and values of RT_{NDT}.

The curves presented in Figure H-18 are useful because they can be used to calculate failure probabilities for different crack depth distributions. In Table H-1 the conditional failure probability is calculated for a reactor pressure vessel with mean copper content of 0.34% and mean initial RT_{NDT} of $0^{\circ}F$, assuming a flaw distribution less severe than the OCTAVIA distribution assumed in the reference case. The estimated failure probability for the less severe flaw distribution is 4.7 x 10^{-5} compared to 7.5 x 10^{-5} for the OCTAVIA distribution. The relatively small difference in the estimated failure probabilities results because the flaw distributions considered are not significantly different in the range of flaw depths that contribute most to the failure probability. An advantage of this approach to evaluating sensitivity to the assumed flaw distribution is that it allows easy identification of the range of flaw depths that contribute most significantly to the failure probability.

H.3.2.7 Shift in RT_{NDT}

A sensitivity study was conducted using the fluence versus ΔRT_{NDT} relation from Regulatory Guide 1.99, "Effects of Residual Elements in Predicted Damage to Reactor Vessel Materials." Use of the upper bound trendlines presented in Regulatory Guide 1.99 is not considered appropriate in a probabilistic analysis but was considered in this sensitivity study in an effort to quantify the effect of differences in assumed trendlines. Figure H-19 presents the results generated assuming ΔRT_{NDT} as predicted by the HEDL trendlines and the Regulatory Guide

1.99 trendlines. Assuming the more severe Regultory Guide 1.99 trendlines increased the calculated failure probabilities by a maximum of nearly two orders of magnitude.

56 C

H.3.2.8 Upper Shelf

As discussed in Section H.2.1, the results presented in this study are based on linear elastic fracture mechanics analysis. In the transients of incerest, however, linear elastic fracture mechanics may not be valid when cracks are predicted to run deep into the vessel wall where the material is operating in the upper shelf temperature regime. In the upper shelf temperature regime, crack extension generally occurs in a ductile mode referred to as tearing rather than in a cleavage mode as predicted by linear elastic fracture mechanics. In the reference case analysis, the mean fracture toughness curves were truncated at an upper value of 200 KSI \sqrt{in} , and it was assumed that if crack arrest did not occur before the applied K reached 200 KSI \sqrt{in} ., the crack would tear through the wall. In reality this problem requires an elastic-plastic or tearing instability type of analysis which has not yet been fully developed and validated for pressurized thermal shock conditions. A study was conducted to evaluate the sensitivity of the calculated failure probabilities to the assumed upper shelf value. In this study the mean upper shelf value was increased to 300 KSIJin., 400 KSIJin. and infinity and a check was incorporated for plastic instability of the remaining section. The assumed higher upper shelf toughness values all resulted in the same calculated failure probabilities, as illustrated in Figure H-20. The calculated failure probabilities with the increased upper shelf values are more than an order of magnitude less than the reference case failure probabilities for mean values of RT_{NDT} less than approximately 240°^{-,} At a mean RT_{NDT} value of 250°F the failure probability associated with the ncreased upper shelf toughnesses is approximately a factor of four less than the reference case; and at a mean RT_{NDT} value of 275°F or greater the calculate failure probabilities are the same. Thus upper shelf mate. I behavior may decrease the probability of catastrophic vessel failure for mean RT_{NDT} values of 250°F or less but provides very little additional margin at higher values of RT_{NDT}. Two notes of caution are in order. First, recent information sugn sta that the gradient in fluence attenuation may not be as steep as assumed in these analyses, and a different model assuming greater radiation damage deeper

in the vessel wall may bring the reference case and increased upper shelf toughness failure probabilities closer together at a lower value of mean RT_{NDT}. Second, the calculated probabilities of crack initiation, which are significant from an economic point of view, are unaffected by the assumption regarding upper shelf toughness.

201

H.3.2.9 Cladding

1

For surface cracks as assumed in this evaluation, the stainless steel cladding will increase the applied stress intensity value due to differential thermal expansion between the clad and base metal. This effect has not yet been included in the fracture mechanics code used in the probabilistic analysis, although it has been evaluated deterministically. A study was conducted to estimate the magnitude of the effect of the increased K_T due to cladding on the calculated failure probabilities. In this study the thermal component of the applied stress intensity factor, K_{It}, was increased by 10% and 20%. This is a gross approximation since the actual increase in ${\rm K}_{\rm I}$ will be a function of crack depth and time in the transient. However, calculations indicate that for the Rancho Seco transient the maximum contribution to the thermal component of the applied K_{T} is less than 10%. Therefore, the case of a 10% increase in K_{It} should be bounding for the Rancho Seco transient as analyzed deterministically. The case of a 20% increase in K_{1+} gives some insight into sensitivity of the assumptions regarding initial stress in the cladding at normal operating temperature. The results of the study are presented in Figure H-21. For an increase in K_{T+} of 10% there is essentially no change in the calculated failure probabilities for mean surface RT_{NDT} values less than approximately 250°F. Above a mean RT_{NDT} of 250°F the failure probabilities increase by less than a factor of three. Figure H-22 illustrates the factor of increase in conditional failure probability assuming a 10% increase in the thermal component of the applied stress intensity factor due to the affect of cladding. For a 20% increase in K_{It} the calculated failure probabilities increase by a maximum factor of approximately 4.

It should be noted that the differential thermal effect between the cladding and base metal may be more significant for more severe thermal shocks, and caution must be exercised in extending the results of this study to those transients.

H.3.3 <u>Transient Sensitivity Studies</u>

In addition to the reference Rancho Seco transient, postulated MSLB and turbine trip with stuck-open bypass valve transients were evaluated using the probabilistic code. The same transients were analyzed deterministically by ORNL in Reference H.1 and were selected for probabilistic analysis to provide some estimate of the conservatisms in the deterministic calculations. Also, a set of hypothetical pressurized thermal shock transients with assumed exponential temperature decays and constant pressure levels was analyzed to determine the sensitivity of failure probability to the minimum temperature reached in the transient, rate of temperature drop, pressure level, and heat transfer coefficient.

3.00

H.3.3.1 <u>Main Steamline Break and Turbine Trip With Stuck Open By-pass</u> <u>Valve Transients</u>

Figures H-23 and H-24 present the pressure and temperature time histories associated with the postulated MSLB and stuck-open bypass valve transients, respectively. The solid lines in the figures represent the pressure and temperature time histories calculated by Brookhaven National Laboratory using the IRT Code. Reference H.1 provides details of the assumptions made in performing the thermal hydraulic calculations. The solid lines in each figure represent the pressure and temperature time histories calculated by the IRT analysis. The dashed lines represent the fourth order polynomial fits to the IRT pressure and temperature time histories used for performing closed form heat transfer and stress analyses. The applied stress intensity values resulting from these polynomial fits agree well with those calculated by ORNL using the OCA-1 numerical heat transfer analysis. Figures H-25 and H-26 present the calculated failure probabilitie for the MSLB and stuck-open bypass valve, respectively, for a longitudinal beltline weld with a mean initial RT_{NDT} of 0°F and mean copper contents of 0.22% and 0.34%. The failure probabilities are very high for both of these severe thermal transients.

H.3.3.2 <u>Hypothesized Transients with Exponential Cooldowns and Constant</u> <u>Pressures</u>

320

Table H-2 presents the failure probabilities for a set of hypothesized pressurized thermal shock transients. The temperature time history in each transient is assumed to follow an exponential decay defined by

$$T(t) = T_{f} + (550 - T_{f})e^{-\beta t}$$

where T is the temperature in °F, t is time in minutes, T_f is the final temperature of the transient in °F, and β is the decay constant in min-¹. Three values of T_f , 150°F, 225°F, and 300°F; three values of β , 0.05 min-¹, 0.15 min-¹, and 0.50 min-¹; and five constant pressure levels, 0 psig, 500 psig, 1000 psig, 1500 psig, and 2000 psig were considered for a total of 45 different transients. Each of these transients was then evaluated for five levels of fluence, 0.5 10^{19} neut/cm², 1.0 x 10^{19} neut/cm², 2.0 x 10^{19} neut/cm², 3.0 x 10^{19} neut/cm², and 4.0 x 10^{19} neut/cm² assuming a mean copper content of 0.30% and a mean initial RT_{NDT} of 20F. The data presented in Table H-1 have been used to evaluate the sensitivity of failure probability to the normalizing factor $T_f - RT_{NDT}$, β , and pressure.

H.3.3.2.1 T_f-RT_{NDT} Sensitivity Study

Figure H-27 presents failure probability versus $T_f - RT_{NDT}$ for the three different values of β considered and a constant pressure of 1000 psig. An ideal normalizing factor would combine the significant transient parameters in such a way that one curve of failure probability versus the normalizing factor could be used to estimate the probability of failure for any arbitrarily defined transient. Several factors combining T_f , β , pressure, total temperature drop, and RT_{NDT} were considered but no combination of these factors yielded a perfect normalizing factor. However, for the range of transients considered here, $T_f - RT_{NDT}$ is a fairly effective normalizing factor for any specific β and constant pressure level. Figure H-28 indicates that failure probability is highly sensitive to the value of $T_f - RT_{NDT}$. For example, considering a β of 0.15 min⁻¹, a
decrease in $T_f - RT_{NDT}$ from -20°F to -70°F results in a factor of approximately 150 increase in failure probability.

34.41

H.3.3.2.2 Cooldown Rate Sensitivity Study

Figure H-27 indicates a much greater increase in failure probabilities when β is increased from 0.05 to 0.15 than when β is increased from 0.15 to 0.50. This observation is more clearly illustrated in Figure H-28 where failure probability is plotted as a function of β for several values of $T_f - RT_{NDT}$ and 1000 psig constant pressure. The curves illustrate that failure probability is very sensitive to β in the range below 0.15 min⁻¹ while increasing β beyond 0.15 min⁻¹ increases the failure probability by less than a factor of five. This is most likely a result of the assumed thermal inertia of the system, and the sensitivity curves will change if different thermal characteristics are assumed in the heat transfer analysis.

H.3.3.2.3 Pressure Sensitivity Study

Figure H-29 is a plot of failure probability versus pressure for several values of the parameter $T_f - RT_{NDT}$. The failure illustrates increasing sensitivity to pressures as the parameter $T_f - RT_{NDT}$ increases. For example, for a $T_f - RT_{NDT}$ value of -25°F an increase in pressure from 500 psig to 2000 psig results in approximately a factor of 200 increase in failure probability while a similar pressure increase for a $T_f - RT_{NDT}$ value of -120°F increases the failure probability by only a factor of 5. Thus pressure is a more important parameter in the transients where the minimum temperature is near the value of RT_{NDT} rather than well below it. It should be noted that for a pressure level of 0.0 psig, the failure probability is zero. Thermal Shock Experiment 6 recently completed at ORNL demonstrated that although severe cracking may occur under the condition of no pressure, thermal stresses alone are not sufficient to drive a crack through the RPV wall.

H.3.3.2.4 Heat Transfer Coefficient Sensitivity Study

Figure H-30 presents the results of a sensitivity study conducted on heat transfer coefficient. The two curves in the figure present RPV failure probability versus

H-20

heat transfer coefficient, h in BTU/hr/ft² °F, for two different hypothetical exponential cooldowns. One has a final transient temperature of 150°F while the other has a final transient temperature of 200°F. A constant pressure level of 1000 psig was assumed and the RPV material was assumed to have an adjusted RT_{NDT} of 250°F. When the thermal conductivity of the cladding is considered, the range of the effective heat transfer coefficient for the thermal hydraulic transients under consideration is between 200 BTU/hr/ft² °F and 400 BTU/hr/ft °F. The results indicate that over that range, the assumed heat transfer coefficient can make as much as an order of magnitude difference in the calculated RPV failure probabilities. The results presented in this study were generated assuming an effective heat transfer coefficient of approximately 300 BTU/hr/ft². The assumed thermal diffusivity in this study was 0.98 in²/min and a constant value of 0.332 was used for the parameter $(\frac{E\alpha}{1-\nu})$. Where E is Young's Modulus,

303

 α is the coefficient of thermal expansion, and ν is Poisson's ratio.

H.3.4 Inservice Inspection Sensitivity Study

ł

Sensitivity studies were conducted using Figure H-18 to evaluate the effect of various levels of non-destructive examination (NDE) reliability on reactor pressure vessel failure probability. Three different functions of flaw non-detection probability were considered. The first function for probability of flaw nondetection was taken from Reference H.2. This function was based on a survey of NDE experts. The other two flaw nondetection probability functions assumed probabilities of non-detection of 0.5 and 0.05, respectively, over the entire range of crack depths. The latter two functions were selected primarily for the purpose of evaluating the sensitivity of failure probability to a wide range of NDE reliabilities. However, they were also intended to correspond to condition of rough surface finish and smooth surface finish, respectively. It was assumed for all functions of NDE reliability that cracks of greater than 2.0 inches in depth would be detected with certainty. The results of these evaluations are preserced in Tables H-3A through H-3C. The first column in these tables gives the flaw depth, a, in inches; the second column is the probability of existence of a crack of depth a as estimated by the OCTAVIA flaw distribution; column three is the probability of non-detection; column four is the probability

of existence of a crack of depth a after performing an NDE (the product of columns two and three); column five is the conditional probability of failure given the Rancho Seco transient and existence of a crack of depth &; and column six is the contribution to the conditional failure probability of the reactor vessel weld for each crack depth (the product of columns four and five). The conditional failure probability of the reactor vessel weld given that the Rancho Seco transient occurs is given by the sum of the probabilities in column six.

The conditional failure probabilities of a reactor pressure vessel weld following inservice inspection can be compared to the conditional failure probability of 7.5x10⁻⁵ before the inservice inspection, from Figure H-8. This comparison indicates inservice inspections conducted with reliabilities corresponding to the Reference H.2 report probability of non-detection function or the constant 0.5 probability of non-detection problem will do very little to improve reactor pressure vessel reliability under pressurized thermal shock conditions. However, if a probability of non-detection of 0.05 can be achieved, even for small flaws, then a substantial decrease in failure probability, approximately a factor of 20, will result.

H.3.5 Warm Prestressing Sensitivity Study

A study was conducted to determine the effects of warm prestressing on the calculated conditional failure probabilities for the idealized Rancho Seco transient that was considered as the reference transient in Section H.3.1. The warm prestress phenomenon was modelled by simply not allowing crack initiation at any time step in the transient for which the applied K value for the simulated crack depth was greater at the previous time step. No allowance was made fcr a possible increase in the allowable K_I to K_{IC} ratio above 1.0 resulting from warm prestressing.

For the Rancho Seco transient warm prestressing was very effective in inhibiting crack extension. The conditional failure probabilities calculated assuming warm prestressing were less than 10⁻⁵ for mean RT_{NDT} values less than 290°F. (See Table H-3.)

H-22

1.1

<u>P(a)</u>	P(Non-Detuction)	P(a Inspection)	P(Failure)	<u>P(Failure)</u>	
0.83	. 69	. 57	0	0	
0.16	. 49	. 78	5x10- ⁵	3.9x10- ⁶	
4.2×10^{-3}	. 24	1.0x10- ³	1.0×10^{-2}	1.0x10- ⁵	
4.1x10-4	.061	2.5x10- ⁵	5.4×10^{-2}	1.4x10- ⁶	
1.3×10^{-4}	. 018	2.3x10- ⁶	5.6×10^{-2}	1.3x10-7	
4.2×10-5	8.1×10^{-3}	3.4×10-7 CONDITIONAL FAILU	4.5x10-2 RE PROBABILITY	<u>1.5x15-8</u> 1.5x10-5	
	<u>P(a)</u> 0.83 0.16 4.2×10- ³ 4.1×10- ⁴ 1.3×10- ⁴ 4.2×10- ⁵	$P(a)$ $P(Non-Detection)$ 0.83.690.16.494.2x10-3.244.1x10-4.0611.3x10-4.0184.2x10-5 $8.1x10^{-3}$	P(a) $P(Non-Detection)$ $P(a Inspection)$ 0.83.69.570.16.49.784.2x10-3.241.0x10-34.1x10-4.0612.5x10-51.3x10-4.0182.3x10-64.2x10-58.1x10-33.4x10-7CONDITIONAL FAILUE	$P(a)$ $P(Non-Detraction)$ $P(a \text{ Inspection})$ $P(Failure)$ 0.83.69.5700.16.49.78 $5x10^{-5}$ 4.2x10^3.241.0x10^31.0x10^24.1x10^4.0612.5x10^{-5}5.4x10^21.3x10^4.0182.3x10^65.6x10^24.2x10^{-5}8.1x10^{-3}3.4x10^{-7}4.5x10^2CONDITIONAL FAILURE PROBABILITY	

TABLE H-3A: A Marsh#71 Report Probability of Nondetection

*,

ł

TABLE H-3B Constant 0.50 Probability of Non-Detection

<u>a P(a)</u>		P(Non-Detection)	P(a 1 Inspection)	P(Failure)	<u>P(Failure)</u>	
0.125	0.83	0.50	0.42	0	0	
0.25	0.16	0.50	0.08	1.5×10-4	1.2x10- ⁵	
0.50	4.2×10^{-3}	0.50	2.1x10- ³	1.0×10^{-2}	2.1x10- ⁵	
1.00	4.1x10-4	0.50	2.1×10-4	5.4x10-2	1.1x10-5	
1.50	1.3x10-4	0.50	6.5x10- ⁵	5.6x10- ²	3.6x10-6	
2.00	4.2x10-5	0.50	2.1×10-2	4.5x10-2	9.5x10-7	
			CONDITIONAL FAILURE	PROBABILITY	4.9×10-5	

TABLE H-3C: Constant 0.05 Probability of Non-Detection

a	<u>P(a)</u>	P(Non-Detection)	P(a 1 Inspection)	<u>P(Failure)</u>	<u>P(Failure</u>)
0.125	0.83	0.05	4.2×10^{-2}	0	0
0.25	0.16	0.05	8.0x10-3	1.5x1-4	1.2x10-6
0.50	4.2×10^{-3}	0.05	2. 1×10^{-4}	1.0×10^{-2}	2.1x10-6
1.00	4.1x10-4	0.05	2.1 \times 10 ⁻⁶	5.4×10^{-2}	1.1x10-7
1.50	1.3x10-4	0.05	6.5x10- ⁶	5.6x10-2	3.6x10-7
2.00	4.2x10-5	0.05	2.1x10-6	4.5x10-10-2	9.5×10^{-8}
			CUNDITIONAL FAILURE	PRUDADILIT	2.0X10.

Based on the above studies it can be concluded that for transients whose thermal hydraulic characteristics ensure warm prestressing ditions, the probability of RPV failure can be significantly reduced.

H.3.6 Flaw Orientation Sensitivity Study

Results presented thus far have concentrated on the longitudinally oriented beltline welds. The volume and orientation of weld material in the reactor vessel beltline region depends on whether the beltline shell was fabricated from rolled plates or forged rings as illustrated in Figure H-1. Several operating vessels are fabricated from ring forgings or have limiting values of RT_{NDT} associated with circumferential welds.

The orientation of the beltline welds is significant in the evaluation of pressurized thermal shock transients because flaws oriented in a circumferential direction have a lower propensity for extension than those oriented parallel to the longitudal axis of the vessel. The circumferentially oriented crack has a lower propensity for crack extension because it is subject to a pressure stress only half as great as the longitudinal flaw and because the applied stress intensity factor is lower due to the increased bending stiffness of the cylinder about its azimuthal axis. In addition, these two factors also create a greater propensity for crack arrest in a circumferentially oriented flaw. Because flaws in the veld material are generally assumed to be oriented in the direction of the weld, reactor vessels fabricated from forged rings with circumferential welds are expected to have a greater tolerance for pressurized thermal shock loadings than reactor vessels fabricated from rolled plates with longtudinal welds.

A study was conducted to evaluate the relative differences in integrity between longitudinally and circumferentially oriented welds. Both determinstic and probablistic calculations were performed for two different transients. The transients were the idealized Rancho Seco Transient illustrated in Figure H-7 and the MSLB accident illustrated in Figure H-24. Two dimensional (infinitely long longitudinal and 360° circumferential) flaws were evaluated using linear elastic fracture mechanics analysis.

H-24

The results of the deterministic calculations indicate that the Rancho Secontransient will not cause catastrophic failure of the reactor pressure vessel for a surface RT_{NDT} less than 350°F (calculated by R.G. 1.99). For a surface RT_{NDT} of 350°F or lower, the deterministic calculations predict crack arrest less than halfway through the vessel wall in the linear elastic regime. For a surface RT_{NDT} of 370°F, crack arrest is predicted approximately three-fourths of the way through the vessel wall. Although some margin still exists for circumferentially oriented flaws, this depth of crack extension is approaching the condition where the vessel would fail due to plastic instability of the remaining ligament. Furthermore, this amount of crack extension leaves little margin for tearing of the crack which could occur in low upper shelf materials.

シート

Tig probabilistic analysis of the Rancho Seco transient generally supports the conclusions from the deterministic calculations. The failure probabilities calculated for the Rancho Seco transient assuming that a 1.0-inch flaw existed with certainty were less than 10^{-5} for a mean surface RT_{NDT} values of 275°F or less and approximately 3.2×10^{-5} for mean surface RT_{NDT} of 290°F. Comparable failure probabilities for longitudinally oriented flaws were 7.5 x 10^{-4} , 10^{-2} , and 4.5×10^{-2} for mean surface RT_{NDT} values of 250°F, 275°F, and 290°F, respectively. Thus, for the Rancho Seco transient, the failure probability of a circumferentially oriented flaw is at least three orders of magnitude less than that of a longitudinally oriented flaw for mean surface RT_{NDT} values of 290°F or less. A comparison of the crack initiation probabilities for longitudinal and circumferential flaws indicated that the probability of initiation of a circumferential flaw ranges by approximately a factor of 1000 to 25 less than that of a longitudinal flaw for a corresponding range in mean surface RT_{NDT} values of 215°F to 290°F.

Deterministic calculations for the MSLB indicate that vessel failure due to extension of circumferential cracks will occur at RT_{NDT} surface values of 226°F (calculated by Regulatory Guide 1.99) or greater. Since 226°F was the lowest RT_{NDT} evaluated, vessel failure might be predicted at even lower values of RT_{NDT} . The probabilistic analysis of the MSLB indicated that the probability of failure of a circumferentially oriented flaw can be as little as a factor of 12 to 3 less than that for a longitudinal flaw for a corresponding range in mean surface values of RT_{NDT} between 250°F and 290°F. Figure H-32 presents the factor decrease

H-25

in failure probability for circumferentially versus longitudinally oriented flaws of 1.0-inch and 0.5-inch depths. For a mean surface RT_{NDT} value of 215°F, no failures were generated in the simulation analysis. The probability of crack initiation for the postulated MSLB accident was essentially equal over a range in mean surface RT_{NDT} values of 215°F to 290°F for the flaw sizes considered.

In summary, both deterministic and probabilistic evaluations indicate that for transients as severe as those which have been observed (the Rancho Seco transient being considered the most severe) circumforential flaws will not lead to catastrophic vessel failure for relatively high values of RT_{NDT} . Furthermore, the probability of initiation of circumferentially oriented flaws is significantly less than that of longitudinal flaws until relatively high values of RT_{NDT} are reached. However, for much more severe postulated transients, deterministic analyses predict that catastrophic vessel failure can result from circumferentially oriented flaws at relatively low values of RT_{NDT} . In addition, probabilistic analyses indicate a relatively small difference in failure probabilities between circumferential and longitudinal flaws and essentially no difference in the probability of crack initiation for more severe transients.

H.4 Application of Probabilistic Analyses in Establishing Regulatory Criteria

Probabilistic analysis is a very powerful technique for gaining insight and understanding of complex technical issues and when used correctly can result in effective regulation without excessive conservatism. However, misapplication of the results of probabilistic analyses which may occur due to inadequate understanding of the bases upon which they were developed could compromise safety and economic objectives. In this context, the purpose of this section is to identify some of the limitations of the work performed and to define how the results presented previously can be most approximately used in developing a regulatory position on the pressurized thermal shock issue.

H.4.1 Limitations of Probabilistic Fracture Mechanics Analyses

As indicated in Section H.2.3, the statistical distributions used to generate the results presented in Section H.3 are based largely on expert opinion and are subjective in nature. Efforts are currently in progress to assemble improved

data bases and develop more rigorous statistical distributions. However, results generated using improved input data will not be available to assist in developing a short-term position on the pressurized thermal shock issue. Uncertainty in the statistical distributions used in the model is one of the main reasons for conducting the sensitivity studies presented in Section H.3.2. The results of these sensitivity studies, in which the variability and form of the statistical distributions were varied, indicate that uncertainties in the statistical distributions for copper content, initial RT_{NDT} , and fluence could contribute as much as an order of magnitude uncertainty to the results presented in Section H.3.

11

Flaw depth is the random variable with the greatest uncertainty. The sensitivity studies on flaw depth distribution and inservice inspection indicate that the calculated failure probabilities for the Rancho Seco transient are relatively insensitive to changes in the distribution for crack depths greater than approximately one inch. This is because relatively small flaws can dominate the failure probability due to the nature of the stresses and toughness gradient associated with pressurized thermal shock events. The sensitivity studies also indicate that the calculated failure probabilities could change substantially given a significant change in the distribution of crack depths. When the probabilities of all crack depths are altered by a constant factor, the calculated failure probabilities change by approximately the same factor. Thus, the uncertainty in the calculated failure probabilities is directly related to the uncertainty in the same crack distribution. Unfortunately, little data exist from domestic operating reactor vessels that would allow a rigorous determination of the flaw depth distribution, particularly in the range of crack depths less than one inch. The distribution of crack depths has large uncertainty associated with it and could easily contribute plus or minus an order of magnitude or more uncertainty to the calculated failure probabilities.

The sensitivity studies conducted on fracture toughness indicate that the calculated failure probabilities are very sensitive to the assumed variability in the fracture toughness data. At high values of RT_{NDT}, relatively small increase in the variability of the fracture toughness can increase the calculated failure probabilities by well over an order of magnitude.

H-27

The above discussion suggests that the calculated failure probabilities could be underestimated due to uncertainties in copper content, initial RT_{NDT}, and fluence; the calculated failure probabilities could be over or underestimated due to uncertainty in the crack depth distribution; and the calculated failure probabilities could be underestimated due to uncertainties in the fracture toughness distribution. In addition to these uncertainties, there exist uncertainties due to elements not considered in the probabilistic model. Specifically, the toughness of the stainless steel cladding which may be great enough to inhibit the initiation of small flaws and warm prestressing which may inhibit crack extension were not considered. If, in fact, the vessel cladding does maintain high toughness in the range of fluence levels of interest, the extension of finite cracks could be inhibited and the failure probabilities may be greatly overestimated. Similarly, warm prestressing which will be effective for a large class of pressurized thermal shock events would greatly reduce the calculated failure probabilities for such events.

3,2--

12.4

Work is continuing to better quantify the confidence levels that can be associated with the calculated failure probabilities. However, based on the currently available data and analysis, it appears that plus or minus two orders of magnitude is a reasonable estimate of the uncertainty associated with the calculated failure probabilities.

H.4.2 Application of the Probabilistic Fracture Mechanics Results

The discussion of the previous section suggests that the results which have been presented are most appropriately used in a relative sense for identifying significant variables and variable interactions. Because of the uncertainties associated with the calculated failure probabilities, use of the results in an absolute sense to establish an RT_{NDT} screening limit would be inappropriate. Nonetheless, there does exist a tendency to view the results in an absolute sense when evaluating proposed regulatory requirements. Furthermore, there is a desire to view the results in an absolute sense when performing a probabilistic risk assessment. Utilization of the results in these manners is useful in evaluating a regulatory position, but the limitations of the analysis as discussed in the previous section must be kept in mind.

H-28

٩.

In conclusion, it is suggested that the regulatory criteria should be based on deterministic fracture mechanics analyses and that the probabilistic analyses not be used as the basis for developing such criteria until such time as greater confidence in the probabilistic analyses can be attained. It is suggested, however, that the probabilistic analyses be used, with caution, to check deterministically derived criteria relative to desired margins of safety.

REFERENCES

.

- H.1 Kryter, R. C. et al. 1981. <u>Evaluation of Pressurized Thermal Shock</u>. NUREG/CR-2083, ORNL/TM-8072, Oak Ridge National Laboratory, Oak Ridge, Tennessee.
- H.2 Study Group Report, "An Assessment of the Integrity of PWR Pressure Vessels," UKAEA Report, October 1976.

a(inches)	P(a)	P(F1a)	$P(F) = P(F \rightarrow)P(a)$		
0.125	-	0	0		
0.25	4×10^{-2}	1.5 x 10-4	4 x 10- ⁶		
0.50	3×10^{-3}	1.0×10^{-2}	3 x 10- ⁵		
1.00	2 × 10-4	5.4×10^{-2}	1 x 10- ⁵		
1.50	1×10^{-5}	5.6 x 10- ²	6 x 10- ⁷		
2.00	1×10^{-6}	4.5×10^{-2}	5 × 10- ⁸		

Table H-1 Flaw distribution sensitivity tuc.

ちゅう

34

۲.

H-31

Table H-2 $Cu_{\mu} = 0.30$ $Cu_{\sigma} = 0.025$ $F_{\sigma} = 30\%$ Ni = .75 IRT_{NCTµ} = 20°F IRT_{NDTσ} = 15°F OCTAVIA FLAW DISTRIBUTION

HEDL MEAN ΔRT_{NDT} T = T_f + (T_i - T_f)e^{-St}

		T₄	150°F			225°F		300°F		
Pressure	F 5 a (a) b (a) b (b)	β'.05	. 15	. 50	. 05	. 15	. 50	. 05	. 15	. 50
(ps1)	rivence (neut/cm ²)									
0	0.5 x 10 ¹⁹	0			Q	0	0	0	0	Q
	1.0×10^{19}	0			0	0	0	0	Ō	ŏ
	2.0 x 10 ¹⁹	2 x 10-®			0	Ó	4 x 10-6	Õ	õ	Õ
	3.0×10^{19}	5.8 x 10-6			Ď	Ō	1×10^{-5}	õ	ถ้	ñ
	4.0×10^{19}	2.1×10^{-4}	3.1×10^{-2}	9.1 x 10-2	Ō	-		õ	ŏ	õ
500	0.5 x 10 ¹⁹	0		8.3×10^{-2}	G	0	0	0	0	0
	1.0×10^{19}	2.2 × 10-5	2.6 x 10-3	1.1×10^{-2}	ñ	ñ	ñ	ñ	0	ŏ
	2.0 x 10 ¹⁹	2 9 x 10-4	2×10^{-2}	6 3 x 10-2	Ô	1 - 10-5	A 9 V 10-5	ŏ	0	0
	3.0 x 1019	9 3 x 10-4	5 x 10-2	$1 3 \times 10^{-1}$	0	Q v 10-5	4.0 X 10	0	0 0	0
	4.0 x 10 ¹⁹	1.7×10^{-3}	8 3 9 10-1	21×10^{-1}	0	A 3 v 10-4	2 2 2 10-3	0	0	0
		1.7 ~ 10	0.J × 10	2.1 X 10	Ŭ	4.3 X 10	2.3 X 10	U	U	U
1000	0.5×10^{19}	2.4×10^{-5}	1.2×10^{-3}	4.1×10^{-3}	0	2×10^{-6}	4×10^{-6}	0.	0	0
	1.0×10^{19}	2.6 x 10-4	1×10^{-2}	3 x 16-2	0	2 x 10-6	8 x 10-6	0	0	0.
	2.0×10^{19}	1.7×10^{-3}	4.9×10^{-1}	1.2×10^{-1}	4×10^{-6}	1.3×10^{-4}	5 x 10-4	Õ	õ.	Ō
	3.0×10^{19}	5.1×10^{-3}	9.8×10^{-2}	2.2×10^{-1}	3.2 x 10-4	8.1 x 10-4	3.4×10^{-3}	0	Ō	Ō
	4.0×10^{19}	1.1×10^{-2}	1.5×10^{-1}		7 x 10-5	2.6×10^{-3}	9.6×10^{-3}	Ō	0	2 x 10-6
1500	0 5 x 1019	1 Q v 10-4	3 8 - 10-3	1 - 10-2	A v 10-6	1 - 10-5	1 A v 10-5	0	2 - 10-6	4 - 10-6
	1.0 2 1019	1.3×10 1.1×10 -3	25 - 10 - 2	5 5 4 10-2	4 X 10	2 9 - 10-5	1.4 × 10 6 6 × 10-5	0	2 X 10-*	4 X 10-5
	2 0 x 1019	5 2 y 10-2	Q Y + 10-2	3.0×10^{-1}	20 - 10-5	2.0 X 10-*	0.0 X 10-3	0	2 x 10-6	4 X 10-5
	3 0 2 1019	$1.7 - 10^{-2}$	1.1.1.10-1	1.3 × 10	2.0 A 10	0.1 X 10-3	0 1 10-2	U O	2 X 10-5	4 X 10=*
	$A = 0 \times 10^{19}$	1.7×10^{-2}	1.3×10^{-1}		1.0 X 10-*	3 X 10	9.1 X 10	U	4 X 10-5	2 0 1 X 8
	4.0 X 10-	3.4 X 10	2.4 X 10		4.4 X 10	8.2 X 10-5	2.3 X 10-4	U	1.2 × 16-5	3.2 × 10-3
2000	0.5 x 10 ¹⁹	5.5 x 10-4	1.0×10^{-2}	2.1×10^{-2}		2×10^{-5}	3.6 x 10- ⁵	2 x 10-6		1×10^{-5}
	1.0×10^{19}	2.7×10^{-7}	4.9×10^{-2}	9.4 x 10-2	1.8 x 10-5	9 x 10-5	2.7 × 10-4	2 x 10-6	1×10^{-3}	1.4 x 10-5
	2.0×10^{19}	1.6×10^{-2}	1.5 x 10-1	2.8×10^{-1}	1.7 x 10-4	2.1 x 10-3	5.1 x 10-3	2 x 10-6	1 x 10-5	1.4 x 10-5
	3.0×10^{19}	4.0×10^{-2}	2.6 x 10-1		6.7 x 10-4	8 7 x 10-3	2 x 10-3	2 x 10-6	2 x 10-3	3 x 10-5
	4 0 x 1019	6.6 × 10-2	2 6 v 10-1		1 6 - 10-3	2 1 - 10-2	A A U 10-2	1 - 10-5	c a u 10-3	1 2 10-4

ົ້



•

FIGURE H-1 PWR Beltline Shell Fabrication Configurations



FIGURE H-2

¥



*

FIGURE H-3 K_{Ic} Reference Toughness Curve with Supporting Data:

*SEE A.B. LIDIARD, "A SIMPLIFIED ANALYSIS OF PRESSURE VESSEL RELIABILITY," BR NUCL ENERGY Soc. JULY 1977.



FIGURE H-4 K_{Ia} Reference Toughness Curve With Supporting Data'

\$



FIGURE H-5

34

^e mini pi ser





35.

Ì



FIGURE H-7: RANCHO SECO TRANSIENT IDEALIZED PRESSURE AND TEMPERATURE TIME HISTORIES

~

\$



FIGURE H-8: CONDITIONAL FAILURE PROBABILITY FOR THE RANCHO SECO TRANSIENT MEAN Cu = 0.34%

.

•

đ



FIGURE H-9: CONDITIONAL FAILURE PROBABILITY FOR THE RANCHO SECO TRANSIENT MEAN Cu = 0.32%

· .



FIGURE H-10: CONDITIONAL FAILURE PROBABILITY FOR THE RANCHO SECO TRANSIENT MEAN Cu = 0.30%

۰.

47

٩

.



٩\$

2.0

FLUENCE, n/cm² x 10¹⁹

3.0

4.0

FIGURE H-11: CONDITIONAL FAILURE PROBABILITY FOR THE RANCHO SECO TRANSIENT MEA!! Cu = 0.28%

1.0

۰.

10⁵ L



FIGURE H-12: CONDITIONAL FAILURE PROBABILITY FOR THE RANCHO SECO TRANSIENT MEAN Cu = 0.26%

• .

\$

•*



FLUENCE, n/cm² x 10¹⁹



۰.





ŧ٢



FIGURE H-15: FLUENCE SENSITIVITY STUDY MEAN Cu = 0.34% MEAN RT_{NDT0} = 0°F

.

ART_{NDT} IN *F

1

*



FIGURE H-16: FRACTURE TOUGHNESS DISTRIBUTION SENSITIVITY STUDY MEAN Cu = 0.34% MEAN RT_{NDT0} = 0°F

.

ART_{NDT} IN *F



ARTNDT IN *F

ľ

FIGURE H-17: SIMULTANEOUS INCREASE IN THE VARIABILITY OF THE RANDOM VARIABLES MEAN Cu = 0.34% MEAN RT_{NDT₀} = 0°F

*

•

.



FIGURE H-18: CONDITIONAL FAILURE PROBABILITY P(F/a) FOR THE RANCHO SECO TRANSIENT MEAN Cu = 0.34% MEAN RTNDTO



..



* • `

FIGURE H-20: UPPER SHELF TOUGHNESS SENSITIVITY STUDY MEAN Cu = 0.34% MEAN RT_{NDT0} = 0°F





ARTNDT IN *F

è

**



FIGURE H-22: FACTOR OF INCREASE IN CONDITIONAL FAILURE PROBABILITY DUE TO CLADDING FOR THE RANCHO SECO TRANSIENT

"BASED ON 10% INCREASE IN KIt AND RANCHO SECO TRANSIENT

.



FIGURE H-23 IRT MAIN STEAM LINE BREAK PRESSURE AND TEMPERATURE TIME HISTORIES





3.11


FIGURE H-25: CONDITIONAL FAILURE PROBABILITY FOR THE IRT MSLB MEAN RT_{NDT0} = 0 °F



FIGURE H-28: CONDITIONAL FAILURE PROBABILITY FOR TURBINE TRIP/STUCK OPEN BYPASS VALVE MEAN RT_{NDT0} = 0 °F

.

.

1



۶

۲.

FIGURE H-27: SENSITIVITY OF CONDITIONAL FAILURE PROBABILITY TO T, RT NDT



Conditional Failure Probability for a single Lognitudinal Beltline Weld 24 1

\$





FIGURE H-30: HEAT THANSFER SENSITIVITY STUDY

4

-4



FIGURE H-31: CRACK ORIENTATION SENSITIVITY STUDY

 \mathbf{h}_{i}

.

2

\$-

APPENDIX I

· 4 m

FLUENCE RATE REDUCTION TO PWR PRESSURE VESSELS

I.1 Introduction

₽.

The NRC staff, as part of its evaluation of the Pressurized Thermal Shock (PTS) problem for PWR pressure vessels (PV), has undertaken a survey of domestic and foreign PTS experience and an evaluation of various fast neutron fluence rate reduction concepts (Ref. I.1). The survey included all three PWR vendors and the eight most affected PWR plants,* that is, those with significant vessel fluence. The staff found general agreement among those surveyed as to the techniques available for fast fluence rate reduction. The reason for this agreement is the generic similarities of the PWR plants of different manufacture and limited number of options which are considered viable.

The staff evaluation includes concepts for: (1) fluence rate reduction (by factors of 2 to 3) employing low leakage fuel loading, and (2) reductions (by factors of 10 or more) using select fuel assembly replacement on the core periphery with nonfueled assemblies containing stainless steel rods. The impact of implementing any of these schemes is so plant dependent that it was not possible to do more than estimate the impact on the total peaking factor as part of this study.

The low leakage fuel loading schemes are also characterized as an "in-out" fuel loading scheme in contrast to the usual "out-in" loading scheme. The "in-out" ("out-in") refers to fuel assembly movement during refueling from the core interior (periphery) to the core periphery (interior). In a low leakage fuel loading scheme, therefore, twice or thrice burned fuel assemblies (or even poisoned fuel assemblies) are placed on the core periphery. In our evaluation

^{*}Fort Calhoun, San Onofre, Oconee-1, Maine Yankee, Calvert Cliffs-1, H. B. Robinson-2, Turkey Point-4 and Three Mile Island-1.

We used stainless steel rods in the nonfueled assemblies. Other choices, however, could be made for the stainless steel rods. 4

4

This report also includes a survey of foreign reactor experience with respect to fluence rate reduction to the pressure vessel.

I.2 Survey of Licensees, Owners' Groups, and Vendors

The staff visited Combustion Engineering (CE) and Westinghouse (\underline{W}). Lengthy discussions were held with cognizant personnel in reactor physics, thermalhydraulics, fuel management, and licensing. A visit could not be arranged with Babcock & Wilcox (B&W) so that information was obtained with a telephone conference call. The vendors were asked to discuss (1) the reduction of peak and longitudinal weld steam fluence accumulation rates by factors of 2, 3, 5, or 10, (2) the corresponding impact of fluence rate reduction schemes, and (3) the estimated cost of implementation of various schemes. The same questions that we asked the vendors were also asked the licensees of the eight most affected plants through the NRC project managers (Ref. I.2). These licensees had little information to offer and, generally, referred us to the respective vendors.

Limited cost estimate data was obtained from our survey. Low leakage fuel loading schemes (in-out) may result in overall cost savings to licensees because of the benefits of extended cycle operation which could accompany such schemes. However, extremes of low leakage loading schemes could cost from 1 to \$5 million. Replacement of fuel assemblies with stainless steel rodded assemblies on the core periphery could cost up to \$20 million per year due to derating plus a one-time engineering cost of \$15 to \$25 million.

I.3 Survey of Foreign Experience

Several foreign reactor plants with radiation-induced pressure vessel embrittlement have been modified or modifications are planned. Such modifications include raising the temperature of the high pressure injection water and reducing the fluence accumulation rate (i.e., lowering the fast flux to weld

1-2

seems or plate material of the vessel). In the following we will deal with modifications related to fluence rate reduction to the pressure vessel. The information gathered was the result of a questionnaire directed to several countries around the world in the summer of 1981 shortly after the PTS task force was formed by NRC (Ref. I.3).

I.3.1 Finland

Loviisa-1 The Soviet built, 420 MWe Finnish reactor was put into operation in 1977 (Ref. I.4). The loading consists of about 360 hexagonal fuel assemblies. The reactor had operated for about 3 years when it was determined that the radiation-induced weld seam embrittlement was higher than originally estimated. In 1980, with only 3 years of operation, the estimated nil-ductility transition temperature increase was 76°C. The originally predicted increase for 40 years of operation was 85°C.

It was decided to remove 36 fuel assemblies on the periphery of the core and replace them with hollow steel rods in the hexagonal shroud identical to that of the fuel assemblies. The assemblies that were removed represented 10% of the core inventory. However, there was no reduction in the power level because the plant had adequate thermal margin. Due to the hexagonal shape of the fuel assemblies the azimuthal flux distribution was fairly uniform varying from .73 to 1.00. The peak fast neutron flux decreased by a factor of about .7. The new flux peak appeared in the location of the previous minimum, reduced by a factor of 2.8 from .73 to about .25 (estimated nonpeak value between .22 to .30). This modification along with an increase in the temperature of the emergency injection water and changes to the emergency operating procedures is expected to provide adequate protection for the remaining life of the plant.

Loviisa-2 This is a sister plant to Loviisa-1 that was put into operation in 1980. The Finns could not decide from cost-effectiveness considerations whether a modification similar to that for Loviisa-1 should have been implemented during the first cycle in Loviisa-2. Nevertheless, the same modifications could be made in a later cycle.

1-3

35.

No information is available to us on surveillance programs, neutron transport calculations, uncertainties, or specific fluence values.

1.1

4

I.3.2 Germany (Obrigheim and Stade)

The PWR plants in the Federal Republic of Germany (FRG) have pressure vessels with only horizontal weld seams, hence, the azimuthal position of the peak fluence is immaterial* and the concern is in the irradiation of the base metal (Ref. I.8). An extensive surveillance program has been instituted in all FRG PWRs. Present estimates indicate that at the end of the 40 EFPY of operation there would be excessive irradiation of the pressure vessel of Stade and Obrigheim and that fuel assembly substitutions to lower the projected peak fluence would be needed. These reactors are very similar to Westinghouse plants, hence, we surmise that the azimuthal distribution has localized peaks. Becuase there is no discussion of potential consequences we assume that element substitution will be of a limited extent with no power derating. The stade reactor has been using a low leakage loading (Ref. I.6) for the last few cycles. The estimated end of pressure vessel life fluence for Obrigheim is somehwat higher than that estimated for Fort Calhoun and for Stade is considerably lower than most American pressure vessels (Ref. I.5). The Federal Ministry of Internal Affairs of Germany in its August 10, 1981 reply to the NRC questionnaire indicated that nonfueled assembly replaced was contemplated (Ref. I.7) for these two reactors.

I.3.3 France

Recent information received from the French (Central Service for the Safety of Nuclear Installations) (Ref. I.8) indicates that a program for the study of material embrittlement was instituted about 10 years ago. This program has only recently been expanded to include pressure vessel dosimetry. No definitive plans are known at this time for pressure vessel fluence rate reduction modifications.

*The PWR at Gundremmingen, currently under construction, has longitudinal welds.

11/13/82

I.3.4 Other Countires

Replies to the NRC questionnaire have been received also from Italy, Spain, Sweden, Korea, and Japan. However, none of the operating utilities have taken any steps to lower fluence rate to the pressure vessel. All show awareness of the problem. Surveillance programs have been established in Sweden and Japan.

252

I.4. Evaluation of Fast Fluence Rate Reduction Schemes

In order to assess independently a number of fluence rate reduction schemes, an evaluation was performed for the staff by its consultants at BNL (Ref. 1.9). From the eight most affected PWR reactors, three plants, Oconee-1, Fort Calhoun,

and Robinson-2 (one from each PWR vendor), were selected for the staff evaluation. These plants were selected because of the availability of plant-specific data and the relatively large vessel fluence. Table I-1 presents some pertinent information concerning these plants. Included in the table are the vendors' and our consultant's estimate of the present and end of vessel life fluence.

The approach taken by the staff in performing this evaluation was:

- (a) To use the transport theory code DOT 3.5 to calculate the fast fluence to the pressure vessel. The calculations were two-dimensional and used
 16 neutron energy groups. The BNL methods have been benchmarked to a number of tests and are comparable to those used by the vendors.
- (b) To use as-built dimensions, material compositions, and measured values of the neutron source to evaluate H. B. Robinson-2, Oconee-1, and Fort Calhoun.
- (c) To calculate for each of these plants the (1) current values of the peak fluence at the longitudinal welds, (2) projected value of the peak fluence to the end of 32 effective full power years (EFPY), (3) fluence attenuation through the pressure vessel, (4) fluence time spectra at various wall thicknesses, (5) pressure vessel fluence azimuthal distribution, and (6) end of vessel life fluence value for various fluence rate reduction schemes.

11/13/82

(d) To evaluate the impact of these modifications in terms of the potential increase in the total peaking factor.

5. 8

(e) To compare the staff's calculations to similar calculations from the licensee or vendor when possible.

Since the fluence to the pressure vessel is caused primarily by the fast neutrons in the peripheral fuel assemblies, schemes for reducing fluence accumulation rate to the pressure vessel fall into two main classes. The first class is designed to lower the neutron leakage from the periphery of the core by lowering the power level of the peripheral fuel assemblies. The second class is designed to lower the fluence rate to the pressure vessel by placing a thick metal shield between the core periphery and the pressure vessel. This second class of fluence rate reduction schemes will not, however, be discussed further because of the lack of space between the core and vessel to accommodate large thicknesses of metal.

The first class of fluence rate reduction schemes consider the lowering of the peripheral fuel assemblies' powers by (1) using low leakage fuel loadings, and (2) removal of fuel assemblies and replacement with assemblies containing stainless steel rods. Note that the use of nonfueled assemblies contains elements of both classes of fluence rate reduction schemes. The power of the reactor could also be lowered in order to reduce the peripheral assemblies' powers. This power derating was not considered in our evaluation. Instead, the assumptions in the staff evaluation are (1) the total power of the reactor is constant, (2) the shape of the power distribution remains the same from the periphery toward to center of the core, (3) the maximum linear heat generation rate is assumed constant, and (4) the core flow is assumed constant.

Since the PIS problem solution will be plant-specific, no attempt was made to optimize core fuel loading patterns on a cycle-by-cycle basis to lessen the impact on fuel cycle economics or to assess the impact on normal operation, transients, and accidents. Only a rough estimate was made of the impact in terms of a potential increase in the total peaking factor.

11/13/82

1-6

Some of the plant-specific factors include, among others things, the location of the weld seams in the pressure vessel, the cooper and nickel content of the weld seams, the core power and size, the peripheral fuel location, the presently accumulated fluence, the fuel management scheme presently employed, and the location of the peak fluence on the vessel. An example of one of these plantspecific items is the weld seam location for the three plants. Figure I-1 shows the three weld seams at Oconee-1 folded onto a quarter core. The Fort Calhoun weld seams are shown in Figure I-2 folded onto an eighth of the core. Figure I-3 shows the weld seams at H. B. Robinson-2 folded cnto an eighth of the core.

154

Calculations were performed by BNL for the three plants to evaluate the low leakage fuel loadings and peripheral element replacement with assemblies containing stainless steel rods. Similar results were obtained for all three plants. The conclusions of our analysis agreed with statements made by the parties we talked to in our sruvey. These BNL calculations will be reported in a forthcoming BNL-NUREG report (Ref. I.13).

Table I-2 shows some results from the BNL calculations for Oconee-1 for three cases in which the ratio of the peripheral assembly power to the core average power was varied. One should roughly assume the 0.910 power ratio to be representative of normal out-in fuel assembly loading, the 0.527 ratio to represent in-out low leakage fuel loading using partially burned or poisoned assemblies, and the zero power ratio to represent peripheral fuel assemblies for which the fission source was artificially zeroed (not achievable in practice). Table I-3 shows the same results in a different format giving the fluence for the remaining 28 EFPY in terms of the relative fluence rate to the peak longitudinal weld seam for the original out-in fuel loading. Two additional cases are also shown in Table I-3. Both of these cases are representative of fuel element removal and replacement with stainless steel assemblies. In one of the cases the stainless steel rods are spaced in the same way as fuel rods while in the other case the rods are more closely packed with additional stainless steel rods. Both Tables I-2 and I-3 clearly demonstrate the fluence rate reduction factors that are possible for the two fluence rate reduction schemes. Table I-3 further demonstrates the effectiveness of including stainless steel rods in the replacement assemblies.

11/13/82

I-7

PTS RPT APP I

Figure 1-4 shows in graphical form the fluence rate reduction factor for the remainins 28 EFPY (data from Table I-2) for the original peak vessel fluence location as a function of peripheral fuel assembly power. The results are linear as a consequence of our assumptions and modeling.

The fase neutron flux attenuation through the pressure vessel is shown in Figure I-5. The curve is nonlinear but shows more than a factor of 10 reduction in flux on the outside wall of the pressure vessel. This figure allows the estimation of fluence rate accumulation at various positions within the pressure vessel when the fluence rate is known on the inside wall of the vessel.

Figure I-6 provides a summary of the staff's evaluation for Oconee-1 showing results for a number of fluence rate reduction schemes as a function of effective full power years of operation. Shown in the figure are the licensee's FSAR value as well as the vendor's (B&W) estimate of the vessel fluence for the current in-out low leakage fuel loading scheme. Note that the staff's evaluation for low leakage fuel loadings closely agrees with the B&W results and both results are about half of the FSAR estimate. Three other evaluations for element removal and replacement with stainless steel assemblies are shown in the figure. Pattern 1 refers to the removal and replacement with stainless steel assemblies are shown in the fluence rate to the weld seams could be reduced to a minimum number of assembly substitutions. Pattern 3 was chosen to reduce the power peak at the core flats caused by Pattern 2. For Patterns 1, 2, and 3 fuel assembly removals and substitutions numbered 40, 20, and 32, respectively.

Figure I-7 provides a summary similar to that of Figure I-6 of the staff's evaluation for Fort Calhoun. Shown in the figure are the licensee's FSAR value as well as the vendor's (CE) estimate of the vessel fluence for the current fuel loading scheme. Note that the staff's evaluation for the current fuel loading scheme is in reasonable agreement with the CE results and both results are about a factor of 2 larger than the FSAR estimate. Staff results are also shown for two in-out low leakage schemes; in one the peripheral power is 0.41 of the core average power and in the other the peripheral power is zero. Three additional cases are shown in the figure for fuel assembly removal

I-8

4

and replacement with stainless steel assemblies. The three cases are for the removal and replacement of 40, 24, and 16 assemblies.

1. 5

Figure I-8 provides a summary similar to that of Figure I-6 of the staff's evaluation for H. B. Robinson-2. Shown in the figure is the vendor's (\underline{W}) estimate of the vessel fluence for the current fuel loading scheme. Note that the staff's evaluation for the current fuel loading scheme is a factor of about 1.14 larger than the vendor's evaluation. Staff results are shown for the out-in loading scheme for which the peripheral power to the core average power ratio is 0.89 as well as for two in-out loading schemes for which this ratio is 0.45 and zero.

Table I-1 summarizes the staff survey and evaluation of the peak vessel fluence for the three plants for various schemes and the associated decrease in the total peaking factor. The table also gives present and end-of-life estimates of vessel fluence by BNL, the vendors, and the FSAR value for the current fuel management scheme.

I.5 <u>Cost Estimate for Fuel Assembly Substitution to Lower Neutron Flux to</u> the Pressure Vessel

Cost estimates of fuel assembly substitution require consideration of plant-specific factors such as, azimuthal flux distribution, longitudinal weld seam location, thermal margin, operating history, and future fuel management. A proper and accurate cost estimate study of fuel substitution requires core redesign, fuel management analysis, and fuel management decisions by the licensee. The staff has not attempted plant-specific analyses, but has approached the problem on a generic basis, using the following assumptions:

(1) The peak fluence to the critical weld will be the criterion for flux reduction requirements. This is a reasonable assumption for the older reactors under consideration because the peak fluence location has received considerably more irradiation than other azimuthal locations. For the newer plants, methods exist which allow fuel assembly management

1-9

11/13/82

6

with low leakage schemes to even out the fluence accumulation in the belt region or minimize fluence to the welds. The latter considerations are beyond the scope of this study.

- (2) We assume that a core redesign could take advantage (in part or in total) of some relaxation of Appendix K, lowering of MDNBR, or power derating depending on the required flux reduction and the plant-specific parameters.
- (3) To estimate the effect of 10 CFR 50.46 and Appendix K, we assume that the linear heat generation rate could be increased by 20 percent if a best estimate evaluation model were used rather than the conservative evaluation models specified by Appendix K (Ref. 1.14). The 20 percent figure may be true for some plants (those limited by decay heat or stored energy) but is not necessarily true for all plants to the same extent. No effort has been made to distinguish whether specific plants are Appendix K limited or not.
- (4) To estimate the effect of the DNBR limits (for those plants which are DNBR rather than Appendix K limited) we assume that lowering the DNBR criterion by 10 percent would allow raising the average heat generation rate by about 20 percent (Ref. I.15). This assumption is based on a sensitivity analysis of MDNBR to eight DNB parameters, one of which is power. A standard Westinghouse, Combustion Engineering, and Babcock and Wilcox plants were used with the corresponding correlations. The suggested relationship is accurate to better than 1.5 percent. The study was carried out by Battelle, Pacific Northwest Laboratories.
- (5) We assume that the daily power replacement cost (hence, plant shutdown cost) is \$0.3 M. Daily power derating costs are derived from this by prorating. It is assumed that fuel cycle costs caused by the new core (due to increased enrichment, more frequent refuelings, possible redesign of the control rods, etc.) are included in the above cost. Plant redesign, power replacement, etc., are plantspecific quantities, and the above figure is for reference purposes only.

11/13/82

I-10

1

٩.

(6) A one-time core redesign cost of \$20 M is assumed for any core needing fuel assembly replacement. This cost will depend on the number of fuel assemblies to be replaced, the size of the core, the objectives of the redesign, etc. This cost is only an approximate figure and should be appropriately adjusted for individual plants. It includes fabrication of dummy replacement assemblies and other necessary core changes; for example, flow adjustments.

Table I.4, Operating Reactor RT_{NDT} Evaluation, was compiled using currently available information (Ref. I.16). The table shows the critical element of the pressure vessel, i.e., circumferential weld, axial weld, or the plate and the remaining increase in RT_{NDT} before the screening criteria are reached. The EFPYs of operation and the fluence at the end of 1981, and the total fluence at which the screening criteria would be reached are also listed. Using the above information, the required annual fluence to just reach the screening criteria at the end of 32 EFPYs was computed. Finally, from this information, the factor by which the flux to the pressure vessel must be reduced so as not to exceed the screening criteria at the end of the 32 EFPY was calculated.

The Flux Reduction Factor (FRF) permits the staff to distinguish the following categories of plants (a) those with FRF(1 which indicates that no flux reduction would be needed; (b) plants with $1 \le FRF \le 2$, which indicates that the flux reduction necessary not to exceed the screening criteria before the end of life of the plant can be achieved with ordinary low leakage fuel management (which can result in economic benefits due to longer fuel cycles and increased burnup); (c) plants with $2 \le FRF \le 3$, which indicates that combined low leakage loading and azimuthal flux distribution management can achieve the goal of meeting the criteria at 32 EFPY (depending on the particular plant under consideration, loss of thermal margin and operational flexibility may result, which may offset part of the economic gains of low leakage fuel loading); and (d) plants with FRF >4. There are four plants in this last category, i.e., H. B. Robinson 2, Turkey Point 3 and 4, and Fort Calhoun. Due to the large flux reduction required to meet the screening criteria at the end of the 32 EFPY, these plants are discussed separately.

I-11

11/13/82

۵

PTS RPT APP 1

9. E

The effect of low leakage and of fuel assembly substitution has been studied by the staff in some detail. The azimuthal variation of the end of life fluence in a variety of peripheral assembly substitutions has been studied, including zero power assemblies, nominal stainless steel rods, and closed packed stainless steel rods. These studies were carried out for Maine Yankee, H. B. Robinson-2, Oconee-1 and Fort Calhoun (Ref. I.17). Analysis of San Onofre, Turkey Point, and Three Mile Island is well under way.

554

H. B. Robinson 2

The applicant's end of life fluence to the pressure vessel for H. B. Robinson-2 was 6.4 x 10^{19} n/cm². Staff sponsored calculations indicated a corresponding value of 7.4 x 10^{19} n/cm². This brought the estimated fluence to the end of 1981 to 1.64 x 10^{19} . In the meantime H. B. Robinson-2 has refueled in 1982 with a low leakage core (extended burnup) which lowered the flux to the peak location by about 50 percent. The current cycle will last about 1.0 EFPY with an accumulated total of 8.0 EFPY. The peak fluence location will be at 1.76 x 10^{19} n/cm² which brought the available margin at the end of 1982 to .31 x 10^{19} n/cm², i.e., RT_{NDT} = 290°F (10° F remaining) and at the end of the current cycle to .17 x 10^{19} n/cm², i.e., 296°F (4°F remaining).

The required peak flux reduction factor to meet the screening criteria at the end of the plant life >9.5 with the current low leakage scheme or >20 without a low leakage core. The maximum obtainable with the removal of the outer row of assemblies is about 9.5

Assuming removal of the outer row of assemblies and substitution with close packed stainless steel assemblies, the plant will reach screening criteria at 14.2 EFPY. At the current rate of fluence accumulation the plant will reach the screening criteria in 2.8 EFPY from the end of 1981, or 1.7 EFPY from the end of the current cycle.

Removal of all peripheral assemblies and substitution with closed packed stainless steel rods will bring the end-of-life vessel fluence to about 1.9×10^{19} , i.e., at the required value, shown in Table I.4. The removals will amount to 36 peripheral assemblies and will result in an increase of

about 10 percent in the absolute peak power assuming the same power level and fuel distribution. Recent experience indicates that a factor of two in the peak power difference can be accommodated with fuel loading measures and careful fuel management. For the remaining 15 percent there are three options or a combination thereof:

- (a) If the plant is decay heat or stored energy limited, relaxation of the Appendix K requirements would allow an increase of up to 10 percent;
- (b) If the plant is limited by MDNBR; a reduction of the criterion by 0.10 would allow increasing the linear power by about 20 percent; and
- (c) Derating by 15 percent.

.

Options (a) and (b) will cost about \$20 M for the design of the new core and will shorten refueling intervals by about 24 percent, if the maximum enrichment remains the same. Increased fuel enrichment might add to the new core costs. Option (c) will cost about \$20 M for the design of the new core and an additional cost of \$9.9 M per EFPY for power replacement.

Turkey Point 3 and 4

The required flux reduction factor is 4.5 from the present fuel loading mode. Since these plants are identical with respect to core arrangement, size, dimensions, material and neutron source distribution with H. B. Robinson-2, we have used the results of the Robinson study. Removal of the three assemblies at the flats with careful fuel management, will reduce the flux by the required amount. It is unlikely that derating would be required. The maximum number of removed peripheral assemblies is 12, i.e., 7.6 percent of the fuel inventory, which produce about 4.3 percent of the power.

The cost is estimated at \$20 M for core redesign. Power replacement cost is not significant.

I-13

Fort Calhoun

Total required flux reduction factor is 4.0. This can be accomplished by the removal of two central assemblies in the flats or the periphery for a total of 8 assemblies. (However, while removal of the central assemblies would reduce the flux by the required amount, the remaining two assemblies may not be functional in their position with the central assemblies removed.) The 8 assemblies represent 6 percent of the fuel inventory and about 5.5 percent of the power. Actual power derating may not be necessary.

The cost is estimated at \$20 M for core redesign. Power replacement cost is not significant.

I.6 Conclusions

- (1) All vendors and licensees provided similar responses to our survey inquiries.
- (2) Presently employed in-out, low leakage loading schemes provide about a 30% reduction of the fast neutron fluence rate as a side benefit derived from extended cycle core designs and may represent overall cost savings to licensees.
- (3) In-out, low leakage loading schemes using twice or thrice burned fuel assemblies on the core periphery can provide a factor of 2 to 3 reduction in the fluence rate to the pressure vessel.
- (4) If in addition to twice or thrice burned fuel, peripheral assemblies loaded with burnable poisons are used, a factor of 5 reduction in fluence rate can be achieved with in-out, low leakage loading schemes.
- (5) If one attempted to maintain the core power rating while implementing low leakage schemes, the power distribution would become more centrally peaked and would require core redesign and fuel rearrangement to flatten power and probably would result in plant derating depending on available plant thermal margin.

¢

- (6) In-out, low leakage schemes can be used to reduce locally fluence rates in areas of peak welds, but may result in slightly higher fluence elsewhere and the appearance of peaks at new locations.
- (7) The exact impact of in-out, low leakage loading schemes is plant dependent and cannot be generalized.
- (8) The effectiveness of in-out, low leakage loading schemes is greatest for plants with large azimuthal flux peaks (CE & <u>W</u>). Implementation in B&W plants would probably involve a large numer of assemblies because of the more uniform azimuthal flux distribution.
- (9) Reduction of the fluence by factors of 10 or higher can be affected by peripheral assembly replacement with nonfueled assemblies (e.g., stainless steel). This can be done locally or uniformly, as needed, depending on the azimuthal flux distribution and location of weld seams.
- (10) use of nonfueled assemblies would result in significant loss of heat transfer area (10-15%), reduced core size, increased thermal peaking, increased linear power generation rates, and increased rod worths. It would require a new core design, with different fuel enrichment and new transient and accident analyses. New limiting safety system setpoints would have to be generated and fuel management philosophies would change.
- (11) Selected replacement would provide local reductions of fluence by a factor of 10 or more. If core symmetry is not maintained, however, the normal means of monitoring core power distribution based on neutron detectors and 1/8 core symmetry would have to be changed.
- (12) Use of nonfueled assemblies cou'd result in power derating of perhaps 30%.
- (13) Effectiveness of any of these fluence reduction schemes depends on previous vessel exposures and materials and is less significant once significant fluence has been accumulated.

11/13/82

I-15

PTS RPT APP I

* 1

(14) The staff is continuing to study the effect of low leakage core and fuel assembly substitution on the fracture toughness properties of operating reactor pressure vessels. There have been four vessels identified which may require large flux reduction to prevent exceeding the proposed screening criteria prior to the projected EOL. Our cost estimate for core redesign and fuel assembly substitution is \$20 M plus power replacement cost.

11/13/82

PTS RPT APP I

REFERENCES

361+

- I.1 Memorandum from R. J. Mattson to L. S. Rubenstein, March 13, 1982.
- I.2 Letter from T. Novak to Licensees, August 21, 1981.
- I.3 Letter from T. LaFleur to: Belgium, Finland, France, Germany, Italy, Japan, Korea, Netherlands, Spain, Sweden and Switzerland, with whom we have bilateral information exchange agreements, July 29 and August 20, 1981.
- I.4 Institute of Radiation Protection, "Embrittlement of Loviisa-1 Reactor Pressure Vessel," Confidential Memorandum, May 15, 1981.
- I.5 MPA Stuttgart, Replies to NRC Questionnaire, September 18, 1981.
- I.6 Reactor Safety Commission, "Safety Against Brittle Fracture of the Reactor Pressure Vessel," NRC Translation 8-23-A, January 1981.
- I.7 Letter from Schnurer of the Federal Ministry of Internal Affairs, West Germany, to LaFleur, NRC, August 10, 1981.
- I.8 Letter and Attachments to the NRC, March 19, 1982.
- I.9 Letter from R. Mattson to Gregory Ogeka of DOE, "BNL Technical Assistance to the Division of Systems Integration, NRR, NRC, Pressure Vessel Radiation Embrittlement Calculations" (FIN A3387), March 19, 1982.
- I.10 Letter, W. V. Parker to H. R. Denton, January 15, 1982 (Oconee Nuclear Station, Docket Nos. 50-269, 50-270, and 50-287).
- I.11 Letter, W. C. Jones (Omaha Public Power District) to T. M. Novak, November 13, 1981; Letter, W. C. Jones to R. A. Clark, November 12, 1981; Letter, W. C. Jones to H. R. Denton, January 23, 1981.

1

I.12 H. B. Robinson Steam Electric Plant, Unit No. 2, Docket No. 50-261, License No. DPR-23, Letter, E. G. Utley to D. G. Eisenhut entitled, "Thermal Shock to Reacto: Pressure Vessels," January 25, 1982 and subsequent communications.

100

- I.13 Report on the PTS Audit Calculations by BNL in a BNL-NUREG report to be published.
- I.14 H. Richings, N. Lauben, personal communication, October 13, 1982.
- I.15 G. M. Hesson and J. M. Cuta, "Analysis of the Sensitivity of Calculated MDNBR to Eight Selected DNB Parameters," Battelle, PNL, FATE-79-101, March 1979.
- I.16 P. N. Randall, personal communication, October 14, 1982.
- I.17 A. Aronson, et al., "Evaluation of Methods for Reducing Pressure Vessel Damage Fluence," to be issued by BNL as a NUREG.

PTS RPT APP I

	Oconee-1	Ft. Calhoun	Robinson-2		
Total Effective Full Power Years of Operation (EFPY) as of 12/81	5.1	5	7.1		
(a) Out-in loading SFPY (b) In-out low leakage EFPY	4 1.1	5	7.1		
Present Vessel Fluence Using Current Fuel Loading Scheme (x10 ¹⁸ n/cm ²)					
(a) BNL calculation (b) Vendor calculation	2.70 2.55 (Ref. I-10)	7.24 6.60 (Ref. I-11)	18.8 13.8 (Ref. I-12)		
End of Vessel Life Fluence (x10 ¹⁸ n/cm ²)					
I. Using Current Fuel Loading Scheme					
(a) BNL calculation(b) Vendor calculation	12.1* 12.5	45.9 42.0	85.0 65.6		
(c) FSAR value	(Ref. I-10) 22.0	(Ref. I-11) 20.0	(Ref. I-12) 51.0		
*II. Using In-out Low Leakage Loading Scheme/ (Increase in Total Poaking	11.2	30.0	53.7		
Factor (%))	(7)	(17)	(17)		
**III. Using Stainless Steel Assemblies on Periphery/ (Increase in Total Peaking	1.90	12.7	28.1		
Factor(%))	(23)	(30)	(23)		

Table I.1 Vessel fluence for Oconee-1, Ft. Calhoun, and H. B. Robinson-2

*Out-in loading scheme value is $18.5 \times 10^{18} \text{ n/cm}^2$. **BNL calculation.

.

•

.

Ø

PTS RPT APP I

4

.

366

		Weld* SA-1430	Weld SA-1493	Weld SA-1073	Peak Wall Location
I.	Flux (x10 ¹⁰ n/cm ² -sec)				
	Fuel Loading Method (a) Out-in, P/P = 0.910 (b) Low leakage, P/P = 0.527 (c) Low leakage, P/P = 0.0	1.59 .984 .125	1.37 .915 .188	1.45 .911 .125	1.84 1.11 .188
Π.	Fluence for 28 EFPY (x10 ¹⁸ n/cm ²)				
	Fuel Loading Method (a) Out-in, P/P = 0.910 (b) Low Leakage, P/P = 0.527 (c) Low Leakage, P/P = 0.0	13.5 8.35 1.05	11.7 7.76 1.60	12.3 7.73 1.06	15.6 9.41 1.60
III.	Fluence for 32 EFPY (x10 ¹⁸ n/cm ²)				
	Fuel Loading Method (a) Out-in, P/P = 0.910 (b) Low leakage, P/P = 0.527 (c) Low leakage, P/P = 0.0	16.1 9.93 1.25	13.9 9.23 1.90	14.6 9.20 1.26	18.5 11.2 1.90

Table I-2 Staff evaluation of flux and fluence to weld seams and peak fluence location for Oconee-1

5 • · •

¢

4

*See Figure I-1 for weld seam location.

.

11/13/82

PTS RPT APP I

Table I-3	Staff evaluation of the ratio of the fluence rate to the weld seams and the original peak fluence location to the fluence rate of Weld Seam SA-1430 for the remaining
	28 EFPY for Oconee-1

	Relative Fluence and Fluence Rate Reduction Factor*										
Fuel Loading Method	Weld SA-1430	Weld SA-T493	Weld SA-1073	Original Peak Fluence Location							
Out-in, $P/\overline{P} = 0.91$	1.00 / 1.00	.863 / 1.16	. 909 / 1.10	1.15 / .87							
In-out, low leakage P/P = 0.527	.618 / 1.62	.575 / 1.74	.572 / 1.95	.697 / 1.43							
In-out, low leakage P/P = 0.0	.078 / 12.8	.118 / 8.47	.078 / 12.8	.098 / 10.2							
Stainless Steel Assemblies, P/P = 0.0	.049 / 20.4	.103 / 9.71	.057 / 17.5	.077 / 13.0							
Stainless Steel Assemblies,** P/P = 0.0	.033 / 30.3	.081 / 12.3	.040 / 25.0	.061 / 16.4							

*First number is the fluence rate ratio; the second number is the fluence rate

reduction factor.
**This case includes nonfueled assemblies with additional stainless steel rods in a
close packed array.

I-21

PTS RPT APP I

		- 1 - As of 12/31/81												
Plant, Vendor/ PV Fabri- cator/ MWe	Controlling element Axial or Circumferential weld or plate	^{KT} NDT Betow Screening Criteria (°F)	Total Fluence E>1 MeV to Meet Screening Criteriag n/cm x10	EFPY	Fluence (n/cm ² x10 ¹⁹)	Fluence per EFPY (n/cm ² x10 ¹	Additional Fluence To Reach Screening Criteria n/cm ² x10 ¹⁹	Remaining EFPY in Plant Life (32 EFPY)	To Reach S <u>Criteria a</u> Fluence per_EFPy (n/cm_x10 ¹⁹)	creening t 32 EFPY Flux Reduction Factor	1 FRF~ 1	1< FRF< 2 	2 <frf<3< th=""><th>FRF> 4</th></frf<3<>	FRF> 4
Robinson-	2 circ.	19	1.95	7.1	1.41	. 199	54	24 0	0217		[1		1
W/CE/065 Fort Calh CE/CE/48	oun axial 6	12 44	1.95 1.18	7.1 5.07	1.64	.230	.31 .67	24.9 26.93	.0124 .0249	$\frac{9.2(1)}{18.5(1)}$ 4.0	; 1	1		X
Turkey Pt W/B & W/	4 círc. 666	41	1.85	5.67	.91	. 160	.94	26.33	.0357	4.5(2)	1 1	1	 	Y
Turkey Pt W/B & W/	3 circ.	41	1.85	5.67	.91	. 160	.94	26.33	.0357	4.5(2)	1	1	· ·	Ŷ
Maine Yan CE/CE/82	kee axtal 5	54	1.18	5.90	.41	.069	.77	26.10	.0295	2.3	1	t 1	x i	•
Calvert Cliffs-1 CE/CE/85	axial 0	131	8.22	4.65	. 68	. 146	7.54	27.35	.276	.53 ⁽³⁾	1 1 X	₽ ₽		
M/CE/965	3 plate	58	1.04	2.98	.167	.056	.873	29.02	.0301	1.9	r •	X		
Vankee Ron W/B & W/1	we plate 175	59	4.48	14.56	1.14	.078	3.34	17.44	. 1915	.4	1 4 X	1		1
Rancho Sec B & W/B &	co axial W/913	63	.77	3.54	. 205	.059	.565	28.46	.0199	2.9	τ 1	l [·]		
Island-1 B & W/B &	e axial & W/792	66	.75	3.52	. 187	.053	.563	28.4 8	.0198	2.7	1	t I	i X I	,
Oconee-2 B & W/B &	circ. W/860	69	. 99	4.71	.287	.061	.703	27.29	.0258	2.4	,] !	l	x 1	i
Zion-1 W/B & W/1	circ.	75	1.25	4.97	.313	.063	.937	27.03	.0347	1.8	1	1 X	1	•
Point Beac W/B & W/4	:h-l avial 197	72	3.48	8.07	,734	.091	2.750	23.93	.1149	.8	X	1	1 I 1	ł
Oconee-1 B & W/B &	axial 1 W/86C	73	1.33	5.04	.273	.054	1.057	26.96	.0392	1.38	1	X	1	•
Indian Pt. W/CE/873	2 axial	81	1.18	4.40	.22	.050	.9 60	27.60	.0348	1.4	; •	ι χ	, 1 ; 1	1

.

			As of 12/31/81											
Plant, (Vendor/_ele PV Cii Fabriwel cator/ MWe	Controlling element Axial or Circumferential weld or plate	RT NDT Betow	Total Fluence E>1 MeV to Meet Screening			Fluence	Additional Fluence To Reach	Remaining	To Reach S Criteria a	creening t 32 EFPY	 	 1 <frf~2 </frf~2 	2-FRF-3	FRF>4
		Screening Criteria (°F)	Criteria n/cm²x10 9	EFPY	Fluence (n/cm ² x10 ¹⁹)	per EFPY (n/cm ² x10 ¹	19 Screening) Criteriag n/cm ² x10 ¹	EFPY in Plant Life (32 EFPY)	Fluence per ₂ EFPY (n/cm ² x10 ¹⁹)	Flux Reduction Factor	ł			: 1 ! .
Ginna W/R & W/A	circ.	87	4.95	8.18	.949	.116	4.000	23.82	. 1679	.7	<u>, X</u>	I	.	┝───
Point Bea W/B & W, Arkansas	ch-2 circ. CE/497	. 85	4.65	7.54	. 935	. 124	3 .69 7	24.46	.1511	.8	x I	· ·		
ANO-1 B & W/B	axia1 & W/836	82	1.23	4.42	.199	.045	1.031	27.58	.0374	1.2	1 1	x	1 j	! ! .
San Onofre W/CE/436	e-1 axial	89	12.23	9.04	2.71	. 300	9.520	22.96	.4146	.7	+ , Χ		1	:
Zion-2 B & W/B	axial & W/1040	93	.77	4.49	.09	.02	.680	27.51	-0247	.8	i x		1 I	; I (]
Palisades CE/CE/740 Crystal	axial O	93	2.33	4.12	.478	.116	1.852	27.88	.0664	1.7	# { *	 X		
River-3 B & W/B	axia] & W/825	93	1.18	2.48	.136	.055	1.044	29.52	.0354	1.6	1	x		i i
Surry-1 W/B & W/2	circ.	100	5.50	4.88	.761	. 159	4.74	27.12	.1748	.9	X		1	. ł
Cock-1 W/CE/1054	circ.	100	1.94	4.56	.287	.063	1.653	27.44	.060	1.05	1	X	· ·	· •
North Anna W/RG,'865	a-1 plate	108	11.57	2.41	.442	. 183	11.13	29.59	.376	. 486	x	 	1 1 1 1	
Beaver Val W/CE/833	lley circ.	118	2.06	1.87	.316	.169	1.744	30,13	.058	2.91			· · · ·	1 i 1
North Anna W/RD/890	a-2 plate	118	10.1	.77	.138	.179	9.962	31.23	.319	.56	X	ļ 		, I
Salem-1 W/CE/1090	axial)	130	3.68	2.26	.148	.065	3.532	29,74	.119	.55	I X	ī	1	1
Oconee-3 B&W/B&W/8	circ. 360	129	5.04	4.78	.292	.061	4.748	27.22	.174	.35	X		1	1 1
Surry-2 W/686/775	axial	130	14.8	4.83	.754	.156	14.05	27.17	.517	.30	; X	l L	, 1	ł

•

		As of 12/31/81												
Plant, Co Vendor/ elem PV Circ Fabri- weld cator/ HWe	Controlling element Axial or Circumferential weld or plate	RT _{NDT} Below Screening Criteria (°F)	Total Fluence E>1 MeV to Meet Screening Criteriag n/cm x10	EFPY	Fluence (n/cm ² x10 ¹⁹)	Fluence per EFPY (n/cm ² x10 ¹¹	Additional Fluence To Reach Screening) Criteria n/cm ² x10 ¹⁹	Remaining EFPY in Plant Life (32 EFPY)	To Reach Screening Criteria at 32 EFPY Fluence Flux per_EFPY Reduction (n/cm ² x10 ⁹) Factor	FRF4 1]= FRF- 	2 2 FRF- 3	3 FRF>4	
St. Lucie-1 CE/CE/777	axial	135	3.02	3.52	.222	.063	2.798	28.48	.098	.64	<u> </u>	1		-
Calvert Cliff CE/CE/850	s-2 axial	140	8.48	3.63	.534	.147	7.946	28.37	- 280	.53	I x	ļ	1	
Trojan W/CBI/1130	plate	147	16.2	3.00	.207	.069	16.00	29.00	.552	.13	i x	1	1 I	1
Davis Besse-1 B&W/B&W/906	circ.	156	5.25	1.68	.111	.066	5.14	30. 32	. 170	.39	X	1		1
Haddam Neck W/CE/582	axial	161	36.48	10.92	1.190	.109	35.29	21.08	1.674	.07	X	1		
Kewaunee W/CE/535	circ.	168	17.14	5.87	. 786	.134	16.35	26.13	.626	.21	X	1	1	
Farley-1 W/CE/829	axial	168	11.61	2.19	. 370	. 169	11.24	29.81	.377	.45	X	1		
MT11stone-2 CE/CE/870	circ.	183	7.62	3.91	.219	.056	7.40	28.09	.263	.21	X	i I		
Prairie Island W/SFAC/520	d-2 axial	216	90.7	5.62	.753	.134	89.95	26.38	3.41	.04 i	X	· .		
Prairie Island W/SFAC/520	d-1 axial	237	292.1	5.90	.790	.139	291.3	26.10	11.16	.01	X	1		
										Summary,	23	8	5 1	 4

(1)The lower line in H. B. Robinson lists the staff's calculations. The differences have not been resolved as of 11/10/82. Note that the current cycle of the Robinson plant (after 12/31/81) is a low leakage cycle, therefore, the FRF would currently be lower by a factor of 2, i.e., 4.6 and 9.2 for licensee and NRC calculations.

(2) The staff has been notified by Florida Power and Light (Ref. 5) that the fluence at 5.67 EFFY was .91 x 10^{19} n/cm² and the projected fluence for the next 3 EFPY will be 1.39 x 10^{19} n/cm² i.e., the fluence accumulations rate will remain the same. These values have not been reviewed by the staff.

(3) The value of the fluence required to reach the screening criteria is based on .21% copper content in the weld seam (Ref. 6).



FIGURE 1-1 LONGITUDINAL WELD LOCATIONS FOR OCONEE-1

.



FUEL ELEMENT GRID AND LONGITUDINAL WELD SEAMS FOR FORT CALHOUN FIGURE 1-2



FUEL ELEMENT GRID AND LONGITUDINAL MELD SEAM LOCATIONS FOR H. B. ROBINSON-2 FIGURE I-3

power for oconee-1 (for remaining 28 effys)

.





375-





,


FIGURE I-6 Peak Fluence As A Function of EFPY For Various Fluence Rate Reduction Schemes For Oconee-1

Pattern 1 Removal and replacement of outer row of assemblies with 40 stainless steel assemblies Pattern 2 Removal and replacement of 20 assemblies opposite weld seams Pattern 3 Modification of Pattern 2 with additional removals and replacement of peripheral assemblies (32 total)





Pattern 1Low leakage scheme(peripheral to core average power = 0.41)Pattern 2Low leakage scheme(zero peripheral power)Pattern 3Assembly replacement with 40 stainless steel assembliesPattern 4Assembly replacement with 24 stainless steel assembliesPattern 5Assembly replacement with 16 stainless steel assemblies



Figure I-8 Peak Fluence As A Function of EFPY for Various Fluence Rate Reduction Schemes for Robinson-2

```
Pattern-1 Out-in fuel loading, P/P = 0.89
Pattern-2 Low leakage loading, P/P = 0.45
Pattern-3 Low leakage loading, P/P = 0.0
Pattern-4 Fuel assembly replacement with 12 stainless steel assemblies
Pattern-5 Fuel assembly replacement with 20 stainless steel assemblies
Pattern-6 Fuel assembly replacement with 36 stainless steel assemblies
```

APPENDIX J

31.

SUMMARY OF NRC STAFF POSITION ON REVIEW OF CONTROL SYSTEMS

J.1 Introduction

.

The following summarizes the staff philosophy on the review of control and protection systems and delineates actions completed or planned to address the effects of control systems on plant safety. The following also specifically discusses the possible impact of control system failures on pressurized thermal shock and actions which should be considered to minimize the possibility of control system failures resulting in an excessive plant cooldown transient.

J.2 Philosophy of Separation of Protection and Control Systems

The philosophy on the separation of protection systems and control systems was developed in the 1960's and early 1970's through interactions between the regulatory staff and industry. The interactions occurred primarily through the development of industry standards such as IEEE-279, "Criteria for Protection Systems for Nuclear Power Generating Stations." The staff did not dictate a particular philosophy, but rather explored through the standards committees and early plant licensing reviews various approaches which could be taken toward reactor protection.

A brief, simplified description of the approach toward protection and control is as follows. A nuclear power plant must satisfy utility requirements for the economic production of power. These requirements include plant operation with a limited number of operators, high plant availability with few unplanned shutdowns, and the ability to follow the utility grid load demand. The requirements for operation are based largely on matching the capabilities of nonnuclear plants. Plant control systems to accomplish the desired economic operational characteristics are established. The control systems, of course, have to be capable of allowing the plant to perform normal operations with margin to plant safety limits.

To assure that safety limits are not exceeded should any system used for normal operation fail, various protective functions such as reactor trip and decay heat removal have been established in the Commission regulations. Systems whose primary purpose is to accomplish the protective functions are provided to fulfill these requirements.

One, thus, has two somewhat differing objectives. The first is to allow normal plant operation within a utility grid which is also supplied by many non-nuclear plants. For this, control functions have been established. The second objective is to ensure that even with failures of the operational equipment, safety limits are not exceeded. For this, protective functions have been established to assure plant safety.

Once control functions and protective functions are defined, a decision has to be made as to whether the same systems should be used for both or whether separate systems should be used. The philosophy developed through the standards committees was one in which the protection systems were treated separately. This allowed a set of guidelines to be established with the intent of ensuring that protection functions are accomplished with a very high degree of reliability. Having a specific, well-defined group of protection systems to accomplish required safety functions allows both industry and the regulatory agency to concentrate their efforts and make effective use of limited resources in accomplishing safety goals.

In development of the philosophy, it was recognized that some limited ties between protection systems and control systems are appropriate and even unavoidable. For example, the systems will always be interrelated through the fluid process systems. Additional interfaces such as the use of the same sensors for protection and control were considered acceptable providing appropriate rules are followed. General Design Criterion 24 and IEEE-279 permit limited interconnections between protection and control — tems and define rules for implementing these interconnections.

J.3 NRC Staff Reviews of Control Systems

,

NRC staff reviews have been performed on currently licensed plants with the goal of ensuring that control system failures will not prevent automatic or manual initiation and operation of any safety system equipment required to trip the plant or maintain the plant in a safe shutdown condition following any "anticipated operational occurence" or "accident." The approach has been to either provide independence between safety and nonsafety systems or to require isolating devices such as isolation amplifiers between safety and nonsafety systems such that failures of nonsafety system equipment cannot propagate through the isolating devices to impair operation of the safety system equipment. In addition, a specific set of "anticipated operational occurrences" and "accidents" have been analyzed to demonstrate that plant trip and/or safety system equipment actuation occurs with sufficient capability and on a time scale such that the consequences are within specified acceptable In these analyses, conservative initial plant conditions, core physics limits. parameters, equipment availability, and instrumentation setpoints have beer assumed. Conservative parameters (for example, heat fluxes, temperatures, pressures, and flows) which could result in core or reactor coolant system pressure boundary damage are also assumed. Where active control system operation would mitigate the consequences of a transient, in general, no credit is taken for the control system operation. In some cases, credit has been allowed for the operation of specific control systems in mitigating the consequences of particular "anticipated operational occurrences." Where this has been allowed, special design features and/or technical specification requirements such as periodic testing have been provided.

33.0

Where active control system operation would not mitigate the consequences of a transient, no penalties are taken in the analyses for incorrect control system actions caused by control system failures. In the case of control systems, for example, the loss of forced reactor flow is analyzed assuming the reactivity control systems either operate properly or do not operate at all, whichever is the worst case. A loss of forced reactor flow occurring simultaneously with an inadvertent rod withdrawal is not considered. Among the specified set of "anticipated operational occurrences" analyzed are occurrences resulting from both mechanistic and nonmechanistic control system failures. The conservative

analyses performed are intended to demonstrate that the potential consequences to the health and safety of the public are within acceptable limits for a wide range of postulated events even though specific actual events might not follow the same assumptions made in the analyses.

In general, until approximately one year ago systematic evaluations of control systems designs had not been performed to determine whether single event induced multiple control system actions could result in a transient such that core or reactor coolant system pressure boundary limits established for "anticipated operational occurrences" are exceeded. Single failures or events which could include multiple control system actions such as discussed above do indeed exist, but experience with operating plants indicates that incidents resulting in transients more severe than currently analyzed as "anticipated operational occurrences" have a low probability. Recent operating plant license applicants have been required to address the possibility of multiple control system actions such as a power supply failure or sensor impulse line failure.

The app is is have been required to identify any power sources, sensors, or sensor impulse lines which provide power or signals to two or more control systems and demonstrate that failures of these power sources, sensors, or sensor impulse lines will not result in consequences more severe than those bounded by the analyses of "anticipated operational occurrences" in Chapter 15 of the FSAR. At this time, similar reviews have not been required of operating plant licensees. However, the effort on the current license applications will provide general guidance on whether significant problems may exist on operating plants.

Until approximately two and one-half years ago systematic evaluations of control system designs had not been performed to determine whether postulated accidents could cause control system failures resulting in control actions which would make accident consequences more severe than presently analyzed. Accidents could cause control system failures by creating a harsh environment in the area of the control equipment or by physically damaging the control equipment. Licensees have, however, now reviewed the possibility of consequential control system failures which exacerbate the effects of high

1,5 3

energy line breaks and taken action, where needed, to assure that the postulated events would be adequately mitigated. Similar efforts are also being performed on plants currently under operating license review.

.

It should be emphasized that the issue is not whether reactor trip or safety system equipment action would be defeated by control system failures, but whether control system failures could cause a transient or accident to proceed in a manner potentially more severe than currently analyzed. Systematic reviews of safety systems have been performed with the goal of ensuring that control system failures (single or multiple) will not defeat trip or safety system action, and both industry standards and staff regulatory guides are quite clear that this is a design requirement for safety systems including those used for reactor trip.

As part of the staff's ongoing activities to access the adequacy of non-safety grade control systems, Unresolved Safety Issue (USI) A-47 "Safety Implications of Control Systems" was established to perform an indepth evaluation of the control systems that are typically used during normal plant operation and (1) to verify the adequacy of current licensing design requirements or (2) to propose (if necessary) additional guidelines and criteria to assure that nuclear power plants do not pose an unacceptable risk due to inadvertent nonsafety grade control system failures. This activity will review the plant designs of the manual and automatic control systems for each of the four nuclear steam system supplier designs (B&W, CE, GE and W).

The activity will evaluate and identify any non-safety grade control systems whose failure (1) may lead to transients or accidents more severe than those analyzed in Chapter 15 of the plant FSAR, or (2) could produce unacceptable frequency of occurrence of those transients bounded by Chapter 15 analysis. As specific subtasks of these reviews, failures that could lead to steam generator or reactor vessel overfill or overcooling transients will be evaluated.

It is anticipated that the technical resolution of this safety issue will be completed in 1984.

J.4 Instrumentation and Control System Impact on Pressurized Thermal Shock

Control system failures can cause inadvertent reactor coolant system cooldowns and inadvertent increases in reactor coolant system pressure. Whether any credible control system failures can cause unacceptable reactor coolant system temperature/pressure combinations, however, requires further analyses.

There are control system failures which can cause excessive feedwater flow or abnormally low feedwater temperature, either of which could lead to reactor coolant system cooldown. If it is found necessary through review of limiting transients, feedwater flow can be terminated automatically with safety-grade equipment following detection of an excessive cooldown. If the problems of concern are found to be only with the control system (and not, for example, with feedwater valve failures) then safety-grade interlocks could be used to redundantly override the control system and terminate feedwater. If there is a concern with excessive feedwater caused by valve malfunction (such as a feedwater control valve failing open) feedwater could be terminated with safety-grade equipment by closing redundant valves or by tripping feedwater pumps and closing a single set of valves for redundancy. This method of terminating feedwater flow could, however, require the addition of expensive equipment on some plants. Also, analyses would have to be performed to determine if feedwater pump trip or valve closure could be accomplished sufficiently rapidly to mitigate any transient of concern.

There are control system failures which can cause excessive steam flow through electric, air, or hydraulic operated steam valves which could lead to reactor coolant-system cooldown. As with the feedwater flow, steam flow could be terminated with safety-grade interlocks or safely-grade isolation valves following detection of excessive cooldown. If a cooldown transient, however, is initiated by a "stuck open" safety valve, it could not be terminated by safety system equipment since design codes prohibit isolation valves in series with safety valves.

Inadvertent reactor coolant system pressure increases caused by control system failure can be terminated by redundantly turning off pressurizer heaters or redundantly terminating charging flow if shown to be necessary. However, it

should be noted that inadvertent cooldowns of sufficient magnitude will, in general, result in eventual automatic initiation of safety injection which, in turn, results in an increase in reactor coolant pressure if operator action is not taken.

A number of plants currently employ interlocks and valves which are redundant and at least "quasi-safety-grade" to automatically terminate feedwater flow and/or steam flow under conditions which could lead to inadvertent cooldown, overfill of steam generators (PWRs), or overfill of reactor vessels (BWRs).

In addition to inadvertent cooldowns or increases in pressure which can be caused by control system failures, actuation of certain emergency safeguards systems can cause inadvertent cooldown and consequential increase in reactor coolant pressure. For example, actuation of auxiliary feedwater on a PWR following a reactor trip can cause an inadvertent reactor coolant system cooldown, contraction of water in the reactor coolant system, depressurization of the reactor coolant system, automatic actuation of safety injection, and then a repressurization of the reactor coolant system. This could occur if operator action is not taken to manually control auxiliary feedwater after its automatic initiation. During recent operating license reviews, the Instrumentation and Control Systems Branch has been reviewing the circuits, equipment, and indications used by the operator to control auxiliary feedwater after automatic initiation with the goal of ensuring that a single failure will not cause uncontrolled auxiliary feedwater flow. A staff position on the design of the auxiliary feedwater system, including instrumentation and controls, has been proposed and is currently under review by the Division of Safety Technology. Implementation of this position would significantly improve the failure tolerance of the auxiliary feedwater system from the standpoint of failures which could result in excessive plant cooldown.

J.5 Actions Completed or Underway to Detemine Potential Consequences of Control System Failures

The consensus judgment of the NRC staff continues to be that the risk associated with control system failures is not sufficient to require immediate corrective

actions. However, to provide added assurance, the following actions are being or have been taken:

4.11

- (1) The resolution of Unresolved Safety Issue A-47, "Safety Implications of Control Systems" will systematically determine if current licensing practices with respect to control systems are adequate. The plan for resolution of this issue specifically addresses evaluations to determine any actions required to prevent control system failures from causing unacceptable reactor coolant system cooldown or overfill of a steam generator (PWR) or reactor vessel (BWR).
- (2) Standard Review Plan Section 7.7 calls for staff reviews to assure that failures of control systems will not impair the capability of the protective system in any significant manner or cause plant conditions more severe than those for which the plant safety systems are designed. The staff has pursued these reviews primarily to ensure that electrical interconnections between protection systems and control systems are implemented such that failures in control system equipment cannot impair the operation of protection system equipment. The Chapter 15 design-basis event analyses have also been reviewed to assure that sufficient conservatism has been assumed so that these analyses adequately bound the consequences of single control system failures. The Instrumentation and Control Branch has been reviewing control system designs of operating license applicants to confirm that the Chapter 15 design bases analyses also bound multiple control system failures initiated by credible failures of common power sources, sensors, or sensor impulse lines. In addition, operating license applicants have been requested to review the potential for control system malfunctions caused by high energy line breaks. Section 7.7 of the Standard Review Plan was revised in 1981 to be more explicit on criteria applicable to control systems. Specifically, the criteria shown in Table J.1 are now delineated in Section 7.7 and reviews of plants currently under licensing review are performed with the goal of verifying that the criteria are met.
- (3) In September 1979, all licensees were asked to review the possibility of consequential control system failures which could exacerbate the effects

of high energy line breaks and identify appropriate actions, where needed, . to assure that these events would be adequately mitigated. The review was requested as a result of postulated scenarios involving consequential control system failures identified by Westinghouse. All licensees responded to the request and the responses were screened. On the basis of the review, no specific event leading to unacceptable consequences was identified and, in general, control equipment locations were such that consequential failures would be unlikely. Some licensees, however, did make changes to operating procedures to address the possibility of control failures.

- (4) I&E Bulletin 79-27 was issued to licensees requesting that evaluations be performed to ensure the adequacy of plant procedures for accomplishing shutdown upon loss of power to any electrical bus supplying power for instruments and controls. In their response to the bulletin, licensees have indicated that corrective action has been taken including hardware changes and revised procedures, where required to assure that the loss of any single instrument bus would not result in the loss of instrumentation required to mitigate such an event. As part of operating license reviews, we are requesting similar verification by operating license applicants.
- (5) Implementation of Regulatory Guide 1.97, "Instrumentation for Light-Water-Cooled Nuclear Power Plants to Assess Plant and Environs Conditions During and Following An Accident," and NUREG-0737, "Clarification of TMI Action Plan Requirements," will significantly upgrade both the quantity and quality of information available to the operator to diagnose and respond to control system failures.
- (6) In 1979 B&W completed a failure modes and effects analysis and review of operating experience for their Integrated Control System (ICS) and reported the results in B&W Report BAW-1564, "Integrated Control System Reliability Analysis." B&W made several recommendations regarding control system improvements which could be made to improve overall plant performance. Licensees with B&W plants were requested to evaluate the B&W recommendations and report their follow-up actions to the staff. Responses were received and reviewed. Based on the review of BAW-1564 and the responses

to the B&W recommendations, the staff has not identified any specific control system failures or actions that would lead to unacceptable consequences.

· ~ ^ i

(7) The Office of Nuclear Regulatory Research is coordinating efforts with the IEEE to establish design criteria for systems important to safety which are not covered by and do not need to meet all of the rigorous standards for safety system equipment but nevertheless may be sufficiently important to safety to be included in the NRC review process.

J.6 Conclusions

At this time, the staff knows of no specific control system failures or actions which would lead to unacceptable consequences. A variety of efforts are still underway to determine the potential safety consequences of control system failures including their impact on pressurized thermal shock. Should these reviews indicate that additional criteria for control system designs are necessary or that specific problems require resolution, appropriate action will be taken for plants in the licensing process and for plants now in operation.

Table J.1

- 1 e

Standard Review Plan Guidance for Control System Review

- (1) Confirm That The Plant Accident Analyses in Chapter 15 of the SAR Do Not Rely On The Operability Of Control Systems To Assure Safety.
- (2) Confirm That The Safety Analyses Include Consideration Of The Effects Of Both Control Systems Action And Inaction In Assessing The Transient Response Of The Plan For Accidents And Anticipated Operational Occurrences.
- (3) Confirm That Consequential Effects Of Anticipated Operational Occurrences And Accidents Do Not Lead To Control Systems Failures Which Would Result In Consequences More Severe Than Those Bounded By The Analyses In Chapter 15 Of The SAR.
- (4) Confirm That The Failure Of Any Control System Component Or Any Auxiliary Supporting System For Control Systems Will Not Cause Plant Conditions More Severe Than Those Bounded By The Analyses Of Anticipated Operational Occurrences In Chapter 15 Of The SAR (The Evaluation Of Multiple Independent Failures Is Not Intended).

APPENDIX K

EFFECTS OF HEATING ECCS WATER ON PRESSURIZED THERMAL SHOCK

Increasing the temperature of the ECCS water can have a positive effect on PTS for LOCA events, where the dominant overcooling results from the injection of the cold ECCS water.

K.1 Large- and Small-Break LOCAs and Secondary Side Effects

It can be shown by analysis that large-break LOCAs are not considered to be a serious PTS problem. This is because in the unlikely event of a large break in the primary system, high pressure cannot be maintained in the reactor pressure vessel. Small-break LOCAs (less than two inches equivalent diameter) also are not a PTS problem because breaks in this size range result in cooldown rates of less than 100°F per hour. Such transients do not cause large thermal stresses. Breaks in the range of two up to possibly as large as six inches arr of concern. These breaks are capable of removing all of the decay heat generated in the core and do not require or establish natural circulation for decay heat removal. Mixing of ECCS water in the downcomer is minimized in this case. (See Section K.3.) In addition, reactor system pressure can remain relatively high $(\sim 1200 \text{ psi})$ for the considerable amount of time required to uncover the break (i.e., steam discharge out of the break), or repressurization can occur after initiation of the break for some plants with high head HPI pumps. This scenario, loss of natural circulation with high pressure, at present appears to be the one most likely to benefit from heating ECCS water in order to reduce the PIS problem. For secondary side events (e.g., main steam line breaks), rapid cooldown and depressurization of the primary system can occur. ECCS actuation will repressurize the primary system. However, since there is no primary system LOCA, only a limited volume of ECCS water will be injected into the primary system by the operator to make up for shrinkage due to cooldown. Therefore, as far as PTS is concerned, the cooldown is not affected as much by ECCS injection as by primary to secondary heat transfer. However, for pertain secondary side events (e.g., steam and feedline breaks) including steam generator tube rupture, interruption of circulation and consequent temperature transients

K-1

that could be influenced by ECCS water temperature could be possible. At this time, sufficient analysis has not been done, and conclusions regarding these events would be premature.

K.2 Plants Which Have Raised ECCS Water Temperature

Several plants have the capability to heat the ECCS water. Connecticut Yankee heats the refueling water storage tank (RWST) to 50°F in the winter to prevent water in the outdoor tank from freezing. Maine Yankee has a technical specification to maintain the RWST at a minimum temperature of 40°F. The water currently is heated no higher than 80°F. Yankee Rowe is the only U.S. plant that Leats its ECCS water substantially above normal ambient temperatures, even in the summer. The safety injection tank water temperature is maintained at 120°F (130°F maximum) to minimize any PTS problem. A review has been conducted by Yankee Atomic Electric Company to ensure that the increased water temperature would not adversely impact protulated accidents.

The Loviisa plant in Finland maintains the ECCS water temperature between a minimum of 113°F and a maximum of 140°F. One of the reasons for this is because the low pressure ECCS system injects through nozzles directly into the reactor vessel. There is no mixing in the cold leg, so the ECCS water is heated to minimize the thermal shock.

K.3 Mixing of ECCS Water

Mixing of ECCS water with water in the reactor vessel has been and continues to be evaluated through analysic and experimentation (Ref. K.1-K.3). As long as adequate reactor coolant flow is maintained, good mixing of ECCS water in the cold leg downcomer is expected, and heating the ECCS water is expected to be of little benefit from a PTS standpoint. In the event that loop flow stagnated, the degree of mixing of ECCS water injected into the cold leg is less certain. If mixing were minimal, colder ECCS water could contact the reactor vessel wall, and therefore, heating the ECCS water would be beneficial in reducing thermal stresses.

K-2

Investigations of the mixing phenomena under stagnant loop flow conditions are underway in order to better quantify the degree of mixing.

K.4 Maximum ECCS Water Temperature

.

The maximum heating that could be allowed without causing other problems with ECCS operation has not been calculated. The impact on containment sprays, pump net positive suction head, and ECCS performance are examples of factors that could limit the water temperature. Evaluations such as these would have to be done on a plant-specific basis.

REFERENCES

Ø · · ·

- K.1. Rothe, P. H., and W. D. Marsher, <u>Fluid and Thermal Mixing in a Model</u> <u>Cold Leg and Downcomer with Vent Valve Flow</u>, EPRI NP-227, March 1982.
- K.2. Rothe, P. H., and M. F. Ackerson, <u>Fluid and Thermal Mixing in a Model</u> <u>Cold Leg and Downcomer with Loop Flow</u>, EPRI NP-2312, April 1982.
- K.3. Levy, S., <u>An Appropriate Prediction of Heat Transfer During Pressurized</u> <u>Thermal Shock</u>, SLI-8213, June 1982.

APPENDIX L

NONDESTRUCTIVE EVALUATION METHODS

L.1 Detectability of Underclad Cracks

6

In order to have confidence that an inservice inspection (ISI) could detect near surface flaws in reactor pressure vessels that would be of interest in a pressurized thermal shock incident, it is necessary to demonstrate high probabilities of detection for 6.0 mm and larger cracks. Cracks of interest are those normal to the inside surface and oriented both parallel and perpendicular to the direction of clad lay. Weld defects within the first 25 mm as well as cracks resulting from clad deposition are of interest. European techniques using longitudinal waves are generally accepted as providing optimum detection results and have been shown to be effective in detecting 3.0 mm or smaller underclad cracks under the more ideal conditions of smooth clad and cracks predominantly perpendicular to the clad lay found in European pressure vessels. Most circumferential welds in U.S. pressure vessels have been clad using the manual metal arc (MMA) process. This welding process creates rough and noisy inspection conditions that inhibit inspection effectiveness. The NRC has, therefore, requested the Pacific Northwest Laboratory (PNL) to evaluate the reliability and effectiveness of these techniques for inspecting U.S. vessels. (See Section L.2.) Results of tests show that light grinding of the clad surface (specifically improving the surface roughlass by a factor of 2, from 0.012 in. RMS to 0.006 in. RMS; improves the crack detectability confidence level from low to very high.

Further work is planned to refine the measurement methods for clad conditions, develop appropriate examination (and calibration) methods, determine crack detection probabilities for various inspection techniques, and to establish performance of techniques for crack sizing. Hence, the surface roughness and cladding noise under field conditions could be quantified, a criteria established for determining if the cladding conditions permit a valid inspection to be performed, and a procedure given for an effective inspection.

515

L-1

L.2 Influence of Improved NDE Techniques

PNL has developed estimates to predict the influence of improved vessel examination techniques on vessel failure and allowable RT_{NDT}.

Table L-1 summarizes the results of this investigation. Using "best estimates" on probability of flaw detection, we have attempted to provide bounds for adjustments in allowable RT_{NDT} to reflect the benefit of optimized vessel inspection techniques. Table L.1 shows that the probability of flaw detection using optimized techniques varies from 50 to 95%, depending on clad type and surface finish. The corresponding benefit from inspection expressed as an increase in allowable RT_{NDT} varies from 10 to 33°F. In addition, we have provided supporting material for fracture mechanics and NDE in Sections L.2.1 and L.2.2 that indicate methodology used to derive Table L-1.

L.2.1 Fracture Implications of Improved Inservice Inspection

The results of probabilistic fracture mechanics calculations were available to PNL from the work of Mr. J. Strosnider of NRC (see Appendix H). These results were used to estimate an allowable increase in RT_{NDT} which could be justified on the basis of the estimated probability of crack detection for inservice inspection (ISI).

Figure L-1 shows trends of the NRC results for failure probability as a function of RT_{NDT}. Results for the NRC cooling rate curves for parameters $\beta = 0.051$, 0.15, and 0.50 are shown along with other results for the temperature/pressure curves of the Rancho Seco transient. It was assumed that the range of interest was a failure rate of 10^{-4} given the occurrence of a transient. In Figure L-1, Po is the probability of failure at a given transient and RT_{NDT}, and P is the probability of failure for the same transient but increased value of RT_{NDT}. All calculations here were based on the $\beta = 0.15$ cooling rate parameter.

L-2

TABLE L-1

ESTIMATED DETECTABILITY OF UNDERCLAD CRACKS AND ESTIMATED INCREASES IN ALLOWABLE RT_{NDT}

CLAD	FINISH	FLAW DIRECTION WITH RESPECT TO CLAD	PROBABILITY OF DETECTION	FACTOR OF IMPROVEMENT (1) IN RELIABILITY	ALLOWABLE INCREASE IN RT _{NDT} , °F
Strip	Smooth	Perpendicular and Parallel	95%	20 to 40	27 to 33
Single Wire Strip	Smooth Unground	Perpendicular Perpendicular	85%. 0.5"-1.0" Flaw 90%, 1.0" or Greate Flaw	v 7.4 to 14.8 er	17 to 24
Single Wire Strip	Smooth Unground	Paralle1 Paralle1			
Manual	Ground	Perpendicular and Parallel	75%, 0.5"-1 .00" Fla	aw 4.3 to 8.6	13 to 19
Single Wire	Unground	Perpendicular and Parallel	80%, 1.0" or Greate Flaw	er	
Manual	Unground	Perpendicular and Parallel	50%, 0.5"-1.0" Flaw 75%, 1.0" or Greate Flaw	v 2.8 to 5.6 er	10 to 15

(1)Factor of Improvement = Probability of Failure without Inspection/Probability of Failure
with Inspection.

Lower bound assumes flaws are isolated and independent occurrences. Upper bound assumes possible occurrence of multiple flaws in a given weld.

TAB	LE	L-	·2
_			

ESTIMATE OF FAILURE	PROBABILITY	WITH AND	WITHOUT	INSERVICE	INSPECTION

				Failure F P(A)•P(F/A)	Probability P(A)·P _{ND}
A	P(A)	PND	P(F/A)	(without ISI)	(with ISI)
0.125	8.3 x 10-1	0.5	0	0	Ω
0.25	1.6×10^{-1}	0.05	1.5 x 10-4	2.4×10^{-5}	1.2×10^{-6}
0.50	4.2×10^{-3}	0.5	1.0×10^{-2}	4.2×10^{-5}	21×10^{-6}
1.0	4.1×10^{-4}	0.05	5.4×10^{-2}	2.2×10^{-5}	1.1×10^{-5}
1.5	1.3 x 10-4	0.05	5.6 x 10^{-2}	7.3 x 10-6	3.5×10^{-7}
2.0	4.2×10^{-5}	0.05	4.5×10^{-2}	1.9×10^{-6}	35×10^{-8}
2.5	1.3×10^{-5}	0.05	-	-	-
3.0	5.0 x 10- ⁶	0.05	-	-	-
3.5	3.3×10^{-6}	0.05	-	-	-
				Po(F) = 9.7	$\times 10^{-5} P(F) = 4.8 \times 10^{-6}$

Based on data from status report by Jack Strosnider on "Failure Probability of a RPV Subject to Pressurized Thermal Shock," March 5, 1982 nuces.

For "Rancho Seco Iransient Reference Case," mean copper = 0.34, mean RTN^{DT} = 0.0 and mean fluence $= 3.0 \times 10^{10}$ for smooth strip clad Probability of flaw nondetection (P_{ND}) for smooth strip clad (2)

ء : ارد م

(3)





٤

Ļ

In PNL's calculations, the decrease in vessel failure probability due to ISI was first estimated. A trade-off between this decrease with an offsetting increase in failure probability due to relaxation in RT_{NDT} requirements was then performed. Table L-2 illustrates the estimate of failure probability as a function of probability of nondetection of a flaw (P_{ND}). In Table L-2:

- P(A) = Probability of a flaw of depth A in the critical weld
- P(F/A) = Probability of failure for the Rancho Seco transient given the presence of a flaw of depth A
- P_{ND}(A) = Probability of not detecting a flaw of depth A based on PNL
 estimates
- $P(A) \cdot P(F/A) =$ Probability of failure without ISI given the occurrence of the Rancho Seco transient
- $P(A) \cdot P(F/A) \cdot P_{ND} = Probability of failure with ISI given the occurrence of the Rancho Seco transient$

Table L-2 used the best detection capability corresponding the more favorable conditions of PNL's flaw detection studies. Results for other inspection conditions are given in Table L-1. The ratio of failure probabilities in Table L-2 was 20:1 for the no ISI case versus the ISI case. Turning to Figure L-1, an increase in RT_{NDT} of 27°F will give an offsetting 20:1 factor in failure probability. Therefore, it is estimated for this particular example that the allowable RT_{NDT} can be increased by 20°F with no net increase in failure probability provided that inservice inspection is performed.

It is recognized that the flaw size distribution in the NRC probabilistic analyses is subject to considerable uncertainties. Therefore, the estimated flaw size distribution as modified by ISI is subject to the same uncertainties. However, the relative improvement in reliability due to ISI is believed to be significantly more accurate than the absolute values of failure probability.

L-6

The upper bound estimate of the allowable increases in RT_{NDT} is an attempt to consider the statistical nature of underclad cracks. Evidence suggests that one can expect either no cracks at all or a large number of cracks. Given a large number of cracks is indeed very small ($P_{ND} = 0.05^{10} = 10^{-13}$), thus, one can arrive at vastly different conclusions regarding the benefits of ISI, depending on the assumption on the stochastic structure of the flaw distribution. The upper bound estimate as shown in Table L-1 on the benefit of ISI conservatively assumes that half the flaws in vessels are random occurrences and that the remaining flaws occur in groups so to be readily detectable. The assumption that all flaws are random occurrences will tend to greatly underestimate the potential benefits of ISI. On the other hand, it is unreasonable to assume that random flaws will not occur, since one can be led to accept any level of embrittlement in a vessel provided that an ISI reveals no flaws.

L.3 Flaw Detectability Measurements

.

Flaw detectability experiments have been carried out on strip clad, single wire sub arc clad, and manual clad. Both ground and unground surfaces were evaluated. The test blocks used for this evaluation were: a 750-mm-dia. strip clad pressurizer dropout, two 600-mm square blocks with strip and single wire clad with one side ground and the other as welded*, two small blocks with ground and unground manual clad. The pressurizer dropout contained through clad notches as well as actual thermal fatigue underclad cracks. The two EPRI blocks contained unclad notches and the manual clad samples contained two reference reflectors for evaluation of general noise level. The measurements reported here were taken using a 2-MHz dual beam longitudinal (SEL) 70° transducer, with 10- by 15-mm elements and focal cross over point of 17 mm. This unit was considered optimum for the clad conditions and thicknesses (6 to 9 mm) terled. All measurements were performed manually.

The results of signal amplitudes compared to the signal amplitude of a 3 mm flat bottom reference reflector are shown in Table L-3. In addition, a blind test was conducted. This blind test used the pressurizer dropout sample that contained nine actual underclad cracks generated by a thermal fatigue process. The

^{*}Access to these two samples was made possible through J. R. Quinn, Electric Power Research Institute (EPRI), Palo Alto, CA.

cracks were oriented both parallel and perpendicular to the direction of the cladding. The cracks ranged in depth from 0.25 to 0.75 inch through the wall. Although none of the three operators had prior knowledge of crack location, each operator detected every crack. The probability of detection data reported in Table L-1 are estimates based on an optimized inspection system, our flaw amplitude measurement and our blind test.

.....

t

TABLE L-3

FLAW AMPLITUDE RESPONSE

SENSITIVITY STANDARD: 3MM FLAT BOTTOM REFERENCE REFLECTOR

SAMPLE TYPE	FLAW DEPTH RANGE	FLAW RESPONSE RANGE (+) GREATER REFERENCE REFLECTOR	
Ground; Strip Clad; Underclad Notch	5mm to 18mm	0 to +9dB	
Unground; Strip Clad; Underclad Notch	5mm to 18mm	0 to +8dB	
Ground, Single Wire; Underclad Notch	5mm to 18mm	-1 to +10dB	
Unground; Single Wire; Underclad Notch	5mm to 18mm	-1 to +12dB	
Ground; Strip Clad Pressuirzer Dropout Underclad Cracks	5mm to 18mm	0 to +11 dB	



Annealing of the beltline region of reactor vessels is a potential remedial measure for the PTS problem for vessels that have suffered considerable radiation embrittlement.

Time-Temperature Effects of Recovery of Properties

There is a fairly good experimental basis for choosing the annealing temperature and time. From the Naval Research Laboratory, research funded by the NRC has revealed the effects of annealing at 650°F and 750°F and the effects of reirradiation and reannealing (Ref. M.1-M.3). Research at Westinghouse funded by the Electric Power Research Institute (EPRI) has revealed the effects of annealing at temperatures of 650, 700, 750, 800 and 850°F (Ref. M.4). As expected, there is a clear trend toward better recovery of properties at the high temperatures and at longer times up to one week. In this discussion, therefore, we will assume that annealing would be done at 850°F for one week, and that the resulting recovery of fracture toughness properties would be about 80 percent.

Reirradiation Effects

With regard to the rate at which ΔRT_{NDT} increases upon reirradiation, the data are scattered and somewhat conflicting. The rate of reembrittlement should be as low as that just prior to annealing, and is almost certainly significantly lower than that at the start of life. Thus, a plant that annealed its vessel after, say 8 EFPY should **expect much** more than 8 additional EFPY before reaching the same ΔRT_{NDT} . Obviously, a better estimate of the reirradiation rate is desired for economic considerations before undertaking annealing; but for purposes of safe operation in later years, there will be additional information ifrom test reactor programs and from plant surveillance data. One technical question that has yet to be thoroughly investigated is the verification test program for a specific plant, which will be required to measure the effects of the annealing operation and the reirradiation.

Appendix G of 10 CFR 50 requires that the degree of recovery be measured "...by testing additional specimens that have been withdrawn from the surveillance program capsules and that have been annealed under the same time-temperature conditions as those given the beltline material."

The specimens in most capsules have been irradiated substantially more than the vessel; hence, measurement of ΔRT_{NDT} for those specimens after annealing should give a conservative estimate of the condition of the vessel. Their use as a guide to the rate of reembrittlement is not well understood. One alternative is to test "reconstituted" Charpy specimens from earlier surveillance capsules, i.e., fabricate Charpy specimens by welding ends on the broken halves of specimens that have lower fluences because they were withdrawn from the vessel early in life. Another alternative is to irradiate archive material to the desired fluence in test reactors and then check the effects of annealing and reirradiation.

With regard to the feasibility of annealing, NRC staff has the results of the EPRI study (Ref. M4) and the (potential) advice of vessel fabricators who have experience in post-weld heat treatment after field fabrication and after repairs. The EPRI study developed a means of heating by electric resistance elements supported on a frame that would be lowered into the vessel before the water is removed. No insurmountable difficulties were reported, but many engineering details remain to be resolved.

From the standpoint of risk, the main concern seems to be the potential for distortion of the vessel and the economic risks associated with problems in reinstallation of the core support structure and the closure head. At 850°F, some creep and relaxation could occur at regions where there are significant stresses caused by differential expansion during heatup and cooldown, by residual stresses, and by the stresses near the supports caused by the dead weight of the vessel. These problems have not been dealt with very carefully or completely as yet. From what has been done, it dos not appear that the piping would have to be separated from the vessel. Again, the experience of field fabricators of vessels must be tapped.

.

Other components that require study of the risks of annealing are the vessel insulation, the adjacent concrete and the supports. The movement of the vessel relative to the support when heated to 850°F will of course be greater than that at the design temperature of 650°F. Also, for those supports where the concrete is only a short distance below the vessel nozzle that must carry the load, the structural integrity of the concrete must not be impaired.

In conclusion, it appears that from the safety standpoint the benefits of annealing are quite clearcut and the risks are low. The risks of annealing are economic risks. There is, of course, a cost in man-rem and dollars if everything goes as planned. The largest uncertainty remains the economic and exposure risks associated with correction of distortion of the vessel or other damage if things do not go as planned.

M-3

REFERENCES

- M.1. J. R. Hawthorne, "Status of Knowledge of Radiation Embrittlement in USA Reactor Pressure Vessel Steels," NUREG/CR-2511, NRC Memo Rept. 4737.
- M.2. C. Z. Serpan, H. E. Watson, and J. R. Hawthorne, "Interaction of Neutron and Thermal Environmental Factors in the Embrittlement of Selected Structural Alloys for Advanced Reactor Applications," Nuclear Engineering and Design, V.II, 1970, pp. 368-380.
- M.3. C. Z. Serpan and J. R. Hawthorne, "Yankee Reactor Pressure Vessel Surveillance: Notch Ductility Performance of Vessel Steel and Maximum Service Fluence Determined from Exposure During Cases 2, 3, and 4. ASME Journal of Basic Engineering, V. 89, 1967, pp. 897-910.
- M.4. EPRI Thermal Annealing Program, Research Project 1021-1. Development of a Generic Procedure for Thermal Annealing on Embrittled Reactor Vessel Using a Dry Annealing Method. EP Interim Report July, 1982, T. R. Mager and R. D. Rishel, PN-2483. Only drafts of other parts of the final report were available as of August 15, 1982.

APPENDIX N

FUTURE CONFIRMATORY STUDIES

N.1 Introduction

\$

The following issues relating to pressurized thermal shock require confirmatory study.

- 1. Applicability of Linear Elastic Fracture Mechanics (LEFM) for initiation, propagation and arrest for reactor pressure vessels subjected to a pressurized thermal shock scenario.
- 2. Effectiveness of Warm Prestress
- 3. Vessel failure under nonpressurized thermal shock conditions.
- 4. Behavior of small finite flaw when subject to PTS conditions.
- 5. Cladding-flaw interaction; bimetallic effects.
- 6. Irradiated cladding material and fracture properties.
- 7. Arrest on the upper shelf.
- 8. Postarrest performance for a deep crack in upper shelf material toughness.
- 9. Definition of margin when using RT_{NDT} to set fracture toughness curves.
- 10. Variation of through-wall fracture toughness degradation.
- 11. Validation of fracture toughness degradation as a function of fluence for ferritic welds.
- 12. Effect of trace elements (copper, nickel, phosphorus) on the embrittlement rate of RPV steels at reactor operating conditions.

- 13. Effectiveness of thermal annealing on fracture toughness recovery and reembrittlement rate.
- 14. Establishments of criteria and standards to be applied to any proposed, in situ thermal annealing of operating reactor vessels.

N.2 Summary of Prior Studies

Thick section pressure vessel materials have been characterized to form the basis for fracture toughness and crack growth data in the ASME B&P codes. Crack arrest methodology has been extensively evaluated and preliminary specimen designs developed. Methods of elastic-plastic fracture analysis have been developed and evaluated. Irradiation effects on pressure vessel plate, forging and weldments, including low-shelf thoughness weldments, have been studied using compact specimens up to 4 inches thick. Thirteen intermediate tests have been performed on nine vessels to validate methods of fracture-failure analyses, to demonstrate the capability of NDE methods and repair procedures in thick sections. Seven thermal shock (unpressurized) experiments have been performed on thick-section cylinders to demonstrate the applicability of LEFM in predictions of flaw behavior and to establish the applicability of small specimen toughness determinations in fracture analysis. Unique crack arrest data have been developed in these tests. Small scale stainless steel cladding tests have been performed to determine the influence of cladding on flaw development. Computer codes have been developed to evaluate fracture potential to define and quantify the principal variables that need to be considered in operating systems. The effect of trace elements, such as copper, nickel and phosphorus, on the embrittlement potential of commonly used reactor pressure vessel steels when subject to different levels of neutron bombardment has been determined. The effect of thermal annealing, at various temperatures on the fracture toughness recovery of neutron embrittlement steels has been defined and quantified. Elastic-plastic material fracture toughness testing procedures have been developed and elastic-plastic fracture data basis are being developed for unirradiated and irradiated reactor pressure vessel steels. Extensive participation with NRR, code writing bodies (ASME, ASTM), information dissemination through formal and informal exchanges, and international cooperative efforts have been maintained.

N-2

N.3 Present Programs Addressing Issues

ŧ

The following programs are underway or are planned to address the issues identified in Section N.1. The numbers in parentheses refer to issues in Section N-1.

- Complete 3 dimensional finite element fracture computer codes [ORFLAW-3D and ORVIRT-3D] (4)(5)(6)
 completion date: March 1983
- Complete evaluation of finite flaw behavior (4)
 -completion date: December 1982
- Complete development of unified LEFM-EPFM methodology considering all regimes of toughness (1)(2)(4)(5)(7)(8)
 -completion date: September 1983
- Complete testing of low-shelf weldments (8)(11)
 -completion date: December 1983
- Complete testing ITCT irradiated specimens of present practice steel (11)
 -completion date: September 1983
- Complete irradiation of cladding material (6)
 -completion date: June 1983
- Complete testing of irradiated cladding material (6)
 -completion date: December 1984
- Complete material procurement for K_{IC} (4T) study (7)(9)(11)
 -completion date: December 1983
- Complete irradiation for K_{IC} (4T) study (7) (9)(11)
 -completion date: June 1985

- Complete testing for K_{IC} (4T) study (7)(9)(11) -completion date: September 1985*
- Complete development of irradiated crack ar. ,t data base (6)(7) (9)(10)(11)
 -completion date: September 1986
- Complete probabilistic fracture mechanics version of Computer Code OCA-2
 (9)
 -completion date: September 1982
- Complete Thermal Shock Experiment TSE-7 (1)(3)(4)
 -completion date: March 1983
- Complete Thermal Shock Experiment TSE-8 (1)(3)(4)(5)
 -completion date: March 1984
- Complete Thermal Shock Experiment TSE-9 (1)(3)(4)(5)
 -completion date: March 1985
- Complete feasibility study and system design for Pressurized Thermal Shock Experiments (1)(2)(4)(5)(7)(9)
 -completion date: September 1982
- Complete PTSE facility construction checkout (1)(2)(4)(5)(7)(9)
 -completion date: April 1983
- o Complete PSTE-1 (1)(2)(4)(5)(7)(9)
 -completion date: March 1983

^{*}Interim data from testing program will be available at earlier dates in 1984 and 1985.

o Complete PSTE-2 (1)(2)(4)(5)(7)(9)
-completion date: March 1984

- Complete development of crack arrest specimen and test procedures (1)(7)
 -completion date: March 1983
- Complete construction of capsules and begin irradiation of specimens in dose rate study. (11)(12)(13)
 -completion date: October 1982
- Complete dose rate study [show irradiation more closely simulating operating reactor experience] (11)(12)(13)
 -completion date: October 1985
- Complete testing of SSC-2 and PSF dosimetry specimens (10(11)(12)
 -completion date: March 1983
- Complete variable radiation sensitivity study (11)(12)
 -completion date: May 1983
- Complete high temperature (454°C) annealing study (13)(14)
 -completion date: March 1985
- Complete high temperature (454°C) annealing
 -reembrittlement rate study (12)(13)(14)
 -completion date: May 1986
- Complete study on the effectiveness of drop weight method of determining
 NDT and applicability of RT_{NDT} (9)
 -completion date: October 1984
- Complete program on System Requirements and Standards development for annealing of reactor pressure vessels (12)(13)(14)
 -completion date: October 1984

Completion of testing irradiated material from KRB Block A pressure vessel wall (10)(11)(13)(14)
 -completion date: October 1983

N.4 Applicability of Research

The research program is integrated with the needs of NRC licensing in addressing the issue of pressure vessel integrity, both under normal and accident or upset operating conditions. Every element of the described program is based upon the need of NRR to define and quantify methods use for evaluating pressure vessel safety issues. Every element of the described program is reviewed frequently by NRR, from the U.S. industry, and American and International technical community for its appropriateness and applicability to known or anticipated safety issues. The timeliness of this ongoing research is such that approximately 70 percent of the issues to be resolved in PTS will be addressed and results obtained by the research effort within Fiscal Year 1983. The remaining 30 percent of the intially needed information should be available as follows: 20 percent in FY 1984, 5 percent in FY 1985, and 5 percent in FY 1986. The planned funding effort is as follows:

FY 1982	FY 1983	<u>FY 1984</u>	<u>FY 1985</u>	FY 1986
\$6,650K	\$6,850K	∿\$6,000K	~ \$6,0 00K	∿\$6,000 K

It should be noted that though most of the initial data will be developed as described above, a considerable confirmatory effort must be continued during the years 1983-1986 to ensure that the results obtained are statistically valid. Another reason for the extension of the program through FY 1986 is the time required to carry out an effective irradiation study.

The funding shown above is committed to four contracts through FY 1983 and thereafter to three contracts.

- 1. HSST program (ORNL)
- 2. Pressure Boundary Integrity for Water Reactor (ENSA)
- 3. Pressure Vessel Simulation (ORNL)
- 4. Systems Requirements for Annealing (EG&G/INEL, terminates FY 1983)

N-6
N.5 Confirmatory Studies on Fluence Trend Curves

ŧ

N.5.1 Refinement of Chemistry and Fluence Factors

Immediate steps must be taken to scrub down the PWR surveillance data base and add data from BWR surveillance. Then a reanalysis is required to refine the copper and nickel terms and determine what the exponent on fluence should be, and whether it should be constant over the whole fluence range. Probably, test reactor data should be omitted until later when a time-temperature parameter is better understood. This represents a change in attitude from that on which Regulatory Guide 1.99 and the MPC trend curves are based. The change reflects the increasing number of surveillance reports in recent years, more than it reflects any increased suspicion that test reactor data and surveillance data are separate populations. There is now an EPRI data base in which the Charpy curves have been fitted by a hyperbolic tangent function and new values of Charpy shift calculated. These values must be compared with the existing data base, which was obtained from curves drawn by eye, and differences reconciled where possible. After these steps are taken; and the new regression analysis is performed, the results will be incorporated in Revision 2 of Regulatory Guide 1.99.

N.5.2 Long-Range Effort

There are two refinements that require further input from research efforts before incorporation in further revisions of Regulatory Guide 1.99. One is the change from fluence measured in terms of neutrons/square centimeter, (E > 1 MeV) to fluence measured in terms of a damage function that considers the effects of different energy spectra, probably displacements per atom (dpa). The other refinement to be expected is a time-temperature parameter that accounts for irradiation temperature and exposure time. Both refinements are needed to permit the inclusion of test reactor and surveillance data in the same data base with complete confidence that they belong in the same population.

N.6 <u>Study of PTS Event Scenarios</u>

N.6.1 Objective and Scope

The objective of this study is to provide an independent probabilistic analysis of PTS at a representative B&W, CE, and \underline{W} PWR. The results will estimate the likelihood of vessel cracking due to PTS, identify what is important (dominant sequences, important operation actions, etc.) and will identify major uncertainties. The results will also provide a comparison of the risk-reduction effect#weness of alternative corrective actions.

414

The scope of the study is limited to addressing the reliability of pressure vessel integrity and does not address the consequences of vessel failure. The study of the three plants, Oconee 1, Calvert Cliffs 1, and H. B. Robinson 2, will be plant specific. Extension of this study to a generic analysis of classes of plants is beyond the scope of this study.

The study will support resolution of USI A-49 in four ways:

- (1) Confirm understanding of PTS; e.g., how likely is vessel failure? What are the important event sequences, operator actions, and control features? How effective are various proposed measures for reducing the likelihood of vessel failure?
- (2) Improve methods for analyzing PTS.
- (3) Provide a plant-specific analysis of PTS for three plants.
- (4) Provide an improved basis for staff evaluation of plant-specific analyses.

N.6.2 Study Plan

The study will use a functional approach rather than a detailed component-bycomponent approach. Conceptually the plan is first to identify phenomena that

N-8

could cause overcooling such as too much feedwater or too much ECC; second, to identify initiating events; and then to analyze the reliability of functions that prevent overcooling.

. . .

The study involves the following steps for each of the three plants.

Ŧ

First, the analysts (ORNL for probabililistic and fracture-mechanics analysis and LANL/INEL for thermal hydraulic analysis) obtain information on the plant and understand how the plant operates regarding overcooling transients.

Then ORNL performs an event-tree analysis to systematically delineate event sequences that could lead to overcooling and estimates the frequency of occurrences of these sequences.

About a dozen of these sequences are sclected for detailed analysis by LANL using TRAC or by INEL using RELAP-5 to calculate temperature and pressure in the downcomer during the transient. These dozen cases are selected to cover a range of severity. Initially, in the Oconee study, both TRAC and RELAP-5 are used to compare and help check out the codes. Subsequently TRAC will be used to analyze Calvert Cliffs, and RELAP-5 will be used to analyze H. B. Robinson.

The method for assessing these TRAC and RELAP-5 models of specific plants (including secondary and control systems) is still being developed. Tentative plans are to calculate plant behavior during a transient such as a turbine trip and compare the results with plant data regarding behavior of turbinebypass valves, feedwater flow, steam generator levels, reactor coolant temperatures, etc. The intent is to verify that the code behaves reasonably in transients of interest.

For each transient the coolant temperature and pressure calculated by TRAC or RELAP-5 will be used in a fracture mechanics calculation of the conditional probability of vessel failure given that transient occurs. Based on these results, ORNL will estimate the consequence to vessel integrity for each of the transient sequences in the event trees. Each of the sequences will then be sorted into one of a half dozen or so damage bins. These bins will be

N-9

identified in terms of how many years the plant could operate before the transients in that bin could crack in the vessel. Bins, for example, would be 0-5 yrs., 6-10 yrs., etc. The likelihood of vessel cracking will be added up for all the sequences in each bin to obtain the frequency of vessel-cracking vs. effective-full-power years. Dominant sequences will be apparent in the results.

N.6.3 Status and Schedule

Following a preliminary survey of available information in the Summer of 1981, the Oconee probabilistic study started in FY 1982. The analysis is scheduled to be completed in February 1983, and the draft report in March 1983.

In July 1982, the owners of Calvert Cliffs and H. B. Robinson agreed to participate in the study. This Calvert Cliffs analysis began in August 1982 and should be completed in October 1983 with a draft report completed in January 1984. The schedule for the H. B. Robinson study has not yet been established.

N.7 Reactor Vessel Annealing Study

N.7.1 Status of NRC Annealing Program

In a letter from Harold R. Denton to Robert B. Minogue dated June 15, 1982, NRR requested RES to perform a study to determine the feasibility of conducting a demonstration of in situ annealing of an irradiation-embrittled reactor pressure vessel to restore the fracture toughness properties. The teasibility study would consider potential candidate vessels, the value and generic applicability of information to be gained from a demonstration, and the costeffectiveness of such a program.

RES engaged EG&G Idaho, INEL, to conduct the study. A survey of the reactor vendors, architect engineering firms, and consultants was conducted by EG&G Idaho during the past two months. The tentative conclusion from the survey is given in the draft interim report EGG-FM-6083, October 1982, "Evaluation of a Reactor Pressure Vessel Anneal Demonstration." The consensus view is that in situ annealing of an irradiation-embrittled reactor vessel is practical.

However, a demonstration anneal is advisable only if the vessel selected possesses the appropriate geometry, materials of construction, and damage level of concern. The review has indicated that Humboldt Bay, Indian Point 1, Shippingport, and BR-3 vessels were not ideal for the demonstration, and therefore a final recommendation has not been made.

The work is continuing to meet the objective of the NRR request.

Ŧ

APPENDIX O

SUMMARY OF ORNL FRACTURE-MECHANICS ANALYSIS FOR SEVERAL PWR RECORDED OCA TRANSIENTS*

Fracture-mechanics calculations were made recently for several PWR overcoooling accidents (OCAs) that have occurred since 1970, including the 1978 Rancho Seco transient (Ref. 0-1). Information pertaining to these transients is presented in Table 0-1 and Figures 0-1 to 0-6.

		<u>Vessel Di</u>	mensions (in.)	RTNDTo (°F)		
Plant	Date of Accident	Inner radius	Wall thickness	Cir. weld	Long. weld	
H. B. Robinson H. B. Robinson H. B. Robinson	4/28/70 11/5/72 5/1/75	78	9.31	- 20	-20	
Rancho Seco	3/25/78	86	8.5	b	+60	
TMI-2	3/28/79	86	8.5	b	+20	
Ginna	1/25/82	66	6.5	+20	С	

TABLE 0-1 PWR OCA DATAª

^aData obtained from Nuclear Reactor Regulation, NRC, 6/16/82.

^bData not available.

^CForged vessel (no longitudinal welds).

Figures 0-1 to 0-6 describe the primary-system-pressure transient and the coolant-temperature transient in the cold leg upstream of the point where the emergency core coolant (ECC) is injected. Because of the location of the temperature measurement, the recorded temperatures are not necessarily accurate indications of the coolant temperatures in the downcomer. For instance, the injection of ECC would result in a lower temperature, and recirculation of

^{*}Contribution by R. D. Cheverton, D. G. Bolls, and S. K. Iskandera of ORNL.

core coolant through the vent valves in a B&W plant would result in higher temperatures. However, the fracture-mechanics calcuations have been made using the recorded temperatures in Figures 0-1 to 0-6 as downcomer temperatures. The curves in Figures 0-1 to 0-6 were digitized for input purposes, using enough time steps to describe the curves accurately; essentially no smoothing of the curves was necessary. Thus, the analysis reflects the effect of nearly all of the irregularities in the curves, except perhaps for the pressure curve in Figure 0-2. For this case it appears that the pressure dropped below 1700 psi but was not recorded. In the calculation, it was assumed for this particular case that the minimum pressure was 1700 psi.

The fracture-mechanics calculations were performed using OCA-II and the basic input data shown in Table 0-2.

In the process of making the fracture-mechanics calculations, a search was made for threshold values of the nil-ductility reference temperature at the inner surface (RT_{NDTS}) corresponding to incipient initiation (II). Results of the analysis are presented in the form of sets of critical-crack-depth curves for the threshold conditions (Figs. 0-7 to 0-18). A summary of the data is shown in Table 0-3.

In Figures 0-7 to 0-18, the existence of minimum points in the constant K_I curves indicates that the requisite conditions for warm prestressing (WPS) exist (dK_j/dt < 0). However, the existence of more than one minimum would indicate that K_I fluctuated with time, and under these circumstaces it is not clear that WPS would actually be effective. It was ignored, therefore.

0-2

Parameter	Value					
Vessel dimensions ^a	See Table 1					
Cladding thickness, a/w	0.025					
Flaw Type	Long axial and continuous circumferential on inner surface and extending through cladding					
Range of flaw depths included in analysis, a/w	0.01-0.95					
Limits imposed on critical crack depths, a/w	0.025-0.15					
K _{IC} and K _{Ia}	ASME Section X	I				
(K _{Ia)} max, ksi √in.	200					
$\Delta RT_{NDT} = f(Cu, Ni, F)$	αF°·27					
Fast neutron fluence (F)	$F = F_0 \exp(-0.24a \text{ in.}^{-1})$					
$\Delta RT_{\rm MDT}$ (a) ^b	= $\Delta RTNDT_c^c e^-o$.	0 ⁶⁵ a in. ⁻¹				
ART _{NDTs} , °F	<u>≤</u> 500					
Fluid-film heat transfer coefficient (h _f), Btu/hr·ft ² ·°F	300 ^d					
		Cladding	Base Material			
Thermal conductivity (k), Btu/hr·ft·°F		10	24			
Thermal coefficient of expans (α) , $^{\circ}F^{-1}$	ion	10×10^{-6}	8.04×10^{-6}			
Modulus of elasticity (E) lbs/in. ²		28×10^6	28×10^{6}			
Specific heat (c _p), Btu/1b.ºF		0.12	0.12			
Density (p), lbs/ft ³		489	489			
Poisson's ratio		0.3	0.3			

Table U-2. Input Data for ULA Ana	alvses	5
-----------------------------------	--------	---

...

 $a_{\rm Sets}$ of K^{*} values were calculated for each set of dimensions. $b_{\Delta RTNDT}$ at the tip of the flaw.

 $^{C}\Delta RTNDT$ at the inner surface of the vessel. $^{d}Corresponds$ to main circulating pumps off.

Tr	ansien	t	Weld ^a	RTNDT _s , ^b °F	a _c ,d in	
Robinson Robinson Robinson Robinson Robinson Robinson Robinson Rancho Seco TMI-2 TMI-2	1970 1970 1972 1972 1975 1975 1975 1975 1975 1978 1979 1979	broken loop broken loop loop C loop C loop B loop B loop B	L C L C L C L C L L L	321 (F) 351 (A) 381 (F) >480 354 (F) 372 (A) 395 (F) 440 (A) 295 (F) 209 (F) 225 (F)	0.93 0.93 1.4 1.4 0.93 1.4 1.2 1.3 1.3 1.3 1.3	
Ginna	1982	loop B	C	378 (A)	0.91	

TABLE 0-3 Results of OCA Analyses

^aL and C refer to longitudinal and circumferential.

^bF and A in parentheses refer to failure and arrest.

 $c_{Small increase in RTDNT_{S}}$ would result in failure. $d_{Critical crack depth.}$



47 L



42.5

<u>.</u>..



)

+74





47 L



121

٠,



FIGURE 0-7

11.44



s17 5.....

5

1.4

 .

-t /



\$ 1 7 .



(

1.1.1

֥.



and a second second



معري کې ۱۰ مېغو درواد د اردې او به مېلورو د ولکې د ماکنورو مېلو د د. د مرو کې ۱۰ مېغو درواد د اردې موله مېلورو د د ولکې د مېلو ورو د و د ولکې د ولکې د د د ولکې د د د د د د د د د د



were with the base of the second second and and and and and a second second

F.IGURE 0-12

. .



، وهوا در فرها المالية المالية

7. T.



FIGURE 0-14

• • • •

FIGURE 0-15

ູ



्राः



.



. .

1

u ", 1





431

:

24

FIGURE 0-17



;



4.54

•--• ;

REFERENCES

-

-14 0

.

¢. •

0.1 R. D. Cheverton et al., "Thermal-Shock Investigations," <u>Heavy Section Steel</u> <u>Technology Program Quart. Prog. Rep. for July-September 1981</u>, ORNL/TM-8145, pp. 69-86.

APPENDIX P CALCULATED RT_{NDT} VALUES FOR PLANTS

P.1 RT_{NDT} Screening Values for All Plants

Table P.1 contains the results of the calculations described in Appendix E for 40 operating PWR plants comprising all of those having significant radiation damage plus all others for which information was readily available. As of June 28, 1982, there were 7 recently-licensed PWRs omitted.

In the column headed "Recommended RT_{NDT} Value for Screening" separate values are given for circumferential and axial welds, because the stress intensity factors produced by certain transients are different for the two cases. For many transients for which pressure is high, the critical value of RT_{NDT} is at least 30 degrees higher for circumferential cracks. Plants are listed in descending order of RT_{NDT} , taking that difference into consideration. For plants where the plate or forgoing governs, its RT_{NDT} value is listed in both columns. Repeating from Appendix E, the recommended RT_{NDT} is the sum of the mean initial RT_{NDT} , the mean ΔRT_{NDT} at the inner vessel wall and the "2-sigma" term.

The sources of information from the various plants are as follows. The EFPY ae calculated from data submitted monthly to the NRC for total megawatt hours thermal. This value is divided by the rated thermal power to get effective full power hours. For fluence, copper, and nickel content, the 8 plants that had been submitted reports containing the results of a recent review of all available data. These 8 plants can be identified in the Table by their values for "Licensee's RT_{NDT}." Most of the other plants had submitted detailed information on their vessel beltline materials and fluence at the critical locations in response to an inquiry from the NRC in May 1977. Finally, there were surveillance reports for a number of plants, which contained updated calculations of fluence at the vessel wall.

P.2 Comparison of NRC and Licensee's Values

For the 8 plants, Table P.1 shoes licensee's values of RT_{NDT} . For the three CE plants, Table P.1 also shows the values calculated by CE in Appendices to CEN-189 (Ref. E.2.). These CE values range from 28 to 39 degrees F above the licensee's values, largely because of differences in the estimates of initial RT_{NDT} . For the CE plants, the NRC value of RT_{NDT} fall, between the licensee's value and the CE value for Fort Calhoun, falls 18 degrees F above the CE value for Maine Yankee, and Falls far below the CE value for Calvert Cliffs 1, because we recently accepted a lower estimate of copper content. For the three Westinghouse plants, the NRC value of RT_{NDT} is 9 degrees lower for Robinson 2, and 12 degrees lower for San Onofre. For Turkey Point 4, the NRC value is 48 degrees higher, because the licensee used a surveillance value that happens to fall well below the Guthrie mean trend curve. For the B&W plants, the NRC value is 48 degrees higher, because the licensee mean trend curve. For the B&W plants, the NRC value of RT_{NDT} gives a higher value than the trend curve from Regulatory Guide 1.99, which B&W used. For Three Mile Island 1, the NRC value of RT_{NDT} was 59 degrees higher than the licensee's value because: (a) they used an initial value of RT_{NDT}of -14°F whereas the NRC used a mean value of 0°F and 2 sigma of 34 degrees F, and (b) they used Regulatory Guide 1.99 as described above for Oconee 1. Actually, B&W did not give the values quoted in the Table. Those values were calculated by the NRC, using copper and fluence values from proprietary references given by B&W. These differences will have to be resolve for those plants that fail the screening criterion.

11/13/82

P-2

ę

Plant NSSS/Vessel	EFPY as of	Fluence n/cm ²	Copper %	Nickel X	Mean Initial	Mean ART	2108+02	RT _{NDT} , °F,	, as	Licensee's
Fabricators	12/31/81	×10 ¹⁸			RTNDT, "F	of NOT	(5)	eircus. 31,	AXATIO)	RENDIT
Robinson 2 W/CE	7.10	(14.1)(3)(8) 14.8 (3)(8)	(0.35) 0.27	(1.20) 0.20	(-56) -56	(303)(4) 151	34 (4) 59	281	154	290 220
Turkey Point 4 W/B&W	5.67 No Axia	9.1 (9) 9 Welds	(0.32)	(0.57)	(0)	(200)	59	259		211
Turkey Point 3 W/B&W	5.67 No Axia	(9.1)(9) Welds	(0.32)	(0.57)	(0)	(200)	59	259		
Fort Calhoun CE/CE	5.07	(7.04) 5.1	(0.35) 0.35	0.99 0.99	(-56) -56	(2.64)(4) 248 (4)	34(4) 34 (4)	242	226	(7) 209 (239)
Indian Point 3 W/CE	2.98 Plate Gov	(1.67) erns	(0.2 4) 0.24	(0.52) 0.52	(+74) +74	(90) 90	48 48	212	212	
Yankee Rowe W/B&W/B&W	14.56 Plate Gov	(11.35) erns	(0.24) 0.20	(0.52) 0.63	(+74) +30	(90) 133	48 48	212		
Rancho Seco B&W/B&W	3.54	(2.33) 2.05	(0.31) 0.35	(0.59) 0.59	(0) 0	(135) 148	59 59	194	207	
Three Mile Island 1 B&W/B&W	3.52	(1.87) (1.87)	(0.31) 0.35	(0.59) 0.60	(0) 0	(133) 145	59 59	192	204	(129) 145
Oconee 2	4.71	(2.87)	(0.35)	(0.71)	(0)	(172)	59	231		

.

•

Table P.1 RT_{NDT} Values for All Plants⁽¹⁾ Calculated Per the Recommendations of the Working Group on RT_{NDT} ⁽²⁾ for the Vessel Inside Surface.

.

٠

See footnote(s), last page of table

These values are subject to change when plant-specific analyses yield better information.

Plant NSSS/Vessel Fabricators	EFPY as of	Fluence n/cm ²	Copper X	Nickel X	Mean Initial	Mean ART _{NOT}	2√03+02 ∆	RT _{NDT} , °F, Efress, ³¹ ,	as 1981(6) Axia(6)	Licensee's BINDT
	12/31/81	x10 ¹⁸			RTNDT . "F	oF UDI	(5)			
Point Beach 1 W/B&W	8.07	(10.01) 7.34	(0.24) 0.24	(0.57) 0.57	(0) 0	(151) 139	59 59	210	198	-
Oconee 1 B&W/B&W	5.04	(2.32) 2.73	(0.26) 0.31	(0.61) 0.55	(0) 0	113 138	59 59	172	197	160
Indian Point 2 ₩/CE	4.40	No Circum Data 2.2	0.34	1.2	-56	211 (4)	34		189	
Arkansas ANO-1 B&W/B&W	4.42	(2.70) 1.99	(0.31) 0.31	(0.59) 0.59	(0) C	129 129	59 59	199	188	
Point Beach 2 <u>W</u> /B&W, CE	7,54	(9.35) No Axial Welds	(0.25)	(0.59)	(0)	(156)	59	215		
Ginna W/B&W	8.18	(9.49) No Axial Welds	(0.25)	(0.56)	(0)	(154)	59	213		
San Onofre W/CE	9.04	(33.45) 27.12	(0.27) 0.27	(0.20) 0.20	(-56) -56	(188) 178	59 59	191	181	203
Zion 2 B&W/B&W	4.49	(2.83) 0.90	(0.26) 0.35	(0.61) 0.59	(0) 0	(119) 118	59 59	178	177	
Palisades CE/CE	4.12	(4.78) 4.78	(0.25) 0.25	(1.2) 1.2	(-56) -56	(174) 174	59 59	177	177	
Crystal River 3 B&W/B&W	2.48	(1.44) 1.36	(0.35) 0.31	(0.59) 0.61	(0) 0	(134) 118	59 59	193	177	

Table P-1 (Continued)

Plant NSSS/Vessel Fabricators	EFPY as of	Fluence n/cm ²	Copper A X X	Nickel %	Mean Initial RTNDT ^{,°F}	Mean al ART _{NOT}	2√0 ² +0 ²	RT _{NDT} , ^o F,	as 1981(6) AxTa	Licensee's 8TNDT'
	12/31/81	×10 ¹⁸				•F	(5)	Circon. Jr.		
Surry 1 W/B&W	4.88	(7.61) 1.66	(0. 25) 0.21	(0.51) 0.59	(0) 0	(141) 81	59 59	200	140	
Cook 1 W/CE	4.58	(2.87) 1.55	(0.40) 0.13	(0.82) 0.99	(-56) -56	(222) (4) 58	34 59	200	61	
North Anna 1 W/RD	2.41	(4.42) No Axial Welds	(0.14)	(0.80) Forging	(+38) Governs	(76) 48	48	162	162	
Beaver Valley ₩/CE	1.87	(3.16) 0.47	(0.37) 0.36	(0.62) 0.62	(-56) -56	(179) 104	59 59	182	107	
North Anna 2 W/RD	0.77	(1.38) No Axial Welds	(0.83)	(+56) Forging	(52) Governs	48		152	152	
Salem 1 W/CE	2.26	(1.49)	(0. 24) 0.24	(0.51) 0.51	(+51) Plate Governs	(87) 87	48 48	150	150	
Oconee 3 B&W/B&W	4.78	(2.92) No Axial Welds	(0.24)	(0.63)	(0)	(112)	59	(171)		
Surry 2 W/B&W, RD	4.83	(7.54) 1.64	(0.19) 0.21	(0.56) 0.59	(0) 0	(108) 81	59 59	167	140	
Calvert Cliffs 1 CE/CE	4.65	(6.84) 6.84	(0.30) 0.21	(0.18) 0.85	(-56) -56	(135) 136	59 59	138 .	139	(7)
St. Lucie CE/CE	3.52	(2.22) 2.22	(0.31) 0.30	(0.11) 0.64	(-56) -56	(98) 132	59 59	101	135	

.

Table P-1 (Continued)

.

.

Plant EFPY NSSS/Vessel as of Fabricators 12/3	EFPY	Fluence	Copper Nicke X X	Nickel T	l Mean Initial RTNDT' ^{°F}	Mean ART _{NDT} °F	2√02+02 2	RT _{NDT} , ^o F, as	Licensee's
	12/31/81	×10 ¹⁸		~			(5)	efregs. 31, 1981(6)	BINDT
Calvert Cliffs 2 CE/CE	3.63	(5.34)	(0.30) 0.30	(0.18) 0.18	(-56) -56	(127) 127	59 59	130 130	
Trojan W CBI	3.00	(2.07)	(0.16)	(0.62)	(+10) Plate	(65)	48 48	123 123	
Davis Besse 1 B&W/B&W	1.68	(1.11) No Axial Welds	(0.24)	(0.61)	(0)	(85)	59	144	
Haddam Neck <u>W</u> /CE	10.92	(14.30) 11.90	(0.22) 0.22	(0.10) 0.10	(-56) -56	(111) 106	59 59	114	
Kewaunee W/CE	5.87	(7.86) No Axial Welds	(0.20)	(0.77)	(-56)	(129)	59	132	
Farley 1 W/CE	2.19	(3.70) 0.83	(0.24) 0.27	(0.60) 0.60	(-56) -56	(117) 89	59 59	120 92	
Millstone 2	3.91	(2.19) No Data for Axial	(0.37) Welds	(0.60)	(-56)	(117)	59	117	
Prairie Island 2 W/SFAC	5.62	(7.53) No Axial Welds	(0.19)	(0.13)	(-56)	(81)	59	84	
Prairie Island 1 W/SFAC	5.90	(7.90) No Axial Welds	(0.14)	(0.17)	(-56)	(60)	59	63	

Table P-1 (Continued)

Footnotes

- (1) Arranged in descending order of the RT as of December 31, 1981 considering circumferential to be 30°F less severe than axial orientations.
- (2) Memorandum, M. Vagins to S. Hanauer, August 30, 1982.
- (3) raises shown in parentheses on top line are for circumferential welds, bottom line is for axial welds. When plate governs--both lines.
- (4) Determine by Reg. Gude 1.99, Rev. 1, Upper Limit Line, σ_{Δ} = 0.
- (5) σ_0 (17°F) and σ_{δ} (24°F) are the standard deviations of the initial RT_{NDT} and ΔRT_{NDT} , respectively. If palte or forging governed, acutal initial RT_{NDT} wavelable and $\sigma_0 = 0$
- (6) The sum of the Mean Initial RT_{NDT}, the mean WRT_{NDT} and $2\sqrt{\sigma_0^2 + \sigma_\Delta^2}$, as of Dec. 31, 1981.
- (7) Initial RT NDT assumed by licensee to be -50°F and by CE to be -20°F. Values in parentheses are by CE.
- (8) Fluence is per letter from CP&L Co., Sept. 24, 1982, pending agreement on final value.
- (9) Fluence reduced from 11.16 n/cm² per letter from FPL Aug. 31, 1982, on TP 4. TP 3 tentatively assumed to be the same as TP 4.
- (10) Fluence reduced to 0.73 x peak per letter from Omaha PPD, Sept. 1, 1982.

SSINS NO.: 6835 IN 82-42

UNITED STATES NUCLEAR REGULATORY COMMISSION OFFICE OF INSPECTION AND ENFORCEMENT WASHINGTON, D.C. 20555

November 5, 1982

IE INFORMATION NOTICE NO. 82-42: DEFECTS OBSERVED IN PANASONIC MODEL 801 AND MODEL 802 THERMOLUMINESCENT DOSIMETERS

Addressees:

All NRC licensees

Description of Circumstances:

Many licensees use thermoluminescent dosimeters (TLDs) for personnel radiation monitoring. One of the major suppliers of these TLD systems is the Panasonic Industrial Company. It has come to the attuntion of the NRC that Panasonic Model 801 and Model 802 dosimeters have exhibited a defect that could reduce the response of affected elements. Panasonic has sent a letter to its customers describing the problem and providing actions to be taken in case the defects are found in the customers' dosimeters. A copy of this letter is attached. The purpose of this Information Notice is to inform NRC licensees of the defect and its significance and how to respond if their dosimeters are affected.

Discussion:

In September 1982, during an interlaboratory calibration check, the NRC noted a decrease in the response of a number of the dosimeters which it uses in its TLD Direct Radiation Monitoring Network. Inspection of the dosimeters indicated that the affected elements had bubbles on the carbon-coated polyamide backing material and the aluminum substrate. The bubbles affect the heat transfer and thus the calibration. When these defects were discussed with the manufacturer. NRC staff learned that four other customers had also observed the same defect in their dosimaters. The precise cause of the defect is not yet known with certainty, but the following characteristics have been observed thus far:

- 1. Affected dosimeters apparently lose sensitivity to a variable degree, thus underestimating radiation exposure. The NRC has observed response reductions as high as a factor of two and other users have reported even greater reductions.
- The effect appears to occur randomly. Though two dosimeters may be handled 2. identically and have the same production date, one may produce a bubble in one or more elements and the other may not have any.
- 3. The bubbles usually appear on the carbon-coated polyamide backing material but have also been seen on the aluminum substrate. Changes in element calibration factors may also indicate the presence of a defect; however, bubbles have been observed that have not significantly affected element calibration factors. The bubbles are large and easily seen.
IN 82-42 November 5, 1982 Page 2 of 2

4. The effect has been seen on both lithium borate and calcium sulfate elements, but more frequently on the latter.

Although Panasonic believes that this defect is the result of manufacturing problems at the factory, the possibility that the defect may be reader-related has not been ruled out, because of possible TLD element overheating.

Guidance:

NRC licensees are required to perform personnel radiation monitoring in accordance with 10 CFR 20.202. Licensees who use Panasonic Model 801 or Model 802 TLD systems should verify that their dosimeters continue to respond accurately and should be prepared to take compensatory actions if the defects described herein appear on their dosimeters. Other Panasonic models that employ calcium sulfate elements may also be affected and should be checked as well. The only way to ensure that one's dosimeters are not affected is to visually examine each dosimeter. Therefore, recalibrations and visual examination of dosimeters should be sufficient to identify damaged dosimeters. The dosimeters should be recalibrated whenever bubbles are observed. Where damaged dosimeters are detected, personnel exposure should be re-evaluated using data from undamaged elements or by using other available dosimetry such as selfreading dosimeters.

No written response to this information notice is required. If you need additional information regarding this matter, contact the Administrator of the appropriate NRC Regional Office.

L. I. Cobb, Director Division of Fuel Facilities, Materials and Safeguards Office of Inspection and Enforcement

Technical Contact: J. Metzger 301-492-9747

Attachments:

- Ltr. fm Panasonic Industrial Co. to its customers signed by Panasonic Natl. Sales Manager.
- 2. List of Recently Issued Information Notices

4. The effect has been seen on both lithium borate and calcium sulfate elements, but most frequently on the latter.

While Panasonic believes that this defect is the result of manufacturing problems at the factory, the possibility that the defect may be reader-related has not been ruled out, because of possible TLD element overheating.

Guidance

NRC licensees are required to perform personnel radiation monitoring in accordance with 10 CFR 20.202. Licensees who use Panasonic Model 801 or Model 802 TLD systems should verify that their dosimeters continue to respond accurately and should be prepared to take compensatory actions if the defects described herein appear on their dosimeters. Other Panasonic models which employ calcium sulfate elements may also be affected and should be checked as well. The only way to ensure that one's dosimeters are not affected is to visually examine each dosimeter. Therefore, recalibrations and visual examination of dosimeters should be sufficient to identify damaged dosimeters. The dosimeters are detected, personnel exposure should be re-evaluated using data from undamaged elements or by using other available dosimetry such as selfreading dosimeters.

No written response to this Information Notice is required. If you need additional information regarding this matter, contact the Administrator of the appropriate NRC Regional Office.

L. I. Cobb, Director Division of Fuel Facilities, Materials and Safeguards, IE

Technical Contact: J. Metzger 301-492-9747

Attachments:

- Ltr. fm Panasonic Industrial Co. to its customers signed by Panasonic Natl. Sales Manager.
- 2. List of recently issued Information Notices

bcc: G. Wayne Kerr, SP

gginbotham '82

IN 82-42 November 5, 1982 Page 2 of 2

4. The effect has been seen on both lithium borate and calcium sulfate elements, but more frequently on the latter.

Although Panasonic believes that this defect is the result of manufacturing problems at the factory, the possibility that the defect may be reader-related has not been ruled out, because of possible TLD element overheating.

Guidance:

NRC licensees are required to perform personnel radiation monitoring in accordance with 10 CFR 20.202. Licensees who use Panasonic Model 801 or Model 802 TLD systems should verify that their dosimeters continue to respond accurately and should be prepared to take compensatory actions if the defects described herein appear on their dosimeters. Other Panasonic models that employ calcium sulfate elements may also be affected and should be checked as well. The only way to ensure that one's dosimeters are not affected is to visually examine each dosimeter. Therefore, recalibrations and visual examination of dosimeters should be sufficient to identify damaged dosimeters. The dosimeters should be recalibrated whenever bubbles are observed. Where damaged dosimeters are detected, personnel exposure should be re-evaluated using data from undamaged elements or by using other available dosimetry such as selfreading dosimeters.

No written response to this information notice is required. If you need additional information regarding this matter, contact the Administrator of the appropriate NRC Regional Office.

L. I. Cobb, Director Division of Fuel Facilities, Materials and Safeguards Office of Inspection and Enforcement

١.

Technical Contact: J. Metzger 301-492-9747

Attachments:

- Ltr. fm Panasonic Industrial Co. to its customers signed by Panasonic Natl. Sales Manager.
- 2. List of Recently Issued Information Notices

bcc: G. Wayne Kerr, SP

*See previous sheet for concurrences.

WPU: JD	IE:F/MB*	IE:FFMB:C*	5:)FMS:D
11/01/82	J.Metzger/DM	L.Higginbotham	LVI. Cobb
5520	10/ /82	10/ /82	11/ /82



Panasonio industriel Company Division of Matsushita Electric Corporation of America

One Penesonic Way Becaucus, New Jersey 07084 201,248,5200 Cable: Mecanece NJ

Executive Offices

Attachment / IN 82-42

November 5, 1982

Sear Sealth Thysicist:

that the authom coated polyamide backing matarial in the elements of our dosimeters have developed bubble type imperfections. Our factory expineers traced the problem back to the production of the dosimeter inserts. The production environment should be virtually humidity free at this time, however, the air system was shut down at the interface between the carbon coated polyamide and the aluminum substrate holding the phosphor. In the heating cycle the element is hanted more aniatithe phosphor. In the heating cycle the element is heated very quickly creating vater vepor instantaneously which stresses the layers present in the element structure beyond their limits creating a bubble condition.

Permasonic has analyzed the situation and surfronmental conditions have been resolved in the production process. Also, a new hedge has been developed which contains a single layer where so water could be trapped, This new bedge contains a single polyzedde/carbon backing which is a new meterial. There will no longer be an alumin substrate. These new bedges will be completely tompatible with the old bedges. The carget date for the availability of these bedges is February 1983. be an aludinum

50% or less volative humidity. If the dosimeters are worn in a high humidity area especially during the summer months a 12 hour or longer staring period in a 50% relative humidity area before reading is recommended. Papasonic recom ands that docimeters be stored and read it a controlled environ mat of

two production runs. five years without problems. as as seen as possible and we will so however, that you wilt until the new requested if this is possible. leveral of the facilities have had acces of the 150,000 docimeters sold for as long as obless. Therefore, this sight be an isolated If you find this has happened to your dowing will replace the dosimeters from of charge. We support the new badyes are available before replacement is an isolated one from one and for the 2

nevid Katuman Antional Gales Manepus Industrial Aquipment Department Industrial Sales Division

Attachment 2 IN 82-42 November 5, 1982

LIST OF RECENTLY ISSUED IE INFORMATION NOTICES

عدقرا والمراجع

•

Information Notice No.	Subject	Date of Issue	Issued to
82-41	Failure of Safety/Relief Valves to Open at a BWR	10/22/82	All power reactor facilities holding an OL or CP
80~35	Leaking and Dislodged Iodine-125 Implant Seeds	10/6/82	Medical licensees holding specific licenses for human use of byproduct material in sealed sources
82-40	Deficiencies in Primary Con- tainment Electrical Penetra- tion Assemblies	09/22/82	All power reactor facilities holding an OL or CP
82-39	Service Degradation of Thick Wall Stainless Steel Recircu- lation System Piping at a BWR Plant	9/21/82	All BWR facilities holding an OL or CP
82-38	Change in Format and Distri- bution System for IE Bulletin Circulars, and Information Notices	9/22/82 s	All NRC licensees
82-34 Rev. 1	Welds in Main Control Panels	09/17/82	All power reactor facilities holding an OL or CP
82-37	Cracking in the Upper Shell to Transition Cone Girth Weld of a Steam Generator at an Operating Pressurized Water Reactor	9/14/82	All power reactor facilities holding an OL or CP
82-36	Respirator Users Warning for Certain 5-Minute Emergency Escape Self-Contained Apparatus	9/2/82	All powc, reactor facilities holding an OL or CP, fuel facilities and Priority I material licensees

************	XXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXX	XXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXX	XX AXAAAAAAAAAAAAAAAAAAAAAAAAAAAAAAAAA	
6460P # 01 TYPE LARLLS - 3	SELECTION - DUUDDU SETS - UZ	NG# 999999 WÜ VATE - 110982	UATE COMPLETED - 11/10/82	
***************************************	**************************************	*****************************	TINE LUNPLETED - 12546 - 12546 - 1254 - 1254 - 1254 - 1254 - 1254 - 1254	
6RUUP # 01 TYPE LABELS - 3 99999	56LECTIUN - 000000 SETS - 02	MO# 999999 WO DATE - 110982	DATE FunditTED 11/0/82	
,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,	\$ XXXXXX	*******************************	TINE CUMPLETED - 1254	
GROUP & OL TYPE LABELS - 1 99999	SELECTION - 000000 SETS - 02	мО# 999999 мО DATE - 110982	10-11-11-11-11-11-11-11-11-11-11-11-11-1	
***************************************	**************************************	**************************************	TIME CUMPLETED - 1254	6
GROUP # UL TYPF LABELS - 3 99999	SELECTION - 000000 SETS - 32	#0# 999999 #0 DAFE - 110982	DATE FOMPLETED - 11/10/82	
***************************************	******************************	,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,	TINE COMPLETED - 1254 - 1255 - 1254 - 12554 - 12554 - 12554 - 12554 - 12564 - 12556 - 12556 - 12556 - 12556 - 12556 - 12556 - 12556 - 12556 - 12556 - 12556 - 12566 - 12566 - 12556 - 125566 - 12556 - 125666 - 125666 - 125666 - 12566 - 1256666 - 1256666 - 125666666666666666666666666666666666666	
GROUF # 01 TYPE LABELS - 3 99999	SELECTION - 000000 SETS - 32	wD# 999999 WO DATE - 110982		
***************************************		***************************************		
GROUP # 01 TYPE LABELS - 3 99999	SELECTION - JOGOOD SETS - 02	#0# 999999 #0 DATE - 110782	LABELS 2,592	patine capeta
***********************************	***************************************	***************************************	TIME CUMPLETEU - 1254 - 1264 - 12554 - 12546 - 12546 - 12546 - 1254 - 1254 - 1254 - 12	
GROUP # 01 TYPE LABELS - 3 99999	5ELECTION - 000000 SETS - UZ	wuis 999999 wu DATE - 110982	DATE FOMBLELS 21592	
**************************************	**************************************	≠≠≠≠≠≠≠≠≠≠≠≠≠≠≠≠≠≠≠≠≠≠≠≠≠≠≠≠≠≠≠≠≠≠≠ ХХХХХХ	TINE COMPLETEU - 1254 - 2000 - 12546 - 12546 - 1254 - 1254 - 1254 - 1254 - 1254 - 1254	
GROUP # 01 TYPE LABELS - 3 59999	SELECTION - 000000 SETS - 02	#0# 999999 MD DATE - 110982	MATE COMPLETERS 1, 21592	
**********************	**************************************	**************************************	TIME CUMPLETED - 1254,000 **********************************	•
GROUP # 01 TYPE LABELS - 3 94999	SLLECTION - 000000 SETS - J2	40# 999999 %U DATE - 110982	DATE (1940) FLED - 11/10/82	
***************************************	*****************************	**************************************	TINE CUMPLETED - 1254	
GRGUP # 01 TYPE LAGELS - 5 94549	SELECTION - 000000 SETS - 02	wG# 999999 wJ DATE - 110982	MATE CLUMPLETED - 11/10/82	. •
***************************************	************************************	\$ XXXXXXX	TIME CUMPLETEU - 1254 - 125	
6x0UP + 01 1YPE LABELS - 3 99494	SELECTICN - 000000 SETS - 02	404 99999 NJ DATE - 110982	11ATE COMPLETED - 11/10/42	
•	• · · • · • • • • • • • • • • • • • • •		TIME CUMPLETED - 1254000	

(1

ł ŕ

4

GROUP # 01 TYPE LAUELS - 3	SELECTION - 000000 SETS - 02	113+42 - 112 - 113+42	4 (106) 1 5 503
**************************************	***************************************	***************************************	DATE CUAPLETED - 11/10/82 TIME COMPLETED - 1254 AXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXXX
GROUP # 01 TYPE LABELS - 3 99999	SELECTION - 000000 SETS - 02	₩C# 999999 ₩O DATE - 110982	# LABELS 2,592 DATE CUMPLETED - 11/10/82
***************************************	***************************************	**************************************	TIME CUMPLETED - 1254
GROUP # 01 TYPE LABELS - 3 99999	SELECTION - 000000 SETS - J2	NO#	# LABELS 2,592 Date cumpleted - 11/10/82
*****	***************************************	*** *********************************	TIME COMPLETED - 1254
GROUP # 01 IYPE LABELS - 3 99999	SELECTION - 000000 SETS - 02	₩0# 999999 ₩0 DATE - 110982	# LABELS 2,592
***************************************	***************************************	***************************************	
GROUP # 01 TYPE LABELS - 3 99999	SELECTION - 000000 SETS - 02	wD# 999999 WO DATE - 110982	# LABELS 2,592
0193200D5423 1 99999 Makjgrie n Aamudt R D #5 Cdatesville pa 19320	002109000444 1 99999 WILLIAM S ABBOTT ATTORNEY & COUNSELLOR AT LAW 50 CONGRESS STREET SUITE 925 BUSTOM MA 02109	*****	DATE COMPLETED - 11/10/82 TIME COMPLETED - 1254 ************************************
017120005319 1 99999 RUBERT M AULER ASST ATTNY GEN BUR OF REG CNCL 505 EXEC HOUSE PO BUX 2357 HARRISBURG PA 17120	093401000949 1 99999 ELIZABETH APFELBERG 1415 COZADERO SAN LUIS OBISPO CA 93401		
043220081505 1 99999 THOMAS APPLEGATE 3950 WOODBHIDGE KD Golumbus oh 43220	032217006296 1 99999 MITCHELL ATTALLA 4028 PUNCE DELEUN AVE JACKSONVILLE FL 32217		
077C98005144 1 99999 BRYAN L BAKER 1 1923 Hawthurne 1 Houston TX 77091	049720005503 1 99999 JIJANN BIER 204 CLINIGN ST CHARLEVOIX MI 49720		
065661006491 1 99999 Samull J Birk R al Bux 243 Murrisun M0 65061	027608005490 1 99999 Shelly Blum 1716 Scales Street Raleigh NC 27608		
06/801001587 1 99999	011778066814 1 99999		
CÁLLÊN BŮCK ESQUIRE PO BŮX 342 URDANA IL 61801	RÚNALD BÖMMER 10 WALNUT ROAD RUCKY PUINT NY 11778		
017102007111 1 99999 LOUISE BRADFURD 1011_GRLEN ST	065262006490 1 99999 EARL BRUWN School District Superintendent		
HARRISBURG PA 17102	PU BUX 9 KINGOOM ETTY MO 45267		

P

r - 1

	NH PI SUN 243 NHRISUN	MD 65061	1716 SCALES STREE RALEIGH	t NC 27608
•	061801001587 1 C Allen Buck Esquire PC 2018 PC 2018 Ukdana	11 91301 89898	OLL779J06814 1 RIVALD DOMMER LJ WALNUT ROAD RICRY PUINT	59939 Ny 11775
•	OLTIJZJJTILL I LOUISE SKAJFUKJ IOII GREEN ST HARKISBURG	99499 Pa 17102	D05262006690 1 EARL BRUWN School District Sc Pu Bus 9 Kingdom City	99999 IPERENTENDENT MO 65262
•	029076000867 1 BREIT & PUPSEY RGUTF 1 85x 93-C LITTLE MOUNTAIN	9999 4 SC 29076	008080000246 1 DAVID A CACCIA HD 2 BUX 70 Sewfil	08060 LN
•	092672001263 1 CITY OF SAN CLEMENT GEURGE CARAVALHU CITY MANAGER 100 AVENIDU PRESIDI SAN CLEMENTE	99999 E ^U Ca 92672	092037005012 1 A S CARSTENS 2071 CAMINITO CIRC MT LA JOLLA	99999 ULD NURTE CA 92037
•	061108006510 I UIANE CHAVEZ 602 UAK STREET APT RCCKFURD	99999 1L 61108	066839005472 1 Manda Christy 515 N IST St Burlington	99999 KS 66839
•	002038005482 1 ALAN R CLEETON 22 MACKINTOSH ST FRANKLIN	99999 Ma 02038	011788005094 1 CLUNTY EXEC/LEGISL/ PETER COHALAN SUFFOLK COUNTY EXEC VETEKANS MEMORIAL M HAVPPAUGE	99999 NTIVE BLDG LUTIVE 1164844 1164844
•	017120005444 1 Mark Cohen 512 E-3 Main Capital Harrisburg	99999 BLDG PA 17120	008070005643 1 ALFRED C COLEMAN ELEANUR G COLEMAN 35 K DRIVE PENNSVILLE	99999 NJ 08070
•	077043005148 1 Carolina conn 1414 Scenic ridge Houston	99999 TX 77043	015228055532 I JAMES COOKINHAM IOU ROYCROFT AVE PITTSBURGH	99999 PA 15228
•	033433006203 l Frederick P CC%An 6152 Verde Trail Boca Raton	99999 FL 33433	003833057219 1 June M Daigneault 14 Woodlawn Circle Exeter	99999 NH 03833
•	074119005186 1 ANDREK T DALTEN JR ESU 1437 SOUTH HAIN STRET TULSA	99999 Et Ok 74119	047259001461 1 TMONAS M DATTILU ESUIRE 311 EAST MAIN STREE MAUISON	99979 I IN +7250
	047250030048 1 THOPAS M DATTILU ATTOPNEY AT LAM 311 E MAIN STREET Madison	99999	021613005335 1 KALPH 5 DECKER ROUTE 4 BOX 190D CAMBRIDGE	99999 99999

· - .

		003833057219 1 99999
033433006273 l Frederick p (laan 6152 vende trail Boca Ratcy	99993 FL 33433	14 MOOULAWN CIRCLE NH 03033 Exeter
074119005186 1 Andren Tualten Jr Esg 1437 Sruth Main Stre Tui Sa	99499 E1 Ck 74119	U47250001461 1 99474 THOMAS M DATTILU ESQUIRE 311 EAST MAIN STREET MADISON
047250030048 1 THOPAS M DATTILU ATTOUNEY AT LAW BIL E MAIN STREET MADISUN	99949 In 47250	021613005335 1 99999 RALPH 5 DECKER ROUTE 4 ROX 190D CAMBRIDGE MD 21613
085027032286 1 STEPHEN DI CENSU 19203 N 29TH AVE N PHCENIX	99999 0 465 AZ 85027	028461005902 1 99999 CHARLES R DIETZ PLANT MANAGER PU ROX 458 NC 28461 SouthPort NC 28461
077021005108 1 John F Duugherty 4327 Allonbury St Houston	99499 Tx 77021	063130040077 L 99999 LEO DREY SIS WEST POINT AVENUE UNIVERSITY CITY MO 63130
046514066158 L STEPHEN J DRISCOLL 3444 E LAKE UR NOR ELKMART	99999 IH IN 46514	027705007049 L 99999 WELLS EDDLEMAN 718-A IREDELL ST NC 27705 DURHAM
O27602000487 1 THOMAS ERWIN ESUUIRF 115 W MORGAN ST RALEIGH	99999 NC 27602	060630080001 1 99999 Jumn J Finerty 5055 W WINDSOR AVE CHICAGD
• 085282062527 1 • WILLIAM G FISHER 2039 EAST BROADWAY SUITE 112	99999 SUITE Y RD AZ 85282	020006005993 1 99999 DAVID S FLEISCHAKER 1735 I STREET NW SUITE 709 WASHINGTON DC 20006
093440000948 1 RAYL FLEMING 1920 MAITIE RUAD SHELL BEACH	99999 Ca 93440	080210001343 1 99999 John W Flora Nuclear Consultant 3076 South High Street Denver
• 097702006255 1 RICHARU F FUSTER PU BUX 4263 SUNRIVER	99999 Or 91702	097110006134 1 99949 J C FREEDMAN BOX 553 Z CANNON BEACH (JR 97110 Z CANNON BEACH
002107064657 1 D FPEEMAN B PENNALUTIK ST NGPFULN	99999 MA (1210)	055108005172 l 99999 STEVE GADLER 2120 CARTER AVENUE 7 ST PAUL MN 55108

•

J 1

¥

 \mathbf{v}

U

فننده

003710077497 1

		1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	. •		***	
002107064657 1 D FPEEMAN	99363	05510800517. Steve GADE	2 1 Ek	999	199	
B PENNALODK ST Nopfola	MA 021	2120 CARTER 07 ST PAUL	AVENUS	24 M	55108	
092701001268	99999	09371007749	7 1	99 9	99	
ESQUIRE	MT EVE	5232 N FRES	NO	. .	61710	
SUITE 222	CA 921	01		LA	43110	
011788005095 COUNTY EXEC/LEG	SLATIVE BLDG	02070600607 ANDREW C GD	COHOPE	999	99	
SUFFELK COUNTY A	ATTERNEY MICHWAY	WHEATON	E HERRALE	HD	20906	
HAUPPAUGE	NY 117					
060611005344	99999	07747100514	51	999	99	
ONE INM PLAZA		1034 SA-LY ROSENBERG	ANN	Tx	77471	
CHICAGU	IL 606	11		• ~		
0701190070+0	00000	02020100201	e 1	001	00	
GARY L GROESCH 2257 BAYKON RGAI	·	ROBERT GUII	ເວ ເ	799	133	
NEW ORLEANS	LA 701	.19 314 PALL MAI COLUMBIA	LL	sc	29201	
097258006279	99999	00211100656	3 I	999	99	
WILLIAM L HALLMA	IRK	151 TREMONT RM 1045				
UNE SOUTHWEST CO	LUMBIA	BOSTEN	I	MA	02111	
PORTLANO	DR 972	58				
033138000804) MARTIN H HODDER ESCHIRE	99999	085040005300 PATRICIA_L	HOURIHAN	999	99	
1131 N E BOTH ST MIAMI	REET FL 331	PHOENIX 38	COIU 21	AZ	85040	
048072005002	99999	09458600653	2 1	999	99	
ESQUIRE	E MILLE	PO BOX 63	ARD .	~ •	0460-	
BERKLEY	MI 480	SUNGL		ĻA	74280	
077043006246 1 M & JM 81 SHUP	49999	061107001584 Philip B Jo	4 1 HNSON	999	99	
HEUSTON	TX 770	HRS 1907 STRATE	DRD LANE		61107	
		RUGKFURU			01101	
037830005105	99999	097258005944	4 1	99 9	99	
BALIER H JURDAN 881 W DUTER DRIV		FRANK JOSSI UNE SOUTHWE	LISON ST COLUMPI	A		
UNN RIUUE	4 M 3 6 M	JU PUKILAND	(1K	97258	
03761/00/11						
UZ7314006154 1 JAMES C LAMB 313 NUCDHAVEN PC	999994 IAD	UL7319005422	2 1	999	99	
CHAPEL HILL	NC 275	14 ETTERS	¢۲.	PA	17319	

•

0

θ

с,

•

•

-	CLUMPIA	41 21345	identifi in a statistist.	• • • •
•	043604-06124 1 Teary Ludie Atturney FGA Applic 915 Spitzer Autuin Telede	99999 ANTS Gright 43604	927514007048 1 Pmyllis Lotchin 195 Sridle Run Chaple Hill	99949 NC 27514
•	020850055217 1 R R MACLARY 9366 OVERLEN DP RGCK/ILLE	99999 MD 20859	049720005384 1 CHRISIA MAKIA Ruute 2 Bux 108C CHARLEVOIX	99979 MI 49720
•	077401005137 i D Marrack 420 Mulberry Lane Bellatat	99993 Tx 77401	018707001497 1 Collesn Marsh Rux 5584 Ru4 Muuntain Tup	99999 Pa 18707
•	048640091591 1 Mendall Marshall Route ID Midland	99 999 MI 48640	048640001423 1 WENUELL MAPSHALL RUDTE 10 MIDLAND	99999 MI 4864U
•	092653006388 1 Charles F McClung 24012 Calle DF LA PI Suite 330 Laguna Hills	99999 ATA CA 92653	077074001170 1 Brenda a McCorkle 6140 Darnell Hduston	99999 Tx 77074
•	053711006541 1 GARY L MILHOLLIN CHAIRMAN 1015 JFFFERSON ST MADISON	99999 WI 53711	049720005386 1 Jim E Mills Route 2 Box 108 C Chaplevoix	99999 HI 49720
•	077035006262 1 MR C MRS FRAMSUN 4822 WAYNESBORD DR MCUSTUN	99999 Tx 77035	085022007300 1 BRUCE NORTON ESQUIRE 3216 N THIRD ST SUI SUITE 202 PHDENIX	99999 TE 202 AZ 85012
6 U	094602064669 1 B E NYE Records specialist 4308 Everett Avenue Oakland	99999 Ca 94602	049664005389 1 John O'Neill II Route 2 Box 44 Maple City	99999 MI 49664
	070808055135 I Roy A Parkek 5061 Abelia drive Baton Kuuge	99999 La 70308	095814077536 1 Alan D Pasternak 455 Capitol Mall Suite 380 Sacramento	99949 Ca 95814
n Nati	027605007047 1 TAAVIS PAYNE ESU 73 W JUHNSON ST BUX 12643 Raleigm	99999 NC 27635	077025005147 1 Hilliam Pennengd 4070 Mekrick Houston	99999 Tx 77025
` .	003051004402 I RINA PETIT RFD 4 132 PAGE RD Litchfield	99999 NH 63051	027101006419 1 William G Pfffferkc Pu Rux 43 Winstun-Salem	99999 IRN NC 27101

• ---• ¢, • • 1

5

•	RED 4 132 PADE HO LITCHFIELD	NH 63051	PU AUX 43 AINSTUN-SALEM	NC 27101
•	OTCR93007081 1 JAMES W FIENCE PO BOX 23571 BATCN RUUGL	99949 La 70293	D77080006247 I F H POTTHOFF 1814 PINE VILLAGE DR HUUSTUN	99999 Tx 77080
•	019C87006118 1 PAUL W PURUOM 245 GJLPH HILLS READ RADNER	99994 Pa 19087	019348005334 1 LAWRENCE R QUARLES KENDAL AT LONGHOUD AF KENNETH SQUARE	99939 97 51 PA 19348
	065262006110 1 John G Reed RFD #1 Kingdom City	99999 MD 65262	016801005400 1 Forrest J Remick 305 E Hamilton Ave State College	99999 Pa 16801
•	077471005138 1 Mayne Rentfro PO Box 1335 Rosenberg	99999 Tx 77471	019401001254 1 ROGER B REYNOLDS ESQUIRE 324 Swede Street Norristown	99999 Pa 19401
•	014614006178 1 WARREN BROSENBAUM 1 MAIN ST EAST-WILDER ROCHESTER	99999 BLDG 707 NY 14614	008402006300 l Willard W Rosenberg 8 North Rumson Ave Margate	99999 NJ 08402
•	002146000437 L NGRAN ROSS 30 FRANCIS STREET BRUDKLINE	99999 NA 02146	066839005470 1 Maryellen Salava Route 1 Box 56 Burlington	99999 KS 66839
•	060416064406 1 Rubert Sayers RR #1 Box 158 Coal City	99999 IL 60416	048103005173 1 Milliam J Scanlon 2034 Pauline Blvd Ann Arbor	99999 Ni 48103
•	077043005141 I William J.Schuessler 5810 Darnell Mguston	99999 1x 77043	019475006439 1 John Shniper Meeting House Law Bli Mennonite Church RD-F Spring City	99999 G T 724 PA 19475
•	060172064217 1 Keith H Siegel Sii dee Lane Roselle	99999 11 60172	093401000945 2 Gordon A Silver Sandra A Silver 1760 Alisal Street San Luis Obispo	99999 Ca 93401
	066205006523 1 JOHN M SIAPSON ATTCRNEY FCR INTERVEI 4400 JOHNSON ORIVE SUITE ILO SHANNEE MISSION	99959 NURS KS 66205	048640021864 1 Mary Sinclair 5711 Sumerset Drive Midland	99999 MI 48640
°u	O14618005399 1 MICHAEL SLADE 12 TRAILWUDD CIRCLE NGCHESTER	99999 Ny 14618	019404001255 1 JUSEPH A SMYTH Assistant County Soli Muntgumery County Co	99999 ICITUR RTHUUSE

•	MESELLE	lL sul€i	SAN LUIS BRISPU	CA 15411
•	066205006523 1 Jühn M Sl4PSON Atterney Fer Interve 4400 Johnson Orive	94997 NURS	04864UJ21864 l Mary Sinclair S711 Sumerset Drive Midlanj	99999 M[48640
•	SUITE 110 SHANNEE MISSICN	KS 66205		
-	014618005398 1	99999	019404001255 I Joseph & Smyth	99999
•	12 TRAIL WOOD CIRCLE ROCHESTER	NY 14618	ASSISTANT COUNTY SOL Montgomery County Cl Nurristonn	LICITOR JURTHOUSE PA 19404
-	048623005178	99999	065024006487 1 Howard Steffen	99999
•	5795 N RIVER FREELAND	MI 48623	MAYOR CHAMOIS	MO 65024
•	U49503005388 1	99999	006114081138 1	99999
•	PETER A STEKETEE 505 PEOPLES BUILDING GRAND KAPIDS	MI 49503	1947 BROAD ST Hartford	CT 06114
•	028422006569 1 FRANKY THOMAS CHAIRMAN-BD GF CUMMI PU BUX 249 BGLIVIA	99999 SSIONERS NC 28422	OCANADO79706 1 CHRIS TODMEY 37 LOCKDARE STREET- ONTARIO MIS 225	99999 - AGINCOURT CANADA
•	094302006000 1	99999	060602006323 1 Robert J Vollen	99999
•	321 LYTTON AVENUE PALO ALTO	CA 94302	109 NORTH DEARBORN Chicago	IL 60602
•	060601705360 1 N Regijn Envir Conte John V Vranken	99999 Rol DIV	020815029135 1 SEYNDUR WENNER 4807 MDRGAN DRIVE	99999
•	188 WEST RANDOLPH 51 Chicago	IL 60601	CHEVY CHASE	MD 20815
•	019010006461 1 JUSEPH_H HHITE	99999	019010066205 1 JOSEPH H WHITE	99999
•	II SOUTH MERIUN AND	PA 19010	BRYN MAWR	PA 19010
•	008221006299 1 JOHN _ #ILLIAMSON	99 999	023883005666 1 J L MILSUN	99999
	211 FOREST DRIVE LINHGOD	NJ 08221	PO BOX 315 SURRY	VA 23883
U	037503007046	00000	044108006126	99999
U	RICHARD D WILSON 725 HUNTER ST ABEY	NC 27502	DANIEL D WILT PD BCX 08159 Cleveland	QH 44108
U.	N F L A	40 EI/0E		
	020015001629 1	393 33	027560079702 1 William M YFAGF#	99999
U	3409 SHEPHERD STREE	T MD 20015	BUX 33 ROUTE 1 MORRISVILLE	NC 27560

•

. *

ر

ني)

U

	211 First ST UNLES		AA AUL -	
	LINWGOD	NJ 08221	PG BGX 315 Surry	VA 238H3
	077502007046	00001	044108006176	00000
^	RICHARD D WILSON	,,,,,	DANIEL D WILT	****
	APEX	NC 27502	CLEVELAND	OH 44108
•				
•	020015001629 1 JGHN F WOLFE FSU	99999	U27560079702 1 Mili IAM M YEAGER	49999
	3409 SHEPHERD STREET	MO 20015	BUX 33 ROUTE 1	NC 375/0
-	CHEVY CHASE	ND 20013	TURRISVILLE	NC 27560
	043112005185 1	99999	005354007093 1	99999
A	ILENE H YOUNGHEIN 3900 CASHION PLACE		HALTER ZALUZNY CHAIRMAN-BOARD DE SE	I FI THAN
	OKLAHOMA CITY	OK 43112	PO BOX 116	VI 05354
•				
	005400052014	00000	A45333A44434 1	
•	ABEX CURP	*****	ACCESS CORP	39344
	R STRAIT	_	NORTHEAST MARKET MAN	AGER
	934 S_MAIN ST	D	4815 PARA DRIVE CINCINNATI	OH 45237
-	WILLITS	CA 95490		
	077042055303 1 ACF INDUSTRIES INC	99999	077459053152 1 ACF INDUSTRIES INC	99939
-	NKM VALVE DIVISION R V HJPKINS		W-K-M VALVE DIVISION	l
_ '	MGR-RELIABILITYCPROD	UCT ASSR	OA MANAGER	
-	HUUSTON	TX 77042	MISSOURI CITY	TX 77459
-	099352056854 1	99999	092672000874 1	99999
•	RICHARD WONG	LIANIS INC	LYN H HICKS	
	1955 JADWIN AVE	•	3908 CALLE ARIANA	
•	RICHLAND	WA 99352	SAN CLEMENTE	CA 92672
	072203063658 1	99999	019002006465 1	99999
۲	AFCO STEEL INC RUBERT & DATES		AIR É WÂTER POLLUTIO FRANK R ROMANO	N PÁTRÓL
	UA MANAGER		CHAIRMAN AVENUE	
	LITTLE ROCK	AR 72203	AMBLER	PA 19002
	03/ 10/00F0/ 5			
6	ALABANA DEPT OF PUBL	IC HEALTH	ALAGAMA DEPT OF PUBL	IC HEALTH
-	STATE OFFICE BUILDING	G	AUBREY GODWIN	ADMIN
6 3	MONTGOMERY	AL 36104	LHIEF-BUREAU OF RADI	O HEALTH
•			PONTGORERY	AL 36130
4 x	036420006551 1 ALABANA FLECTRIC CO	99999	035291081005 1	99999
-	CHARLES & LOWMAN		PO HOX 2641	AL 35301
	ANDALUSTA	AL 36420	BIKHINGHAH	AL 35291
•				
	035291005828 1	99999	035291006787 1	99999
•	ALABANA POYER CO F L CLAYTON		ALASAMA POWER CO POWER GENERATION	
	SENIUR VICE PRESIDEN	Ť	F L CLAYTON JR	
U .	BIRMINGHAM	AL 35291	PU BUX 2641	
	034313007405	00005	DIKEINGHAM	AL 35291
U	ALABAMA PUNER ()	<u> </u>	ALABAMA POWER CU	99999
-	W G HAIRSTON		NUCLEAR ENGINEERING DLIVER D KINGSLEY	
	PLANT MANAGER		MANAGER	

-

 \frown

•

•

8

2

Ô

4

3

. **_**`

л. • т и			I
	035291006550 88000		~
· · ·	R P NCUUNALU	035291002953 1 99999	
	PO BOX 2641	SAFETY AUDIT & ENGRNG REVIEN	
-	AL 352	91 B-300 2641	
-	035291002958 1 89800	BIRMINGHAM AL 35291	
	NUCLEAR GENERATION	036312080905 1 99999	
	GENERAL ANAGER	SAFETY AUDIT & ENGRG REVIEW	
•	BIRMINGHAN AL REDO	SUPERVISOR DRAWER ATO	_
	035291005822 1 89900	A SHECRO AL36312	7
•	W O WHITT	099811011192 1 99999	
•	PU BUR 2641	EKNST N MUELLER CLAMISSIONER	
-	AL 3529	POUCH CONCERNING AND A	
	007006079661 1 99869	AK 99811	
-	JANES O MICELI	029812025953 I ALLIED GEAFMAL NUM	-
	35 DANIEL ROAD WEST	J A BUCKHAN NUC SERVICES PO BOX 847	^
	NJ 07006	BARNWELL SC 29812	
	029812064600 1 99999		^
•	GEFFIELDS SUPVICES	017405052918 1 99999 ALLIS CHALMERS (OPP 99999	
Ъ	PU BCX 847 BARNELL	VALVE DIVISION H.L.GOLDHAN	
•	SC 29812	UNCOLN AND HARTIFY STREET	•
0	017405052917 1 99999	PA 17405	
	PRECISION COMPONENTS DIVISION	AMERACE CORP 1 99999	
.,	MGR-QUALITY METHODS	R J BERLING	
	YGRK PA 17405	LUO HILKURY STREET GRAFTON STREET	
	AMERICAN ELEC POWER 599999	010000000000 WI 53024	
	JOHN & DOLAN	ANERICAN ELEC POWER SVC CURP	· · · · · · · · · · · · · · · · · · ·
Ĺ	VICE CHAIRMAN-ENGINEERING	CHIEF NUCLEAR ENGINEER	•
	NEW YORK NY 10004	NEW YORK NY 10004	•
	AMERICAN ELEC POWER SWO 99999	006032037944	•
	RUBERT W JURGENSEN	ANENICAN NUCLEAR INSURERS	•
L	VICE PRESIDENT 2. BROADWAY	NICHAEL P FERRANTE	•
	NEW YURK NY 10004	SUITE 245 FARMINGTON	•
0	AMERICAN NUCLEAR INSUPERS	092021053502	•
	DOTTY SHERMAN	AMÉTER INC 199999 SIRAA US	•
	SUITE 245	N J PEET CA DIRECTOR	•
	DIRECTON CT J6032	790 GREENFIELD DRIVE	•
	ANACUNUA CO FLINDA CO	CA 92021	-
	QUALITY ASSURANCE MANAGER COR	ANACONDA-ERICSSON INC	•
-	YORK PA 1240	MGR-ENVIRON SAFETY 6 HEALTH	-
	01770100201	GREENWICH CT 06830	u • •
•	ANCHUR JANLING VALVE CU	094402052923	-
	TECHNICAL DIRECTOR	ANCHON DAREING VALVE CU	•
	WIELIAMSPURT PA 17701	K T MEEK VICE PRESIDENT	
		The second se	

	week of the second s	
•	5425 SOUTH RICE AVENUE HOUSTON TX 77036	HELLAIRE TX 77401
•	006877031760 1 99999 Anti-C() INC J D MURPHY 904 Ethani Alice Michael	ULTALYU31551 1 99999 Ansaluu spa Jbgv Tullu bading
٠	PD AUX 43 RIDGEFILLU CT U6877	DÚP MÁNAGER VIALE SARCA 336 MILAND 20126 - ITALY
•	O17404005427 I 99999 ANTI-NUCLEAR URLUP REPRES YURK GAIL P BHADFORD 245 W PHILADLLPHIA STREET YURK PA 17404	U2040U007119 L 99999 ANTIOCH SCHOOL OF LAW URBAN LAW INSTIT HERBERT SEMMEL CUUNSEL FOR CHHISTA MADIA
-		2633 16TH ST NU DC 20460
٠	U2046U0U6162 1 99999 ANTIOCH SCHOUL OF LAW URBAN LAW INSTITUTE Hernert Semme	025117002966 1 99999 APPALACHIAN POWER CO F Z COMBS
•	COUNSEL FUR CHRISTA-MARIA ETAL 2633 16TH ST NW WASHINGTUN DC 20460	SAFETY/SECURITY COORDINATOR 818 SKINNER DRIVE ST ALBANS WV 25117
•	060935001635 l 99999 APPLESEED LUURDINATUR BRIDGLT L KOREM 117 NUKTH LINDEN STREET ESSEX IL 60935	095030066339 1 99999 APPLIED ENGINEERING ASSOC ROBERT PETROKAS PRESIDENT 15625 LUS GATOS BLVD
•		LUS GATOS GATOS BLVD CA 95030
•	029115055443 1 99999 APPLIED ENGINEERING CO INC NUCLEAR PRUJECTS MANAGER PO 80X 1327	029115052926 1 99949 APPLIED ENGINEERING CO INC S METTA DA WANAGED
•	DRANGEBURG SC 29115	1525 CHARLESTOWN RUAD URANGEBURG SC 29115
•	030080067150 1 99999 Applied Physical Technulugy Nuclear Data Power Div Sue Maltese 2830 Scuim Cubb Drive	094303063067 1 99999 Aptecm Engineering Services Robert Cargill 795 San Antunia Road
•	SMYRNA GA 30080	CA 94303
•	094303056998 1 99999 Aptecm Engineering Services Steve A Lefton 795 San Antonig Road Palo Alto CA 94303	038118064584 1 99999 ARCADIA ASSOC K R ASCHNER PRESIDENICOT
1		SEATT NA 98118
,)	098105064757 1 99999 ARCADIA ASSOC F H KLUNDEP VICE PRESIDENT 4732 JATH AVE NURTHEAST 5	083401002213 L 99999 Argonne national Laboratory Idaho Facilities Librarian
)	SEATTLE WA 98105	P 0 80% 2528 IDAHU FALLS ID 83401
J	060439006414 1 99999 Arggrne Natiunal Laburatory NHC Assistance Project L. Charak	060439006423 I 99999 Argonne national Laboratory Martin J.Steindler
J	MANAGER 9700 SUUTH CASS AVENUE Argonie IL 60439	ARGONNE IL 60439
J	060443055307 1 99999 Argonne National Laboratory Office of Uperational Safety Edward a Wimunc	085007001376 1 99999 Akizona Attorney General Charles 5 Pierson Assistant Attorney General
	ARGONNE IL 60443	1700 WEST WASHINGTON PHUENIX AZ 85007
	085037301377 1 97997 ARIZC'A GUVERNURS OFFICE ECUM PLAN & DEV - ENERGY PRUG MA'GARET FALKER U_PUTY DIRECTUR	095036381095 1 99999 Akizona Public Service Co G C Andognini YP OF ELECTRIC OPERATIONS
	PHGENIX ASHINGTON AZ 85007	PHOENIX AZ 85036

0

0 - L

MARGANET ALLE TERMS TO THE OFFICE ALLE TERMS TO TREETING TO TREETI	VP OF ELECTRIC OPERATIONS STATION 1740 PHIENIX AZ 8
065036001550 1 99499 Akiziya Puhlic Service Cu Nuclear Prujects E E VAN Prunt Vice Pfesident Pu Ruk 21646 Phoenix AZ 85036	DB5C36002947 L 9999 Amizuna Purlic Semvice CJ Nucleam Projects E.E. Van Arunt Jr Vice Phesident Po Eux Z1666 Phoenix AZ de
085281064198 1 99999 ARIZUNA RADIAT REGULATORY AGCY CMARLES F TEUFCRU DIRECTUM 925 STRUTH 52ND AVE SUITE 2 TEMPE AZ 85281	072201002410 I 9999 Arransas board of mealth DIV of Environ mealth prote E F milson Director 4915 best markham St Little Rock ar 72
072201205814 1 99999 Arkansas Bureau Envir HLTH SVC Directur 4015 West Markham Street Little RGCK Ar 72201	072801007100 1 99999 Arransas NUL Plan 6 Respons S L Smith Uperations officer PU BCX 1749 Russellville ar 72
072801005809 1 99999 ARKANSAS NUCLEAR DNE JAMES P GHANLCN GENLKAL MANAGER PO BOX 608 RUSSELLVILLE AR 72801	072203005815 1 99999 ARKANSAS POWER & LIGHT CU ENERGY SUPPLY WILLIAM CAVANAUGH SR VICE PRESIDENT PO JUX 551 LITTLE ROCK AR 72
072203005004 I 999999 Arkansas pumer & Light Cu Energy Supply M Cavanaugh III Sr Vice President P O BOX 551 Little ROCK Ar 72203	072801064687 L 99999 ARKANSAS POWER & LIGHT CU JESSALYN DAVIS ASST ÜFC SERVICES SUPERVISU PO EDA 608 RUSSELLVILLE AK 72
072801079376 1 99999 Arkansas Pomer & Light Cu Arkansas Mullear Dne JIM M Levine Plant Manager Pu BOX 608 Russellville Ar 72801	072203005808 1 99999 ARKANSAS POWER & LIGH CO Juhn Marshall Manager-Licensing Po Box 55 Little Rock AR 723
032785052928 1 99999 ARMCO INC STAINLESS STEEL DIV MUBERT W SCHULTZ QUALITY ASSURANCE SUPV RT 2 ROX IA WILDWOOD FL 32785	095003C67159 1 99999 AKREMONY ASSOCIATES NDE SUPPORT SERVICES DUN AKREMUNY 3060 VALENCIA AVE SUITE G8 APTOS CA 950
020853079640 1 99999 ARTHUK M CARTER JR & ASSOC 14632 CRUSSWAY RD RUCKVILLE ND 20853	044047006445 1 99999 ASHTABULA COUNTY JUHN G CARDINAL PROSECUTING ATTURNEY COUNTY COURTHOUSE JEFFERSON OH 440
090224052929 1 99999 ASSUC PIPING & Engineering CO Div of Johnson Controls 1707 m Compton RLV0 Cumpton CA 90224	084015052930 1 99999 ASSUC PIPING & ENGINEERING C T J JONES VICE PRESIDENT 851 FRFEPORT INDUSTRIAL PRWY CLEARFIELD UT 840
065801CU5183 1 99999 ASSUCIATED ELEC COOP INC GERALD F DIDDLE GENERAL NANAGER PU BUX 754 SPRINGFIELD MD 65801	007013055837 1 99999 ASSOCIATED TECHNOLOGIES INC T KALLU 1200 ROUTE 46 CLIFTON NJ 0701
030603081899 I 99999 ATHENIANS FUP CLEAN ENLNUY TCM CLEMENTS PU BUX 97 ATHENS GA 30603	008404005789 1 99999 ATLANTIC CITY ELECTRIC CU FRANK J FICANDENTI Sk vice president 1400 Partfir Avf

•

)

 \mathbf{c}

`

Э

THIL OPERATIONS AZ 85336 97 1 99999 ALIC SEMVICE CJ JECTS INT JR JEMI AZ 85036 0 I 99999 Ard of Health Ron Health Protectn ARKHAM ST AR 72201 D 1 99999 PLAN & RESPONSE FFICER AR 72801 5 1 99999 NER & LIGHT CU LY VANAUGH SIDENT AR 72203 7 L 99999 WER & LIGHT CU NVIS NVISSUPERVISUR AK 72801 3 1 99999 1ER & LIGH CO ILL INSING AR 72203 OCIATES SERVICES 99999 A AVE SUITE GB CA 95003 I UNTY NAL ATTURNEY HOUSE 999999 0H 44047 1 99999 6 ENGINEERING CO T INDUSTRIAL PRWY UT 84015 L 99999 CHNDLOGIES INC NJ 07013

.

~	· · · · · · · · · · · · · · · · · · ·	માન રંકેટ ⊄	CLIFICN	NU UTULI
•	OTAINAÙG7161 I ATCMIL ENFRÙY COUNCI VICTUR CHEN-HNA CHE Administratùr 67 Lane 144 Reelung TAIPEI 107	99999 IL NG RD SEC 4 TAIWAN	OCANADOS6819 I ATOMIC ENERGY OF CAN LIARARY Susan NISH 1600 DURCHESTER BLVD H3H 1P9 Montreal Quebec	99999 IADA LTD WEST CANADA
•	020814040062 1 ATOMIC INDUSTRIAL FO TECHNICAL BEPT PATRICK C HIGGINS MANAGER 7101 WISCONSIN AVENU BETHESDA	99999 IRUM IÉ MD 20814	007932074111 1 AUTOMATIC SWITCH CO WILLIAM M BROWN SUPERVISOR-GOVT C NU HANOVER ROAD FLORMAM PARK	99999 CLEAR SALE NJ 07932
•	020910038054 I Autumation Industrie Vitro Lahuratokies d Jay Tebu 14000 geurgia Ave 4-1107 Silvek Spring	99999 S INC IVISION MD 20910	OCANADO53197 1 B & W CANADA LTD GALT ONTARIO	94949 Canada
•	062249052949 1 B LINE SYSTEMS INC E D MATLECK G A MANAGER 509 WEST MENROE STRE HIGHLAND	99999 ET IL 62249	044203052935 1 BABGUCK & WILCOX CO NED-JIST 91 STIRLING PO BOX 2 BARBERTON	99999 71 0H 44203
•	024505065892 3 BABLUCK & WILCX CO Nuclear Power Genera Library Po Bux 1260 Lynchburg	99999 NTION DIV VA 24505	024505001577 I BABGOEK & WILCOX CO RESEARCH & DEVELUPME LICENSE ADMINISTRATO PU BUX I260 LYNCHBURG	99999 NT DIV R VA 24505
•	024505080963 I RABCOCK & WILCOX CO VIRGINIA OPERATIONS R A ALIO Manager-NUC Matls & PO BLX 800 LYNLHBUKG	99999 Manu Div Va 24505	020814001431 I BABCUCK & WILCOX CO POWER GENERATION GRO R B BORSUM 7910 WOODMONT AVE SUITE 220 Bethesda	99999 IUP MD 20814
•	020814005115 1 BABCUCK & MILCCX CO NUCLEAR POWER GENER/ RUFENT B BONSUM 7910 NOUDMUNT AVE SUITE 220 Bethesda	99999 NTION DIV MD 20814	044601081396 1 BABCOCK & MILCOX CO Alliance Research CT E H Fisher 1562 Beeson ST Alliance	99999 R DH 44601
•	024505001578 1 Rábčiick & Wilcox Co Licensing NP gen dii J H Taylur P O Hox 1260 Lynchburg	99999 / Va 24505	024505001343 1 BABCUCK & WILCOX CO R A WALLIN PO BOX 1260 Lynchburg	99999 Va 24505
•	024505001346 1 BABCUCK & MILCOX CU MPPSS - PEBBLE SPRIT E G MARU Senidk Phuject Manag PU BUX 1260	99999 NGS JER	077002000120 1 BAKER & BOTTS J G COPELAND ESWUIRE UNE SHELL PLAZA HUUSTON	99999 Tx 77002
•	020006006266 1 BAKER & BUITS GURDUN GUUCH ESWUIRE 1701 PENNSYLVANIA AV MASHINGTUN	99999 16 NW DL 20006	077002000168 1 Baker & Rotts Melbert D Schwarz Esquire Une Shell Plaza Hgustun	99999 TX 77002
	035201005824 1 Bally Bingham Baker Rurekt a Buettner Pu Bux 306 Birmingham	99999 E HANTHORN AL 35201	021203005069 1 BALTIMORE GAS & ELEC JAMES A BIDUISON GENERAL COUNSEL PC 2CX 1475 BALTIMURE	99999 TRIC CU MU 21203

j

1

1.1

•	GIRDEN & BOITS	077002000168 1 99999 BAKEN & ROTTS MELBERT D SLMMARZ
~	ESQUIRE 1701 PENNSYLVANIA AVE NW WASHINGTUN DL 20006	UNE SHELL PLAZA TX 77002 HOUSTON
•	035201005824 1 99999 BALCH HINGHAM BAKER & HAWTHORN RUBERT A RUETNER PUBLY 300 A1 35201	021203005869 L 99999 Baltimone GAS C Electric Cu Janes A Biddison General Counsel Primeral Counsel
•	BIRAINGARA AL JILOI	BALTIMURE MD 21203
•	021203081567 i 99999 Baltimume gas 6 electric CO General Services Mumman Hummaken	020657007022 1 99999 BALTIMUKE GAS & ELECTRIC CO CALVERT CLIFFS NPS S M DAVIS
•	VICE PRESIDENT PC BUA 1475 BALTIMUNE MD 21203	SUPERVISOR-OPERATIONS CA MARYLAND RTES 2 6 4 LUSBY MD 20657
•	020657006431 L 99999 BALTIMERE GAS E ELECTRIC CU QUALITY ASSURANCE UEPARTMENT S M LAVIS	020657007023 1 99999 BALTIMURE GAS & ELECTRIC LU CALVERT CLIFFS NPS R E UENTON-TRAIN & TECH SERV
•	SUPERVISOR Clyrt Cliffs NUC PWR Plant Lusby MD 20657	MARYLAND RTES 2 & 4 LUSBY ND 20657
•	0212030A1568 1 99999 Baltimure LAS & Electric Co Security Services R. C. Dernoga, June Con	021203005875 1 99999 Baltimure GAS & Electric CO Quality Assurance Dept R M Douglass Sum Tsmallsood Road Complex
•	CENERAL SUPERVISOR PC BUX 1475 BALTINORE NO 21203	BUX 1475 BALTIMORE ND 21203
•	021203005830 1 99999 BALTIMORE GAS & ELECTRIC CO A E LUNDVALL VILE-RESSIONT-SUPPLY	021203006139 1 99999 Baltimore GAS & Electric CO Supply A F Lundyall Jr A F Lundyall Jr Supply
•	BALTINGRE ND 21203	PU BOX 1475 BALTIMORE NO 21203
•	021203005870 1 99999 BALTIMORE GAS & ELECTRIC CO R C ULSUN R DOM 922 GGE BUILDING BOX 1475 MD 21203	020657JJ5871 1 99999 BALTIMORE GAS & ELECTRIC CO CALVERT CLIFFS NUC POWER PLANT LEUN 8 RUSSELL PLANT SUPERINTENDENT LUSAT SUPERINTENDENT
•	NALTINCKE NU 21203	
•	020657007021 I 99999 BALTIMURE GAS & ELECTRIC CU CALVERT CLIFFS NPS J A TIERMAN MANAGEREMUCI FAR POWER OFPT	DELECTRONE GAS & ELECTRIC CU JA TIERNAN Manager-Muclear Power Po Box 1475 Charles Center
•	MARYLAND RTES 2 6 4 MD 20657	LÜSBY MD 21203
•	095060063660 1 99999 BARUN & ASSOCIATES INC STEPHEN L RARUN ZUT MISSIUN STREET CA 95060	091749052940 1 99999 BARTON INSTRUMENTS UNIT DF ITT G R MELT JA DIMECTOR
•	SANTA CRUE CA FJOOD	400 SUUTH TURNBULL CANYON RD UITY OF INDUSTRY CA 91749
•	D432010H2125 1 99999 BATTILLE COLUMBUS LAB NUCLEAR TECH & PHYS SCI DEPT WALTER J ELGIN Phoject Manager	043201064637 1 99999 BATTELE COLUMBUS LAB Library R à Rawles Mgr Admin & Info Services Gue Ming Admine
•	SUS KING AVE OH 43201	COLUMNUS OH 43201
	099352006415 1 99999 BATTLLLF PACIFIC NORTHWEST LAB WALT APLEY C/D MAX CLAUSEN BATTLLLE BLVD-SIGMA IV BLDG	099352006329 1 999999 BATTELLE PACIFIC NORTHWEST LAE Spencer H BUSH PQ BUX 999 ACKS
•	RICHLANU WA 99352	KICHLAND BA 99337

U

~

1

~

9

1

-

•	020657007021 1 94949 BALTIMULE GAS & ELECTRIC LU CALVERT CLIFFS NPS J A TIEKMAN Manaufr-Nuclear Power Dept Maryland RTFS 2 6 4 10 10057	021203074435 3 HALTIMOHE GAS & ELECTHIC CU J A TIEMAN MANAGEN-NUCLEAN POWER PO RUX 1475 CHARLES CENTER LUSUY MU 21203
	LUSBY 40 20037 Ogsoboub 1060 1 99999 Baruh C Associates Inc Stephen L Harun Zut Missiun Street Santa Cruz CA 95060	U91749052940 1 99999 DANTON INSTRUMENTS UN17 DF ITT G R WELT JA DIRECTUR 900 SULTH TURNBULL CANYON RO EITY OF INDUSTRY CA 91749
	0432010H2125 1 99949 BATTELLE COLUMPUS LAB Huclear Tech & Pmys Sci Dept Walter J Elgin Project Manager SJ5 King Ave Columbus DH 43201	043201066637 1 99999 BATTELLE COLUMBUS LAB LIBRARY H A RAMLES MGR ADMIN & INFO SERVICES SUS KING AVENUE COLUMBUS OH 43201
	099352006415 1 99999 BATTELLE PACIFIC NORTHWEST LAB WALT APLEY C/U MAX CLAUSEN BATTELLE BLVD-SIGMA IV BLDG RICHLANU WA 99352	099352006329 1 99999 BATTELLE PACIFIC NORTHWEST LAB SPENCER H BUSH PO BUX 999 ACRS RICHLAND NA 99352
•	048106001113 1 99999 BECHTEL PUWER CURP CENTKAL LINKARY LIEKARIAN P C BUA 1000 ANN ARBUR MI 48106	094119064724 I 99999 BECHTEL PUWER CORP NFO LIBRARY PU BOX 3965 SAN FRANCISCO CA 94119
. •	020760005872 1 99999 Bechtel Pumer Corp J H Bell Chief Nuclear Engineer 15740 Shady Grove Ruad Gaitheksburg MD 20760	094119064588 1 99999 BECHTEL POWER CORP M H BRUINS STAFF ASST ADMIN SVCS PO BOX 3965 SAN FRANCISCU CA 94119
	020760026694 1 99999 BECHTLL PUWER CURP NANLY & CHAPMAN 15740 Shady Grove RD Gaithersburg MD 20760	048106052944 1 99999 BECHTEL POWER COMP J I DUISON PROJECI MANAGER POST OFFICE BOX 1000 ANN ARBUR MI 48106
	090060001126 1 99999 BELHTEL POWER CURP LUS ANGELES POWER DIVISION JEAN UKEGURY LIUKARIAN TERMINAL ANNEX PO BOX 60860	077001007067 1 99999 BECHTEL POWER CORP CHARLES MALLIGAN PU BUX 2166 TX 77001 HUUSTON TX 77001
	CUS ANGELES CA 40080 077252064612 1 99999 BECHTEL PUNER CUPP V L HICKMAN RECOROS MANAGER PO BUX 2160 MOUSTON TX 77252	020877064676 1 99999 BECHTEL PLYER CORP F P JUNES Supervisor Rec MGMT 15740 Shady Grove Road Gaithersburg ND 20877
	094119006457 I \$9999 BECMTEL POWER CURP G H JUNES PROJECT MANAGER 50 BEALE STREET - PO BOX 3965 SAN FRANCISCI CA 94119	020877006748 1 99999 BELHIEL POWER CONP GA DEPT D C KANSAL 15760 SMADY GROVE ROAD 141 GAITHERSBURG MO 20877
	094119064719 1 99999 BFCHTEL POWER CORP D H LANIER PRUJECT AUMINISTRATOR-SUPP GRP PO BOX 3965 SAN FRANCISCU CA 94119	O90650064620 1 99999 BECHTEL PUWER CORP C Y LOPINSKY STAFF ENGINEER 12400 E IMPERIAL HWY NORMALK CA 90650
	094119006280 1 99999 HFCMTHL PCAER COMP	077252064631 1 99999 Rechtel Puwer Cokp

O

ز.)

J

Û

Ü

U

Q 100

С

0

R.

-	QQ211000658 1 BERMAN & LEWENBERG FHANLIS S WHIGHT 211 CLNGRESS STHEET BUSTON	999999 Ma 92110	JIBOIO03445 I HETHLFHIM STEEL CURP MEMEDITH HEMPHILL ASSISTANT GENERAL COUP TOI EAST THIND STREED HETHLEHEM	99999 NSEL PA 14016	۹ ۹	لر. سرب
•	04972005343 1 BIG RUCK PUINT NUC PI PLANT SUPFHINTENDENT CMARLEVUIX	99939 DHER PLANT MI 49720	D97210052948 1 Hingham Willamette Cu Guy F Atkinson Cu G I Rove La Manager 2800 N W Front Avenue Portland	99999 98 97210	•	
•	002125052954 1 BIW CARLE SYSTEMS IN E MANCHESTER PRESIDENT 65 BAY STREET BOSTON	99999 C MA 02125	064114001128 1 HLACK & VEATCH LEU HACK LIBRARIAN 1500 MEADOW LAKE PKHY KANSAS LITY	99999 10 04114	•	
•	0641140051H1 1 Black & Veatch M H Nubinsun Po Rox 8405 Ransas Lity	49949 MD 64114	06414021969 BLACK & VEATCH PUWER M J HOBINSON P J BUX 9405 KANSAS CITY	99999 40 64114	•	
•	011801001412 1 BLAU & COHN Howard L Hlau Esquire 217 Newbridge RD Hicksville	99999 Ny 11801	OWGERNUO5318 I BMFT KEACTOR SAFETY HEINZ SEIPEL AUMINISTRATOR HEINEMANNSTRASSE 2 S300 JUENN 2	99999	•	-: tag
•	098124052953 1 BUEING CO ENGR & CONSTRUCTION GEORGE I LOULBOURN PO RCX 3707 MS 94-73 Seattle	99999 WA 98124	022100064704 1 BUEING CUMPUTER SERVIC ENEMGY TECH APPL S.G.LINEKER 7480 GALLUWS COURT CV-31 VIENNA	99999 LES VA 22180	•	
•	097208029759 1 BUNNEVILLE PUWER AUM PO HUX 3621 PURTLAND	99999 IN OR 97208	ULANADO52964 1 BURG WARNER CORP HYRUN JALKSON DIV JALK ANDERSON SUPVR-REM ENGINEFRING PO BUX 180 STATION H UNTARIO M4C 5J1	99999 TORUNTO CANADA	•	
•	040749078990 1 Bokg Warner Corp Byrun Jackson Pump D Merritt Hugins Po Bux 4966 M/S 74 Carson	99999 IVISION CA 90749	OVI409053679 1 BURG WARNER CORP NUCLEAR VALVE DIVISIO P L MILINAZU Manager quality Assur. 7500 Tykune Avenue Van Nuys	99999 N Ance Ca 91409	•	
ີ ຍ . ບ	090749077817 1 BURG WAKNEK CURP BYRUN JACKSON PUMP D JUHN MILLAN ENERGY SYSTEMS DEVEL PO BUX 4966 CARSDD	99999 IVISION OPMT_LTR CA 90/49	U90058052965 1 BIRG WARNER CORP Byrin Jackson Pump D1 R L Rumeru 2300 EA47 Vernon Aven 06 LDS Angeles	99999 V UE CA 90058)	
	002199064706 1 BOSTUN LUISUN CU W L AMDERSUN NUCLEAR ENGINEER BOD RULYSTON STREET BCSTON	99999 Ma 02199	002199001541 1 Rustun Edison Cu Nuclear Projects R M Ruiler Manager Rug Buylstun Strelt Buston	99999 99999	ر ر ر	
L L	002184006262 1 BOSTON EDIS'IN CO NUCLEAR OPERATIONS JUHN F FRASER 25 HRAINTREE HILL PA P 260 BRAINTREE	99979 RK Ma 02184	002199064607 I BUSTUN EDISCN CO B A GURUNN ENGR PERMANENT FILE S BUO BUYLSTON STREET RUSTUN	99994 YS Ma 02199	-	
	002199080929 1	99949	002199001437 1	99999		

ſ

. . . .

:

R

			and the second	
	M/S 79	CA 90749	VAN NUTS	CA #1****
	CARSON	ÇA JULI	10 105 805 2965 L	44444
	090749077817 1	94 944	BORG WARNER CORP	n I V
	BORG WARNER CURP	DIVISION	BARTIN JACKSON POMP	
	JOHN MILLAN	GEORME CTR	2300 EAST VERNON AV	ENUE
	ENERGY SYSTEMS UC	C. 00360	06 ANGELES	CA 90058
)	CARSON	LA YUINY	202 0001541 1	99999
	002199064706 1	99934	AUSTUN EUISON CO	
	BOSTON EUISUN CU		NULLEAN PROJECTS	
-	NUCLEAR ENGINEER	. T	MANAGEP	r
	ACCTON	C' MA 02199	BOD BUYLSIUN SINCE	MA 02199
	505101		000101	99934
	002184006262 1	49979	SUSTON LOISON CO	
	BUSTON EDISON CO	s	B A GURUGN	ESVS
	JEHN & FRASER		BUO BOYLSTON STREE	1 MA 02199
	25 BRAINTREE HILL	PARA	BUSTUN	MA VELT
	BRAINTREE	MA 02184		99999
	002199080929 1	99949	002199001437 L	
	BUSTON EDISON CO		PILGRIM NUCLEAR ST	ATION
	M/C NUCLEAN HILLIAM D HARRING	TON	HILLIAM P HAURET	
	SENTION V PRESIDE	IT OF NUCLEAR	BUO BOYLSTON STREE	T HA 02199
	BOO BUTLSTUN STAT	MA 02199	BUSTEN	00000
-	0005425	99999	002360081566 1	44111
-	ACTION EDISUN CO		L J MATHIS	
D	RICHAND U MACHON	MANAGER	STATION MANAGER	10
	RED I ROLKY HILL	ROAD A 07360	PLYMOUTH	MA 02360
	PLYNOUTH	AN ULIUU		
-		99999	002184005624	99999
	002184029310 4		BOSTON EDISON CO	
•	HIC NUCLLAR		A V MORISI	DAWK
	A V MORISI Manalfik		25 BRAINIREE HILL	MA 02184
	25 BRAINTREE HIL	L PARK MA 02184		
•	BRAININEC	00000	302199006073 1	49999
	002199005443 1		BUSTUN EDISON CO	
	H A SIDES		BOD BOYL STUN STRE	ET
	QUALITY ASSURANC	L MAMAGEN	BUSTON	MA VELVA
•	BOSTEN	NA 02194		
•			002141062526 1	99999
	094601063657	NETAL PRODUCTS	BP TECHNICAL SER	ICES
•	D RIGGE		LAMERIDGE	MA 02141
	TA MANAGER	AY	-	
-	DAKLAND	CA 94601		
•			075081063099 1	99994
	006226052955	1 94944	BRANT & ASSOCIAT	ES
•	WIRE & CASLE UI	VESTONS	WILLIAM A DRAM	1 . 16081
	P PETRUCHIK	DIVS	RÍCHARDSUN	14 13001
-	PG BGX 448	ET 06226		
•	WILL IMANTIC	CT VOLLO	00.001052057 1	99944
	020036006105	1 99999 TTNER E MONADE	BRISTOL STEEL C	IRON WORKS
•	BRICH MURIUN B	LISTELE & CONTRACTOR	PLANT #3	
-	1140 CONNECTIC	UT AVE NW	UA MANAGER	INTY INDUST PK
_	SUITE TIOU	DC 50036	, ARISTUL/WASH CUI	VA 24201
			0011010% 3310128041081	99999
	011973005037	1 99999	UIIVIJUGOBJI BRUDKHAVEN NATI	UNAL LABURATURY
-	BROUKHAVEN NAT	TUNAL LABORATOR	REACTOR DIVISIO	N
-	LISKAHIAN		UPTON	
		NY 1197	3 BLUG 703	NY 11973
- 🗳	UP I Die			

(

(

ς.

NY 11973 00003

-

•

•

0

۰

4

•

0

Ô

•

۲

-

	ALCONTRACTOR OF ALCONTRACTOR		State Contraction of the	(4) (2) (2)
	ASSUL UNIVERSITIES IN		MANAGLE CONSI	** 77:01
	LEUNARD L LITTUE DIRE	C108	HOUSTON	14 11001
	ASSISTANT TO THE	MA TTALS		
	OPICA		117001064659 L	99444
		99999	LADAN L ROOT INC	
	07700106 105 INC		HICHARD P HEGRI	F K
	BRUNN L ROST		ASSISTANT WA MARINE	
	PO BUX 3	TX 77001	PO 30X 2	IX FIOUL
	HOUSTON	•	HC0310.	
			1 05 205 8	99994
	· · · · •	49999	OTTOOLUSENOT INC	
	077001064728		BRUNK G HOU -	
•	BRUNN & MUUT ENO		M HILE DOSELLI	ENT
-	JE FNU DOCUMENTATIO	N	GRUUP VILLE BOX 3	17001
	PO BUX 3 BLDG 41	TX 77001	PUST UPFILE UUA	1X 11001
	HÖUSTUN		HUUSIUM	99999
		9000	ONGERMOS3327 BEAK	TOR GHBH
	014477053734	177777	BRUNN BUYER	
-	BROWN BUYER LEFELS	NIV.	ABT MHUCHEREL	
-	SWITCHGEAK STOTENS		DUDENSTR 44	MGERMANY
	D D DUVALSIDENT		6800 MANNHEIM I	
•	ROUTE 309 L NORRIST	PA 19477		00000
	SPRING HOUSE		006856057905 1	44444
		99949	BURNDY CURP	
	CHOZOLOUDO C HEYBUR	N	POWER	
	IDSEPH B HELM		H R HUNAJAN	S AVE
	ESQUIRE CITIZENS	PLAZA	203	CT 06856
	16TH FLOOR CATTER	KY 40202	NDRWALK	
•	LUUISVILLE			99999
	1	99999	0993220 RDE INC	
	007649006080 INC		B G CONN	100
	BUKNSEN		N/M GROUP SUPERV	
-	TOD KINUERKAMACK R	JAU NI 07649	601 WILLIAMS BET	WA 99352
	ORADELL		RICHLAND	
				99999
	•	00099	007649006424 INC	
1	007649064616	,,,	BURNALA S KC STA	ANDARDS SUPA
	BURNS & ROE INC		H C JUNG	
	S A JUNUAR AUNINISTRATO	R	700 KINDERKAMAL	K KU NJ 07649
	ADD KINDERKAMACK R	UAU N.1 07649	URADELL	
	DRADELL			
		00000	007649064639	1 44477
	007649006828 1	99994	AURNS & RUE INC	
	AURNS & ROE INC		H L KIESS CONTRO	IL SUPVR
	KENHETH LISTACK	RD	TOO KINDERKAMAC	K KD NI 07649
	700 KINDERAM	NT 01044	ORADELL	110 0.00
	URADCEE			
			015077005708	
1	007-49001160	99994	BURRUUGH UF SH	the tracent
	AURNS & RUE INC		THOMAS J LICENE	
	NUCLEAR LICENSING		00 80x 26	DA 15077
1	STANLEY	RD	a SHIPPINGPORT	PA LIGHT
	IDANELS	NJ UTO-	•	
10	URANEEL			1 99999
		99999	COLE BRAUN & CO) -
	007622UCFRVICE G	ROUP	HILLIAM A JACH	
	AL FXANDER J MACD	QUGALL	VICE PRESIDEN	MONT
	VILE PRESIDENT-2	ALES 0 001 460	1000 50010 184	CA 91802
- 1	110 SUMMIT AVE P	NJ 0764	S ALHANDRA	
	MONTVALE			, 99999
	–	00494	006450020109	•
1	091802052470	*****	C N FLAGE CO	A REALD AND F
4	C F BRAUN L CU		MANAGER-QUALL	14 VP20KWWCE
		UIVISION	02 450 HURDUCK A	VENUE CT 06450
1	ALHANBKA	CA YIS	MERIDEN	T T T T
1	and the second s			00000
	V		095814006285	1 99999
	045814006335	1 99999	CALIFORNIA DE	LEALTH SECTION
	CALIFURNIA UEPT	CF CONSERV	RADIOLOGICAL	RAD CONTROL UNIT
<u>P</u>	UTV UF MINES 4	GEOLUGI	THEF ENTER	
11	PERKA AWINDIN	1	ROOM 498	
			· · · · · · · · · · · · · · · · · · ·	

1.2

•

•

007649006828 1 BURNS & ROE INC	Adddd	007649064639 1 HURNS & KDE INC	44999
TOO KINDERKAMACK RU	NI 07.40	H L RIESS DUCUMENT CONTROL SUP	VR
		URADELL	NJ 07649
OUTO490011AU 1 BURNS & FLE INC NUCLEAR LIGENSINS STANLEY S HOSEN	99999	015077005708 1 Durruugh uf Shipping Thomas J Czerpah Mayor	99499 Port
BUG RINDERKAMÁCK AD Urajeli	NJ 07649	PO BOX 26 SHIPPINGPORT	PA 15077
OD7645038055 1 BUTLER SERVICE GROUP ALEXANDER J MACDUDULA VICE PRESIDENT-SALES	99999 LL	091802008724 1 C F BRAUN & CO William & Jack Vile President	99999
TTO SUMMIT AVE P C B MUNTVALE	DX 460 NJ 07645	1000 SCUTH FREMONT ALHAMBRA	CA 91802
091802052970 1 C F BRAUN E CO A H WHITAKER	99494	006450020109 1 C n Flagg CD Glenn R Lacev	99999
VICE PRES-PRIMER DIVI ALMAMBKA	SIDN CA 91802	MANAGER-QUALITY ASSU 450 MURDUCK AVENUE MERIDEN	RANGE GT 06459
095814006335 1 CALIFURNIA UEPT OF C DIV UF MINES & GEDLU PERKY AMINOTO 1416 9TH STREET HOUM 1341	99999 DNSERV GY	095814006285 1 California dept of H Radiological Health Chief Envir Kad Cont: 714 P Street Rudm 498	99999 EALTH SVCS SECTION ROL UNIT
SACRAMENTU	CA 95814	SACRAMENTU	CA 95814
095814012056 2 CALIFURNIA DEPT OF H NADIGLUGIC HEALTH BR JUE MARU CHIEF STUEST	99999 EALTH SVCS ANCH	095825080939 1 California Energy CD Siting & Environment E R Detter Div Chief	99999 MMISSIUN AL
SACRAMENTO	CA 95814	SACHAMENTO	CA 95825
095825006094 1 CALIFURNIA EMERGY CO CHRISTOPH ELLISUN 1111 MOWE AVENUE SACRAMENTO	99999 MISSION CA 95825	095814005986 1 California Legal Affi Bryon S Georgiou State Capital-Governi Sagramentu	99999 AIRS SECY DRS OFC CA 95814
-			
O95814057457 1 CALIFORNIA OFFICE OF ENERGY C ENVIRONMENT PHILLIP A GREENDERG	99999 Guvernor	094102006009 1 California public ut. Lawrence u garcia 5066 State Bldg	99999 ILITY COM
	C C C C C C C C C C	SAN FRANCISCO	CA 94102
094102007066 1	CA 95814		
CALIFURNIA PUBLIC UTI JANICE E KERN JSD MCALIFERD ST	ILITY COM	CALIFORNIA RESOURCES	AGENCY
SAN FRANCISCU	CA 94102	SECRETARY FOR RESOUR(1416 91M STREET ROOM 1311	
920003005766 1	99999	3468491010 895610078956 1	00000
CALIFURNIA STATE OFFI ELEANUR SCHWARTZ 600_PENNSYLVANIA AVE	SE	CALIFORNIANS FOR SAFE ELLIS S TRAUBER CHAIRNAN	EENERGY
RM 201 WASHINGTON	DC 20003	3673 MESA VERDE AVE EL DORADO HILLS	CA 95630
010016001415 I CAMMER & SHAPIRO Ralph Shapirg Esquire 9 East 4014 Storey	99499	091361079030 1 LANYON RESEARCH GHUUP ERNEST J CONWAY 741_LAMEFIELD RDAD	99999 INC
V NEW YURK	NY 10016	SUITE R WESTLAKE VILLAGE	CA 91361
019004052967 I CAPITAL PIPE & STEEL	99999 PRODUCT	028208000669 1 Carolina Frvirin Stud	99999 99999

Í

Q

•	027602006083 1 99999 CARULINA PUWEK & LIGHT CU POWEP SUPPLY CONSTRUCT & ENG	U19603U52968 1 99444 CARPENTER TECHNOLOGY COPP N R RÉCERISE
•	E E UTLEY EXEC VICL PRESIDENT PU BCX 1551 RALEIGH NC 270	CEN MGR AUCLEAR PHUDUCTS 101 WEST BERN STREET Heading Pa 19603 302
•	060521079443 1 99999 CATALYTIC INC 2001 SPKING RD	019102028466 1 99799 CATALYTIC INC PAUL KLICHART
•	DAK HPUUK 11 605	1500 MARKET STREET 521 PHILADELPHIA PA 19102
•	090064001221 1 99999 Genter for law public intere John R Phillips	OSPAINO23959 1 99999 ST CENTRAL NULLEAR DE ALMAKAZ JUAN M BLANCO DE INCECA 4
•	10951 WEST PICO BLVD LOS ANGELES CA 900	MADRID - 8 SPAIN 064
•	078403001440 1 999999 Central Power & Light Co M L Rokchelt Pu Bux 2121	OBELGIO57562 1 99999 Centre d'Etude de l'Ener Nucl Puwer plant ex 3 François moite
•	CORPUS CHRISTI TX 784	03 ADMINISTRATOR BUERETANG 200 B-2400 MOL BELGIUM
•	OFRANCOO3524 1 99999 CENTRE D'ETUDES NUCLEAIRES DEPT DE SURETE NULLEAIRE GABRIELLE KATZ	063192006104 1 99999 CHACKES & HUARE Kenneth M Chackes Ainy_For Joint Intervendrs
•	92.260 FONTENAY-AUX-ROSES FRANCE	ST LOUIS MD 63102
•	005404053041 1 99999 CHAMPLAIN LABLE CORP WIBE & CABLE DIVISION C L MARTIN C L MARTIN	027514007042 L 99999 LHANGE Daniel F Read President
•	WINDUSKI VT 054	04 CHAPEL HILL NC 27514
٠	002199067190 I 99999 CHARLES T MAIN INC THERMAL PUNER CIVISION J H NGBLE PRUDENTIAL CENTER	025304063988 1 99999 Charleston Area Nedical Center Memorial Div Ridgely Conant Director of Nuc Medicine
٠	BOSTON HA 021	99 3200 MÁĽCÚŘKLĚ AVĚ ŠĚ Charleston vy 25304
6	028207007020 1 99999 CHARLUITE MECKLENBURG ENV CO HENRY PRESLER CHAIRMAN 943 HENIY DIAFE	029210066210 1 99999 IAL CHEM NUCLEAR SYSTEMS INC INFORMATION SPECIALIST 240 STEMERIDGE DRIVE
L	CHARLOTTE NC 282	07 COLUMBIA SC 29210
C	060602005200 1 99999 CHERRY & FLYNN MYRUN CHERRY THREE FIRST NATIONAL PLAZA	021218006421 1 99799 Chesapeake Energy Alliance Rubert y Pollard 609 Munipelier Street
-	SUITE 3700 CHICAGU IL 606	DALTINORE MD 21218
J	038113052969 1 99999 Chilagu Bridge & Ikon Cu 2700 Channel Ave P G Box 133 Memphis TN 391	035201052971 1 99999 CHICAGU BRIDGE & IRON CO 08 WELDING & WA DEPT 13 PU 80X 277
•		BORMENGHAM AL 35201

•

.

68

 \mathbf{U}

3

 060611005731
 1
 99997
 045219032023
 1
 99999

 CHICAGU METRU SANITARY DIST
 LINCINNATI ALLI FOR RESPUN ENG

 CECIL
 LUE MING
 TUM CARPENTER

 DIRECTOR OF MESEARCH & DEV
 DIRECTUR

 IOU
 LAST ENTEST
 2270 VINE STREET

 CHICAGU
 II
 50511
 CINCINNATI

	0 38	113	0525	169 (1061	ł,	KCN (999 11	99	,	0352 CHIC	0105 109	297 881	DGE		IRUN	944 Cù	1. J. J.
•	270	5°C) 2H1	ANN S	IEL I	IVË P	6 50 T	3 X 4	13308 33113		9184 90 9 9184		277 A4	, m U			▲L	35201
	0600 CHIG	11(AG.)057) ##	131 180 181	SANE	TARY	599 D1	97 ST		0452 L1NL T0M	1903 INN2 CAR	3202 AT 1 RPEN	23 ALL ITER	1 ¹ F	OR R	599 ESPU	149 IN ENG
	CECI DIRE 100 CHIC	LTI	ja () ST E	RIE	SEAK	СН С ЕТ 11	υE -	¥ 83511		01RE 2270 CINC	VIN INN		TRE	ET		DH	45219
•	0452 1 NC NUCL	DIC INF LHI	1791 ATI SE	20 GAS RVIC		LECT	299 21 C	99 CU		0451 CINC WM H J R PLAN	5307 INN EIP SCHL T M	1092 111 446 111 111	GAS GAS NU GER		ELEC AR S	999 TRIC TAT	999 : CC (UN
•	PUP		960 VA T I)		01	μ	45201		PO B HUSC	ON 4 OW	201				DH	45153
•	074 1 L I T I C ARF	1070 2 E I	0051 NS 01	INA CTT CKE	DR S	AFE	999 Ent	199 R INC		0452 CITI 2699 CINC	2001 2EN CL	8149 5 A(1FT(ATI	95 Gain Dn A	ST VE	A RJ	1010 0H	ENVIR 45220
•	CLA	tem(DRE	•		C	ĸ	75107				•					
•	018 111 1407 1407 1407 1407	501 (ZE) (AS) (RP) (SC)	0014 NS 4 J 1 ERST	97 NG - 11 HALLI DN	L NST N IGAN	IUC D	999 ANG	999 JERS		0276 (111 CU-C PATR SLAT	120 2EN COR 1CT ER	070 S AI DIN A E N	43 GAIN ATÚP T NE EWM/	1 ST WM/	NUCI	99' E A R	999 POWER
•	SČRJ	ÎNTI	DN			P		18501		RALE	IGH		0111			NC	27612
•	075 C11 JUA	224 1251 111	001 NS 1	600 AS 500 ELL I	ç fur	sou	999 ND	ENGY		0782 CITI LANN 5106	330 ZEN Y	061 5 C 51N 54	50 ONCE K I N DRO	RN	ED AI	99' BT N	999 UC PaR
	1420 DAL	SIU 5 Si LAS	DUTI	H POI	LK	T	x	75224		SAN	ANT	ONL	Ō			TX	78233
P.	060	.05	006	604			99	999		0170	3430	055	14	1		99	999
		IZE	NS I NS I H L FI	FOR INTZ	A BEI Etiti Urfn	ITER IUNER STRE	ÉŃI S FT	VIRON		CITI VIRC CHAI 264	ITEN INI IRMA WAL	S F A N TON	SOU SOU	N S. FHAI	RÐ RÐ	ENVI	RON
٠	ŠÚI CHI	TË Cag	160	5		1	L	60605		ĩ.ÉM(DYNE					PA	17043
•	077 C11 PEG	422 12E GY	005 NS RU	309 FDK CHOK		ABLE	999 U	999 TILS		078 CIT RIC:	7670 7 DF 1880	1070 AU	39 STI BAL ATT		H F Y	99	999
	RUU	LUI TE ZOR	I B	ox'î	6610) 684	` 1	x	11422		AUS	BUX TIN	iós	8		-	TX	78767
•	078 Ç11	767 Y Q	001	481 USTI	1 N 1	560A4	99 T M	999 ENT		049 (11)	1060 Y UF	056 BR BEM	91 10G	1 Man K		99	999
	NIC	HOL	AŠ	ĞER	Eiszi	ER				BRI	ÖR DGMA	LN .				HI	49106
•	AUS	8C) 1 [N	10	88		1	x	78767									
•	019	107	1306 3F P	437 HILA		AIA	99	999		O15 CIT MAR	2190 V DF VIN	0051 F P1 FE		EL EL	GH	99	999
•	PHI	LAU	DEEP	HIA		1	•	19107		313 P11	- ČÍ 1 T 58 (IY C JRGH	OUN 1	ĨŸ	BUIL	PA	15219
•	092 C I 1	267) 17 (101)	2004 DF 1	5376 SAN C	LEME	NTE	99	999		060 011 011	0490 V UF N B	081 21 5 Pt	553 IUN ENCE	1 R		99	9999
•	SAP	i ĉi	LEME	ENTE		1	LA	92672		MAY 282 210	OR B SI N	HER	IDAN	RC	•	IL	60034
	044 ÇĻļ	EVE	105 LANI	1501 2 EL	ec lu	LUMI	99 NA 1	999 ING CO		044 LLE	101 ¥EL	DU4!	553 ELE NGIN			94 11NA 1 6 CI	9999 FING LU DNSTR

	sfiul1 +1 53511	
•	006095038343 1 99999 CCMRUSTION ENGINEERING INC PSG LIBRARY-TEPT 6435-405 Librarian-Atin XGD OOCUMENTS LODU PRUSPECT FILL ROAD WINDSUR CT 06095	020014005904 1 99999 COMBUSTIUM ENGINEERING INC CHE POBER SYSTEMS CHARLES B BRINKMAN 4853 CURDELL AVE SUITE A-1 MD 20014
•	006095064587 L 99999	006095006333 [99999 COMPUSTION ENCINEERING INC
•	H D BROSMAN HGR GROUP RECOS & SECPTY	V C HALL 1000 PROSPECT HILL RDAD
	1000 PRÚSPEČŤ HILL RÚAJ 6110–1901	WINDSOR CT 06095
•	HINDSOR CT 06095	
	006095005873 1 99999 COMBUSTION ENGINEERING INC	CUMBUSTION ENGINEERING INC
•	P & KRUSE MANAGER-ENGINEERING SERVICES	P L MCGILL VICE PRESIDENT
•	WINDSGR CT 06095	WINDSOR CI 06905
-	120014038052 1 09999	006095001660 1 99999
•	CLABUSTION ENGINEERING INC	COMBUSTION ENGINEERING INC
	4853 CORDELL AVE BETHESDA ND 20014	1000 PROSPECT HILL ROAD WINUSOR CT 06095
•		
	006095026584 1 99999	037402052975 1 99999
•	A E SCHERER	COMBUSTION ENGINEERING INC
_	1000 PRUSPECT HILL ROAD	911 W NAIN ST
•	WINDSUR CT 06095	CHRITANUUGA IN STAUZ
•	006095001299 1 99999 Combustion Engineering inc	OMEXICO66478 1 99999 Comisión Federal Electricidad
-	CE POWER SYSTEMS	APDO PUSTAL NO 5-214 MEXICO 5-0F MEXICO
	1000 PROSPECT HILL ROAD	
	WIND508 CT 06095	
•	WINDSUR CT 06095	
•	WINDSUR CT 06095 OMEXICO64504 1 99999 COMISION NACIONAL SEGURIDAD	OBRAZIOO8064 1 99999 Comissao Nacional de Energia
•	WINDSUR CT 06095 OMEXICO64504 1 99999 COMISION NACIONAL SEGURIDAD NUCLEAR Y SALVAGUARDIAS INSURGENTES SUR ND 1806	UBRAZIOOBO64 1 99999 Comissao Nacional de Energia International affairs office Elve M. De Castru
•	WINDSUR CT 06095 OMEXICO64504 1 99999 COMISION NACIONAL SEGURIDAD NUCLEAR Y SALVAGUARDIAS INSURGENTES SUR ND 1806 MEXICO DF 01030 MEXICO	OBRAZIOO8064 1 99999 COMISSAO NACIONAL DE ENERGIA INTERNATIUNAL AFFAIRS OFFICE ELVE M DE CASTRU ADMINISTRATOK RUA GENERAL SEVERIANO 90
• •	WINDSUR CT 06095 OMEXICO64504 1 99999 CONISION NACIONAL SEGURIDAD NUCLEAR Y SALVAGUARDIAS INSURGENTES SUR NO 1806 MEXICO OF 01030 MEXICO	OBRAZIOOBO64 1 99999 COMISSAO NACIONAL DE ENERGIA INTERNATIONAL AFFAIRS OFFICE ELVE M DE CASTRU ADMINISTRATOR RUA GENERAL SEVERIANO 90 RIO DE JANEIRO 20000 BRAZIL 06060006340 1 999990
•	WINDSUR CT 06095 OMEXICO64504 1 99999 COMISION NACIONAL SEGURIDAD NUCLEAR Y SALVAGUARDIAS INSURGENTES SUR NO 1806 MEXICO DF 01030 MEXICO 049201029240 1 99999 COMMONWEALTH ASSOCIATES INC PLAFE DIVISION	UBRAZIQOBO64 1 99999 COMISSAD NACIONAL DE ENERGIA INTERNATIUNAL AFFAIRS OFFICE ELVE M DE CASTRU ADMINISTRATOK RUA GENERAL SEVERIAND 90 RIO DE JANEIRU 20000 BRAZIL 060690064240 1 99999 COMMONMEALTH EDISON CO NUCLEAN SECURITY AUMIN DEFICE
•	WINDSUR CT 06095 OMEXICO64504 1 99999 COMISION NACIONAL SEGURIDAD NUCLEAR Y SALVAGUARDIAS INSURGENTES SUR ND 1806 MEXICO DF 01030 MEXICO 049201029240 1 99999 COMMONWEALTH ASSOCIATES INC PGWER DJVISION H RANDALL SR PEG CUNSUITANT	OBRAZIOO8064 1 99999 COMISSAO NACIONAL DE ENERGIA INTERNATIUNAL AFFAIRS OFFICE ELVE M DE CASTRU ADMINISTRATOK RUA GENERAL SEVERIANO 90 RIO DE JANEIRU 20000 BRAZIL 060690064240 1 99999 COMMONHEALTH EDISON CO NUCLEAR SECURITY ADMIN OFFICE PO BOX 767 72 WEST ADAMS - ROOM 1230 F
• • • • • •	WINDSUR CT 06095 OMEXICO64504 1 99999 COMISION NACIONAL SEGURIDAD NUCLEAR Y SALVAGUARDIAS INSURGENTES SUR NO 1806 MEXICO OF 01030 MEXICO 049201029240 1 99999 COMMONWEALTH ASSOCIATES INC PGWER DIVISION W H RANDALL SR PEG CUNSULTANT 209 E WASHINGTON AVE JACKSON MI 49201	OBRAZIOOBO64 1 99999 COMISSAO NACIONAL DE ENERGIA INTERNATIUNAL AFFAIRS OFFICE ELVE M DE CASTRU ADMINISTRATOR RUA GENERAL SEVERIANO 90 RIO DE JANEIRU 20000 BRAZIL O60690064240 1 99999 COMONWEALTH EDISON CO NUCLEAR SECURITY AUMIN DFFICE PO BOX 767 72 WEST ADAMS - ROUM 1230 E CHICAGO IL 60690
• • • • • •	WINDSUR CT 06095 OMEXICO64504 99999 COMISION NACIONAL SEGURIDAD NUCLEAR Y SALVAGUARDIAS INSURGENTES SUR NO 1806 MEXICO OF 01030 MEXICO OF 01030 MEXICO OF 01030 MEXICO OF 01030 MEXICO DF 01029240 1 99999 COMMONHEALTH ASSOCIATES INC PCMER UJVISION W H RANDALL SR PEG CUNSULTANT ZOY E WASHINUTUN AVE JACKSON MI 49201 060690079384 1	OBRAZIOOBO64 1 99999 COMISSAO NACIONAL DE ENERGIA INTERNATIONAL AFFAIRS OFFICE ELVE M DE CASTRU ADMINISTRATOR RUA GENERAL SEVERIAND 90 RIO DE JANEIRU 20000 BRAZIL O60690066240 1 99999 COMMONNEALTH EDISON CO NUCLEAR SECURITY ADMIN OFFICE PO BOX 767 72 WEST ADAMS - ROUM 1230 E CHICAGO IL 60690
• • • • • • • • • • • • • • • • • • • •	WINDSUR CT 06095 OMEXICO64504 1 99999 COMISION NACIONAL SEGURIDAD NUCLEAR Y SALVAGUARDIAS INSURGENTES SUR NO 1806 MEXICO DF 01030 MEXICO 049201029240 1 99999 COMMONWEALTH ASSOCIATES INC PGER DJ VISION W M RANDALL SR PEG CUNSULTANT 209 E WASHINGTON AVE JACKSON MI 49201 060690079384 1 99999 COMMONWEALTH EDISON CO R. CUSARU.	OBRAZIOOBO64 1 99999 COMISSAO NACIONAL DE ENERGIA INTERNATIUNAL AFFAIRS OFFICE ELVE M DE CASTRU ADMINISTRATOK RUA GENERAL SEVERIANO 90 RIO DE JANEIRU 20000 BRAZIL 060690064240 1 99999 COMMONMEALTH EDISON CO NUCLEAR SECURITY ADMIN DFFICE PO BOX 767 72 MEST ADAMS - ROUM 1230 E CHICAGO 060690055268 1 99999 COMMONMEALTH EDISON CO L DELGEORGE
•	WINDSUR CT 06095 OMEXICO64504 99999 COMISION NACIONAL SEGURIDAD NUCLEAR Y SALVAGUARDIAS INSURGENTES SUR NO 1806 MEXICO DF 049201029240 99999 COMMONWEALTH ASSOCIATES INC PCWER DJVISION W H RANDALL SR PEC CUNSULTANT 209 E MASHINGTON AVE JACKSON MI 49201 060690079384 1 99999 COMMONWEALTH EDISON CO R LUSARU SR PEC CUNSULTANT 209 E MASHINGTON AVE JACKSON MI 49201 060690079384 1 99999 COMMONWEALTH EDISON CO R LUSARU SITE CONSTRUCTION SUPERINTENDT PO BUX 767 IL 60800	OBRAZIQOBO64 1 99999 COMISSAD NACIONAL DE ENERGIA INTERNATIUNAL AFFAIRS OFFICE ELVE M DE CASTRU ADMINISTRATOR RUA GENERAL SEVERIANO 90 RIO DE JANEIRU 20000 BRAZIL O60690064240 1 99999 COMMONMEALTH EDISON CO NUCLEAR SECURITY AUMIN OFFICE PO BOX 767 72 MEST ADAMS - ROUM 1230 E CHICAGO O60690055268 1 99999 COMMONMEALTH EDISON CO L DELGEORGE DIRECTOR UF NUCLEAR LICENSING PO HUX 767
• • • • • • • • • • • • • • • • • • • •	WINDSUR CT 06095 OMEXICO64504 1 99999 COMISION NACIONAL SEGURIDAD NUCLEAR Y SALVAGUARDIAS INSURGENTES SUR NO 1806 MEXICO DF 01030 MEXICO 049201029240 1 99999 COMMONWEALTH ASSOCIATES INC PCWER DIVISION W H RANDALL SR PEG CUNSULTANT 209 E WASHINUTUN AVE JACKSON MI 49201 060690079384 1 99999 COMMONWEALTH EDISON CO R CUSARU SITE CCNSTRUCTION SUPERINTENDT PO BUX 767 CHICAGD IL 60690	OBRAZIOOBO64 1 99999 COMISSAO NACIONAL DE ENERGIA INTERNATIUNAL AFFAIRS OFFICE ELVE M DE CASTRU ADMINISTRATOR RUA GENERAL SEVERIANO 90 RIO DE JANEIRU 20000 BRAZIL O60690064240 1 99999 COMMONNEALTH EDISON CO NUCLEAR SECURITY AUMIN DFFICE PO BOX 767 72 MEST ADAMS - ROUM 1230 E CHICAGO IL 60690 060690055268 1 99999 COMMUNMEALTH EDISON CO L DELGEORGE DIRECTOR UF NUCLEAR LICENSING PO HOX 767 CHICAGO IL 60690
•	WINDSUR CT 06095 OMEXICO64504 99999 COMISION NACIONAL SEGURIDAD NUCLEAR Y SALVAGUARDIAS INSURGENTES SUR NO 1806 MEXICO OF 01030 MEXICO OF 01030 MEXICO OF 01030 MEXICO OF 01030 MEXICO DF 01030 MEXICO OF 01029240 GOMMONHEALTH ASSOCIATES INC MI 49201 O60690079384 99999 COMMONHEALTH EDISON CO R CUSARU SITE CONSTRUCTION SUPERINTENDT PO BUS ARD SITE CONSTRUCTION SUPERINTENDT PO IL 60690 O60690006231 99999 COMMONNEALTH EDISON CO	OBRAZIOOBO64 1 99999 COMISSAO NACIONAL DE ENERGIA INTERNATIUNAL AFFAIRS OFFICE ELVE M DE CASTRU ADMINISTRATOR RUA GENERAL SEVERIAND 90 RIO DE JANEIRU 20000 BRAZIL 060690066240 1 99999 COMMONNEALTH EDISON CO NUCLEAR SECURITY ADMIN DFFICE PO BOX 767 72 WEST ADAMS - ROUM 1230 E CHICAGO IL 60690 060690055268 1 99999 COMMONNEALTH EDISON CO L DELGEORGE DIRECTOR OF NUCLEAR LICENSING PO HOX 767 CHICAGO IL 60690 061341081556 1 99999
• • • • • • • • • • • • • • • • • • • •	WINDSURCT06095OMEXICO6450499999COMISION NACIONAL SEGURIDAD NUCLEAR Y SALVAGUARDIASINSURGENTES SUR NO 1806MEXICO DF 01030MEXICO DF 0102040MEXICO DF 01020400MEXICO DF 01020400MEXICO DF 01020400MEXICO DF 01020400MEXICO DF 010204000MEXICO DEGEDRGEDIKECTUR DF NUCLEAR LICENSING	OBRAZIOOBO64 1 99999 COMISSAD NACIONAL DE ENERGIA INTERNATIUNAL AFFAIRS OFFICE ELVE M DE CASTRU ADMINISTRATOR RUA GENERAL SEVERIANO 90 RIO DE JANEIRU 20000 BRAZIL 060690064240 1 99999 COMMONMEALTH EDISON CO NUCLEAR SECURITY ADMIN DFFICE PO BOX 767 72 MEST ADAMS - ROUM 1230 E CHICAGO IL 60690 060690055268 1 99999 COMMUNMEALTH EDISON CO L DELGEORGE DIRECTOR UF NUCLEAR LICENSING PO HOX 767 CHICAGO IL 60690 061341081556 1 99999 COMMUNMEALTH EDISON CO LA SALLE COUNTY STATION G J DIEDERICH
	WINDSURCT06095OMEXICO6450499999COMISION NACIONAL SEGURIDAD NUCLEAR Y SALVAGUARDIASINSURGENTES SUR NO 1806MEXICO DF 01030MEXICO DF 0104231MEXICO DIX T67CHICAGOMEXICO DIX T67 <td>OBRAZIQOBO64 1 99999 COMISSAD NACIONAL DE ENERGIA INTERNATIUNAL AFFAIRS OFFICE ELVE M DE CASTRU ADMINISTRATOR RUA GENERAL SEVERIAND 90 RIO DE JANEIRU 20000 BRAZIL O60690064240 1 99999 COMMONMEALTH EDISON CO NUCLEAR SECURITY AUMIN OFFICE PO BOX 767 72 MEST ADAMS - ROUM 1230 E CHICAGO IL 60690 O60690055268 1 99999 COMMONMEALTH EDISON CO L DELGEORGE DIRECTOR UF NUCLEAR LICENSING PO MUX 767 CHICAGO IL 60690 O61341081556 1 99999 COMMONMEALTH EDISON CO L DELGEORGE DIRECTOR UF NUCLEAR LICENSING PO MUX 767 CHICAGO IL 60690</br></td>	OBRAZIQOBO64 1 99999 COMISSAD NACIONAL DE ENERGIA INTERNATIUNAL AFFAIRS OFFICE
	WINDSURCT06095OMEXICO6450499999COMISION NACIONAL SEGURIDAD NUCLEAR Y SALVAGUARDIASINSURGENTES SUR NOINSURGENTES SUR NO04920102924099999COMMONWEALTH ASSOCIATES INC PCWER DIVISION W H RANDALL SR DEG CUNSULTANT 209 E WASHINGTON AVE JACKSON99999COMMONWEALTH EDISON CO R CUSARD99999COMMONWEALTH EDISON CO R CUSARD99999COMMONWEALTH EDISON CO R CUSARD99999COMMONWEALTH EDISON CO SITE CONSTRUCTION SUPERINTENDT PO BUX 76799993COMMONWEALTH EDISON CO LCUIS DELEGENGEE99993COMMONWEALTH EDISON CO LCUIS DELEGENGEE99993D60690J0623199993COMMONWEALTH EDISON CO LCUIS DELEGENGEE99993D60690J06231199993COMMONWEALTH EDISON CO LCUIS DELEGENGEE11DIKECTUR OF NUCLEAR LICENSING PU BOX 76711CHICAGOIL 60690	OBRAZIOOBO64 1 99999 COMISSAD NACIONAL DE ENERGIA INTERNATIUNAL AFFAIRS OFFICE ELVE M DE CASTRU ADMINISTRATOR RUA GENERAL SEVERIAND 90 RIO DE JANEIRU 20000 BRAZIL O60690064240 1 99999 COMMONWEALTH EDISON CO NUCLEAR SECURITY AUMIN DFFICE PO BOX 767 72 MEST ADAMS - ROUM 1230 E CHICAGO 1L 60690 O60690055268 1 99999 COMMONWEALTH EDISON CO L DELGEORGE DIRECTOR UF NUCLEAR LICENSING PO HOX 767 CHICAGO IL 60690 O61341081556 1 99999 COMMONWEALTH EDISON CO LA SALE COUNTY STATION G J DIEDERICH SUPERINTENDENT RR I PO BOX 24 MARSEILLES IL 61341
	WINDSURCT06095OMEXICO6450499999CDMISION NACIONAL SEGURIDAD NUCLEAR Y SALVAGUARDIASINSURGENTES SUR NDIB06MEXICO DF01030MEXICO DF01030O49201029240199999CDMMONWEALTH ASSOCIATES INC PCWER DIVISION W H RANDALL SR PEG CUNSULTANT 209 E WASHINGTON AVE JACKSCN060690079384199999CDMMONWEALTH EDISON CD RCUSARD COMMONWEALTH EDISON CDR CUSARD COMMONWEALTH EDISON CD CHICAGD060690006231199997CDMMONWEALTH EDISON CD LOUS DELGEORGE DIKECTUR OF NUCLEAR LICENSING PO BUX 767CHICAGDIL 60690060690006231199999CDMMONWEALTH EDISON CD LOUS DELGEORGEDIKECTUR OF NUCLEAR LICENSING PO BUX 767CHICAGDIL 606900606900793891060690079389	08RA/100806499999COMISSAD NACIONAL DE ENERGIAINTERNATIUNAL AFFAIRS OFFICEELVE M DE CASTRUADMINISTRATORRUA GENERAL SEVERIAND 90RIO DE JANEIRU 20000 BRAZIL060690066240199999COMMONNEALTH EDISON CONUCLEAR SECURITY ADMIN DFFICEPO BOX 76772 MEST ADAMS - ROUM 1230 ECHICAGO06069005526899999COMMONNEALTH EDISON COL DELGEORGEDIRECTOR UF NUCLEAR LICENSINGPO BOX 767CHICAGOCOMMONNEALTH EDISON COL DELGEORGEDIRECTOR UF NUCLEAR LICENSINGO61341081556SUPERINTENDENTRR I PO BOX 24MARSEILLESMARSEILLESIL 613410606907938006079380079999CUMMONNEALTH EDISUN CO
	WINDSURCT06095OMEXICO6450499999COMISION NACIONAL SEGURIDAD NUCLEAR Y SALVAGUARDIASINSURGENTES SUR NO 1806MEXICO DF 01030MEXICO DF 0102040MEXICO DE DEGEDRGEDIACTUR DF NUCLEAR LICENSINGPU 80X 767CHICAGOMEXICO IL 605900606090793870606090793870606090793870606090793870606090793870606090793870700000000000000000000000000000000000	08RA/100806499999COMISSAD NACIONAL DE ENERGIAINTERNATIUNAL AFFAIRS OFFICEELVE M DE CASTRUADMINISTRATORRUA GENERAL SEVERIAND 90RIO DE JANEIRU 20000 BRAZIL060690064240199999COMMONWEALTH EDISON CONUCLEAR SECURITY ADMIN DFFICEPO BOX 76772 WEST ADAMS - ROUM 1230 ECHICAGO060690055268060690055268060690055268060690055268060690055268060690055268060690055268060690055268060690055268060690055268060690055268199999COMMUNWEALTH EDISON COL DEGENGEDIRECTOR UF NUCLEAR LICENSINGPO HOX 767CHICAGOCHICAGO06134108155699999COMMUNWEALTH EDISON COLA SALLE COUNTY STATIONG J DIEDERICHSUPERINTENDENTRR I PO BUX 24MARSEILLESIL 61341060690079380199999CUMMONWEALTH EDISUN COJ F GUDACSTATIUN SUPERINTENDENTPO HUN 26PO HUN 26PO HUN 26FATIUN 50PO HUN 26FATIUN 50PO HUN 26PO HUN 26
	WINDSURCT06095OMEXICO6450499999COMISION NACIONAL SEGURIDAD NUCLEAR Y SALVAGUARDIASINSURGENTES SUR NO 1806 MEXICOO4920102924099999COMMONWEALTH ASSOCIATES INC PCWER DIVISION W H RANDALL SR PEG CUNSULTANT 209 E MASHINUTUN AVE JACKSON99999O6069007938499999COMMONWEALTH EDISON CO R CUSAND SITE CONSTRUCTION SUPERINTENDT PO BUX 767 CHICAGO99999O6069000623199999COMMONWEALTH EDISON CO LOUTS DELGFORGE11 60690O6069000623199999COMMONWEALTH EDISON CO LOUTS DELGFORGE11 60690O6069007038999999COMMONWEALTH EDISON CO LOUTS DELGFORGE11 60690O6069007038999999COMMONWEALTH EDISON CO LOUTS DELGFORGE11 60690O6069007038999999COMMONWEALTH EDISON CO LOUTS DELGFORGE11 60690O6069007038999999CHICAGOIL 60690O6069007038999999CHICAGOIL 60690	OBRAZIOOBO6499999COMISSAD NACIONAL DE ENERGIAINTERNATIUNAL AFFAIRS OFFICEELVE M DE CASTRUADMINISTRATORRUA GENERAL SEVERIAND 90RIO DE JANEIRU 20000 BRAZIL060690064240199999COMMONMEALTH EDISON CONUCLEAR SECURITY AUMIN DFFICEPO BOX 76772 MEST ADAMS - ROUM 1230 ECHICAGO060690055268060690055268060690055268060690075700100000000000000000000000000000000000

0.2020

A

* ~ *

- <u>-</u>	• • •	to a firm and the state of
•	ALIFICAL FRANCE PRESIDENT 4 IRVING PLACE NEW YORK NY 1.003	LUGENE INUMU SR RADIATION POGTECT INSTRUCT BUCHANAN NY LOSIL
•	010511005718 11 99999 CUNSULIDATED EDISGN CD OF NY CMARLES W JACKSUM VICE PRESIDENT-NUCLEAR PUWEP BRUADMAY ANU BLEAKLEY AVES	010003005724 I 99959 CONSGLIDATED EDISUN CO DF NY LUUIS F LIBERATURI NUCLEAR LICENSING ENGINEER 4 IRVING PLACE
•	BUCHANAN NY 10511	NEN YURK NY 10003
•	010511005717 11 99999 CONSOLIDATED ECISON CO UF NY FRANK MATRA Resident Construction Managek Broadway and Pleakley Aves Buchanan Ny 10511	010003005719 1 99999 CONSULIDATED EDISCN CO OF NY NUC AFFAIRS & QUALITY ASSURANC JOHN D OTODIE VICE PRESIDENT 4 IRVING PLACE NEW YORK NY 10003
_		049201005343 1 99999
•	STEVEN PHERSHEY STEVEN PHERSHEY JUNIPER & LUCUST STREETS PHILADEL PHIA PA 19107	JUJD L BACON 212 WEST MICHIGAN AVE JACKSON MI 49201
•	049201001421 1 99999 CONSUMERS PIWER CO	069201064735 1 99999 CONSUMERS POWER CO
•	JAMES E BRUNNER	D M BUDZIK PROJ RECORDS MGMT SUPVR
•	212 WEST MICHIGAN AVENUE JACKSON MI 49201	1945 W PARNELL ROAD JACKSON MI 49201
•	049201006276 1 99993 Consumers Power Co J & Couk Vice President	049201058220 1 99999 Consumers Power Co J & Couk Vice President
	1945 MÉST PARNALL RD JACKSCN MI 49201	1945 W PARNALL RD JACKSON MI 49201
•	049201003026 1 99999 CUNSUMERS POWER CD MILMA G FOGG NUCLEAR LICENSING ANALYST	049201064581 1 99999 Consumers Power Co J R Frey Managment Systems Admin
۲	JACKSON MI 49201	JACKSON MI 49201
•	049720081510 L 99999 CONSUMERS POWER CO BIG RUCK POINT NPS D P HUFFMAN PLANT SUPERINTENDENT	D20B14006416 1 99999 Consumers Power Co Ruger W Huston 7910 Woodmont Ave Su <u>i</u> te_220
•	RGUTE 3 BGX 591 CHARLEVUIX MI 49720	BETHESDA ND 20814
•	049043005337 L 99999 CUNSUMERS PCWER CO PALISADES PLANT Romert W Muntkuss Plant Manager	049201005391 1 99999 CUNSUMERS POWER CO PAUL A PERRY SECRETARY 212 WEST MICHIGAN AVE
•	COVERT MI 49043	JACKSON MI 49201
•	060450001512 I 99999 CUNSUMERS POWER CD DRESDEN NPS D J SCOTT STATION SUPERINTENDENT	048640066519 1 99999 Consumers Power CD Nuclear training center Peggy Stolž PD RDX 1407
•	RR 1 NURPIS IL 60450	MIDLAND MI 48640
•	049201005497 1 99999 CONSUMERS PCWER LG DAVID VANDEWALLE NUCLEAR LIL AJMINISTRATOR 212 W MICHIGAN AVF JACKSON MI 49201	092714052938 1 99999 CUNTREL COMPONENTS INC 2567 SE MAIN STREET INVINE CA 92714
	D20852064749 L 99999 CONTROL DATA CORP R L PRIEM MARKETING ASSUCIATE ACO3 FYF(UTIV, J VO	043050052981 1 99999 COOPER INDUSTRIES INC CUOPER ENERGY SERVICES G W MIZER TECH SERVISIO ON WANAGED

•

•

A

æ

۲

۲

.

0

J

	Phillippi Phillippi	1	MA JOEK HELPHULATIN	y thu
•	BRUNNVILLE	NE 68321	AUKIN SAADUSKY SIREE	UH 43050
•	016423052942 1 CGPES VULCAN ADV MANAGER	99999 9	002919052983 1 CGRNER & LADA CO INC M T ALLEN	94999
•	LAKE CITY	PA 16423	QUALITY ASSURANCE MN 1341 Elmwuod Avenu <u>e</u> Cranston	GR RI 02910
•	027622057209 1 CGRPUFATE CUNSULTING STEPHIN A LEHRMAN	99997 E DEV CO	054601006426 1 Cuulee Region Energy Geurge R Nygaard	99999 CUALITION
•	PC BUX 30096 RALEIGH	NG 27622	LACROSSE	₩I 54601
•	031513005790 L CCUNTY UF APPLING CCUNTY COMMISSIONERS	99999	065251006106 I County of Callaway Robert G WR Ight	99999
•	COUNTY COURTHOUSE	GA 31513	ROUTE #1 COUTY COURT FULTON	ERN DIST MO 65251
•	020768005876 I COUNTY OF CALVERT BD OF COUNTY COMMISS	99995 IONERS	049720005390 1 County of Charlevoix County Board of Supe	99999 RVISORS
•	PRESIDENT PRINCE FREDERICK	MD 20768	CHARLEYUIX	NI 49720
•	036250005796 1 County of citrus BDARD of commissione Wilbur Langely Chairman	99999 RS	097501006345 1 County of Columbia BD of County Commiss Rubert M Hunt	99999 IONERS
•	IVERNESS	FL 36250	ST HELENS	OR 97501
•	033130006555 L County DF Dade M R Stierheim County Manager-Metro Miami	99999 DADE CTY FL 33130	029535006554 1 County of Darlingtun Board of Supervisors McCuen Morrell Churty Courthouse	99939
•	065066006489 1	99999	DARLINGTON 017101005433	SC 29535
9	COUNTY OF DASCONADE Hakold Lottman Presiding Judge Routf 1		CUUNTY OF DAUPHIN EMERGENCY PREPAREDNES COURT HOUSE FRONT & MARKET STREET	34444 22
•	OWENSVILLE	NO 65066	HARRISBURG	PA 17101
٠	017101007003 1 County of Dauphin Lakky Hochendgner Commissioner	99999	017101005434 1 CRUNTY OF DAUPHIN DAUPHIN COUPY COURTH EARL B HOF N	99999 IQUSE
•	PO BOX 1295 Harkisburg	PA 17101	COMMISSION FRUNT & MARKET STREET MARRISBURG	PA 17101
•	017101005437 1 CCUNTY OF DAUPHIN DAUPHIN COUNTY COURTH JOHN E MINNICH	1002F 99998	055066005941 1 CUUNTY OF GOODHUE R L TANNER F CUUNTY AURITON	99999
•	CHAIRPERSON BO UF CON FRUNT & MARKET STREET HARRISBURG	MISSIUNER PA 17101	RED WING	MN 55066
•	060450005402 1 CUUNTY OF GRUNDY HOARD OF SUPERVISURS CHAIRNAN	99999	044077006128 1 COUNTY OF LAKE DUNALD T EZZUNE	99999
•	GRUNDY COUNTY COURTHO	IUSE 11. 60450	105 MAIN STREET PAINSVILLE	DH 44077
•	052406005646 1 COUNTY OF LINN BCARD OF SUPERVISORS CHAIRMAN CEDAR RAPIOS	9999y	065069006488 1 COUNTY OF MUNTGUMERY FRED LUEKEY PRESIDING JUDGE	99999
۲			RHINELAND	MU 65069

5

.....

A

	1			
•	Paris and Bhummyslith Ne	64321	ME VERNUN	1 1314 - 4-5115-13
•	016423052982 1 999 CGPES VULCAN ADV MANAGEN	69	00291G052983 1 CURNER & LADA CO IMC W T ALLEN	94999
•	LAKE CITY PA	16423	UGALITY ASSURANCE MN 1341 ELNWUGD AVENUE CRANSTUN	RI 02410
•	027622057209 1 999 CORPORATE CONSULTING & DE STEPPEN & LEHRMAN MANAGEN SEISMIE DEDIECTS	199 V 20	U54601006426 1 CUULEE REGION ENERGY GEORGE R NYGAARD 2307 FAST AVENUE	99999 CUAL IT ION
•	PO BUX 30096 RALEIGH NC	27622	LACROSSE	≈I 54601
•	U31513005790 1 999 COUNTY UF APPLING COUNTY COMMISSIONERS CMAILMAN	99	065251006106 1 COUNTY OF CALLAWAY ROBERT G WRIGHT	99999
•	COUNTY COURTHOUSE BAXLEY GA	31513	ROUTE #1 COUTY COURT	NO 65251
•	020768005476 1 999 COUNTY OF CALVERT BD OF COUNTY COMMISSIONER	2 99	049720005390 1 CUUNTY UF CHARLEVOIX CUUNTY BOARD OF SUPER	99999 1 VI SORS
•	PRESIDENT PRINCE FREDERICK MD	20768	CHARLEVUIX	MI 49720
•	J36250005796 1 999 County of citrus Board of commissioners Wilbur Langely Chairman Langely	99	097501006345 1 COUNTY UF COLUMBIA BD DF CDUNTY COMMISSI RUBERT M HUNT	99999 IONER S
•	IVERNESS FL	36250	ST HELENS	GR 97501
•	033130006555 I 999 County of dade M R Stierheim County Manager-Netro dade Miani Fl	99 CTY 33130	029535006554 1 CDUNTY OF DARLINGTUN BOARD OF SUPERVISORS MCCUEN MORRELL CHAIRMAN CJUNTY CDUA (HOUSE DARIINGTON	99999 56 20535
•	065066006489 1 999 CLUNTY UF DASCONADE MARGLU LOTTMAN PRESIDING JUDGE RGUIF 1	99	017101005433 L CUUNTY OF DAUPHIN EMERGENCY PREPAREDNES CUURT HUUSE FRUNT_C MARKET STREET	99999 55
•	OWENSVILLE MU	62066	NUUN 7 HARRISBURG	PA 17101
•	OTTIGIO7073 I 999 County of Jauphin Larky Huchendoner Commissioner Po aux 1295	99	OL7101005434 1 COUNTY OF DAUPHIN DAUPHIN COUNTY COURTH EARL B HOFFMAN COUNTY COURTH	99999 IDUSE
•	HARRISBURG PA	17101	FRUNT & NARKET STREET HARRISBURG	PA 17101
Û	OL7101005437 1 999 CCUNTY OF DAUPHIN DAUPHIN COUNTY COURTHOUSE JOHN F MINNICH	99	055066005941 1 CUUNTY UF GODDHUE R L TANNER CUUNTY AUDITOR	99999
Ci l	CHAIRPERSON UD OF CONMISS FRUNT & MARKET STREET MARRISBURG PA	ICNER 17101	RED WING	MN 55066
ر	060450005402 1 9999 CUUNTY OF GRUNDY BOARD OF SUPERVISORS CHAIRMAN	99	044J77006128 1 County of Lake Dunald T Ezzune Assistant Prusecuting	99999
ں ا	GRUNDY COUNTY COURTHOUSE MURRIS	50450	105 MAIN STREFT PAINSVILLE	OH 44077
Ĺ	052406005646 1 9996 COUNTY OF LINN BCARD OF SUPERVISORS CHAIRMAN CFDAR RAPIDS 14	99	065069006488 1 CHUNTY OF MONTGUMERY FRED LUEKEY PKESIDING JUDGE	99994
£		~~~~	RUMAL RUUTE Rhinelang	Mil 65069

Y

^	the state of the second	ತಿಗೆ ಸ್ಥಾಳಿಯಲ್	KHINELAND	4u objov
$\hat{}$	068305005855 1 CGUNTY OF NEMAMA BOAKD OF COMMISSIONE WILLIAM SIEMENT	99499 85	029621005804 1 CUUNTY UF UCONEF JAMES M PHINNEY CUUNTY SUPERVISOR	99999
•	COMMISSIONER Nemana County Counth Auburn	NUSE NE 63305	WALMALLA	SC 29621
•	043452005779 1 Courty UF Uttawa Buaru OF Cuphissione President Purt Clintun	99999 RS DH 43452	072801005811 1 COUNTY CF POPE ERMIL GRANT ALTING COUNTY JUDGE POPF COUNTY COURTHOU RUSSELLVILLE	99999 ISE AR 72801
•	DEL201005850 1 County of Kuck Islam Buard of Suderviscas Paul Mulcaney Chairman Ruck Island County C	999999 ID GURTHQUSE	095414007101 1 Crunty 9F 537,RAMENTC BUARD OF SU ERVISOR 827 7TH SI ROOM 424 SACRAMENTU	99999 CA 95814
•	ROCK ISLAND 092101005377 1 CUUNTY UF SAN DIEGO BDARD DF SUPERVISDRS CHAIRMAN SAN DIEGO	IL 61201 99999 ; CA 92101	033450005903 1 COUNTY UF ST LUCIE WELDON B LEWIS CUUNTY ADMINISTNATUR 2300 VIKGINIA AVENUE ROGM 104 FURT PIERCE	99999 FL 33450
•	023683005668 1 CDUNTY OF SURRY BUAKD OF SUPERVISORS SMEPLOCK HOLMES CHAIRMAN SURRY COUNTY COURTHO	99999 USE VA 23683	068008005922 1 COUNTY OF WASHINGTON NOARD OF SUPERVISORS JACK JENSEN CHAIRMAN COUNTHOUSE-16TH & CO BLAIR	99999
•	080631006315 1 GUUNTY UF WELD BD OF CDUNTY COMMISS CHAIRMAN GREELEY	99999 IONERS CO 80631	055313005664 1 COUNTY OF WRIGHT BOARD OF COMMISSIONE O S DOUGLAS AUDITUR BUFFALC	99999 RS MN 55313
•	007094037904 I COURTER & CO INC VA/OC JAMES T SCHMIT I 50 MEADOWLANDS PKWY SECAUCUS	99999 NJ 07094	060521052984 I CRANE CO Director of Engineer 649 Executive Blvd Willow Brook	99999 ING IL 60521
•	OOII51053737 I CRANE CO INDIAN URCHARD PLANT D E KUHN UA MANAGER 203 HAMPSHIRF STHEET INDIAN URCHARD	99999 Ma 01151	063104052985 1 CRANE MIDWEST FITTIN D F BETZ VA NANAGER 1450 SUUTH SECOND ST ST LOUIS	99999 165 REET MU 63104
•	002093052986 1 GRUSBY VALVE & GAGE 43 KENDRICK STREET WRENTHAM	99999 Co Ma 02093	019468052203 1 Crouse nuclear energ W J Randles Vice President Upper Lewis Ruad Linfield	99999 97 Services PA 19468
•	019468056822 I CROUSE NULLEAR ENERG J T STEINMETZ UPPER LEWIS RUAD LINFIELU	99999 Y SERVICES PA 14468	043216064467 1 CVI INC MALLACE R RUSKIN VA MANAGEM PD RDX 2138 CULUMRUS	99999 OH 43216
•	092075066557 1 CYGNA ENERUY SERVICE 225 STEVENS ST SUITE 1J2 SOLANA BEACH	99999 S CA 92075	002210066214 1 Cygna Energy Service Geurge D Bradlfy Licensing Engineer 600 Atlantic Ave 247 duston	99999 5 H FL HA 02210

~

^	018917036575 1 999 UEL-AWARE JAMES M NEILL Assulate Luunsel Rox 51	19-1	019931006459 DELAWARE DEPT (JIV OF ENVIRON) THOPAS P ELCHU JIRFFIGR	1 99999 DF NATL RESCURCE Mental Control Er
•	DURLIN PA	13917	P-3 PUX 1401 DUVER	DE 19901
•	019901036537 1 999 UELAMARE DEPT OF NATL FES ENVIRONMENTAL CONTHOL HARKY UTTO MANAGER-TELMNICAL SERVICE	iga SUURCE	019901001261 Delamare Dept (F M Parkowski Deputy atturne Tatvall Duildi(L 99999 DF NATL RESOURCE Y GENERAL NG DF LOUGH
•	DOVER DE	19901	UUYER	DE 14401
•	019901005639 1 999 DELAWARE DEPUTY ATTORNEY JUNE D MACARTUR DEPUTY ATTORNEY GENERAL	199 GEN	019839003040 DELMARVA POWER F A CUUR SOO KING STREE	1 99999 6 LIGHT CO T
•	BOX 1401 DEVER DE	19901	MICHINGIUM	0E 14844
•	OBO203006316 L 999 UENVER DEPT OF LOCAL AFFA DIVISION OF PLANNING DIRECTION 1865 SHERMAN-615 CULUMBIN DENVER CO	199 1 - S 1	OWGERMOII714 DER BUNDESMINI REGULATION & L HELMUT SCHNUR ADMINISTRATOR REFERAT RS I 2 5300 BONN 1	1 99999 STER DES INNERN IC OF NUC REACT EH POSTFACH 170290 NGERMANY
	048226006501 1 999 DETRUIT EUISON CO FERMI 2 NPS WILLIAM J FARNER PRUJECT MANAGER 2000 SECOND AVE	199	048226006502 Detroit edison Wayne Jens 2000 Second Avi Detroit	L 99999 CO E MI 48226
	DETREIT MI	48226	04 83 34 0005 3 0	1 00000
•	DETROIT EDISON CU L E JUNES DIN PRUJ REC MGMT 2000 SECOND AVENUE		DETROIT EDISON PETER A MARQUA ESQUIRE 2000 SECOND AVI	
•	DETROIT MI	48226	DETROLT	MI 48226
•	048226079139 1 999 DETRUIT EDISON CO R G KATEICK 2000 SECUND AVE	1 39	048084006500 DETPOIT EDISON LARRY E SCHUERI 3331 WEST BIG	1 99999 CO Man Heaver Road
•	DETRUIT MI	48226	TROY	MI 48084
•	048226006228 1 999 DETROIT EDISON CO ENGINLERING & CONSTRUCTIO HARRY TAUBER VICE PRESIDENT 2000 SECOND AVE DETROIT NI	199 IN	048226004989 DETROIT EDISON ENGINEERING ANN HARRY TAUBER VICE PRESIDENT 2000 SECOND AVI OFTHOIT	3 99999 CO CONSTRUCTION E MI 49334
	012301081110 1 999	99	015077001767	1 99999
•	DÉVÉRSIFIÉD NUCLEAR INC Radiulogical Engng Stephen e miller Manager PD Bur 2000		DICK CORP J A LONG JA CUORDINATOR PU BCX 49 Smirpingort	DA 15077
•	SCHENECTADY NY	12301	SHEFFINGEGRE	
•	002138021665 1 999 DISZADLPIPE INC WALLACE B WEIGHT 55 WHELLER ST	99	022101065227 Dominion Engine M J Pantridge 6862 ELM STREE	1 99999 EERING INC I
•	CAMBHIDGE MA	02138	NCLEAN	VA 22101
Ű	049106005694 1 999 Dunald C Cuuk Nuclear Pla W G Smith Plant Manager Pu Bus 458	99 NT	094102005654 Dunchul Junes (Linda J Brown Lug Van Ness An 1914 Fluur	L 99999 BROWN & CLIFFORD /E
•	BRIDGMAN MI	49106	SAN FRANCISCO	CA 94102

†

A

•

•

4

-

•

.

٠

9

•

•

4

٤.,

.

		and the second	ALC: 1.1. A. P. P.
	068643331693 [99799 DOm (HEMICAL CC	045540081494 1 99444 UNH CHEMICAL CC	
	REALTUR OPERATIONS CATE	C # KGCMÊŘ HEALTON SUPERVISOR	
. <u>.</u>	RESEARCH MANAGER & CHAIRMAN	1604 BUILDING ME 4312	
	MIDEANJ MI 48640	-1356483	u
	048640081492 1 99979	090650052991 1 99999	
	DLW CHEMICAL CC Radiation Saffty Cmif	FOBERI E BOND	
	L W HAMPY	VICE PRESIDENT	
	1803 BUILDING	NURWALK CA 9065	0
	DASTSUUSZAAZ I AAJAA	DRESDEN NUCLEAR POWER STATION	1
-	PIPE FABRICATION DIVISION Renneth B Anderson	PLANT SUPERINTENDENT Rural route #1	
	GÉNERAL MANAGER 1115 GIENAN AVENUE	MORRIS IL 6045	0
-	MARIETTA OH 45750		
	060450006422 1 99999	029342080967 1 99999	
•	PLANT SUPERINTENDENT	CHEROKEE_NUCLEAR STATION	
	RURAL ROUTE #1 MORKIS IL 60450	N E BAILEY Site Manager	
4 .41		PU 80X 400 GAFENEY SC 2934	
	UUKE PUNER CD	DUKE POWER CO	
	NICHARD P BUGERT	NUCLEAR PRODUCTION DEPT K S CANADY	
-	PU BIT 1436 SENECA SC 29678	422 S CHURCH ST CHARLOTTE NC 2424	2
			-
	028242007094 1 99999	028242081033 1 99999	
•	STEAM PRODUCTION DIV	DESIGN ENGINEERING	
	6 A COPP PO 80X 33189	L C DAIL VICE PRESIDENT	
۲	CHARLOTTE NC 28242	PU BUX 33189 CHARLOTTE NC 2824	2
	028262001567 1 99999	029710080963 1 90000	-
	DUKE PUWER CO	DUKE PUHER CO	
	L C DAIL	J W HAMPTON	
	PO DUC 33139	PO BOX 392	
•	CHARLOTTE NC 28242	CLOVER SC 2971	0
	028242065237 L 99999 DUKE POWER CO	028031080896 1 99999 Diske Power Co	
-	STEAM PHUDUCTION	MCGUIRE NPS I E 2	
	PO BUX 33189	PLANT MANAGER	
	LMAKLUTTE NC 28242	PU BOX 488 CURNELIUS NC 2803	1
	028242006327 1 99999	028242028120 1 99999	-
۲	DUNE PUMER (U MEGUTAE MAS LE 2	DUKE POWER CO	
	J T MOURE	WILLIAM C PARKER	
	PC DUX 33139	422 SUUTH CHURCH ST BUX 33189	
-	GMAKLEITTE NC 28242	CHARLOTTE NC 2824	Z
	028242000453 1 99999 Duke Puwer CD	029710007427 1 99999 DUKE POWER CO	
-	WILLIAM PORTER	J C RUJERS	
_	CHARLOTTE NC 28242	CLUVER 5C 2971	0
	029678003050 1 99999	U28242000622 I 99999	
	DURE POWER CO DCONEE NUELFAK STATION	HIKE POWER CO	
	J E SMITH STATISME MAGALER	VILE PRESIDENT-NUCLEAR PROD	
<u> </u>	PO 808 1439	922 SUUIN LNUKEN SI 908 33189	
	arurra 20 56 58	UNARLUITE NC 2824	Z

. 1

^

1

 \mathbf{a}

0

0

Ø

Ű

Ø

Ο

Э

C

.

U

J

4

9. .

•	PU	LHARLUTTE NG 28242
•	028242081247 1 99999 DURE Pliate Cli M 38 THERE Cli	015077003061 1 59999 DUQUESNE LIGHT CU MANNEN VALLEY DIMER STATION
•	VICE PRES-NUCLEAR POWER DEPT 422 S CHURCH ST CHARLUITE NC 28242	F J BISSERT P L BUX 4 SHIPPINGPORT PA 15077
•	015077064696 1 99999 Dugursne Light Cu	015217005713 1 99999 DUQUESNELIGHT_CU
•	J J LARFY VICE PRESIDENT-NUCLEAR PO 50x 4	JOHN J CAREY VI/L PRESIDENT 435 SIXTH AVENUE
	015077007012 1 99999	015077079396 1 99999
•	DUQUESNE LIGHT CO QUALITY ASSURANCE DEPT C.E.EWING	DUQUESNE LIGHT CO QUALITY ASSURANCE DEPT C E EWING
•	MANAGER PG BUX 186 Shippingpurt pa 15077	GUALITY ASSURANCE MANAGER Pu Box 186 Shippingport pa 15077
•	015077006546 1 99999 Duguesne light cu Nuclear Operations	015205079395 1 99999 Duquesne Light Co E F Rurtz
•	T D JUNES MANAGEN PD BOX 4	DIRECTOR OF LICENSING RUBINSON PLAZA BLDG #2-RTE 60 SUITE 210
-	SHIPPINGPORT PA 15077 015205007013 1 99999	PITTSBURGH PA 15205 015077080919 1 99999
•	DUQUESNE LIGHT CO Bruhn Valley 2 E F Kurtz	DUQUESNE LIGHT CO W S LACEY Chief Engineer
•	ROBINSON PLAZA BLDG 2 RT 60 Pittsburgh pa 15205	PU BOX 4 SHIPPINGPORT PA 15077
•	015219030948 1 99999 DUQUESNE LIGHT CO R M MAFRICE MUCLER EN ENCLOSED	015219081569 1 99999 DUQUESNE LIGHT CO R MARTIN NUCLESNE LIGHT CO
•	435 SIXTH AVENUE PITTSBURGH PA 15219	435 SIXTH AVE PITTSBURGH PA 15219
•	015077062825 1 99999 Duquesne light co Shippingport atomic power sta J V McGee Station office manager	015077079147 1 99999 Duquesne light CD Nuclear Safety & Licensing DPT J D Sieber Manager
•	BOX 57 Shippingport PA 15077	PU BOX 4 Shippingport PA L5077
•	015205079394 1 99999 Duquesne light CD H N Siegel	015205007014 1 99999 Duquesne Light CO Beaver Valley 2
•	DIRECTOR OF ENGINEERING ROBINSON PLAZA BLDG #2-RTE 60 SUITE 210 PITTSBURGH PA 15205	12 M SIEGEL DIRECTUR UF ENGINEERING RUBINSUN PLAZA BLDG 2 RT 60 PITTSBURGH PA 15205
•	015077079148 1 99999 Dugulsne light co Beaver Valley NPS-Nuclear	015205006499 1 99999 Dughesne Light Co H J #Ashabaugh Dughesne Light Co
•	MANAGER-NUCLEAR ENGINEERING PO ROX 4 Shippingpopt pa 15077	RUBINSON PLAZA BLDG Z-ROUTE 60 SUITE 210 Plitsburgh PA 15205
•	015205079434 1 99999 DUGUESNE LIGHT CII NUELESNE LIGHT CII	015077005709 1 99999 DUDUESNE LIGHT CU NEAVEN VALLEY DUED STATION
	R MASMABAUGH PROJECT MANAGER DI ULAT MANAGER	A P WILLIAMS STATION SUPERINTENDENT
•	PITTSBURGH PA 15205	SHIPPINGPORT PA 15077
•	UI205079460 1 99999 Duguesne Light (0 Beaver Valley TWO Project Earl J Wollever Rubirson Plaza Slog 2-pa RT 60	DIS205001563 1 99999 Diguesne Light Co Nuclear Cunstruction Earl J Wogleyer Vice President
~	SUITE #210 PITTSBURGH PA 15205	RUBINSUN PLAZA NO 2 RTE 60 PITTSBUPG PA 15205

	ALALMA	LH 44202	1 • • • • •	
	030042033060 1 EBASCU SERVICES INC	99 994	011375000979 1 EBASCO SERVICES INC	99999
-	R H DILTZ SUPERVISING ENGINEER 145 TLCHNDLUGY PAPK NORCRUSS	GA 30092	110-34 73RD ROAD FOREST HILLS	NY 11375
C	01004805553J 1 EBASCU SERVICES INC DIONE KROBITZKY	99999	010005006264 1 EBASCO SERVICES INC RAY MATZELLE DAVISET MANAGER ACNOL	99999
^	Z MUKED TRADE CENTER BOTH FLOOR NEW YORK	NY 10048	19 RECTOR ST NEW YORK	NY 10005
<u>^</u>	030092026013 1 EBASCU SERVICES INC JAMES C O'HARA	99999	010048038050 1 EBASCO SERVICES INC NUCLLAR LICENSING DEI	99999 90 70
^	LUNSULTING ENGINEER 145 TECHNOLOGY PARK/ NGKCRUSS	GA 30092	THO WORLD TRADE CENTI 15TH FLOOR New YORK	ER NY 10048
^	011753022462 EBASCU SERVICES INC L PHELAN LIBRAHIAN	99949	010006007978 1 EBASCO SERVICES INC L REICHEL VICE PRESIDENT NUCLEA	99999 Ar
•	I JERICHO PLAZA JERICHO	NY 11753	2 RECTOR STREET New York	NY 10006
r i	010048064464 EBASCU SERVICES INC MATL ENGR & QA B E TENZER VICE DESENTENT	99999	094612081439 1 ECHU ENERGY CONSULTAN Manager Of Quality A: 1499 Franklin St Daki And	99999 NTS SSURANCE
0	TWD WORLD THADE CENT	ER NY 10048	DAREAND	
•	015219006305 1 ECKENT SEAMANS CHERI BARION & CGWAN 600 GRANT STREET 42ND FLUOR PITTSPUKGH	99999 N MELLOTT PA 15219	019355008722 1 ELOLAIRE SYSTEMS INC ENGINEERING CARL L NEWMAN VICE PRESIDENT TWU COUNTRY VIEW RDAN	999999
•	020036079255 1	99999	MALVERN 094598057367 2	PA 19355 99999
" "	EDISON ELECTRIC INSI Federal Regulatory Ci Larry Logan Federal Regulatory Ci 1111 Nineteenth St Ni Aschittor	DURDINATE	EDS NUCLEAR INC LIBRARIAN 350 LENNUN LANE WALNUT CREEK	CA 94598
•	094104026885 1 EDS NUCLEAR INC LIBRARY	99999 99999	094598057030 1 EDS NUCLEAR INC PROJECT DEVELOPMENT	999999
•	220 MONTGOMERY STREE	T CA 94104	350 LENNON LANE WALNUT GREEK	CA 94598
b	094104053000 1 EDS NUCLEAR INC R A AYRES Da Manager	9 99 9 69	030092066906 1 EDS NUCLEAR INC DEL BRITT	99999
•	220 NUNTGOMERY ST SAN FRANCISCO	CA 94104	333 TECHNOLOGY PARK NORCRUSS	GA 30092
۳	094104005480 1 EDS NUCLEAR INC R A FORINEY 201 MUNICONERY ST	99999	094104064767 1 EDS NUCLEAR INC C M FRANK III	99999
¥	SAN FRANCISCO	CA 94104	220 PONTGOMERY STREE	TCA 94104
v	094104064768 1 EDS NUCLEAR INC N HALLADAY RECURDS ADMINISTRATO	99999 K	094104065746 1 EDS NUCLEAR INC Auvanced Analysis di Arun Seiken Suere	99999 V
Ú.	SAN FRANCISCO	CA 94104	220 MUNTGOMERY ST SAN FRANCISCO	CA 94104
	083415063554 1	99999	083401064611 1	99999

~

نے

.

_	P & PAT TOL -	10 83415	РО 50X 1625 15А КОО 10440 FALLS	10 83401
•	083415031484 1 EGG IDAHO INC INEL TECHNICAL LIB Refnt & Jacubsen	99999 Raky	083415064221 1 EGEJ IDAHO INC QUALITY DEPT M J KIRSCHENMANN	99449
•	LIBRARY SUPERVISOR PU BOX 1625	10 83415	PU BUX 1625 1DAHU FALLS	10 83415
•	104HC FALLS 083401064744 1 EGEG 104HQ INC	99999	U83401062537 1 EGEG IDAHO INC DAVID B VAN I FUVEN	99999
•	F J ROCSIS 111 CONFIG DOC CONTROL PU BUX 1625 IDAHU FALLS	MGR 10 83401	PU BOX 1625 TRA 649 IVAHU FALLS	ID 83401
•	083415062748 1 EGEG IDAHO INC POWER REACTORS SPE	99999 CIAL PRCJECT	046320901446 1 Eichnorn Morrow & Ei William H Eichnorn	99999 ICHHGRN
•	M S VARGO Manager Po Box 1625 Idahu Falls	ID 83415	5303 HOHMAN AVENUE HAMMOND	IN 46320
	OSWITZO57566 1 EIDG AMT FUR ENERG	99999 IEWIRTSCHAFT ISIGN	094303064766 1 ELECTRIC POWER RESEA	99999 Arch Inst
•	ROLAND NALGELIN ADMINISTRATOR 5303 WUERENLINGEN	SWITZERLA	JOSEPHINE FISHER 3412 Hillview Avenus Box 10412 Palo Alto	E Ca 94303
•	094303054728 1 ELECTRIC PUWER RES NSAC RUBERT LEYSE PRUGRAN NANAGER PD BOX 19412	99999 EARCH INST	020008079698 1 EMBASSY OF GREAT BR ALISON COULTER 3100 MASSACHUSETTS WASHINGTON	99999 11AIN AVE NW DC 20008
•	020037000346 1 EPBASSY OF JAPAN SCIENLE SECTIUN TETSUMISA SHIRAK 600 NEW HAMPSHIRS SUITE 900 MASHINGTON	99999 AWA AVE NW DC 20037	OITALYOOBO74 1 ENEA External relations Piero vanni Administrator Viale regina marghe OO190 Rome	99999 DISP RITA 125 ITALY
٠	015222062525 L ENERGY CONSULTANT CORPUPATE LIBRARI 121 SEVENTH STREE	99999 S. INC AN T	015222065752 1 ENERGY CONSULTANTS OPERATION SERVICES LIBRARIAN	99999 INC DIV
<u>,</u>	PITTSBURGH	PA 15222	800 PENN AVENUE PITTSBURGH	PA 15222
	020850051943 1 ENERGY INC KAY DOVE	99999	083402002752 1 Energy Inc Technical Library	99999
	118 MONROE ST 430 ROCKVILLE	MD 20850	PO BOX 736 Idaho Falls	ID 83402
	002154032003 1 Energy Research (Rubert Capstick	99999 GROUP INC	002154001413 1 Energy Research Gru Marc W Guldsmith	99999 10P INC
	400-1 TUTTEN PCN Waltham	D ROAD MA 02154	400-1 TOTTEN POND F WALTHAM	MA 02154
υ	095159055107 1 Enter Curp Donald Baul	99999	095126054738 1 Entor Curp Jack Meber	99999
•	PO BOX 28450 SAN JOSE	CA 95159	1885 THE ALAMEUA San Jose	CA 95126
	016801028533 1 Environ Coalitio Judith Johnsrud	99999 N ON NUC POWER	016801001495 1 Environ coalition Chauncey Kepfurd	99999 Dn Nuc Power
•	433 ORLANDO AVE State cullege	PA 16801	433 ORLANDO AVENUE State college	PA 16801
	040140045477 1	00909	040475057097 I	90000
UK TEURSES		A MARKET AND A MARKET		
--	------------------------------	---	-----------------------------	
AGMIN1ST6A100 P0 B0X 268 00101 HELSINKI 10	FINLAND	DIRECTOR DIRECTOR U BOX 268 OVIDI MELSINKI 10	FINLAND	
● 050150053007 1 ● FISHER CONTROLS CC J R BOVEE_	99999	050158021209 1 FISHER CONTROLS CO PUY R BRUDIN LNG CODES E STOS	9994 9	
PO BOX 190 MARSHALLTONN	IA 50158	PU BUX 11 MARSHALLTOWN	IA 50158	
O91720057035 1 FLEET ANALYSIS CENTER GIDEP	99999	032304005799 1 FLORIDA ATTORNEY GENE DEPT OF LEGAL AFFAIRS	99999 RAL	
DILLIAM ARNITZ PROGRAM DIRECTCR CORONA	CA 91720	TALLAHASSEE	FL 32304	
032301005802 1 FLURIDA CLEARINGHOUSI CTATF PLANNING & DEVI	99999 ELOPMENT	032301005798 1 FLORIDA DEPT OF ENVIR PUWER PLANT SITING SI	99999 R REG ECTION	
CAPITAL BLOG EXEC OF TALLAHASSEE	FC OF GOV FL 32301	2600 BLAIR STONE ROAL TALLAHASSEE	FL 32301	
O32301004853 1 FLORIDA DEPT OF ENVIO HAMILTON OVEN JR ADMINISTRATOR ADMINISTRATOR	9999 9 R Reg	032301012054 3 FLORIDA DEPT OF HLTH RADIOLOGICAL HEALTH LYLE JERETT ADMINISTRATOR	99999 5 REHAB PROGRAM	
• TALLAHSSEE	FL 32301	1317 WINEWOOD BLVD TALLAHASSEE	FL 32301	
O32231009733 1 FLORIDA DEPT OF HLTH JOHN LANHAM PUBLIC HEALTH PHYSIC	99999 6 REHAB 15T	0323040057977 I FLORIDA OFFICE OF PU JACK SCHREVE HOLLAND BLDG	BL COUNSEL	
PO BOX 210 JACKSONVILLE	FL 32231	TALLAHASSEE	FL 32304	
033450080956 1 FLGRIDA POWER & LIGH ST LUCIE 2 NPS B J ESCUE PLANT MANAGER	99999 IT CO	033408081117 1 FLORIDA POWER & LIGH Advanced Systems & T Robert Euhrig Vice President 90 Hox 14000	99999 IT CO ECHNOLOGY	
 PO BOX 128 FT PIERCE 	FL 33450	JUNG BEACH	FL 33408	
O33454064618 1 FLORIDA POWER & LIGH J P LEWIS WAULT CUSTODIAN	99999 IT CO	FLGRIDA POWER & LIGH D F MORGAN MGR-CURP RECORDS SER	IT ČÓ IVICE	
PO BUX 128 PSL-1 QC FT PIERCF	FL 33454	PO BOX 529100 MIAMI	FL 33152	
D33152006241 1 FLORIDA POWER & LIG ADVANLED SYSTEMS & 1 ROBERT & UMKIG	99999 HT CO LECHNOLOGY	033450080957 L FLORIDA POWER & LIG ST LUCIE 1 NPS NAT WEEMS NAT WEEMS	IT CO	
VILE PRESIDENT PU BOX 529100 MIAMI	FL 33152	PO BOX 128 FT PIERCE	FL 33450	
• 033454079153 1 FLCRIDA POWER & LIG ST LUCIE UNITS 1 & C. M. METHY 1000	99999 ht CO 2	033101006556 1 FLORIDA POWER & LIG Turkey Point NPS Henry YAEGER Diant Manager	99999 HT CO	
PLANT MANAGER PO BOX 128 FORT PIEKCE	FL 33454	PO BOX 013100 MIAMI	FL 33101	
• 033152064656 1 FLORIDA POWER & LIG K N YERK DOCUMENT CONTROL SU	99999 HT CO PVR	033733079322 5 FLORIDA POWER CORP NUCLEAR LICENSING MANAGER PU BUX 14042	99999	
MIAMI	FL 33152	MAC H-2 ST PETERSBURG	FL 33733	
O33733005795 1 FLCRIDA PGMER CORP S A BRANDIMURE VICE PRESIDENT & GL P€ B0x, 14042	99999 N COUNSEL	033733005803 1 FLURIDA PUWER LORP NNCLEAR OPERATIONS J & HANCUCK PU BOX 14042 MAC H-2		

~	FLLEY , LAFORE STEVEN E REANE 777 FAST DISCUNSIN AVENUE MILWAUREE WI 53202	NEFLANS UN BUARU LEUYU K MARBEI 19142 S BAKERS FERKY KOAD BUHING OR 97009
~	075001053010 1 99999 FORNEY ENGINEERING CU QUALITY CONTROL RAY NUEPHY PG BOX 189 3405 WILEY PUST RD ADDISUN TX 75001	DU7039053011 I 99999 FOSTER WHEELER ENERGY LORP W J FINAN UUALITY ASSURANCE 110 SOUTH DRANGE AVENUE LIVINGSTON NJ 07039
~	007039056039 1 94999 FGSTER WHEELER ENERGY CORP CHARLES MASH 110 SLUTH ORANGE AVENUE LIVINGSTON NJ 07039	017110005449 1 99999 FDX FARR & CUNNINGHAM JJRDAN D CUNNINGHAM 2320 N SECUND SI HARRISBURG PA 17110
•	QQ2Q35064465 L 99999 FLXBURD CO SYSTEMS OPERATIONS SYSTEMS O ALITY SERVICE MECHANIC ST D961 Envented MA 02035	002035053012 3 99999 FUXBURD CD NEPGNSET PLANT MARGLU D DENZER JR 38 NEPONSET AVENUE DEPT. 137 FOXBORD MA 02035
•	OC2333064466 1 99999 FOXERO CO GUALITY CONTROL ENGINEERING GEORGE DUSIMAN 6JG N BEDFURD ST D-784 PRIDCEMATER MA 02333	015235011881 1 99999 FRAMATOME C70 WESTINGHOUSE NES CLAUDIA GRANCHE 300 PENN CENTER BLVD SUITE 600 PITISBURGH PA 15235
•	CAST DAIDGE MATCH AN OLDED FRANKLIN INSTITUTE FRANKLIN RESEARCH CENTER S P CARFAGNO NRC CGNTRACT 20TH & RACE STS PHILADELPHIA PA 19103	020037006298 1 99999 FRIED FRANK HARRIS ET AL HAROLD P GREEN THE WATERGATE 600 600 NEW HAMPSHIRE AVE NW SUITE 1000 WASHINGTUN DC 20037
•	019065006440 1 99999 FRIENDS OF THE EARTH DELAWARE VALLEY ROBEKT L ANTHONY 103 VERNON LANE BOX 186 MOYLAN PA 19065	094111006026 1 99999 FRIENDS OF THE EARTH ANDREW BALDWIN 124 SPEAR ST SAN FRANCISCO CA 94111
•	043603007102 1 99999 Fuller & Menry Paul M Smart 300 Madison ave PO Box 2088 TGLEDU DH 43603	OBRAZIO28680 1 99999 FURNAS CENTRAIS ELECTRICAS SA DIV BIBLIUTECA KUA REAL GRANDEZA 219 2038 RIO DE JANEIRO 22283 BRAZIL
•	085010052916 1 99999 GARRETT TURBINE ENGINE CO LIGKARY ILL SOUTH J4TH STREET PHUENIX AZ 85010	094123064672 1 99999 GATEWAY SYSTEM INC H L ANDERSON PRESIDENT 2271 UNION ST SUITE 5 SAN FRANCISCO CA 94123
•	002149053014 l 99999 GAULIN CORP 44 GARDEN ST EVERETT MA 02149	092138001584 1 99979 GENERAL ATOMIC CO NUCLEAR MATLS CONTROL DIV LICENSING ADMINISTRATOR P 0 BUX 81608 SAN DIEGO CA 92138
•	020006006319 1 99999 GENEFAL ATUMIC CO EAST CDAST OFFICE JAMES B GRAHAM 2021 K STRFET NW SUITF 709 WASHINGTON DC 20006	092138064750 1 99999 GENERAL ATUMIC CO L M RABJOHNS SUPERVISCM REC MGMT PU BUX 81609 SAN DIEGO CA 92138
u	092138001466 1 99999 General Atumic Co Frank a Wafner P 0 Box 81608	006340020097 2 99999 GENERAL DYNAMICS CORP FLLCTRIC BUAT DIVISION MAINFILE-DEPT 633

)

ر

SULLE JUY	1 2.1406		
MASHINGTON	00609	006340020097 2	99999
GENERAL ATUNIC CO	****	GENERAL DYNAMICS CORI	2 2 N
FRANK A WAPNER	_	MAINFILE-DEPT 633	
SAN UIEGU	CA 92138	GROTON	CT 06340
		020202001616	00000
095125001598 I	44494	GENERAL ELECTRIC CO	
NUCLEAR ENLAGY DIV		MANAGER-LICENSING &	NAM
175 CURTNER AVE		P 0 80X 780	
MC 528 SAN JUSE	CA 95125	WILMINGTON	NC 28402
006602053735 1	99999	019142053020 1	99999
GENERAL ELECTRIC CO	S NEPT	SWITCHGEAR BUSINESS	DEPARTMENT
1285 BUSTON AVENUE	CT 06407	C E BEDFORD General Hanager	
BRIDGEPGKI	C1 00002	6901 ELMWOOD AVE	PA 19142
			00000
001910053023 1 CENERAL ELECTRIC CO	99999	GENERAL ELECTRIC CO	77777
R F BUCKLEY		LAWRENCE J CHGCKIE	
40 FEDERAL STREET		M/C 827	CA 95125
WEST LYNN	WY OLAIO	SAN JUSC	
028402053025 1	99999	019142055290 1	99999
GENERAL ELECTRIC CO	MENT	GENERAL ELECTRIC CO PWR SYS MGMT BUS DEP	T
L W DOAFE		E J FIERKO	
POST OFFICE BOX 780		6901 ELMHOOD AVE	
WILMINGTON	NC 28402	PHILADELPHIA	PA 19192
095125007112 1	99999	052601053024 1 GENERAL FLECTRIC CO	99999
D L FUREMAN		SWITCHGEAR BUSINESS	DEPARTMENT
PROJECT MANAGER	C 682	NGR BURLINGTON PLANT	OPERATION
SAN JUSE	CA 95125	BURLINGTON	IA 52601
040531045305	00000	095125064718 I	99 999
GENEPAL ELECTRIC CO		GENERAL ELECTRIC CO	
814 CUMMERCE DR		SPECIALIST OP SUP EN	ſG
SULTE 305 Dak Brock	IL 60521	714	CA 05175
		SAN JUSE	CR 77127
095125006263 1	99999	OZ8402064770 1 General Electric Co	43333
RAY LEBRE	~ 5	F M MALIGA MGR DUAL VERT & RECI	DRDS
175 CURTNER AVE	0J CA 05175	PO 80X 780 M/C 872	NC 28402
SAN JUSE	CA 42152	#1CUIVAIN4	
095125006442 1	99999	094086066484 1	99999
GENERAL ELECTRIC CO		GENERAL ELECTRIC CO LICENSING & REACTOR	SYSTEMS
PROJECT MAMAGER	1 CONE 395	W W PHELAN	
SAN JUSE	CA 95125	PO 80X 508	
		SUNNT VALC	00000
017355063069 1 GENERAL ELECTRIC CO	99999	GENERAL ELECTRIC CO	77777
POWER SYS HEAT BUSI	NESS DEPT	GLENN SHERWOOD 175 CURTNER AVENUE	
205 GREAT VALLEY PK	Y	N/C 682	CA 95125
MD-1106 MALVERN	PA 17355	SAN JUJE	
095125006384 1	99999	020014039641 1	99999
GENERAL ELECTRIC CO	SION	GENERAL ELECTRIC CO JOAN STARR	_
JUHN M SKARPELDS		7910 WOODMUNT AVENU	E
NCI17	<i>.</i>	RETHESDA	MD 20014
SAN JOSE	CA 95125		00000
095125065236 1	99999	061701038048 1	44444

. . .

E

	1. A. A				
~	GENERAL ELECTRIC CC NUCLEAR ENERGY DIVISION JUHN N SKANPELGS		JEJI4J39641 L General Electric CD JOAN STARR 7910 WCCDMUNT AVENUE	** * *	F 9
•	ITS CURTNER AVENUE NC117 SAN JOSE CA 9	95125	SULTE 203 RETHESDA	MD 2	0014
•	095125065234 1 9999 General Electric CC Howard T Watanabe	99	D61701038048 1 GENERAL ELECTRIC CO CHARLES WETTER	9999	9
•	MG 682 SAN JUSE CA 9	5125	BLCOMINGTON	IL 6	1701
•	074036055265 1 9999 GENERAL ELECTRIC CO NUCLEAR ENERGY	99 (D37415026373 1 GENERAL PHYSICS CORP HATTANOGGA DIVISION	9999	9
	ROUTE I BOX 378		ECHNICAL INFORMATION	CENT	ER
•	INULA UK 7	4036 (HATTANOGGA	TN 3	7415
٠	021044079621 1 9999 GENERAL PHYSICS CORP TECHNICAL INFERNATION CENT N L KNIGHT ADMINISTRATOR	ier R	20144064653 1 ENERAL PHYSICS CORP R TOUZIN ENIOR SPECIALIST	9999	9
•	CCLUMBIA ND 2	1044 ^Ĉ	OLUMBIA	ND 2	0144
•	030334055207 1 9999 GEORGIA DEPT OF HUMAN RESO RADIOLOGICAL HEALTH SECTIO BOBBY RUTLEDGE 1256 BRIARCLIFF RD NE ROUM 425-5 ATLANTA CA 3	IN E	3033403C991 1 EORGIA DEPT OF NATL NVIRONMENTAL RADIATI AMES C HARDEMAN 70 WASHINGTON ST SW 9, 855	9999 RESOUI DN	9 RCES
	un j	0334 A	ILANIA	GA 3(3334
•	030334004860 1 9999 GEORGIA DEPT OF NATL RESOU ENVIRCNMENTAL PROTECTION D JLLEDBETTER	90 RCESG IVN R	30332000551 1 Edrgia institute df Uclear research cent S kirkland	999999 TECH ER	•
•	270 HASHINGTON ST SW ATLANTA GA 30	0334 ^R	EACTOR SUPERVISOR TLANTA	GA 30	332
•	030332080964 1 9999 Georgia Institute of Tech A P Sheppard Acting VP for Research 225 North Avenue Atlanta GA 30	9 0 G 2 0332 A	30334005793 1 Eurgia DFC Planning H Marles H Badger 70 Mashington Street DDM 610 Tlanta (99999 6 BUDO 5W	ET
-					
•	030302005794 1 99999 GEURGIA POWER CO J T BECKHAM VICE PRESIDENT-NUC GENERATI PU BOX 4545 ATLANTA GA 30	9 0 G N ION J 9302 3	30302008706 1 Eurgia Pomer Co Uclear Generation T Beckhan Jr D Box 4545 33/16	99 999)
		A.	TLANTA C	FA 30	302
•	031513080897 1 99999 GEORGIA POWER CO C E BELFLOWER SITE OA SUPERVISOR	7 0. Gi P(30302001549 1 EORGIA POWER CO DWER SUPPLY UNG DUITON	99999	ł
•	BAXLEY GA 31	1513 Pi	ICE PRESIDENT-PRODUCT D BOX 4545 ILANTA (MGNT	302
•	030302006493 1 99999 GEURGIA POWER CO D D FCSTER BO BOY SEC	9 0: Gi D	30302080901 1 Eurgia Power Co D Foster	99999	
•	ATLANTA GA 30	0302 Pi A1	RUJECT GENERAL MANAGE) BDX 4545 [Lanta g	R A 30	302
•	030830080903 1 99999 GEURGIA PUNER CO H H GREGORY	O O I Gi	0630080902 1 ORGIA POWER CO D_GROOVER	99999	
•	PO BOX 282 WAYNESBORD GA 30	IK QA PL 1830 WA	SITE SUPERVISOR 3 BOX 282 VNESBORD G	A 30	830
•	030302001431 1 99999 GEORGIA POWER CO L T GULWA CMIFF NUCLEAP ENCINEED	O3 GE R	0302080904 1 Druia Power Co J Kelly	99999	
		· •			

θ 6

80

J.

		ULUHUIA PLWEN CL
	L T GUCHA	R J KELLY FXFCUTIVE VICE PRESIDENT
	CHIEF NUCLEAP ENGINEER PU BUX 4545 ATLANTA GA 30302	PC BCX 4545 ATLANTA GA 30302
	030830004711 1 99999 Georgia Powlr Cu Yugilf.NP5	031513005791 1 99999 Geurgia Power Co Ejwin I Hatch Plant Max Manry
•	M MARKT Plant Marager PD BUX 282 GA 30930	PO BUX 442 BAXLEY GA 31513
-		030830064729 1 99999
9	O31513080898 I 59999 Georgia Puwer Co Match NPS I & 2 M C NIX	GECRGIA PUWER CO B R QUICK Site Records MgMt Supvr Bo Boy 242 PIANT VOGTIF
•	PLANT MANAGER PO RUX 439	WAYNESBORD GA 30830
	BAXLEY GA 31513	
9	020830064731 1 99999 Georgia Power Co W L Shuman	030334080966 1 99999 Gedrgia Public Service Con Hugh S Jordan
	SITE DUCUMENT SUPERVISOR	244 WASHINGTON ST SW
	WAYNESBORD GA 20830	ATLANTA GA 30334
•	OWGERMO81467 1 99999 GESELLSCHAFT FUR REAKTURSICHE	010001001192 1 99999 GIBBS & HILL INC
	E LINDAUER GLOCKENGASSE 2 HOERMAN	RALPH F DEANGELIS
	5000 CULUGNE I BUCKHAN	11 PENN PLAZA NY 10001
NN		
4	010001022168 1 99999 GIBBS & HILL INC N.N.KEDDIS	GIBODIOBIS77 I 99999 GIBBS & HILL INC D C PURDY MANAGER-ADVANCED TECH
	MANAGER QA 393 7TH AVE	11 PENN PLAZA
	NÉW YORK NY 10001	NEW YORK AT LOUDI
	095110064501 1 99999 GIBBS & HILL INC	010001006517 1 99999 GIBBS_E_HILL INC
	MESTERN REGIONAL OFFICE JACQUELYN L RICHARDSON SR TECH ADN SPECIALIST	H R RUCK 393 SEVENTH AVE New York Ny 10001
۲	SAN JOSE CA 95110	
3	0:0001008707 1 99999 GI885 & HILL INC PETER H SMITH	019603008708 1 99999 Gilbert Associates Inc James R Stoudt
•	PŘESIDENT 393 SEVENTH AVENUE New York Ny 10001	PRESIDENT PU BOX 1498 READING PA 19603
		010(03007083 3 00999
•	019603053027 1 99999 Gilbert Associates Inc Ind C Energy Res DIV	GILBERT ASSOCIATES INC LIBRARY BUDTH
	N H INAFFAS 525 LANCASTER AVENUE LAZZO PA 19603	P 0 BOX 1498 PA 19603 Reading Pa 19603
		019603066712 1 99999
	GILBERT ASSOCIATES INC	GILBERT COMMONWEALTH
	RITA WORTH PD BDX 1498	RECORDS HEAT CONSULTANT
	READING PA 19603	PO 80X 1498 Reading PA 19603
i T		01040404436 1 00009
	GILBERT COMMONWEALTH R O PUTTS R D PUTTS	GILRERT COMMONWEALTH Information Mgnt Dept A M DRATHER
-	PO BOX 97 DUPERAISUR	MANAGER POL ROX 1498
:	PERKT UN 44081	READING PA 19606
	070116001282 1 99999 Gillespie & Jünes Luke Fontana	031520026178 1 99999 Glynn County Health Dept Jerky Morris Radiation Safety Officer
-		

t. ...

ບ ຍ ບ

	A second s	way that a watter of the second
•	LILENSING MATAGEN 100 INTERSPACE PARKWAY PARSIPPANY NJ 07054	IN INTERPACE PARKAAY PARSIPPANY NJ 07054
•	007054080995 l 99999 GPU SEPVICE CURP N & PASTOR	019603064751 1 99399 GPU SERVILE CORP L B SMATTUCK CORMINICATION/STANDARDS
•	PRDJECT MANAGER 260 Cherry Hill RD Paksippany nj 07054	PC BUX 1018 PA 19603 READING PA 19603
•	048640000968 1 99999 GREAT LAKES ENERGY ALLIANCE MARY SINCLAIR DOLME	OGREECO57557 1 99999 GREEK ATOMIC ENERGY COMMISSION NKC DEMUKRITOS IDWN KOLLAS
•	STIL SUMMERSEL DELVE MI 48640 MIDLAND	NUCLEAR PEGULATORY SERVICES AGHIA PARASKEVI ATTIKI GREECE
•	074103005187 1 99999 Green Feldman Hall & Woudard Inseph R Farris	090040053034 1 99999 GREER HYDRAULICS INC DONALD DIPADLA
•	ESO BIG ENTERPRISE BLOG TULSA DK 74103	SUPERVISOR CONTRACTS ADMINISTR 6500 E SLAUSON AVENUE CITY UF COMMERCE CA 90040
•	008876064658 1 99999 GULF & WESTERN MFG CO TAYLOR BONNEY DIV N B GRUHUL GA MGG-IAYLOR FORGE STAINLESS	052803055300 1 99999 Gulf & Western MFG CD Eagle Signal Division Jay Hankinson Q A Manager The Ecceptic Street
•	READINGTON KO Somerville nj 08876	DAVENPORT IA 52803
•	060690053033 1 99999 GULF E WESTERN MFG CO ENERGY PRODUCTS GROUP DAVID L SMIRA P O ROX 485 CMICLEG 11 60690	CULF & MESTERN MFG CO FLUID SYSTEMS JOHN F WALTER GENERAL MANAGER 25 GRAYSTONE STREET
•		MÁRWICK RI 02886 070775062571 1 99999
•	GULF POWER CO B M GUTHRIF EXECUTIVE VICE PRESIDENT	GULF STATES UTILITIES CO RIVER BEND STATION LIBRARY PU BOX 220 ST FRANCISVILLE LA 70775
•	PENSACULA FL 32520	
•	077704064583 1 99999 Gulf States utilities CO T E Amerine Supy Nuclear Document CTR	077704001519 1 99999 Gulf States utilities CO R Bend Nuclear Group William J Cahill Sem Jur Vice Paesident
•	BEAUMUNT TX 77704	PU BOX 2951 BEAUMONT TX 77704
	077704081877 1 99999 Gulf States Utilities Co	077701064636 1 99999 Gulf States utilities CO R & Ramsey
	WILLIAN J CAFILL SENIUR VICE PRESIDENT PO BOX 2951 TX 77704	NUČLEAR ŠTAFF ASSISTANT PO BUX 2951 BEAUMUNT TX 77701
	077704006483 1 99999 GULF_STATES_UTILITIES CO	077701064765 1 99999 GULF STATES UTILITIES CO
•	WILLTAN J REED Director-Nuclear Licensing Po Box 2951	NUCLEAR RECORDS REPRESENT
	BEAUMUNT TX 77704	BEAUMONT TX 77701
٠	092627053035 1 99999 Gultun Industries Inc Gultun S C D Bruce Fitzhugh President	007029053513 1 99999 GUYON ALLOYS INC GEURGE GRUNTHALER 959 SOUTH FUURTH STREET HARRISON NJ 07029
•	1644 WHITTIER AVENUE COSTA MESA CA 92627	
J	006424005415 1 99999 HADDAM NECK PLANT SUPERINTENDENT	020006007117 1 99999 HARMON & WEISS .UNION UF CONCERNED SCIENTISTS
	PUSI UPFILE BUA 1270 RFD #1 Cast Mampton CT 06626	SUITE 506 DC 20006

<u>^</u>	ΟΕΡΟΊΥ ΟΙΝ ΓΟΛ ΈΛΑΙΝΟΥ ΗΕΑΕΤΗ Ρυ ουχ 3378 Hondeulu ΗΙ 96801	GA MANAGLA PHST OFFICE DOX 492 BURLINGTUN VT D5401
^	060507053039 1 99999 HENRY PRATT CO A K WILSON VICE PRESIDENT ENGINEERING 401 South Highland Avenue Aurora IL 60507	020036005460 1 99999 Hill Christopher & Phillips Herbert H Brunn Lawrence Cde Lanpher 1900 M St NW Washington DC 20036
-	021045063964 I 99999 Hittman Nuclear & Develup Lurp Enginger Cmarles W Mallory Vice President 9190 Red Branch Road	OENGLAO53198 1 99949 HOPKINSUNS LTD BKITANNIA WORKS R MATSUN PU BUX B27 HD2 20RJ
•	COLUMBIA MD 21045 U77002001156 1 99999 HUUSTUN BAR CENTER CARRU HINDLASTEIN 723 MAIN STREET SUITE 500 HEUSTON TX 77002	HUÖDERSFIELD ENGLAND 036301005823 1 99999 HOUSTON COUNTY COMMISSION CHAIRMAN J DUTHAN AL 36301
•	077001006243 1 99999 Huustun Lighting & Power Co Exfcutive Vice President Po Bux 1700 Houstun tx 77001	077001001479 1 99949 Houston Lighting & Powek Co South Texas project D & Barker Manager PO Box 1700 Huuston Tx 77001
•	077002064711 1 99999 HOUSTON LIGHTING & POWER CO T DZIALO LEADER RECORDS MANAGEMENT 611 WALKER HOUSTON TX 77002	077001006267 1 99999 HOUSTON LIGHTING & POWER CO NUCLEAR ENGINEERING & CONSTRUC J H GULDBERG VICE PRESIDENT PD BOX 1700 HOUSTON TX 77001
•	077001081032 1 99999 HOUSTUN LIGHTING & POWER CO NUCLEAR ENGING & CCNSTRUCTIUN J H GOLDBERG VICE PRESIDENT PO BOX 1700 HGUSTON TX 77001	077001006265 1 99999 HOUSTON LIGHTING & POWER CO P & HURN PO BOX 1700 HOUSTON TX 77001
te	077001001528 1 99999 HGUSTUN LIGHTING 6 POWER CO G w DPREA VICE PRESIDENT PU BOX 1700 HGUSTUN TX 77001	077001003119 1 99999 HUUSTON LIGHTING & POWER CU G W DPREA JR VICE PRESIDENT P D BOX 1700 HOUSTON TX 77001
L	077001064730 1 99999 HOUSTON LIGHTING & POWER CO P NEID Records Management Supyr PD BOX 1700 HOUSTON TX 77001	077001J05308 1 99999 HOUSTON LIGHTING & POWER CO CLOIN ROBERISON MANAGER-NUCLEAR LICENSING PO ROX 1700 HUUSTON TX 77001
⊌ ບ	077001064651 1 99999 HOUSTON LIGHTING & POWER CC P A SHERINGEN JR GEN SUPVR PFOJ ADMIN PG BCX 1700 HCUSTUN TX 77001	083642053042 1 99999 HUICO INC PUST OFFICE BOX 208 MERIDIAN ID 83642
	061856063895 1 99999 Human Factors Society Charles G H(PKINS 28 Fouthill RD Munticello IL 61856	023212005665 1 99999 HUNTUN & WILLIAMS MICHAEL W MAUPIN PO BOX 1535 RICHMOND VA 23212
	023212000793 1 99999 HUNTUN G WILLIAMS W T REVELEY III ESQUIRE PO BUX 1535 RICHMUND VA 23212	C49770007061 1 99999 HURST C HANSON COUNSEL FOR INTERVENORS 311 1/2 E MITCHELL PETOSKY MITCHELL MI 49770

С

and the second		•	
PC HER 1700 HEUSION	TX 77001		
061856063895 1 HUMAN FACTOLS ST	99999 GIETY	023212005665 1 HUNTUN E WILLIAMS MICHAEL & MAUPIN	99999
Z8 FOUTHILL HD MUNTILELLO	IL 61856	PC BLX 1535 Richmund	VA 23212
023212000795 MUNTUN 6 WILLIA	L 99999 15	049770007061 1 Hurst & Hanson Founsel For Interve	49994 NUR S
H T REVELEV III ESQUIRE PD BOX 1535 RICHMOND	VA 23212	311 1/2 E MITCHELL PETCSKY	MI 49770
094105064599 Ibm Curp	1 99999	077704064605 1 IBM CORP T D GANDY 48	434 34
L A EVERS SENIOR INDUSTRY 425 MARKET ST 1 SAN FRANCISCO	SPECIALIST BTH FLUOR CA 94105	INDUSTRY SPECIALIST PD BOX 831 BEAUMONT	TX 77704
083720026180 10440 DEPT OF H	I 99999 EALTH & WELFARE DI SECTION	062701005970 1 Illingis Asst Attor Reed Neuman	99999 INEY GENERAL
ROBERT FUNDERB MANAGER STATEHOUSE	10 83720	500 SOUTH SECOND ST SPRINGFIELD	IL 62701
060610007029 ILLINUIS ASST A ENVIRONMENTAL C PHILIP L WILLMA	1 99999 TTORNEY GENERAL ONTROL DIV	060601079171 1 Ill Indis Attorney 1 Environmental Cunti Karen Borgstadt 188 m Bandolph	99999 GENERAL UFC RUL DIV
ESQ 188 M RANDOLPH CHICAGO	ST - 2315 IL 60610	SUITE 2315 CHICAGO	IL 60601
060601005489 ILLINOIS ATTORI DEAN HANSELL 188 WEST RANDOI SUITE 2315 CHICAGD	1 99999 EV GENERAL OFC .PH ST IL 60601	060601005403 1 Illindis Attorney Environmental Cont Susan K Sekular 100 m Randolph Str Suite 2315 Chite 2315	99999 GENERAL OFC ROL DIVISION EET IL 60601
062704005506 Illinuis Dept 1035 Guier Par Springfield	1 99999 DF NUCLEAR SFTY L DR STH FLOOR IL 62704	062704005728 1 ILLINDIS DEPT DF N MANAGER-NUCLEAR FA 1035 GUIER PARK DR SPRINGFIELD	99999 WCLEAR SFTV CILITY SFTV LIVE -5TH FL IL 62704
		042706080974	99999
062704064197 Illindis dept Philip F Gusta 1035 duter Par Springfield	L 62704	ILLINGIS DEPT OF N NUCLEAR FACILITY S GARY N WRIGHT MANAGER 1035 DUTER PARK OF	NCLEAR SETY SAFETY
		ŠPŘÍNGFIELD 062525007035 1	IL 62704 99999
DEI//DE45%/ ILLINGIS PUNER D S FUNES ADMINISTRATIVE R R 3 BUX 22 CLINTON	CO SUPERVISOR 8 IL 61727	ILLINGIS POWER CO JULIUS GEIER Sou South 27th Ste Decatur	REET IL 62525
062525007034	1 99999 CO	061727064734 1 ILLINCIS POWER CO	99799
L J KOCH VICE PRESIDENT 500 SOUTH 27TH DECATUM	STREET IL 62525	CLINION NPS C & WALDEN SUPYR SITE DOCUME PU BUX 678 (LINION	NTS IL 61727
0625250047C1 ILLINGIS PUNE	1 99999 CU	062525007036 1 1LLINOIS POWER CO	99999
NUCLEAR STATI G E WULLER SUPERVISCR-LI 500 SOUTH 271	JN CNUR DEPT CENSING 4 SIRFLT , LDEDE	SUPERVISOR OF LIC 500 SUUTH 27TH ST DECATUR	ENSING REET IL 62525
FRE F. A TESU	11 0/3/3	•	

	An attack in the second second second second	غید میں دریار دکھر ± ادریک
	ALENT L DRUGS RESIVENT MANAGER	JUHN OCLAN
	BUCHANAN NY 10511	PO BOX 18 BOWLING GREEN STA NEW YORK NY 10004
^	OLOCO4005689 1 99999 Indiana & Michigan Power Co John Dulana	049106004699 1 99999 INDIANG E MICHIGAN POWER CU DENALD C CUOK NUCLEAR PLANT
_	VILE PRESIDENT BEWLING GREEN STATION	PLANT MANAGER PLANT MANAGER PLANT 458
A	NEW YOPK NY 10004	BRIDGHAN MI 49106
•	046206012128 1 99999 INDIANA BUARD OF HEALTH RALPH C PICKARD ASST CUMMISSIONER-ENVIR HEALTH	046234064700 1 99999 Indiana Public Service Cum Rubert Glazier Chief Engineer
•	1330 WEST MICHIGAN STREET INDIANAPOLIS IN 46206	1NDIANAPOLIS IN 46204
6	047401000516 1 99999 Indiana sassafras Audubon Soc David G Frey MRS	008650064452 1 99999 INDUSTRIAL ENGINEERING WORKS H K MEKNITT UA/CONTROL NGR
•	2625 S SMITH ROAD Bloomingtun in 47401	TRENTON NJ 08650
•	020877064650 1 99999 INFO & BUSINESS SYSTEMS INC D R SUPGEN PRES CALVER AD BOY 435	092805064702 1 99999 INFODETICS CORP STEVE GILHEANY SYSTEMS ENGINEER IDALIS CLAUDINA STHEET
•	GAITHERSBURG MD 20877	F193 ANAHEIM CA 92805
•	D66110064696 1 99999 Inland Vital Records Center S. Schultz	008060065236 1 99999 Innovation Technoldgy inc Irwin Spandau
•	MANAGER 8500 INLAND DRIVE Kansas City KS 66110	MUUNT HOLLY NJ 08060
•	010020067165 1 99999 INSIDE NRC DON MARTIN IZZI AVE CF THE AMERICAS	020045037906 1 99999 INSIDE NRC MARGARET RYAN MASHRITON EULTOR
•	NEW YORK NY 10020	433 NATIONAL PRESS BLUG HASHINGTON DC 20045
•	ODENHAOG6999 1 99999 INSPECTORATE OF NUC INSTALLATN AAGE Administrator PD Bux 217 Riso Muse 11	OYUGOSO64849 1 99999 INSTITUT JOSEF STEFAN BORUT MAVKO ADMINISTRATOR JAMOVA 39
•	DŘ-4000 RUSKÍLDE DENMARK	61001 LJUBLJANA YUGUSLA
•	OYUGOSO67273 1 99999 Institui Rude" Bosković Petar Tomas Bijenicka C34 41000 Zagreb Yugosla	020009081538 1 99999 INSTITUTE FOR POLICY STUDIES GDYT ACCOUNTABILITY PROJECT THOMAS DEVINE ASSOC DIRECTOR 1901 O STREET NW
•		MÁŠHINGTÚN DC 20009
•	021403081456 1 99999 Institute for resource mgt Bobby Lednard 428 Fourth St	030339066104 1 99999 Institute of Nuclear PWR OPER Technical Library 1920 Water Place
•	ANNAPULIS ND 21403	ATLANTA GA 30339
•	OPDLANOO7288 L 99999 INSTITUTE OF NUCLEAR RESEARCH REACILR ENGINEERING DEPT. A STRUPCZEWSKI ASST PKJF	094566080830 1 99999 IMTERSTATE NUCLEAR SERVICES VIUGHN OUTMAN FACILITY MANAGER
1	05-400 Swierk Near Warsaw Polanu	CLEASANTUN CA 94566
	092U56040034 1 79999 INTL CONSULTANTS JAY S LYER AND S LYER	UJ2114063070 1 99999 Intl Energy Engineering Inc Richard L Green 77 N Mashington Street
	DED HADUN HHEEL DRIVE DELANNIHE CA 92056	BESTON MA 02114

•	TALALLAIL IN SUL INSTALLAIN AGE JENSEN ADMINISTRATON PU BUR 217 MISO HUSE 11 OK-4000 RUSKILCE DENMARK	LISTITUL JISLE STEFAN BURUT MAVKU AJMINISTAATOR JANGVA 39 61001 LJUBLJANA VUGDSLA
•	OVUGOSO67273 1 99999 INSTITUT RUDER BOSKOVIC PETAR IONIS BIJENICKA (54 41000 ZAGREM VUGOSLA	020009081538 I 99999 IASTITUTE FOR POLICY STUDIES JUNT ACCOUNTABLEITY PROJECT THOMAS DEVINE ASSCC DIRECTUR 19919 STREET NW OF 19900
•	021403081456 1 99999 INSTITUTE FOR RESDURCE MGT BOBBY LEDNARD 428 FOUGIN ST MO 21403	WASHINGTON DC 20009 030339066104 1 99999 INSTITUTE DF NUCLEAR PWR OPER TECHNICAL LIBRARY 1920 WATER PLACE ATLANTA GA 30339
-	OPULANOD7258 L 99999 INSTITUTE OF NUCLEAR RESEARCH REACTOR ENGINEERING DEPT.	074566080830 L 99979 INTERSTATE NUCLEAR SERVICES
•	A 3TRUPCZEWSKI ASST PRUF OS-400 SWIERK NEAR WARSAW POLAND Q92U56040034 1 99999	FACILITY MANAGER PII BUX 218 PLEASANTUN CA 94566 UJ2114063070 1 99999
•	INTL'CONSULTANTS JAY S IVER ISZO WAGON WHEEL DRIVE OCEANSIDE CA 92056	INTL ENERGY ENGINEERING INC Ricmand e green 77 n Washington Street Buston Ma O2114
•	050319081532 1 99999 IONA COMMERCE COMMISSION THOMAS HOUVENAGLE LUCAS STATE OFFICE BLDG DFS MOINES IA 50319	050319055550 1 99999 IOWA COMMERCE COMMISSION ANDREW VARLEY CHAIRMAN STATE CAPITOL
•	052406008660 1 99999 JONA ELECTRIC LIGHT & POWER CO	DES MOINES IN 50319 052406005645 1 99999 IOWA ELECTRIC LIGHT & POWER CO D L MINECK
•	DUANE ARNOLD ASST VICE PRESIDENT PO BOX JSI ELDAK RAPIDS IA 52406	PO 802 351 CEDAR RAPIDS 1A 52406
•	DRASTICINUIS GAS CELEC CU DR STICHNUIM PRESIDENT 206 LAST SECUND AVENUE DAVENPORT IA 52801	IÓWA ILLINOIS GÁS E ELEC CÓ DEAN H STICHNOTH CHAIRMAN PÚ HOX 4350 DAVENPORT IA 52608
•	950319005647 1 99999 Iuma defice for plnng 6 prug 523 east lith street Des Muines IA 50319	044708000122 1 99999 IRWIN STEEL FABRICATORS GULLEY NEG DIV C A DUEAY GUALITY ASSURANCE MGR 1545 MHIPPLE AVE SW CANTON DH 44708
•	060603005419 1 99999 ISHAM LINCULN & HEALE GUUNSELURS AF LAW ONE FIRST NAIIONAL PLAZA 42ND FLUCR CMICAWU IL 60603	060603005204 1 99999 Ismam Linculn & Beale Alam S Farnell One First National Plaza 42ND Fe LHICAGC IL 60603
•	060603006559 1 93999 Ismam Linucin & Beale Ruyert & Fitzgirbens 3 First National Plaza Suite 5200 Chicago IL 60603	020036005179 1 99999 ISHAM LINCOLN & BEALE JUSEPH GALLO ESO IIZO CONNECTICUT AVE RIUM 325 MASHINGTUN DC 20034
•	0606060070H2 1 99999 Isham Linculn C Beale M I Millem L FIRST NATIONAL PLAZA	060602081540 1 99999 Ismay Lincoln & Beale Micheal Miller 3 fikst National Plaza

*

-

• SZ3 EAST 1. TH STYCET 1A 50319 DES MUINES	LA DUEAY LA DUEAY UALITY ASSURANCE MGR 1345 WHIPPLE AVE SW CANTON CANTON
060603005419 1 99999 1 SHAM LINCULN & HEALE CUUNSELUAS AT LAW ONE FIRST VATIONAL PLAZA 42ND FLUCH CHICAWU IL 60603	060603006204 1 99999 IJHAM LINCULN 6 BEALE Alan 5 Farnell Une First National Plaza 42ND FL IL 60603 CHICAGC IL 60603
060603006559 1 99999 I SHAM LINCOLN & BEALE ROYERT & FITZGIBBONS 3 FIRST NATIONAL PLAZA SUITE 5200 IL 60603 CHICAGD	U20036005179 1 99999 ISHAM LINCOLN & BEALE JUSEPH GALLO ESQ 1120 CONNECTICUT AVE ROUM 325 WASHINGTUN DC 20036
0606060070H2 1 99999 ISMAN LINCULN & BEALE M I NILLEH 1 FIRST NATIONAL PLAZA SUITE 4200 CHICAGO IL 60606	060602081590 1 99999 ISMAM LINCOLN & BEALE MILHEAL MILLER 3 FIRST NATIONAL PLAZA 51ST FLOOR IL 60602 CHICAGO IL 60602
060670001344 1 99999 ISHAN LINCULN & BEALE PAUL M MURPHY ESJUIKE ONE FIRST NATIONAL PLAZA 42ND FLUDR CHICAGO IL 60670	UGUGO3005730 1 99999 ISMAN LINCOLN E BEALE PHILLIP P STEPTOE UNE FIRST NATIONAL PLAZA SUITE 4200 IL 60603 CHICAGO IL 60003
060603006202 1 99999 ISHAM LINCULN E BEALF RUNALD G ZAMARIN UNE FIRST NATIONAL PLAZA 42ND FL CHICAGU IL 60603	OJAPANO53044 1 99999 Ishikawajipa Harina Heavy Ind Nuclear Power Division K Jin-Nakahara-Cho No 235 Isogo ku Yukohama Japan
OISRAEOS7568 1 09999 ISRAEL ATUMIC ENERGY CCM LICENSING DIVISION DAN LITAI ADMINISTRATUR PU BUX 7061 TEL AVIV 61070 ISRAEL	091201055299 1 99999 Itt General Controls Sid Lehnan Cuurd & O C Eng Bol Allen Avenue CA 91201 Glendale CA 91201
002901038043 1 99999 ITT GRINNELL CORP EDWARD REFFA 260 WEST EXCHANGE STREET PROVIDENCE RI 02901	027284053047 1 99999 ITT GRINNELL CORP INJUSTRIAL PIPING R & BERLIEN VP & DIR-QA I250 W MOUNTAIN STREET RERNERSVILLE NC 27284
O 17603053048 1 99999 ITT GRINNELL CORP DIA-FLO DIVISION E L BROUNE CONTRACIS MGR 33 CLNTERVILLE ROAD I ANTERVILLE ROAD	002901U38044 1 99999 ITT GRINNELL CORP CHARLES MCKENNA 260 WEST EXCHANGE ST PROVIDENCE RI 02901
DO2888053049 1 99999 ITT GRINNELL CORP ITT HAMMEL DAME DONALD A UELLETTE Q A MANALEE ITS POIST RUAD MARWAICK RI 02889	002916053046 1 99999 ITT GRINNELL COMP PIPE HANGER DIVISION THOMAS SMITH MGR APPLICATION ENG 260 m Exchange SI PROVIDENCE RI 02916
DIGIISOBILUG I 99999 J E LONERGAN CO T A NICKEY QA MANAJER RED LIUG KD WEST OF VERREE RD PHILADELPHIA PA 19115	094041065237 1 99999 JACK R BENJAMIN & ASSOC LUURT HOUSE PLAZA BLUG FRED A MEDITER MIN BAY PLAZA 444 CASTRO ST SUITE 501 MUNTAIN VIEW CA 94041
O32201003136 1 99999 JACK JUNY ILLE FLEC AUTHURITY A M HARK INGTON ASSOC MANAGING DIRECTOR	U32201006295 1 99999 JACKSONVILLE JOURNAL REG CROMDER Pij BUX 1949 JACKSONVILLE FL 32201

-

•

8

•

0

0

•

۲

•

•

J.

J

3

Ú

ں

. مەربە

Ż

•	PHILADECPHIA DA LILA	AJUNTALA VEEM LA 94941
•	032231003136 1 99999 JACK JUNYILLE FLEE AUTHURITY A M HARKINGTON ASSUE MANA JING DIRECTOR P (1 41X 53015	032201006295 1 99999 JALKSUNVILLE JOURNAL REG CRUMDER PJ RUX 1949 JACKSUNVILLE FL 32201
•	JAČKSÚNVILLE FL 32201	077777067763 i 99999
•	001605053050 1 99999 JAMESHUKY LUKP F A KKEUZINZER DA WAEUZINZER	JANDL MEG CO KALPH WEIGHTMAN 24131 HARDY HD
•	640 LINCOLN STREET MA 01605	SPRING 1X 11313
•	037919064537 1 99999 JRF Assuciates inc Myrcn Casaja 1430 Diamitina rest blyd	UD8731006394 I 94949 JERSEY CENTRAL PHR & LIGHT CU UYSTER CREEK NUL GENERAT STA LICENSING SUPERVISOR
•	KNOXVILLE TN 37919	PÔ ĐƯX 308 Furked River NJ 08731
•	008731006493 1 94999 JERSEY CENTRAL PWP & LIGHT CO Dyster Creek Nuclear Station J I Carroll Jr	008731001562 1 99999 JERSEY CENTRAL PHR & LIGHT CO UYSTER CREEK NUC GENERATING ST P. H. FLEDLER.
•	DINECTON ÖYSTER CREEK OPS PU BUX 388 FORKED RIVER NJ 08731	VICE PRESIDENT-GENERATION PU BUX 388 FORKED RIVER NJ 08731
•	008731005324 1 99999 Jersey Central Pwr & Light Co Dyster Creek NUC Pwr Station M Laggart	010005026889 1 99999 Johnson & Higgins Nuclfar Advisury Group
•	PLAÑT SUPERINTENDENT Pû bûx 388 Furked River nj 08731	NEW YURK NY 10005
•	OTOL39038381 1 99999 Johnson & Higgins Nuclear Advisury Group	053201066231 1 99999 Juhnson Cuntrols Inc Frank Reck For Last Michigan St
•	THEDUURE RUDDUCK ONE SHELL SUUARE STH FLOOR NEW URLEANS LA 70139	MILWAUKEE WI 53201
•	091740053052 1 99999 JOHNSTON PUMP CO	044115066699 1 99999 JCNES DAY REAVIS & POGUE Poneett Brunn
•	GLENDURA CA 91740	1700 UNION CUMMERCE BLDG CLEVELAND OH 44115
•	094623001049 1 99999 KAISER ENGINEERS POWER CORP	020036032157 1 99999 Kansai Electric Power Co Inc Kunihiku Uta
	CHARLES D CUFFER MANAGER PU BUX 23210 CA 94623	MÁNÁGER 1100 17TH ST N W Súite 606 Masmington DC 20036
	067201005474 1 99999 KANSANS FOR SENSIBLE ENERGY	064141080935 1 99999 KANSAS CITY POWER & LIGHT CU
	PU 60X 3192 WICHITA KS 67201	D I MCPHEE VICE PRESIDENT PO BUX 679 KANSAS CITY NO 64141
•	064141001348 1 99999	066620012140 1 99999
٠	RANSAS LITY PUNER & LIGHT CU DUNALD T MCPHEE VILE PRESIDENT - PRODUCTION 1330 DALTIMORE AVENUE	RUREAU OF RADIATION CONTROL Gerald W Allen Directur A 100 S Toppka Avf
۳		TUPEKA KS 66620
•	UGOGUGUDDU/ 1 94949 KANSAS LLEC PUNER LOUP INC JUE MULHCLLAND Manager-Puner Supply C Engrg PC HDA 49/1 - Gagi CTR_STATION	KANSAS GAS & ELEC CU LECAL HAI PH FOSTER CSUURE
	TEPEKA KS 00004	WICHITA KS 67201

1. 1.

00000

1011001554

4 فسا 0 •

•	OUND SCOLUTE AS POWER CLAP RAISER ENGTHERES UNLITY ASSURANCE LICENSING LUALITY ASSURANCE CMARLES C LUFFER MANAGER PU BUR 23210 CA 94623	ANDAL LEUTHT STA NUNHHU UTA MANAGER 1100 ITTH ST N M SUITE 606 MASMINUTUN DL
•	DAKLAND 067201005474 1 99449 KANSANS FOR SENSIBLE ENERUY PU BUX 3142 KS 67201 WICHITA	U6414100J935 1 94 NANSAS LITY POWER & LIGH D T MCPHEE VICE PRESIDENT PO BUX 679 MO KANSAS CITY MO
•	064141001348 1 99999 KANSAS CITY PODER & LIGHT LU DUNALO T MCPHEE VICE PRESIDENT - PRODUCTION 1350 BALTIMORE AVENUE KANSAS CITY MO 64141	U66620012140 1 9 KANSAS DEPT OF HEALTH & HUREAU UF RADIATION CON Gerald W Allen Directur 6700 S Topeka Ave KS Topeka KS
•	06660400650 1 99994 KANSAS ELEC PUMER COUP INC JUE MULMCLLAND MANAGER-PUMER SUPPLY E ENGRG PC BUSCH 4977 - GAGE CTR STATION PC BUSCH KS 6604	067201001353 1 KANSAS GAS G ELEC CO Legal Raiph foster Eswure Po bux 208 Wichita
•	067201005268 1 99999 KANSAS GAS & ELEC CU GLENN L RUESTER VICE PRESIDENT-NUCLEAR P C BGX 209 KS 67201 WICHITA KS 67201	067201001554 1 KANSAS GAS & ELEC CO NUCLEAR GLENN L KOESTER 201 NURTH MARKET STREE PO BOX 208 K WICHITA K
•	D66612005471 1 99999 KANSAS STATE CORP CONMISSION MICHAEL C KEENEK WDIF CREEK PROJECT DIRECTOR STATE OFFICE BLDG KS 66612 TOPEKA	066612007010 1 KANSAS STATE CORP COMM Legal Divisiun C I Petenson State Office Building 4Th Flouk Topeka
•	D66612006075 1 99995 KANSAS STATE CORP COMMISSION TERRI SCULLEY STATE OFFICE BUILDING TOPEKA KS 66612	040601001462 1 KENTUCKY ATTORNEY GEN UAVID K MARTIN ASSISTANT ATTURNEY GEI CAPITUL BLDG FRANKFORT
Ċ	040578064892 1 99999 KENTUCKY DEPT OF ENERGY BUREAU OF ENGY PROD L UTIL ANTHUR F NICHOLSON IRON WURKS PIKE POB 11998 KY 40578	040621009337 I RENTUCKY DEPT OF HUM Bureau for Health Ser Dunald R Hughes Manager-Radiatn Contr 275 East Min Street Frankfurt
	LEXINGTON 006483053054 1 99999 KERITE CU ENGINEERING DEPT R E FLEMING NUCLEAR DEVELUPMENT ENGINEER NUCLEAR DEVELUPMENT ENGINEER 90DAY ST PJ BUX 452 90DAY ST PJ BUX 452 90DAY	015222064450 1 KLRUTEST MANUFACTURIN Makarczyk UA MANAGER 2525 LIBERTY AVE PITTSBURGH
	SETHION OT3102011701 L 99999 KERR MGGEE CORP NUC LICENSING & REGULATION W J SHELLEY VICE PRESIDENT VICE PRESIDENT KERR-MGUEF CENTER KERR-MGUEF (IN) 0K 73102	054216005831 1 REWAUNEE COUNTY BOAR DONALD L OUISTROFF CHAIRMAN KIWAUNEE COUNTY COUR REWAUNEE
•	OKLAHUMA CTIT DI9104006438 1 99999 KEYSTONE ALLIANCE ALAN J NUGLE 3700 JHESTNUT STREET PHILADELPHIA PA 1910	020036007062 1 KIRNPATRICK LOCKHART HERBERT H URUWN ESU M ST NW 9 TH FLOUR WASHINGTON
·	020036007063 1 09999 Kihkpatkick Lockhart et al Labrence c Lanptier	020036007064 1 Kirkpatrick Lockhar Karla J Letsche LSU

TH ST N H	DL 20036
DELON LE	99999 99999
PHEE RESIDENT	M8 64141
012140 1	99999 FM 6 ENVIRN
, DEPT OF HEAL 1 OF RADIATION 1 W ALLEN	CONTROL
TOPEKA AVE	KS 66620
GAS C ELEC C	0
RE X 208	KS 67201
1001554 L S GAS & ELEC C	99999 D
AR L KOESTER Urth Market St	IREET
TA 1200	KS 67201
S STATE CORP I DIVISION PETERSON	1NC
E OFFICE BUILD FLOUK KÅ	KS 66612
01001462 1 UCKY ATTORNEY D K MARTIN	GENERALS OFC
STANT ATTURNET TOL BLOG KFORT	KY 40601
21009337 1 UCKY DEPT DF I	99999 NUM RESOURCES SERVICES
AU FUR HUGHES AGER-RADIATN CI FAST MAIN STR	ONTROL BRANCH
VŘFURT 222064450 1	99999 10 THG CURP
MAKARCZYK Manager S i trekty ave	
TSBURCH	PA 15222
AUNEE COUNTY I	SOARD
AUNEE COUNTY	COURTHOUSE WI 54216
NO 36007062 1 REPATRICK LOCK	99999 HART ET AL
UO M ST NW H FLOUR H TNGTON	DC 20036
0036007064 1 RKPATRICK LOCK	99999 HART ET AL
RLA J LETSCHE	

/

	ANDER THE REPORT OF A STATE OF A	CARD LISERTY AVE
~	- 49 DAY 51 P3 65X 452 - Seymonia - Cy orange	PLITSBURGH PA 15222
	073102001701 1 99999 #FRA MCCFF (SWP	
	NUC LICENSING & REGULATION	UDNALD L CUISTROFF
	M J SHELLET Will F PH 1 1 1 1 1 1	CHAIRMAN
•	KERR-MOUEE CENTER	REMAUNEE COUNTY COURTHOUSE
	DELAHEMA LITY DE 731J2	
	019104006438 1 99999	020036007062 1 49949
A	REVSTORE ALLIANCE Alan i Sucie	KIRKPATRICK LOCKHART ET AL
	3700 CHESTNUT STREET	ESU
	PHILADELPHIA PA 19104	1900 M ST NW
		STM FLOUR SASHINGTON OF DODA
	0.2007/00.2013	
	KIRKPATRICK I CCKHART FT AI	
	LANRENCE C LANPTIER	KARLA J LETSCHE
	ESU Igun m st nm	ESU M ST NU
	WASHINGTON DC 20036	WASHINGTON DE 2003A
	ONGERNUS 3057 1 99999	020096067283 1 99999
^	KLUCKNER-WERK AG	KMC INC
	W AUSTEL	JUAN BELAUT 1767 Dennsvivanja aug au
-	U A MANAGER	WASHINGTON DC 20006
~	PU BUA 27 BU Di 4500 Disnabruck - Mgermany	
	012301007985 1 999999 KNCLIS ATOMIC DOUCD LAD	017108005438 1 99999
	LIBRARY	KUNERT I KNUDD
	LIBRARIAN	ASSISTANT SOLICITOR
	SCHENFCTADY AV 12203	407 N FRONT STREET
-		HARRISBURG PA 1710H
	010023024754 I 00000	
	KUMANOFF ENERGY ASSOCIATES	UI5222057893 1 99999 I K CONSTOCK E CO INC
-	CHARLES KOMANDEF	TECHNICAL SERVICES
	14IH FLOOR	JAMES W DELL
•	NEW YORK NY 10023	920 FORT DUQUESNE BLVD
-		PITTSBURGH PA 15222
	060604001451 1 99999	053209053059 1 99994
•	LAKE MICHIGAN FEDERATION	LAKESIDE BRIDGE & STEEL CO
	ESQUIRE	DIRECTOR-DA
-	53 WEST JACKSON BOULEVARD	5300 NORTH 33RD STREET
•	CHICAGO IL 60604	MILWAUKEE WI 53209
	002030052040	
•	LAMED INDUSTRIES INC	049423081887 1 99999
-	1546 NUNTH JCHNSON STREET	475 E 16TH ST SHITE 20
	EL LAJUN CA 92020	HELLAND MI 49423
•		
-		
_	094550079656 1 99999	063114006112 1 00000
•	LAWKENUE LIVERMORE LABORATORY	LFAGUE OF WUMEN VOTERS
	CAROL A MELER	MISSUURI CHAPTER
-	PC BOX BON L-46	2138 HOUDSON ROAD
•	LIVERMURE LA 94550	ST LOUIS MU 63114
	063140006111 1	
	LEAGUE OF NUMER VATENC	063114006113 1 99999
•	UNIVERSITY CITY CHAPTER	NESSEURI CHAPTERS
	MARJUNIE REILLY Energy Chairman	BARHARA SHULL
•	7065 PER SHING AVE	ST LOUIS MUUDSUN RUAD
-	UNIVERSITY CITY NO 63130	
	020036079364 1 99999	1200 360061 79 t power
	LEBOLUF LANB LEIBY & MACRAL	LI BOLUF LAMB LEIBY & MACHAF
	TRENE C CARSTENS	RUBERT S FARON
	1333 NEW HAMPSHIRE AVE NW	SUITE 1100
۳	WASHINGTUN DC 20024	WASHINGTON DC 20036
	ure c (1030	

"

ſ

T

٩.,

Here is a second s	الهائي والمحاص		
LIVER MURY	** ***********************************	11 1 999	99
9999		HUMEN VUTERS	
A FALLE OF NUMER VETERS	นี้เริรัต้มุ่ม (HAPTER	
UNIVERSITY LITT COMPTON	JARMARA JILA WGCD	SUN RUAD	63114
MAN JUNIT CHAIR MAN	STLOUIS	FU FU	03114
TU65 PERSHING AVE NO (3130	a	00
UNIALSZIA CIAL	aa jagu 36006	179 . I WALL MALL	KAE
020036079364 LETAY & MACR	AE LEBUEUF L	FARIN	
LEBOLUF LANG LEVEL	1133 NEW	HAMPSHIRE AVE N	H
TRENE & CARSTENS	ŠŲĮTE 110	0 00	20036
1335 NEW HAMPSHIKE HTL	WASHINGIG		
MASHINGTON DC	20036	120 1 99	999
999	020036000 LEACEUE L	AND LEIBY & MAC	RAE
LENGEUF LAMB LEIBY & MACH	AL MARILYN A	SHAW	W
CARL D HOBELMAN	1333 NEW	HARPSHIRE HE	10014
1313 NER HARFSITTE	20036 WASHINGT	ON DC	20030
WASHINGTUN UL	20030		
	02003600	1368 MA	CRAE
OZOUJOUOLAS IFINY & MAC	RAE LEPGEUF	VUIGT	
LEUNARU N THOSTEN	1335 NEW	HANDSHINE AVE	
1333 NEW HAMPSHIKE AVE H	- SULTE 11 MACHINGT	00 DC	20036
WASHINGTON DC	20030 ##3/11/01	-	
	01945400	2167 1 9	4999
020036065264	LEEDS 4	NURTHRUP LU	
LEMBEUF LAMB LEIDT & MAC		AL MANAGER	
1333 NEW HAMPSHERE AVE N	SUMNEYTU	WN PIKE DA	19454
SUITE 1000 DC	20036 NORTH W	ALES CO	
WASHINGIUN			0000
99	999 0636110	DI447 I DIANN MAYER	LUDH
	EDWARD	M OSANN	
D'H DURFER	ESQUIRE		
QUALITY SYSTEMS ENGINEER	UNE IBM	- YEAZA	
PARSIPPANY NJ	UTUDA CHICAGO	11	60611
	0364110	05841 1	99999
019107006463 1	1999 USSOLLU 1 IMESTO	NE COUNTY COM	
LINERICK ECULOGY ACTION	CHARLES	R CHRISIUPHEN	
1315 WALNUT STREET		185 .	1 36611
SUITE 1032 DA	19107 ATHENS	A	L 39011
PHILADELPHIA			00000
	9459 0245020	164449 1	44444
024506038042 I T	LINITON	LUSZUSKY	
F K DENHAM	SC MAN	AGER	
VICE PRESIDENT	11318 5114 4	LUDALL RU	A 24502
I YNCHBUKG	24506 LYNCHB	UNU	
A taking an		ALLER7 1	99999
006108052979 1 9	020854	SCCIATES	
LITTON INDUSTRIAL PRUD		RTHSBY COURT	10 2085
CUNTRUMATICS DIV	PUTUNA	Ľ	
JUALITY ASSURANCE MANA	GER		
ZZZ ROBERTS ST	1 06108		
CADI MANTCHED	011501	081574	99999
011501081575 1 151 ANU 1 1047 ING L	O LUNG 1	SLAND LIGHTING	
EDWARD M BARRETT	JEFFRE GENERA	ATTORNEY	
GENCHAL LOUNSEL	250 04	O COUNTRY RD	NY 1150
MINEULA N	Y 11501 MINEOL		,
118 ILL 0	- ·		99999
011801081573 1	99949 01179	ISLAND LIGHTING	c ú
LENG ISLAND LIGHTING (ີ່ ເປັນ ເປັນ ເປັນ ເປັນ ເປັນ ເປັນ ເປັນ ເປັນ	ATSCN	
T F GEREGRE Managep-ja dept	RECOR	US MANAUER X 604	
175 E CLU CLUNTRY HD	NY 11801 WADIN	G RIVER	NT IIC
HILKSVILLE			00000
011501081130	99999 01130	1006066 1	699999 CD
011201001120 1.4.4.	e6	ISLAND LIGHTING	

1.13

٠ ۲ . .

1.5 1

AL 35611 99999

VA 24502

ND 20854

NY 11501

NY 11792

~	₩ Λ } ‡ιφα ΗξψΈμη	NY 11792	HICKSVILLE N	II 11001
-	010808001396 1 LOUISIANA ATTORNEY LANDS & NATURAL RES	99999 GENERAL	070802007078 1 LUUISIANA CONSUMERS LE DUPIS FALKENHEINER SAS NURTH 61M ST	99999 AGUE
~	NILLIAM J GUSTE Aitn: J David McNei 7434 Perkins Kuad Batun Kuugf	LL 111 LA 70808	BATON ROUGE	A 70802
^	UTUI12001397 1 LOUISIANA DEPT OF RICHARD M TRUY	99999 JUSTICE	070808007079 1 Luuisiana dept of Jus Linda d Hatkins Staff Atny	99999 11CE
~	ASSISTANT ATTURNEY 234 LOYOLA AVENUL New Grleans	LA 70112	7434 PERKINS RD SUITE BATCN RUUGE	C LA 70808
r	073804032453 1 LOUISIANA DEPT OF GFC OF ENVIRON AFF B J PORTEF	99999 NATURAL RES AIRS	070898021458 LUUISIANA DEPT OF NATI NUCLEAR ENERGY DIVISI WILLIAM M SPELL ADMINISTRATOR	99999 URAL RES ON
~	PD BDX 44066 BATUN RUUGE	LA 70504	F O BOX 14690 Baton Mouge	LA 70898
~	07CI74005949 L LDUISIANA POWER & NUCLEAR	99999 Light Co	070174079375 1 Louisiana power & Lig T gehrets La Manager	93999 HT CU
^	F J DRUMMUND PROJECT SUPPORT MA 142 DELAHUNDE ST New Urleans	NAGER-NUC LA 70174	142 DELARONDE ST New URLEANS	LA 70174
ſ	070174001280 1 LUUISIANA PUHER C WATERFURD 3 SES	99999 LIGHI CC	070174005951 1 Louisiana power & Lig Nuclear Operations L V Maurin	99999 Ht Cu
ņ	PRODUCTION ENGINEE 142 DELARUNDE STRE New Orleans	R ET LA 70174	VICE PRESIDENT 142 DELARONDE ST NEW ORLEANS	LA 70174
•	070174028293 L LOUISIANA POWER C NUCLEAR OPERATIONS L V MAURIN PO BUX 6008 142 DE NEW FORIEANS	99999 LIGHT CO LARONDE ST LA 70174	070808007080 l Luuisianians for Safe Gretchen R Rothschil 1659 glenmore ave Haton Rouge	ENERGY D LA 70808
•		00000	020036000010 1	99999
۲	LOWENSTEIN NEWMAN MAURICE AXELRAD 1025 CONNECTICUT	REIS ET AL	LONENSTEIN NEWMAN REI Robert Lowenstein Fsuure Loss connecticut Avei	IS ET AL
•	SUITE 1214 WASHINGTUN	DC 20036	SUITE 1214 WASHINGTUN	DC 20036
6	020036007037 1 LOWENSTEIN NEWMAN JACK R NEWMAN	99999 REIS ET AL	020036C07089 1 LUWENSTEIN NEWMAN RE Makuld F Reis 1025 Connecticut Ave	99999 IS ET AL
6	WASHINGTON	DC 20036	SUITE 1214 MASHINGTON	DC 20036
	020036001632 1 LOWENSTEIN NEWMAN	99999 Reis et Al	045214053062 1 Lunkenheimer CO	99999
•	KATHLEEN H SHEA 1025 CONNECTICUT WASHINGTON	AVENUE NW DC 20036	PU HOX 14360 CINCINNATI	OH 45214
	090723053090 1	99999	016145053732 1	99999
.	Ř Ř KELLUGG CO PULLMAN POWER PRO 14507 S PAPAMOUNT PARAMOUNT	DULTS RLVD CA 90723	INDUSTRIAL CONTROL D L C MCCLELLAND CONTROL BUARD MGR	IV
• 🙂			SANDY LAKE	PA 16145
•	00433336579 1 Maine Dept of Hum Div of Health Eng Dic Hoxie Diserton	99999 IAN SERVICES INIERING	004330005434 L MAINE EXECUTIVE DEPA STATE PLANNING OFFIC IR9 STATE STREET AUGUSTA	RTMENT ER NE 04330
-	STATE HOUSE STA	TION IL		

а Э	02003600061+3 1 9799 Lowenstein newaan beis et al Maukile axelrad Tobe Connecticut ave nu	J2 J336000 J10 L Lumpstein Newan Reis et al Rubert Lumenstein Fsuure
•	SUITE 1214 DL 20036	1025 CONNECTICUT AVENUE NH SUITE 1214 WASHINGTUN DC 20036
	020036007037 1 97999 LOBENSTET: NEBMAN HEIS ET AL JACK P NEWMAN 1025 CUNNECTICUT AVENUE WASHINGTUN OC 20736	D20036007089 1 99999 LUMENSTEIN NEWMAN REIS ET AL MANUED F HEIS 1025 CEMNECTIEUT AVF NH SUITE 1214 MASHINGTON DE 20036
•	0200360J1632 1 99999 LOWENSTEIN NEMMAN REIS ET AL KATHLEEN H SHEA 1025 CONNELTICUT AVENUE NM MASHINGTON DC 20036	045214053362 1 99999 LUNKENHEIMER CO PL BUX 1436J CINCINNATI OH 45214
•	090723053090 1 99999 M M ELLUGG CO PULLMAN POWER PRODUCTS 14507 S PAPAMOUNT BLVD PARAMOUNT CA 90723	Ol6145053732 1 99999 Magnetics Industrial control DIV L C MCCLELLAND Control Buard NGR Po EOX 457 Sandy Lake PA 16145
•	004333336589 1 99999 MAINE DEPT UF HUKAN SERVICES DIV OF HEALTH ENGINEERING DIKECTOR STATE HOUSE STATION 11 AUGUSTA ME 04333	004330005939 1 99999 MAINE EXECUTIVE DEPARTMENT STATE PLANNING DEFICER 189 STATE STREET AUGUSTA ME 04330
•	004336006478 I 99999 MAINE YANKFE ATOMIC POWER CO NUCLEAR ENVINEERING & LICENS JOHN H GAR ITY SENIUR DIFECTOR 83 EDISON DRIVE AUGUSTA ME 04336	001701005940 1 99999 MAINE YANKEE ATOMIC POBER CO RUBERT H GROCE Senior Engineer-Licensing 1671 Wurdester RD Framingham Ma 31701
•	004336008719 L 99999 MAINE YANKEE ATUMIC POWER CD JUHN R RANDAZZA VICE PRESIDENT 83 EDISON DRIVE AUGUSTA NE 04336	004336006575 1 99999 Maine Vankee Atomic Power CC F n Thurlow President Edison Drive Me 04336 Augusta Me 04336
•	004330004692 1 99999 MAINE YANKFE ATOMIC POWER CO ELWIN W THURLOW PRESIDENT 9 GREEN STREET AUGUSTA NE 04330	004578006479 1 99999 MAINE YANKEE ATONIC PONEM CO E G NOJO Plant Manager PU Rux 3270 MISCASSET ME 04578
•	092025064747 1 99999 Management Analysis CO J L Nickle Consulting Associate Bid Luna Vista Jrive Escondidu ca 92025	010471079405 1 99999 Manhattan College Mechanical Engineering Dept Runald Kane Realtur Administrator Manhattan Cullege Parkway 9runx ny 10471
•	OWGERM053063 1 99999 Mannesmann Anlagenbau Ag Ru F Bruehl Director PD BGX 100307 4000 Duesseldupf FRG Wgermany	UD7005053731 1 99999 MARUITA SCIENTIFIC CONTROL INC DUNALD L DENISE 4 MANAGER BOUNTON AVENUE BUUNTON NJ U7005
	090302J53730 1 99999 MARVIN LNGI (EERING CO LTD G FRILUMAN Executive vice president 251 West Reach Avenue Inglémuud ca 90302	U21201009739 2 99999 MARYLAND DEPT HLTH & MNTL HYG DIVISION OF RADIATION CUNTRGL HUREAT E CORCURAN CHIEF 201 w PRESTON STREET PALTIMORE MD 21201
	UZ1201080909 1 99999 MARYLAND DEPT HLTH & MNTL HYG ENVIRUNMENTAL PROGRAMS	021204005879 1 99999 MARYLAND DEPT OF NAT RESOURCES ENERGY & CUASTAL ZUNE AUMINIST ADVIVISTRATOR-PHR PLNT SITING

	BESTER A CZ202	
^	OG21030A1565 1 99999 MASSACHUSETIS ENVIRON PROTOIV OFC UF ATTURNEY GENEFAL JOANNE SHOTWELL ASST ATTURNEY CENERAL LASHDWITON PL 191H FLOOR	002138051912 1 99999 MASSACHUSEITS INST UF TECM Science Technology and Suciety Building 200-222 MA 02138 CAMBRIDGE MA 02138
•	BOSTON NA CZION	002108030662 1 99999
\sim	DOZISHUTHUB I Massachusetts Inst of Telm Linuuln Clark Director of Reactor Laboratory Isr Alhany Street	MASSACHUSLTIS DEC ATTORNEY GEN ENVIRO PROT DIV Laurie Burt 1 Ashburton Pl
^	CAMBRIDGE MA 02139	RM 1902 BUSTG4 MA 02108
^	002108006069 1 99999 MASSACHUSETTS CFC ENERGY RES PATRICK J KENNY 73 TREMUNT STREET MA 02108	002108006068 1 99999 MASSACHUSETTS DFC ENERGY RES EDWARD L SELGRADE 73 TRENUNI STREET MA 02108
<u>~</u>	BOSTON	
•	002108006070 L 99999 MASSACHUSETTS PUB PROTECT BUR ENVIKUN PRUTECTIUN DIVISION Stephen N Leonard UNE Ashburtun Place Typh Fluch	002108006071 1 99999 MASSACHUSETTS PUB PRCTECT BUR UTILITIES DIVISION MICHAEL B METER ONE ASHBURTON PLASE 19TH FLUUR
-	BOSTON NA 02108	BUSTON MA UZIUB
•	002108006067 1 99999 MASSACHUSETTS PUB PRDTECT BUR OFC OF ATTNY GENERAL JD ANN SHUTWELL	MASSACHUSETTS WILDLIFE FEDERAT LESTER B SMITH DIRECTOR UF CONSERVATION
•	ONE ASHBUKIDA PLACE 1914 FLOOR MA 02108	NATICK NA 01761
•	002101066215 1 99999 Material Handling Equipment Steven ParkHurst Hommun 114	029210081602 I 99999 Maisco David A Harley Kegiunal Manager
	ROSTON MA 02101	SUITE 201 SC 20210
•	090807055298 1 99999 MCC PACIFIC VALVES INC	COLUMBIA SC 29210 002062053064 1 99999 MLGRAW EDISON CO
•	3201 WALNUT AVENUE CA 90807 Long reach ca 90807	ASUNATAN STREET NA 02062 NORWOOD NORWOOD
•	060007053065 1 99999 MEA INC T & LASIEWICZ PRESIDENT CAN LANK	091311055297 1 99999 Metal Bellows Corp Div Metal Bellows-Sharon Ma Don Williams Markfing Mgr
•	ELK GROVE VILLAGE IL 60007	2096U KNAPP ST CHATSWURTH CA 91311
•	077036081119 1 99999 Metter Management Systems inc Jeff Gramm	068106007026 1 99999 Metry Area Planning Agency Ungnia Prieditis
•	ŠĀBA PUINT HEST DRIVE BOX 770020 HOUSTUN TX 77036	7000 W LENTER RU 200 NE 68106 ;44M4 NE 68106
•	OLTUSTUOS442 L 99499 Metrupulitan Edison Co TMI-2 Autou	017057005513 1 99999 Metropolitan edison Cu Gale Huvey Mp f Dimetror of TM1-2
۲	J J HARION Acting directur of TMI-2 Pu Rix 440 Midulfturn pa 17057	PU BUX 480 MIDULETUWN PA 17057
٠	017057066309 1 99999 Métropulitan Edison Cu Gpu Nucleam Meney 13 Hukili	017057006413 1 99999 METROPOLITAN EDISON CO SITE OPERATIONS THI-2 L KING
	VICE PRES & DIRECTOR-THIL PL BLX 440	ALTING MANAGER PU BOX 480
•	MEDULETUWN PA 17057	105125006063 1 99999
	1999 # # # 1 19/19/19/19/19/19/19/19/19/19/19/19/19/1	

	ME I STATE		Satte A	1 A A A A A A A A A A A A A A A A A A A
	TMA-2 1 F LARSEN	LNSTNG	SAN JOSE	
	SUPERVISUR-UNIT 2 LIC			
	PO BCX 480 MIGOLETUNN	PA 11051		99999
		99949	NUH TECHNICAL ASSOC	
	095127031049 ASSOC		DALE BRIDENBAUGH	
	TECHNICAL CUNSULTANTS		SHITE K	CA 95125
	1723 HAMILION FT	CA 95125	SAN JOSE	
	SAN JUSE	CA ///		00000
•	_	00003	048909005690 1 PU	BLIC HEALTH
	095125005948 ASSOC	44447	MICHIGAN ULLI	
•	HHB TECHNICAL MARD		DIRECTOR	
	1723 HAMILTON AVE		PO BOX SUUSS	41 48909
	SUITE K	CA 95125	CANSTING	
	JAN JOJE	_	048913001422	99999
	048009001425	99999	MICHIGAN UFC OF ATT	DIVISION
_	MICHIGAN DEPT OF PUE	HEALTH	ENVIRON PROTECTION	
-	DIV OF RADIOCOULDE		ASST ATTORNEY GENER	(AL
	CHIEF		720 LAW BUILDING	MI 48913
~	PO 80X 33035	MI 48909	LANSING 2	99999
	LANJING I	99949	048913005392	GUVERNOR
	048909019440	GOVERNOR	WILLIAM & RUSTER	
	COMPREHENSIVE ENERG	ENCT BUST	CAPITOL BUILDING	
	DENNIS PATHE - CAP	ITOL BLDG	LANSING	MI 48413
_	ROCH 1	MI 48909	-	. 2000
$\hat{}$	LANSING	00000	094043064753 1.	51999 TENS
	048909004886	VICE COM	MICROFORM DATA ST	
$\hat{}$	MICHIGAN PUBLIC JC.		PRODUCT MANAGER	
	G		830 MAUDE AVENUE	CA 94043
	LOX 30221	MI 48909	HUGHTATIONS	
ר	LANSING		044576053066 1	99999
	070161006470 1	99999	MIDLAND ROSS	ISTON
	MIDDLE SOUTH SERVI	CES .	NETAL FRAMING DIV	
,	D C GIBBS		UUALITY ASSURANCE	SPECIALIS
	225 BARONNE STREET	_	BOZ EISENHUWER UM	IN 46526
3	BOX 6100	LA 70161	GUSTEN	99999
•		99999	055101065220 L	B & INDUSTRY
	ONETHEODILOZ	FAIRS & EMPL	BOILER DIVISION	
	HEAD-NULLEAR DEPT	1	HENRY BARUN	
	ADMINISTRATUR		444 LAFAYETTE RD	MN 55101
	PC 80x 69	NETHERL	ST PAUL	00009
	ADOKBOKO	00009	055440079447 1	F HEALTH
	055440005663 L	HEALTH	ENVIRUNMENTAL HE	ALTH DIVISION
•	COMMISSIONER OF H	EALTH ET SF	HUGER L DERUUS	
	717 DELAWARE STRE MINNEAPOLIS	- AN 55440	717 DELAWARE STR	EET SE NN 55440
	H I HILL HILL DO T		MINNEAPULIS	00009
-		69999	055113024874	ION CONT AGEY
	055414005660	ONT CITZ ASS	C MINNESOTA FOLLO	
	ENERGY TASK FORCE		NUCLEAR ENGINEE	(20
	RUSSELL J HATLING	•	1935 W. LUUNIT .	MN 55113
-	144 MELBOURNE AVI	E SE MN 5541	4 ROSEVILLE	
	MINNEAPUL15		055113005661	1 99999 TION CONT AUCY
	055108031221	10N CONT AGEY	MINNESUTA POLLU	IIUN CUNT ADD.
	MINNESOTA POLLUI		EXECUTIVE UIREC	TOR
-	2120 LARTER AVE	MN 5510	08 1435 W COUNTY K	
	ST PAUL		RÖSEVILLE	MN 55113
	•		055407021926	1 99999
: -	055113005908	99999 100 CONT ACC	V MINNESUTA POWER	LE LIGHT CU
	MINNESOTA POLLUT	LUN CONT ACC	J F RUWE	DR STREET EEROS
U	JUCELYN 7 ULSUN SDFLIDI ASST ATI	URNEY GENERA		RN 33802
	1935 W CUUNTY R) ¹¹² WN 551	13	

•	
•	
•	
•	-
•	c ₁ - Line
0	
•	
•	
٠	
۲	
۷	
ŵ	

		- 25115 (* * * * * * * * * * * * * * * * * * *
•	LAG MELOUUTNE AVE 25 MINNEAPOLIS MN 55414	A-2 ROSEVILLE MN 55113
•	055108031221 1 99999 MINNESOTA PULLUTION CONT AGGY STEVEN J GADLER 2120 CARTEY AVE ST PAUL MN 55108	055113005661 1 99999 Minnesuta Pollutio", Cont Aucy Terry Hoffman Executive Director 1935 W County Road
~		NZ NN 55113
-	055113J05908 1 99999 MINNESCTA POLLUTIU4 CUNT AGCV JLEFLYN F ULSUN SPELIAL ASST ATTURNEY GENERAL 1935 M COUNTY RD B2 RUSEVILLE MN 55113	US5302021926 1 99949 MINNESUTA PUWER & LIGHT CU J F RUHF 30 MEST SUPEKIOR STREET 30JLUTH MN 55802
	039205004299 I 99999 MISSISSIPTI BOAHD OF MEALTH DIV OF NAUIULUGICAL HEALTH EDDIE S FUENTE DIFECTOR PO BOX 1700 JACKSON MS 39205	039201064188 l 99999 MISSISSIPPI DEPT ENGY & TRANSP Wilbur G Ball 510 george St Watkins Blug Jacksun MS J9201
	0391500R0965 1 99999 MISSISSIPPI POWER & LIGHT CO GRAND GULF NPS 1 & 2 C K MCLY PLANT MANAGER PU BOX 756 PCPI GIASON MS 39150	039205008656 1 99999 MISSISSIPPI POMER & LIGHT CO J P MCGAUGHY ASST VICE PRES - NUCLEAR PROD PD BCX 1640 MS 39205 JACKSON MS 39205
-	039205001515 1 99999 MISSISSIPPI PONER & LIGHT CO PRUDUCTION NUCLEAR J.P. MCGAUGHY	039205005304 1 99999 MISSISSIPPI POWER & LIGHT CO John Kichardson Po Box 1640 Markson MS 39205
-	ASSISTANT VICE PRESIDENT PC BDX 1640 JACKSON MS 39205	JACK3UN (10 DICO)
c	039205064683 1 99999 MISSISSIPPI POWER & LIGHT CU L F SUDRECK NUCLEAR RECORDS ADMIN PU BOX 1640	065102064891 1 99999 MISSDURI DEPT OF NAT RESOURCES RUNALD A KUCERA DEPUTY DIRECTOR BUX 176 00000000000000000000000000000000000
C	JĀCKŠON MS 39205	JEFFERSUN CITY HU BILLE
6 4	065101001355 1 99999 MISSUUKI PUBLIC SERVICE COM A S LAUGER ASST GENERAL COUNSEL PO BOX 360 JEFFERSUN CITY MO 65101	065102031055 1 99999 MISSOURI PUBLIC SERVICE CUM ELECTRIC DEPT RONALD M FLUEGGE DIRECTOR PO BOX 360 JEFFERSON CITY MO 65102
•	063130006108 1 99999 MISSUURIANS FOR SAFE ENERGY DONALD BOLLINGER MENBER 6267 DELMAR HIVD	095030066310 1 99999 Mollerus Engineering James D Gilcrest 455 LOS Gatos Blyd Suite 208 LOS Gatos CA 95030
C	UNIVERSITY CITY MO 63130	
•	07G130001279 1 99939 MCNKOE & LEMANN W M STEVENSON Esquire 1424 Whitney Building New Ukleans La 70130	J59620005498 1 99999 MONTANA DEPT OF HLTH & ENV SCÌ RADIGL & OCCUP HEALTH PROGRAM LARRY L LLOYD COGSWELL BUILDING R ^M Alija MT E0420
•		HELENA HI 39020 020026005331 1 99999
•	09306305285 ISPVICE MCCRE TECHNICAL SERVICE RUGER M MCCRE 5998 FEAPING STHEET SIMI VALLEY CA 93063	MORGAN LEWIS & BOCKIUS D'S HEISIAND JR ESJUIRE 1800 M STREET NW WASHINGTUN DC 20036
•	027801055296 1 99999 MORRISON KNODSEN CO INC 1400 VANLE STHEET NC 27501 ROCKY MUUNT NC 27501	083729064641 1 99999 Murrison Knudsen CO Inc Power Group Thumas Sauerbrun Yanager Special Projects Two Plaza Po Bux 7808 Doise ID 83729

_

-

-4

•

•

•

.

.

-

U

Y

ت

Aleria	م به مور	and the second sec
027603006553 1	99999 10456_	J27603006527 1 99999 N. LARULINA CLEARINGHOUSE
DIV OF POLICY DEVEL 116 WEST JUNES STREE RALETON	DPMENT T NC 27603	CHRYS BAGGETT 116 WEST JUNES STREET RAIFIGH NC 27603
0225140070+5 1	99 999	027605002503 1 99999
N CAPOLINA CONSERVA	TION COUNCE	N CAROLINA DEPI OF HUN RES DIV OF FACILITY SERVICE DAVNE H BROWN
EXECUTIVE COORDINAT	NC 27514	CHIEF-RADIATN PROTECTION SECTN P U 80X 12200
UPAPEL HILL		RALEIGH NC 27605
027602005817 1 N CAROLINA DEPT OF	INPLICE	N LARULINA ELEC MEMB CORP 333 NURTH BLVD
JUSTICE BUILDING	NC 27602	BUX 27306 RALEIGH NC 27611
027625005487 1 N CAROLINA MPA-1	99999	N CAROLINA GEC OF INTGOVT REL
PO BUX 95162 RALEIGH	NC 27625	RALEIGH NC 27603
027602012785 1	99999 5 1 0M	058501004306 1 99999 N DAKOTA DEPT OF HEALTH
ROBERT FISCHBACH	PUB STAFF	DIV OF ENVIRONMNTL ENGINEERING DANA NOUNT DANA NOUNT
PO BOX 991 RALEIGH	NC 27602	RM 304 BISMARCK ND 58501
028658053729 1	99 999	ODENMA066991 1 99999
RANK J NAPOLI		ENERGY OFFICE BJORN THORLAKSEN
PO BUX 730 NEWTUN	NC 28658	ADMINISTRATOR 29 STRANDGADE 29 JAOL COPENHAGEN K DENMARK
037330045276	99999	043229065206 1 99999
NATIONAL BOARD OF	JOILER INSP	NATIONAL BOARD OF BUILER INSP R JAGGER R JAGGER AVE
ĜIJ RĈĈHELLE DR NASHVILLE	TN 37220	COLUMBUS OH 43229
010102006632	99999	UMEXIC054851 1 99999
NATIONAL LAWYERS C DONALD S BRUNSTEIN	UILD	NATIONAL NUC SETY & SEGUS CUR Roberto Trevino
1425 WALNUT STREET THIRD FL	PA 19102	AVENIDA INSURGENTES SUR 1806 MEXICU 20 DF MEXICO
	00000	016020066661 1 99999
NATIONAL NUCLEAR C SAFETY & LICENSING	GROUP	NATIONAL UNDERGROUND STORAGE
J SCRIVINS CANUKIDGE PD WHET	STUNE	BUYERS PA 16020
LEICESTER LEG DER		010005 105 305 1 99999
NATIONAL VALVE 6 P	IFG CO	NATURAL RESOURCES DEFENSE CACL
BOX 100 PITTSBURGH	PA 15230	WASHINGTON DC 20005
094108007015 1 NATURAL RESOURCES	DEFENSE CNCL	NEBRASKA DEPT OF ENVIR CONTROL DIRECTUR
25 KEAKNY STREET SAN FRANCISCO	CA 94108	PO SUX 94877-STATE HOUSE STATN Lincoln ne 68509

• • • • •

J

0

-

-

		The second of the second se
~	L C LESSEN STATION SUPERINTENDENT P G BOX 98	J M PILANI JIVISICN NANAGER P C 90X 494
	BROWNVILLE NE 68321	COLUMBUS NE 68001
^ ^	068601005851 1 99999 NEBRASNA PULLIC PUWER DISTRICT LICENSING & GUALITY ASSURANCE J M PILANT DIRECTIM PU BUR 499	058501005852 1 99999 NEHRASKA PURLIC PUHEN DISTNICT G D NATSCN GENERAL COUNSEL PJ RUX 499 CJLU-BUS NE 68601
	COLUMNUS NE 68601	
•	DA971U028226 1 99999 Nevada Cunsumer Heth Pout SvC Raciolugical Health Jenn Valen for Futhers	UB9710039883 1 99999 Nevada dept of Energy Dept of Human Resources Ace Martelle
•	KINKEAD BLUG SUS E KING ST	ROOM 600
	CARSON CITY NV 89710	CARSON LITY NV 89710
•	089158012227 1 99999 NEVADA UFPT OF MFALTH & WELFRE BUR UF LUNSUMER HLTH PRUTEC SV LARRY FRANKS RACIATION CUNTROL SPECIALIST 620 RELROSE ST LAS VEGAS NV 99158	005346005884 1 99979 NEW ENGLAND CGAL DN NUC POLLUT HILL & DALE FARM WEST HILL-FARAWAY RD RD #2 HCX 223 PUTNEY VI 05346
•	ODI862063071 I 99999 NEW ENGLAND NUCLEAK CDKP RAU WASTE ENGINEERING GKOUP KEHRY BENNERT 601 TREBLE CUVE RGAD BILLENICA MA 01862	003301001402 1 99999 New HAMPSHIRE ATTORNEY GENERAL E T KINDER ASSISTANT ATTORNEY GENERAL 208 STATE HOUSE ANNEX CUNCORD NH 03331
•	0033J1080913 1 99999 NEW HAMPSHIRE CIVIL DEF AGENCY EILEEN FOLEY DIRECTOR DIRECTOR DNE AIRPORT RUAD CONCORD NH 03301	003301002479 L 99999 New Hampshire Div of Pub Hlth Ofc of Radiation Control Diane Teffi Health & Welfare Bldg-Hazen Dr Concord NH 03301
•	OO3105001410 1 99999 NEW HAMPSHIRE PUB SVC CO BRUCE B BECKLEY PROJECT MANAGER 1000 ELM STREET PG BOX 330 MANCHESTER NH 03105	OJ3105005351 I 99999 NEW HAMPSHIRE PUB SVC CO D G CAMERDN GENERAL COUNSEL IOOO ELM STREET PO BOX 330 MANCHESTER NH 03105
•	003105007087 L 999999 NEW HA4PSHIRE PUB SVC CD STEPHEN D FLOYD PD BOX 330 MANCHESTER NH 03105	003105007099 L 99999 NEW HAMPSHIRE PUB SVC CO STEPHEN D FLOVD PO BOR 330 MANCHESTER NH 03105
•	003105091129 1 99999 NEW MAMPSHIRE PUB SVC CO SIEVEN D FLOYD 1000 ELM STREET PO BGX 330 MANCHESTER NH 03105	003874064759 1 99999 NEW HAMPSHIRE PUB SVC CO DCC H G LIZOITE DCC SUPERVISOR PU BOX 300 SLABROOR NH 03874
•	003105001549 1 99999 NEW MAMPSHIKE PUB SVC CO W C TALL MAN PRESIDENT 1000 LLN STREET PO BCX 330 MANCHESTER NH 031C5	003105022006 1 99999 NEW HAMPSHIRE PUB SVC CO W C TALLMAN CHAIRMAN 1000 ELM STREET PU BUX 330 Manchester NH 03105
•	OD387406476J 1 99999 NEW HAMPSHIKE PUB SVC CU D CC VINCENT VERMAL ADMIN DEPT SUPERVISUR SEABHODK STA PO BOX 300 SEABHOCK NH U3874	OU8628005322 1 99999 NL# JERSEY BUR OF RAD PROTECT GENE FISHER BUREAU CHIEF 380 SCOTTS ROAD TRENTON NJ 08628
L	007102005323 1 99999 NEW JERSEY DEPT OF ENERGY CLMMISSILNER 101 CLMMERCE STREET NEWARK NJ 07102	007102004899 1 99999 NLA JERSEY DEPT OF ENERGY DIV OF ENERGY PLANNING CONSERV LFUNARD S COLEMAN 1100 RAYMOND BLVD

Ĵ

6

Ű

(

Ű

_	ASSISTED LANGE LIVE STATES AND ASSISTED AND A DESTRICT OF STATES AND A	SZÍ EAST STATE STATE NJ. DUDEL TRENTON NJ. DUDEL
•	008625006302 1 99999 NEW JERSEY DEPT OF PURL ADVOCT HERIAM N SPAN ASST DEPUTY PUBLIC ADVCCATE 520 EAST STATE STREET PD RUX 141 NI DAA25	DG8625000042 1 99999 NEW JERSEY DEPT DF PUBL AUVOCT DIV GF PUBL INTEREST ADVOCACY 4004RT WESTREICH PO BOX 141 NJ 08625 THENTUN NJ 08625
•	TRENTLN 19 00000 007102005324 1 99999 New Jeksey Dept of Public Util Cumissioner 101 commerce Street Nemark NJ 07102	008625006389 1 99999 NEW JERSEY DEPUTY ATTNY GENER DEPT UF LAN & PUBLIC SAFETY DEPUTY ATTORNEY GENERAL 36 W STATE ST CN-112 CN-112 THENTON NJ 08625
•	DD8625006301 1 99999 New Jersey Deputy Attny Gener Richard M Hluchan Remecca # Fields State House Annex nj 08625 Trenton nj 08625	007102006455 1 99999 New Jersey Nuclear Energy CNCL Makk L FIRST Deputy Atturney general 36 Mest State Street Trenton NJ 07102
•	087503000003 1 99999 NEW PEXICU ASST ATTY GENERAL RAND L GREENFIELD ASSISTANT ATTURNEY GENERAL BATAAN MEMORIAL BUILDING SANTA FE NM 87503	D87504D31209 1 99999 NEW MEXICO DEPT OF HLTH & ENV RADIATIUN PRUTECTIUN BUREAU Alphonso A TOPP CHIEF PO BUX 968 NM 87504 Santa FE NM 87504
•	070160005708 1 99999 New Grleans Public Service CC M L Purstell Vice President-eng & Prod 317 Barunne St New Grleans La 70160	010047081681 1 99999 NEW YORK ATTORNEY GEN OFFICE DIV-CONSUMER FRAUD-ENGY & UTI RUBERT ROTH 2 WORLD TRADE CENTER RODM 4615 NEW YORK NY 1004
•	010013055936 1 99999 New YORK CITY DEPT OF HEALTH BUREAU FOR RADIATION CONTROL LEONAKD SOLON DIRECTOR 65 WERTH ST NY 10013	012233002487 1 99999 NEW YURK DEPT OF ENVIRON CONS TOXIC & RADIATION SECTION THOMAS J CASHNAN CHIEF 50 WGLF ROAD ALBANY NY 1223
•	NEW TOAR 012237009723 1 99999 NEW YORK DEPT OF HEALTH FNV RADIATION PROTECTION BERNARD A HEALD CHIEF-ENVIRUN RAD SECTION CHIEF-ENVIRUN RAD SECTION EMPIRE STATE PLAZA TOWER BLDG AI RANY NY 12237	010047009730 1 99999 New Yurk Dept of Labor Radiological Health UNIT Francis J Bradley Principal Radiophysicist 2 World Trade Center New York NY 1004
G	010047005352 1 99999 New York Dept of Law Environmental Protection Bur Exa & Bialik Assistant Attorney General 2 Nord Trade Center New York	012223005396 1 99999 New Yurk Energy Office Technical Development Program Director Empire State Plaza Aggy Blug Albany Ny 1222
•	012223059613 1 99999 NEW YORK ENERGY UFFICE WILLIAM DAVIS DEPUTY COMMISSIONER AGCY BLOG 2 EMPIRE STATE PLAZA 10TH FLUOR ALHANY NY 12223	012223001367 1 99999 New York Energy Office Technological Development Pri t & Debuer Director Empire State Plaza Albany Ny 122
	012223005892 5 99999 New YGRK ENERGY OFFICE TECH DEVELOPMENT PRÜG J Dunkleburger Acting Dikeltor 2 Rockefeller Plaza Agcy BLOG2 Aluany Ny 12223	012223000090 1 99999 NEW YORK ENERGY OFFICE JAY DUNKLEBURGER DEPUTY CUMMISSIONER & COUNSE EMPIKE STATE PLAZA AGY BLOG ALBANY NY 122
ل ب	012223010637 5 99999 New Yurk Erda West Valley Program T K Derolr Director 2 Rockefeller Plaza Alony 4 12223	UIUUI9005733 1 99999 NEW YORK POWER AUTHONITY NUCLEAR GENERATION J P RAYNE Senidr Vice President 10 Columbus Circle New Yurk NY 100

i. J

....

99999 F PUBL AUVOCT EST ADVOCACY NJ 08625 99999 ATTNY GENER HIC SAFETY FENERAL HI2 NJ 08625 99999 AR ENERGY CNCL GENERAL Reet NJ 07102 99999 DF HLTH & ENV Tiun Bureau NM 87504 99999 Y GEN OFFICE UD-ENGY & UTIL NTER NY 10047 99999 ENVIRON CONSV N SECTION NY 12233 L 99999 FLABOR ALTH UNIT FYSICIST ENTER NY 10047 1 99999 OFFICE DPMENT PROGRAMS AZA AGCY BLUG Z Ny 12223 1 99999 OFFICE EVELDPMENT PROG AZA NY 12223 1 99999 OFFICE ER DNER & COUNSEL AZA AGY BLUG 2 NY 12223 I 99999 AUTHORITY FLON ESIDENT RCLE NY 10019

-

^

۵

•

8

•

•

•

0

0

 \odot

Ü

•

.

و

•	NEW YURK NY IGUIS	BUCHANAN NY ISSI
4	010511079439 1 99999 New York Pluer Authority Indian Pcint 3 NUC Par Plant	DIODI9006537 I 99499 Nëm Yukk Pomer Authority A Alausmann Vice Prestdent-Cuality Assuf
•	N D MARLEA ASST TU RESIDENT MANAGER PU 64X 215 BUCHANAN NY 10511	10 COLUMPUS CIRCLE NY 10019
•	013093036535 1 99999 New York Purer Authority James & Fitzpatrick NUC Pwr Pl Curbin & McNell	D10019005735 1 99999 New YCRK POWEH AUTHORITY CHARLES M PHATT ASSISTANT GENERAL COUNSEL
•	RESIDENT MANAGER PO BOX 41 LYCUMING NY 13093	NEW YORK NY 10019 New York NY 10019
•	010019003250 1 99999 New York Power Authority Lergy Sinclair President	DIGD19005736 1 99999 New York Puwer Authority Lerdy & Sinclair President & Chief Operate DFC
•	10 COLUMBUS CIRCLE NY 13019 New Yürk Ny 13019	NEW YORK NY 10019
•	010019035628 1 99999 New York Power Authority Technical Library Saretta Spaner Lu Columbus Circle New York NY 10019	010019001229 1 99999 New YORK POWER AUTHORITY GEDRGE M WLYFRDING CHAIRMAN-SFTY REVIEW COMMITTEE 10 COLUMBUS CIRCLE New YORK NY 10019
~	010019006539 1 99999 New York Power Authurity Geurge M Wilverding Chairman-Safety Review Comm 10 Cilumbus Circle Ny 10019	O13093G64674 1 99999 New YURK POWER AUTHORITY JAMES A FITZPATRICK NUC PWR PL E L ZUFELT UPERATING DOCUMENT COORD PD BOX 44 NY 13093
	D10037031760 L 999999 NEM YURK PUBLIC INT RES GROUP INDIAN POINT PROJECT 9 HURRAY ST NEW YORK NY 10007	010038005688 1 99999 NEW YCRK PUBLIC INT RES GROUP JOAN HOLT PRUJECT DIRECTOR 5 BLEKMAN STREET NEW YORK NY 10038
ر ب ب	012223081572 1 99999 New York Public Service CCM Power Division Director 3 Empire State Plaza Albany ny 12223	012223001605 1 99999 New York Public Service Cum Matthew J Kelly Rockefeller Empire State Plaza Albany Ny 12223
tu	013902003164 l 99999 New York State Elec & Gas CC Richard L Anderson 4500 Vestal Parkway E Binghanton Ny 13902	013902064610 1 99999 New York State Elec L GAS CO J J Marvilla Project Engineer 4500 Vestal Parkway East Binghampton Ny 13902
•	023607038036 1 99999 NEHPURT NEWS INDUSTRIAL COMP BUSINESS DEVELOPMENT T G REEFE 230 AIST STREET NEMPURT NEWS VA 23607	023607053075 I 99999 NEWPORT NEWS SHIPBUILDING R T CLARK JR 4101 WASHINGTON AVE NEWPORT NEWS VA 23607
•	D11747037563 1 99949 NEWSDAY STUART DIAMOND 235 PINELAWN PGAD MELVILLE NY 11747	090017357903 1 99999 NHK SPRING COLTU 700 SDUTH FLOWER ST SUITE 1900 LOS ANGELES CA 90017
•	013093030942 1 99999 Niagara Mghawk Pumer Curp MmPC Ja Pii 80% 53 Lycuming Ny 13093	OI3093012251 1 99999 NIAGARA MOHAWK POWER CORP NINE MILE POINT UNIT 1 John Aldrich Supervision-upperations PO HCX 32 NUMBER 1500
_		· · · · · · · · · · · · · · · · · · ·

•

0

ر

	PI, 35,8 52 LYCOMING NY 13093	SVRALUSE AV AND AV ISECO
•	013093079362 1 99999 NIAGANA MEMANK PUNER CLRP W. DRENS SUDERINTENDENT	013202007118 1 99999 Niagara Muhawk Power Corp Juhn W Keið Fsj
•	PU BUX 32 LYCUMING NY 13093	300 ER IE BOULEVARD W Syracuse ny 13202
•	013202003191 1 99497 NIAGARA MUMAWK POWER CGRP NUCLEAR GENERATION THUMAS E LEMPGES 300 ENTE BLVO WEST	013202004687 1 99949 NIAGARA MOMAWK POWER CORP NUCLEAR ENGRN & LICENSING C V MANGAN VICE PRESIDENT
•	CI SYRACUSE NY 13202	300 FRIE BOULEVARD WEST SYRACUSE NY 13202
n	D13093064721 1 99999 NIAGARA MUHAWK POWER CURP D L MAXWELL RECORUS MANAGEMENT SUPERVISUR PU BOX 63 LYCOMING NY 13093	013093079363 1 99999 NIAGARA MUMAWK POWER CORP NULLEAR GENERATION I PERKINS GENERAL SUPERINTENDENT PU BUX 32 LYCOMING NY 13093
•	013093007113 1 99999 NIAGARA MUHAMK POWER CORP THOMAS PERKINS STATION SUPERINTENDENT PU BOX 32 Lycoming NY 13093	DI3202001371 I 99999 NIAGARA MOMAWK POWEH CORP ENGINEERING G RHODE VICE PRESIDENT 300 ERIE BUULEVARD WEST SVRACUSE NY 13202
•	013202081571 1 99999 NIAGARA MOMAWK POWER CORP System Project Management Gekald & Rhode Senicr Vice President 300 Erie Buulevard W Syraluse Ny 13202	D13202028142 1 99999 NIAGARA MOHAWK POWER CORP System Security Joseph J Sunser 300 Erie Boulevard West 8-1 Sykacuse ny 13202
•	OITALYOBOB92 I 99999 NIRA LIBRARY VIA DEI PESCATURI 35 GENOVA 16129 ITALY	006385005407 1 99999 NORTHEAST NUCLEAR ENERGY CO Millstene plant Superintendent PU BOX 128 Materford CT 06385
•	006101001556 1 99999 NORTHEAST NULLEAR ENERGY CO NULLEAR ENGR & OPERATIONS W G LOUNSIL SENICR VICE-PRESIDENT PD BUX 270 MARTFORD CT 06101	006101006571 1 99999 NORTHEAST NUCLEAR ENERGY CO D G DIEDRICK Manager of Guality Assurance PO BOX 270 Martford CT G6101
•	DOGIOIO79359 1 99999 NORTHEAST NUCLEAR ENERGY CO D G DIEDRICK MALACER OF JUALITY ASSURANCE PD BUX 270 MARTFORD CT 06101	006101080932 1 99999 NORTHEAST NUCLEAR ENERGY CU GENERATION CONSTRUCTION E R FOSTER DIRECTUR PU BOX 270 MARTFORD CT 06101
U	OU6385080931 1 99999 NORTHEAST NUCLEAR ENERGY CO CONSTRUCTION QUALITY ASSURANCE K N GRAY CUBERSISON	006385026372 1 99999 Nurtheast Nuclear Energy CD Jack Provencal PU Box 123 Materond CT 06385
Ð	PO BOX 128 WATEKFJKD CT 06385	
U L	006101006803 1 99999 NOPTHEAST UTILITIES SERVICE CO GENERATION ENGR & CONSTR DIV M G CUUNSIL SH VICL PRESIDENT PO BUX 270 MARTHORD CT 06101	006101064595 1 99999 NORTHEAST UTILITIES SEFVICE CU R M DAVENPORT MGR-GENERATION FACILITIES RECS PU BOX 270 HARTFURD CT 06101
•	ODDIDIDO6249 1 99999 NORTHEAST UTILITIES SERVICE CO GENERATION FAC LICENSING RICHAPD T LAUDEVAT MANAGER PD RUX 270 HANTERPD CT 06101	006101005409 1 99999 NORTHEAST UTILITIES SERVICE CO JUHN F UPEKA Systems Superintendent PU BCX 270 HARTFORD CT 06101

eð J J

J

^

7

0

ŋ

-

		e de la companya de la
•	SH VILL SHEJIJENT PG VILL 273 MANTEGRA	HARTFORD CT UDIL
	006101006249 1 99999 NORTHEAST UTILITIES SERVICE CC LENERATION FAC LICENSING	UD5101005409 1 99999 NURTHEAST UTILITIES SERVICE CC JUMN F UPERA Systems Superintendent
•	RICHARDI LEGULART MANAGER PO BUS 270 MARTEURO CT JOLUL	PL 827 270 HANTEORD CT 96131
•	006101005752 1 99999 NURTHEAST UTILITIES SERVICE CO JOHN DPEKA VICE PHELIDENT-NULLEAR OPERATN PC MOR 270 HARTEURD CT 06101	046325004674 1 99999 NGRTHERN INDIANA PUB SVC CD Hailly I E m Smgrð Vice President 5265 Húhnan Avenue Mamponi
•	D55362005659 1 99999 NURTMERN STATES POMER CO MONTICELLO MUC GENERATING PLNT PLANT MANAGER MUNTICELLO NN 55362	055401079386 1 99999 NORTHERN STATES POWER CO C E LARSON DIRECTUR-NUCLEAR GENERATION 414 NICOLET MALL MINNEAPOLIS NN 55401
•	0554G1003203 1 99999 NDATHERN STATES POWER CU NULLEAP SUPPURI SERVICES LLE D MAYER MANAGER 414 NEGULET MALL-BTH FLUCR 514 NEGULET MALL-BTH FLUCR	055401005657 1 99999 WIRTMERN STATES POWER CC NUCLEAR SUPPORT SERVICES D M MySCLF 414 NICOLLET MALL 814 FLCUR MINN-APOLIS RN 55401
•	NINNEAPULIS 99999 OSSOB9064703 1 99999 NUPIMERN STATES POWER LU T & REDING DOCUMENT CONTROL SUPERVISOR DOCUMENT CONTROL SUPERVISOR	055401005658 1 99999 NORTHERN JATES POWER CO ARTHUR RENGUIST VICE PRESIDENT-LAN 414 NICOLLET MALL
•	WELCH MN 55089	MINNEAPOLIS MN 55401
•	055362081509 1 99999 NORTHERN STATES POWER CO MUNTICELLO NPS W A SHAMLA	020036006476 I 94444 Northern States Power CO Jay & Silberg Counsel
•	PLANT MANAGER MONTICELLO MN 55302	WASHINGTON DC 20036
6	055089006147 1 99999 NURTHERN STATES POWER CO PRAIRIE ISLAND NUC GEN PLANT F P TIERNEY PLANT MANAGER	055089081554 1 99999 NTRTHERN STATES POWER CO PRAIRIE ISLAND NPS E L WATZL PLANT MANAGER
L	RGUTE 2 MN 55089	RC 2 WELCH NN 55089
6	098033064609 1 99999 NURTHWEST ENERGY SERVICES CO J F HANNUM RECORDS MANAGER PU BOX 1090	098033006268 1 99999 NORIMEST ENERGY SERVICES CO ENVIRONMENTAL LICENSING J E MICCA MANAGER
•	KIRKLANÓ WA 98033	PU 30K 1090 KIRKLAND HA 98033
•	098033004556 l 99999 Northwest Emergy Sfrylces CO Nucleak Licnsing C Safety James Mecla	04003007017 L 99999 Northwest Energy Services CO Frank Spangenberg Assi Pruject Manager-Nuclear
٠	MANAGER PU BUX LOPO KIRKLAND wA 98033	KIRKLAND WA 98033
•	078758053077 1 99999 NPS INDUSTRIES INC NULLEAR POMER SERVICE INC TERNY J OCUNNELL PLANT MCR SE Q A ISBAT MCR SE Q A	030329064094 1 99999 Muclear Assurance Corp Information Center Manuy Reinhold 24 Eaecutive Park West Ailanta Ga 30329
•	AUSTIN TX 79758	006810026537 1 99999
	UDBIDDITES LEVICES INC ANDULAR ENERGY SERVICES INC ALBERT & RUEFL MANADIR-INSPECTION PROGRAMS SHELTER RUER NUAD IT DANID	NUCLÉAR ÉNÉRGY ŠERVICEŠ ÍNČ Adam m Levin Staff Nucléar Engineer Smelter Rock Röad Oanher Ct 06910

.

~

		-
~	006810028328 1 99999 NUCLEAR ELEGY SERVICES INC OTTU SAALHORN SMELTER ROCK PORD DANNUE: CT 06410	DOBBLDOBBOBT 1 99999 "DILLEAM ENERGY SERVICES INC MARKETING RUNALD ZEMPER DIRECTUR OF MARKETING SHELTER RUCK RUAD DANBURY CT 36810
•	037650001707 1 99999 NULLEAR FULL SERVICES INC W C MANSER PLANT MANAUER ERMIN TN 37650	DENGLADG3669 I 99999 NULLEAN INSTALLATIONS INSPECT HEALTH & SAFETY EXECUTIVE DH MARBISUN ADMINISTRATUR THANES HOUSE NORTH-MILLBANK LUNDON SWIP 40J ENGLAND
•	OENGLAOS7563 1 99999 NUCLFAR INSTALLATIONS INSPECT DEPT OF MEALTH & SAFETY 5 HAKBISUN OVERSEAS LIASON OFFICER TMAMES HUUSE NORTH MILLBANK LONDON SWIP 46J ENGLAND	UD1742066518 1 99999 HJCLEAN METALS INC DAVID J ALLARD SUPERVISOR OF HEALTH PHYSICS 2229 MAIN ST CURCURD MA 01742
•	001742063894 I 99999 NUCLEAR METALS INC FRANK J VUMBACO Managen-mealth g Rauiatn Sfty 2229 Main St Concord Ma 01742	DENGLA065242 1 99999 NUCLEAR POWER CO LTD L R KATZ CAMBRIDGE RD - WHETSTONE LEICESTER LEB 3LH ENGLAND
	030075066030 1 99999 NUCLEAR POWER CONSULTANTS INC ROBENT M COMPTUN 230 THISTLEWOOD LANE ROSWELL GA 30075	080012006790 1 99999 NUCLEAR POWER EXPERIENCE INC JAMES F FRANKS Managing Editur PO BUX 2612 Denver CO 80012
•	092668066347 1 99999 NUCLEAR POWER SERVICES INC BANK OF AMERICA BLDG ENGINEERING MANAGER ONE CITY BLVD WEST SUITE 1423 DRANGE CA 92668	DO7094029005 1 99999 NUCLEAR POWER SERVICES INC TECHNICAL LIBRARY SUSAN COHEN ONE HARMON PLAZA SECAUCUS NJ 07094
•	018976057283 1 99999 Nuclear Research Corp NRC Industries Made Paffrath Product Manager 125 Titus Ave Marrington pa 18976	OJAPANO30174 1 99999 NUCLEAR SAFETY BUREAU STA DIV-REACTOR REGULATION KENICHI MURAKAMI 2-1 KASUMIGASEKI 2-CHOME CHIYUDA-KU IGKYU 100 JAPAN
ບ ປ	095131055479 1 99999 NUCLEUNICS CORP JOHN KANDALL AJMINISTRATIVE SUPVR 2240 LUNDY AVE SAN JOSE CA 95131	033515005806 1 99999 NJS CORP Manager LIS Manager LIS CLEARMATER FL 33515
ل ر	020878064606 1 99999 NUS CORP I L GUDWIN 910 CLUPPER RD GAITHERSBURG MD 20878	033515006269 1 99999 NUS CORP J E HOUGHTALING 2536 COUNTRYSIDE BLVD CLEARWATER FL 33515
	033515005436 1 99999 NUS CURP J C Plunkett 2536 Cuuntryside Blyd Clearwater Fl 33515	020878064440 l 99999 NUS CORP R C ROSSI Manager Management Systems 910 Clopper RD Gaitmersburg ND 20878
	020878066977 1 99999 NUS CSRP STEVE SAGATIES 910 CLOPPER ROAD GAITHEESBUNG MD 20078	020878064660 1 99999 NUS CURP M P SENASALK STAFF ANALYST 910 CLOPPER RD GAITHERSBURG MD 20878
	00000	005119057203 1 99999

n ۲ ŵ

1

~	ELEAKNATEN +2 33515	US ELOPPER RE GAITHERSBURG MD 20073
•	020878066377 1 99999 NUS CCKP STEVE SAGATIES	020878064660 1 99999 NUS CUPP M P SENASACK
-	910 CLOPPER ROAD GAITHERSBURG ND 20278	STAFF AMALTST 910 CLOPPER RD GAITHERSBURG MD 20878
•	020814066371 1 99999 NUTECH INC MARCIA EDMARDS	095119057203 1 99999 Nutech inc Rosemary fox
•	SUITE 1300 BETHESDA MD 20814	SAN JOSE CA 95119
•	003105001411 1 99999 D'NEILL BACKUS SPIELMAN ET AL ROBERT A BACKUS 116 LOWELL STREET MANCHESTER NH 03105	037830051560 l 99999 DAK RIDGE NATIONAL LAB NUCLEAR SAFETY CENTER PO BOX Y BLDG-9711-1 DAK RIDGE TN 37830
•		017030044515 1 00000
•	DAK RIDGE NATIONAL LAB C A BURCHSTED BLDG 9204-1 MS-10 PD BUX Y UAK RIDGE TN 37830	OAK RIDGE NATIONAL LAB A L LOITS Director—NRC Programs PO BOX X Gak Ridge TN 37830
•	044060006447 I 99993 OCRE SUE HIATT INTERIM REPRESENTATIVE B275 MUNSON MENTOR OH 44060	017120005699 1 99999 UFFICE OF CONSUMER ADVOCATE IRWIN A POPOWSKY 1425 STRAWBERRY SQUARE MARRISBURG PA 17120
•	032211006307 l 99999 OFFSHORE POWER SYSTEMS VINCENT W CAMPBELL VICE PRESIDENT & GEN COUNSEL 8000 ARLINGTON EXPRESSWAY BUX 8000 JACKSONVILLE FL 32211	032211006308 1 99999 OFFSHORE POWER SYSTEMS A R COLLIER PRESIDENT 8000 ARLINGTON EXPRESSWAY 80X 9000 JACKSONVILLE FL 32211
•	032211006306 1 99999 UFFSHURE POWER SYSTEMS THOMAS M DAUGHERTY 8000 ARLINGTON EXPRESSWAY BOX 8000 JACKSUNVILLE FL 32211	032211008703 1 99999 UFFSMURÉ POWER SYSTEMS A P ZECHELLA PRESIDENT 8000 ARLINGTON EXPRESSWAY JACKSONVILLE FL 32211
•	043216005786 1 99999 DHIU DEPT OF HEALTH Radiolugical Health Program Director PU Bux 118 Columbus OH 43216	043216012023 1 99999 DHID DEPT OF HEALTH RADIOLOGICAL HEALTH PROGRAM ROBERT M QUILLIN DIRECTOR 246 N HIGH ST PD 80X 118 CULUNBUS DH 43216
•	043216031506 l 99999 DMIO DEPT OF INDUST RELATIONS DIV OF POMER GENERATION HELEN W EVANS DINECTOR PD BLX 825 COLUMENUS DH 43216	D43085066715 1 99999 UHID DISASTER SERVICES DIV JAMES R WILLIAMS HUCLEAR PREPAREDNESS OFFICER 2825 GRAVVILLE RD WURTHINGTON DH 43085
6. Li	044308008704 1 99999 UMID EDISON CU LYNN FIRESTONE VICE PRESIDENT 76 5 MAIN ST AKRON OH 44308	043216005705 1 99999 UHIO ENVIRON PROTECT AGCY DIVISION OF PLANNING ENVIRONMENTAL ASSESSMENT SEC PO BOX 1049 COLUMBUS OH 43216
۔ ر	0432150057R0 1 99999 DHIO UFC CF ATTORNEY GENERAL DEPARTMENT OF ATTORNEY GENERAL 30 EAST BRUAD STREET COLUMMUS OH 43215-	043216005781 1 99999 Umid Power Siting Board Marold Kamn Staff Scientist 361 e Bruad Street Culumbus OH 43216

0

Ö

U

Û

U

J

 \odot

JU EAST SHEAT STALLT COLUMPUS OH 43215	STAFF SCIENTIST 361 E BRUAD STREET CULUMBUS OH 43216
073105005189 1 79999 OKLAHCHA ATTOPNEY GENERAL UFC JAN L GARTWRIGHT ATTOPNEY GENERAL 112 STATE CAPITUL BUILDING UKLAHUMA CITY OK 73105	073152012022 1 99999 UKLAHUMA DEPT OF HEALTH NCCUP & RADIOLCGICAL HETH SERV J UALE MCHARD CHIEF PU BOX 53551 OKLAHUMA CITY OK 73152
073191022851 1 99999 UKLAHOMA GAS & ELECTRIC TOM HOKE PC BUX 321-MC444 UKLAHUMA CITY OK 73101	074102005180 1 93999 UKLAHOMA PUBLIC SERV CC VAUGHN L CONRAU PU BUX 201 TULSA OK 741J2
068102005777 1 99999 OMAMA PUBLIC POHER DISTRICT ENVIHON & REG AFFAIRS DIV MANAGER 1623 HARNEY ST UMAMA NE 68102	068023003214 I 99999 Umama Public Poner District Fort Calhoun Station W G Gates Nanager PG Box 399 Furt Calhoun Ne 68023
068102006282 1 99999 OMAHA PUBLIC POMER DISTRICT PRODUCTION OPEKATIONS W C JONES DIVISION NANAGER 1623 HARNEY STREET OMAHA NE 68102	068102022680 1 99999 OMAMA PUBLIC POWER DISTRICT PRODUCTION OPERATIONS WILLIAM C JONES DIVISION MANAGER 1023 HARNEY STREET UMAHA NE 68102
068102064619 l 99999 Omama Public Power District B R Livingston Manager Admin Services 1623 Harney Street Omama ne 68102	D68102080934 1 99939 UMAMA PUBLIC POWER DISTRICT FURT CALMOUN STATION S C STEVENS MANAGER PC BOX 399 FURT CALMOUN NE 68102
037219000773 1 99999 DPER ELLIS & BRABSON LEROY J ELLIS 111 ESQUIRE CHANCERY BLDG - 421 CHARLDITE NASHVILLE TN 37219	OCANADO64709 1 99999 GNTARIO HYDRO NUCLEAP GENERATION DIVISION R J KELLY REACTOR SAFETY ENGINEER 700 UNIVERSITY AVE-TGRGNTO GNTARIU M5GIX6 CANADA
097310056502 1 99999 DREGEN DEPT OF ENERGY SITING & REGULATION DONALD W GODARG 102 LABUR & INDUSTRIES BLDG ROCM 111 Salen OR 97310	D97207002513 1 99999 OREGON DEPT OF HUMAN RESUURCES DIV OF HEALTH-RADIATN CNTL SVC MARSHALL W PARRUIT SECTION MANAGER P O BOX 231 PORTLAND OR 97207
097234005965 1 99999 Oregon dept of justice Richard M Sandvik 520 SW YANHILL Portland or 97204	077701000770 1 99999 Okgain Bell & Tucker Stanley Plettman Esguire Beaumont Savings RLDG Beaumont Tx 77701
094111006011 1 999999 URRICK MERRINGTON & SUTCLIFFE DAVID R PIGDTT 600 MONIGOMEKY STREET SAN FRANCISCO CA 94111	094111066113 1 99999 DRRICK MERRINGTON & SUTCLIFFE ALAN C WALTNER 6UJ MUNTGOMERY ST SAN FRANCISCO CA 94111
020006031123 L 99999 DVERSEAS ELEC INDUS SURVEY INS MASHINGTON UFFICE M TAKAHASHI ASSISTANT GENERAL MANAGER IOLS 18TH STREET NN Ø E20 MASHINGTUN DC 20006	094106064589 1 99999 PALIFIC GAS & ELECTRIC CO P C BURGESS RECORUS MANAGMENT SUPVR 77 BEALE STREET ROOM 2404 SAN FRANCISCO CA 94106
094106064688 1 99999 PACIFIC GAS & ELECTRIC CU R P CHAN PRUJECT MANAGER 77 BEALE STREET ROOM 2459 SAN SERVEISC TO 94106	094120005996 1 99999 PACIFIC GAS & FLECTRIC CU PHILIP A GRANE 77 BEALE ST 31ST FLUOR SAN FRANCISCO CA 94120

ور

 $\widehat{}$

e

٤.

1	1993 1200 - 200 200 22 March 6 - 200 13 - 100 265 13	SAN FRANCISCU LA MAIZU
•	SAN FRANCISCU CA 94106	
•	094136003235 1 99999 PALIFIC GAS & ELECTRIC CU Balific GAS & ELECTRIC CU	020036006524 1 99999 PACIFIL GAS & ELECTRIC CO DWEN H DAVIS
	VILE PRES & GENERAL COUNSEL 77 BEALE ST BIST FLOOK	DIRECTUR-FEDERAL AGENCY RELAIN 1950 17TH STREET NH SUITE 1160
-	SAN FRANCISCU CA 94106	HASHINGTON UL 20036
•	094120007065 I 99999 Pacific Gas 6 Electric CO Malcclm H Furbush Malcclm H Furbush	PACIFIC GAS & ELECTRIC CU ENGR RESEARCH DEPT
•	77 REALE ST PO BOX 7442 SAN FRANCISCO CA 94120	3400 CRUM CANYON RO SAN RAMON CA 94583
	093424003219 1 99999	094106064764 1 99999
	PACIFIC GAS & ELECTRIC CO Diablo canyon power plant W B NAEFER plant work	ENGG CUALITY CONTROL C E RALSTON
	TECH ASST TO PLANT MOR P U BOX 56 AVILA BEACH CA 93424	ZOG 215 NARKET ST SAN FRANCISCO CA 94106
ſ	094120005994 1 99999 Pacific Gas & Electric CO Nuclear Generation Dept	094106005747 1 99999 Pacific GAS & Electric Co Nuc plant operater IIM Shiffer
•	VICE PRESIDENT PD BOX 7442	77 BEALE ST RUOM 1485 SAN ERANTISCH CA 94106
	SAN FRANCISCO CA 94120 005501057689 1 99999	097403007015 1 99999
۲	PACIFIC GAS & ELECTRIC CU HUMBOLOT BAY POWER PLANT F D WEEKS	PACIFIC NORTHWEST RESUURCES TERENCE L THATCHER LAW_CENTER 1101 KINGAID
•	PLANT SUPERTENDENT 1034 oth Street Fureka ca 95501	EUGENE UK 97403
	090255053081 1 99999	092803053082 1 99999 PACIELC SCIENTIFIC CO
•	DIVISIUN OF DRESSER INDUSTRIES D.B.HARNEY	KIN TECH DIVISION J F DOWOY NAMES NUMBER PRODUCTS
5	PRESIDENT 7515 BICKETT STREET MUNTINGTON PARK CA 90255	1346 SOUTH STATE COLLEGE BLVD ANAHEIM CA 92803
	032020052945 1 99999 PACIFIC SCIENTIFIC CO	090807053083 1 99999 PACIFIC VALVES INC
•	BÉĽFAB DÍV R R FAIR O A MCR	UA MANAGER 3201 WALNUT AVENUE Long Beach ca 90807
٠	P 0 BUX 9370 305 FENTRESS BLVD Daytuna beach fl 32020	
	029205007019 1 99999 PALMETTO ALLIANCE	007094064697 1 99999 Panasonic Co
•	2135 1/2 DEVINE ST COLUMBIA SC 29205	NATIONAL SALES MANAGER ONE PANASONIC WAY
•		SECAUCUS NJ 07094
	017057007108 1 99999 PANE	035810081026 I 99999 PATEL ENGINEERS PATEL ENGINEERS
•	BDARD OF DIRECTORS PU BOX 268 MIDDLETUUM PA 17057	BUX 3531 HUNTSVILLE AL 35810
•	MIDDLEIDWA (A 1103)	
	017105005431 1 99999 PATH LUT NEWS	092668065239 1 99999 PAUL MUNKOE HYDRAULICS INC
-	RICHARD RUBERTS BLZ MARKET STREET Marketsung pa 17105	MGR-ENERGY PRODUCTS DIV 1701 W SECUDIA AVE
۲	HARLSOUND IN 11107	ŪRÁŇGE CA 92668
-	019016053114 1 99999 PENN SHIP CU	017105005429 1 99999 PENNSYLVANIA DEPT OF ENVIR BES
•	INDUSTRIAL PRODUCTS DIVISION RICHARD H HAGAN	UFFICE OF PADIOLUGICAL HEALTH Director Po Rox 2063
ų.	NURTUN AVE DA 1901A	HARAISBURG PA 17105

.

A

.

2

-

•

•

•

•	019016053114 I 99999 PENN SHEP CO INDUSTRIAL PROCUCTS DIVISION RICHAND H HAGAN MGR-PROGRAM CONTROL	U17105005429 1 99999 PENNSYLVANIA UEPT UF ENVIR KES UFFICE UF PADIOLUGICAL HEALTH DIRECTOR PO BUX 2063 PA 17105
•	CHESTER PA 19016	
•	015120005468 1 99999 PENNSYLVANIA DEPT OF ENVIH RES RUHERT & ADLER 505 LEKEUTIVE MOUSE PU B 2357 HARRISBURG PA 15120	OL7120006097 I 99999 PENNSYLVANIA DEPT OF ENVIR MES HUREAU OF ADMIN ENFORCEMENT KAPIN M CARTER SPECIAL ASST ATTY GENERAL 505 EXECUTIVE MUUSE MARRISBURG PA 17120
•	017120001446 1 95999 PENNSYLVANIA DEPT OF ENVIR RES BUKEAU OF KADIATION PROTECTION THOPAS M GERUSKY DIKELTOR PC 854,2063	017120031447 1 99999 PENNSYLVANIA DEPT UF ENVIR RES BUREAU UF RADIATION PROTECTION THOMAS M GERUSKY DIRECTOR PO BOX 2063
-	HARRISBURG PA 17120	HARRISBURG PA 1/120
•	D17120007086 L 99999 PENNSYLVANIA DEPT OF ENVIR RES DFC OF ENVIRONMENTAL PLANNING DAVID HESS PU BOX 2063 HARRISBURG PA 17120	UL7120005420 1 99999 PENNSYLVANIA DEPT OF JUSTICE WALTER W COMEN CONSUMER ADVOCATE 1475 STRAWBERKY SQUARE HARRISBURG PA 17120
•	OI7120006441 I 99999 PENNSYLVANIA EMERG MGMT AGENCY DIRECTOR TRANSPURTATION & SFTY BLDG BASEMENT MARRISBURG PA 17120	017120081576 L 99999 PENNSYLVANIA OFC CONSUMER ADV MARTHA BUSH L475 STRAWEERRY SQ HARRISBURG PA 17120
~	Q17120005451 1 99999 PENNSYLVANIA OFC OF GOVERNOR PLANNING & DEVELOPMENT COURDINATOM: PA CLEAKINGHOUSE PU BOX 1323 ADDISERVES	018603064720 I 99999 PENNSYLVANIA POWER & LIGHT CO SUSQUEHANNA STEAM ELEC STATION SUPVN NUCLEAR RECORDS SYS PU BOX 467
Ç	HARRISBURG PA 17120	BERWICK PA 18603
C B	018101001489 I 99999 PENNSYLVANIA POWER & LIGHT CO WILLIAM E BARBERICH SUPV-NUCLEAR LIC 2 NORTH NINTH STREET ALLENTUWN PA 18101	DIBIOIGOG626 I 99999 PENNSYLVANIA POMER & LIGHT CO N W CURTIS 2 NORTH 9TH STREET ALLENTOWN PA 1BIOI
•	OIBIOLUOBTUS I 99999 PENNSYLVANIA POWER C LIGHT CO ENGINEERING C CONSTRUCTION NURMAN W CURTIS VICE PRESIDENT 2 NGRIM NINIH STREET ALLENTOMN PA 18101	O18101064665 1 99999 Pennsylvania powek & light Co D M Dewalt Supv-records systems & proced 2 North Ninth Street Allentown PA 18101
•	018603009311 1 99999 PENNSYLVANIA POWER & LIGHT CO H W KEISEK SUPT UF PLANT PU BUX 467 BER-17 PA 18603	DIBIOIOBISGO I 99999 PENNSYLVANIA POMER & LIGHT CO BRUCE D KENYON VICE PRESIDENT-NUCLEAR OPER 2 N NINTH SI ALLENTOHN
•	SCHWICK PR 10003	ALCOLUNG PA 18101
•	O18101064722 1 99999 PENNSYLVANIA PGWER G LIGHT CO V M MCNABB SUPVR NUC RECURDS SYSTEM TWO N NINTH STREET ALLENTOWN PA 18101	018101001490 1 99999 PENNSYLVANIA POWER & LIGHT CO EDWARD M NAGEL GENERAL COUNSEL & SECRETARY 2 NORTH NINTH STREET ALLENTOWN PA 18101
•	O18101064629 1 99999 PENNSYLVANIA PLWER & LIGHT CC C A DLENWINE COMPUTER SYSTEMS ANALYST 2 NORTH NINTH STREET ALLENTUWN PA 18101	018101064664 1 99999 PENNSYLVANIA POWER & LIGHT CO R M ROSENDALE MGR TECH RECORDS & SYS 2 NGRIH NINIH STREET ALLENTOWN PA 18101
_	018101079441 1 99999	016103005706 1 99999

ω

.

•	PHILADELPHIA PA 14101	VILEPRESIDE I 2301 MAEKET STREET PHILADELPHIA PA 19101
•	0491J1008341 1 99999 PHILADELPHIA FLECTKIC CO NUCLLAP LOFPATIONS VINCENT BUYEP	019101000660 1 99999 Philadelphia Electric CD Eugene J Bradley Asst Heneral Counsel
•	SÉNIOR VILL PRESIDENT 2301 MARKET ST PHILADFLPHIA PA 49101	PHILADELPHIA PA 19101
•	019101081562 1 99999 Philadelphia Electric Cu Eugene J Rradley 2301 Nakket St Philadelphia Pa 19101	U191010U71J4 1 99999 PHILADELPHIA ELECTRIC CG NUCLEAR GENERATION DIV M J COUNEY SUPERINTENDENT 2301 MARKET ST
•	019101005639 1 99999 PHILADELPHIA ELECTRIC CO ELECTRIC PRODUCTION DEPARTMENT S L DALTROFF VICE PRESIDENT 2301 MARKET ST PHILADELPHIA PA 19101	PHILADELPHIA PA 19101 0191J1J006432 1 99999 PHILADELPHIA ELECTRIC CO ENGIMEERING & RES DEPT John S KEMPER 2301 MARKET STREET PHILADELPHIA PA 19101
•	O17314006576 1 99999 PHILADELPHIA ELECTRIC CU PEACH ROTTOM APS W T ULLRICH PA 17314 DELTA PA 17314	OPHILIO57567 1 99999 PHILIPPINE ATOMIC ENERGY COM ZOILO BARTOLOME CUMNISSIONER Don Mariano Marcos Ave Diliman Quezon City Philippin
•	024019063893 1 99999 PHYSICS ASSOCIATES 5346 PETERS CREEK RD NW RUANOKE VA 24019	010007062815 1 99999 Pinkertuns inc Security Division G J Decard Director Nuclear Energy 100 Church Street New York NY 10007
•	084601053087 1 99999 Pittsburgh des Moines Steel CC E S Hill Welding C Ga Manager 510 East 6th St PD Box 1447 Provo UT 84601	035218053089 1 99999 PITTSBURGH DES MOINES STEEL CO Southern Division J M Jennings Radiation Safety Officer Pu Drawer e Birmingham AL 35218
•	015225053086 1 99999 PITTSBURGH DES MOINES STEEL CO A J MUELLER QA MANAGER NEVILLE ISLAND PITTSBURGH PA 15225	002360005859 1 99999 Plymouth ad of Selectnen David F tarantino Chairman 11 linculn street Plymouth MA 02360
•	002360081563 1 99999 Plymguth Civil Defense Director II Linguln St NA 02360 Plymguth NA 02360	077471005143 1 Cod9 Pollan Nichulsun & Doggett Stephen & Doggett Po Box 592 Rosenberg tx 77471
•	097204005958 1 99999 Purtland General Electric Co James W Durham 121 SW Salmun Street	097204005960 1 99999 Portland General Electfic CC Warren Hastings 121 SB Salmon Street 15 13
٠	PORTLAND DR 97204	PURTLAND DR 97204
•	097204004657 1 99999 Portland General Electric Co Bart D Mithers Vice President Nuclear 121 S W Salmon Street Purtland Gr 97204	097205006347 1 99999 PURTLAND GENERAL ELECTRIC CO HART D WITHERS VICE PRESIDENT-NUCLEAR 121 SW SALMON STREET PURTLAND OR 97205
•	097204081459 1 99999 Portland General Electric Co C P Yungt 121 Sm Salmon St Portland Cr 97204	UD6359026457 1 99999 POSI SEAL INTERNATIONAL INC DUNALD J GUIRK QA MANAGER RUUIES 49 & US95
		N STUNINGTUN CT 06359

•	006001032077 1 99499 Pús corp Corp 71055	061820007030 1 99999 PRAIRIE ALLIANCE P1 B02 2424 STATION A
	FRANK L KELLY 55 BYRON DR	CHAMPAIGN IL 61820
•	AVON LE US	551
•	061820081497 1 99999 PRAIPIE ALLIANCE RANDALL L PLANI	022102064742 I 99999 PRC IMAGE DATA SYSTEMS CO J F FORNEY
	PO BOX 2424 STATION A CHAMPAIGN IL 61	UIVISION MANAGER 820 7600 DLU SPRINGHOUSE ROAD MUIFAN VA 22102
•		
•	063147053069 1 99999 PROGRESSIVE FABRICATORS INC A M HUSS	PROJECT ASSISTANCE CORP C BOLKNER
	QUALITY ASSURANCE MANAGER	UFFICE MANAGER 100 N WINCHESTER BLVD 147 SAN JOSE CA 95128
•		
•	037830064617 1 99999 PROJECT MANAGEMENT CORP R 1. KIMK	PROTO POWER MANAGEMENT CORP P E JALBERT
_	CHIEF QUALITY ENGIN PO BOX U DAY BIOCE TN 37	TREASURER 591 POUUONNOCK ROAD 830 GROTOS CT 06340
•	UAK RIDGE IN ST	
•	070802001283 1 99999 PUBLIC LAW UTILITIES GROUP STEPHEN W IRVING	PUBLIC SERVICE CD NEW MEXICO PLANING AND RESOURCES
-	535 NORTH 6TH ST BATON RUUGE LA 70	ELY YAO 802 Manager Special Projects Alvarado Su
•		ALBUQUERQUE NM 87158
•	080651052349 1 99999 PUBLIC SERVICE CO UF COLORA FORT ST VRAIN NUC GENERATIN	DO PUBLIC SERVICE CO OF COLORADO
	J & GAHM 16805 WELD COUNTRY RD 19 1/ PLATTEVILLE CO 80	SUPERVISUR-NUC ENGRG RECORDS 2 5909 E 38TH AVE 651 DENVER CC 30207
•		000120005753 1 99999
•	PUBLIC SERVICE CO OF COLORA	DO PUBLIC SERVICE CO OF CÓLCHADO ELECTRIC PRODUCTION
•	VICE PRESIDENT PD BCX 840	VICE PRESIDENT 1800 WEST SHERI LANE
•	DENVER CO 80	201 LITTLETON CO 80120
•	PUBLIC SERVICE CU OF COLGRA FORT ST VRAIN NUCLEAR STATE	DO PUBLIC SERVICE CO OF COLORADO
	D WARENBOURG MGR NUC PRODUCTION 16805 MCR 19 1/2	NUCLEAR PRODUCTION MANAGER 16805 WELD COUNTY ROAD 19.1/2
¥	PLATTEVILLE CO RC	651 PLATIEVILLE CO 80651
Ŭ	PUBLIC SERVICE CU OF N INDI	ANA PUBLIC SERVICE CO OF OKLAHOMA
	FIRST VICE PRESIDENT 5265 HOHMAN AVENUE HANNOND IN 46	PO BUX 201 G0-45-17 TULSA OK 74102
U U		- 0.74102007095 1 99999
C	PUBLIC SERVICE CO OF DELAHO BLACK FUX STA NUCLEAR PROJE	MA PUBLIC SERVICE CO UF OKLAHUMA
	G W MUENCH Manager PD Bux 201	PD BOX 201 OK 74102
-	TULSA DK 74	
	PUBLIC SERVICE ÉLEC & GAS (C/D AECHTEL PUWER CURP	D PUBLIC SERVICE ÉLEC L GÁS CO
	E F DEVOY PRINCIPAL ENGR-HOPE CREEK 50 BEALF STREFT - PO BOX 30	VANAGER-LICENSING & ANALTSIS 30 Park Plaza T1 60 965 Newark NJ 07101
	SAN FRANCISCO CA 90 007101000245 1 9000	6119 9 007101063654 1 99999
-		

• • •

•

 $\widehat{}$

-

• |

J

•	NICHARD ASTLING ASST GENERAL SULICITUR BO PAPR PLAZA TSE NEWARK NJ 07101	LIUNARY FLURINE CURPORA 30 PARK NEWARK
•	OU7101006342 1 99999 PURLIC SERVICE ELEC & GAS CO PIERRE & LANDRIEU PRUJELT MATAGOR BU PARK PLAZA T 17 A NE-ARK NJ 07101	0071010 PUBLIC ECHIN A Mahager PU BUX TIGD NF BARK
•	007101079360 1 99993 PUBLIC SERVICE ELEC & GAS CO NUCLEAR LICENSING & REGULATION EDWIN A LIJEN PC BCX 570 TI60 NEWARK NJ 07101	0071010 PUBLIC T J MAR 80 PAKK ROOM T1 NEWARK
•	DOB03B079361 1 99999 PURLIC SERVICE ELEC 6 GAS CO Salem Openations H J Miduna General Manager PO HOX E Manlolks Bridge NJ 08038	0080380 PUBLIC SALEM D HENRY J GENERAL PU BOX HANCOCK
•	007101004741 1 99999 PUBLIC SERVICE ELEC & GAS CO COMPORATE QUALITY ASSURANCE ROBERT L MITTL PO BOX 570 T160 Newark NJ 07101	0071010 PUBLIC CORPORA ROBERTI PU BOX T160 Newark
•	UJB038006458 1 99999 PUBLIC SERVILE ELEC & GAS CO MOPE CREEK PRODUCTION R S SALVESEN GENERAL MNGR-HOPE CREEK OPER PO BOX A HANLOCKS BRIDGE NJ 08038	0080380 PUBLIC R A UDEI VICE PRI PU BOX TISA HANCUCK
•	008038028096 1 99999 PUBLIC SERVICE ELEC G GAS CO NUCLEAR DEPARTMENT RICHARD A UDERITZ VICE PRESIDENT PO BOX 236 HANCUCKS BRIDGE NJ ORO38	04716200 PUBLIC S J BREI PO BOX NEW WASI
•	046168001458 I 99999 PUBLIC SERVICE INDIANA CHARLES & CAMPRELL VICE PRESIDENT & GEN COUNSEL 1000 EAST MAIN STREET PLAINFIELD IN 46168	04616800 PUBLIC R L HAGO Systems 1000 E P PLAINFII
•	047162065984 1 99999 Public Service Indiana B Hensley Executive Secretary PD BDX 190 New Washington IN 47162	0471620(PUBLIC NUCLEAR S W SHII SENIDR V PO ADX I WASHING
•	047162007399 1 99999 PUBLIC SERVICE INDIANA NUCLEAR DIVISION S N SHIELDS SR V P P O BOX 190 NEW MASHINUTON IN 47162	UUO94JO PUERTO EDUARDO DIRECTOP MINILLAS SANTURCE
D D	OOU936008697 1 99999 PUERIU RICU WATER RES AUTH OKLANDU ANGLEPC DIVISION HEAD ENV PROTECTION PO BOX 4267 SAN JUAN PR 00936	00093608 PUERTU H ELECTRIC J A BUNN ASST EXE GPU BUX SAN LUAN
•	098009007036 1 99999 Puget Sjund Puwer & Light Cu Rubert V Myers Vice Pres-General Resources Puget Pumet Meug	0177010 PULLMAN ALLAN E U A MANA PO MOX

LI NAPY PSC FLUMINE HUNT CUPPORATE LIBRARIAN 30 PARK PLAZA NJ 07151
007101005637 1 99999
PUBLIC SERVICE ELEC & GAS CG
MAHAGER-NUC LICENSING & REGULA PU BUX 570
NEWARK NJ 07101
007101003281 1 99999 Public Service Elec E GAS CO
T J MARTIN 80 Park Plaza
NEWARK NJ 07131
008038005630 1 99909
PUBLIC SERVICE ÉLEC & GAS CO SALEM OPERATIONS
HENRY J HIDURA General Manager
PU BOX 168 HANCOCKS BRIDGE NJ 08038
007101006343 1 99999 PUBLIC SERVICE FLEC 5 645 CO
CORPORATE QUALITY ASSURANCE
PO BOX 570 1160
NEWARK NJ 07101
PUBLIC SERVICE ÉLEC & GAS CO
VICE PRESIDENT-NUCLEAR PU BOX 236
TISA MANCUCKS BRIDGE NJ 08038
047162062895 1 99999 PURI LC SERVICE INDIANA
S J BREWER PO BOX 190
NEW WASHINGTON IN 47162
046168064713 0 90999
PUBLIC SERVICE INDIANA R L HAGGARD
SYSTEMS PROJECT SUPYR 1000 E MAIN STREET
PLAINFIELD IN 46103
047162006238 1 99999
NUCLEAR DIVISIUN S W SHIELDS
SENIOR VICE PRESIDENT
WASHINGTUN IN 47162
PUERTO RICO ENERGY OFFICE
VINELTOR VINILLAS STATION BOX 41089
SANTURCE PR 00940
000936080958 1 99999
ELECTRICAL PLAN RES & CONSTRUC
ASST EXECUTIVE DIRECTOR GPU BOX 4267
SAN JŪAN PR 00936
01//01053091 1 99999 PULLMAN PUMER PRODUCTS
PO HOX 3308 REACH ROAD

•

.

•

•

•

•

^	tan kan ang kan tang kan tang Tang kan tang	AN UAN	• • • • • • • • • • • • • • • • • • •
•	098009007096 1 99999 PUGLT SJUND POWER & LIGHT CU ROBERT V MYEPS	017701053091 1 Pullman Puwer Produc Slian Bair	99999 15
•	VILE PRES-GENERAL RESUDREES PUGET PUMER delle Bellevue va 90009	PU BUX 3338 REACH RO WILLIAMSPURT	AU PA 17701
•	002210053092 L 94999 PX ENGINEERING CO INC GEORGE S SURVICH	061242005847 1 CUAD CITIES NUC POWE NICK KALIVIANAKAS	99999 R STATION
-	VICE PRESIDENT MARKETING 660 S-IMMER STREET MA 02210 BUSTON MA 02210	22710 206TH AVENUE-N CURDIVA	ORTH 11 61242
•	015108079693 1 99999 WADREX COMP	015108079694 1 NUADREX CORP	99999
-	LIBRAGIAN 1995 DEAVER GRADE ROAD Coraupulis pa 15108	HENRY D DARBEE Manager of Operation 1005 Beaver grade RC Suite 201 Copaugui 15	IS IAD PA 15108
-	005008038038 1 99999	095008020505 1	99999
•	QUADREX CORP ENGINEERING A F KITZ VP ENLINEERING	GUAUREX CORP NUCLEAR SERVICES COP Margaret Ma Librarian	(P
•	I TOO DELL AVENUE CAMPRELL CA 95008	ÎÎO DELL AVENUE Campbell	CA 95008
•	048124077486 1 99999 Quantum Enterprises Bruce D Cargi President	095030067156 1 R F REEDY INC RICK SWAYNE 236 N SANTA CRUZ AVI	99999 E
٠	24925 RDSS DEARBORN MI 48124	SUITE 247 Lus gatos	CA 95030
•	015077081392 1 99999 Rad Services Inc James & Orr	094025026771 1 Raychem Corp F e Lafetra	99999
	ASST PROJECT SUPERVISOR PU UUX 85 Shippingport pa 15077	300 CUNSTITUTIUN DR. MENLG PARK	CA 94025
	001803064740 1 99999 RayTheon CO Technical Info Operation	001803064598 l Rayimeon Co P L Edsall	99999
•	A M DELLO RUSSO MANAGER	TECHNICAL INFO SPEC	IALIST
۲	Z WAYSIDE ROAD Buklington Ma U1803	BURLINGTON	MA 01803
•	001803064746 1 99999 Raytheon Co R_MULLHOLAND	095128057907 1 REACTOR CONTROLS IN JACK C MILLETT	99999 C
	RECORDS MGMT CONSULTANT 2 WAYSIDE KOAD JURLINGTUN MA 01803	SUITE 222 SAN JOSE	CA 95128
•	013057065240 1 99999 Reactor controls inc	012181003805 3 Rensselaer Polytech	99999 NIC INST
	5854 BUTTERNUT DR EAST SYKACUSE NY 13057	FULSON LIBRARY	NY 12181
	ODENMAD11715 1 99999	044128012807 1	99999
	F LIST ADMINISTRATOP	MANAGER QUALITY ASS RUBERT L SIMON	URANCE
. 🌒	DK-4000 RUSKILDE DENMARK	18530 SOUTH MILES P WARRENSVILLE HEIGHT	ARKWAY S UH 44128
•	002916079409 L 99999 RMGDE ISLAND ATONIC ENERGY COM JOHN S PASCALICES	002908023165 1 Rhude Island Dept D Div Occupatni Hith	99999 F HEALTH & RAD CNTL
	COMMISSIONER 18 CHAUNCEY STREET RUNFORD RI 02916	JAMES E HICKEY CHIEF DAVIS STREET-CANNUN PROVIDENCE	I BLDG BL 02908
		DERANCO53096 1	91979

-	·		· · · · · ·	
•	HALLINGFORD CI	1 36492		
•	091304077487 L RUCKWELL INTERNATIONAL ENERGY SYSTEMS GHOUP A CARNAZZULA	99999	027602053099 1 RUCKWELL INTERNATIONA FLUW CONTRUL DIV J V GRASSO BLANT MANAGER	99979 L
•	CODES L STANDARUS 8900 DE SUTG AVENUE CANUGA PARK CA	A 91304	P U BOX 1961 KALEIGH	NC 27602
•	035801081550 1 RUCKWELL INTERNATIONAL NEILL LANGFURD 3322 S MEMURIAL PARKWAY	59999 Y	002110067268 1 RUPES 6 GRAY THOMAS DIGNAN ESQUIRE	99999
•	SUITE 39 HUNTSVILLE AL	L 35801	BUSTON	MA 02110
•	002110000449 L NOPES & GRAY JUHN A RITSMER ESQUIPE	99999	055344053727 1 RDSEMOUNT INC NUCLEAR QUALITY ASSUR GERALD D ANDERSON	99999 ANCE
•	BOSTUN MA	02110	AI EDEN PRAIRIE	1 MN 55344
•	030339064286 1 S RCSEMOUNT INC RANDY HALEY 290 INTERSTATE NORTH SUITE_150	99999	055435065512 1 Rusemount Inc Lyle Lofgren Senior Engr Pû Box 35129	99993
•	ATLANTA GI	A 30339	MINNEAPOLIS	MN 55435
•	092701000876 1 RUURKE L HUDURUFF ALAN R MATTS ESQUIRE	99999	095201055291 1 RTE DELTA CORP 705 NORTH CARLTON AVE STOCKTON	99999 NUE CA 95201
•	1055 NORTH MAIN STREET SUITE 1020 SANTA ANA CI	A 92701		
•	064012064461 1 Ruskin Manufacturing Ci Quality Assurance Dept QA Manager	99999 D	002181056882 1 RV & ASSOCIATES AVRUM SILVERMAN 69 EDGEMOOR AVENUE	99999 MA 02181
0	GRANDVIEW M	0 64012	WELLESLET	NA VZICI
۲. ۲	029201009340 1 S CARGLINA DEPT HLTH E DIVISION RADIOLOGICAL HEYWARD SHEALY	99999 NVIR CTL HEALTH	029201001392 1 S CARULINA DEPT HLTH RICHARD P WILSON ASSISTANT ATTORNEY GE PD H0X 11549	99999 Envir Ctl Neral
(260) BULL ST-J MARION COLUMBIA	SIMS BLD C 29201	COLUMBIA	SC 29201
	029065002890 1 S CAROLINA ELEC & GAS Summer 1 NDS O S Bradham Station Manager Publy Bur Bur F	99999 CU	029218006033 I S CAROLINA ELEC & GAS U M DIXON VICE PRESIDENT-NUCLEA PO BOX 764 MAIL CODE COLUMBIA	99999 CU R_OPER F=04 SC 29218
	JENKINSVILLE S 029218062856 1 S CAROLINA ELEL & GAS SUMMER I NPS J P HARKISON	94949 CU	029065080954 1 S CARULINA ELEC & GAS Summer 1 NPS U A LAVIGNE	99999 5 CD
U I	DIRECTOR OF SECURITY PD BUX 764 CULUMBIA S	C 29218	ŠITE QĀ ČUORDINATUR PU BOX BB JENKINSVILLE	SC 29065
U	029218004542 1 S CARDLINA ELEC & GAS NUCLEAR UPERATIONS T C NICHOLS PO BUK 764	99999 CD	029218080953 1 S CAROLINA ELEC & GAS SUMMER 1 NPS C A PRICE NUCLEAR ENGING MANAGEF	99999 i CO
V	CULUMBIA S	C 29218	CULUMBIA	SC 29218
C	029065081353 1 S CAROLINA ELEC & GAS A A SMITH Site ya Courdinator	60 60	029218006034 1 S CAROLINA ELEC E GAS MARK WHITAKER POST DEFICE BOX 764	99999 5 CU
-	PU BOX 9 JENKINSVILLE S	iC 29065	COLUMBIA	SC 29218
	many constraint	00030	057501064189 1	09939

ø J ---
PU BUR 100 NILL SAU LAN AND MARKED	
CULUABIA CO 20210 PU BUX 764	
029055041353 1 9999 2 029219006035 1 S CARULINA FLEF 5 GAS CO	99949 SC 29218
A A SHITH CLEAR AND CONSTRUCTION CARULINA ELEC E GAS SITE JA COURDINATOR MARK WHITAKER DO ADA COURDINATOR PUST OFFICE BOX 764	Cu
SUBARINSVILLE SC 29065	SC 29218
029461U06234 1 99999 057501064189 1 S CARULINA PUB SERV COM S DAROTA STATE PLANNIN HENRY CYRUS SENION VICE PRESIDENT EXEC POLICY ANALYST 233 NORTH LIVE DAN DRIVE CAPITOL BUILDING MONCKS	99999 Ng bureau
095008079603 3 D0000 000000000	50 57501
S LEVY INC S M STCLLER CORP LINDA SMITH LIBRARIAN SUITE 500	99999
SUITE 725 BOULDER C CAMPBELL CA 95008	0 80302
080302052913 1 99999 OBELGIO81072 1 S M STOLLER CORP SA INTERCOM NY ERIC ULSON PLACE DU IRONE 1 BI SUITE 500 COLORADO 8LOG BRUSSELS IOUO B BOULDER CO 80302	99999 0607 Elgium
O95813064615 1 99999 O95813006095 1 SACRAMENTO MUNICIPAL UTIL DIST SACRAMENTO MUNICIPAL U J JEWETT SENIOH ENGINEER SECRETARY & GENERAL CO 6201 S STREET 6201 S ST SACRAMENTO CA 95813 B0X 15830 SACRAMENTO CA 95813 B0X 15830	99999 TIL DIST UNSEL
O95813006531 1 99999 O95813007423 1 J J MATTINUE ASST GENERAL MGR & CHIEF ENGR 6201 S STREET BDX 15830 SACRAMENTO CA 9581300723 1 J J MATTINOE SACRAMENTO SACRAMENTO CA 9581300723 1 J J MATTINOE SACRAMENTO SACRAMENTO CA 9581300723 1 SACRAMENTO CA 9581300723 1 SACRAMENTO CI	99999 TIL DIST NGR 4 95813
O95813064384 1 9999 O95813079430 1 SACRAMENTO MUNICIPAL UTIL DIST RANCHU SECO NUCLEAR PWR PLANT L G SCHWIEGER MANAGER 6201 S STREET BOX 15830 SACRAMENTO CA	99999 Fil dist 95813
O49508063180 1 99999 053226063184 1 SALEM ENGINEERING CORP SALEM ENGINEERING CORP 2186 44TH STREET SE 2360 N 124TH STREET B GRAND RAPIDS NI 49508 WAUWATDSA WI)9999 53226
O77092063179 1 99999 O01803063181 1 9 SALEM ENGINEERING CORP SALEM ENGINEERING CORP SALEM ENGINEERING CORP SALEM ENGINEERING CORP SALEM ENGINEERING CORP SALEM TECHNICAL SERVICE UNPORATE PLACE 1 HOUSTON TX 77092 99 S REOFORD ST SUITE 2 BURLINGTON MA	99999 S 07 01803
O29360006278 1 99999 SALUDA HIVER ELEC LG-UP INL 207 SHERWUUD DH LAURENS SC 29360 J R POSTON ASSISTANT GENERAL MANAG PU BOX 1771 SAN ANTORIO TY	9999 ICt BD ER 78294
092112001537 1 99999 092112008694 1 SAN DIEGD GAS & ELECTRIC CU SAN DIEGO GAS & ELECTRIC SAN DIEGO GAS & ELECTRIC PUWER SUPPLY SAN DIEGO GAS & ELECTRIC U U CUTUN PUWER PLANT ENGR & CONS L BERNAIH PU BOX 1831 PD BUX 1831 SAN UIEGO CA	9999 C CU TR 92112

ند. •

-

1

-

-

۲

•

0

•

۲

•

•

Ü

W

4540 m sele stert HOUSTON ●	TX 77092	U RELATE PLACE 1 99 S BEDFORD ST SUIT BURLINGTON	E 207 MA 01803
029363036278 1 SALUDA KIVER ELEC LU 207 SHERWUGD DK LAURENS	49949 -UP INC SC 24360	078295001482 L San Antuniu Public S Uperatiuns J P Pustun Assistant General Mai Pu Bux 1771 San Antuniu	99999 ERVICE 8D NAGER TX 78296
092112001537 1 SAN UIEGD GAS & ELEC POWER SUPPLY G U CUT!∪ ² L BERNAIH PD 50X 1831 SAN DIEGU	59999 TRICCU CA 92112	U92112008694 1 San Dieud Gas & Elec Pumer Plant Engr & C Gary D Cotton Pu Box 1831 San Diego	99999 TRIC CO ENSTR CA 92112
092112006330 1 SAN DIEGO GAS 6 ELEC HENRY PETERS PO BUX 1831 SAN DIEGO	99999 TRIC CO CA 92112	092112005375 1 San Diego Gas & Elec Harry B Stoerr Po Box 1831 San Diego	99999 TRIC CG CA 92112
092112005374 1 SAN DIEGO GAS & ELEC JALK E THOMAS PU BOX 1831 SAN DIEGU	99999 Tric CC CA 92112	087185010382 1 SANDIA LABORATORIES DIVISION 4446 L L BONZON ALBUQUERQUE	94999 NM 87185
060603006509 1 SARGENT & LUMDY NUCLEAR SFGDS & LICE E DWARG R CRASS S5 EAST MONKOE STREE CHICAGO	99999 NSING DIV T IL 60603	060603064604 1 Sargent & Lundy P J Frdemike Dgcumentation Sectio 55 East Monroe Stree Chicago	99999 N LDR T 11 60603
060091064675 1 SARGENT & LUNDY J & GILBERT CORPORTE RECORDS MAN 1225 DARTMUUTH WILMETTE	99999 AGER EL 60091	060603007032 1 SARGENT & LUNDY R C HEIDER PRUJECT MANAGER 55 EAST MONROE ST CHICAGO	99999 11 60603
O60603066681 1 SARGENT & LUNDY E KABUT SENIOR VA CONTROLLER 55 EAST MONRUE STREE CHICAGO	99999 T IL 60603	U60603030583 1 SARGENT & LUNDY DIV-NUCLEAR SAFEGUAR DAN KANE ASST HEAD 55 EAST MONRUE ST CHICAGO	99999 DS & LIC 1L 60603
064659081094 1 SARGENT & LUNDY ROGELIO & PENASALES 6322 N AVERS AVE CHICAGO	99999 IL 60659	060603064686 1 SARGENT & LUNDY G PRAPUOLENIS DOCUMENTATION SUPERY 55 EAST MONROE STREE CHICAGO	99999 150R 1L 60603
OGOGO3OG4662 L SARGENT & LUNDY M RLSA JR DOCUMENTATION SUPERV 55 EAST MUNRUE STREE CHILAGU	79999 1508 11 60603	060603065213 1 SARGENT & LUNDY MARTIN E SCHUSTER HFAD-QUALITY CONTROL 55 E MUNROE ST FLUOR 18 CHICAGO	99999 DIVISION IL 60603
060603053105 1 SARGENT & LUNDY H S TAYLOR HEAD-QUALITY ASSURAN 55 EASI MUNRUE ST CHICAGU	99999 ICE DIV IL 60603	U60603064671 1 SARGENT & LUNDY G J VANCINA U0CUMENTATION SUPER 55 E MONROE STREET CHICAGO	99999 11508 11 60603
060603006506 1 SARGENT & LUNDY P L WATTELET 35 EAST MURRE STREE	999999 ET	U45241065233 1 Savage Malker Schul 10880 INDECD UR CINCINNATI	99949 FASSOC INC DH 45241

•	093105000947 1 94999 Scenic Shumelive Presenv Cump Frederica fisier Santa Barbara Drive Ca 93105	01522007155 1 99999 Schneider Cunsulting Engineers Ray Kathiresan Pou Pean Ave Pittsburgh	OJAPANO57564 I 99999 SCIENCE & TECHNOLIGY AGENCY ATOMIC ENERGY BUREAU AUNINSTRATOR BUREAU CHIYUDA-KU TOKYO 100 JAPAN OFRANCOBUG9 I 99999 SESIN-MINISTRY OF INUUSTRY ANDRE MESSIAH ANDRE MESSIAH ANDRE MESSIAH ANDRE MESSIAH ANDRE MESSIAH ANDRE MESSIAH ANDRE MESSIAH ANDRE MESSIAH	053U66045537 1 09999 SENTRY EJUIPMENT COMP UPERATIONS DEPT MANAGER UPERATIONS DE PARIMENT MANAGER UPERATIONS DE PARIMENT CUNOMOMOC MUR ROAU MI 53066 094108005987 1 99999 SETTOUM MILLIS 099999 SUITE 2100 SUITE 2100 SUITE 2100 SUITE 2100 SUITE 2100	02 0036000650 0 1 99999 EMAM PITTMAN POTTS TROMURIDGE ENDER ELLARE 1800 m Streft NM DC 20036 MASHINGTON DC 20036 220036010518 0 1 5 TROMURIDGE GERALD CHARNOFF ESU 020036010518 0 1 5 TROMURIDGE GERALD CHARNOFF ESU MASHINGTON DC 20036	020036000038 1 99999 Sham Pittman Poits Irowaridge Jay E Silbeng 1300 m Street NM DC 20036 Mashingidn DC 20036	011752006326 1 99999 Shikemam nucleak pomer statium Jeff Smith Pis Bux 618 My 11792 Mading Kiver Ny 11792	020007051946 1 99999 517 3421 R 51 NH UC 20007 MaShingTun UC 20007
Coletan IL alous	020036356038 1 99999 Scandinavian decument center F-12779-6 11201974 street am DC 20336 Washingtun	060605007033 1 99999 Schiff Harjin & Maitl Sheluon Zaull Maitl 7200 Sears Tumer 233 Aacker Dr Chilagu	015222038031 LTING ENLINEFKS SCHWEIDER CLASULTING ENLINEFKS GORP US MANGEK FORP US MANGEK T25 SEVENTH STPEFT PA 15222 037892023285 THE PA 15222 03789023285 THE PA 15222 SCHENCE APPLICATIONS INC TECHNICAL INFURMATION LIANY RIDGE TMPK SUITE 1000 DAX RIDGE TMPK SUITE 1000	OSPAINUJ34J9 I 99999 SENEK INCENIERIA V SISTEMAS DOCUMENTION SERVICE ALBISU APARTADU 8 LAS ARENAS(VIZCAVA) SPAIN ODI867000863 I 99999 SERVIAR CUV EUNARU M KELLV MA OI867 REAUINU	020036005762 1 99999 5HAW PITTMAN PUTTS THOMBRIDGE 1000 M S BAXTER 1000 M S BAXTER 1000 M S BAXTER MASHINGTON 0200360U7093 1 99999 5HAM PITTMAN PUTTS THOMBRIDGE 1900 M STHLET NM DC 20036 100 M STHLET NM DC 20036	020036005863 1 99999 Smam Pittman Poits Trownkijge Brug E Ceukenill 1800 m Statei nm DC 20036 Mashingtun	020036005190 1 99949 Sham Pittman Potts trombmidge Geurge F trombmidge Esu 1800 m Street NM DC 20036 Mashington	094115005653 1 99999 51644 CLUM 16641 UETENSE FUND INC MICHALL & SMEMMOND 2044 FLLAUME STAEL 2044 FLLAUME STAEL 54N FRANCISCU

مروري وروي المراجع . به معهد بر دمر استبد بد 1.2.

a good a second

) (

11

ຸດ

1 1

(

	É a d	
•	1600 M STREET NW WASHINGTUN DC 20036	WADING RIVER NY 11792
•	094115005653 1 99999 STERRA CLUM	020007051946 1 99999 STI
•	MICHAEL R SHERWOHD 2044 FILLMUHE STREET SAN FRANCISCU CA 94115	3421 R ST NW WASHINGTUN DC 20007
-	085073000937 1 99999	020850001347
	SNELL & WILMEH ARTHUR C GEHR ESQUIRE	SNUPPS NICHULAS A PETRICK EXECUTIVE DIRECTUR
•	PHOENIX AZ 85073	S CHUKE CHERRY ROAD RUCKVILLE MD 20850
•	020850021731 1 99999 SNUPPS NIEHGLAS A PETRICK Exel utive director	093420005458 1 99999 Suuth Cuunty Publishing Cu Richard E Blankenburg
9	S CHILLE CHERRY KD KOCKVILLE MD 20850	ARREYO GRANDE CA 93420
•	091770000761 1 99999 Suuthern Califunnia Edison Co Lan Dept	091770058231 I 99999 Suuthern California Edisun Co Nuclear Engeneration Co
•	JANES A BEOLETTO Po Bix Buo Rusemead ca 91770	R DIETCH VICE PRESIDENT 2244 WALNUT GRUVE AVE BOX 800
•	091770001533 1 99999 Southern California Edisun Co Rohert Difte	CA 91770 091770005372 1 99999 SOUTHERN_CALIFORNIA EDISON CU
•	VIČE PREŠIDENT 2246 val nut grove avenue	ASSISTANT GENERAL COUNSEL
	ROSEMEAD CA 91770	ROSEMEAD CA 91770
•	U91770006332 1 99999 Suuthern California Edison Cu Mark Medfurd 2244 Walnut Grove Ave	091770079429 1 99999 Suuthern California Edison Co L T Pápay Vice President
•	RUSEMEAU CA 91770	PU BOX BOO RUSEMEAD CA 91770
•	092672064663 2 99999 Suutmern California Edison Cu Corporate Documentation MGMT R.L.preston	092672081541 I 99999 Suuthern California Edisun Co San Onofre NPS Stephen I Honow
1	SUPERVISUR COM PO BOX 700 SAN CLEMENTE CA 92672	PU BUX 700 UNITS 2/3 OB-2 SAN CLEMENTE CA 92672
0	U35202001433 1 99999 SOUTHERN CI SERVILES INC J A BAILY PRUMERT I CENTING MANAGER	U35202005789 1 99999 Southern Co Services inc Uzen Batum
0	PO BUX 2625 BIRMINGHAM AL 35202	PU BOX 2625 BIRMINGHAM AL 35202
J	030346064769 L 99999 Southern Cu Services Inc R L Kime	035202001236 1 99999 Southern Co Services inc Kuble A Thunas
1	SUPVR ENG DATA PROCESSING PG BCX 720071 ATLANTA GA 30346	VICE PRESIDENT PU BOX 2625 BIRMINGHAN AL 35202
-	035202008695 E 99999 Southern Cu Services Inc Ruble A Thumas	030367003262 1 99999 Southern Engineering CD
.	VICE PRESIDENT PU BUX 2625 BIRMINGHAM AL 35202	NGR GENERATION SERVICES 1400 PEACHTREE ST NW ATLANTA GA 30367
-	033528039976 1 99999 Southern Science Applications Office of Olack & Veatch E J Cheling Linkanian	013528004318 1 99999 SUUTHERN SCIENCE APPLICATIONS WALTER MITCHELL III EXECUTIVE VICE PRESIDENT
	₩Ej H1 # \$2\$	

***	A Constant of the Constant	
	078244064614 1 99999 SOUTHWLST FLSEARCH INSTITUTF G F HUMPHHLSS SENTUM KESEARCH ANALYST PO BUX 24510 SAN ANTUNIU TA 78284	078284023410 1 99999 SOUTHMEST RESEARCH INSTITUTE DIVISION 17 LIBRARY C.A. KOYAL LIMPAKJAN 0220 CULEBRA RD PO DRAMER28510 SAN ANTONIO TT 18244
P P	046222053726 I 99999 Speeday Machine & Tigleg Inl Wanada 1802 North Luftt St Indianapolis In 40222	U27611363655 1 99999 SJUARE D.CD I.C.DIVISION HILLIAN J.FIGHTMASTER JUALITY ASSURANCE MANAGER PU JUA 27466
Þ	046970053110 1 99999 Sudake D Lü Pomer Equip Div Lapry D mest Wa Mur	NALEIGH NC 27611 060120081903 1 99999 51 JUSEPHS HOSPITAL APPARAU DEVATA 77 N AIRLITE ST
•	252 N TIPPECANCE ST PERU IN 46970	ELGIN IL 60120
7	080217053111 1 99999 Stearns Ruger Inc Roy Davis Corpurate Da Manager	033131006038 1 99999 Steel Hector & Davis Nurman & Cull 1400 Se First NATL Bank Bidg
•	DENVER CO A0217	MIAMI FL 33131
Ϊ k	077001J66465 1 99999 STEWART & STEVENSUN SERVICES NUCLEAR CONTRACTS M M ARBUCKLE PU BCX 1637	020006001101 1 99999 Stune & Mebster Engr Corp Washington Operations 1475 Eve Street NW
	HOUSTUN TX 77001	WASHINGTON DC 20006
	OOBD34064668 I 99999 STONE & WEDSTER ENGR CORP L J ALDRIDGE ADMINISTRATIVE ASSISTANT 3 ERELUTIVE CAMPUS BOX 5200	008034064592 1 99999 SIDNE & WEBSTER ENGR CCRP J J CARPINO SUPVR RECORDS MANAGEMENT PU BOX SZOU
	CHERRY HILL NJ 08034 002107064708 1 99999 STONE & WEBSTER ENGR CURP S & COMEFORD PROJECT RECURDS ADMIN 245 SUMMER ST BOSTON NA 02107	O20006063157 I 99999 STDNE 6 WEBSTER ENGR CURP MILARY COUSTER FNGR CURP 1875 EYE STREET NW SUITE 550 WASHINGTUN DC 20006
	070775064727 1 99999 Stune & Webster Engr Corp Dunald P Hopkins Record Aumin Po Bux 217 St Francisville IA 70775	CU2107065623 1 99999 STONE & WEBSTER ENGR CORP STEPHEN HUNT 245 SUMMER ST BOSTON MA 02107
1	D02107038032 1 99999 STUNE G WERSTER ENGH CURP R B KELLY VICE PRESIDENT G QA MGR PU HIX 2325 BUSTUN NA 02107	OU8034064667 1 99999 STONE & WEBSTER ENGR CORP U D MARTIN PRUJECT RECORDS ADMINISTRATOR PU 80X 5200 CHERRY MILL MAL DAGG
	002107064726 1 99999 STGNE & WEBSTER ENGR LORP M A MURJAN JX PRUJELT MANAGER 245 SUMMEP STREET BUSTON MA 02107	OU3C45064680 1 99999 STUNE 6 WEBSTER ENGR CURP ADMINISTRATIVE SERVICES CARUL A NORCHUSS ASSISTANT SUPERVISOR RED BI PATTER HILL RD CUEESTONN
¥	002107064630 1 99999 STUNE & KENSTER ENGK LUNP D M CMAN ASST MGH KEC MGMT GIV 245 SUMMER SIREET BOSTON NA 02107	002107001069 1 99999 STUNE & MEBSTER ENGR CCKP TECHNICAL INFURMATION CENTER NANCY PELLINI P U BOX 7325 BUSTUN MA 02107

nutte thasana

 \mathcal{O}

• rine e

. .

JQ2107304720 1 STONE 6 DEUSTER ENGR M.A.MORUAN JR PRUJET MANAGEN 245 SUMMEP STREET BOSTON	59999 LLHP Ma 32107	UUJUAJAADAADHU I STUNE L WEBSTER ENGR ADMINISTRATIVE SERVI CARLL A NURCHUSS ASSISTANT SUPERVISUP RFU WI PATTER HILL R GUFFSIOWN	990999 CCCRP ICES ID NH 113065
002107064-30 1 STUNE E VENSTEN ENGR D M LMAN ASST MGR REC MGMT D1: 245 SUMMER STREET BESTEN	99999 Cump V MA 02107	002107001069 1 STONE & HEBSTER ENGH TECHNICAL INFURMATIC NANCY PELLINI P U HOX 2325 BUSTUN	99999 CLAP IN CENTER NA 02107
OUZIG7035292 1 STENE & NEMSTER ENGR LICENSING INFORMATIU ARLENE & PORT SUPERVISUR PC. BUX 2325	99999 Curp N	002107006497 1 Stune & Wenster Engr C & Richardson Po Bux 2325 Buston	98999 CGRP MA 02107
BUSTEN UQ2107064642 1 STUNE & MEDSTER ENGR C F SCHILT JR SUPERVISUR-NECURDS MO 245 SUMMER STREET BOSTEN	MA 02107 99999 Curp GNT MA 02107	002107064733 1 STUNE & WEBSTER ENGR 4 N VECCHLARELLO SR MGMT SYSTEMS ENGR PO COX 2325 BOSTON	99999 CURP
OO2107005849 1 STUNE & MEBSTER FNGR NUCLEAK ENGT DEVELO PIRECTOR OF ENGINEERI PUD BDX 2325	99999 CUHP DPMENT ING	020906055441 1 STRIDE INC GAJANAN M SABNIS 13721 TOWN LINE ROAD SILVER SPRING	99999 MD 20906
BOSTON 094583066632 1 STRUCTUR AL MECHANICS JAMES J JOHNSON VICE PRESIDENT 2400 DLD CROW CANYON SAN RAMON	MA 02107 99999 ASSOC INC RD CA 94583	008071053724 1 Struthers Dunn Inc F P Merkel Quality Control Engi Lambs Road Pitman	99999 NEER NJ 08071
016365053112 I STRUTHERS WELLS CORP J P FANARTIS Exec vice pres PO BOX 8 WARKEN	99999 PA 16365	OSWEDEUO3323 1 STUDSVIK ENERGITEKNI STUDSVIK LIBRARY INGHIC PETTERSSON LIBRARIAN 611 82 NYKOPING	999999 K AB
019107006526 1 Sugarman & Denburth North Anerican Buildi 121 South Broad Stree Suite 510 Philauelphia	99999 Ing T PA 19107	OSHITZOO3423 1 Sulzer Brothers Inc Rubert Sagesser Va Manager Dozo CH-8401 JINTERTHUR	99999 Switzerla
OLB 703001492 1 SUSUUEMANNA ENVIRON A GERALD R SCHULTZ PU BGX 1560 WILRES-BARRE	99999 DVUCATES PA 18703	063101053115 L Sverdrup Parcel & As: M E Blumer President Bol North 11th Blyd St Louis	99999 50C inc M0 63101
007013053116 1 SWEPCO TUBE CORP UNL LLIFTON BLVD CLIFTUN	99999 NJ 07013	OSWITZO66988 I SWISS FEDERAL DFFICE International Energy Llaude Zangger Administrator 3003 Bern	99999 OF ENERGY AFFAIRS SWITZER
049801064460 1 Systems Control Inc PU BOX 788 IRON MOUNTAIN	99999 MI 49801	037630065216 1 Systems development C N 5 Markowicz 903 Oak Ridge Turnpik UAK Ridge	99999 Gurp (<u>e</u> TN 37830
• TATWANG 3568 1 TATWAN PUWER CC	99339 · •	U11735053118 1 TARGET_ROCK_CDRP	99999

0

12

•

.

•

a

6

	OTATHAUO3568 1 99999 TATHAN PUNCK CO	
	ATUMIC POHEN DEPARTMENT P C LIU No G LIU	
٠	SEEVIUN S TAIMAN	
	019355057363 1 99999	,
•	TARGET TECHNOLOGY LTD GREAT VALLEY LERPERATE CENTER	
	DAN WALSH 844 UKFAT VALLEY PARKWAY	
۵	MALVERN PA 19355	1
	QSPA10009874 1 99999	
•	LECNATUM S A	
-	RUSE H HELFNUEL LIBRANIAN Can fluartian de fris beves	1
•	KM 19 CNI MAURIO-IRU SPAIN	
•	093406005461 1 99999 Telechan tribune	
-	SAN LUIS UBISPE COUNTY	ļ
	1321 JOHNSUN AVE Bux 112	
-	ŠĀN LŪĪS OBISPO CA 93406	
	037203031321 1 99999 TENNESSEE DEPT OF PUBLIC HLTH	
-	BUPEAU OF ENVIRON AFFAIRS R. NULLE	
	150 9TH AVE NORTH 1ST FL TERRA	,
-	RASHVILLE IN 37203	
•	TENNESSEE VALLEY AUTHORITY	
-	J W COFIELD	
٠	PO BCX 2000 SURGOINSVILLE TN 37873	1
	038852004633 1 99999	1
•	TENNESSEE VALLEY AUTHORITY Vellow creek nuclear plant	
	L S COX PRCLECT_MANAGER	1
۲	PO 90X 2000 10ka - NS 38852	į
-	037902001295 1 99999	,
•	249-A HAMILTUN BANK BLDG	
	400 COMMERCE AVENUE	
•		1
	037401080960 L 99999 TENNESSEE VALLEY AUTHORITY	
•	H J GHEEN Directur of Nuclear Power	
•	GHATTANDUGA TN 37401	i
-	015402079056 1 99699	
•	TENNESSEE VALLEY AUTHORITY	
	POWER PLANT SUPERINTENUENT	
•	DECATUR AL 15602	
	037401001380 1 99999	,
•	TENNESSEE VALLEY AUTHURITY DAVID LAMHERT	
_	GU CHESTNUT STREET TUNER LI CHATTANGUGA IN STAUL	

A	1 8	Ğ	35 E 1 P 4	.) T	5 11	31	1	8 NI	DA	P			5	34	ų ç	ł		
1	C	F	P	'n	ŧ	51	Ū	ËŇ	1-	EN	GI	NE	ERI	N	G			
\$0 A	X S	T	F	A	R	M (IN	GD	٩L	Ł		1	NΥ		11	17	35	
22	r	9. HI	22	0	9	14	3	8 8	2	1 FN	FR	. v	90)9 :0	99	•		
ų	Š	A !	V.		Ģ	Įί	D	ĒŇ	-									
1	Ē	N	Ē	G	Ŷ	ີເ	E	NT	ER	ł			_					
. N	U.	× I	/ 1	L	L	E							I N		37	19.	22	
ļ0	2:	2	54	0	4	сļ	6	8		1	• • • •	~ .	.9	99	99	2	-	
N	F	ò	Ř	14	Ĩ	ığ	jŅ	Č	ĒĀ	iΓê	R		251	ι¥	14	. C	2	
3	ö	į	56	C	0 0	NĈ) t	KV AV	Ĕ	. 2 3								
íA	L	T	44	M	ł							1	MA		0;	22	54	
• •	,	2	• •	20	A	1 <		,		1			9	90	90	>		
É	Ń	Ņ	Ē	ŝ	Ĕ	Ê	Ö	ĔΡ	I.	ŎF	PI	UB	ĻĮ		Ĥ	T	H	
ij	č	н	Ľ	ÌĻ	н. 	H	Ä	08	LE	Y	AL	-	CAL		The second secon			
2	R -	2		20	C	OA	10	EL	ι	HU	LL	8	10	5				
iΑ	5	H!	V I	I L	L	E							TN		3	72	19	
13	7.	2	15	10	P	6(F	2	9 FC	r		AT	•	2	99 F N	99	? 2 A 1		
į	Ļ	Ļ	į	Į,		₿	Ă	ųě	83	RD	0	• • • •						
ü	è	Ř	Ē	Ē		ţά	າບົ	ņ	^u e	501	LD	ĬŇ	<u>ה</u> בי			• •		
4	Ş	H	¥ 1	L	L	Ł							TN		5	72	19	
13	7	31	8 1	0	7	90)5	5		1			94	99	94	,		
E	NI	Ņ	ÈŚ	ŝ	E	Ē	V	ÀL PS	LĘ	¥,	AU	TH	OR.	11	۷			
	Ì		ŝç	Ì	Ţ	Ë	5	 	•		-							
ğ		ē,),	ļ	Ż	ιç	įõ	E 48	•				•					
	-							_					* **			-	91	
Ē	Ń	N	ËŜ	ŝ	ŧ	Ē	et Y	ÅL	ĻE	Y	AU	TH	DŔ	íï	Y	•		
(.	T N	1	ς Cι	۳ ال	Ŷ	R El	۱,	3	2									
0	g.	-	ĹĊ)M H	MB	Êf B	38	£	A۷	/E								
N	Ō	X	V	L	L	Ē							Th		3	79	02	
13	5	7	į	Q	P	44 c	54 U	3.		Į.		7 M	99 119	99 1 1	9	9		
ò	N	Ş,	Ì	เบ	č	ţ	ļò	Ñ	4.6 w 8		-0				•			
R	ğ.	5	ĒÇ	T	Ē	H,	Ń	ÂG	EH	i								
, 10	Ľ	Ľ		10	0	Č,	0	0					AL		3	57	52	
13	7	0	74	0	0	46	53	7		1			99	99	9			
	친	S	Ē	i S	Ē	Ę	V LU	AL	LE	Y	AU	TH	OR	11	¥			
l M	Ĩ	1	H/	Ĭ	Ĥ	Č	ĴŤ	Ê		,								
ÿ	٦	Ş	į	į .	Z	0	0	~~		•			TN		a 1	20	76	
							-	-					 	00				
Ē	N	N	ĒŠ	ŝ	Ĕ	Ē	Y	ÁĻ.	L E	Ŷ	AU	TH	ON.	ii	Y			
10	ü	R (ł	2	00 JC)0	٤S										
)£	C	8	۴ı	m									AL		3:	6	02	
													_		_	_		
13	7 N:	3	19 E	10	1	Q3 E	36 V	3	LF	ł	AU	Тн	9' DR	99 1	99	•		
ĥ	ç	U	u)	A	H	<u>,</u> ,	١Ř	S.	ĩ	Ĺ	2							
4	Ă	N	ſ	្រី	ŭ	21	R	IN	TE	ND	EN	ĩ						
1.	1.1	÷ •	_			_												

037401006275 1 99999

B

037401081040 1

99499

	شي ب ما المعالية	N	الأم الإيرانية (14 مجروعية) عنها من الأمريكي (14 محرومية) عنها المراجعة (14 محرومية) من الأمريكي المراجعة المراجعة (14 محرومية) من الأمريكي من المراجعة (14 محرومية) من الأمريكي من المراجعة (14 محرومية) من الأمريكي من المراجعة (14 محرومية) من المراجعة (14 م المراجعة (14 محرومية) من المراجعة (14 محرومية) من المراجعة (14 محرومية) من المراجعة (14 محرومية) من المراجعة (14	
			DAISY	TN 37319
**	0374010H1040 1 TENNESSEE VALLEY AUTH MATTS DAR 1 5 2 D D FUNCAV	99999 Chilla	US7401006275 1 TENVESSEL VALLEY AUTO DAVID DEMSBY	99949 Hurity
-	PROJECT ENGINEER 400 CHESTNUT STREET T CHATTANOUGA	GWER 11 TN 37401	CHATTANDOGA	TN 37401
-	037401001554 1 Tennessee Valley Auth H & Parkis Manaler de Pourr	99999 IGRITY	037401023241 1 TENNESSEE VALLEY AUTI HUGH & PARRIS	99999 Hority
~	SOUA CHESTNUT STPEET CHATTANOUGA	TOWER 11 TN 37401	5 104 CHESTNUT ST TOW CHATTANUUGA	ER 11 TN 37401
~	035752080959 L TENNESSEE VALLEY AUTH A M CUALLS DIAM SUBSEINTENDENT	39999 Iarity	037902005175 1 TENNESSEE VALLEY AUTI NUCLEAR ENGINEERING A	99999 HURITY BRANCH
^	PO BUX 2000 HOLLYHOUD	AL 35752	ADD SUMMIT HILL AVE #10C126C-K KNUXVILLE	W10C126 TN 37902
^	037401005840 1 TENNESSEE VALLEY AUTH RCN ROGERS	99999 Chity Cher II	037902000459 1 TENNESSEE VALLEY AUTH HERBERT S SANGER JR GENERAL FOUNSEL	99999 HOR I T Y
•	CHATTANUUGA	TN 37401	400 COMMERCE AVENUE E11833 C KNGXVILLE	TN 37902
•	037319080955 1 TENNESSEE VALLEY AUTH Sequiyah NPS 1 E 2 G G STALK	99999 IOR [] Y	037401001342 1 TENNESSEE VALLEY AUTO D TERRILL 1 TENSING ENGINEER	99999 107179
0	PRÚJECT HANAGER PO BOX 2000 DAISY	TN 37319	400 CHESTNUT STREET	TOWER 11 TN 37401
0	037381080949 1 TENNESSE VALLEY AUTH WATTS BAR HPS 1 G 2 G WADEW ITZ PRUJECT MANAGER	99999 Iority	037902005007 1 TENNESSEE VALLEY AUTI DUNALD L WILLIAMS 400 W SUMMIT WILL DR W10885C.	99999 HORITY
C	SPRING LITY	TN 37381	KNUXAILLE	IN 37902
с 5	037902057407 L TENNISSEE VALLEY AUTH DCRALD L WILLIAMS 400 - Summit Hill Dr W10H85 Knuxyille	99999 IDRITY TN 37902	037401001296 1 TENNESSEE VALLEY AUTI JERRY & WILLS 400 CHESTNUT ST TOWEI CHATTANOOGA	99999 HORITY R II TN 37401
•	000014044700	00000		
ر	TERA ADVANCED SERVICE J LUNG VP ADVANCED SERVICES 7101 WISCONSIN AVENUE	S CCRP	TERRA TECHNULOGY CORI SALES & MARKETING STEPHEN D PORELL SYSTEM SPECIALIST	, ,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,
-	DETTESUA	HD 20014	REUMUND	WA 98052
-	D06095026904 1 TERRY CURP GEURGE ENIEP SENTUR ENGINEER PU BOX 555 WINUSUR	99999 CT 06095	078711006153 1 TEXAS ATTORNEY GENER. ENVIRON PHOTECTION D BHIAN BERWICK ASSISTANT ATTORNEY GI PUR 12548 CAPITOL	99999 ALS OFFICE IVISIUN ENERAL STATION
-	078711005467	0000	AUSTIN	17 78711
	TERAS ATTORNEY GENERA ENVIRON PRUTECTION DI SUSAN PLETTMAN PO HUX 12548 CAPITAL AUSTIN	STATION TX 78711	TLASS ATTCRNEY GENERAL ENVIRONMENTAL PROTECT DAVID J PREISTER ASST ATTOPNEY GENERAL PU EUX 12548 CAPITOL	LS OFFICE
-	078756002533 3 TEXAS DEPT HE HLALYH BUR LF RADIATILN CLNT DAVID K LALKER DIRECTUR LIOD WEST 49TH STREFT	99979 RUL	AUSTIN 076101002903 1 TEXAS ELECTRIC SERVIO P U AUX 970 FUKT WORTH	13 78711 999999 Le Cu Tx 76101
	AUSTIN	TX 78756		

	ULVETTER 1100 NEST 44TH STREET AUSTIN TX 78756		
	077012053119 1 99999 TEALS PIPE SENUING CO SUBSUY-MENDIX CURP STEVE SPITH	077074006245 1 TEXAS PIRG UNIVERSITY OF HOUSTO JAMES SCOTT	99999 N
-	HGR U A PO BUX 5183 HGUSTEN TX 77012	HOX 237 UC HOX 211 UC	TX 77074
-	075201006229 1 99997 Texas utilities generating Cu R J Gary Frights uply a classifier of the same sp	075201628147 1 TEXAS UTILITIES GENE H H GARY	59999 Rating Lu
~	ZODI MAYAN TOWERS DALLAS TA 75201	ZJOI BRYAN TOMER CALLAS	TX 75201
~	075201006518 1 99999 Texas utilites services inc Humer C Schwidt Manager-Nuclear Services	UUBB69U24729 1 THOMAS & BETTS CU FRANK J REYNOLDS 92U ROUTE 202	99999
^	2001 GRYAN TUWER Dallas - Tx 75201	PARITAN	NJ U8864
^	OIBIO2006464 1 99999 Thomas & Hair Charles & Ellict 123 North Fifth Street	017108005698 1 THOMAS & THOMAS CHARLES & THOMAS 212 LOCUST STREET	99999
~	ALLENTOWN PA 18102	BUX 999 HARRISEURG	PA 17108
~	035630081373 1 99999 TIMES DAILY LAURA SIMMONS PO RIX 797 ELECTOR	020045026885 I Tuleuu blade newspap Washington bureau Michael J Wouds Scientes Entor	99999 ERS
.	PLUKENCE RC 33030	1280 NATIONAL PRESS	BUILDING CC 20045
0	043652004958 I 99999 Toleun Edisin Co Richard Crouse Vice President Nuclear 300 Madison Ave -Stop 712	043652005787 1 Toledo Edison Co Kilmard P Crouse Vice President-Nucle Edison Plaza-300 Mag	99999
0	STOP 712 TOLEDO OH 43652	TOLEDO	UH 43652
	043449081508 1 99999 Tileuu Edisun Co T d Murkay Station Superintendent 5501 n State Roufe 2	043652007103 1 TULEDU EDISUN CO TED MYERS MANAGER-NUCLEAR LICE 300 MADISON AVE EDIS	99999 NSENG ON PLAZA
<i>'</i>	DAK HARBOR OH 43449	TOLEDO	OH 43652
•	054216005830 1 99999 TGWN OF CARLTON Stanley Lacrosse Chairman	054632005366 1 Tuwn CF Genda Tuwn Chairman Ruute 1	99999
~	AUUTE I KEWAUNEE WI 54216	GEUNA	WI 34632
L.	006103005738 1 99997 Town Of Haddam Blard of Selectmen Town Hali	014519005395 1 Tuwn UF ONTARIO Supervisur 1350 RIDGE ROAD	99999
	HAUDAN CT 06103	ONTARIO	N7 14519
-	006062020056 I 99999 TGWN UF PLAINVILLE UIV UF EMERGENLY SERVICES	001367005412 1 Tuwn GF Rowe Buard OF Selectmen	99999
•_	RADITIUSICAL DEFENSE DEFECER 28 LOCUST ST PLAINVILLE CT 06062		MA 01367
-	OI31260J6536 1 99999 TUWN UF SLPIEA RUHERT P JUNES SUPERVISLR	054241005865 1 Toan OF Two Creeks Girdun Mlaha Tuwn Chairman	99999
	KU #4 OSWEUD	RUUTE 3 THO RIVERS	WI 54241

÷.

. ...

MA 01307	49999 45 49999	4 INC 99993 4 INC 99993 41 SUPVR MI 44106	т .0к М 49043	10164 IW 66666	44604475 Allohats *J 08038	ромоб (жут рез Ссн яса о 17057	10901 AN	99997 R VA 22041	41MG 1NC Иле илс 5	PA 1/815 59995 1GR 1GR 63141	1165 1AC
	054241005865 1547 01 140 CKE 1547 01 140 CKE 1547 04 1444 1547 04 1448 1416 1 141 15 5 141 15 5	04810604401 19mnseng 60111 (19mnseng 60110 (1 f112patricu (1 f12batricu (1 f02 1363 66461 Mn arduk	049043005381 1UMN5H1P UF COVER 1UMN5H1P SUPERVIS RUULE 1 BUX 10 VAN BUREN COUNTY	049101005692 1 10280541P OF LAKE 8481E SCHULER 50PERV150R	008038005441 I TOWNSHIP OF LOBER MARY U HENDERSON MUNICIPAL BULLDIM MANCUCKS BRIDGE	01 7057005430 1 1040541P 0F 5 1040 0441D 0 maxwell 0441D 0 superil 8040 0f 500ervis 810 01 6 1040	010601053120 1 Ransnuclear Inc Prestien Une Mith Broadway White Plains	022041080951 1 14 ANSNUCLEAR INC 6 STUNEBURNER 18AFFIC COUNDINATO 5205 LEESBURG PIRE 1ALLS CHUNCH	017415063795 1/1 Valley Enginee Paul Valley Enginee Vice President Pigmland Drive Sma Pigmland Drive Sma	81.00450046 063141036029 1180044 5161 NC 418044 5161 NC 4181 7 45504 ANC 6 90 ADX 2/370 51 LUUI 5	099352066741 [0%C NUCLFAR & NDUSTR P EARING RECORDS SUPY EARING RECORDS
1 LT 06962	93744 13126	99994 [] CI U6385	NC 99999 NC 89999 REET #106	99999 LN 08723	99999 LOMAYS NJ 08035	11104 11104 PA 17315	99999 186 8 D1 / 1510N CA 94621	99999 Afions Ny 10601	49949 46 INC CA 94550	99994 Ance NY 11237	66666
KALNU, I J	013126006536 1 10mm CF SCP1FA Robert P JURES SUPERVISCR SUPERVISCR	DOBJESSUSAUH I TOWN OF WATERFORD Mall UF BELECUSSS Fast Selections Fast Ford Post RUM	0481060667JU 1 TUWNSENU E HUTTUM I K G HFRUKN 2245 STUTH STATE ST 87X 1968 ANN ARBUR	008723005321 1 10885417 15 14CE7 84705 818 6551 14CE7 8040 5088651 14CE7 8040	DOBOJROUS632 TC#NSHIP OF LOWER AL Savel L E DUNELSCN Mayur Municipal Mall Manucicis Bridge	017314005775 J Dunskip of Pealm Bo Boad uf Supenyisons Albent Steel Chaiman Steela	004621052968 1 Ransamerica delaval Engine and Courre Ssu Courre Ssu Courre Ssu Courre al Age Courre al Age Darland	010001055002 1 TRANSAUCLEAN INC JOHN AANGUST NERL MANAGER-(IPER I NORTH MANAGER-(IPER) MMITE PLAINS	9445540056865 1. TRL VALEY CNCINEER 1.324 CUMANGUN BLVD LIVENMJRE	911237053122 1 5.01622100 5.017842114 8.01784214 123 VAPICK AV(011901005351 1 TWCMEY LATHAM L SHFA Stephen Latham L Shfa Stephen Latham Steelt PL 441 194

1.14

)

ດ

	2	s a g	
P	Mihayyo Pu Hux 490 RICHEAND & & 99352	MAK HIDGE IN STUSS	
Þ	UI0987081115 1 99999 UNION CARAIDE CORP RACIUCHEMICAL PRODUCTION J J MCGUVERN	014150022591 1 59999 UNION CARBIDE CORP T SMIST P-D- RCT 44	
ר	MANAGER PD AUX 324 TUXEDD NY 10987	TGNAWANDA NY 14150	
ר	063166001356 1 99999 UNICN ELECTRIC CU J E HINK ASSISTANT TU TUC CIN COMMON	Du 3166006486 1 999.jy UNION ELECTRIC CO NUCLEAR ENGINEERING	
า	PU BUX 149 SAINT LUUIS MU 63166	DUNALD W CAPUNE MANAGER PU BGX 149 ST LOUIS MO 63166	
`	063166064877 1 99949 UNICA FLECTRIC LC J. GRUSS	065251064692 1 99999 UNIUN ELECTRIC CD UFNU-DOCUL CONTROL REFERENCE OM	
^	RECORDS MANAGEMENT SUPVR DP BCX 149 ST LOUIS MD 63166	J P KANYO PO BUX 620 CA 460 FULTON MO 4626	
C	065251081504 1 99999 UNIGN ELECTRIC CO STEVE MILTENBENGER	063166079475 1 99999 UNION ELECTRIC CO	
^	PL 8CX 620 FULTON MD 65251	SUPERINTENDENT LICENSING PII RDX 149 - MAIL CODE 470 ST LOUIS MG 03166	
^	063166002916 1 93999 UNION ELECTRIC CO NUCLEAR ENGINEERING DEPT	063166001553 1 99999 UNION ELECTRIC CO DONALD F SCHNELL	
•	VICE PRESIDENT-NULLEAR PO BCX 149 MAIL CODE 400 ST LUUIS MO 63166	VICE PRESIDENT - NUCLEAR PU BUX 149-MAIL COJE 400 ST LOUIS MO 63166	
,	063166079379 1 99999 UNION ELECTRIC CO NUCLEAR CONSTRUCTION M H MEHER	020036005756 1 99999 UNION OF CONCERNED SCIENTISTS STEVEN C SHOLLY	
Ċ	MANAĞER PO BOX 149 - MAIL CODF 400 ST LOUIS MO 63166	SUITE 1101 WASHINGTON DC 20036	
с	048184053126 1 99999 UNISTKUT CORP C L LCDK MANAGER JUALITY CONTROL 4110 South Wayne RDAU Wayne MI 48184	019101008692 1 99999 UNITE' ENGINEERS & CONSTRS INC SAFETI LIC ADVANCED ENGR DEPI HLNRY L BERMANIS 30 SOUTH ITTH STREET BOX 8223	
	019101005492 1 99999 UNITED ENGINEERS & CONSTRA INC.	PHILADELPHIA PA 19101 019101001345 1 99999 UNITED ENCIMEERS CONSTRUCTION	
-	A M LANER Phojett Manager 30 Suuth 17tm St Pu Bux 8223 Philauelphia pa 19101	V MANI 30 SUUTH 17TH STREET PHILADELPHIA PA 19101	
~	099352064745 1 99999 UNITED ENGINEERS & CONSTRS INC J F MCCALEH SUPERVISCR-DOCUMENT CONTROL 1814 STEVENS DRIVE RICHLAND #A 99352	019101001238 1 99999 UNITED ENGINEERS & CONSTRS INC AUV ENGR DEPT ADULPH T MOLIN PO ROX 8223 PHILADELPHIA DA 10101	
-	006506001439 1 99999 UNITED ILLUMINATING CO JLMA D FASSETT PRESIDENT BO TEMPLE STREET MEM INVESTIGET	0063820J8901 1 99999 JNITED NUCLEAR CORP NULLEAR & IND SAFETY DEPT WILLIAM F KIRK ANAGER	
-	055330008693 t 99999	UNCASVILLE CT 06382	
	UNITED POWER ASSOCIATION JACOBB NONDBERG PRESIDENT	UNITED STATES TESTING CO GFNE BASILE 1415 PARK AVENUE	
	CAN FITCH AN 5533C	HUBOKEN NJ 07030	

 \mathfrak{I}

۰.

-
-
•
•
•
•
٠
•
•
٠
•
•
•
٠
•
٠
•
J
i •

	ELK - Lut -	بافترد ما	March Lts	، ایر ش
	046312025359	99999 05 8 4 6 5 5 6 8	DCANADUS1594 1	99999
£	LUCAL 1010 CLIFFURD MEZO	OF ARCHICA	LE CENTRE HOSPITALIE JEANLOUIS BENEDETTI 2705 BIVD LAUPIER	R
	ACTING PHE IDENT 3703 EUCLID AVE EAST CHICAGE	IN 46312	TURICULOGY QUEBEL LIV-4G2	CANADA
-	085721006037 1 UNIVEPSITY OF ANIZUM DAVID L HETRICK	49999 A	032611080950 1 UNIVERSITY DF FLORID DEPT-NUCLEAR ENGNG S	99999 A CIENCES
-	PROF OF NUCLEAR ENGI TUCSUN	NEERING AZ 85721	A H JACOBS CHAIRMAN 202 NUCLEAR SCIENCES GAINESVILLE	5 CTR FL 32611
-	D66044021426 1 UNIVERSITY OF KANSAS Schubl of Engineerin	99995 G	001854080944 1 UNIVERSITY OF LOWELL LEON E BEGHIAN	99999
~	RÚSSELL MESLEP Reactur supervisor 102 Nuclear Reactor Lawrence	CENTER KS 66044	ACTING DIRECTOR OF P 1 UNIVERSITY AVE LOWELL	MA 01854
^	DO1854006416 1 UNIVERSITY CF LOWELL NUCLEAR CENTER	99999	020742079419 1 UNIVERSITY OF MARYLA	99999 ND
^	T WALLACE REACTOR SUPERVISOR] UNIV AVE LOWELL	NA 01854	COLLEGE PARK	MD 20742
<u> </u>	065211006485 1 UNIVERSITY OF MISSOU MISSOURI/KANSAS SEC- MILLIAM H MILLER	99999 R1 AM NUC SOC	087131066750 1 UNIVERSITY OF NEW MI DEPT CHEMICAL & NUCH J & ROBERTSON	99999 EXICO LEAR ENGR
0	DEPT OF NUCLEAR ENGL 1026 ENGINEERING BLO COLUMBIA	NEERING G HD 65211	CHIEF REACTOR SUPERY ALBUQUERQUE	NM 87131
0	027514007044 1 UNIVERSITY OF NORTH ENVIRUNMENTAL LAW PR GEORGE JACKSON	99999 Carul Ina Uject	073019080062 1 UNIVERSITY OF OKLAHO SCHOOL-CIVIL ENG & I EDWIN H KLEHR	99999 DNA ENVIRON SCI
0	CHAPEL HILL	NC 27514	202 W BOYD NORMAN	OK 73019
0	092110000875 1 UNIVERSITY OF SAN DI SCHOOL OF LAW RILHAHD J WHARTON	99999 EGO	078712079370 1 UNIVERSITY OF TEXAS MECHANICAL ENGRG THUMAS BAUER	99999
с U	EÑVIRON LAW CLINIC A San Diego	LCALA PARK (A 92110	SUPERVISOR-NUC ENGR TAY 167 AUSTIN	TEACH LAB
C	078712080912 1 UNIVERSITY OF TEXAS BUREAU OF ECONOMIC (WILLIAM FISHER	99999 EOLOGY	079712079372 1 University of texas College of Engineer E f gluyna	99999 I NG
Ċ,	GEOLUGY BLOG ROOM ! AUSTIN	129 1x 78712	COCKRELL HALL 10 31 AUSTIN	0 TX 78712
С.	078712079371 1 UNIVERSITY DF TEXAS Mechanical Engineer: D E Klfin	99999 14G	084112000172 1 UNIVERSITY OF UTAH NUCL ENGINEERING LA SALT LAKE CITY	99999 BORATORY UT 84112
<u> </u>	DIKECTOR-NUC ENGR TI Tay 167 Austin	TX 78712		
-	022901081042 1 UNIVERSITY UF VIRGI J S BRENIZER DIRECTOR UF REACTOR	99999 VIA FACILITY	098195006055 1 UNIVERSITY OF WASHI Department of Ocean George C Anderson	99999 Ngtun Dgraphy
٠.	CHARLOTTESVILLE	VA 22901	SEATTLE	WA 98195
:	084602079366 1 UNIVERSITY-BRIGHAM CDLLEGE PHYS & MATH	99999 Young Matic Sci	U20064U79399 1 UNIVERSITY-CATHOLIC W KEENE RADIATION SAFETY OF	99999 ELC IAI
	FACILITY CHIEF	UT 54602	PANGBORN HALL HASHINGTON	DC 20064

0

J

Paul n Habar	4 (1) (1) (1) (1) (1) (1) (1) (1) (1) (1)
DINECTUR 32NG G CHESTNUT STREETS PHILADELPHIA PA 19104	PUCATELLU IU 83209
038152030193 1 99999 UNIVENSITY-MEMPHIS STATE CEVTEN FOR NUCLEAR STUDIFS D M JONES D M JONES	048823081067 1 99999 UNIVERSITY-MICHIGAN STATE LOLLEGE OF ENGINEERING JAMES CARKICK DIV-ENGINEERING RESEARCH
MEMPHIS TN 38152	EAST LANSING NI 49823
039762080968 1 99999 UNIVERSITY-MISSISSIPPI STATE UEPT UF MECH E NUCLEAR ENGNU C T CARLEY MEAD ARE ME	014214080941 1 99499 UNIVERSITY-NEW YORK STATE Buffalo DIR-NUC SCI & TECH Facility 14214 Rotary Road Ruffalo Vy 14214
MISSISSIPPI STATE NS 39762	
014260079414 1 99999 UNIVERSITY-NEW YORK STATE RES & GKAD STUDIES DONALD W RENNIE VICE PRESIDENT S52 CAPEN MALL NY 14260	060115021248 l 99999 UNIVERSITY-NORTHERN ILLINGIS BRUCE W VON ZELLEN BIOLOGICAL SCIENCES BI57531753 DEKALB IL 60115
ARTERSI AT 1000	043210079377 1 99999
UNIVERSITÝ-NORTHERN ILLÍNÓIS DEPT OF BIOLUGICAL SCIENCES BRUGE VUN ZELLER DEKALB IL 61107	ŬNĴVËRSITY-OHIO STATE Radiation Safety George A kave Health Sciences Staff Rep Rm 232-370 m 9th Ave Columbus oh 43210
097331006086 1 99999 UNIVERSITY-UREGON STATE School uf Oceandgraphy Robert L Hulton Corvallis or 97331	016802031127 1 99999 UMIVERSITY-PENNSYLVANIA STATE HEALTH PHYSICS OFFICE RUDGER W GRANLUND HEALTH PHYSICIST 228 ACADENIC PROJECTS BLDG UNIVERSITY PARK PA 16802
016802005515 1 99999 UNIVERSITY-PENNSYLVANIA STATE WILLIAM LUCHSTET 104 DAVEY LABDRATORY UNIVERSITY PARK PA 16802	00855R066980 1 99999 UNIVERSITY-PRINCETON Schoul of Engr & Applied Sci Rubert C Axtmann Acrs NI 08558
	PRINCEION
047907007317 1 99999 UNIVERSITY-PURDUE DEPT OF NUCLEAR ENGINEERING PAUL LYNDYDIS IN 47807	047907031569 1 99999 UNIVERSITY-PURDUE School of NUCLEAR ENGR ELDON R STANSBERKY PLACTOR SUPERVISOR
MEST CAPATELLE IN 41101	WEST LAFAYETTE IN 47907
014778081879 1 99999 UNIVERSITY-ST BUNAVENTURE SCHOUL OF BUSINESS S M HAI	013210001373 1 99999 UNIVERSITY-SYRACUSE COLLEGE OF LAW RICHARD GOLDSMITH RICHARD GOLDSMITH
ASSOC PROF OF FINANCE ST BUNAVENTURE NY 1477B	SYRACUSE NY 13210
019140066216 1 99999 University-temple RDIATION SAFETY UFFICE	077843080937 1 99999 UNIVERSITY-TEXAS AGM GFC-UNIVERSITY RESEARCH R & COCHRAN
ASST RAUIATION SAFETY GFFICER 3307 NURTH BRLAD ST PHILADELPHIA PA 19140	CHIEF ADMINISTRATIVE OFFICER COLLEGE STATION TX 77843
026505080948 1 99999 UNIVERSITY-WEST VIRGINIA COLLEGE OF ENGINEERING	001923066627 1 99999 URS JOHN A BLUME & ASSOC ENGR BOSTUN OFFICE 130 Sylvan (freet
REACTOR DIRECTOR WV 26505 Morgantown WV 26505	DANVERS MA 01923
0944C2U8U837 1 99999 URS JOHN & BLUME & ASSOC ENGR MARKETING DIVISION 951 MAREMERS ISLAND BLVD	074105066497 1 99999 URS JOHN A BLUME E ASSOC ENGR Ravindra P Mistry Pruject Manager 130 iessie Stefet
SAN MATEU CA 94402	SAN FRANCISCU CA 94105

٦.

7

C

3

ţ

02654500Tue5 1 19973	097310005969 1 99994
US DUE COURDINATION & SPEC PROJECTS HERBERT FEINPLIM HERBERT FEINPLIM	US DOE SITING & PEGULATION DUNALD W GUDARD
NE-550	HOW 111
	SALET UN 47
OZC545005065 I 99999 US DEE DEC SAFETY & QUALITY ASSURANCE ANDREW J PRESSESKY DIRECTUR	043401028822 1 99999 US DICE IDAHO OPERATIONS K J BEERS 550 2ND ST IJAHU FALLS ID 83
WASHINGTON DC 20545	
083401005974 1 99999 US DOE TUAHD OPERATIONS R F TILLER	040207026663 1 99999 US ECOLCGY INC H M HIPSHER
550 SECUND STREET	MANAGER CORPORATE SAFETY
	LOUISVILLE KY 40
060604006242 1 99999 US ENVIR PRUTECTION AGENCY	080203006314 1 99999 US ENVIR PROTECTION AGENCY
FEDERAL ACTIVITIES BRANCH Regiunal rad representative	RADIATIUN PROGRAMS Regional representative
230 SOUTH DEARBORN ST CHICAGO IL 60604	1860 LINCOLN STREET DENVER CO 80
002203005405 1 99999	010007005326 1 99999
REGION 1 OFFICE	KEGION 11 OFFICE
REG RADIATION REPRESENTATIVE JFK FEDERAL BUILDING	REGIONAL RADIATION REP 26 Federal plaza
BOSTON MA 02203	NÉN YÜRK NY 10
019106005452 1 99999 US ENVIR PRUTECTION AGENCY	030308005673 L 99999
REGION ITI CFFICE Regional Radiation Rep	REGION IV REGIONAL RADIATION REPRESEN
6TH E WALNUT STREET CURTIS BLD 6TH FLOUR	345 COURTLAND ST NW ANK 460
PHILAUELPHIA PA 19106	ATLANTA GA 30
US ENVIR PRLIECTION AGENCY	094111006284 1 99999 US ENVIR PROTECTION AGENCY
REGIONAL RADIATION REP	REGION IX OFFICE REG HADIATION REPRESENTATIV
ATLANTA GA 3030R	SAN FRANCISCO CA 94
060604081503 1 99999	060604005942 1 99999
US ENVIR PROTECTION AGENCY Region V	US ENVIR PROTECTION AGENCY REGION V OFFICE-FED ACTIV BI
EIS LUCRDINATOR 230 S DEARBORN ST	REGIONAL RADIATION REP 230 S DEARADRN ST
CHICAGO IL 60604	CHICAGO IL 600
075270006534 1 99999	075270005813 1 99999
REGIÓN VI	REGIUN VI OFFICE
1201 ELM STREET	EIS COORDINATOR 1201 ELM STREET-1ST INTERNA'
DALLAS TX 75270	DALLAS TX 752
064106005926 1 99999 US ENVIR PROTECTION AGENCY	020472005740 1 99999
REGION VII REGIONAL RADIATION NLO	GENERAL COUNSEL
324 E 11 STREET MO 44104	1725 I STREET NH
	Shinding - DC 204
020426064596 1 99999 US FED ENERGY REG CUM	020426006102 1 99999 US FED ENERGY REG COM
E M DAWSUN ACLUUNTANT LUDE-FERC-DCAL	UUINTEN LAWSON 885 NURTH CAPTTOL NE
825 NORTH CAPITOL ST NE WASHINGTON	ROOM F611 DF 300
020515006403 1 99999 US_HUUSE_IIF_REPRESENTATIVES	020515001253 1 99999 US HOUSE OF REPRESENTATIVES

GULATION DARD JUSTRIES BLDG OR 97310 22 1 99999 10 OPERATIONS 10 83401 3 1 INC 49999 PORATE SAFETY KY 40207 4 1 99999 DTECTION AGENCY RUGRAMS PRESENTATIVE N STREET CO 80203 6 1 99999 OTECTION AGENCY IFFICE DIATION REP PLAZA NY 100 NY 10007 3 1 99999 DTECTION AGENCY DIATION REPRESENT GA 30308 4 1 99999 UTECTION AGENCY FFICE ON REPRESENTATIVE T STREET CO CA 94111 2 1 99999 DTECTION AGENCY FICE-FED ACTIV BR DIATION REP OKN ST IL 60604 3 1 99999 UTECTION AGENCY FFICE Alor Reet-1st Internate TX 75270 0 1 99999 GENCY MGMT AGENCY NSEL ÈT Nw DC 20472 2 I GY REG COM WSON Apitol Ne 99999

DC 20426

.

	المانية من الأراف العامية. توقيق الاستعال التي تعالي المنافع المانية العامية العامية العامية العامية العامية الع	UN FEITENCE OF FEITEN DUINTER TABUTE
) _	ACCOUNTANT (DHE-FERCHUCA) 825 NORTH CAPITCL ST NE MASHINITUN	875 NORTH CAPITOL NE KOCH F611
		HASPINUIUN DC 20420
/	020515006603 I 99989 US HUUSE OF REPRESENTATIVES	020515001253 1 99999
-	TUM CURCURAN WASHINGTON DC 20515	LAWRENCE COUGHLIN
•		
	020515081495 1 99999	030515041531
•	US HOUSE UP REPRESENTATIVES Envir Eyergy & Nate Resourses John R Gallonay	US HUSE OF REPRESENTATIVES MTE CN INTERIOR & INSULAR AFF KICHARD UDFLI
•	SUBEMTE STAFF DIRECTOR Rayburg Hob Room 8-371	SUBCHTE ON OVERSIGHT & INVEST
•	WASHINGTON DC 20515	WASHINGTUN DC 20515
٠	020380079658 1 99999 US MARINE CORPS HEADQUARTERS	020590006830 1 99999 US MARITIME ADMINISTRATION OFC OF ADVANCED SHIP OPERATIOS
	GLAMANDANT OF THE MARINE CORPS	Z LEVINE NAR-930 ROOM ALOO
۲	LMA 3 WASHINGTON DC 20380	WASHINGTON DC 20570
	020555063622 1 99999	020555003911 1 00000
•	US NAL BRANCH LHIFF	US NRC
	SAFEGUAROS BRANCH	EXECUTIVE DIRECTOR
•	NASHENGTON DC 20555	WASHINGTON DC 20555
	020555052567 1 00000	
•	US NRC	U2 U3 35 U3 146 / 1 99999 US NRC
	CHAIRMAN	ACRS PROJECT REVIEW ASST EXECUTIVE DIRECTOR
•	WASHINGTON DC 20555	H-1014 WASHINGTON DC 20555
	020555012329 105 99999	020555051515 1 (00000
۲	ADM DIV OF RULES & RECORDS	US NRC
	BRANCH CHIEF	DIRECTOR
•	7414-MNR8 NASHINGTON	ENS-263A WASHINGTON DC 20555
•	U12 501005810 1 99999	020555003909 1 99999 US NRC
-	ARKANSAS NPS RESIDENT INSPECTOR	
	PO BLA 2090 RUSSELLVILLE AR 73801	EN529
•	A 120VI	HASHINGTON DE 20555
•	020555006053 1 99999 US NRC	020555003908 1 99999
•	ASLAP	ASLAP
	Ew-529	CHAIRNAN Fuldag
•	WASHINGTON DC 20555	NASHINGTON DC 20555
	020555006052 1 99999	020555028228 1 00000
•	US NRC ASLBP	US NRC
	WASHINGTON DC 20555	LIBRARY
•		WASHINGTON DC 20555
	020555967119 1 00000	015077005407
•	US NRC ASLBP	US NRC
	TECHNICAL ADVISER	RESIDENT INSPECTOR
•	WASHINGTON DC 20555	PU BOX 298 Shippingport pa 15077
	015077077514 1 99000	1135752005157
۲	US NKC BEAVER VALLEY 2 NPS	US NKC 99999
	RESIDENT INSPECTOR	PESIDENT INSPECTUR

	TERRITAR CONTROL			
	とおえる3月 ニネトル1900月10~	86 20555	SHIPPINGPORT	PA 15077
	## 744 1 160 1 174			
	015077077514	99999	035752005157 1	99999
-	USINKC		US NEC ALLEFONTE NOS	
	BEAVER VALLEY 2 NPS		PESTOFNT INSPECTOR	
	PU BCX 181		PC GLX 677	A1 36767
~	SHIPPINGPORT	PA 15377	HULLTHUUU	ML 33132
		00000	AL 04 07 004 51 3 1	00034
-	049720005385 1	494 99	US NRC	*****
,	BIG ALLA POINT NPS		ARA TOWGOD NPS	
	RESIDENT INSPECTOR		REAL BOX 79	
-	CHARLEVOIX	MI 49720	BRACEVILLE	IL 60407
	035611005457 1	99999	028461005901 1	99999
-	US NRL		US NRC Arunswter NPS	
	RESIDENT INSPECTOR		RESIDENT INSPECTOR	
	RT 2 BOX 311	41 35671	PG BUX IV>7 SouthPont	NC 28461
	ATHENS	AL SODII	SUUTHFORT	
			0 3065 5004754 I	69999
	061010066939 1	99999	U2U333UV4734 4	*****
	BYRCN NPS		CA DIGESTON	
	RESIDENT INSPECTOR		UFFILE DIRECIUM	
	AAAB NUKIH GERMAN G Byron	ÎL 61010	NASHINGTON	DC 20555
	U FRUN			
	020457005#77 3	99999	029745005345 1	99999
-	US NRC		US NRC	
	CALVERT CLIFFS NPS		RESIGENT INSPECTOR	
	PO BOX 437		ROUTE 2 BOX 179N	FF 307/F
	LUSBY	ND 20657	YURK	36 29143
	020555051558 1	99999	061727605303 1	99999
	US NRC		US NKC CLINTON NPS	
	WASHINGTON DC	20555	RESIDENT INSPECTOR	
			RR 3 BOX 229-A	IL 61727
			UL INTUN	
			0334 2000 EAOL 1	69999
	068321005856 1	99999	USZ6290USRUL I	33334
	COOPER NPS		CRYSTAL RIVER NPS	
	RESIDENT INSPECTOR		RESIDENT INSPECTOR	
	MU BUX ZIU BROWNSVILLE	NE 6832?	BOX 717	
			'RYSTAL RIVER	FL 32629
	A 34 4 00 0 5 7 8 4 1	00000	049127005693 1	99999
	043449005754 l US NRC	AAAAA	US NRC	
	DAVIS-BESSE NPS		DC CUOK NPS Resident inspector	
	SSUEN STATE ROUTE	2	7730 RED ARROW HIGH	AY
	OAK HANBOK	0H 43449	STÉVENSVILLE	MI 49127
	093424005301 1	99999	060450005404 1	99999
	US NRC		US NRC DRESDEN STATION	
-	RESIDENT INSPECTOR		RESIDENT INSPECTOR	
	PO BOX 364	CA 03434	RK #1 MORRIS	11 60450
-	AVILA BEACH	64 93424	THE REPORT OF A	12 00470
-			A 205556002040 1	68899
	052324005644 1	34333	NS NKC 1522003204 1	77777
-	DUANE ARNOLD NPS		ELD RULEMAKING & ENI	FORCE DIV
	RESIDENT INSPECTOR		9604 9604	
	RUFAL RUUIE I Pals	14 2324	WASHINGTON	DC 20555
-				
	036419005925	40334	013093005154 1	99999
	US NRC		US NRC	
	FARLEY NPS		REFERENCES MAD	
	ROUTE 2		PO ACX 136	
	BGX 24	A1 36110	LYCOMING	NY 13093

	RESIDENT INSPECTOR PO BUX 395	RESIDENT INSPECTUR RH #1 MODELE	U 60450
	AVILA BEACH LA 93024		00000
بر	052324005544 1 99937 US NPC DUANE ARNEED NPS	- US NAC - U	RCE DIV
•	RESIDENT INSPECTOR RUMAL RUUTE 1	UIVISION DIRECTOR 9604 HISTOR	7 20555
-	PALG IA 52324	MA \$111 NG 1 UM	
	036319005825 1 99999 US NRC FARLY NAS	013093005154 1 US NRC FITZPATRICK NPS	99999
	RESIDENT INSPECTUR ROUTE 2	RESIDENT INSPECTOR PO RGX 136	
<u>^</u>	BUX 24 COLUMBIA AL 36319	LICURIAG "	41 13043
~	080651006317 1 99999 US NRC	068023005925 1 US NRC	8999 9
	FORT ST VRAIN NPS Resident inspector 14805 — Flo County Road	PI LALHUUN NPS Resident inspector Po bux 309	
^	19 1/2 PLATTEVILLE CO 80651	FORT CALHOUN	NE 68023
	020555003920 1 99999	014519005394 1	99999
	US NRL GC General Counsel	GINNA NPS RESIDENT INSPECTOR	
~	H-1035 WASHINGTON DC 20555	1503 LAKE ROAD ONTARIU	NY 14519
	039150005160 1 99999	006423005406 1	99499
	US NRC GRAND GULF NPS	US NRC HADDAN NECK NPS	
~	RESIDENT INSPECTUR Route 2 Box 399 Plat Gibson MS 39150	EAST HAUDAN POST OFFI	CE CT 06423
**		037074008010 1	00000
0	027562006334 I 99999 US NRC Markis NPS	US NRC HARTSVILLE NPS	77777
-	RESIDENT INSPECTOR REUTE 1 BOX 3158	RESIDENT INSPECTOR PU NGX 468	Th: 17074
0	NEW HILL NC 27562	HARTSVILLE	1N 37074
4.4	031513051338 1 99999 US_NRC	008038005109 1 US NRC	99999
U	HATCH NPS Resident inspector	RESIDENT INSPECTOR	
U	RAXLEY GA 31513	HANCUCKS BRIDGE	NJ 08038
	020555063603 1 99999 115 NRC	020555063634 1 US NRC	99999
<u>с</u>	TE DIRECTOR	DEPUTY OFFICE DIR	
C	EM359 WASHINGTON DC 20555	HASHINGTON	DC 20555
-	020555064784 1 99999 US NFC	020555078987 1 US NRC	99499
1 4-	IE STAFF DIRECTOR ENGDLE EMENT STAFF	IE SECTION CHIEF UPERATIONS SUPPORT E	DUTY SEC
	Em359 WASHINGTON DC 20555	EH359 WASHINGTON	DC 20555
	020555056764 1 99999	020555056767 1	99999
·••	US NRC IE-CIV ENERGENCY PREPAREDNESS BRANCH CHI!F	IE-DIV EMERGENCY PREP DIVISION DIRECTOR	PAREDNESS
	ENERGENCY PREPAR LICENSING BR	EW-359 WASHINGTON	DC 20555
	MASHINGIAN UL 20000 020555063633 1 99999	020555063639 1	99999
-	US NEC IE-UIV ENGEG G QUALITY ASSUR	US ARC IE-DIV ENGRG & QUALIT	TY ASSUR
-	DIVISION DIRECTOR 3354 FWS	BRANCH CHIEF EVENTS ANALYSIS BRANC	CH

	all all the same of the first and a same	IL-ULY FROM I
	BUSA ENS	BRANCH CHIEF
•	WASHINGTON DC 20554	EVENTS ANALYSI
		MASHINGTON
_	020555463669 1 00000	
•	US NKE 99799	020555076814
	LE-DIV ENGRU & QUALITY ASSUR	US NRC TEADLY ENCOR
	ENGINEERING E THEM EUROPEAN	BRANCH CHILE
	JIZA FAS	UUALITY ASSURA
	WASHINGTON DL 20555	JUSA Ens
	020555078980 1	AND PLACE UN
	US NKL 99999	020555078981
	LE-ULY ENGRE & QUALITY ASSUR	US NKC
	SECTION CHIEF	SECTION CHICE
	EN-S 3054	ENGRU & TECH SH
	WASHINGTON DC 20555	E-5 305A
	0205550-1407	#A SHENGTON
	US NHL 99999	020555063643
-	IE-DIV FUEL FACIL MATLS SPUDS	US NRC
	DIVISION DIRECTOR	ALLY FUEL FAC
	WASHINGTON	FUEL FACTUATION
-	DC 20555	EW 359
	03065501 0.00	WASHINGTON
1	US NRC 99999	020555063664
•	IE-DIV OF REACTOR PROCEAME	US NAC
	DIVISION DIRECTOR	IE-DIV OF REACT
2		BRANCH CHIEF
	DC 20555	EN-S 300
		HASHINGTON
\mathbf{r}	020555063645 2 99999	037611070000
,	LE-DIV OF REALTON DEBORING	US NR(
	BRANCH CHIFF	IE-REACTOR TRAIN
~	REACTUR CONSTRUCTION PROGES AP	CENTER DIRECTOR
	MASHINGTON	CHATTANDOCA
	DC 20555	UT T TANUUGA
	010511005737 1 79999	010511064
	INDIAN POTHE NO.	
	RESIDENT INSPECTOR	INDIAN PUTNE 2 M
~	PO 80X 38	RESIDENT INSPECT
-	BUCHANAN NY 10511	PU BCX 38
		DUCHANAN
-	010511080042 1 99999	6766 -
-	INDIAN POINT 2 NOT	USU332003989 1
	RESIDENT INSPECTOR	IP III
	PO RUA 66	OFFICE DIRECTOR
-	BULHANAN NY 10511	EWS 406
	· · · · · · · · · · · · · · · · · · ·	
	054216005833 1 99999	Af i i i a a a a
-	KEWAIINEE MOR	12+632005367 1
	RESIDENT INSPECTOR	LA CRUSSE NOC
	ROUTE I	RESIDENT INSPECTO
-	BUX 999	RURAL ROUTE 1 BOX
	NCWAUNCE W1 54216	GENUA
	061354000003 1 99899	
	US NRC 17777	019464005110 1
	RESIDENT INCOURTON	LIMERICE NDC
	PO BUX 224	RESIDENT INSPECTO
	MARSEILLES IL AIRAA	PU BOX 47
		SANATUGA
	004578005938 1 80000	
	US NRC 77799	028031005364 1
	RESIDENT INSPECTOR	MLGUIKE Mac
	PU HOX E	RESIDENT INSPECTOR
	WISCASSET ME OAS70	PG BUX 216
		CURNEL IUS
	048640005017 1 00000	
	US NRC 97499	006376066702 1
	RESTRENT INSPECTOR	MILLSTONE TTT
	KOUTE 7	RESIDENT INSPECTION
	MEGLAND HE CAUS	PU BUX 416

L WALLEY ASSUR SIS BRANCH DC 20555 1 99999 E QUALITY ASSUR RANCE BRANCH OC 20555 1 99999 & QUALITY ASSUR SUPPORT BR-SEC A DC 20555 1 99994 ACIL MATLS SEGDS ES & NATERIALS BR DC 20555 1 99999 CTUR PROGRAMS TOR PROGRAMS BR DC 20555 L 99999 AINING CENTER TN 37411 1 99999 NPS NY 10511 1 99999 DC 20555 1 99999 CTOR BOX 276 WI 54632 99999 1 TOR PA 19464 99999 IGR. NC 28031 99999

3

à

٦.	NIANTIL	(1 ubás)/	maticato	* *,
	037650051306 1 US NEC	99444	013093007003 1 US NRC NINE MILE NPS	9999 5
,	RESIDENT INSPECTOP		RESIDENT INSPECTOR	
~	ERMIN	IN 37650	LYCUMING	NY 13043
~	013093080043 I US NRC Nine Mile Puint NPS	43959	020555004023 1 US NRC NMSS	99549
	RESIDENT INSPECTOP PU BCX 99		95855 UPFILE DIRECTOR	00 20555
-	LYCOMING	MA 13043	M 7 344 1, 40 4 0.4	
_	020555004773 1	99999	020555035625 1 US NRC	99999
	NHSS JIV FUEL CYCLE	MATL SETY	NHSS DIV DE SAFEGUA BRANCH CHIEF	DEDCRAMS RK
~	39655 WASHINGTUN	DC 20555	BBL SS WASHINGTON	DC 20555
	020555055502 1	99949	020555029836 1	99999
•	NASS DIV OF SAFEGUAL	RDS	NMSS DIV OF WASTE	HANAGEMENT
	BB1-SS WASHINGTON	DC 20555	623-SS WASHINGTON	DC 20555
	023117007002 1	99999	023117051334 1	99999
	ŬŜ NRC North Anna NPS Resident inspector Rte 2 box 78a Mineral	VA 23117	US NAC North Anna NPS Resident inspector Rt 2 Box 784 Mineral	VA 23117
		00000	020555056815 l	99999
-	US NRC NRR-UIV HUMAN FACTO	RS SAFETY	US NRC NRR-DIV HUMAN FACT DIFECTOR	URS SAFETY
	OPERATOR LICENSING	BRANCH	242 MASHINGTON	DC 20555
~	WASHINGTON	DC 20555		00000
-	020555056729 1 US NRC NRR-DIV OF ENGINEER CHIEF	99999 ING	D20555056730 US NRC NRR-DIV OF ENGINEE SLCTION LEADER-CUR CHEM ENG BR	RING ROSION ENG
-	P-234 WASHINGTON	DC 20555	P-1000 WASHINGTON	DC 20555
	020555056731 1	49949	020555056736 1	97979
U	US NRC NRR-DIV OF ENGINEER Asistant director Materials & Qualif	ING ENGINEERING	ARR-DIV OF ENGINEE SECTION LEADER-B STRUCTURAL ENG BRA	R ING
-	WASHINGTON	DC 20555	WASHINGTON	DC 20555
	020555056739 1 US NRC NRR-UIV OF ENGINEEF SECTION LEADER-SDY	99999 LING	020555056743 1 US NRC NRR-DIV OF ENGINEE Section Leader-Che Section Leader-Che	99979 Ring M Jech
	538 MACHINGTON	nr 20555	P-802 WASHINGTON	DC 20555
	020555056745	44999	020555056746 1	99999
	US NAC NAR-DIV UF ENGINEE CHIEF Materiais Fn., Brani	KENG TH	US NRC NRR-UIV OF ENGINEL ASSIST DIRECTOR COMPONENTS & STRU	ER ING LTURE ENG
۰ <u>ـ</u>	P 328 WASHINGTON	DC 20555	P-530 WASHINGTON	DC 20555
	020555056747 1 US NHC NRR-DIV OF ENGINEE	99994 K ING	020555056755 1 US NEC NER-UIN DE ENGINEL	99999 Er ing
	DIRECTOR P-202 WASHINGTUN	DC 20555	SECTION LEADER-EN EQUIP QUALIF BRAN 440 WASHINGTON	DC 20555

۰.

•

Þ	LULIP JUALIN GRANDA SBE WASHINGTON DE 200	DICTURE LHEMICA P-802
•	020555056745 1 99999 US NPC	0205550 0205550
	NAR-DIV OF ENGINEERING	NRP-DIV ASSIST
7	P 326 WASHINGTUN DC 205	5
2	020555056747 1 99999	0265550
•	NRR-UIV OF ENGINEERING DIRECTOR	US NRC NKR-UIV SECTION
,	WASHINGTON DC 2051	55 440 #ASHING
2 [°]	020555056757 1 99999 US_NHC	02055505
	NRR-UIV OF ENGINEERING Chief Chenical for Drames	NKR-UIV Chief
-	P-302 NASHINGTUN DC 2058	STRUCTUR P-1022
~	020555056779 1 99999	02055505
	NRR-DIV OF ENGINEERING	US NRC NRR-DIV
~	QUALITY ASSURANCE BRANCH	DEPUTY D
	#ASHINGTON DC 2055 020555056299 1 00000	5
^	US NRC 1 99990 NRR-DIV OF LICENSING	020555050 US_NRC
_	CHIEF LICENSING BRANCH #2	CHIEF DRI
	WASHINGTON DC 2055	MASHINGTO
0	020555056811 1 99999 US_NRC	020555056
	CHIEF ORB #2 416B	NRR-DIV C
	WASHINGTON DC 2055	430 WASHINGTO
_	020555056816 1 99999 US_NRC	020355056
	NRR-DJV DF LICENSING CHIEF DRB #5	US NRC NRR-DIV O Chife
•	WASHINGTON DC 20555	LICENSING 316 WASHINGTO
-	020555058236 1 99999 US_NKC	020555063
	CHIEF OF LICENSING	NKR-DIV O
	124 MASHINGTON OF 2055E	SYSTEMATI 516
	020555063682 1 99999	WASHINGTO
	NRR-DIV OF LICENSING	US NRC NRR-UTV OF
	DPERATING REACTORS ASSESS BR	STAN & SPE
	020555064843 t 99999	HASHINGTON
	NRR-UIV UF LICENSING	0205550664 US_NRC
	S28 MASHINGTON	CHIEF DRB 416A
		WASHINGTON
	US NAL NRK-DIV SAFETY TECHNOLOGY	0205550567 US_NRC
	TECHNICAL ASSISTANT	NRR-DIV SA DIRECTUR
	MASHINGIUN DC 20555	HASHINGTON

÷.

HEMICAL ENG BRANCH GTON DC 20555 5056746 1 99999 IV UF ENGINEERING F DIRECTOR NENTS & STRUCTURE ENG GTON DC 20555 056755 I 99999 V DF ENGINEERING IN LEADER-ENV GUAL QUALIF BRANCH GTON DC 20555 056770 1 99999 V OF ENGINEERING URAL ENG BRANCH TUN DC 20555 56724 1 99999 OF LICENSING STON DC 20555 56810 1 99999 OF LICENSING TON DC 20555 56812 1 99999 OF LICENSING I ON DC 20555 56817 1 99999 OF LICENSING NG BRANCH #1 TON . DC 20555 3023 1 99999 OF LICENSING IC EVALUATION PGM BR ON DC 20555 3900 1 99999 OF LICENSING PECIAL PROJECTS BRANCH ON DC 20555 6420 L 99999 DF LICENSING 8 #4 JN 🗌 DC 20555 5782 1 99999 SAFETY TECHNOLOGY

OF POSSS

	020555056753 1 99999	020555056782 1 99999
* }	ÚŠÍNŘC NRH-DIV SAFETY TECHNOLDGY TECHNILAL ASSISTANT	NER-CIV SAFFTY TECHNOLOGY Directur
7	278 WASHINGTON DC 20555	MASHINGTON DC 20555
	A 20555056786 1 99999	020555056787 1 99999
~	US NRC	US NRC NRR-DIV SAFETY TECHNOLOGY
	ASSISTANT DIRECTOR	CHIEF
~	278 DE DE DE DE DE DE DE	266 DC 20555
	WASHINGTON DC 20333	020555056791 1 99999
~	020555056788 1 99999 US NRC	US NRC
	NRR-DIV SAFETY TECHNOLUGY SECTION LEADER-STSS	ASSIST DIRECTOR
~	LICENSING GUIDANCE BRANCH	266 DC 20555
	WASHINGTON DC 20555	WASHINGTON DC 20000
~	020555056792 L 99999 US NRC	US NRC
	NRR-DIV SAFETY TECHNOLOGY	CHIEF
_	SAFETY PROG EVALUATION BRANCH	ZIG
C	MASHINGTON DC 20555	WASHINGTON DC 20000
_	020555056748 1 99999 US NRC	020555056771 1 37777 US NRC
$\hat{\mathbf{C}}$	NRR-DIV SYSTEMS INTEGRATION	SECTION LEADER/RAD PROTECTION
_	INSTRU & CONTROL SYS R	RADIOLOGICAL ASSESSMENT BRANCH P-712
0	WASHINGTON DC 20555	WASHINGTON UL 20555
	020555056772 1 99999	020555056790 1 99999 US NRC
?	NRR-DIV SYSTEMS INTEGRATION	NRR-DIV SYSTEMS INTEGRATION SPEC ASST-POLICY DEVELOPMENT
	EFFLUENT TREATMENT SYS BRANCH	P-802 WASHINGTON DC 20555
0	WASHINGTON DC 20555	
	<u>(20555056796 1 99999</u>	020555063630 1 99999
· •	CARTOIN SYSTEMS INTEGRATION	NRR-DIV SYSTEMS INTEGRATION
	HAUIOLOGICAL ASSESSMENT BRANCH	P-8028
	MASHINGTON DC 20555	
	020555056740 1 99999	017057056750 1 99999
	US NRC NRR-PLAN & PROG ANALYSIS STAFF	NRR-THI PROGRAM OFFICE
	CHIEF Technical support branch	THI SITE - PO BOX 311
	P-408 WASHINGTON DC 20555	HIDDLEIUMA PA LIGHT
	020555056741 2 99999	020555056749 1 99999
Ċ	US NRC NRR-THI PROGRAM OFFICE	NRR-INI PROGRAM OFFICE
	SECTION LEADER Envik Review	SAFETY REVIEW
C .	AR-SO31 WASHINGTON DC 20555	WASHINGTON DC 20555
	026555031469 1 99999	020555038064 1 99999
<u> </u>	ŬŜ NRC DEM	US NRL OCH OCH
	TECH ASST TO COM ASSELSTINE	1149-H
	WASHINGTON DE 20555	#ASHINGTON DC 20555
	020555038065 1 99999	020555038066 1 99999
· •	US NRC UCM	UCM TEC ASST TO CHAIRMAN PALLADIND
	TECH ASST TO LOW GILINSKY	1159-H LASHINGTON DE 20555
2	WASHINGTON DC 20555	HERITAL OF CASES
	029678051339 1 99999	008731005328 1 99999 US NRC
		and the second

٠,

7

 \sim

 \sim

 \bigcirc

P

3

• .

0

ŧ

		Трана араба балада 1149-на малатана	0435.1 07 20555	- есл. 4597 - Ей. Сел. (не. - 1159-н - Бахиялатам	01 24555
-		## 241 40 FUA			
	~	029678051339 1	99999	000731005328 1 US NKC	99539
		OCONEL NPS		UTSTER CREEK NPS	
ممسيعين		RT 2 HUX 610		PU BUX 445	
1	^	SENECA	SC 29678	FORKED RIVER	NJ 08731
•		020555003928 1	99999	049043005342 1	99999
	^	ŬŜ NŔĊ		US NRC PALISADES NPS	
		DEFICE DIRECTOR		RESIDENT INSPECTOR	
	~	WASHINGTON	DC 20555	COVERT	MI 49043
				0.3.566600.303.6 A	00000
	~	US NRC	44444	US NRC	44444
		PALO VERDE NPS RESIDENT INSPECTOR		PE OFFICE DIRECTOR	
		PU ROX 21324	47 86001	H 1013	00 20565
	-	PHUENIX	100C8 1A	MASHINGIUM	UC 20555
		017314005776 1	99999	044081005311 1	99999
	^	US NRC PEACH BOTTON APS		US NRL PERRY NPS	
		RESIDENT INSPECTOR		RESIDENT INSPECTOR	
	~	DELTA	PA 17314	PERRY	DH 44081
			00000	A54341005847 1	00000
	-	US NPC	44444	US NRC	*****
		PILGRIM NPS Resident inspector		POINT BEACH NPS RESIDENT INSPECTOR	
	_	PG BCX 867	NA 02360	6612 NUCLEAR ROAD	MI 54241
	С	P C HOVIN	NA 62300		41 34641
		055089005910 1	99999	061242005848 1	99999
	2	DE NEC PRAIRIE ISLAND NPS		QUAD CITIES NPS	
		RESIDENT INSPECTOR		RESIDENT INSPECTOR	ORTH
	0	WELCH	MN 55089	CORDOVA	IL 61242
		095638006533 1	00000	020555066690 1	99999
	<u>_</u>			US ARC	
	-	RESIDENT INSPECTOR		DIRECTOR	ETALUATION
		14410 TWIN CITIES PD HERALD	CA 95638	LIJO SS WASHINGTON	DC 20555
	-				
		020555063080 1 US NKC	99999	020555063085 1 US NRC	99999
		RES-DIV OF ENGINEERIN	NG TECH	RES-DIV OF ENGINEERI	NG TECH
		1130 55		HECHANICAL/STRUCTURA	L ENGR BR
		MASHINGIUN	0. 20555	WASHINGTON	DC 20555
		020555066611 1	99999	020555063568 1	99999
	U U	US NRC Res-DIV CF Engineeri	NG TECH	US NRC RES-JIV OF FACILITY	OPERATIONS
		BRANCH CHIEF	ENCE BR	CHIEF INSTRUMENTATION & CO	NTROI AR
		1130 55	DC 20555	1130 SS	00 20555
		020555063571 l US NKC	99999	U/U//5006453 [US NRC	99999
	-	RES-DIV OF FACILITY (OPERATIONS	REVER BEND NPS RESIDENT INSPECTOR	
-		HUMAN FACTORS BRANCH		PU EGX 1051	
•	'-	WASHINGTON	DC 20555	DI TRANGIDILLE	LA TUTTS
		020555003945 1	44444	029550005820 1	99999
•	7	US NKC RMZOHI		US NRC Rubinson NPS	
``		BRANCH CHIEF		RESIDENT INSPECTOR	
		MASHINGIGN	0C 20555	HUX 266 1A	,

And a set of the set o

, Mart

į

С

SAN CLEMENDS PJ 80X 4329 SAN CLEMENTE CA 920. 020555003921 94999 US NRC SECY SECHETARY H149 H149 DC 2055 011778005092 99999 US NRC 99999 US NRC 99999 US NRC 99999 SAN CLEMENT IN SPECTOR 99999 PU AUX B NY 1177 020555056728 99999 US NRC 99999 WS NRC 99999 US NRC 99999 US NRC 99999 US NRC 99999 US NRC 9000 SP STATE L LICENSEE RELATIONS AN 2016 MASHINGTON DC 2055 033457005156 9999
PU BOX 1 129 SAN CLEMENTE CA 926 020555003921 94939 US NRC SECY 94939 US NRC 96000000000000000000000000000000000000
SAN CLEMENTE CA 926. 020555003921 94999 US NRC SECY SECY SECY SECHETARY H149 MASHINGTUN DC 2055 011778005092 99999 US NRC 99999 US NRC 99999 US NRC 99999 SHUREMAN NPS 45106NT NUS NRC 99999 SHUREMAN NPS 1177 020555056728 99999 US NRC 99999 SP STATE & LICENSEE RELATIONS ANE-2016 DC 2055 033457005156 99999
020555003921 1 94999 US NRC SECY SECHETARY HI149 MASHINGTUN DC 2055 UI1778005092 1 99999 US NRC SHUREHAM NPS KESIGENT INSPECTOR PU AUX B RUCKY POINT NY 1177 020555056728 1 99999 US NRC SPIATE & LICENSEE RELATIONS ANSISTANT DIRECTUN AN-2016 MASHINGTON DC 2055 033457005156 1 99999
020555003921 1 94999 US ARC SECY SICHETARY HI149 WASHINGTUN UC 2055 011778005092 1 99999 US NHC SHUREMAM APS KESIVENT INSPECTOR PU AUX B RUCKY POINT NY 1177 020555056728 1 99999 US NHC SP STATE & LICENSEE RELATIONS AN-2016 WASHINGTON DC 2055 033457005156 1 99999
US KRC VILL VILL SEC SEC SEC SECKETAPY HI43 UC 2055 UL1778005092 99999 US NEC US NEC SHUREMAN NPS SHUREMAN NPS SHUREMAN NPS NESION NY 1177 O20555056728 1 999999 SP STATE C LICENSEE RELATIONS ASSISTANT DIRECTOR ASSISTANT DIRECTOR DC 2055 Q33457005156 1 99999
SECY SECHETARY H143 WASHINGTUN UC 2055 011778005092 99999 US NRC 99999 SHUREHAM NPS HUREHAM NPS HUREHAM NPS NT PU AUX B NY 1177 020555056728 99999 US NRC 99999 US NRC 99999 US NRC 99999 SP STATE & LICENSEE RELATIONS ASSISTANT DIRECTUR MASHINGTON DC 2055 033457005156 99999
SECHETARY H149 MASHINGTUN DC 2055 011778005092 99999 US NRC SHURHAM NPS SHURHAM NPS SECTOR PU AUX B NY 1177 020555056728 99999 US NRC 99999 US NRC 99999 US NRC 99999 SP STATE & LICENSEE RELATIONS AN-2016 MASHINGTON DC 2055 033457005156 99999
PLIAP DC 2059 011778005092 99999 US NRC SHUREHAM NPS KESIDENT INSPECTOR PU PUX RUCKY POINT O20555056728 99999 US NKC SPS STATE & LICENSEE ASSISTANT DIRECTOR AASSISTANT DIRECTOR MASHINGTON DC 033357005156 99999
UL UL 2053 011778005092 1 99999 US NRC 99999 SHURLMAN NPS KESIDENT INSPECTOR 9000000000000000000000000000000000000
011778005092 1 99999 US NRC SHUREHAM NPS KESIGENT INSPECTOR PU AUX B RUCKY POINT NY 1177 020555056728 1 99999 US NRC SP STATE & LICENSEE RELATIONS AN-2016 WASHINGTON DC 2055 033457005156 1 99999
US NRC SHURHAM NPS KESIJENT INSPECTOR PU AUX B INSPECTOR PU AUX B RICKY POINT NY 1177 O20555056728 1 99999 US NRC SP STATE & LICENSEE RELATIONS ASSISTANT DIRECTUR AN-2016 WASHINGTON DC 2055 033457005156 1 99999
SUBLEMAM NPS KESIDENT INSPECTOR PU HUX B RCCKY POINT NY 1177 O20555056728 1 99999 US NKC SP STATE & LICENSEE RELATIONS ASSISTANT DIRECTUR AN-2016 WASHINGTON DC 2055 033457005156 1 99999
RESIDENT INSPECTOR PU AUX B RECKY POINT NY 1177 O20555056728 1 99999 US NKC SP STATE & LICENSEE RELATIONS ASSISTANT DIRECTUR AR-2016 WASHINGTON DC 2055 033457005156 1 99999
PU AUX B PU AUX B RUCKY POINT NY 1177 020555056728 99999 US NC SP STATE C LICENSEE RELATIONS ASSISTANT DIRECTUR AR-2016 MASHINGTON DC 2055 033457005156 99999
020555056728 1 99999 US NKC SP STATE & LICENSEE RELATIONS ASSISTANT DIRECTUR AN-2016 WASHINGTON DC 2055 033457005156 1 99999
020555056728 1 99999 US NKC SP STATE & LICENSEE RELATIONS ASSISTANT DIRECTUN AN-2016 WASHINGTON DC 2055 033457005156 1 99999
US NEC SUIZE L 99999 SP STATE & LICENSEE RELATIONS ASSISTANT DIRECTUM AN-2016 WASHINGTON DC 2055 033457005156 1 99999
SP STATE & LICENSEE RELATIONS ASSISTANT DIRECTON AN-2016 WASHINGTON DC 2055 033457005156 1 99999
ASSISTANT DIRECTUR AR-2016 WASHINGTON DC 2055 033457005156 1 99999
AN-2016 WASHINGTON DC 2055 033457005156 1 99999
WASHINGIUN DC 2055 033457005156 1 99999
033457005156 1 99999
033457005156 1 99999
USING
ST LUCIE NPS
RESIDENT INSPECTOR
7900 S ALA
JENSEN BEACH FL 3345
029065035011 1 00000
US NRC 44999
SUMMER NPS
KESIDENT INSPECTOR
FU BUX 64 ROUTE 1
SC 2906
018455001500
NE NEL 1 99999
SUSQUEHANNA NPC
RESIDENT INSPECTOR
PU BUX 52
SHICKSHINNY PA 18655
017057007106 1 99999
US NRC
171-1 50 05510544 1455555
PD ACK THE INSPECTOR
MIDDLETOWN
PA 17057
097501006344
US NRC 99999
TROJAN NPS
RESIDENT INSPECTOR
PU HCX 0
DR 97501
00515500000
007453005898 1 99999
VEDNONT VANVEL
RESIDENT INSPECTOR
PG 80X 176
VERION VT 05453
070066005312 1 99999
US NKC
WAIERFORD NPS
PU RUX A22
KILLUNA IA TOUL
L= /0066
037831005000
US NKC 1 99999
US NKC HATTS PAR NPS

ż i. ie

)

	US C MODIAL NAS			WATERFLER APS			
	RESIDENT INSPEC	TOR		RESIDENT INSPEC	TUR		
	WAYNESBURD	GA	30830	KILLUNA		LA	70666
	037381053194	1 59	999	037831005099	1	999	99
2	WATTS BAR NPS	T		HATTS PAR NPS	TAL		
~	RT 2 EDX 300 CONING CITY	TN	37381	POULE 2 HOX 300		TN	37831
		••••					
~	098541006223	1 99	933	099352006274 US NKC	1	999	199
	RESIDENT INSPEC	TOR		RESIDENT INSPEC	TOR		
~	ELMA	wA.	98541	PICHLAND		нA	99352
	099352006325	1 99	999	001350005371	1	999	99
	US NRC WPPSS-2 NPS			US NRC YANKEE ROWE NPS			
	RESIDENT INSPEC	TGR	00353	PU BOX 28	TOR		01350
^	RICHLANU	MA	44325	MUNKUE BRIUGE		TA .	01350
~	045153054715	1 99	999	020555003884	1	999	199
	ZIMMER NPS RESIDENT INSPEC	TOR		NRA-CLINCH RIVE CARDIS L ALLEN	R BREE	DER	REAC
~	BÖX 2021 ÚS RŤ Mescuw	52 OH	45153	AR5003 WASHINGTON		DC	20555
	0 2055504453	1 00	000	020555000813	,	000	000
<u> </u>	US NKC	. ,,		US NRC ASL BP	•		
	MARKLEY L AU HUMAN FACTURS B	R		CHARLES BECHHO	DEFER	_	
. .	1130 SS WASHINGTON	DC	20555	WASHINGTON		00	20555
	020555007027	1 99	999	020555006129	1	999	999
0	ATUMIC SAFETY & CHARLES BECHHO	LICENSIN	6 80	ASLBP PETER B BLOCH			
-	ESQ WASHINGTON	DC	20555	CHAIRMAN WASHINGTON		DC	20555
·	0 30555005769	1 00	000	020555000562		000	000
		1 77	***	US NRC	1		
	ELIZABETH S BOW	IERS		LAWRENCE BRENT	IER		
	WASHINGTON	DC	20555	WASHINGTON		DC	20555
	020555003845	1 99	999	020555003839	1	999	999
	ASLEP GLENN D BRIGHT			ASLAP JOHN H BUCK			
,	EW439 WASHINGTON	DC	20555	VICE CHAIRMAN EN529			
~	0 20555 005 30 7			WASHINGTON		DC	20555
-	US NRC ASLAP	r 44	777	US NRC ASLAP	•	332	. , ,
	JOHN H BUCK			JUHN CHO COUNSEL			
	WASHINGTON	DC	20555	EN-529 WASHINGTON		DC	20555
	045153005318	1 99	1999	020555000017	1	999	999
-	ZIMMER NPS	TANSON		ASLEP RICHARD F CDIE			
	US ROUTE 52 RE	01		WASHINGTON		DC	20555
<u>.</u>	NUSCOW	OH	45153				
{	U26555007774 US NKC	1 99	1943	U20555065759 US_NRC AFUD	1	99'	999
	RICHAPD CULL			JUHA L CROOKS			

Э

. . . .

.

<u>ن</u>و r ----

•

. :

U

:

9

نگر

Þ	US AND STATES	020555505261 1 ууууу US NKC ASL8P Снамі FS I батта
า	CHAIRMAN Eu-529 MASHINGTON DC 205	ASSISTANT EXECUTIVE SECY FW439 SS WASHINGTON DC 20555
ſ	DG1C1UQU6511 1 99999 US NRC BYRCM NPS WILLIAM FOURNEY RESIDENT DURNEY	020555001226 I 99999 US RRC A SLBP JUMN H FRYE
	6448 GERMAN CHURCH ROAD BYRON IL 610	LHAIRMAN HASHINGTON DC 20555
	020555007110 1 99993 US NRC ASLAP REGINALD L GUTCHY EW-529	020555051908 1 99999 US NRC IE/PSB L, GREHER
-	WASHINGTON DC 205	55 WASHINGTON DC 20555
~	020555005341 1 99999 US NRC ASLBP MEMBERT CROSSMAN	U205550 38234 1 99999 US NRC ASLUP
~	CHAIRMAN GRUSSHAN MASHINGTON DC 205	HERBERT GROSSMAN En439 55 WASHINGTON DC 20555
	020555007028 1 99999 US NRC ATOMIC SAFETY & LICENSING BD JERRY MARBOUR	020555056993 1 99999 US NRC ASLBP JAMES HARD
	WASHINGTON 2055	EM-439 5 WASHINGTON DC 20555
	047147005349 1 99999 US NRC MARBLE HILL NPS J J MARRISON RESIDENT INSPECTOR	037411006528 1 99999 US NRC REACTOR TRAINING CENTER GLIVER HAVENS
0	3738 MARBLE HILL ROAD NABB IN 4714	SUITE 200 7 CHATTANOOGA TN 37411
0	037411066452 1 99999 US NRL REACTUR TRAINING CENTER IRENE IRENE IRENE IRENE	020555006247 1 99999 US NRC ACRS FLIPTOLO 6 LONG
D	RIC CHATTANDOGA TN 3741	HIOIG MASHINGTON DC 20555
D	020555005819 1 99999 US NKC ASLBP	- 020555009812 1 99999 US NRC ASLAP
با	W R JUHNSON Ew-529 WASHINGTON DC 2055:	W REED JOHNSON ADMINISTRATIVE JUDGE-TECH 5 E4529
-	020555026886 1 99999 US NRC AUM DSB NICRUGRAPHICS	ASLAP
J	042 MASHINGTUN DC 20555	CHRISTINE N KOHL MASHINGTON DC 20555
-	020555003890 1 99999 US NKC IP	020555007810 1 99999 US NRC
	MILLIAM S LAVINE 403A EWS WASHINGTON DC 20555	ROBERT M LAZO Ew439 MASHINGTON DC 20555
	020555007809 1 99999 US NKL ASLBP	048166006227 1 99999 US NRC FERMINPS
	EW439 WASHINGTON DC 20555	BRUCE LITTLE RESIDENT INSPECTOR 6450 W UIXIE HIGHWAY NEWPORT
•	020555007808 1 99999 US NKL A SLHP Frequencies	020555000542 1 99999 US NRC ATOMIC SAFFTY & LICENSING OD

ł.

•

1

3

HIL	L ROAD) IN	47147	SUITE 200 CHATTANOOGA TN 37411
	1	999	99	020555006247 1 99999 US_NRC
NIN IAYC CE	IG CENT RAFT CTR SU	JITE	200	ACRS ELPIDID G IGNE H1016 WASHINGTON DC 20555
÷.,	_	TN	37411	000555000913 1 00999
	1	999	199	U2U555UU98I2 I 99999 US NRC ASLAP W REED JOHNSON ADMINISTRATIVE JUDGE-TECH
		DC	20555	EW529 WASHINGTON DC 20555
5	L	999	999	020555006035 1 \$9999 US NKC
ROGF	RAPHIC	s nc	20555	ASLAP CHRISTINE N KOHL WASHINGTON DC 20555
			20333	
0	1	999	999	020555007810 1 99999 US NRC ASLAP
AVI	NE	DC	20555	ROBERT M LAZO EW439 WASHINGTON DC 20555
9	1	99	999	048166006227 1 99999 US NRC
RGE	R JR			FERMI NPS BRUCE LITTLE
		DC	20555	RESIDENT INSPECTOR 6450 W DIXIE HIGHWAY NEWPORT MI 48166
8	1	99	999	020555000542 1 99999 US NRC ARC SAFETY & LICENSING BD
EBK	E			EMMETH A LUEBKE
		DC	20555	WASAINGION 00 20000
4	1	99	999	020555030757 1 99999 US_NRC
CRE	LESS			R E MCKINLEY
		DC	20555	WASHINGTON DC 20555
1	1	9 9	1999	020555006399 1 99999 US_NRC
E MI	LLER			THOMAS S MOURE WASHINGTON DC 20555
		DC	20555	

•

**

	Marijar iskali i a	LICE			a the state of the states of			
•	EN437 NASHENGTON		ÜC	20555	WA SMENG FUN		υC	20125
Ð	020555052087 US NHC ASLAP	ł	99	9 49	020555007785 US ARC ASL9P	1	999	999
•	THOMAS S HUURE ADMINISTRATIVE EM-529	JUDGE-	-LEG	AL	DORIS M MCRAN EW439 WASHINGTON		DC	20555
-	**************************************			20333	0204 78005 605			
•	US NHL CALLAWAY NPS JUMN NEISLER RESIDENT INSPE	CTOR	49	• • •	US NKC WILLIAM T ORDER SR RESIDENT INS HOUTE 2	I SPECTO		***
7	STEEDMAN		40	65077	SENECA		sc	29678
า	020555006548 US NRC ASLBP	1	99	999	017057006382 US NRC TMI SITE	1	99	999
า	WASHINGTON		DC	20555	PU BUX 311 MIDDLETOWN		PA	17057
n	020555007109 US NRC Aslap Lawrence P QU	1 ARI ES	99	999	020555081463 US NRC SECY JEAN RATHIE	2	999	999
")	EN-529 WASHINGTON		DC	20555	H ST LOBBY WASHINGTON		DC	20555
ı	031513005792 US NHC R F RCGERS RDUTE 1	1	99	999	020555007091 US NRC Aslap Alan S Rosentha	1	999	999
Э	PO BOX 279 BAXLEY		G▲	31513	EN-529 WASHINGTON	-	DC	20555
0	020555007090 US NRC Aslap Richard S Salt	1 ZMAN	999	999	019406006409 US NRC G Sanborn 631 Park Avenue	1	999	199
Э	EW-529 WASHINGTON		DC	20555	KING OF PRUSSIA	l.	PA	19406
C	030830006492 US NRC VOGTLE NPS MILLIAN S SAND RESIDENT INSPE	1 ERS CTOR	999	999	020555029190 US NRC ACRS RICHARD SAVID 1016.MSI	1	999	999
C	WAYNE SBURD		GA	30830	WASHINGTON		oc	20555
C .	020555081604 US NRC EDU WALT_SCHWINK	1	999	999	D20555063102 US NRC ADM TIDC STEVE SCOTT	1	999	99
L.	PROGRAM MANAGE 6106 MNBB WASHINGTON	ĸ	DC	20555	LAND-1210 WASHINGTUN		DC	20555
U U	020555081107 US NRC AEDD K V SEVERIT	1	99	999	020555000023 US NRC ASLRP FREDERICK J SHO	1	999	99
-	263A EWS WASHINGTON		DC	20555	E-W 450 WASHINGTON		DC	20555
C.	020555009718 US NRC ASLBP FREDERICK SHU FM619	1 N	99	999	U20555005114 US NRC ASLBP IVAN & SMITH FSD	1	999	99
<u>.</u>	WASHINGTON		0C	20555	EA-450 WASHINGTON		00	20555
·	020555007804 US NRC ASLUP IVAN W SMITH	I	99	994	076043005008 US NRC CUMANCHE PEAK A PUBERT & TAYLOR	1 IPS	999	999
	WASHENGEDN		0C	20555	PO BOX 38	. FUR		

-	A CARLER AND A CARLE	washington	36 C2555
	020555007804 1 99959 US_NRL	076043005008 1 US ARC	99999
-	IVAN W SMITH	POBERT G TAYLOR	
•	HASHINGTON DC 201	55 PU DOX 38	TX 76043
	0444300/5350 1 00909	015077036496	99399
•	US NAC MOIE CHEEK NPS	US NRC GLENN HALTON	
	TEM VANJEL RESIDENT INSPECTOR	PU Brix 181 Shippingport	PA 15077
•	PO BOX 311 BURLENGTON KS 664	337	
	020555028206 1 99999	020555005402 1	97997
•	US NKC SP	US NRC Aslap	
	SUE WEISSBERG STATE LIAISUN OFFICER	CHRIS WILLIAMS Ew529	
•	AK5037 WASHINGTON DC 201	WASHINGTON	DC 20555
	020555000144 L 99999	020555007803 1	99999
•	ASLAP	ASLBP	
	ESQUIRE	EW439	0/ 20565
•	WASHINGTON DC 20	i55	
•	019406003570 20 99999	019406006387 1 US NRC REGION 1	99999
•	REGIONAL ADMINISTRATOR 631 PARK AVENUE	REGIONAL ADMINISTE 631 PARK AVENUE	LATOR
•	ŘÍŇG OF PRÚŠSÍŘ – PA 194	06 KING OF PRUSSIA	PA 19466
-	019406006340 1 99999	019406026175 1	99499
•	ÚŠ NŘČ ŘĚGIÚN I Karl Abraham	ÚŠ NRČ ŘEGION I Faith brennemán	
	PUBLIC AFFAIKS OFFICER 631 PARK AVE	REGIONAL STATE LI 631 PARK AVENUE	LISON OFFICER
	KING OF PRUSSIA PA 194	606 KING OF PRUSSIA	PA 19406
	019406081630 1 99999	030303006386 1	99999
•	NYU CANPBELL State Acrestants Deetrew	REGIONAL ADMINIST	ATOR
-	631 PARK AVE KING OF PRINSTA DA 19	SUITE 3100	
•			
	030303056497 1 99999 US NRC REGIUN II	030303081629 I US NRC REGION II	99999
•	REGIONAL STATE LIAISON OFFI 101 MARIETTA STREET	ER RICHARD WOODRUFF STATE AGREEMENTS D	DFFICER
	SUITE 3100 ATLANTA GA 30	IUI MARIETTA ST SU 303 ATLANTA	JITE 3100 GA 30303
•			
•	060137006385 1 99999 US NRC REGION 111	US NRC REGION III	99999
•	799 ROUSE VELT ROAD	799 KUDSEVELT RD	
•	GLEN ELLIN IL BU		12 00131
	076011006384 1 99999	076011056496 2	99999
۲	US NAC REGION IV Regional Achinistrator	US NHC REGION IV HEGIUNAL STATE LI	ALSON OFFICER
	611 RYAN PLAZA DRIVE SUITE 1000	611 RYAN PLAZA OR Suite 1000	IAF
۲	ARLINGIUN IX 76	UIT WALTMOINN	17 10011
-	076012003589 1 99999	076011007009 1 US NRC REGION IV	99999
•	REACTOR PREJECT BRANCH 1	JOHN T COLLINS	
	GII PYAN PLAZA DR STE 1000	SUITE 1000 ARLINGTON	TX 76011
•	ARLINGTUN TX 76	012	
	0.7/0100011/07 1 00000	076012081628 1	99999

 $\sim \gamma_{\rm c}$

Ŋ

•

-

ال

J

J

P	US NEC REGION III REGIONAL ADVINTSTRATUR	US NAC REGION ITT REGIONAL STATE LIATSON OFFICER
•	GLEN ELLYN IL 60137	GLEN ELLYN IL 60137
•	076011006384 1 99999 US NRE HEGI'IN IV Regilal Acministrator 611 Hyan Plaza Orive	076011056496 2 99999 US NKC REGIUN IV REGIUNAL STATE LIAISON OFFICER 611 RYAN PLAZA DRIVE
7	SUITE 1000 Arlington TX 76011	SUITE 1000 ARLINGTUN TX 76011
7	076012003589 1 99999 US NRC PEGIUN IV Reactor Pruject Aranch 1 Chiefman Dista Ing	076011007009 1 99999 US NRC REGION IV John T Collins 611 Ryan Plaza
7	AKLINGTUN TX 76012	ARLINGTON TX 76011
1	076012081627 1 99999 US NRC REGIUN IV Robert Doda Statf Agreements officer 611 kyan Plaza DR Suite 1000 Arlington tx 76012	076012081628 1 99999 US NRC PEGION IV RALPH S HEYEN STATE AGREEMENTS OFFICER 611 RYAN PLAZA DR SUITE 1000 ARI INGTON
	094596006383 1 99999	
1	US NRC REGION V REGIONAL ADVINISTRATOR 1450 MARIA LANE SUITE 210	US NRC HEGION V Regional State Liaison Officer 1450 Maria Lane Walnut Creek CA 94596
7	WALNUF LKEEK LA 94396	
0	D9459606567 1 99999 US NRC REGION V JANES MANCHETT Public Information Officer 1990 N California Blyd Mainut (Meen	094596081626 1 99999 US NRC REGION V JACK HORNUR STATE AGREEMENTS OFFICER 1450 MARIA LANE SUITE 210
3	US VETERANS ADMINISTRATIUN IRODI R L TURCOTTE HOSPITAL DIRECTOR 4101 DUN HURET AVE	084110079448 1 99999 UTAH DEPT OF HEALTH DIV OF ENVIRONMENTAL HLTH ALVIN E RICKERS DIRECTOR BOX 2500
	007081064468 1 00000	SALT LAKE CITY UT 84110
э э	VALCOR ENGINEERING CORP G J BURGU VA DIRECTOR NO 2 LAWEENCE RD Springfield NJ 07081	VALORE MCALLISTER ET AL CAML VALORE MAINLAND PROFESSIONAL PLAZA 535 TILTON ROAD PU BOX 152
•	D84663053128 1 99999 VALIEN INC G C CLARK UUALIIY ASSURANCE MNGR MUUNTAIN SPRINGS PARKWAY SPRINGVILLE UI 84663	NURINFIELD NJ 08225 OCANAD053129 1 99999 VELAN INC A K VELAN PRESIDENT 2125 WARD AVE HEMITIA
•	005495053130 1 99999 VELAN VALVE CORP	NONTREAL QUEBEC CANADA Ous602005891 1 99999 Yermunt Lattorney General
J.	PLANT MANAGEN Ave C Griswold Ind Park Williston vt 05495	109 STATE STREET-PAVILION BLDG MUNTPELIER VT 05602
,	005602006557 L 99999 VERMUNT DEPT OF HEALTH DIV OF UCCUPATIONAL & PAD MLTH RAYMOND N MCCARDLLSS ADMIN HLUG-10 HALDWIN STPEFT MUNTPELIEM VT 05602	005602011804 1 999999 VERMONT PUBLIC SERVICE RICHARD H SAUDEK CJMMISSIONER 120 STATE ST MUNTPELIER VT 05602
-	005602005895 1 99999 VERNINT PUHLIC SERVICE BOARD 120 STATE STRELT MCTTULIC VT 05602	005301005896 1 99999 VERMONT VANKEE DECOM ALLIANCE 53 FRIST STREET

Э

•	AVE C GRISHULD IND PARK WILLISTON VI 05495	LUY STATE STRET-PAVILLY OL - MUNTPELIER VI USUUR
•	005602006557 L 99999 VERMONT DEPT OF HEALTH DIV UF DICLUPATIONAL & PAD MLTH RAYMOND & MCCANDLUSS ADMIN MLUG-10 HALDWIN STREET MUNTPHILEN VT DS602	005602011804 1 99999 VERMENT PUBLIC SERVICE FICHARU M SAUUEK CIMMISSIDNER 120 STATE ST MUNTPELIER VT 05602
)	D05602005895 1 99999 VERMONT PUSLIC SEPVICE BCARD 120 STATE STREET MONTPELIER VT 05602	005301005896 1 99999 VERMONT VANKEE DECOM ALLIANCE 53 FROST STREET BRATTLEBGRO VT 05301
, , ,	005602005897 1 99999 VERMCRT VANKEE DECOM ALLIANCE 5 STATE STREET BOX 1117 MUNTPELIER VT 05602	005301006252 1 99999 VERMUNT VANREE NUC PUWER CGRP W F CORMAY PRESIDENT & CHIEF EXEC OFFICER RJ 5 BUX 169 FERRY RD BRATTLEBURD VT 05301
า า	DO1701005883 I 99999 VEKMUNT VANKEE NUC POWER CORP LGUIS MEIDER VICE PRESIDENT 1671 MIRCESTEK HD FRAMINGHAM MA 01701	DO5354064682 1 99999 VERMONT YANKEE NUC POWER CORP A F MILLIGAN ADMINISTRATIVE SUPERVISOR PO BOX 157 GOV HUNT RD VERNON VT 05354
7	005301005886 1 99999 VERMUNI VANKEE NUC POWER CGRP W P MURPHY VP G MANALER UF CORRATIONS KD 5 MUX 169 FER; : RD BRATILEBORU VI 05301	005301005618 1 99999 VERMUNT YANKEE NUC POWER CORP J B SINCLAIR LICENSING ENGINEER RU 5 BOX 169 FERRY RD BKATTLEBORO VT 05301
3	005301007004 I 99999 VERMONT VANKEE NUC POWER CORP VANKEE ATOMIC ELECTRIC PWK CO J B SINLLAIR LICENSING ENGINEER RUSSING ENGINEER RUSSING FERRY RD	044104053725 1 99999 Victoreen Inc K E Stafford Ga Manager 10101 Huddland Ave Cleveland OH 44104
с с	DRATTLEBURG VI USDU DO1950065238 1 99999 VIKEM INDUSTRIES INC VAL BUUCHARD VICE PRESIDENT & TECH NGR 42 PLEASANT ST NEWBURYPURT MA 01950	010511005686 1 99999 Village of Buchanan George Degany Mayor 236 Tate Avenue Buchanan ny 10511
ງ ງ	010520005722 1 99999 Village uf Buchanan Carl H Dalvia Village Atturney 395 Suuth Riverside Ave Groton un Hudson ny 10520	023219005670 1 99999 Virginia Attorney General Liginia Broad Street Richmond VA 23219
)	023209005671 1 99999 Virginia Cummunmealth State Curpuration Cummission James R mittine PD rok L197 Richmund VA 23209	023209011800 1 99999 VIRGINIA CURP COMMISSION JAMES R WITINE DIRECTON PU RUX 1197 PICHMOND VA 23209
•	023228081537 1 97999 Virginia Curp CF Richmond 5804 Lareside Ave Pob 9474 Richmond VA 23228	023117080970 1 99999 VIRGINIA ELECTRIC C PUWER CU NORTH ANNA NPS 1 & 2 W R CARTWRIGHT STATIUN MANAGER PU BUR 402 VA 2011
, ,	023204064594 1 99999 VINGINIA ELECTRIC & POWER CU J C DANYLG SUPVY AJMIN & REGCROS MG PU JUX 564 PICHMGND VA 23204	MINEKAL VA 23117 023261005672 99999 99999 VIRGINIA ELECTRIC & POWER CC 99400 J H FERGUSON EXECUTIVE VICE PRESIDENT-PUWER FU HCX 26666 KICHMUNU VA 23261 VA 23261

• • • •

• •

₽

A	CALESCE HERE AT 10520	
•	023209005671 1 99999 VIRGINIA CUMMONDEALTH STATE CURPHATICA COMMISSION JAMES R DITIGE PU ROX 1197 NICHMOND VA 23209	023209011800 1 99999 VIRGINIA CUMP COMMISSION JAMES R WITTINE DIRECTOR PID 4UX 1197 FICHMOND VA 23209
•	023228081537 1 90999 VIRGINIA CURP OF HICHMCHD 5809 LARESIDE AVE PUB 9474 RICHMCND VA 23228	023117080970 1 97999 VIRGINIA ELECTPIC & PUWER CU NURIM ANNA NPS I & 2 N R CARTWRIGHT Statiun Manager PU dux 402 Mineral VA 23117
•	U23204064594 1 99999 VIRGINIA ELECTRIC & POWER CU J C DANYLO SUPV4 ADMIN & RECCRDS MG PU BUX 564 RICHMUND VA 23204	023261005672 1 99999 VIRGINIA ELECTRIC & POWER CG J H FEPGUSON EXECUTIVE VICE PRESIDENT-POWER PU BGA 26666 FICHMUNO VA 23261
•	D23261004558 1 99999 VIRGINIA ELECTRIC & POWER CO Lihkary A R Flanagan Lirkarian PU buz 26666 RICHMONU VA 23261	023261080952 1 99999 VIRGINIA ELECTRIC & PCWER CC SURFY & NORTH ANNA NPS I 6 2 P G GODWIN UIRECTUR OF NUCLEAR SECURITY PC BOX 26666 RICHMOND VA 23261
•	023261006544 1 99999 VIRGINIA ELECTRIC & POWER CO R H LEASBURG VICE PRESIDENT-NUC OPERATIONS PU BOX 26666 RICHMOND VA 23261	U23261005619 1 99999 VIRGINIA ELECTRIC & POWER CO NUCLEAR OPERATIONS R H LEASBURG VICE PRESIDENT PU BOX 26666 RICHMOND VA 23261
•	023204080945 1 99999 VIRGINIA ELECTRIC & POWEF CO NURTH ANNA 3 NPS È LIFRAGE PROJECT COURDINATOR PO BUX 564 Richmunu VA 23204	023117080946 1 99999 VIRGINIA ELECTRIC 6 POWER CO NURTH ANNA 3 NPS P G PERRY SITE MANAGER PU BUX 38 MINERAL VA 23117
•	023883079289 L 99999 VINGINIA ELECTRIC & POWER CO SURRY NPS L & 2 J L WILSON STATION MANAGER PU POX 315 VA 336	020006006552 1 99999 VULPE BUSKEY G LYONS D B MACGUINEAS 918 16TH STREET NW WASHINGTON DC 20006
-	060521053721 1 99999	020874066211 1 99999
•	UA MANAUER 1315 WEST 22ND ST	NUCLEAR INDUSTRY CONSULTANTS
•	DAKBROUK IL 60521	GERMANTOWN MD 20874
•	DJ2173066783 1 99999 M R GRALE & CU DRGANIC CHEMICALS DIVISION JOHN J EAGAN RESEARCH SPECIALIST SESUAVULN AVE	061820001585 1 99999 MAALER EVANS & GORDON THOMAS J GORDON ESQUIRE 2503 S NEIL CHAMPAIGN II 61820
•	LEXINGTION MA 02173	
•	046224006240 1 99999 Marash Valley Pimer Assue E P Martin General Manager PO BUX 24700 Indianapulis In 46224	010007057894 1 99999 Wall Street Juurnal Jumn Emschwiller 22 CCRTLANDT STRFET New Yurk Ny 10007
•	007036053131 1 99999 WALWURTH LU ALCVCU PLANT JCHN L MARLEY	019481053133 1 99999 WALWURTH CU CORPCRATE ENGINEERING # U HUNT
U U	NULLEAK ENGN CUNSULTANT 1400 w Elizabeth ave Linuen nj 07036	QUALITY ASSURANCE MANAGER PU BUX 873 Valley Furge pa 19481

6

8

6

Э

;

	المتحافظ والمتحافظ والمتحاف والم	الاي الايران يرتد الفرار الأنفارين الفتر فتنو لانقتافه
	RADIATIUN LUNTHEL PHOGRAM	- I N SINUNG - NEAD-EADEATION CONTROL SECTION
	SUPERVISOR	OLYMPIA WA 98504
n		
	ULTHPIA AN 75504	
_	098504004311 1 99999	098504005964 1 99999
0	RADIATION CUNTRUE SEC	NICHCLAS DIFWIS
	TERRY STRING	CHAIRMAN
•	MEAD Atabustrial Park LF+13	ULYMPIA WA 94504
	OLYMPIA #A 98504	
	098504080853 1 99999	099352006466 1 99999
•	WASHINGTUN ENERGY FACIL SITE	HASHINGTON PUB POWER SUPL SYS
	NICHOLAS D LEWIS	DEPUTY DIRECTOR-SETY & SECURTY
	CHAIRMAN	PO 80x 968-MD 650
•	OLYMPIA HA 98504	RICHLAND HA 99302
	000353091539 1 00000	008641004224 1 00000
	WASHINGTON PUB POWER SUPL SYS	WASHINGTON PUB POWER SUPL SYS
-	C S CARLISLE	KENNETH W COOK
	PU BUX 968	ELMA WA 98541
•	RICHLAND WA 99352	
-		
_	098541079424 1 99999	099352006469 1 999999
•	D E DUBSON	G E DOUPE
	PRUJECT MANAGER-WNP-3/5	3000 GFORGE WASHINGTON WAY
•	ELMA WA 98541	RICHLANU WA 99352
•		
	099352006225 1 99999	099352008688 1 99999
•	WASHINGTON PUB POWER SUPL SYS	HASHINGTON PUB POWER SUPL SYS
-	MANAGING DIRECTOR	MANAGING DIRECTOR
	3000 GEORGE WASHINGTON WAY	PO 80X 968
•	PU DUA 968 RICHLAND WA 99352	KICHLAND WA 99352
		0//01F3070433
	WASHINGTON PUB POWER SUPL SYS	WASHINGTON PUB POWER SUPL SYS
•	G L FONES	R B GLASSCOCK
	SK PROJECT VA ENGINEEK Pri Hux 1923	PO BOX 968
•	ELMA WA 98541	RICHLAND WA 99352
•		
-	098541079422 1 99999	098541064621 1 99999
•	WASHINGTON PUB POWER SUPL SYS	R D MADDEN
	PRUGRAM DIRECTOR - WNP 3/5	QUALITY ASSURANCE ENGINEER
	PU BUX 1223 FLMA VA 98541	PU BUX 1223 FIMA MA 98541
•		
	099352081530 1 00000	099352006468 1 99999
•	MASHINGTON PUB POWER SUPL SYS	HASHINGTON PUB PUWER SUPL SYS
-	R G MATLLLR Program Director-WNP-2	KUUEK NELSUN LICENSING MANAGER
_	PO BOX 968	PU BUX 968
•	NECHLAND WA 99352	RICHLAND WA 99352
	00000	000353003530 1 00000
	WASHINGTUN PUB POWER SUPL SYS	WASHINGTON PUB POWER SUPL SYS
-	RICHARD O JUIGLEY	R W ROUT
	SOUD GEURGE WASHINGTON WAY	PO 80X 968
•	RILHLAND WA 99352	RICHLAND WA 99352
-		
-	099352057188 1 99999 WASHINGTON PUB POWER CUPI SYS	099352064626 1 99999 WASHINGTON DUB DOWED CUDA CHE
•	LICENSING ENGINEERING	C L WHITCHNE
	G C SURENSEN	SENTUR ENG REC ANALYST
•	PO BOX 908	RICHLANU WASHINGTUN WAY
-	RICHLAND WA 99352	
	099219053134 1 99999	025305005701 1 99999
•	WEEK BRUS METAL PRODUCTS INC.	HEST VIRGINIA DEPT OF HEALTH
	SPUKANE WA 99219	STATE DIRECTOR OF HEALTH
		REAG & ROAM 20A

¹ς,

. ۰ ۲ • • • Ψ.

•

S S ANDES GUALITY ASSUMANCE ENGINEER PU BUX 1223 BLMA WA 98541	UG9352UDB468 1 99999 Masmington Pub Pumer Supl Sys Vicia Melson Licensing Manager Pu Bux 308 Michlanu	049352081529 1 Mashington pub Pomer Supl Sys Ry Bran Directur-WNP-1/4 Pu Bux 468 Richland WA 94352	099352004626 HOWER SUPL SYS Washington Pub Power Supl Sys C L Whitcomb C L Whitcomb Statur Ence Rec Analysi Stou George Washington Way Xichlanu	025305005701 l 99999 Mest Virgimia Dept OF Mealim Llark Mansbarger State Directur of Mealtm Blog 3 Ruum 206 WV 25305 Charlesfon	025305004355 1 0FC OF EMER SVC MEST VIRGINIA OFC OF EMER SVC DIRECTOR DIRECTOR STATE CAPITOL COMPLEX KUOM E880 CHARLESTON MV 25305 026505081876 10 MV 25305 MC 1RGINIA UNIVERSITY MOUNTAINCLAIR SUM MV 20506	073005005182 ILEC 99999 Western Farmers Elec Cuop Inc Maynard Human Ceneral Manger Pu Box 429 Anadarko	007662079039 I #ESTERN STRESS INC 374 NURTH MIDLAND AVE 2 AUDLE BRUUK	015230006610 1 99999 MESTINGHUUSE ELECTRIC CORP NUELER ENERGY SYSTEMS LIBKARY MANAGER PO BUX 355 PA 15730 P1TTSBURGH	032576053722 1 WESTINGMOUSE ELECTRIC CURP NUCLERCUMPONENTS DIVISION C M ADVINS PRUJICT MANAGER BRCP PRUJICT MANAGER BRCP PRUJICT 12831 FL 32576	ULS230065223 1 99999 WESTINNHOUSE ELECTRIC CURP W BANFUNG PU BUX 355 PI BUX 355 PI TTSBUXGM
PRUGIAN ULHELICH - WHP 3/5 PRUGIAN ULHELICH - WHP 3/5 PO EUX 1223 ELMA - 78541	044352081530 1 94944 Masmingtun Pun Punek Supi Sys C Maillek Prugem Uike(ICR-WWH-2 PC Bur 900 Richland	099352006407 1 99999 Mashingtun Pub Pumér Supl Sys Richaru U Guigley Esuur Surge Masmington Way Richlanu	099352351498 LOWER 99999 Masmington Pur Power Supl Sys Litensing Engine Eping Solo Selinge Washington Way Boons Job Ritenang Washington Way 99352 Ritenang	099219053134 1 Méla Brus Métal Pruducts Inc Po Bux 19099 Spukanf 99219	O25305006545 I F 99999 MEST VINGINIA OLPT OF LABOR LAMRISSIC BAKKER COMMISSIC BAKKER SIMTE LAPITOL COMPLEX-BLDG 6 STATE LATIOL COMPLEX-BLDG 6 ROUM ASI VIRGINIA OFFICE OF GUY D25305005704 I OFFICE OF GUY VIRGINIA OFFICE OF GUY CHARLESTUN	OI9103028069 I 99999 MESTEL SERVICES INC APPLIEU SCIENCES I AN SAKUNI I AN WORTH 20TH STREET A 19103 PHILADELPHIA	094124053135 1 99999 MESTERN PIPING L ENGINEERING R M MAYIACK UA MALAGEMITE AVE CA 94124 SAN FYANCISO	014845053148 1 99999 Westinghouse electric corp indust courtebuild quality assukance manager Westinghouse circle ny 14845 Monsemeaus	015224001629 L 99999 MESTINGHOUSE ELECTRIC CUMP MNI LIGAAKY Parkway CFNTER PA 15220 P111594464	D6U099UR1500 1 99979 Mestinumusf fiftetaector Instruct & imairing Reactor C mairing

(

E.

b

Ì

•	014845053148 1 99999 WESTINGHTUSE ELECTRIC CORP INDUST & GUYT TUME DIV QUALITY ASSURANCL MANAGER MESTINGHTUSE CIFCLF	015 WES NUC LTH
•	MURSEHEADS NY 14845	PIT
•	015270001629 1 99999 MESTINGHUDSE FLECTRIC CGRP WNI LIBRARY 2 PAKKWAY CFNTER	032 NES NUC
٠	PETTSHURGH PA 15220	PRU PU PEN
•	060099081500 1 99999 WESTINGHOUSE ELECTRIC CORP INSTRUCT & TRAINING REACTOR C HACH Managir	015 WES W H PU
•	SOS SHILOH BLVD ZICN IL 60099	P110
٠	015230038027 1 99999 Westinghouse Electric Corp Nuclear Safety Dept R D Hurch Licensing Fingineer	015 HESI POWE H A
•	PO BGX 355 PITTSBURGH PA 15230	PITI
•	014240053138 1 99999 Mestinghuuse Electric Comp Large Motor Divisiun T M Commay	0852 WEST 1.0MP
•	MANAGER QUALITY ASSURANCE 4454 GENESEE STREFT BUFFALU	PO D
•	015230006311 1 99939 WESTINGHOUSE ELECTRIC CURP BOB FAAS PO BOY 345	0152 WEST NUCL
•	PITTSRURGH PA 15230	₩ ₩ ₽0 ¶ ₽1TT
•	015230001222 1 99999 MESTINGHUUSE LLECTRIC CURP MC GANGLOFF	0148 WEST ELEC
•	PLTTSBURGH PA 15230	WALL WEST 111 HURS
•	015230006498 1 99999 Mestinghouse electric corp Power systems J_C Hoebel	U142 WEST PUWE R G
•	PO 80X 355 PITTSBURGH PA 15230	GENE PO B BUFF
•	014240053723 1 94999 Westinghouse Electric Cump Medium Mutur & Gearing Div D L Houver Wenemal Manageh	0378 WEST ANGU PU B
•	PG BUX 225 BUFFALD NY 14240	
•	015230305522 L 99999 WESTINGHUUSE ELECTRIC CORP ATCMIC POWER DISTRIMUTION WILLIAM KORTIEN PUST OFFICE HOX 355 PITTSBURGM PA 15230	0152 WEST W LU PU BU PLTTS
-	015643053142 1 94999	(11)44
٠	WESTINGHNÜSE ELECTRIC CÖRP ARD-AALIZ MILL SITE A G MARTINI QUALITY ASSURANCE MANAGER	HESTI PENSI T D M
	· · · · ·	

LA HA JALEN 1485 VUSEMITE AVE SAN FRANCISU

.

0

î

CA 94124

0152300066 WESTINGHUU	IO I ISE ELECTRI		1999 JKP
- NUCLEAN EN - LIBRARY MA - PO BLI 355	NAGER	45	
PITTSBURGH	t	PA	15230
0325760537	22 1	. 99	9999
NUCLEAR LU	MPONENTS D	ivis	IUN
PRUJECT NA PU BOX 128	NAGER BRCP		
PENSACOLA		FL	32576
WESTINGHOU WESTINGHOU	SE ELECTRI	c 20	1999 IRP
PU BOX 355 PETTSBURGH		PA	15230
0152300816 WESTINGHOU	12 1 SE ELECTRI	c 29	999 8P
POWER SYST H A CLAWSO	EMS N		
PI TTSBURGH	8	PA	15230
0852820331	40 1	99	999
RESTINGHOU	SE ELÉCTRI INSTRUMEN	Ç ČÓ Tati	ŘPÍ On div
PU BUX 220	GER 05 1441 w /	AĻĄM	EDA DR
ICHPC		AL.	85282
01523000670 WESTINGHUU	09 1 SE ELECTRIC	99 101	999 KP
NULLEA? SEI W H FURFAR	RVICE DIVI:	SION	
PITTSBURGH	5	PA	15230
0148450015	35 1	99	999
ELECTRONIC	COMPUNENT		₹ <i>₽</i> V
WESTINGHOUS	SECIRCLE		
HÜRSEHEADS		NY	14845
01424005314 WESTINGHOUS	E ELECTRIC	994 CO	999 28
R G HULDBRU General Man	IRUNICS & E JOK MAGER	DKTAF	: 545
PO BOX 225 BUFFALO		NY	14740
03783008113	1 . 1	999	99
ANGUS KIMM PU BOX M	INS	(0)	(P
OAK RIDGE		TN	37830
01523000127	а I	007	00
HESTINGHOUS	É ELÉCTRIC	CUR	(h) 12.A
PU BUX 355 PITTSBURCH		PA	15230
03259605314 HESTINGHOUS	4 I E ELECTRIC	999 CDR	99 P

NAJULE BRUK (1000) AVI UT662

VESTINGHOUSE ELECTRIC CÓŘP Pensacula divisiun t d miller Quality Așșurance Manager

۲

•

•

•

•

•

•

•

•

•

•

U

Ú

U

INAD FOR FILE AND SAN + CANCISC	CA 94124	5 b	1 4 y (18 m - 1
014845053148 1 WESTINGHUNSE FELCTRI INDUST 5 GUVT TUME M ULALITV ASSUBANCE M WESTINGHUNSE CIRCLE MURSEMEADS	99799 IC COFP JIV ANAGEP NY 14845	UI5230006610 1 HESTINGHUUSE ELEET NUCLEAR ENERGY SYSTI LIDHARY MANAGER PG 30X 355 PITTSBURGH	99999 16 CORP EMS PA 15730
015220001629 I WESTINGHGUSE ELECTKI WNI LIGKAKY 5 PARKWAY CENTER PIITSBURGH	99999 IC CURP PA 15220	G32576053722 1 WESTINGHQUSE ELECTRI NUCLEAN LUMPONENTS (C R AUXINS PRUJECT MANAGER BRCF PU EUX 12031 PENSACULA	99999 IC CUPP DIVISIUN P FL 32576
060099081500 I WESTINGHOUSE ELECTRI INSTAUCT & TRAINING C BALH Manager 505 Shiloh Blvo Zila	99999 C CORP REACTOR	015230065223 I Méstinghouse Electri W H Banford Pu Box 355 Pittsburgh	99999 IC CORP PA 15230
015230038027 1 WESTINGHUUSE ELECTRI NUCLEAN SAFETY DEPT R D BURCH LICENSING ENGINEER PO BUX 355 PITYSRUHGM	99999 C CORP	015230081612 1 #ESTINGHUUSE ELECTRI POWER SYSTEMS H A CLAWSON PO BOX 2728 PITTSBUNGH	99999 C COHP PA 15230
014240053138 MESTINGHUUSE ELECTRI LARGE MOTOR DIVISION T M CONTAY MANAGEH QUALITY ASSU 4454 GENESEE SIREFI BUFFALU	99999 C CORP RANCE NY 14240	085282053140 1 MESTINGHOUSE ELECTRI GUMPUTER & Instrumen NANCY EUUGGER PU BOX 22005 1441 M TEMPE	99999 C CORP ITATION DIV ALAMEDA DR AZ 85282
015230006311 1 MESTINGHOUSE ELECTRIC BOB FAAS PO BOX 255 PITTSRURGH	99999 C CURP PA 15230	015230006709 1 WESTINGHOUSE ELECTRI NUCLEAR SERVICE DIVI M H FURFAR PO 90X 2728 PITTSBURGH	99999 C CURP Ston PA 15230
015230001222 1 MESTINGHUUSE ELECTRIG C GANGLOFF 20 BOX 355 PITTSBURGH	99999 C CURP PA 15230	014845001535 1 MESTINGHOUSE ELECTRI ELECTRONIC COMPONENT WESTINGHOUSE CIRCLE 111 MURSEHEADS	99999 C CURP S DIV
015230006498 1 LSTINGHOUSE ELECTRIC OWER SYSTEMS I C HCEBEL O BOX 355 FITTSOURGH	99999 Curp FA 15230	UI4240053147 1 WESTINGHOUSE ELECTRIC PUMER ELECTRONICS 6 (R G HÜLDBRUOK GENERAL MANAGER PO BOX 225 BUFFALO	99999 C CORP URIVE SYS
14240053723 1 ESTINGHOUSE ELECTRIC EDIUM MUTUR & GEARIN I HOUVER ENENAL MANAGEN G BUX 225 UFFALO	99999 . Cukp IG DIV NY 14240	037630081131 1 WESTINGHOUSE ELECTRIC ANGUS KIMMINS PO BOX W UAR RIDGE	99999 CORP IN 37830
15230006522 1 ESTINGHOUSE ELECTRIC TOMIC POWER DISTRIBU ILLIAM KORTIEN UST UFFICE BUX 355 ITTSDUKGH	97999 CURP TION PA 15230	015230001379 1 Mestinghouse Electrig Meuce Po 80x 355 Pitisbukgh	99999 Curp PA 15230
15663053142 1 ESTINGHOUSE ELECTRIC RU-MALTZ MILL SITE G MARTINI UALITY ASSURANCE MAN D HOX 156 ADISON	99999 CORP AGER PA 15663	032596053144 1 MESTINGHOUSE ELECTRIC PENSACULA UIVISIUN TO MILLER VUALITY ASSURANCE MAN PU BOX 1313 PENSACULA	99999 CORP AGER

C

1-	NUCLEAR SAFETY JEHT	NEST HUMPHILL LERENTLE LENT POWER SYSTEMS
•	R D HURCH LICENSING ENGINEEN PO BUX 355 PITISBURGH PA 15230	H A CLANSON PO 50X 2728 PITTSBURGH PA 15230
•	014240053138 L 99999 MESTINGHUSE ELECTRIC LGAP LARGE MOIOH DIVISION T.M. CONHAY	085242053140 I 54999 #LSTINGHUUSE ELECTRIC LOKP (.THPUTER C INSTRUMENTATION DIV NANCY DUGGER
•	MANAGLA QUALITY ASSURANCE 4454 GENESEE SIREFI BUFFALTI NY 14240	PU 30X 22005 1441 # ALAMEDA DR TEMPE AZ 85282
•	015230006311 1 99999 WESTINGHOUSE FLECTRIC CURP BOB FAAS PO BOK 355 PITTSBURGH PA 15230	015230006709 1 99999 WESTINGHOUSE ELECTRIC CLKP NUCLEAR SERVICE DIVISION M H FURFARI PO ROX 2728
•		PITTŠBURGH PA 15230
•	015230001222 1 99999 WESTINGHUUSE LLECTRIC CGRP W C GANGLÜFF PO BOX 355 PITTSBURGH PA 15230	014845001535 1 99999 WESTINGHOUSE ELECTRIC CORP ELECTRONIC COMPONENTS DIV WALLACE GILLIES WESTINGHOUSE CIRCLE
	015230006498 1 99999	HORSEHEADS NY 14845 014240053147 1 99999
	WESTINGHDUSE ELECTRIC CURP Power Systems JC Hoebel Po Bux 355 Pittsrurgh PA 15230	WÉSTINGHOÙSÉ ELËCTRIC CORP Puwer Electronics & Drive Sys R G Holdbrook General Manager Po Roi 225
1		BUFFÂLO NY 14240
•	MESTINGHOUSE ELECTRIC CURP MEDIUM MUTUR & GEARING DIV U L MOUVER	WESTINGHOUSE ELECTRIC CORP ANGUS KIMMINS PO BOX W
•	BURTAL MANAGEH PO BUX 225 BUFFALD NY 14240	OAK RIDGE TN 37830
•	015230006522 1 99999 MESTINGHOUSE ELECTRIC CORP ATCMIC POWER DISTRIBUTION WILLIAM KORTIEN	015230001379 1 99999 Westinghouse Electric Curp W Luce Po Box 355
•	PUST OFFICE BOX 355 PITTSBURGH PA 15230	PÎTÎSBURGÎN PA 15230
•	015663053142 1 99999 WESTINGHOUSE ELECTRIC CORP Aru-Maltz Mill Site A.G. Martinil	032596053144 1 99999 HESTINGHOUSE ELECTRIC CORP PENSACULA DIVISION T.D.MILLER
•	UTALITY ASSUPANCE MANAGER PO BOX 158 Madisun pa 15663	QUALITY ASSURANCE MANAGER PU BOX 1313 PENSACOLA FL 32596
•	020014056984 1 99999 WESTINGHCUSE ELECTRIC CUPP LICENSING OPERATIONS OFC ED T MURPHY MANAGER 4901 6410404 T AMERICA	015230079598 1 99999 WESTINGHOUSE ELECTRIC CURP NES LICENSE ADMINISTRATION A J NARDI MANAGER
~	BETHESDA MD 20014	BUX 355 PITTSBURGH PA 15230
6	015024053139 1 99999 WESTINGHOUSE ELECTRIC CURP ELECTRIC HELMANICAL DIVISION C L LWEN	015230006516 1 99999 WESTINGHUUSE ELECTRIC CORP A T PARKER PU_BOX 355
U	CHESHICK AVENUE CHESHICK AVENUE CHESHICK PA 15024	PETTSBURGH P& 15230
U	021030081456 1 99999 WESTINGHOUSE ELECTRIC CCHP NUCLEAR INSTRUMENTATION & CUNT MEDIAR INSTRUMENTATION & CUNT	015230081407 1 99999 WESTINGHOUSE ELECTRIC COMP WATER REACTOR DIV E.P. RAME
	GENERAL MANAGER 1111 SCHILLING RD MS 7422 HUNT VALLEY MD 21030	MANAGEK-NUCLEAR POWER DEPT PC HUX 355 Pittsrurgh pa 15230
	OIS230366433 1 99999 Mestinghuse Electric Curp Monkceville Nuclear Center D H Laitng	U15230064694 1 99999 WESTINGHOUSE ELECTRIC CORP INFO & RECORDS SYSTEMS

•

^

A

.

•

•

•

Ô

0

0

•

•

4

Ø

U

Ø

U

Q

Ú

C
MANAULA UUALIIY AUSUAANUE 4454 GUNESEE STREFT 80FFALU NY 14240	NA 46 4 000000 00000000000000000000000000
015230006311 1 99949 WESTINGHOUSE FLECTRIC CUPP BOE FAAS PO BOX 355 PITTSHUKUM PA 1523C	UIS230006709 1 99999 HESTINGHUUSE ELECTRIC CLAP NUCLEAP SERVICE DIVISION M H FURFARI PO 40x 2728 PITTSBURGH PA 15230
015230301222 1 99999 MESTINUHUUSE ELECTAIC CUPP C GANGLUFF PC BCX 355 PITTSBURGH PA 15230	Ol4845DOl535 1 99999 WESTINGHOUSE ELECTRIC CURP ELECTRUNIC COMPUNENTS DIV WALLACE GILLIES WESTINGHOUSE CIRCLE 111 WURSENSADT
015230006498 1 99399 MESTINGHTUSE ELECTRIC CORP POWER SYSTEMS J C MCEBEL PO BOX 355 PITTSRURGH PA 15230	U14240053147 1 99999 WESTINGHOUSE ELECTRIC CORP PUWER ELECTRONICS & DRIVE SYS R G HOLDBRUOK GENERAL MANAGER PO BCX 225 BUFFALD NY 14240
O 14240053723 1 94999 MESTINGHOUSE ELECTRIC CUMP MEDIUM MUTUR & GEARING DIV U L HUDVER GENEKAL MANAGER PO BUX 225 BUFFALO NY 14240	037830081131 1 99999 MESTINGHOUSE ELECTRIC CORP ANGUS KIMMINS PO BOX W OAK RIDGE TN 37830
O152300J6522 1 99999 WESTINGHOUSE ELECTRIC CORP ATCHIC POWER DISTRIBUTION WILLIAM KORTIEN POST OFFICE BOX 355 PITTSBURGM PA 15230	015230001379 1 99999 WESTINGHOUSE ELECTRIC CURP W LUCE PO BOX 355 PITTSBURGH PA 15230
 Ø15663053142 1 99€99 WESTINGHDUSE ELECTRIC CORP ARU-HALTZ MILL SITE A G MARTINI QUALITY ASSURANCE MANAGER PO BOX 158 MAUISUN PA 15663 	032596053144 1 99999 WESTINGHOUSE ELECTRIC CORP PENSACOLA UIVISION TO MILLER QUALITY ASSURANCE MANAGER PU BOX 1313 FL 32596
O20014056984 I 99999 WESTINGHUUSE ELECTRIC CURP LICENSING OPERATIONS OFC EUI MURPHY MANAGEK MONAGEK BETHESDA ND 20014	O15230079598 1 99999 WESTINGHOUSE ELECTRIC CURP NES LICENSE ADMINISTRATION A J NARDI MANAGER BOX 355 PITTSBURGH PA 15230
015024053139 1 99999 WESTINGHOUSE ELECTRIC CORP ELECTRU MECHANICAL DIVISION C E LWEN PRUDUCT ASSURANCE MANAGER CHESWICK AVENUE CHESWICK PA 15024	015230006516 1 99999 WESTINGHOUSE ELECTRIC CORP A T PARKER PU BOX 355 PITTSBURGH PA 15230
©21030081454 1 99999 MESTINGHOUSE ELECTRIC CORP NUCLEAR INSTRUMENTATION & CUNT M PATALON GENERAL MANAGER 1111 SCHILLING RD MS 7422 HUNT VALLEY MO 21030	015230081407 1 99999 WESTINGHOUSE ELECTRIC COMP WATER REACTOR DIV E P RAME Managek-Nuclear Power Deft PG BDX 355 Pittsburgh DA 15230
015230066433 1 99999 MESTINGHUUSE ELECTRIC COMP MONRGEVILLE NUCLEAR CENTER D H KAMLINS PG 60x 355 BAY 417A PITTSHUKGH PA 15230	015230064694 1 99999 WESTINGHOUSE ELECTRIC CORP INFO & RECORDS SYSTEMS L M RICHMAN MANAGER PU BCX 2728 PUTTSBURGH DA 15330
O15146001625 1 99999 NESTINGHOUSE ELECTRIC CURP PLANT APPARATUS JIVISIGN HARVY H RUSLHBLUM GUALITY ENGINEERING MGR BUX 325	015230G64758 1 99999 WESTINGHOUSE ELECTRIC CORP NUCLEAR ENERGY UE RUSSELL MGR RECORDS & FILES OP PU BOX 355 & FILES OP

3

8

-

.

۲

 \mathbf{a}

 \frown

•

8

	015/24054149 1 94499	015230336516 1 00020
•	WESTINGHOUSE FLECTRIC CUPP	WESTENGHOUSE ELECTRIC CORP
-	C L Latia	Po_uox_355
	PRUDULT ASSUPANCE MAMAGER Cheswick Avenue	PETTSEINGH PA 15230
9	CHESWICK PA 15024	
Þ	D21030081454 1 99499 WESTINGHOUSE ELECTHIL CLHP NUCLEAN INSTRUMENTATION & CONT	015230001407 1 99999 MESTINGHOUSE ELECTRIC COMP WATER REACTON DIV
	W PATALON General Manager	E P RAHE Managen-Nucleak Power offer
D	HINT VALLEY NO RD MS 7422	PC 80X 355 PL 15X 355 PL 15X 355
•	MESTINGHOUSE ELECTHIC COMP MONNDEVILLE NUCLEAR CENTER D.H.NANLINS	HESTINGHOUSE ELECTRIC CORP INFO & RECORDS SYSTEMS L M RICHMAN
•	PU BUX 355 BAY_417A	MANAGER PO BCX 2728
•	PITTSHURGH PA 15230	PÍTÍŠBURGH PA 15230
Ð	D15146001625 1 99999 MESTINGHOUSE ELECTRIC CORP Plant Apparatus Division Harvey H Rusenblum	015230064758 1 99999 WESTINGHUUSE ELECTRIC CORP MUCLEAR ENERGY U E AUSSELL
•	QUALITY ENGINEEKING MGR Bux 425	MGR RECORDS & FILES OP Po box 355
	MÜNKDÉVILLE PA 15146	PITTSBURGH PA 15230
1	015230081499 1 99999 WESTINGHUUSE ELECTRIC CUHP LICENSING SAFEGUARDS & SAFETY A T SABU	033065055288 1 99999 #LSTINGHOUSE ELECTRIC LORP Relay Instrument Div
•	DIRECTUR PO BOX 355	MANAGER JUALITY ASSURANCE
	PITTSAURGH PA 15230	CURAL SPRINGS FL 33065
	015238053145 1 99999	015112053150 1 99999
1	WESTINGHOUSE ELECTRIC CORP INDUSTRY ELECTRONICS DIVISION T E SMITH	WESTINGHUUSE ELECTRIC CORP Puwer Sys Swichgear DIV H 1 Stahr
)	ZOO BETA DRIVE PITTSBUNGH PA 15238	TOO BRADDUCK AVE E PITTSDURGH PA 15112
	015230001417 1 99999	015230005491 1 99999
Ð	POWER SYSTEMS DIVISION	WESTINGMOUSE ELECTRIC CORP. W WRIGHT
	F J THOGUOD .Po BCX 355	PROJECT MANAGER Post Office Box 355
	PITTSBURGH PA 15230	PETTSNURGH PA 15230
•	099352053149 1 99999 MESTINGHOUSE HANFORD CC	U99352077527 1 79999 HESTINGHOUSE HANFORD CC
	RILHLAND WA 99352	MGR-NUCLEAR WA PROGRAM OFC
		PC BCX 1970 W/8-110
•		RECHLAND WA 49352
J	099352055108 1 99999 Mestinumuust manford Cu Rubent L Meadur Bur 1970	099352039954 1 99999 HESTINGHOUSE HANFORD CO W/D 27 4 A R SCHADE
•	W-R-128 #16HLAND	PU_DUX_1970
3		RICHEAND NA 99352
3	045153026371 1 99999 William Zimmer Nuclear The Johnson	004578006480 1 99999 WISCASSET FIRST SELECTMAN MUNITERAL ALOS - 15 COMPA
	US RUUTE 52 PO JOX 201	WESCASSEE ME 04574
	004578007115 1 99999	053702020123 1 99999
.)	WASCASSET PUBLIC LIARARY ASSOCIATION	WISCONSIN DEPT OF JUSTICE Patrick w Walsh
	HIGH STREET WISCASSET ME 04579	ASSISTANT ATTORNEY GENERAL
•	····	MAULSON WE 53702
•	053707080976 1 99999 NTSCHASTR OTENN SAFETY 5 REDG	053702064176 1 99999 MISCONSIN DIV CH STATE ENEAGY

ൗ

<u>,</u>) 1

	· · · · · · · · · · · · · · · · · · ·	enter a serie de la companya de la c
•	015024053139 1 94499 WESTIVGHOUSE FLECTRIC COPP ELECTRY MECHANICAL DIVISION	U15230036516 1 99499 MESTINGHUUSE ELECTRIC CORP A.T. PANKEN
Þ	G E GHEN PRUJULT ASSURANCE MANAGEN GHESNICK AVENUE CHESNICK PA 15024	ри вих 355 Регтервикон ра 15230
Þ	D21030031454 1 99499 MESTINGHUUSE ELECTRIC CURP NULLEAR INSTRUMENTATION & CLNT	015730081407 1 99999 WESTINGHOUSE ELECTRIC CCHP WATER REACTOR DIV
Þ	M PATALIN GENERAL MANAGER 1111 SCHILLING RD MS 7422 HUNT VALLEY MD 21030	E P RAPE Manager-Nuclear Power Dept PD 50x 355 Pittsrurgh Pa 15230
)	015230066433 I 99999 Mestingmouse Electric Comp Monroeville Nuclear Center	U15230064694 1 99999 WESTINGHUUSE ELECTRIC CORP INFC_G_RECORDS_SYSTEMS
•	0 H HANLINS Pû 60x 355 Bay 417a Pittshukgh Pa 15230	L M RICHMAN MANAGER PO BCX 2728 PITTSBURGH PA 15230
•	015146001625 1 99999 MESTINGHOUSE ELECTRIC CORP PLANT APPARATUS DIVISION	015230064758 1 99999 WESTINGHUUSE ELECTRIC COMP MUGLEAM EMERGY
7	MARVET H RUSENBLUH QUALITY ENGINEERING MGR BUX 425 MONROFVILLE PA 15146	D E RUSSELL MGR RELORDS & FILES OP PU BOX 355 PITTSBURGH PA 15230
1	015230081499 1 99999 WestingHuuse Electric Curp Licensing Safeguards 6 Safety	033065055288 I 99999 WLSTINGHOUSE ELECTRIC LORP Relay instrument div
•	DINECTUR PD ROX 355 PITTSBURGH PA 15230	MANAGER JUALITY ASSURANCE 4300 CORAL RIDGE DRIVE CURAL SPRINGS FL 33065
7	015238053145 1 99999 WESTINGHOUSE ELECTRIC CORP INDUSTRY ELECTRONICS DIVISION	015112053150 1 99999 Westinghuuse Electric Corp Puwer Sys Swichgear Div
)	ŃCEP MGR 200 BETA DRIVE PLITSBURGH PA 15238	MANAGER GUALITY ASSURANCE 700 BRADDOCK AVE E PITTSBURGM PA 15112
.)	015230001417 1 99999 WESTINGHOUSE ELECTRIC CORP Power Systems Division	015230005491 1 99999 MESTINGHOUSE ELECTRIC CORP W WRIGHT Dedictor
Þ	PO BCX 355 PITTSBURGH PA 15230	POST OFFICE BOX 355 PITTSBURGH PA 15230
3	099352053149 1 99999 WESTINGHOUSE HANFORD CO PO BUX 1970 RI(HI AND MA 99352	U99352077527 1 99999 Westinghouse Hanford Co J H Maglaren um Procem Dec
•		NG NGLEAR ON PROGRAM OFC PO BOX 1970 W/B-110 RICHLAND WA 99352
•	099352055108 l 99999 WESTINGHUUSE HANFORD CC Rubert L Meadur Box 1970	099352039954 1 99999 Hestinghouse Hanford Co W/D 27 A A R Schade
3	M-R-128 RICHLAND WA 99352	PU DUX 1970 W/C-81 Richland WA 99352
J	045153J26371 1 99999 Hilliam Zimmer Nuclear Tum Johnsun Us Ruute 52 Po adx 201	U04578U06480 1 99999 WISLASSET FIRST SELECTMAN MUNICIPAL RLDG - US ROUTE 1 WISCASSET MF 04578
J	MUSCON 045153	
J	004578007115 1 99999 Miscasset Public Lihrary Assiciation Migh Streef	US3702006123 1 99999 WISCCNSIN DEPT OF JUSTICE PATRICK W WALSH ASSISTANT ATTURNEY GENERAL
•	WİSCASSET ME 04578	114 EAST-STATE CAPITOL MADISON WI 53702
•	053767080976 1 99999 MISCHNSIN DILHR SAFFTY & REDG	053702064176 1 99999 WISCONSIN BLV CE STATE ENERGY

 \mathfrak{O}

G

	NGSTI SPICESE CHEUTETE CHEF LICENSING SPERALINGS OFF ED T MURPHY MANAGER 4901 FAIRMUNT AVENUE	AESTIAGHUUSE ELELIPIU LERI NES LICENSE AUMINISTRATION A J NARUI MANAGER HUX 355
	BETHESDA MD 20014	PITTSBURGH PA 15230
	015024053139 1 99999 MESTINGHOUSE ELECTRIC CUPP ELECTRIC MECHANICAL DIVISION	UI52300J6516 1 99999 WESTINGHUUSE ELECTRIC CURP A.T. PANKER
-	C L CHEN PHUJULT ASSURANCE MANAGER CHESHIFK AVENUE CHESHICK PA 15024	ри вих 355 Ріттівніком — РА 15230
~	021030081454 1 99499 MESTINGHIDUSE ELECTRIC LCKP NUCLEAR INSTRUMENTATION & CLNT M PATALUN	015230081407 1 99999 WESTINGHOUSE ELECTRIC COMP WATER REACTOR DIV E P RAME
~	GENERAL MANAGER 1111 SCHILLING RD MS 7622 HUNT VALLEY MD 21030	MANAGER-NUCLEAR POWER DEPT PC BOX 355 PITTSBURGH PA 15230
^	015230066433 1 99999 WESTINGHLUSE LLECTRIC CORP MONRIFEVILLE D H HAWLINS D H HAWLINS	015230064694 1 99999 WESTINGHOUSE ELECTRIC CORP INFO & RECORDS SYSTEMS L.H.RICHMAN
^	BAY 4174 P1TTSBUKGH PA 15230	PO BCX 2728 PITTSBUNGH PA 15230
^	D15146001625 1 99999 WESTINGHOUSE ELECTRIC CORP PLANT APPARATUS DIVISION Makvey H RDSENALUM	015230664758 1 99999 WESTINGHOUSE ELECTRIC CORP NUCLEAR ENERGY
^	QUALITY ENGINEERING MGR BLX 425 MONKOEVILLE PA 15146	MGR RECORDS & FILES OP PU BUX 355 PITTSBURGH PA 15230
0	015230081499 1 99999 MESTINGHOUSE ELECTRIC CURP LICENSING SAFEGUARDS & SAFETY A T SABU	033065055288 1 99999 #ESTINGHOUSE ELECTRIC CORP Relay instrument div
0	DIRECTOR PO 90x 355 PITTSBURGH PA 15230	MANAGER JUALITY ASSURANCE 4300 CORAL RIDGE DRIVE CURAL SPRINGS FL 33065
Q	015238053145 1 99999 WESTINGHOUSE ELECTRIC CORP INDUSTRY ELECTRONICS DIVISION T.E.SMITH	015112053150 1 99999 WESTINGHUUSE ELECTRIC CORP PUWER SYS SWICHGEAR DIV H I STANK
e -	NCGP MGR 200 BETA DRIVE PITTSBURGH PA 15238	MANAGER QUALITY ASSURANCE 700 BRADDOCK AVE E PITTSBURGH PA 15112
L.	015230001417 1 99999 WESTINGHOUSE FLECTRIC CORP Power Systems Division F_J Twogugd	015230005491 1 99999 WESTINGHOUSE ELECTRIC CORP W WRIGHT PROJECT MANAGER
ί.	PD BCX 355 PITTSdurgh PA 15230	POST OFFICE BOX 355 PLTTSBURGH PA 15230
C	099352053149 1 99999 WESTINGHOUSE HANFORD CO PO BUX 1970 RICHLAND WA 99352	099352077527 1 99999 Westinghouse Hanford CG J H Maclaaren Mgr-Nuclear ga program Ofc Po_box 1970
		N/B-110 RICHLAND NA 99352
U	099352055108 1 99999 WESTINGHUUSE HANFORD CU RUBERT L MEADUR BOX 1970 W-B-128	099352039954 1 99999 WESTINGHOUSE HANFORD CO W/D 27 A A R SCHADE PU DUX 1970
U	KIUHLANU WA 99352	N/C-BI RICHLAND NA 99352
U	045153926371 1 99999 William Zimmer Nuclear Tum Johnson Us Rode 52 PU Jox 201 Moscum On 45153	U04578006480 1 99999 WISCASSET FIRST SELECTMAN MUNICIPAL BLDG - US ROUTE 1 WISCASSET ME 0457H
	004578007115 1 99999 Miscasset Public Liyrary Assuciation Mich Street	053702006123 1 99999 Wiscensin Dept of Justice Patrick w Walsh Assistant Atturney General

N. 4

÷

2 . ٤ ω 1 , •

3

÷

4.5

÷.

	#111	. tr		• 4	▲* - €	1115.	1 50		۰.4	. .
 ,	09931 HESTI PO 86 RTCHI	52053149 INGHOUSE IX 1970 ANU	1 HANFORD	99' CC #A	999 94352	U993520 WESTIN J H MAG MGR-NU PU BGX W/B-110 RICHLA	077527 GHDUSE I LLAREN CLEAK U 1970 J	l Hanford A Progri	999 CC M CF	199 C 99352
 -	09935 WESTI RUBEH BOX 1 W-R-1 R ICHE	2055108 INGHLUSH ITL HEA 1970 28 AND	HANFORD Dur	991 CL WA	999 99352	0793520 HESTING W/D 27 A R SCI PU DUX W/C-81 R1CHLAS	039954 GHOUSE 1 HADE 1970	1 HANFORD	999 CC	199 99352
-	04511 WILLI TUM US RU MOSCO	3026371 IAM ZIMM JUHNSCH IUTE 52 IM	IER NUCLEA Po Box 20	999 1 0H	45153	UO45780 WISCAS MUNICII WISCAS	DO6480 SET FIR: PAL BLD SET	ST SELEG G - US F	999 TMAN IOUTE ME	99 1 04578
~	00457 WISCA ASSUC HIGH WISCA	18007115 SSET PU IATION STREET SSET	I IBLIC LIBR	999 ARY ME	04578	0537020 WISCON PATRICI ASSIST 114 EA MADISU	DUG123 SIN DEP K W WALS ANT ATTO ST-STATO	1 DF JUS SH JRNEY GE CAPITO	999 STICE NERA NL WI	99 IL 53702
^ ^	05370 MISCU JOHN CHIEF PU BO MADIS	07080976 INS IN 01 J DUFFY INSPEC IX 7969	LHR SAFET	999 9 E	999 BLDG 53707	0537020 HISCON DEPT OF STANLE 4802 SP HADISO	064176 SIN DIV F ADMIN Y YORK HEBOYGAI N	1 OF STA1 ISTRATIC N AVE	999 IE EN IN WI	99 IERGY 53702
с с	05320 W1SCU SOL EXECU 231 W MILWA	DIOOG46I DISIN EL BURSTEI DTIVE VI LEST MIC LUKEE	ECTRIC PO N CE PRESID HIGAN	999 WER DENT WI	53201	0532010 WISCON C W FA ASST V 231 WE MILWAU	006250 SIN ELE ICE PRE ST MICH KEE	I CTRIC PO SIDENT IGAN ST	999)#ER WI	99 CU 53201
	05424 MISCO PUINI GLENM MANAL 6610	1006474 INSIN EL BEACH I A REEU FR-NUCL NUCLEAR I VERS	1 ECTRIC PO NULLEAR P EAR OPERA ROAD	990 WER WR S	999 CO STATN NS 54241	0542410 WISCON JUINT JIZA PLANT PUSTAL THO RI	081592 SIN ELE BEACH N CH Manager Route VFRS	CTRIC PO PS 3	999 JWER WI	199 60 54241
ນ ບ	05370 WISLO PETER POE MADIS	3000485 INSIN EN ANDER DX 1105	1 VIRGNMENT SON	994 IAL I WI	53703	053703 WISCON KATHLE GENERA 114 N MADISO	005866 SIN ENV EN N F L CDUNS CARROLL N	I IRONMENI ALK EL ST	999 IAL E WI	53703
	05370 MISCU Chair Hill Madis	2005365 JNJIN PU MAN FARMS S JON	1 IBLIC SERV STATE CFF1	994 /1CE ICE M	999 CCM BLDG 53702	053707 WISCON WILLIA CHIEF Pù Box Madiso	J81507 SIN PUB M SAYL Enginee 7854 N	LIC SERT ES R	999 /ICE #1	ю99 Сом 53707
	05370 MISCO STANI CHALF PO BO MAULS	7039884 INSIN PL LEY YUF IMAN IX 7854 SON	L UBLIC SERV	49 /ICE WI	999 CDM 53707	054305 WISCON C W GI VICE P PU BOX GREEN	005834 SIN PUB ESLER RESIDEN 1200 BAY	1 LIC SERI T-NUCLE	999 VICE NR P(WI	199 CORP JWER 54305
:	05430 W1 SCH NUCL H C W C V ICE 600 P GREE	15002942 UNSIN PU AK PUNJ BIESLEK PRESIDE I ADAMS N BAY 1506476	I UALIC SERV EN DEPT ENT-NUCLE	99 VICE AR PI WI 99	999 CURP GWER 54305 999	054216 WISCON NEWAUN D C HI PLANT RT I B KEWAUN 039205	064224 SIN PUB EE NUC NTZ MANAGER LX 48 EE 001246	1 LIC SERI POWER PI	999 /ICE ANT WI 990	199 CORP 54216
								-	_	

Э

ŀ .

. .

÷

.

•

1

i

		•	and the short of the state	*
	CHIEF INSPECTOR		STANLEY YURK	•
	PU 80x 7969		4802 SHEBUYGAN AVE	NT 51702
	MADISUN	11 22/07	-201304	
		00000	0533030/04250 1	00000
	USJZULUU6461 I MISCUNSIN ELELTRIC PC)	#ER CG	WISCONSIN ELECTRIC PO	#ER CU
	SOL BURSTEIN		C W FAY	
	EXECUTIVE VILE PRESIDE	NT	231 WEST MICHIGAN ST	
~	MILWAUKEE	1 53201	MILWAUKEE	wI 53201
	054241006474 1	99999	054241081592 1	99949
$\hat{}$	WISCONSIN ELECTRIC PO	IER CO	WISCONSIN ELECTRIC PU	WER LO
	GLENN & REED		J J ZACH	
_	MANAUER-NUCLEAR OPERAT	FIONS	PLANT MANAGER	
	THU RIVERS	NI 54241	TWO RIVERS	WI 54241
			053303005844 N	00000
~	- 053703000485 1 MISLINSIN ENVIRONMENTI	AL DECADE	WISCONSIN ENVIRONMENT	AL DECADE
5	PETER ANDERSON		KATHLEEN N FALK	
	P 0 80X 1135	41 53703	HIA N CARROLL ST	
\sim	HAD1 304		MADISON	WI 53703
• ·				
	053702005365 1	99999	053707081507 1	99999
0	WISCONSIN PUBLIC SERV	ICE COM	WISCONSIN PUBLIC SERV	1CE CDM
	NILL FARMS STATE LEFT	CE BLDG	CHIEF ENGINEER	
	MADISON	NI 53702	PO BOX 7854	
\frown			MADISUN	#1 337U/
				00000
-	053707039884 1	99999 166 60W	054305005834 I HISCONSIN PUBLIC SERV	44444
C	STANLEY YORK		C W GIESLER	
	CHAIRMAN		VICE PRESIDENT-NUCLEA	R POWER
	PU BUX (854	1 53707	GREEN BAY	WI 54305
	054305002942 1	99999	054216064224 1	99999
• -	WISCONSIN PUBLIC SERV	ICE CORP	WISCONSIN PUBLIC SERV	ICE CORP
-	NUCLEAR POWER DEPT		D C HINTZ	ANI
	VICE PRESIDENT-NUCLEA	R POWER	PLANT MANAGER	
	600 H ADAMS		RT 1 BUX 48	HT 54216
	GREEN BAT	1 24302	KENAUNEE	MI J4210
	054305064763 1	99999 165 6000	039205001246 1	99999
0	J N MURRISON	ILE CURP	ROBERT B MCGEHEE	ANAWAI
	NUCLEAR ADMIN SUPY		ATTORNEY	
	CREENBAY	KI 54305	JACKSON	MS 39205
~	GREE ADDEN			
	001609079621 1	99999	075201006519 1	99999
i 1	WORCESTER POLYTECHNIC	INST	WORSHAM FORSYTHE & SA	MPELS
C	E CRANCH		SPENCER C RELYEA	
	WURGESTER	MA 01609	DALLAS	TX 75201
	035607057011 1	99999	091760057154 3	99999
	LIBRARY		DEPT 170	
	PD BUX 1008	41 35807	RALPH L CUMMINGS	ANNING
	NU4134162	37841	PO HOX 160	OF1 & FIF1
			NORCO	CA 91760
	035807057010 1	99999	035807053348 1	99999
	WYLE LABORATOR IES		WYLE LARORATURIES	
-	PD 80% 1008		DIRECTOR NUCLEAR PROG	RAMS
	HUNTSVILLE	AL 35807	PU 50X 1008	
. •			UD412AIFFF	AL 35807
	DEZODIOIZZ73 1	99999 5 513 520	02/028005511 1 YADKIN RIVER COMMITTE	99999 F
	RAUTULUGICAL HLTH SVC	- 300 316	MARY & DAVIS	
	THUMAS SCHELL		CHAIRMAN	

VILE PRESIDENT-NULLEAP PUNEH 600 H AUAMS GREEN BAY WI 54305	PLANT MANAGER RT 1 BOX 48 wt 54216 KEMAUNEE wt 54216
054305064763 1 99999 WISCINSIN PUBLIC SERVICE CORP J N MUBRISCI J N MUBRISCI	U39205001246 1 99999 WISE CARTER CHILD & CARAWAY Rubert B MCGEMEE Attorney
PO BOX 1200 GREENBAY NI 54305	PU BCX 651 JACKSON MS 39205
001609079421 1 99999 WURLESTER PULYTECHNIC INST	075201006519 1 99999 WUKSHAM FORSYTHE & SAMPELS SPENER & RELYEA
PRESIDENT MA 01609 WURLESTER MA 01609	2001 BRYAN TUWER TX 75201 DALLAS TX 75201
035807357011 1 99999 Wyle Laburatories	091760057154 3 99999 Wyle Laburatur IES DLPT 170
PO BUX 1008 HUNTSVILLE AL 35807	RĂLPH L CUMMINGS DIRECTOR-STRATEGIC PLANNING PO ROX 160 NORCO CA 91760
035807057010 1 99999 WYLE LABORATORIES	035807053348 1 99999 WYLE LABORATORIES
R [°] HALL PO BOX 1008 HUNTSVILLE AL 35807	N SCALES DIRECTOR NUCLEAR PROGRAMS PO BOX 1008 HUNTSVILLE AL 35807
082001012273 1 99999 WYDMING DEPT OF HITH & SOC SVC	027028005511 1 99999 Yadkin River Conmittee Mary A Davis
THUMAS SCHELL CHIEF HATHAWAY BUILDING CHEFFANG WY 82001	CHAIRMAN ROUTE 4 BOX 261 MOCKSVILLE NC 27028
098948006417 1 99999 Yakima Indian Nation Consclidated Tribes & Bands	001701005714 1 99999 Vankee Atomic Electric CD Library
RUSSELL JIM TRIBAL CUUNCILMAN PO BUX ISI Toppenish na 98948	LIBRARIAN 1671 WURCESTER ST FRAMINGHAM NA 01701
001367036517 1 99999 YANKEL ATOMIC ELECTRIC CC H AUTIU	OD1701007098 1 99999 YANKEE ATOMIC ELECTRIC CO JUHN DEVINCENTIS
PLANT SUPT STAR ROUTE MA 01367 RUNE MA 01367	1671 WORCESTER RD FARMINGHAM MA 01701
001701081394 1 99999 Yanxee Atomic Electric Co Ionn Defincentis	001701008690 l 99999 Yankee Atomic Electric Cu W P Johnson
IG71 WURCESTER RD FRAMINGTON NA 01701	VICE PRESIDENT 1671 WORCESTER ROAD FRAMINGHAM ME 01701
001701005500 1 99999 YANKE ATOMIC ELECTRIC CO	001701081046 1 99999 Yankee Atomic Electric Co K Iskern
JANES A KAY SENIDH ENGINEER 1671 WURCESTER KD ERANINGHAM NA U1701	Î671 MURCESTER RD FRAMINGMAM MA 01701
001701005410 1 99799 YANKEL ATUMIC ELECTRIC CC	UDIS81007024 1 99999 YANKEE ATUMIL ELECTRIC CO DUNALD E VANDENURGH
JAMES E INIBOLL PRESIDENT 1671 MURCESTER RD FRAMINGHAM MA 01701	VICE PRESIDENT OF ENGINEERING 20 TURNPIKE RDAD WESTHORD NA 01581
019422053153 1 99999 Yarway Curp	OL7405064457 I 99999 Yurk Electro Panel Contrul Co
W A VOLGIR Manager quality systems Blue Bll PA 19422	JA MANAGER VORK COUNTY INDUSTRIES PARK PUB 1702

Э

1

 \odot

 \odot

HERE HELE BLUE HELE	PA 19422	44 HAHAGEH VURK COUNTY INDUST POB 1702	RIES PARK			
044509053154 1	99900	074101053155 1	99999 99999			
P G SUX 2401 YOUNGSTUWN	0H 44509	PU BUX 3158 TUESA	OK 74101			
002114080065 1 ZEVITAS ASSOCIATES	99999	060099006542 1 2100 OFC OF MAYOR	99999			
BOSTON	MA 02114	ZION	IL 60099			
055144064622 1 3M CC L R MARQUARDT	99999					
MARKET MGR SERV HEAL 3M CLNTER BUILDING 2 ST PAUL	THCARE 20-9E HN 55144					
11	M)		00 NN	NN BRKRERRRKRR (2222
	MHI MMNI MMNI	N NMH DDDDDDDD N MMNM DU	DD NNN DD NNNN	NN RRRRRRRRR CCC		
		MM MM 00 00	D NN NN NN NN NN NN	RR RR CC	44 44 22 444444444 22	22
		MM 00 00 MM 00 00 MM 00 00 N	NN NN NN NN NN NN R N NNN RH	RARRARRARR CC R RR CC L RR CC	444444444 22 44 22 44 22	
LL LÎLLILLILL LILLILLILL	мм р/ мм м/ мм ми	MM DD DD NN M DJDDDDDDDD NN Ddyddddd NN	NGN RR NN RR N RR	RR CC RR CCCCCCCCCC RR CCCCCCCCCC	44 2222222222 44 2222222222 44 222222222	
			00000		ţţ.	
	لَّلَ د ل الد					
LL.	LL LL					
LL LL	11111					
	00 100000000					
••••E END JOB II	JO LMDNRC42 DO LMDNRC42	NFCDO4AX	RUDM 306 RUDM 306	2.30.02 PM 10 NOV 82 2.30.02 PM 10 NOV 82 2.30.02 PM 10 NOV 82	PRINTERS SYS IPUL JOB 1100 E	ND Lee
CONTRACT OF THE TRACE OF THE TR	CO LMDNRC42 CO LMDNRC42	NRCDO4AX NRCDO4AX NRCDO4AX	RUCM 306 RUOM 306 RUOM 306	2.30.02 PM 10 NOV 82 2.30.02 PM 10 NOV 82 2.30.02 PM 10 NOV 82	PRINTER3 SYS IPOL JOB 1100 E PRINTER3 SYS IPOL JOB 1100 E PRINTER3 SYS IPOL JOB 1100 E	ND L** ND L** ND L**
++++L END JUB 11 ++++L END JUB 11 ++++L END JUB 11	UU LMDNRC42 00 LMDNRC42 00 LMUNRC42	NRCDO4AX NRCDO4AX NRCDO4AX	RCUM 306 RDOM 306 ROOM 306	2.30.02 PM 10 NOV 82 2.30.02 PM 10 NOV 82 2.30.02 PM 10 NOV 82	PRINTER3 SYS IPO1 JOB 1100 E PRINTER3 SYS IPO1 JOB 1100 E PRINTER3 SYS IPO1 JOB 1100 E	ND L** ND L** ND L**
****L END JUB L1 ****L END JGB L1 ****L END JGB L1	UU LMUNKC42 UU LMUNKC42 UU LMUNKC42	NKLDU4AX NRCD04AX NKCD04AX	ROCM 306 RDOM 306 ROOM 306	2.30.02 PM 10 NOV 82 2.30.02 PM 10 NOV 82 2.30.02 PM 10 NOV 82	PRINTER3 SYS IPUI JOB IIOO E PRINTER3 SYS IPUI JOB IIOO E PRINTER3 SYS IPUI JUB IIOO E	ND L** ND L** ND L**
****L FND JOB 11 ****L END JOB 11 ****L FND JOB 11	00 LMDNRC42 00 LMDNRC42 00 LMDNRC42	NRÉDÓ4AX NRÉDÓ4AX NRÉDÓ4AX	RÖĞM 306 ROOM 306 ROON 306	2.30.02 PM 10 NOV 82 2.30.02 PM 10 NOV 82 2.30.02 PM 10 NOV 82	PRINTERI SYS (POI JOB 1100 E PRINTERI SYS (PUI JOB 1100 E PRINTERI SYS (PUI JOB 1100 E	ND L** NU L**
● ● ● ● ● ● ● ● ● ● ● ● ● ● ● ● ● ● ●	OU EMUNACA2 OO EMUNACA2 OO EMUNACA2	NRCDO4AX NRCU04AX NACD04AX	ROUM 306 ROCN 306 ROCN 306	2.30.02 PM 10 NOV 82 2.30.02 PM 10 NOV 82 2.30.02 PM 10 NOV 82	PRINTERS SYS IPOL JOB 1100 E PRINTERS SYS IPOL JUB 1100 E	ND L++
****E END JUB 11	00 LMDNRC42 00 LMDNRC42	NRCDJ4AX NPCDJ4AX	ROUM 306 ROOM 306	2.30.02 PM 10 NOV 82 2.30.02 PM 10 NOV 82 2.30.02 PM 10 NOV 82	PRINTERS SYS IPUL JOB 1100 E	ND L**
••••L END JUB II ••••L END JUB II ••••L END JUB II	UO LMUNRC42 UO LMUNRC42	NRLDJ4AX NKCDJ4AX	RDUM 306 RDUM 306	2.30.02 PM 10 NOV 82 2.30.02 PM 10 NOV 82 2.30.02 PM 10 NOV 82	PRINIERS STS IPUL JUB LLOO E PRINIERS SYS IPUL JUB LLOO E PRINIERS SYS IPUL JUB LLOO E	ND L** ND L** ND L**
	UU LMUNRE42	NHCUUGAX	RUGA 306	2-30-02 PH 10 NOV 82	PRINTERS SYS IPUL JOB 1100 E	ND L***
CONTRACT OF THE	UO LMONRC42	NECDUAAX	ROOM 306	2.30.02 PM 10 NUV 82	PRINTER3 SYS IPUL JUB 1100 E	ND L++
	00 EMONRÉ42 00 EMONRE42 00 EMONRE42 00 EMONRE42 00 EMONRE42	NRCD04AX NRCD04AX NRCD04AX NRCD04AX NRCD04AX NRCD04AX	ROOM 306 ROOM 306 ROOM 306 ROOM 306	2-30-02 PM 10 NOV 82 2-30-02 PM 10 NOV 82 2-30-02 PM 10 NOV 82 2-30-02 PM 10 NOV 82	PRINTER3 SYS IPUI JUB 1100 E PRINTER3 SYS IPUI JUB 1100 E PRINTER3 SYS IPUI JUB 1100 E PRINTER3 SYS IPUI JUB 1100 E	

•

December 3, 1982

MEMORANDUM FOR: Chairman Palladino FROM: William J. Dircks Executive Director for Operations

.

11 E

SUBJECT: ZIMMER MONTHLY STATUS REPORT TO COMMISSION

Enclosed is the Zimmer monthly status report for October.

(Signed) William J. Direk

William J. Dircks Executive Director for Operations

	Enclosure As stated	:					
	CC: Comm Comm Comm SECY OGC OPE OIA OI OCA PDR	Issioner Gilir Issioner Ahear Issioner Rober Issioner Assel	nsky me ts Ist1ne				
	Distributic Dircks Roe Rehm Stello Keppler DeYoung Denton Res. Insp., EDO r/f	on: 821208 PDR AI R , RIII (Tr #	30 595 8 21203 ООСК 0 5 00035 РГ Pransmitted via	B R 5520 from RI	II to OEDO 12	2/1/82)	
OFFICE	RIII ESchweibinz	RIII JKeppler	EDO WJDircks				
DATE	12/1/82	12/1/82	12/ 3 /82		•••••••••••••••	•••••	
NRC FORM 318	(10-80) NRCM 0240		OFFICIAL	RECORD C	OPY		USGPO: 1981-335-960

U.S. NUCLEAR REGULATORY COMMISSION

3

REGION III

ZIMMER MONTHLY STATUS REPORT

October 1982

0

8212080597 821203 PDR ADOCK 05000358 R PDR

1

•

Table of Contents

t

- Summary of the Project for the Month Α.
- NRC Zimmer Section Manpower Availability and Utilization Β.
- С. High Visibility Items
 - Catalytic, Inc. 1.
 - 2. HJK Investigation Report

Quality Confirmation Program (QCP) D.

- Structural Steel Task I. 1. Weld Quality 2. Task II. Task III. Heat Number Traceability 3. Task IV. Socket Weld Fitup 4. Radiographs Task V. 5. Task VI. Cable Separation 6. Task VII. Nonconformances 7. Task VIII.Design Control and Verification 8. Task IX. Design Document Changes 9. Task X. Subcontractor QA Programs 10. 11. Task XI. Audits
- Ε. **Ongoing Activities**
- F. Potential Plant Problems
- G. **FOIA Requests**
- Η. Attachments
 - Summary of the progress of the QCP task areas Ι.
 - II. QCP Status as of October 31, 1982 (1 III. The November 12, 1982 Order to CG&E QCP Status as of October 31, 1982 (Provided by Licensee)

 - IV. Licensee November 15, 1982 response to NRC October 27, 1982 Request for Information

2

A. Summary of the Project for the Month

Meetings were held on October 27, 1982 with senior NRC staff representatives and on October 28, 1982 with the Commission to discuss the status of the Zimmmer project and problems being identified. (These meetings precipitated followup Commission meetings in November and led to the Commission decision to issue an "Order to Show Cause and Order Immediately Suspending Construction" on November 12, 1982 (Attachment III).)

Region III inspection efforts at the Zimmer facility during the month of October were concentrated on the activities of Catalytic, Inc. (CI), the CG&E overview of CI, review of nonconformance reports, and the routine monitoring and inspection of other ongoing activities including the QCP.

The resident inspector conducted a tour of the Zimmer site for the U.S. Attorney, Southern District of Ohio, representatives of his staff, and a representative of the F.B.I.

A management meeting was conducted with the licensee on October 19, 1982, (open meeting) regarding the NRC inspection findings from a Catalytic, Inc. inspection. Details are provided in section C.1 of this report.

The National Board of Boiler and Pressure Vessel Inspectors is continuing inspection efforts onsite. The National Board issued interim reports on May 12, July 1, (August 6 supplement), and September 30, 1982. The findings of the National Board are consistent with and similar to NRC findings. The licensee responded on August 5 and 30, October 20 and 29, 1982, and plans to provide a bi-weekly status report.

In accordance with 10 CFR 50.55(e), the licensee reported the following potential construction deficiencies:

On October 13, 1982, design changes made to the fire protection system piping in the cable spreading room in 1979 to upgrade the piping to have seismic supports was done with no evidence of quality assurance of the modifications or of Sargent & Lundy calculations for the modifications.

On October 26, 1982, the licensee determined that Sargent & Lundy (S&L) dynamic analysis of small bore piping at Zimmer was questionable. The stress analyses are being reviewed by S&L and the licensee. A similar report was submitted concerning Clinton.

On October 27, 1982, the licensee reported a potential problem relating to division separation between non-essential cables and bundling with essential cables during cable pulling and termination.

On October 29, 1982, the licensee determined that fire dampers manufactured by Air-Balance, Inc. have a fusable link held by a "J Hook" and that one prevented the fire damper from closing when the damper was actuated. On October 29, 1982, the licensee determined that pipe support installation procedures did not contain seismic clearance criteria between pipe supports and cable trays or conduit and associated supports as required by the specification. This may indicate a lack of proper interface on design.

Region III is continuing evaluation of the licensee's responses to the investigation report 81-13 and civil penalty, and some progress was made during the month because of a temporary personnel assignment to this project.

٠

Since the July Monthly Status Report was prepared, two petitions, both dated August 20, 1982, have been received from the Government Accountability Project (GAP). The first is for reconsideration of the Commission's Order of July 30, 1982. The staff has considered this petition and made it's recommendations to the Commission. The second petition is to suspend construction of the Zimmer Station. On September 24, 1982, the RIII Regional Administrator issued a "Demand for Information" pursuant to the Commission's authority under Section 182 of the Atomic Energy Act and 10 CFR 50.54(f) of the Commission's regulations. This "Demand for Information" requires the licensee to admit or deny each of the allegations applicable to the licensee's and its principal contractor's or subcontractor's performance contained in paragraphs 19 through 273 of the petition. If the allegations are not admitted, they must explain the basis for not admitting the allegations. In addition, they must identify the manner in which the Quality Confirmation Program (QCP) addresses the type of existing or potential quality assurance or construction deficiencies and problems identified in each of the above allegations. If the QCP does not address such deficiencies or problems, CG&E must describe the manner in which they will ensure such deficiencies or problems are corrected. They have until December 31, 1982 to respond. On October 18, 1982, GAP issued a supplement to their August 20, 1982 petition. The staff is reviewing this supplement.

NRC Region III was advised of allegations that relevant documentation on the welders at the Zimmer site was prepared for, or reviewed at, a meeting between Cincinnati Gas and Electric Company (CG&E) and H. J. Kaiser (HJK) held on July 8, 1982, but that such documentation was not discussed with, or made available to, Region III at a meeting on this subject held on July 9, 1982, between CG&E, HJK and Region III. Substantial documentation was made available to Region III in connection with the July 9 meeting, but additional relevant documentation was allegedly not made available. The matter of the existence of such documentation was informally discussed on October 15 and 20, 1982, between Messrs. B. R. Sylvia of CG&E and R. F. Warnick and D. R. Hunter of NRC Region III. Further information was required to resolve these concerns and, if these concerns were found to be valid, to determine whether enforcement action, including modification, suspension, or revocation of CG&E license, is warranted.

Accordingly, pursuant to Section 182 of the Atomic Energy Act and 10 CFR 50.54(f) of the Commission's regulations, on October 27, 1982, CG&E was required by Region III to submit under oath and in writing, by November 16, 1982, a list of all reports or other documentation prepared by CG&E's or HJK's document reviewers or other personnel in preparation for the meetings of July 8 and/or 9, 1982, or reviewed at those meetings. Copies of any such reports were requested to be made available at the Zimmer site for inspection by the NRC. (The licensee responded to this 10 CFR 50.54(f) "Request for Information" on November 15, 1982 (Attachment IV). This response is currently being reviewed by Region III).

B. Zimmer Section Manpower Availability and Utilization

1. Assigned Manpower

t

Section Chief (Assigned to Zimmer full time. Onsite part-time.)

Project Manayer (Assigned to Zimmer full-time. Onsite as needed.)

Resident Inspectors

Senior Resident (Full-time onsite) Resident Inspector (Full-time onsite) Resident Inspector (Full-time onsite)

Investigators (Office of Investigations (OI))

Five Investigators (Full-time)

2. Summary of Manpower Utilization

Onsite and in office professional effort from January 3, 1982 through October 30, 1982 was approximately 8,949.5 manhours, with 1068 of these manhours occurring between October 3 and October 30, 1982. (These hours do not include those of OI investigators).

C. High Visibility Issues

1. Inspection of Catalytic, Inc. Activities

Special inspection performed by Region III during weeks of August 10 and September 7 and 13, 1982, regarding the adequacy of licensee control of Catalytic, Inc. work activities and to determine if the licensee had violated the December 24, 1980 NRC Immediate Action Letter (IAL) and the December 30, 1980 CG&E Stop Work Order (SWO).

Effort by Licensee

During the NRC inspection a number of corrective actions were taken by the licensee, including the issuance of a Limited Stop Work Order (SWO) on September 10, 1982, on miscellaneous work by Catalytic, Inc.; and a Stop Work Order (SWO) on October 11, 1982, for all essential work activities performed by Catalytic, Inc., following the discussion of the inspection findings with Region III on October 8, 1982. Immediately prior to the inspection, the licensee notified the NRC that a Stop Work Order (SWO) was initiated on August 5, 1982, regarding Catalytic, Inc. work associated with the removal of Control Rod Drive (CRD) hangers and supports.

NRC Effort/Action

ŧ

The NRC inspection identified a number of apparent programmatic concerns as well as individual findings in the field concerning the Catalytic, Inc. safety-related activities.

A management meeting was conducted at the Greater Cincinnati Airport (Open Meeting) on October 19, 1982, to discuss the inspection findings.

Additional working meetings were held on November 2-4, 1982, to discuss the specific technical and programmatic issues in more detail.

Findings to Date

The licensee and the NRC have identified a number of concerns in the areas of organizational description and interfaces, assignment of responsibilities and authority, training, design control, procedures, document control, inspections, nonconformance control, corrective action, records, and audits. A number of concerns are considered repeat items. The inspection results indicate a breakdown in the CG&E management controls regarding Catalytic, Inc. The above items were considered prior to the issuance of the November 12, 1982 NRC ORDER.

2. H. J. Kaiser (HJK) Internal Investigation Report

Region III anonymously received a partial copy of a report of a HJK investigation conducted at Zimmer. The report was mailed from Cincinnati, Ohio on March 23, 1982, and received by Region III on March 26, 1982. The NRC is investigating to determine the safety significance of the matters described in the investigation report and whether or not NRC reporting requirements were met.

Effort by Licensee

The licensee is conducting a review of the report.

NRC Effort/Action

The NRC investigation into the HJK investigation report and its significance has been transferred to the Office of Investigations.

Findings to Date

The findings will be discussed after the investigation is complete.

D. <u>Quality Confirmation Program (QCP)</u>

1

A summary of the progress of the QCP task areas is provided as Attachment I. The CG&E QCP Status Report as of October 30, 1982, is included as Attachment II. This attachment includes more detail than the following summary presents.

1. Task I, Structural Steel

Review of structural beams, beam welds, re-entrant corners, procurement of beams, beam and steel plate heat number traceability, and inspection of cable tray foot connections.

Effort by Licensee

The licensee is continuing to inspect structural steel items including foot connections, drywell steel, control room steel, gallery steel, and switchgear steel.

The task involves 26 personnel and is reported to be 58% complete with an estimated completion date of December 1, 1982.

NRC Effort/Action

Reviewed 2,013 initial issue nonconformance reports (NRs) and 138 dispositioned NRs and inspected a selected number of those reviewed.

Resident and specialist inspectors performed field walkdowns of actual conditions and rework activities.

Findings to Date

The licensee has generated 977 NRs identifying about 9546 weld deficiencies. The majority of the 4963 deficiencies dispositioned have been dispositioned as "rework".

Three construction deficiency reports have been reported to the NRC concerning laminated angle iron, cable tray hangers/weld deficiencies, and cable tray hanger "Nelson stud" deficiencies.

2. Task II, Weld Quality

Review of welds performed, weld rod control, transfer of weld rod heat numbers, and deletion of weld inspection criteria.

Effort by Licensee

The licensee is continuing to review the areas of structural weld cards, welding procedures, welder qualifications, small-bore piping welds, and large-bore piping welds.

The task involves six people and is reported to be 78% complete with the completion date still to be determined. The licensee is reorganizing the task work with both HJK and CG&E performing the reviews to improve efficiency.

NRC Effort/Action

Reviewed 642 initial issue NRs and 74 dispositioned NRs and inspected a selected number of those reviewed.

Findings to Date

The licensee has identified nonconforming conditions regarding weld data sheets, heat numbers, welder qualifications, and welding procedures. Discrepancies being identified include lack of objective evidence, white-outs and cross-outs, signature differences, inconsistent data, and lack of adequate acceptance criteria. (Welder qualifications are being reviewed by a special HJK/CG&E task group.)

Due to the identification of the lack of adequate control of drawings, inadequate acceptance criteria, and restart of the review activity under a new program, the large-bore piping weld review has not progressed.

Three construction deficiency reports have been identified to the NRC concerning weld procedure deficiency, carbon weld rod in stainless steel weld, and lack of weld preheat or post weld heat treatment.

3. Task III, Heat Number Traceability

Review of installed large-bore and small-bore pipe heat numbers, heat numbers on isometric drawings, incorrect or marked-up heat numbers, and purchase orders.

Effort by Licensee

The licensee is continuing to review heat number traceability in the areas of large-bore piping, small-bore piping, and by purchase orders.

The task involves 5 people (the August-September monthly status report incorrectly showed 14 people on this task, the correct number for that time was 6 people) and is reported to be 36% complete with the estimated completion date still to be determined.

NRC Effort/Action

Reviewed 529 initial issue NRs and 36 dispositioned NRs.

Findings to Date

The licensee has identified a substantial number of discrepancies regarding small-bore piping, large-bore piping, and purchase orders.

The nonconforming conditions include documentation deficiencies, lack of design change control, unsigned reports, use of unapproved vendors, and upgrade of materials.

Three construction deficiencies have been identified to the NRC concerning 2400 ft. of SA-106 Grade B piping, some bolting material, and control of "gamma plugs."

4. Task IV, Socket Weld Fitup

Review of socket weld fitup to ensure adequate disengagement for small bore piping.

Effort by Licensee

The task is substantially complete with 29,821 fitups reviewed of a total of 32,000. The final reviews of the identified discrepancies and radiographs are being made.

The task involves one person, and is reported as 98% complete with an estimated completion date of December 1, 1982.

NRC Effort/Action

Reviewed 115 initial issue NRs and 84 dispositioned NRs and inspected a selected number of those reviewed.

Findings to Date

The licensee identified 695 weld joints which lacked evidence of disengagement and the joints have been radiographed. Of the 695 welds 111 have been rejected for lack of disengagement.

5. Task V, Radiographs

Review the radiographs which did not meet ASME Code requirements due to inadequately shimmed penetrameters.

Effort by Licensee

The licensee has completed the review of the radiographs and is preparing a code inquiry to ASME concerning the shimming of the penetrameters.

The selected, qualifying radiographs are being reviewed to assess the weld conditions. Of the 61 welds initially identified, 2 welds have been found to be duplicates, 1 weld had been replaced and was removed from the list, and 12 welds are inaccessible. This reduces the total to 46 welds identified. All 46 have been re-radiographed and accepted.

This activity is reported to be approximately 99% complete with a completion date of November 15, 1982.

NRC Effort/Action

The resident inspector reviewed 97 initial issue NRs and 4 dispositioned NR.

Findings to Date

A substantial number of the M. W. Kellogg radiographs were not shimmed adequately; however, the quality and sensitivity of the radiographs appears adequate.

Two construction deficiency reports have been identified to the NRC concerning radiographs.

6. Task VI, Cable Separation

Review cable separation regarding essential and associated cables.

Effort by Licensee

The licensee is continuing the review and evaluation of wall penetrations/sleeves, associated cables for Class 1E panels, all 1E panels, and responses to engineering evaluation requests (EERs).

The task involves nine persons and is reported to be 50% complete (scope expanded) with a completion date of June 1, 1983 (scope expanded).

NRC Effort/Action

Reviewed 879 initial issue NRs and 285 dispositioned NRs and inspected a selected number of those reviewed.

Findings to Date

Nonconforming conditions have been identified concerning cable separation, identification, and routing. The licensee has written 776 NRs and 345 NRs have been dispositioned (131 rework, 25 repair, and 189 accept-as-is) of which 217 NRs have been closed to date, with an additional 12 NRs cancelled because of duplication.

Three construction deficiency reports have been identified to the NRC concerning electrical cable separation.

7. Task VII, Nonconformance Reports

Review of noncomformances documented in surveillance reports, punch lists, and exception lists; and nonconformances not documented, not entered, and voided rather than adequately dispositioned. Review 300 closed NRs and solicit NRs not entered into the system.

Effort by Licensee

ŧ

The licensee is continuing to review and evaluate nonconformances.

Approximately 200 letters to former QC inspectors soliciting nonconformances not entered into the system return receipt requested were mailed by the licensee. The responses were limited. One report has been received from an individual identifying four potential NRs.

Additional procedures, field walkdowns, and a material impound area are being provided.

The task involves three persons and is approximately 61% complete with an estimated completion date of January 30, 1983.

NRC Effort/ Action

Reviewed 367 initial issue NRs and 257 dispositioned NRs and inspected a selected number of those reviewed.

Findings to Date

Fifty eight NRs have been written due to reopening of previously voided NRs. Of these NRs, 43 have been dispositioned (15 rework, 1 repair, and 27 accept-as-is).

8. Task VIII, Design Control and Verification

Review procedures controlling design calculation completion, S&L program for controlling deviations from FSAR, correctness and consistency of FSAR, and design deviation identification and disposition.

Effort by Licensee

A programmatic audit of S&L design control was needed to complete this task. The audit was conducted on September 1-3, 1982 in Chicago. There were no findings. The audit will not be closed until the two recommendations given by the audit team have been resolved.

The task is 99% complete with an estimated completion date of December 15, 1982.

NRC Effort/Action

The activity is being monitored with a closeout inspection planned to verify completion of the QCP Task VIII items, to identify any generic implications, and to evaluate the licensee conclusions.

Findings to Date

No significant findings have been identified to date regarding the S&L work; although, the S&L system has been made more formal.

Four construction deficiency reports have been identified to the NRC regarding design control.

9. Task IX, Design Document Changes (DDC)

Establish an accurate and complete listing of DDCs, DDC records, and associated QC inspection records.

Effort by Licensee

This task was divided into five phases as follows:

- Phase I Classification of CG&E, S&L, and HJK, DDCs as essential or nonessential.
- Phase IA Classification of WY&B and other site contractor DDCs as essential or nonessential.
- Phase II Review inspection documentation to determine if CG&E, S&L, and HJK DDCs have been incorporated and inspected.
- Phase IIA Review inspection documentation to determine if WY&B and other site contractor DDCs have been incorporated and inspected.

Phase III Electrical inspections in the control room.

The original task (Phases I and II) is 40.8% complete. Phases IA and IIA are 9.2% complete and Phase III is 10% complete.

The task involves 14 persons for Phases I, II, IA, and IIA with an estimated completion date of January 1983. Phase III involves 6 persons with an estimated completion date of June 1983.

NRC Effort/Action

Reviewed 511 initial issue NRs and 3 dispositioned NRs and inspected a selected number of those reviewed.

Findings to Date

A number of deficiencies have been identified concerning missing documentation, misclassification of DDCs, inspection program deficiencies, failure to incorporate deficiencies, premature inspection, and incomplete inspection documentation.

10. Task X, Subcontractor QA Programs

Confirm the quality of the Bristol Steel work and review all subcontractor QA programs or safety-related work to ensure the safety-related activities performed were acceptable.

.

Effort by Licensee

Bristol field welds are being reviewed within Task I and 80 audits of subcontractors were identified encompassing 13 subcontractors. All 80 audits have been reviewed.

.

The task involves two persons and is 78% complete with an estimated completion date of December 1982.

NRC Effort/Action

None during this period.

Findings to Date

Six subcontractor audits may require audits of subsequent activities for confirmation of work that cannot be verified by document review or inspection.

Many of the 80 audits reviewed do not address all applicable criteria.

11. Task XI Audits

Review all past audits of HJK, S&L, GE, GED, EPD, EOTD, and GCD to determine depth and adequacy with respect to Appendix B to 10 CFR 50 and appropriate closeout of audit findings. Justify the acceptability of the areas not audited.

Effort by Licensee

Review of audits, audit summaries prepared, and all of the 296 past audits have been reviewed.

The task involves two persons and is 82% complete with estimated completion date of November 15, 1982.

NRC Effort/Action

None during this period.

Findings to Date

Coverage of the audits was not sufficient to verify adequate implementation of program requirements.

E. Ongoing Construction Activities

Major ongoing construction activities during October included installation and modification of pipe supports, drywell steel modifications, installation of drywell air coolers, seismic modification to switchgear, rework of control room structural steel, and installation of seismic columns. As of November 12, 1982, all safety-related construction activities, including rework of identified deficient construction was suspended by an NRC ORDER.

F. Potential Plant Problems

The following is a list of areas or items which Region III considers as 'potential problems and are being monitored by the inspectors.

- . Rust on the stainless steel liner plate in suppression pool
- . Containment liner leak rate channel leakage (welds)
- . Sacrificial shield weld inadequacies (records and actual weld conditions)
- . Cable tray trapeze weld and support stud inadequacies
- . Past personnel qualifications
- . Past weld procedures
- . Purchase of equipment
- . Structural steel bolting
- . Control panel and electrical switchgear mounting plug weld inadequacies
 - Cable seperation inadequacies
- G. Freedom of Information Act (FOIA) Requests

None during this period.

SUMMARY OF THE PROGRESS OF QUALITY CONFIRMATION PROGRAM (QCP) TASK AREAS AS OF JUNE, JULY, AUGUST, SEPTEMBER, AND OCTOBER 1982

•

.

	TASK AREA	PERCENT COMPLETE/EXPECTED COMPLETION AS OF						
		JUNE	JULY	AUGUST	SEPTEMBER	OCTOBER		
Ι.	STRUCTURAL STEEL*	31% 12/01/82	35% 12/01/82	50% 12/01/82	57% 12/01/82	58% 12/01/82		
11.	WELD QUALITY*	58% **	58% **	62% **	68% **	70% **		
III.	HEAT NUMBER TRACEABILITY*	30% **	30% **	33% **	33% **	36% **		
IV.	SOCKET WELD FITUP	98% 08/01/82	95% 10/01/82	96% 10/01/82	98% 12/01/82	98% 12/01/82		
۷.	RADIOGRAPHS	97% 08/01/82	97% 09/15/82	98% 11/15/82	98% 11/15/82	99% 11/15/82		
VI.	CABLE SEPARATION*	54% 12/31/82	52% 12/31/82	35% 06/01/83	44% 06/01/83	50% 06/01/83		
VII.	NONCONFORMANCES	61% 12/31/82	40% 12/31/82	52% 12/31/82	61% 01/30/83	61% 01/30/83		
VIII.	DESIGN CONTROL AND VERIFICATION	97% 07/15/82	99% 08/15/82	99% **	99% **	99% 12/15/82		
IX.	DESIGN DOCUMENT CHANGES	34% 12/31/82	35% 12/31/82	<32% 04/15/83	<33% 04/15/83	<35% 06/01/83		
Χ.	SUBCONTRACTOR QA PROGRAMS	60% 08/13/82	65% 09/15/82	75% 10/15/82	75% 10/30/82	78% 12/31/82		
XI.	AUDITS	70% 10/08/82	72% 10/08/82	74% 11/15/82	80% 11/15/82	82% 11/15/82		

*Areas viewed by Region III as potentially requiring a significant amount of rework. **Estimated completion date to be determined.

Attachment I

QUALITY CONFIRMATION PROGRAM

٠

. . . .

.

4

STATUS

(AS OF OCTOBER 31, 1982)

BY

J. F. SHAFFER

DIRECTOR, QUALITY CONFIRMATION PROGRAM

.

Attachment II

TASK I: STRUCTURAL STEEL

ACTION BEING TAKEN

.

- A. 100% VISUAL INSPECTION OF ACCESSIBLE STRUCTURAL STEEL BEAM FIELD WELDS, BRISTOL SHOP WELDS, RE-ENTRANT CORNERS AND WELDED CABLE TRAY FOOT CONNECTIONS.
- B. 100% OF ALL BEAMS INSPECTED ARE COMPARED TO THE DESIGN DRAWINGS AND DESIGN DOCUMENT CHANGES (DDC'S).

C. 100% VISUAL INSPECTION OF HVAC SUPPORTS IN THE CONTROL ROOM.

SUMMARY OF TASK

AREA	TOTAL # OF ITEMS	ITEMS COMPLETED THIS MONTH	ITEMS COMPLETED TO DATE	PERCENT COMPLETED THIS MONTH	PERCENT COMPLETED TO DATE	MANHOURS EXPENDED THIS MONTH	MANHOURS REMAININ
1)DRYWELL LESS 525 EL	322 V	0	322	0	100	0	0
2)REACTOR, AUX SRV WATER BLDG	*2992	23	1628	1	55	1452	8184
3)CONTROL RM HVAC SUPTS	105	0	49	• 0	47	0	336
TOTAL	3419	23	1999	1	58	1452	8520

CURRENT STATUS AND RESULTS

1. NONCONFORMANCE (NR) SUMMARY

AREA	NR'S WRITTEN	QUANTITY DEFICIENCIES IDENTIFIED	NUMBER DEFICIENCIES DISPOSITIONED	ACCEPT AS IS	REWORK	REJECT	REPAI
FOOT CONN.	253	975	242	139	88	0	15
DRYWELL	279	1825	314	10	306	4	3
CONTROL RM	173	2488	2331	17	2312	4	1
GALLERY STL	40	122	121	0	121	0	1
SWITCHGR 567'5	" 34	1953	1955	0	1933	0	2
SWITCHGR 546'	66	869	0	8	7	0	1
AUX BLD ROOF 5	91' 1	119	0	0	0	0	0
CABLE SPRD 546	82	78	-	-	. 🛥	-	-
HVAC	49	1117	-	-	-	-	-
TOTAL	977	9546	4963	174	4767	8	23

TASK I: (CONT'D)

- NOTE: MAJORITY OF THE DEFICIENCIES ATTRIBUTED TO: OVERLAP, UNDERSIZE WELD, UNDER-CUT, WELD PROFILE, LACK OF FUSION, AND INCORRECT INSTALLATION.
- 2. BEAMS BEING INSPECTED ARE COMPARED TO DESIGN DRAWINGS AND DESIGN DOCUMENT CHANGES AT PRESENT, THE BEAMS THAT HAVE BEEN INSPECTED, AND HAVE GONE THROUGH THE DESIGN, DOCUMENT REVIEW, APPEAR ON THE DESIGN DRAWING AND/OR A DESIGN DOCUMENT CHANGE (DDC).
- 3. COSMETIC REWORK HAS BEEN REQUIRED ON 70% OF THE BEAMS INSPECTED, TO OBTAIN RESULTS IN COMPLIANCE WITH AWS D1.1 1972 ACCEPTANCE CRITERIA.

COMMENTS

- 1. TASK I PROVIDED INSPECTORS TO PERFORM 152 INSPECTIONS FOR TASK IX ON ELEC-TRICAL CONDUIT SUPPORTS.
- 2. THE STOP WORK ORDER #82-02 STOPPING ALL WORK ON SCAFFOLDING, PAINT REMOVAL, ETC., HAS GREATLY IMPACTED THE TOTAL NUMBER OF INSPECTIONS COMPLETED THIS MONTH.
- 3. THE RANDOM SAMPLE INSPECTION OF 100 CONNECTIONS IN THE SERVICE WATER BUILDING IS IN PROGRESS. DEFICIENCIES NOTED THUS FAR ARE PRIMARILY COSMETIC IN NATURE. COMPLETION IS CONTINGENT UPON THE RESOLUTION OF ITEM 2 ABOVE.

PRESENT MANPOWER SUMMARY	ACTUAL
TASK COORDINATOR	1
QUALITY SPECIALISTS	3
INSPECTORS	15
DOCUMENT REVIEWERS	7
TOTAL	26

STATUS

THIS TASK IS APPROXIMATELY 58% COMPLETE.

ESTIMATED COMPLETION DATE

DECEMBER 1, 1982

TASK II: WELD QUALITY

ACTION BEING TAKEN

A. PERFORM A 100% REVIEW OF CODE PIPING KE-1 WELD DATA SHEETS TO DETERMINE WELD ROD HEAT NUMBERS, INSPECTION STAMPS AND DATES, IDENTIFY MISSED HOLD POINTS, AND MISSING OR ALTERED DOCUMENTATION.

B. VERIFY PROPER WELD PROCEDURE AND WELDER QUALIFICATION.

SUMMARY OF TASK

AREA	TOTAL # OF ITEMS	ITEMS COMPLETED THIS MONTH	ITEMS COMPLETED TO DATE	PERCENT COMPLETED THIS MONTH	PERCENT COMPLETED TO DATE	MANHOURS EXPENDED THIS MONTH	MANHOURS
1)STRUCT. KE-1 REV	11,000 IEW	0	11,000	0	98	0	416
2)SM BORE PIPE	32,000	0	25,841	0	79	*160	837
3)LG BORE PIPE	(APPX) 9,400 (380 PSK'S)	0	0	0	0	0	1373
4)WELD PRO- CEDURE REVIEH	91 (PROCEDURES) 121 REV'S	14	24	15	26	187	1413
5)WELDER QUAL. REVIEW	4600 (DOCUMENTS)	357	2456	7.5	53	80	1223
5)WELD FCD CONTROL		0	0	0	0	0	416
TOTAL	57,121	371	39,321	2	70	427	5678

1) *SEE ITEM #2 BELOW

CURRENT STATUS AND RESULTS

- 1. AREA #1 PRELIMINARY REPORT DRAFTED ON KE-1 REVIEW CER TO BE INITIATED ON AREAS OF CONCERNS.
- 2. AREA #2 *REWRITE OF PROCEDURE 19-QA-14 TO INCORPORATE A FINAL REVIEW OF DOC-UMENTS GENERATED, *MANHOURS EXPENDED IN THE REVISION OF 69 NR'S TO THIS TASK FOR CLARITY AND PROCEDURE REWRITE.
- 3. AREA #3 PROCEDURE TO BE WRITTEN, SEE TO BE RESOLVED, ITEM #1.

TASK II: (CONT'D)

4. AREA #4 24 CER'S HAVE BEEN GENERATED ON PROCEDURE'S REVIEWED.

.

5. AREA #6 PROCEDURE HAS NOT BEEN WRITTEN DUE TO LACK OF PERSONNEL - INVEST-IGATION TO BE STARTED NOVEMBER 1982 WITH PROCEDURE DEVELOPED AT THE SAME TIME FRAME.

TO BE RESOLVED

- 1. COORDINATION WITH HJK REVIEW FOR NPP-1 PROGRAM TO EFFECT A COORDINATED N-5 PROGRAM INCLUDING ITEM (1) IN COMMENTS SECTION.
- 2. CONFIRM ACCEPTABLE HEAT NUMBERS ESTABLISHED BY TASK III.
- 3. COMPLETION OF REWRITE OF 19-QA-14 TO SUPPLY DIRECTION FOR A SECOND REVIEW.

MANPOWER SUMMARY	ACTUAL
TASK COORDINATOR	1/2
INSPECTORS	1
DOCUMENT REVIEWERS	4
CLERKS	1/2
TOTAL	6

STATUS

THIS TASK IS APPROXIMATELY 70% COMPLETE.

ESTIMATED COMPLETION DATE

TO BE DETERMINED BY HJK SCHEDULE OF NPP-1/N-5 PROGRAMS (TO DATE, SCHEDULE NOT POSTED).

QUALITY CONFIRMATION PROGRAM

TASK III: HEAT NUMBER TRACEABILITY

ACTION BEING TAKEN

- A. PERFORM A 100% REVIEW OF SMALL BORE PIPING DOCUMENTATION AND CONDUCT A FIELD WALKDOWN OF THE SYSTEMS IDENTIFIED ON THE QCP LIST OF SAFETY-RELATED AND IMPORTANT TO SAFETY SYSTEMS.
- B. PERFORM A 100% INSPECTION AND/OR REVIEW OF DOCUMENTATION OF LARGE BORE PIPING FIELD MODIFICATIONS OF SYSTEMS ON THE SAFETY-RELATED AND IMPORTANT TO SAFETY SYSTEM LIST.
- C. PERFORM A 100% REVIEW OF ALL CODE AND STRUCTURAL PURCHASE ORDERS TO ESTABLISH A LIST OF ACCEPTABLE HEAT NUMBERS.
- D. PERFORM A REVIEW OF PURCHASE ORDERS FOR STRUCTURAL STEEL AND STEEL SHAPES TO DETERMINE IF PURCHASED ESSENTIAL OR NONESSENTIAL (THESE P.O.'S SHALL BE PART OF ACTION C). RESULTS OF THE REVIEW WILL BE USED TO ANSWER TASK 1 CONCERNS REGARDING ACCEPTABILITY OF STRUCTURAL STEEL.
- E. PERFORM INSPECTION AND REVIEW OF GAMMA PLUGS TO GAIN ADDITIONAL INFORMATION FOR EVALUATION AND DISPOSITION OF TOCFR50.55(e) REPORT M-56.

AREA	TOTAL # OF ITEMS	ITEMS COMPLETED THIS MONTH	ITEMS COMPLETED TO DATE	PERCENT COMPELTED THIS MONTH	PERCENT COMPLETED TO DATE	MANHOURS EXPENDED THIS MONTH	MANHOU REMAIN
A)SM BORE DOC	2691 ISK	27	52	1	2	226	2540
SM BORE WLKDWN	2691 ISK	0	2691	0	100	0	0
B)LG BORE	(APPX) 380 PSK	0	0	0	0	0	74,600
LG BORE WLKDWN	(APPX) 380 PSK	0	0	0	0	0	
C)PURCHASE ORDER REVW	2688	26	44	1.0	1.6	215	2951
D)HJK STRCTL STEEL PO'S ESS/NONESS	1900	0	1900	0	95	0	4C
E)GAMMA PLUG 5D.55(e)M-5	404 56	179	219	44	54	223	209
TOTAL	11,134	232	4906	· 3	36	664	80,340

SUMMARY OF TASK

(5)

TASK III: (CONT'D)

- 1) *TOTAL MANHOURS SHOWN IS FOR BOTH LARGE BORE DOCUMENTATION AND WALKDOWN.
- 2) **INCLUDES STRUCTURAL STEEL PO'S IDENTIFIED IN ACTION D. TOTAL NUMBER OF ITEMS CHANGED DUE TO CHANGE IN METHOD OF ACCOUNTING, I.E. PURCHASE ORDERS RATHER THAN SHIPMENTS RECEIVED.
- 3) ***5% REMAINING TO COMPLETE IS TO PERFORM FINAL REVIEW.

CURRENT STATUS AND RESULTS

- 1. HJK IS VALIDATING CMTR'S FOR STRUCTURAL ITEMS RECEIVED ON NONESSENTIAL PO'S THROUGH USER TESTS OR QUALIFYING SUPPLIER. SIMILAR REVIEW AND ACTION BEING COMPLETED FOR ASME PIPING COMPONENTS.
- 2. 1900 STRUCTURAL STEEL PURCHASE ORDERS HAVE BEEN IDENTIFIED, OF THIS NUMBER, 1142 WERE ORDERED ESSENTIAL AND 758 WERE ORDERED NONESSENTIAL.
- 3. 219 OF 404 GAMMA PLUGS HAVE BEEN INSPECTED. THE DATA HAS BEEN SUBMITTED TO CG&E NUCLEAR ENGINEERING DEPARTMENT. ACTION ON THIS ITEM IS STOPPED PENDING NED EVALUATION.

TO BE RESOLVED

1. A METHOD OF SURVEILLING THE H.J. KAISER N-5 PROGRAM IS BEING RESEARCHED TO ADDRESS ACTION B.

COMMENTS

۰.

• •

1. THE FIGURE SHOWN FOR LARGE BORE REFLECTS AN APPROXIMATE NUMBER OF MANHOURS NECESSARY FOR THE QCP TO PERFORM THE LARGE BORE WALKDOWN AND DOCUMENTATION REVIEW. THIS FIGURE WOULD BE MUCH LESS IF HJK PERFORMED THIS FUNCTION AND QCP PERFORMED A SURVEILLANCE OF THEIR ACTIVITY.

MANPOWER SUMMARY	ACTUAL
TASK COORDINATOR	1
TASK LEAD	1
INSPECTORS	1
DOCUMENT REVIEWERS	2
TOTAL	5

TASK III: (CONT'D)

STATUS

۰.

.

THIS TASK IS 36% COMPLETE.

ESTIMATED COMPLETION DATE

TO BE DETERMINED AFTER COORDINATION WITH HJK AND CG&E DOCUMENTATION REVIEW GROUP.

.....

•

QUALITY CONFIRMATION PROGRAM

TASK IV: SOCKET WELD DISENGAGEMENT

ACTIONS BEING TAKEN

٠,

- A. IDENTIFY SMALL BORE PIPING SOCKET WELDS FOR WHICH VERIFICATION FOR DISEN-GAGEMENT DOES NOT EXIST.
- B. RADIOGRAPH 100% OF THE ACCESSIBLE WELD NOT HAVING VERIFICATION OF DISEN-GAGEMENT.

SUMMARY OF TASK

AREA	TOTAL # OF ITEMS	ITEMS COMPELTED THIS MONTH	ITEMS COMPLETED TO DATE	PERCENT COMPLETED THIS MONTH	PERCENT COMPLETED TO DATE	MANHOURS EXPENDED THIS MONTH	MANHOUR REMAINI
IV SOCKET WELD DISENGAGE- MENT	32,000	0	29,821	0	98	*90	240

CURRENT STATUS AND RESULTS

- A. 348 RADIOGRAPHS TO DATE HAVE BEEN TRANSMITTED TO QADVG.
- B. 111 SOCKET WELDS HAVE BEEN REJECTED FOR LACK OF DISENGAGEMENT.
- C. 695 WELDS HAVE BEEN IDENTIFIED TO DATE AS LACKING EVIDENCE OF A QUALITY INSPECTION.
- D. REJECT WELD HAVE BEEN IDENTIFIED ON NONCONFORMANCE REPORTS.

COMMENTS

1. MANHOURS EXPENDED THIS TASK DUE TO CER 82-240 WRITTEN ON 19-QA-02 FOR NON-COMPLIANCE OF PROCEDURE. RESULTS: A REWRITE OF 19-QA-02 FOR CLARITY, DIR-ECTION AND TO INITIATE A SECOND REVIEW OF DOCUMENTS GENERATED BY TASK IV. IMPACT OF SECOND REVIEW HAS NOT BEEN DETERMINED TO DATE, PROCEDURE WAS REVIEWED BY THE REVIEW BOARD AND COMMENTS ADDED 10-26-82.

MANPOWER SUMMARY	ACTUAL
TASK COORDINATOR	1/2
CLERKS	1/2
TOTAL	1

(8)

QUALITY CONFIRMATION PROGRAM

TASK IV: (CONT'D)

STATUS

THIS TASK IS 98% COMPLETE. INITIAL REVIEW/CER 82-240.

ESTIMATED COMPLETION DATE

DECEMBER 1, 1982 PENDING APPROVAL OF THE H. THIELSCH REPORT. AN ESTIMATED COMPLETION DATE FOR THE SECOND REVIEW WILL BE ADDRESSED AFTER IMPLIMENTATION OF 19-QA-02.



- 14

TASK V: RADIOGRAPHS

ACTION TO BE TAKEN

A. CONFIRM THAT THE EXISTING RADIOGRAPHS OF LARGE BORE PIPING SUPPLIED BY M. W. KELLOGG ARE ADEQUATE TO IDENTIFY WELD DEFICIENCIES.

SUMMARY OF TASK

AREA	TOTAL OF ITEMS	#	ITEMS COMPLETED THIS MONTH	ITEMS COMPLETED TO DATE	PERCENT COMPLETED THIS MONTH	PERCENT COMPLETED TO DATE	MANHOURS EXPENDED THIS MONTH	MANHOURS REMAININ
RADIOGRAPHY REVIEW	4250		0	4250	0	100	0	0
RADIOGRAPHS	46		4	46	27.3	100	40	40
TOTALS	4296	٠	4	4296	1	99	40	40

CURRENT STATUS AND RESULTS

- A. REVIEW OF 4250 WELD RADIOGRAPHS FOR SENSITIVITY IS COMPLETE.
- B. 46 WELDS HAVE BEEN IDENTIFIED TO BE RE-RADIOGRAPHED, COVERING ALL VARIATION IN PIPE AND WALL THICKNESS.
- C. PROGRAM TO CONFIRM RADIOGRAPHS HAS BEEN APPROVED BY THE NATIONAL BOARD OF BOILER PRESSURE VESSEL INSPECTORS AND THE STATE OF OHIO.
- D. 46 OF THE 46 RADIOGRAPHS HAVE BEEN RE-RADIOGRAPHED, AND ACCEPTED.
- E. MOCK-UPS OF THE 4 INACCESSABLE WELDS WERE FABRICATED AND RADIOGRAPHED.

TO BE RESOLVED

COMMENTS

A FINAL REPORT FOR TASK V IS BEING WRITTEN AND WILL BE PRESENTED TO ALL APPLICABLE PARTIES.

MANPOWER SUMMARY

PERSONNEL ASSIGNED TO TASK III WILL SUPPORT THIS ACTIVITY.

STATUS

OVERALL TASK IS APPROXIMATELY 99% COMPLETE.

ESTIMATED COMPLETION DATE

NOVEMBER 15, 1982

QUALITY CUNFIRMATION PROGRAM

TASK VI: CABLE SEPARATION

ACTION BEING TAKEN

- A. 100% INSPECTION OF SLEEVES AND CLASS IE FLOOR PENETRATIONS FOR SEPARATION IDENTIFICATION AND ROUTING.
- B. INSPECT A MINIMUM OF 10% OF THE ASSOCIATED CABLES TO ARRIVE AT A 95% CONFI-DENCE LEVEL THAT 95% OF THE ASSOCIATED CABLES ARE PROPERLY SEPARATED IN TRAYS AND CONDUITS. CONFIDENCE LEVEL WAS NOT MET. THIS INSPECTION IS SUPERSEDED BY THE INSPECTION OF CABLE TRAYS.
- C. 100% INSPECTION OF CABLE TRAYS IN CATEGORY 1 STRUCTURES.
- D. 100% INSPECTION OF CABLES REQUIRING SEPARATION INSIDE PANELS AND TO THE FIRST ASSIGNED RACEWAY TO VERIFY PROPER SEPARATION AND IDENTIFICATION. THIS INCLUDES ALL CABLES INSTALLED BETWEEN THE CABLE SPREADING ROOM AND THE MAIN CONTROL ROOM.
- E. TRACK RESOLUTION OF SIX SAMPLES OF FAILURE TO MEET CABLE SEPARATION CRITERIA IDENTIFIED BY THE NRC.

SUMMARY OF TASK

	TOTAL # OF ITEMS	ITEMS COMPLETED THIS MONTH	ITEMS COMPLETED TO DATE	PERCENT COMPLETED THIS MONTH	PERCENT COMPLETED TO DATE	MANHOURS EXPENDED THIS MONTH	MANHOUR REMAINI
AREA	11615						••
WALL PENETRA- TION	182	0	182	0	100	0	32
SLEEVES	R)						
ASSOC.	392	0	392	0	100	0	0
CABLES				15.3	AA Q	538.5	8172
TRAY SYS. WLKDWN	1778(TR) (12446 UN)	1903(UN)	5589(UN)	12.3	44.5	500.0	
CABLES RE- QUIRING SEP. IN- SIDE PANEL	- 8951 CABI (35,804 UN) LS	LES O	0	0	0	217	6357
DEVIEW OF	0)	0	37	0	100	0	0
EER RESPO	NSES					500 F	1592
ADMINIS-	-	-	-	-	-	589.5	* 5 5 5 6
TRATIVE TOTAL	48,824	1903	6200	6	50	1345	16,153
(TR = TRA (UN = UNI	YS) TS)			(11)			
TASK VI: (CONT'D)

CURRENT STATUS AND RESULTS

- A. A TOTAL OF 776 NONCONFORMANCE REPORTS HAVE BEEN WRITTEN FOR SEPARATION, IDEN-TIFICATION AND ROUTING DEFICIENCIES. 345 NR'S DISPOSITIONED (131 REWORK, 25 REPAIR, 189 ACCEPT-AS-IS). 217 OF THE 345 DISPOSITIONED NR'S HAVE BEEN CLOSED. 12 NR'S WERE CANCELLED BECAUSE OF DUPLICATION. 419 NR'S HAVE NOT BEEN DISPOSITIONED.
- B. WALL PENETRATION AND SLEEVE INSPECTIONS COMPLETE. LEVEL II REVIEW OF INSPEC-TION RECORDS REMAINING.
- C. THE DECISION HAS BEEN MADE TO INSPECT CABLE TRAYS IN PLACE OF CLOSING OUT ASSOCIATED CABLE INSPECTION RECORDS SINCE CONFIDENCE LEVEL HAS NOT BEEN MET. THE PROCEDURE FOR TRAY INSPECTION 19-QA-38 WAS APPROVED ON AUGUST 24, 1982. THE CABLE TRAY INSPECTIONS FOR THE CONTROL ROOM AND THE SERVICE WATER PUMP STRUCTURE ARE COMPLETE AND INSPECTIONS ARE BEING PERFORMED IN THE REACTOR BUILDING.
- D. THERE ARE 4210 CABLES THAT REQUIRE .INSPECTION IN THE CONTROL ROOM AND FROM THE CABLE SPREADING ROOM TO THE CONTROL ROOM. INSPECTIONS CANNOT START UNITL THE SEPARATION REQUIREMENTS ARE FINALIZED.
- E. STATUS OF THE SIX EXAMPLES OF LACK OF SEPARATION: (2) CORRECTED, (1) IN-PROCESS OF BEING REWORKED, (2) ACCEPTED BY ENGINEERING, AND (1) IDENTIFIED ON NR.

COMMENT

MEETINGS WERE HELD WITH S&L TO CLARIFY AND RESOLVE SEPARATION REQUIREMENTS.

MANPOWER SUMMARY	ACTUAL
TASK COORDINATOR	1
QUALITY ENGINEER	1
LEAD INSPECTOR	1
INSPECTORS	4 1/2
DOCUMENTATION (NR'S & SCOPING	1 1/2
PKGS)	
TOTAL	9

(12)

QUALITY CONFIRMATION PROGRAM

TASK VI: (CONT'D)

٠

<u>STATUS</u>

. '

٠,

٠

50% COMPLETE (EXPANDED SCOPE MANHOURS).

ESTIMATED COMPLETION DATE

.

JUNE 1, 1983

QUALITY CONFIRMATION PROGRAM

TASK VII: NONCONFORMANCE REPORTS

ACTION BEING TAKEN

- A. 100% REVIEW OF ALL VOIDED NONCONFORMANCE REPORTS (NR'S), SURVEILLANCE REPORTS (SR'S), PUNCHLIST AND EXCEPTION LIST ITEMS.
- B. 100% SOLICITATION TO PAST AND PRESENT QC INSPECTORS, BY CERTIFIED MAIL, REQUESTING THEIR KNOWLEDGE OF ANY NONCONFORMANCE REPORTS NOT ENTERED INTO THE SYSTEM.
- C. REVIEW 300 RANDOMLY SELECTED, CLOSED NR'S.

SUMMARY OF TASK

AREA	TOTAL # OF ITEMS	ITEMS COMPLETED THIS MONTH	ITEMS COMPLETED TO DATE	PERCENT COMPLETED THIS MONTH	PERCENT COMPLETED TO DATE	MANHOURS EXPENDED THIS MONTH	MANHOURS REMAINING
VOIDED NR REVIEW INITIAL	1318	100	1050	7.58	79.66	125	539
VOIDED REVIEW SECONDARY	1318	64	656	4.85 •	49.77	80	565
SURVEILL- ANCE RPT REVIEW	3500	0	3285	0	94	0	5430
PUNCHLIST REVIEW	(APPX) 25,000	0	0	0	0	10	1190
CLOSED NR REVIEW	300	0	50	0	16.66	0	145
NR'S FROM INSPECTORS	UNKNOWN	0	0	(NO NR'S F	RECEIVED THIS	5 MONTH)	
TOTAL	31,436	0	5,041	0	61	215	7869

CURRENT STATUS AND RESULTS

 A. OF THE 100 NR'S REVIEWED, 50 HAVE BEEN SENT TO HJK FOR ADDITIONAL INFORMATION/ BACK-UP DOCUMENTATION AND 50 ARE READY FOR TRANSMITTAL UPON RECEIPT OF ORIGINAL 50 (EXPECTED 10/26; LETTER FOR ADDITIONAL 50 FOR SIGNATURE).

B. NO NO COMMITTEE HELD THIS MONTH.

(14)

TASK VII: CONT'D.

- C. TWENTY MAN HOURS WERE EXPENDED THIS PERIOD FOR DISCUSSION AND PREPARATION OF PROCEDURES TO CLOSE OUT PUNCHLIST ITEMS.
- D. PREPARATION OF PROCEDURE (19-QA-20 REVIEW OF PUNCHLIST) IN PROGRESS AS OF THIS DATE. PROCEDURE AND FINAL REVIEW CYCLE TO BE COMPLETED BY NOVEMBER 15, 1982.
- E. 58 NR'S HAVE BEEN GENERATED FROM THE REVIEW OF VOIDED HJK NR'S AND SURVEILL-ANCE REPORTS.

MANPOWER SUMMARY	ACTUAL
TASK COORDINATOR	1
DOCUMENT REVIEWERS	_2
TOTAL	3

STATUS

THIS TASK IS APPROXIMATELY 61% COMPLETE.

ESTIMATED COMPLETION DATE

.

JANUARY 30, 1983

QUALITY CONFIRMATION PROGRAM

TASK VIII: DESIGN CONTROL AND VERIFICATION

ACTION BEING TAKEN

VERIFY ADEQUACY OF S&L DESIGN CONTROL AND VERIFICATION PROGRAM. PROCEDURES WERE IN PLACE AND WERE CLARIFIED.

CURRENT STATUS AND RESULTS

THE FINAL REPORT FOR TASK VIII IS BEING WRITTEN.

STATUS

THIS TASK IS 99% COMPLETE TO DATE.

ESTIMATED COMPLETION DATE

DECEMBER 15, 1982

•

٠

•

QUALITY CONFIRMATION PROGRAM

TASK IX: DESIGN DOCUMENT CHANGES

ACTION BEING TAKEN

- 1. COMPILE A COMPUTER LISTING OF ALL DDC'S.
- 2. REVIEW CG&E, HJK, S&L AND OTHER SITE CONTRACTORS' DDC'S TO DETERMINE PROPER CLASSIFICATION.
- 3. REVIEW CG&E, HJK, S&L AND OTHER SITE CONTRACTOR'S ESSENTIAL DDC'S TO DETER-MINE IF DDC'S WERE INCORPORATED IN INSPECTION DOCUMENTATION.
- 4. PERFORM ELECTRICAL INSPECTIONS IN THE CONTROL ROOM FOR CABLE TRAY HANGERS, CONDUIT, AND CONDUIT SUPPORTS. (ADDITION TO ORIGINAL QCP WORK SCOPE)

AREA	TOTAL # OF ITEMS	ITEMS COMPLETED THIS MONTH	ITEMS COMPLETED TO DATE	PERCENT COMPLETED THIS MONTH	PERCENT COMPLETED TO DATE	MANHOURS EXPENDED THIS MONTH	MANHOURS REMAINING
PHASE I	19,711	0	18,725	0	95	0	80
PHASE II	9,111	227	3,303	2.5	36.3	551.5	13,858.5
PHASE IA	2,600	190	754	7.3	29	20	146
PHASE IIA	2,200	68	167	· 3.1	7.6	228	5,030.5
FHASE III DOCUMENTAT	ION 714	24	124*	3.3	17.3	45.0	-
INSPECTION	714	79	98*	11	13.7	491.5	-
OTHER	-	-	-	-	-	1017.0	-
TOTAL	35,050	-	23,171	2	35	2353	19,115

SUMMARY OF TASK

- ***** EQUIVALENT ITEMS
- ** OTHER TIME SPENT FOR TRAINING AND CERTIFICATION OF PERSONNEL, PROBLEM RESEARCH AND RESOLUTION, PROCEDURE PREPARATION, CLERICAL, PHASE II ADMINISTRATION, AND SETTING UP A DDC REVIEW AND HANDLING SYSTEM. (386.5 HRS. FOR DDC SYSTEM)
- PHASE I CLASSIFICATION OF CG&E, S&L AND HJK, DDC'S AS ESSENTIAL OR NON-ESSENTIAL.
- PHASE II REVIEW INSPECTION DOCUMENTATION TO DETERMINE IF CG&E, S&L AND HJK DDC'S HAVE BEEN INCORPORATED AND INSPECTED.
- PHASE IA CLASSIFICATION OF WY&B AND OTHER SITE CONTRACTOR DDC'S AS ESSENTIAL OR NON-ESSENTIAL.
- PHASE 11A- REVIEW INSPECTION DOCUMENTATION TO DETERMINE IF WY&B AND OTHER SITE CONTRACTOR DDC'S HAVE BEEN INCORPORATED AND INSPECTED.
- PHASE III- ELECTRICAL INSPECTIONS IN THE CONTROL ROOM.

(17)

TASK IX: (CONT'D)

RESULTS

- 1. IN PHASES II & IIA, 760 DEFICIENCIES IDENTIFIED TO DATE FOR REPORTING VIA DOCU-MENT DEFICIENCY NOTICES OR NONCONFORMANCE REPORTS AS APPLICABLE.
- 2. IN PHASE III, 95 DEFICIENCIES HAVE BEEN IDENTIFIED TO DATE FOR REPORTING VIA NONCONFORMANCE REPORTS. A MAJORITY OF THESE DEFICIENCIES ARE DUE TO WELD DEFICIENCIES.
- 3. TASK IX GENERATED THE FOLLOWING CER'S DURING THE PAST MONTH:
 - i) DOCUMENTATION FOR MISCELLANEOUS HVAC INSTALLATION
 - ii) RESPONSE TIME FOR CER'S
 - 111) DOCUMENTATION FOR INSPECTION OF GE MECHANICAL FDDR'S AND FDI'S
 - iv) CABLE TRAY HANGER WELDS (VENDORS SHOP VS. FIELD)
 - v) CONDUIT HANGER DEADWEIGHT CALCULATIONS
 - vi) CONDUIT HANGER DIMENSION TOLERANCES
 - vii) NR'S WITH CONDITIONAL DISPOSITIONS

PROMBLEM/DEFICIENCY		ELECT.	•	MECH.	STRUCT.	TOTAL
1.	MISSING DOCUMENTATION	32			203	235
2.	DDC MIS-CLASSIFICATION	13			3	16
3.	INSPECTION PROGRAM DEFICIENCIES & MISC.	26			32	58
4.	DDC NOT INCORPORATED IN DOCUMENTATION	101		10	307	418
5.	INSPECTION PRIOR TO DDC BEING WRITTEN (AS-BUILT)	3				3
6.	INCOMPLETE INSPECTION	<u>8</u>			22	
	TOTAL	183		10	567	760

TO BE RESOLVED

- A. TASK IX IS STILL WAITING FOR HJK TO RESPOND TO CAR'S IN ORDER TO COMPLETE SOME DDC REVIEWS.
- B. THE METHOD USED TO CALCULATE CONDUIT DEAD LOADS, AS CONTAINED IN THE E-189 SERIES GENERAL NOTES, MAY BE INADEQUATE. THIS IS BEING ADDRESSED BY CG&E QUALITY ENGINEERING.

TASK IX: (CONT'D)

- C. THE WELD INSPECTORS HAVE STATED THAT THEY CANNOT DETERMINE WHICH WELDS ON TRAY SUPPORTS WERE MADE BY THE SMAW PROCESS AND WHICH WERE MADE BY THE MIG PROCESS AFTER THE WELDS HAVE BEEN HOT-DIPPED GALVANIZED. THERE IS PRESENTLY NO WAY TO DETERMINE WHICH SUPPORTS WERE FABRICATED BY THE VENDOR AND WHICH WERE FABRICATED HERE ON SITE. THE LATTER REQUIRE INSPECTION OF THE "SHOP" WELDS. THIS PROBLEM IS BEING ADDRESSED BY CG&E QUALITY ENGINEERING, PER CER WRITTEN BY TASK IX.
- D. THERE ARE SEVERAL HUNDRED DDC'S OPEN AGAINST THE E-189 DRAWINGS, OF WHICH MORE THAN 150 APPLY TO THE CONTROL ROOM AREA. A SYSTEM IS CURRENTLY BEING DEVELOPED TO ASSURE THAT ALL APPLICABLE DDC'S ARE INCLUDED IN THE INSPECTION PACKAGES AND PROVIDE FOR VERIFICATION OF THAT FACT.
- E. A LARGE NUMBER OF CONDUIT SUPPORTS IN THE MAIN CONTROL ROOM STILL HAVE OPEN WORK TICKETS FOR LINE REVISIONS. THE PRESENT HJK PROGRAM ONLY REQUIRES INSPEC-TION OF THE ACTUAL REVISION, AND NOT THE ENTIRE SUPPORT. THEREFORE, SINCE THESE SUPPORTS HAVE BEEN PREVIOUSLY ACCEPTED, THEY STILL FALL UNDER 19-QA-35. THIS PROBLEM HAS NOT YET BEEN ADDRESSED, BUT WILL REQUIRE RESOLUTION BEFORE ALL OF THE SUPPORTS CAN BE COMPLETED.
- F. TASK IX IS GOING TO PARTICIPATE ON A STRUCTURAL DDC NR REVIEW BOARD WHICH IS GOING TO BE FORMED TO FACILITATE THE DISPOSITIONING OF DOCUMENTATION DEFICIEN-CIES. THIS REVIEW BOARD NEEDS TO BE FORMED.
- G. WAITING FOR RESOLUTION OF PROBLEMS WITH X & Y DIMENSION TOLERANCES ON CONDUIT HANGERS.
- H. WAITING FOR RESOLUTION/CLARIFICATION OF HANGER SEPARATION CRITERIA.

COMMENTS

 PHASE III OF TASK IX REQUIRES MANPOWER ELECTRICAL INSPECTORS IN ORDER TO ACHIEVE COMPLETION BY THE DATE INDICATED BELOW. A TOTAL OF 14 PEOPLE WILL BE REQUIRED BY NOVEMBER 1, 1982 IN ORDER TO MEET THE COMPLETION DATE.

MANPOWER SUMMARY	PHASES I, II, IA, IIA
ENGINEERS (INCLUDING TASK COORDINATOR)	4
DOCUMENT REVIEWERS	2
CLERKS	<u> </u>
TOTAL	7
	PHASE III
DOCUMENT REVIEWERS	4 **
INSPECTORS	
TOTAL	7 (DOES NOT INCLUDE WELD INS
** NUMBER INCLUDES PHASE COORDINATOR & LE CERTIFIED AS ELECTRICAL INSPECTORS.	AD INSPECTOR. TWO DOCUMENT REVIEWERS ARE A
(19)	

TASK IX: (CONT'D)

STATUS

•

•

PHASE I 95% COMPLETE
PHASE II 36.3% COMPLETE
PHASE IA 29.0% COMPLETE
PHASE IIA 7.6% COMPLETE
PHASE III 10% COMPLETE (ITEMS COMPLETED)
THE ORIGINAL TASK (PHASES I & II) IS APPROXIMATELY 40.8% COMPLETE.
PHASES IA AND IIA ARE APPROXIMATELY 9.2% COMPLETE.
THE OVERALL TASK IS 35% COMPLETE (DOES NOT INCLUDE PHASE III)

ESTIMATED COMPLETION DATE

PHASES I, II, IA, IIA - JUNE 1983 PHASE III ' - JANUARY 1983 *

* BASED ON RESOLUTION OF ITEMS "B", "C", "D", "E", "G" & "H" BY NOVEMBER 15, 1982.

٠

•

QUALITY CONFIRMATION PROGRAM

TASK X: SUBCONTRACTOR/QA PROGRAM

ACTION BEING TAKEN:

- A. ALL CGLE QA AUDITS OF SUBCONTRACTORS/VENDORS HAVE BEEN IDENTIFIED.
- B. MATRICES HAVE BEEN DEVELOPED SHOWING WHICH 10CFR50 APPENDIX B CRITERIAS WERE VERIFIED DURING THESE AUDITS.
- C. AUDIT REPORTS ARE BEING REVIEWED AND SUMMARIZED ON INDIVIDUAL FORMS.
- D. EVALUATIONS ARE MADE OF THE SCOPE AND DEPTH OF THE AUDITS TO DETERMINE WHETHER THESE AUDITS COVERED ALL APPLICABLE 10CFR50, APPENDIX B CRITERIAS IN SUFFICIENT DETAIL.
 - 1. IF EVALUATIONS PROVE SUBCONTRACTORS PROGRAMS ACCEPTABLE THESE CAN BE CLOSED.
 - 2. IF EVALUATIONS CANNOT PROVIDE EVIDENCE THAT SUBCONTRACTOR PROGRAMS WERE SATISFACTORY, ALTERNATE MEASURE TO VERIFY HARDWARE INTEGRITY, SUCH AS REVIEW OF INTERNAL AUDITS PERFORMED BY SUBCONTRACTORS, REVIEW OF QUALITY DOCUMENTATION (INSPECTION RESULTS) TO PROVE ADEQUATE PROGRAM COVERAGE, COVERAGE IN OTHER QCP TASKS, RE-AUDIT OR HARDWARE INSPECTION AND/OR TESTING, WILL BE PURSUED.

SUMMARY OF TASK

AREA	TOTAL # OF ITEMS	ITEMS COMPLETED THIS MONTH	ITEMS COMPLETEE TO DATE	PERCENT COMPLETED THIS MONTH	PERCENT COMPLETED TO DATE	MANHOURS EXPENDED THIS MONTH	MANHOURS REMAINING
SUBCONTRAC QA PROGRAM	TOR 80	0	80	3	78*	20	130

* REMAINING 22% TO COMPLETE EVALUATION

CURRENT STATUS AND RESULTS

- A. PROCEDURE FOR INTERFACE WITH OTHER QCP TASKS TO VERIFY ADEQUACY OF WORK HAS BEEN DEVELOPED. THIS PROCEDURE IS BEING REWRITTEN AND WILL BE SUMITTED FOR REVIEW.
- B. EIGHTY (80) AUDITS OF SUBCONTRACTORS WERE IDENTIFIED AS BEING PERFORMED PRIOR TO APRIL 8, 1981. THESE AUDITS ENCOMPASSED THIRTEEN(13) DIFFERENT SUBCONTRACTORS. OF THE THIRTEEN SUBCONTRACTORS, SIX (6) MAY REQUIRE AUDITS OF SUBSEQUENT ACTIVITIES FOR CONFIRMATION OF THE WORK WHICH CANNOT BE ACCOMPLISHED BY DOCUMENT REVIEW OR INSPECTION. ALL AUDITS HAVE BEEN REVIEWED. THE SCOPE OF MANY AUDITS DO NOT ADDRESS APPLICABLE CRITERIA.

TASK X: CONT'D.

C. EVALUATIONS OF THE AUDITS WILL BEGIN WHEN THE PROCEDURE FOR THIS TASK IS APPROVED.

TO BE RESOLVED

COMMENTS

- A. THE ESTIMATED TIME REQUIRED FOR COMPLETION IS 15 MAN DAYS WHICH INCLUDES EVALUATION OF PAST AUDITS OF SIX SUBCONTRACTORS.
- B. IF THE SCOPE AT THIS TASK IS EXPANDED TO INCLUDE VENDORS WHICH SUPPLIED MATERIAL BUT HAVE NOT BEEN AUDITED, IT IS ESTIMATED THAT A MINIMUM OF 10 MAN DAYS PER VENDOR WILL BE REQUIRED TO COMPLETE AN AUDIT.

STATUS

THIS TASK IS 78% COMPLETE.

ESTIMATED COMPLETION DATE

DECEMBER 1982 FINAL REPORT AND EVALUATION.

TASK XI: AUDITS

ACTION BEING TAKEN

A. REVIEW ALL PAST CG&E QA AUDITS OF HJK, S&L, GE, NPD, NED, EOTD, AND GCD.

B. DEVELOP MATRICES SHOWING WHICH 10CFR50 APPENDIX B CRITERIA WERE IDENTIFIED DURING AUDITS.

SUMMARY

	TOTAL # OF	ITEMS COMPLETED THIS MONTH	ITEMS COMPLETED TO DATE	PERCENT COMPLETED THIS MONTH	PERCENT COMPLETED TO DATE	MANHOURS EXPENDED THIS MONTH	MANHOURS REMAINING
AUDITS REVIEWED	296	0	296	2	82	40	210

CURRENT STATUS AND RESULTS

- A. PRELIMINARY EVALUATION INDICATES THAT MOST AUDITS WERE OF LIMITED SCOPE, HOWEVER, COLLECTIVELY DUE TO THE LARGE NUMBER OF AUDITS PERFORMED, THE APPLICABLE 18 CRITERIA WERE COVERED FOR HJK, EOTD, NED, GCD, AND NPD. THERE WERE AREAS NOT COVERED IN AUDITS OF S&L AND GE SUBSEQUENT TO APRIL 8, 1981. HOWEVER, CURRENT AUDITS OF S&L AND GE HAVE COVERED THE MAJORITY OF APPLICABLE CRITERIA.
- B. PROCEDURE FOR INTERFACE WITH THE QCP TASKS TO VERIFY ADEQUACY OF WORK HAS BEEN DEVELOPED. THIS PROCEDURE IS BEING REWRITTEN AND WILL BE SUBMITTED FOR REVIEW.

COMMENTS

A. IT IS ESTIMATED THAT 30 MAN DAYS WILL BE REQUIRED FOR THE EVALUATION OF THESE AUDITS.

STATUS

THIS TASK IS 82% COMPLETE

ESTIMATED COMPLETION DATE

OCTOBER 15, 1982 - REVIEWS COMPLETE NOVEMBER 15, 1982 FINAL REPORT AND EVALUATIONS.

UNITED STATES OF AMERICA NUCLEAR REGULATORY COMMISSION

CCMMISSIONERS:

Nunzio J. Palladino, Chairman Victor Gilinsky John F. Ahearne Thomas M. Roberts James K. Asselstine

*82 NOV 12 P4:50

i l'AETEI

BLUKE KUNK SERVICE ERMICH

SERVED NOV 121982

In the Matter of

t"

211160392/82111

PDR ADOCK

CINCINNATI GAS & ELECTRIC COMPANY ·

(William H. Zimmer Nuclear Power Station) Docket Nc. 50-358 Construction Permit No. CPPR-88 EA 82-129

ORDER TO SHOW CAUSE AND ORDER IMMEDIATELY SUSPENDING CONSTRUCTION (CLI-82-33)

The Cincinnati Gas and Electric Company (CG&E) holds Construction , Permit No. CPPR-88 which was issued by the Commission in 1972. The permit authorizes the construction of the William H. Zimmer Nuclear Power Station Unit 1, a boiling water reactor to be used for the commercial generation of electric power. The Zimmer plant is located on the licensee's site in Moscow, Ohio.

II.

A. Initial Identification of OA Problems

In early 1981 the NRC conducted an investigation into allegations made by present and former Zimmer site employees and by the Government Accountability Project. The NRC investigation revealed a widespread breakdown in CG&E's management of the Zimmer project as evidenced by numerous examples of non-compliance with twelve of the eighteen quality assurance Criteria cf

Attachment III

NOV 2 2 1982

Appendix B to 10 CFR Part 50. Consequently, CG&E paid a civil penalty of \$200,000 for the failure to implement an acceptable quality assurance program, false quality assurance documents, and intimidation and harassment of quality control inspectors. (See Notice of Violation and Proposed Imposition of Civil Penalties, dated November 24, 1981 and Investigation Report No. 50-358/81-13.) In addition CG&E agreed to take actions to correct identified QA failures and prevent their recurrence and to determine quality of completed construction work.

1. Actions to Correct Identified QA Failures and Prevent Recurrence

A meeting was conducted by Region III on March 31, 1981, and the utility agreed to implement ten actions to correct quality assurance failures identified during the January - March 1981 investigation and to preclude their recurrence. These actions included: (1) increasing the size and technical expertise of the CGSE QA organization; (2) taking action to assure independence and separation of the QA/QC function performed by Kaiser from the construction function; (3) conducting 100% reinspections of the quality control (QC) inspections performed after that date by Kaiser and other contractors; (4) reviewing for adequacy, and revising as appropriate, all QC inspection procedures; (5) training QA/QC personnel on new and revised procedures; (6) reviewing for adequacy, and revising as appropriate, the procedures governing the identification, reporting, and resolution of deviations from codes and Final Safety Analysis Report (FSAR) statements; (7) reviewing for adequacy the procedures governing nonconformance reporting and justifying the disposition of each voided nonconformance

- 2 -

report; (2) establishing an adequate program for control of QA and QC records; (9) performing a 100% review of all future surveillance and nonconformance reports written by contractor personnel; and (10) reviewing and revising the CG&E audit program so that it included technical audits of construction work and more comprehensive and effective programmatic audits. These commitments were confirmed in an Immediate Action Letter to the licensee on April 8, 1981.

2. Actions to Determine Quality of Completed Construction Work

m

Following the identification in 1981 of significant quality assurance problems and related management breakdowns, CG&E agreed to establish a comprehensive program to determine the quality of the completed construction work. The Quality Confirmation Program (QCP) was submitted to the NRC by the licensee on August 21, 1981. The QCP addressed problems identified by the investigation in the following areas: (1) structural steel; (2) weld quality; (3) traceability of heat numbers on piping; (4) socket weld fitup; (5) radiographs; (6) electrical cable separation; (7) nonconformance reports; (8) design control and verification; (9) design document changes; (10) subcontractor QA programs; and (11) audits.

١.

3. <u>Results of Actions Taken by the Licensee to Determine the Guality of</u> Completed Construction Work

Many construction deficiencies have been identified by the licensee during the conduct of the QCP and other quality reviews and reported to

- 3 -

the NRC pursuant to 10 CFR 50.55(e) which could have been prevented or identified in a timely manner by the licensee and its contractors had there been a properly managed QA program. Major construction deficiencies identified to date by the quality reviews are listed in order of identification and include the following:

Welds performed using an unqualified welding procedure for welds greater \ than 0.864 inches.

Unauthorized stamping of fittings and use of "high-stress" stamps.

. ASME structural weld and welder qualification deficiencies.

. Welds performed and welders not qualified for weld thickness range per ASME requirements.

Approximately 2400 feet of small bore piping identified with questionable heat treatment.

Welder qualifications with a substantial number of documentation discrepancies.

Carbon steel weld rod may have been used for a portion of several stainless steel recirculation line welds.

١

Electrical cable tray installation and inspection deficiencies.

- Hangers installed for the control rod drive system are of indeterminate quality.
- Both weld and radiograph quality deficiencies for sacrificial shield welds and radiograph deficiencies identified for the containment monorail and the ventilation stack.
- Deficiencies in the H. J. Kaiser procurement program for structural steel and other materials.
- Inadequate design control by Sargent & Lundy (architect engineer) for electrical separation.

6

- Inadequate weld preparation prior to radiography (ripples not removed) which caused masking of discontinuities in some welds.
 - Reactor control, reactor protection, and neutron monitoring panels, including field installed wiring do not, in some cases, conform to design drawings with regard to cable separation.

Inadequate engagement of "gamma plugs" in large-bore piping and lack of heat number traceability of the "gamma plugs." (During radiography of a pipe weld, a gamma source is sometimes inserted through a small hole in the side of the pipe. After radiography the hole is plugged to provide a pressure boundary.)

- Inadequate inspection program and installation procedures for "Nelson stud" installation for cable tray hangers.
- Concrete and steel coating program not in accordance with the QA Program and the Sargent & Lundy specification requirements.
- Design changes made to the Fire Protection System piping in the cable spreading room in 1979 were inadequately controlled.
 - . The Sargent & Lundy (architect engineer) dynamic stress analysis of small bore piping is questionable.
 - Cable separation problem with regard to division separation between non-essential cables being bundled with essential cables of different divisions.

Pipe support installation procedures did not contain seismic clearance criteria between pipe supports and cable trays or conduit and associated supports as required by the specification.

These deficiencies represent those which the staff considers most significant. There were additional 10 CFR 50.55(e) reports made by the licensee has identified a large number of

nonconformances (which could reflect construction or other types of deficiencies). As of September 30, 1982 the licensee's continuing quality confirmation program reviews had identified approximately 4,200 nonconformances of which about 800 have been "dispositioned", <u>i.e.</u>, the licensee had made a determination as to resolution. (Inspection Report No. 50-358/82-12, report pending.) The large number of noncomformance reports and the significance of the matters being identified corroborate the staff's 1981 finding of significant breakdown in the licensee's quality assurance program.

B. <u>Findings Subsequent to Licensee Actions Taken to Correct OA Failures and</u> Prevent Recurrence

Since the Immediate Action Letter was issued on April 8, 1981 and quality assurance and management deficiencies were brought to the attention of the licensee, hardware and programmatic QA/QC problems have been identified by the NRC and the National Board of Boiler and Pressure Vessel Inspectors. These problems are discussed in the following paragraphs and indicate the licensee and the constructor are still having difficulty implementing satisfactory QA/QC programs:

> During an inspection conducted the latter part of 1981 and the early part of 1982 (Inspection Report No. 50-358/82-01, issued on June 24, 1982) three items of noncompliance were identified. The findings concerned (1) the failure to clearly establish and document the authorities and duties of all QA Department personnel, (2) the failure to provide

- 7 -

ł

adequate certification of qualifications of all QA Department personnel, and (3) the failure to provide adequate procedures. The licensee failed to adequately address the provisions of Regulatory Guide 1.58 (ANSI N45.2.6-1978) concerning personnel in the QA Department. Additionally, inadequately qualified personnel were reviewing and approving quality procedures controlling electrical activities, which contained deficiencies.

Furthermore, as a result of the licensee reviews it was revealed that some weld inspectors involved in the QCP Task I, Structural Steel, were not adequately certified and the task was stopped. The task was restarted following upgrade of the inspectors through training provided by additional certified weld inspectors.

During an inspectio: conducted in March and April 1982 (Inspection Report No. 50-358/82-05, issued on July 1, 1982) two items of noncompliance were identified. The findings concerned the lack of implementation and timeliness of corrective actions and the failure to adequately review and document potentially reportable matters.

During an inspection conducted in April, May, and June of 1982 (Inspection Report No. 50-358/82-06, issued on November 2, 1982) two items of noncompliance were identified. The findings concerned (1) the performance of quality activities required of the welding engineers by inadequately qualified clerks and (2) the failure to perform required calibrations

- 8 -

during a critical quality activity, Induction Heating Stress Improvement (IHSI) program.

A recent inspection conducted during June and July of 1982 (Inspection Report No. 50-358/82-10, report pending) identified a number of signficant concerns. These concerns were discussed with the licensee on July 9, July 15, August 15, and October 19, 1982. Four significant items of concern (potential items of noncompliance) were identified: (1) the inadequate control and documentation of welder qualifications; (2) the failure to take corrective actions following the identification of inadequate records to support welder qualifications; (3) the unauthorized correction, supplementation, and alteration of quality records; and (4) the failure to follow procedures controlling weld filler metal control, logging and control of requests for information/evaluation, and imposition of reporting requirements on contractors. The NRC findings concerning weider qualifications resulted in the requalification of approximately 100 active onsite welders and the need for the licensee to develop a program to evaluate the previous work of the welders whose qualifications were not adequately documented.

An inspection was conducted following notification of the Region III Office that a CG&E Stop Work Order (SWO) had been initiated on August 5, 1982, pertaining to Catalytic, Inc. (CI) activities in the area of the control rod drive system hangers and supports. CI is a contractor of the licensee performing construction work

0

including rework activities identified by the QCP program. During this inspection conducted during August and September of 1982 (Inspection Report No. 50-358/82-13, report pending), significant concerns were identified regarding the implementation of CG&E's quality assurance program and its management program established to control and monitor the activities of Catalytic, Inc. (CI). The concerns involved the areas of (1) the description of organization `and functional interfaces, (2) training of CI personnel, (3) design control measures, (4) procedure content and implementation, (5) document control, (6) inspection and surveillance activities, (7) nonconforming conditions, (8) corrective actions, (9) records, and (10) audits. The findings were discussed with the licensee on August 12, September 10 and 17, and October 19, 1982.

As a result of the inspection findings and subsequent discussions with the licensee, Stop Work Orders were issued by the licensee, stopping all essential work by CI on October 11, 1982, pending resolution of the programmatic problems identified by the NRC and licensee reviews.

The licensee has initiated Stop Work Orders in addition to those affecting CI due to inadequate quality assurance in the areas of application of coatings (October 12, 1982), electrical cable installation (October 12, 1982), and special process procedures (November 1, 1982). The Stop Work Orders involve ongoing activities. The November 1, 1982 Stop Work Order involved procedures not meeting requirements notwithstanding that the procedures had been specifically reviewed by CG&E for adequacy subsequent to the issuance of the April 8, 1981 Immediate Action Letter.

Additionally, during the week of October 10, 1982, the Authorized Nuclear Inspector (ANI) for the H-stamp holder (H. J. Kaiser) recalled ASME work packages then being used in the field because of the performance of ASME code work (hanger attachment removal and piping cutouts) was outside the approved QA Program procedures. The ASME code work was being controlled and performed utilizing an H. J. Kaiser administrative memo which bypassed the ANI's required involvement in the code activities. The NRC was apprised of the required corrective actions during a meeting involving CG&E and H. J. Kaiser on October 15, 1982. The corrective actions taken and planned were considered acceptable by the Authorized Nuclear Inspector.

'n

The National Board of Boiler and Pressure Vessel Inspectors, at the request of the State of Ohio, have been onsite since March 1, 1982. The National Board has issued three interim reports documenting findings regarding ASME code activities. The National Board findings include deficiencies in the following areas regarding on-going ASME code activities: design control, procurement, procedures, special processes, nonconforming conditions, and corrective actions. The findings are generally consistent with past and present NRC findings.

- 11 -

C. Rework Activities

As a result of the information obtained from the licensee's reviews of plant quality, the licensee is proceeding, prior to completion of the relevant QCP tasks, to initiate rework activities. A major example of rework activities is the area of structural steel welding. The feinspection and rework of structural steel welds located in a number of areas of the plant have been in process for a number of months. Approximately 70 percent of the structural welds are being reworked to make the welds acceptable. In the case of these welds, rework is being undertaken prior to the completion of the quality reviews to determine the acceptability of all structural steel welds and beam/hanger materials. The rework of these welds prematurely may result in the addition of new weld material over unacceptable weld material or beam/hanger materials. Following completion of the quality reviews unacceptable areas may require additional rework activities. This approach to rework activities indicates a lack of a comprehensive management program to address rework activities and the safety impact of those activities on the facility.

III.

The foregoing information indicates that: 1) the Zimmer facility has been constructed without an adequate quality assurance (QA) program to govern construction and to monitor its quality, resulting in the construction of a facility which currently is of indeterminate quality; 2) substantial efforts are underway-to determine the quality of past construction activities and numerous construction deficiencies have been

- 12 -

identified and are continuing to be identified such that both reanalysis and rework will be required to bring the facility into conformance with the application and regulatory standards on the basis of which the construction permit was originally issued; and 3) rework of deficiencies identified by the Quality Confirmation Program (QCP) has been undertaken prior to completion of other relevant QCP tasks and other reviews, resulting in the potential for additional reworking of the same item if further deficiencies are found, as has been the case, by the quality reviews. Consequently, the NRC presently lacks reasonable assurance that the Zimmer plant is being constructed in conformance with the terms of its construction permit and 10 CFR Part 50, Appendix B, and that there is adequate management control over the Zimmer project to ensure that NRC requirements are being met.

The verification of the facility's quality and appropriate actions to correct deficiencies in construction are of utmost importance to the public health and safety should the licensee receive a license to operate the facility. Moreover, the licensee must be in a position to assure that its construction activities have been properly carried out in accordance with Commission requirements, as the Commission inspectors are not able to personally verify every individual aspect of construction that may impact on safety. In view of the importance to safety of construction verification and corrective actions and the past pattern of quality assurance deficiencies, the Commission has concluded that safety-related construction, including rework activities, should be suspended until there is reasonable assurance that future construction activities will be appropriately managed to assure that rework activities and all other construction activities will be conducted in

4

- 13 -

accordance with 10 CFR Part 50, Appendix B, and other Commission requirements. The Commission has further determined that in light of the foregoing considerations the public health, safety and interest require suspension of construction, effective immediately pending fyrther authorization.

14

IV.

Accordingly, pursuant to sections 103, 161i, 182 and 186 of the Atomic Energy Act of 1954, as amended, and the Commission's regulations in 10 CFR Parts 2 and 50, IT IS HEREBY ORDERED THAT:

- A. Effective immediately, safety-related construction activities, including rework of identified deficient construction, shall be suspended.
- B. The licensee shall show cause why safety-related construction activities, including reworking activities, should not remain suspended until the licensee:
 - (1) Has obtained an independent review of its management of the Zimmer project, including its quality assurance program and its quality verification program, to determine measures needed to ensure that construction of the Zimmer plant can be completed in conformance with the Commission's regulations and construction permit.
 - (a) The independent organization conducting this review shall be knowledgeable in QA/QC matters and nuclear plant construction and shall be acceptable to the Regional Administrator. The independent organization shall make

recommendations to the licensee regarding necessary steps to ensure that the construction of the facility can be completed in conformance with the Commission's regulations and the construction permit. A copy of the independent organization's recommendations and all exchanges of correspondence, including drafts, between the independent organization and CG&E shall be submitted to the Regional Administrator at the same time as they are submitted to the licensee. In making recommendations, the independent organization shall consider at a minimum the following alternatives for management of the Zimmer project and shall weigh the advantages and disadvantages of each alternative:

15

- 1. Strengthening the present CG&E organization.
- Creation of an organizational structure where the construction management of the project is conducted by an experienced outside organization reporting to the chief executive officer of CG&E.
- 3. Creation of an organizational structure where the quality assurance program is conducted by an experienced outside organization reporting to the chief executive officer of CG&E.
- 4. Creation of an organizational structure with both quality assurance and construction project management conducted by an experienced outside

i:

organization reporting to the chief executive officer of CG&E.

- (b) The licensee shall submit to the Regional Administrator the licensee's recommended course of action on the basis of this independent review. In evaluating the recommendations of the independent organization, the licensee shall address why it selected particular alternatives and rejected others. The licensee's recommendations and its schedule for implementation of those recommendations shall be subject to approval by the Regional Administrator.
- (2) Following the Regional Administrator's approval in accordance with section IV B(1)(b),
 - (a) Has submitted to the Regional Administrator an updated comprehensive plan to verify the quality of construction of the Zimmer facility and the Regional Administrator of NRC Region III has approved such plan. In preparing this updated comprehensive plan, the licensee shall review the ongoing Quality Confirmation Program to determine whether its scope and depth should be expanded in light of the hardware and programmatic problems identified to date. The updated plan shall include an audit by a qualified outside organization, which did not perform the activities being audited, to verify the adequacy of the quality of construction; and

- 16 -

- (b) Has submitted to the Regional Administrator a comprehensive plan, based on the results of the verification program, for the continuation of construction, including reworking activities, and the Regional Administrator has confirmed in writing that there is reasonable assurance that construction will proceed in an orderly manner and will be conducted in accordance with the requirements of the Commission's regulations and the Construction Permit No. CPPR-88.
- (3) The Regional Administrator may relax all or part of the conditions of section IV.B for resumption of specified construction activities, provided such activities can be conducted in accordance with the Commission's regulations and the provisions of the construction permit.

'n

۷.

Within 25 days of the date of this order, the licensee may show cause why the actions described in section IV should not be ordered by filing a written answer under oath or affirmation that sets forth the matters of fact and law on which the licensee relies. As provided in 10 CFR 2.202(d), the licensee may answer by consenting to the order proposed in section IV of this order to show cause. Upon the licensee's consent, the terms of

- 17 ·

section IV.B of this order will become effective. Alternatively, the licensee may request a hearing on this order within 25 days after the issuance of this order. Any request for a hearing or answer to this order shall be submitted to the Secretary, U.S. Nuclear Regulatory Commission, Washington, D.C. 20555. A copy of the request or answer shall also be sent to the Director, Office of Inspection and Enforcement, and to the Executive Legal Director at the same address, and to the Regional Administrator, NRC Region III, 799 Roosevelt Road, Glen Ellyn, Illinois 60137. A request for a hearing shall not stay the immediate effectiveness of section IV.A of this Order.

If the licensee requests a hearing on this order, the Commission will issue an order designating the time and place of hearing. If a hearing is held, the issues to be considered at such a hearing shall be whether the facts set forth in sections II and III of this order are true and whether this order should be sustained.

Commissioners Ahearne and Roberts dissent from this decision. Their dissenting views are attached.

It is so ORDERED.



Dated at Washington, D.C. this 12th day of November, 1982. For the Commission

Acting Secretary of the Commission

- 18 -

DISSENTING VIEWS OF COMMISSIONER AHEARNE

ł

1

I agree with both the substance and the direction for change described in this order. However, I would have simply issued a Show Cause Order and would not have made it immediately effective.

Ŋ

DISSENTING VIEW OF COMMISSIONER ROBERTS

I disagree with the action taken by the Commission majority on several grounds. First, I believe the Commission's action in immediately suspending construction at the Zimmer facility is precipitous. Earlier this year. Cincinnati Gas and Electric Company (CG&E) made substantial changes in its management structure in order to manage more effectively construction activities and to monitor more carefully quality assurance programs. Despite the fact that this new organizational structure is relatively untested, the Commission is now suspending effective immediately all construction and corrective actions at the site. Additionally, the NRC Staff admits that CG&E's enhanced Quality Confirmation Program (QCP) and large quality control staff is effectively identifying existing construction problems. Moreover, to the extent that actual construction deficiencies have been found, CG&E's management has demonstrated its willingness to take strong remedial actions by issuing stop work orders in those areas where construction deficiencies have been found. In a plant that is approximately 98 percent complete, the Commission is requiring the relatively few remaining construction activities and the ongoing corrective actions necessitated by the QCP to stop immediately while additional organizational changes are implemented.

Second, I believe the Commission's action does not comport with its own practice. In <u>Licensees Authorized to Possess</u>...<u>Special Nuclear</u> <u>Materials</u>, CLI-77-3, 5 NRC 16, 20 (1977), the Commission said that "[a]vailable information must demonstrate the need for [such] emergency

actions and <u>the insufficiency of less drastic measures</u>" (emphasis added). <u>See also Consumers Power Co.</u> (Midland Plant, Units 1 & 2), CLI-73-38, 6 AEC 1082, 1083 (1973). I believe that, in this case, some of the less drastic alternatives proposed by the Staff would be adequate to resolve the problems at this facility. For example, the Commission could send CG&E a letter indicating that at this time the Commission does not have sufficient information to conclude that Zimmer has been constructed in substantial conformance with the construction permit. The Commission could request the provision of information on the part of CG&E which, if available, would provide the Commission with the necessary assurance. <u>See</u> 10 CFR 50.54(f).

Third, in the absence of willfulness, the Commission may suspend construction effective immediately in accordance with Section 9b of the Administrative Procedures Act and the Commission's regulations <u>only</u> if the Commission finds that the public health, safety, or interest requires such action. I do not believe that the concerns listed in the Commission's Order show that the public health and safety requires immediate suspension of all construction and corrective actions at the Zimmer site. Indeed, Mr. James Keppler, the Region III Administrator, has stated that CG&E's QCP has been successful in identifying existing construction problems. Transcript of Public Meeting on the Status of Zimmer, October 28, 1982 at 5. Additionally, most of the NRC inspection findings arising cut of the QCP point to administrative or procedural deficiencies, rather than to actual material or construction errors. While the NRC's level of confidence in the adequacy of the plant

6

2

construction has been reduced, it has not been shown by the NRC that problems exist which require immediate resolution to protect the public health and safety. Moreover, I do not believe this action is in the public interest.

I am also concerned that the Order has been approved without consideration for the Applicant's proposal to correct management and construction problems. That proposal, outlined in a letter to the Commissioners dated November 10, 1982, contained all of the essential elements approved by this Order. Specifically, the proposal calls for obtaining new project management, stopping all-rework on quality confirmation matters, and an independent third party review to confirm the acceptability of selected safety systems. In view of the voluntary agreement by CG&E to such drastic measures, I feel that this Order is primarily punitive in nature and does little to correct problems in the interest of public health and safety.

Finally, I disagree with the Commission's Order because of the potential for delay inherent in this procedure. CG&E has an absolute right to a hearing on the Commission's Order. If CG&E avails itself of this right, then other "interested persons" will be entitled to demand a hearing. Once started, the hearing would be difficult to bring to an expeditious close. Even if the Staff and CG&E were to reach agreement on the corrective actions to be taken, litigation of the requirements imposed by the Commission Order would continue. <u>Consumers Power Co.</u> (Midland Plant, Units 1 & 2), ALAB-315, 3 NRC 101 (1976); <u>Dairyland Power</u> <u>Cooperative</u> (LaCrosse Boiling Water Reactor), LBP-81-7, 13 NRC 257, 264-65 (1981).

3

E * # *



THE CINCINNATI GAS & ELECTRIC COMPANY

CINCINNATI, OHIO 45201

B. R. SYLVIA VICE PREBIDENT NUCLEAR OPERATIONS

November 15, 1982

U.S. Nuclear Regulatory Commission Region III 799 Roosevelt Road Glen Ellyn, IL 60137

Attention: Mr. J.G. Keppler Regional Administrator

RE: WM. H. ZIMMER NUCLEAR POWER STATION - UNIT 1 - NRC ALLEGATIONS -WELDER QUALIFICATION - DOCKET NO. 50-358, CONSTRUCTION PERMIT NO. CPPR-88, W.O. #57300, JOB E-5590 - FILE NO. NRC-19 & I.E. INSPECTION REPORT NO.

Gentlemen:

By letter dated October 27, 1982, CG&E was notified by the NRC that "NRC Region III has been advised of allegations that relevant documentation on the welders at the Zimmer Site was prepared for, or reviewed at, a meeting between Cincinnati Gas & Electric Company (CG&E) and H.J. Kaiser (HJK) held on July 8, 1982, but that such documentation was not discussed with, or made available to, Region III at the meeting on this subject held on July 9, 1982 among CG&E, HJK and Region III. Substantial documentation was made available to Region III in connection with the July 9 meeting, but additional relevant documentation was allegedly not made available."

The July 8, 1982 meeting between CG&E and HJK was a routine Zimmer Site management meeting. One of the topics scheduled for discussion near the end of the meeting was the resolution of current welder performance qualification. These concerns over welder performance qualification records were identified earlier in 10CFR50.55(e) Report M-45 and in numerous HJK and CG&E nonconformance and Corrective Action Reports which were available to the NRC. Generic categories of document deficiencies identified in these CARs and NRs were discussed at the meeting for the purpose of

Attachment IV

Mr. J.G. Keppler

determining resolutions acceptable to CG&E and HJK management. The results of this discussion were presented to the NRC in a meeting. on July 9, 1982. Additional supporting documentation was subsequently made available to the NRC.

2

CG&E's approach to formulating a response to the aforementioned allegation was to seek information from each individual who attended the July 8, 1982 meeting. Everyone attending the meeting was asked to identify documentation or reports made available or discussed at the meeting, or used in preparing for the meeting.

On the basis of the information given to me by the meeting participants, and my personal knowledge, I have determined that no reports or documentation other than that already made available to Region III were prepared for use at either the July 8, or July 9, 1982 meeting. A list of documents or reports either used at the July 8 and/or July 9, 1982 meetings or used in preparation for these meetings is provided as Attachment 1. This list is a compilation of the information provided by each of the July 8, 1982 meeting attendees. This list, of course, does not include all information generated as part of the general review of welder performance qualification records not prepared specifically for or discussed at the July 8, or July 9, 1982 meetings. These records have been and are available for NRC review.

Very truly yours,

By BRSishin

B.R. Svlvia Vice President - Nuclear Operations

BRS/bcf Attachments cc: D. Hunter W.F. Christianson

County of Clermont)

Sworn to and subscribed before me this 15th day of November, 1982

Notary Public not an incusation step and -2.3
ATTACHMENT 1

HJK QA Documentation Group

- A) Current Welder Qualification Status Report (July 9, 1982)
- B) Current Welders with Gladstone Laboratories Test
- C) Welder Status Index
- D) Current Welders Status
- E) QA Records Review Welder Status Checklist

HJK Welding Department

A) Welder Qualification List

CG&E Quality Confirmation Program

- A) QCP Welder Qualification Log generated by 19-QA-21
- B) Welder Qualification Record: #50-238
- C) Welder Qualification Record: #9-3174
- D) Welder Qualification Record: Number not recalled

Other documents referenced during the July 8, 1982 meeting

- A) HJK Procedure: WCP-2
- B) HJK Procedure: SPPM 3.2, R.4
- C) HJK Procedure: SPPM 3.2, R.7
- D) Various CARs and NRs referenced in the above documents.

- -

C F BRAUN & CO Engineering and Construction Subsidiary of Santa Fe International Corporation

November 24, 1982

Ì

ş

B R Shelton Commonwealth Edison Co SNED-35 FNB P O Box 767 Chicago, IL 60690

BL-26

Dear Mr. Shelton

ADDITIONAL MATERIAL TESTS LA SALLE STATION ADVANCE PURCHASE ORDER 805-023 BRAUN PROJECT 6356-N

On November 19, we received additional material sampling data from the Nuclear Regulatory Commission Region III office to supplement the material addressed in Section 5.3 MATERIAL of the October 27, 1982, Independent HVAC Review Final Report. The data has been reviewed by our welding and material specialists. His comments, and a copy of the information received from the NRC Region III office, are attached.

Our specialist agrees with, or has no comment on, the NRC data. For attachments 4, 5 & 6 "No substantive comments" means that the information in the letters is technically correct but there are some editorial corrections. For example, 163° pyramid indenter should be 136° per ASIM E92.

Sincerely yours

AJK df

Andrew J Kempiak Project Manager

Dr. A Bournia Office of Nuclear Reactor Regulation U.S. Nuclear Regulatory Commission Washington, D.C. 20555 NR C Docket Nos. 50-373/374

James G Keppler, Regional Administrator -3 copies U.S. Nuclear Regulatory Commission 799 Roosevelt Road G R Bodde Glen Ellyn, IL 60137 Manager 1

Daniel L Shamblin -2 copies Commonwealth Edison Company La Salle County Station Section 1 through 5 only G R Boddeker -w/o attachment Manager Nuclear Projects C F Braun & Co 1000 So Fremont Alhambra, Ca 91802



C F BRAUN & CO Engineering and Construction Subsidiary of Santa Fe International Corporation

B R Shelton

Page 2

Harold R Denton, Director -40 copies Office of Nuclear Reactor Regulation U.S. Nuclear Regulatory Commission Washington, D.C. 20555

Richard Pollock -6 copies SARGENT AND LUNDY 55 East Monroe Chicago, IL 60690

Leonard J Koch Vice President Illinois Power Company 500 South 27th Street Decatur, IL 62525

Walt Bird Consumers Power Company 1945 West Ponnell Road Jackson, Michigan

Cordell Reed, Vice President Nuclear Operations - 37FNW Commonwealth Edison Company 1 First National Plaza Chicago, IL 60690 Project 6356-N BL-26 November 24, 1982

Charles W Schroeder Nuclear Licensing Administrator - 34 FN E Commonwealth Edison Company 1 First National Plaza Chicago, IL 60690

Walter J Shewski Manager, Quality Assurance (Marquette 6th fl) Commonwealth Edison Company 140 South Dearborn Street Chicago, IL 60690

1

ŧ

Gerald J Diederich Superintendent La Alle County Station R.R. 41 Marseilles, IL 61341 CFBRAUN& CO A Subsidiary of Santa Fe International Corporation

	Date	NOVEMBER 23, 1982
ANDY KEMPIAK Power division	From	LEONARD BOYD RESEARCH

On 6356-N, COMMONWEALTH EDISON, LA SALLE 1, MATERIAL PROPERTIES

On November 19 the U S Nuclear Regulatory Commission's (USNRC) Region III office telecopied data from their material sampling program that was not included in their transmittal to C F Braun on October 7. Copies of the six documents are attached. The identification of the documents and my comments are listed below.

ATTACHMENT 1 Letter dated November 19 from US NRC, C E Cornelius, 1 Page to C F Braun, W L Stiebe, transmittal letter for other attachments.

No comment.

ATTACHMENT 2 Sargent and Lundy Responses to NRC Region III "Questions 6 pages on HVAC System"

> Question 1 I agree with S & L's discussion of the effects of increased hardness of bolting material. Although the reported hardness of 287 Brinell is well above the maximum of 241 allowed for A307 Grade A bolts, it is near the middle of the range of 248 to 331 Brinell for A325, High-Strength Bolts for Structural Steel Joints. A325 is a commonly specified bolting material when higher strength is required.

Questions 2 and 3. No comments.

Results of Sample Analysis (56 samples)

ATTACHMENT 3 4 pages

То

My comments on the first 48 samples are contained in my pink letter to you dated October 13.*

Samples 49 and 50 conform to the chemical requirements for A575, Grade M1015, and are acceptable for use in accordance with ASTM A29, paragraph 4.3.1, because no misapplication is indicated.

Samples 51 through 56 are for A307 bolts and A563 nuts. The results shown are within the specification requirements.

*The first 48 samples are discussed in Section 5.3 MATERIAL and Appendix L of the final report.



ļ

UNITED STATES NUCLEAR REGULATORY COMMISSION REGION III 756 ROOSEVELT POAD GLEN ELLYN, ILLINGIS 19107

November 19, 1981

1.8.7.

C. F. Braum & Company ATTN: Mr. Wayne L. Stiebe Assistant Vice-President 1000 South Frencht Avenue Alhembra, California 91802

Gentlemen:

Please find attached additional data we obtained as part of our material sampling program at LaSalle County Nuclear Statics which is an addition to the data provided to you on October 7, 1982.

If you have any questions on the above, please contact bick Jackiw or Roger Lanksbury of my staff at (312) 932-2503.

Sincerely,

C. F. Yi-seun

C. Z. Sorelius, Director Division of Engineering and Technical Programs

Attachments: As Stated

cc: T. Novak, NBR A. Bournia, NRE L. Spessard

PDR

8212080664 821124 PDR ADDCK 05000373 SARGENT & LUNIY ENGINEERS Attachment 2

ł

NRC REGION III

QUESTIONS ON HVAC SYSTEMS

Question 1: The NRC tested 4 bolts which were 3/8 inch A307 Grade A. The ASTM hardness requirement is between 121 and 241 Brinell. One of the 4 bolts tested had a hardness of 287 Brinell. The chemical analysis was OK. The NRC's lab could not conduct elongation or tensile tests on a 3/8 inch bolt. What does the 287 Brinell hardness mean?

S&L Response: With increasing hardness yield and fracture strength increases and ductility decreases. Increased hardness is advantageous from a strength point of view.

> Increased hardness also decreases ductility of the bolt material. However, as no impact loads are expected on HVAC ductwork, this decreased ductility does not affect the safety of the bolts.

Bence, increased hardness is beneficial.

Attached is a list of strasses on the bolts. obtained by conservative analysis of typical MVAC ductwork containing componants (dampers, registers, grills, etc.).

ð

ł

SARGENT & LUNDY EN SINEERS CH CASO

Summary of findings for A307 bolts used in HVAC Duct. Companion flange analysis.

Note: 1) The analysis was conservatively based on highest duct and duct component weights.

2) Analysis was based on 3.'8 inch bolts.

3) Yield stress for A307 is minimum 36000 psi.

Building - Reactor 1 and 2

(

(

١

Service Level: Emergency

Duct Size (W X H inches)	Calculated Stress (ksi)	Duct Size (W X H inches)	Calculated Stress (Ksi)
IO X 6 IO X 6 IO X 6 I2 X 8 I2 X 10 I2 X 12 I4 X 10 I6 X 16 I8 X 8 I8 X 12 I8 X 14 I8 X 18 24 X 24 24 X 18 24 X 20	(ksi) 3.0 10.0 9.6 10.7 9.6 10.6 10.33 9.76 9.35 8.94 9.92 10.05 8.99 7.18 9.16	<pre>(W X H inches) 30 X 14 30 X 20 32 X 20 36 X 30 40 X 20 40 X 36 42 X 18 42 X 18 42 X 36 48 X 16 48 X 32 60 X 40 72 X 60 96 X 40</pre>	Stress (ksi) 7.544 8.323 4.633 9.92 9.102 7.995 7.831 5.95 9.76 5.002 2.624 5.054 6.44
26 X 12 26 X 14 26 X 20 28 X 14 28 X 20	9.184 9.963 9.512 9.061		

3

SARGENT & LUNDY ENGINEERS GH CAGD

Building - Auxiliary

•

1

1

Service Level: Envergency

.

Duct Size (W X H inches)	Calculated Stress (ksi)
10 × 6	10.414
	10.91 -
70 X C	6.91
	10.54
	10.91
	10.62
	6.57
14 X 40	9.31
18 X 44	8.57
20 X 16	9.594
12 X 36	8.4
36 X 18	10 62
26 X 20	0 67
24 X 54	0, <i>31</i> 8 87
48 X 36	0.04
30 X 28	/./3
30 X 38	9.43
22 X 18	7.71
28 X 70	9,512
72 X 72	5.54
A0 Ø	9.8
70 Ø	9,23

4

ł

SARGENT & LUNDY ENGINEERS CHICAGO

.

1

.

Building - Containment	Service Level: Emergency
Duct Size	Calculated Stress
(W X H inches)	(ksi)
12 X 24	8.16
32 X 10	8.1
30 X 32	8.364
18 Ø	9.27

5

SARGENT & LUNDY ENGINCESS CHICASO

. .

•

Question 2: What are SiL's design factors for companies flanges and stiffeners? flanges and stiffeners?

SEL Response: Between companion flanges and duct sheet metal, the sheet metal is the weaker member.

١

Stiffeners are not load carrying members. They are installed to prevent warping of the duct sheet metal. The stiffeners are installed to industry practice.

Attachmemt	2	Ģ

1

SARGENT & LUNDY ENGINEERS CHICADO

Question 3: Was A500 structural tubing ever added to HVAC Spac? S&L Response: A500 tubing was approved for use by Field Change Request (FCR) 34,307. It will soon be incorporated into the HVAC Spac.

.

RESULTS OF SAMPLE AMALYSIS

	AST:*	STN* System	_					Tersile Strength HE		
Sample	Material Type	Removed From	System Component	C		P	Mn 1			
1	A36	TE	Ranger	0.19	0.035	0.003	0.68	-	39	
2	A527	VC	Duct	0_07	0.027	0.005	0.38	N/A	Z (Y	
3	A36	VC	Banger	0.15	0,033	0.006	0.59	63	53	
4	A575	۷C	Stiffner	0.18	0.030	0,009	0.49	8/A	5/k	
5	A36	₹	Bauger	0.23	Q.038	0.012	0.58	-	55	
6	A36	VE	Hanger	0.21	0.026	c.:06	0.58	-	5ē	
7	A527	VZ	Duct	0.07	9.031	0.006	0.35	2/¥	S/A	
8	A575	VE	Stiffner	0.16	0.036	0.018	0.55	N.'A	S/A	
9	A36	۷C	Hanger	0.13	0.019	c.005	0.75	59	55	
10	136	₹C	Eanger	0.21	0.031 0.033	0.001	0.86	-	<u>1</u> ē	
11	A36	VC	Hanger	0.20	0.041	0.027	C.38	•	38	
12	▲527	VX	Duct	0.05	0.019 0.018	0.005	0.41	r/a	87A	
13	A36	YY	Hanger	0,18	0.035	0.012	0.62	-	38	
14	A 36	T V	Hanger	0.15	0.035	0.006	C.éS	63	51	
15	A36	XA	Langer	0.13	0.035	0.008	0.54	-	35	
16	A36	VE	Eanger	0.19	0.046	0.023	0.73	63	38	
17	њЗ6	X V	Sanger	0.19	0.032	0.0051	0.61	-	58	
18	A575	TX.	Companion Flange	9.22	0.031 7.032	0,017	0.38	N: A	N A	
19	2575	C7	Hanger	3.20	5.023	0.007	0.50	2 (¥	N A	

*All ASTM A575 Samples are specified to be Grade M1000, except sample w1 which was M1013

Attachmont 3

8

ŧ

RESULTS OF SAMPLE AVALYSIS

.

	ASTM#	Systep						Tensile Str.	47 275. FS.
Sample	Macerial	Removed	System		Set 2	<u></u>		Calculated	ASTM
Number	Тура	Frez	Composent	1 C	5	<u>?</u>		<u></u>	
20	A527	C7	Duct	0.05	0. <i>323</i> 0.024	360.0	0.35	N/A	5/A
21	A575	VD	Companion Fiange	0.23	0,026	0.005	0.52	8/A	5/A
/ 1	A36	C7	Hanger	0.19	C.038	0.008	0.60	-	58
23	A26	CV	Hangar	0.19	0.036	0.009	0,68	64	53
24	A575	ΔΔ	Stiffner	0.20 0.19	6,026	0.008	0.61	S/X	N, A
25	A36	77	Eanger	0.20	C.C42	0.007	0.64	-	58
26	A575	VŸ	Stiffaer	0.23	0.031	0.009	0.61	N/A	N'A
27	A36	AĀ	Hanger	0.17	0.037	0.007	0.56	-	58
28	A575	VT	Sciffaer	0.21	0,029	0.008	0.58	3/2	5/A
29	A36	VY.	Hanger	0.15	0.049	0.035 0.035	0.72	-	53
30	A36	VT	Hanger	C.21	0.030	0.009	0.72	-	56
31	A527	VD	Duct	0.03	0,02	0,005	0.42	5/A	N/A
32	AS27	VD.	Duct	0,05	0.02	0.009	C.30	N/A	N/A
33	A5 27	VD	Duct	0.06	0,02	0.006	0.43	S/A	8/A
	3527	AA	Duct	0.06	6.03	0.007	0.29	N/A	S.'A
35	AS27	ůā	Duct	0.06	0.02	c.007	Ç.23	5/A	No A
35	A36	VY	Emger	0.17 0.15	0.0-	0.005	0.71	÷€	35

All ASTM A575 Samples are specified to be Grade MICLO, except sample 41 which was MICLE.

.

Ģ

÷

RESULTS OF SAMPLE ANALYSIS

le l										
r		This	helt is techar b							
	Sample	ASTM* Material	System Removed	System		Weizt	<u></u>		Tensile Str Calculated	ASTA
	Number	Туре	Froe	Compensat		<u> </u>			<u></u>	
	37	A 527	VX .	Duct	0.98	0.03	6.005	0.32	8/A	S-A
	38	▲307	VX	Bolt	-	0,03	0.211	0.69	143	4 0
	39	A563	X 7	Nat	0.05	-	0.009	0.39	. 97**	66**
(40	A575	X	Hanger	G.21	C.C3	0.013	0.65 0.62	5/A	No A
	41	A575	5X	Stiffer	0.15	0.03	0.010	0.42	3.'A	N/A
	42	▲ 575	۷C	Hanger	0.19	0.04	0.018	C.44	N/A	5/A
	43	A575	VC	Stiffer	0.19	0.0-	0.012	0.32	8/A	S/A
		A575	۳C	Companion Flange	0.13 9.12	0.03	3.015	0.4 <u>5</u>	N/A	3/A
	45	A527	۷C	Duct	0.04	0.05 0.04	0.009	0.32	N/A	M A
	46	٨527	vc	Companion Flange	0.14	0.04 C.03	0.015	C.41	5/a	N/ A
ŗ	¥ 47	A527	VE	Duct	0.09	0.02	0.006	0.40	3/A	S. A
-	3 48	A527	70	Duct	9.07	0.03	0.011	0.33	N/A	S A
-	49	A575	ΫC	Companion Flange	0.14	0.03	0.217	0.45	S/A	87.A
(50	A575	۳C	Companion Flange	C.13	0.03	0.017	0,45	X.A.	8 s
	51	A307	XV.	Bolt	-	J. 22	0.122	-	* *	60
	52	A363	גד	Nut	0.⊠		0.016	0.37	<u>8</u> .	<u>.</u> :**

*All ASTM A575 samples are specified to be Grade M1021, except sample 41 which was M113.
**Proof load stress.

874=

95

107-*

ţ

50

Samle	ASTM* Material	Systep Repoved	System		Veig	int t	Tensile Strength, Ki Calculated ASTM		
Sample nave Number Type	Туре	ype From	Component	C	ç	F	<u> </u>	Yin.	Men.
53	▲307	VX	Belt	-	0.02	0.214	•	102	51

0.09 0.02 0.009

- 0.02 0.031

0.11 0.01 0.026

C.3G

•

0.34

A563

A307

X563

77

VB

VR

Suc

Bolt

Nut

54

55

56

(

1

RESULTS OF SAMPLE ANALYSIS

*All ASTM A575 Samples are specified to be Grace MillD, except sample 41 which was MillD. **Proof lead stress,

AUG 11 SEE

ARCONNE NATIONAL LABORATORY 9700 South Cass Areve Arcone Illinois 60439

Mephone 72, 772, 5137

August 5, 1982

Roger Lanksbury USNRC 799 Roosevelt Rd. Glan Ellyn, IL 60137

STBJECT: Hardness Tests on A36 Specimens

Dear Rogar:

Rardness tests were performed to determine whether the material samples which meet the chemical requirement for Alf material also meet the tensils requirements for this material. The requirements include a minimum and maximum tensile strength, a minimum yield strength, and minimum elongation requirements. These should be determined by unlaxial tensile tests. However, because of limitations on the abrunt of material available hardness tests offered the only means to estimate these properties. The relationship between hardness and flow stress is well known (see, e.g., F. McClintock and A. Argon. Mechanical Schevier of Materials). Because this material strain hardens substantially and because these specimens were probably subjected to significant deformation when they were removed, the flow stress is best interpreted as a lower bound on the tensile strength.

Five specimens of the nominal A36 vers judged suitable for testing. These specimens were mounted and polished to removed ~~0 mils of the most severaly deformed material from the surface. Fickers hardness tasts were than performed using a 163° pyramid indenter with a loading force of 30 kg. Some exploratory tasts showed that the particular choice of the loading force had little effect on the measured hardness values. The mean diagonal of the impression d, was measured and the Vickers hardness V computed from

$$7 = \frac{P}{A_C} = \frac{1.854P}{d_1^2}$$
 (P kg, $z_1 =$)

where P is the loading force and A1 the contact area. The Meper-Vickers hardness M2 which is based on the projected area A_2 instead of the contist area A_2 was computed from

My =
$$\frac{27}{2.2}$$
 = $\frac{7}{0.527}$ KB =

Since the Meyer-Vickers hardness generally is a petter representation of the average pressure on the area of contact it was used to compute the flow stress Sy from

USDEATIVENT & EBG.

Belandsto S Conservation

8212080668 821124 PDR ADOCK 05000375 P PDR

12/

1

2

 $M_{y} = 3.2 S_{y}$

The results from the five specimens are tabulated below with $S_{\rm E}$ expressed in the usual American engineering units of psi (1 kg/mm² = 1412 psi).

Specimen	V (kg/m ²)		S _F (pei)
▲36-3	136	137	65,000
A36-9	145	156	65,000
A36-14	136	147	65,COC
A36-16	136	147	63,000
A36-23	234	145	54,000

Since the minimum tensile strength for A36 material is 58,000 psi, these results indicate that all these specimens do meet this specification.

The measurements and calculations presented here were actually performed by J. T. Park and D. Perkins. Since I will be away for the next two weeks, if you have additional questions please call Jang Yul Park at 972-5030.

Sinceraly.

William Shack Materials Science Division

WJS:diam

- cc: J. Y. Park
 - D. Perkins
 - R. W. Waeks

Attachment 5

£

Techone \$12/972/5137

ARCONNE NATIONAL LABORATORY 9700 South Case Alence Arcone, 11 nois 60439

September 21, 1982

Hr. Roger Lanksbury U.S. Nuclear Regulatory Commission 799 Roosevelt Road Glan Ellyn, IL 60137

SUBJECT: Hardness Tests on Assorted Specimins

Dear Roger:

Bardness tests were performed to determine whether the material samples met the mechanical requirements of the appropriate ASTM specification. In some cases the requirements include a minimum and maximum tensile strength, a minimum yield strength, and minimum alongation requirements. These should be determined by uniaxial tensile tests. However, because of limitations on the amount of material available hardness tests offered the only means to estimate these properties. The relationship between hardness and flow stress is well known (see, e.g., F. McClintock and A. Argon, <u>Mechanical Behavior of</u> Materials).

Specimens were mounted and polished to remove 350 mils of the most severely deformed material from the surface. Vickers hardness tasts were then performed using a 163° pyramid indenter with a loading force of 30 kg. Some exploratory tests showed that the particular choice of the loading force had little effect on the measured hardness values. The mean diagonal of the impression d₁ was measured and the Vickers hardness V computed from

 $\nabla = \frac{P}{A_{c}} = \frac{1.854P}{d_{1}} (F \text{ kg, } d_{1} \text{ m})$ (1)

where P is the loading force and A_C the contact area. The Mayer-Vickers hardness M₂ which is based on the projected area A₂ instead of the contact area A₂ was computed from

$$M_{\rm Ty} = \frac{2P}{d_{\rm T}^2} = \frac{\nabla}{0.927} k_{\rm F}^2 m^2$$
(2)

Since the Meyer-Vickers hardness generally is a better representation of the average pressure on the area of contact it was used to compute the flow stress $S_{\rm p}$ from

LS DEPARTNENT of EVERY.

THE UNREST OF THEMAS

ARYDRIE UNACHETER ALS PARTIN

8212080670 821124 PDR ADOCK 05000373 P PDR

(3)

ł

My = 3.2 SF

1

Ń,

The results are tabulated below with S_p expressed in the usual American engineering units of ksi (1 kg/mm² = 1.422 ksi).

2

Item	ASTM Specifications*	Measured	Flow	
	•	Q##	<u>ж</u> т	Stress S _F
Plate ASTM A36	Tensile Strength 58-80 ksi	137	1-8	66 ksi
Sergie # 26	Minimum Vield Point 36 ksi			
Bolt ASTH A307 Grade A	Tensile Requirement GOHT ksi	303	327	145 ksi
This that is both that is	Brinell Hardness 121-241 min max			
jangla # 58	Rockwell Hardness 869-8100 min max			
Hut ASTM A563 Grada A	Proof Load Stress 68 ksi	203	219	97 251
5 emple # 19	Brinell Hardness 116-302 min max			
	Rockwell Hardness 369-C31 min max			
Washer Unknown This is	the not still speed	<u> </u>	107	-8 ks1
*Farts of the required for full specification	specifications. See "An	nnual Book .	of ASTM Sca	caris"
**Vickers hardness, Eq.	(D .			-
Meyers hardness, Eq.	(2). 5 q. (2).			

15

,

Ł

ł

1

The flow stress obtained for the A16 plate is consistent with the tensile strength requirements for this material. The requirements for the bolt and nut are given in terms of Brinnell hardness rather than Vickers hardness. However, the two are approximately equal; a more accurate comparison of the two hardness measurements can be obtained from the ASH Metals Handbock, Vol. I, p. 1234. According to these results:

Vickers	is equivalent to	Brineil	or	Rockutli
303	•	287		B1C6
203		193		892
99		94		B56

Thus the nut seems to meet the hardness specification and the calculated flow stress is consistent with the proof load stress. However, the bolt appears to exceed the hardness specification which would tend to reduce the toughness on the bolt. We have made measurements on the washer, but it is not clear what specifications it should meet.

I hope this information will be of help to you. As before the measurements and calculations presented here were actually performed by J. Y. Park and D. Perkins.

Sincerely, 1.1 41 114, - - - -William JC Shack

Materials Science and Technology Division

WJS:dica

۰.

cc: J. T. Fark D. Perkins

R. W. Weeks

3

Attachme	nt 6
٠	12

ARCONNE NATIONAL LABORATORY 9700 South Cass Areve Arcone linos 50439

EDIOC NE/972-5137

Cccober 7, 1962

Mr. Roger Lanksbury U.S. Nuclear Regulatory Commission 799 Roosevelt Road Glen Ellyn, IL 60137

SUBJECT: Hardness Test on Bolts and Nuts

Dear Roger:

Hardness tests were performed on the three bolts (1/2, 3/8, 1/4-in.) and the corresponding nuts to see if they conformed to the appropriate ASTM specification (A307 for the bolts, A563 for the nuts). The results of the tests are summarized below:

					Harda	;e\$\$	
Dur san	,48 's	S	pecimen		Vickers	Brinell	
¢. 2	No. 2	22	3/8-in.	Bolt Nut	169 193	152 182	
رد ۲۱	No.	23	1/4-in.	Solt Nut	211 190	201 181	
41 5	No.	24	1/2-in.	Bolt Nut	157 222	178 211	

The tests were actually performed using a Vickers 163° pyramid intenter with a loading force of 30 kg. The correspondence between the Vickers and Brinell hardness follows the correlation given in the ASM <u>Hetals Handbock</u>, Vol. 1, p. 1234.

The specimens do meet the required specifications, and in most cases are roughly in the middle of the range. The well known correspondence between herdness and flow indicate that they almost certainly must the tensile requirements, but we have not checked them itrattly.

Sincersly, الم يعد وسر مساد بازا William 📶 Shack

William R., Shack Matarials Science & Technology Division

WJS:dkm

8212080673 821124 PDR ADOCK 05000373 007 : 2 592

Ł

LS DEDATIMENT of EVERGY

PDR

THE WARRENT & CHEVREN

AMONE LINE STES ASST LATER

PDK



UNITED STATES NUCLEAR REGULATORY COMMISSION WASHINGTON, D. C. 20555

DEC 6 1982

NOTE TO: File (50-373)

FROM: Anthony Bournia, Project Manager Licensing Branch No. 2, DL

SUBJECT: PROCESS DOCUMENT FROM C. F. BRAUN

Please process this document thru DCS. This is additional information to be included in the C. F. Braun HVAC Independent Review Report for La Salle Unit 1.

Bourne

Anthony Bournia, Project Manager Licensing Branch No. 2, DL

cc: Jeff Bartlett

DEC 6 1982

DISTRIBUTION: Docket File LB#2 File ABournia

NOTE TO: File (50-373)

....

FROM: Anthony Bournia, Project Manager Licensing Branch No. 2, DL

SUBJECT: PROCESS DOCUMENT FROM C. F. BRAUN

Please process this document thru DCS. This is additional information to be included in the C. F. Braun HVAC Independent Review Report for La Salle Unit 1.

Anthony Bournia, Project Manager Licensing Branch No. 2, DL

cc: Jeff Bartlett



Distribution:

MHartzman

3.

 ΔT

Document Control NRC PDR L PDR DEC 1 1982 PRC System NSIC Docket No.: 50-275 LB#3 File HSchierling HDenton GKnighton ECase JLee Philip A. Crane, Jr., Esq. ACRS (16) REngelken Pacific Gas and Electric Company RVollmer LChandler, ELD P. 0. Box 7442 JKnight Taylor, IE San Francisco, California 94120 PKuo Jordan. IE HPo1k DEisenhut

Dear Mr. Crane:

.

At the October 20, 1982 Commission meeting on Diablo Canyon the staff indicated that their consultant, Brookhaven National Laboratory (BNL), would be requested to perform a limited number of additional independent analyses. In this regard, a discussion of the BNL effort was provided in Enclosure 2 of SECY Paper 82-414 dated October 13, 1982.

The purpose of the BNL analyses are to provide the staff with additional insight as to the character of results obtainable by use of state-of-the-art analytical techniques without regard to methods or procedures previously approved in the licensing process for Diablo Canyon. These analyses are therefore not intended as a substitute for the design and evaluation efforts now underway for the Diablo Canyon project; nor are they a substitute for the analytical efforts being performed by the Independent Design Verification Program (IDVP). Our experience has been, however, that such analyses often provide the bases for judgements that expedite the review process.

Attached is a list of the drawings and data that have been identified as necessary to perform the intended analyses. We request that you provide us with this information as soon as possible. Please advise us promptly if any of the requested information can not be provided within 2 weeks after receipt of this letter.

.

1

Sincerely,

Original signed by Decrett G. Eisenhut

Darrell G. Eisenhut, Director Division of Licensing Office of Nuclear Reactor Regulation

Enc	losure:	
As	stated	

8212080697 821201 PDR ADDCK 05000275

cc: See next page

PDR

5	eepterious	Λ		N			
OFFICE	DL:LB#3	DLAAK	DE :	D/CSE	BT. DB	DLOLA	
SURNAME	HSchierling	GKoʻigh	ton JPK	night	. Morak	Disentut	 •••••
DATE)	<u>11/1/82</u>		82	/.\$/.82	11/ 9/B2	.1	 •••••
NRC FORM 318	(10-80) NRCM 0240	└─── ─	OF	FICIAL	RECORD C	OPY	 USGPO: 1901-335-100

Distribution: Document Control NRC PDR L PDR PRC System NSIC LB#3 File JLee Docket No.: 50-275 HDenton ACRS (16) LChandler, ELD ECase REnge1ken Talyor, IE RVollmer Jordan, IE JKnight DEisenhut PKuo HPo1k MHartzman HSchierling GKnighton

Philip A. Crane, Jr., Esq. Pacific Gas and Electric Company P. O. Box 7442 San Francisco. California 94120

Dear Mr. Crane:

The participation of Brookhaven National Laboratory (BNL) in the Diablo Canyon design verification effort as a staff consultant was discussed at the Commission meeting on October 20, 1982. Details of the BNL effort provided in Enclosure 2 of SECY Paper 82-414 of October 13, 1982.

Attached is a letter from BNL to the NRC which includes a request for structural drawings, for other material needed for structural analysis and items needed for piping analysis. We request that you provide us with the information identified as soon as possible. Please advise us promptly if any of the information cannot be submitted within two weeks after receipt of this letter.

Sincerely,

Darrell G. Elsenhut, Director Division of Licensing Office of Nuclear Reactor Regulation

Enclosure: As stated

cc: See next page

	4
SURNAME Schierlingiph GKnighton JKnight TNovak DEisenhut	
OFFICE DL:LB#3 DL:LB#3 DE:AD:SE DL:AD:L DL:DIR	

cc: Mr. Richard Hubbard MHB Technical Associates Suite K 1723 Hamilton Avenue San Jose, California 95125

> Joel Reynolds, Esq. Center for Law in the Public Interest 10951 West Pico Boulevard Third Floor Los Angeles, California 90064

Herbert H. Brown, Esq. Hill, Christopher & Phillips, P.C. 1900 M Street, N. W. Washington, D. C. 20036

_ /

Bruce Norton, Esq. Suite 202 3216 North 3rd Street Phoenix, Arizona 85012

David F. Fleischaker, Esq. P. O. Box 1178 Oklahoma City, Oklahoma 73101

Mr. Georg A. Maneatis Pacific Gas & Electric Company P. O. Box 7442 San Francisco, California 94120

Dr. Jose Rosset 3506 DuVal Road Austin, Texas 78751

E

Dr. Morris Reich Structural Analysis Division Brookhaven National Laboratory Upton, Long Island, New York 11973

Diablo Canyon Nuclear Power Plant

.

•.

(

.

٩.

Structural Drawings Request

.

_ *

438232	Interior Concrete Outline Plans at F1 74'-8" & 91'0"	Rev	* 10
	Containment Structure		
438233	Interior Concrete Outline Plans at El 124'-O" & 140'-O" Containment Structure	Rev.	12
438234	Interior Concrete Outline-Main Sections Containment Structure	Rev.	12
438235 ,	Interior Concrete Outline Miscellaneous Sections Containment Structure	Rev.	6
438239	Interior Concrete Reinforcing Slab at El. 91'-O" North Containment Structure Area F	Rev.	5
438240	Interior Concrete Reinforcing Slab at El. 91'-O" South Containment Structure Area G	Rev.	6
438274	Interior Concrete Details Reactor Nozzles Area Containment Structure	Rev.	4
43827 6	Detailed Plans at El 111 of Steam Generator Supports Containment Structure	Rev.	7
438278	Reactor Coolant Pump Supports at El 106 Containment Structure	Rev.	5
438280	Pressurizer Support Containment Structure Area F	Rev.	4
439571	Equipment Supports Plan Below El 113 Containment Structure Areas F & G	Rev.	8
439572	Layout of Lateral Support for Steam Generators at El 139'-O" Containment Structure Areas F & G	Rev.	2
439573	Steam Generator Support at El 139'-O" Containment Structure Area F & G	Rev.	6
438135	Requirement for 40,000 Gal. Diesel Fuel Oil Storage Tanks	Rev.	6

*It is understood the revision shown is the latest. If not, provide the latest revision.

-1-

Request for Other Material Needed for Structural Analysis

- 1. Equipment mass data similar to that attached to PG&E letter to J. Blume dated 9/13/82. Ref. Diablo Canyon Unit 1 Containment Structure.
- Geotechnical Studies Intake Structure, Water Storage Tanks Diesel Fuel Oil Storage Tanks Diablo Canyon Nuclear Power Plant San Luis Obispo County Conf., HLA Job No. 569,031.04, Harding-Lawson Assoc., April 12, 1978.
- 3. Letter report from Harding Lawson dated October 11, 1982 updating the above (2) report.

_ t

- 4. 7175360 51 series Vertical Steam Generator Outline Sheets 1 of 2 and 2 of 2 Bechtel #'s DC-663206-27-1 and DC-663206-1-10
- 5. The following sections are required for the polar crane:
 - (a) through the horizontal girders
 - (b) at the top of the columns
 - (c) at the bottom of the columns
- 6. 'The horizontal Newmark HOSGRI 7.5 M digitized time history for the Reactor Containment Building (cards plus printed copy of cards).
- 7. Horizontal surface HOSGRI 7.5 M Newmark digitized time history (cards plus card listing) for the buried diesel oil tank.
- 8. Latest sketches and/or drawings of modifications to the connections of the annulus structural steel.
- 9. Total weight and weight distribution of the steam generators.

Items Needed for Piping Analysis

A PG&E piping document package or its equivalent and computer inputoutput listings are requested for each of the following Diablo Canyon Nuclear Power Plant, Unit 1 piping problems;

> A) PG&E No. 4A-122 (4A-26) B) 6-102 (6-11) C) 8-116, 8-117, 8-118 D) Westinghouse Problem RHR Loop 4 (6-4, 6-7)

The document packages to include at least;

- 1. walkdown isometric
- 2. Hosgri envelope spectra
- 3. thermal and pressure operating modes from DCM-M-46
- 4. building displacements from DCM C-28
- 5. pipe specification and material data
- 6. support descriptions

(]

- 7. equipment movements if appropriate
- 8. applied spring forces if appropriate
- 9. list of data preliminary in nature and and subject to change

Please note the data requests for Problem No. 4A-122 and 6-102 will be used to updata-the BNL calculations for Problem No. 4A-26 and 6-11. The data requests for Problem 1-10 and the RHR Loop 4 will be used to perform two new confirma-tory piping evaluations as per SECY-82-414.

In addition the following general information is requested:

- 1. Piping Reverification Analysis Log
- 2. Westinghouse Piping Analysis List
- 3. Stress Combination and Criteria DCM M-42
- 4. ME 101 Computer Code Manual

ţ.

TOMB ONO

	DISIVIDUITON	
	Docket File	ADe Agazio
	NRC PDR	RIngram
	L PDR	Gray File
Docket No. 50-346	ORB#4 Rdg	ASLAB
	DEisenhut	RFerguson
	OELD	EBlackwood
	AEOD	HOrnstein.
Mr. Richard P. Crouse	1E - 2	Tulambach
Vice President, Nuclear	ACRS-10	Y. Beneroya
Toledo Edison Company	TBarnhart-4	O. Parr
Edison Plaza	LSchneider	••••
300 Madison Avenue	OGC	
Toledo, Ohio 43652	OPA	
	DBrink man	
Dear Mr. Crouse:	RDiggs	

SUBJECT: APPENDIX R TO 10 CFR 50 - EXEMPTION FROM CERTAIN TECHNICAL REQUIREMENTS

DICTOIDUTION

By letter dated April 29, 1982 (No. 815), Toledo Edison Company submitted a request for exemption from certain technical requirements of Section III.G of Appendix R to 10 CFR 50. The technical requirements from which exemption is requested are: (1) the requirement for a fixed fire suppression system in the control room, and (2) the requirement for one-hour-rated fire barriers where less than 20 feet of separation exists between redundant trains of equipment in the component cooling water heat exchanger and pump room.

We have completed our evaluation of your request for exemption, and we conclude (1) that the installation of a fixed fire suppression system will not increase significantly the level of fire protection safety in the control room, and (2) that the installation of one-hourrated fire barriers between the component cooling water pumps will not increase significantly the level of fire protection in the component cooling water heat exchanger and pump room. Therefore, the Commission hereby grants your requested Exemption. A copy of our Safety Evaluation is enclosed.

In accordance with 10 CFR 50.12, we have determined that this exemption is authorized by law and will not endanger life or property or the common defense and security and is otherwise in the public interest. We have also determined that this exemption does not authorize a change in effluent types or total amounts nor an increase in power level and will not result in any significant environmental impact. Having made this determination, we have further concluded that the exemption involves an action which is insignificant from the standpoint of environmental impact and, pursuant to 10 CFR \$51.5(d)(4), that an environmental impact statement, or negative declaration and environmental impact appraisal need not be prepared in connection with this action.

AC FORM 318	(10-80) NRCM 0240		OFFICIAL	RECORD C	OPY	L.	USGPO: 1981-335-960
DATE	********	• • • • • • • • • • • • • • • • • • • •	· · · · · · · · · · · · · · · · · · ·		*	*****	** * * * * * * * * * * * * * * * * * * *
					· · · · · · · · · · · · · · · · · · ·		
OFFICE	F ADUCK	05000346	•••••	• • • • • • • • • • • • • • • • • • • •			•••••••
	8212080643	821123			ſ		

15⁰.894 15⁰.894 162

Mr. Richard P. Crouse

We have concluded that: (1) because the exemption does not involve a significant increase in the probability or consequences of an accident previously evaluated, does not create the possibility of an accident of a type different from any evaluated previously, and does not involve a significant reduction in a margin of safety, the exemption does not involve a significant hazards consideration, (2) there is reasonable assurance that the health and safety of the public will not be endangered by operation in the proposed manner, and (3) such activities will not be inimical to the common defense and security or to the health and safety of the public.

-2-

A Notice of Exemption, which is being forwarded to the Office of the Federal Register for publication, is also enclosed.

Sincerely,

Original signed by

Dærrell G. Eisenhut, Director Division of Licensing

Enclosures:

- 1. Safety Evaluation
- 2. Notice of Exemption

cc w/enclosures: See next page

				-in/m		c.mil	CEB RFerguson 8/10/82
OFFICE SURNAME DATE	ORB#4:DLjv RIngram 7/1/82	ORB#4:DIOL ADe Agueio;o 7/2/82	C-QRB#4:DL f JS#61-2 7/ C782	ORB#5:DL TWambach .7/.7/82		0ELD Str. et a s 7/4./82	
NRC FORM 318	(10-80) NRCM 0240		OFFICIAL	RECORD C	OPY	.	USGPO: 1981-335-960



UNITED STATES NUCLEAR REGULATORY COMMISSION WASHINGTON, D.C. 20555

DISTRIBUTION Docket File ORB#4 Rdg RIngram

November 23, 1982

Docket No. 50-346

Docketing and Service Section Office of the Secretary of the Commission

SUBJECT: DAVIS-BESSE NUCLEAR POWER STATION, UNIT 1

Two signed originals of the Federal Register Notice identified below are enclosed for your transmittal to the Office of the Federal Register for publication. Additional conformed copies (**12**) of the Notice are enclosed for your use.

- Solution Notice of Receipt of Application for Construction Permit(s) and Operating License(s).
- Notice of Receipt of Partial Application for Construction Permit(s) and Facility License(s): Time for Submission of Views on Antitrust Matters.
- □ Notice of Availability of Applicant's Environmental Report.
- □ Notice of Proposed Issuance of Amendment to Facility Operating License.
- Notice of Receipt of Application for Facility License(s); Notice of Availability of Applicant's Environmental Report; and Notice of Consideration of Issuance of Facility License(s) and Notice of Opportunity for Hearing.
- □ Notice of Availability of NRC Draft/Final Environmental Statement.
- □ Notice of Limited Work Authorization.
- □ Notice of Availability of Safety Evaluation Report.
- □ Notice of Issuance of Construction Permit(s).
- □ Notice of Issuance of Facility Operating License(s) or Amendment(s).
- X Other: Appendix R to 10 CFR 50 Exemption from Certain Technical Requirements.

Referenced documents have been provided PDR,

Division of Licensing, OR8#4 Office of Nuclear Reactor Regulation

Enclosure: As Stated

OFFICE	ORB#4:DL	 		
	RIngram	 		 an i kanadara
DATE	11/30/82	a i i i i i i i i pagano de Sec	na na sa sina ana sa sa sa sa	

NRC FORM 102 7-79

Toledo Edison Company

cc w/enclosure(s):

Mr. Donald H. Hauser, Esq. The Cleveland Electric Illuminating Company P. O. Box 5000 Cleveland, Ohio 44101

Gerald Charnoff, Esq. Shaw, Pittman, Potts and Trowbridge 1800 M Street, N.W. Washington, D. C. 20036

Paul M. Smart, Esq. Fuller & Henry 300 Madison Avenue P. O. Box 2088 Toledo, Ohio 43603

Mr. Robert B. Borsum Babcock & Wilcox Nuclear Power Generation Division 7910 Woodmont Avenue, Suite 220 Bethesda, Maryland 20814

President, Board of County Commissioners of Ottawa County Port Clinton, Ohio 43452

Attorney General Department of Attorney General 30 East Broad Street Columbus, Ohio 43215

Harold Kahn, Staff Scientist Power Siting Commission 361 East Broad Street Columbus, Ohio 43216

Mr. James G. Keppler, Regional Administrator U. S. Nuclear Regulatory Commission, Region III 799 Roosevelt Road Glen Ellyn, Illinois 60137

Mr. Ted Myers Manager, Nuclear Licensing Toledo Edison Company Edison Plaza 300 Madison Avenue Toledo, Ohio 43652 U.S. Nuclear Regulatory Commission Resident Inspector's Office 5503 N. State Route 2 Oak Harbor, Ohio 43449

Mrs. Julia Baldwin, Librarian Government Documents Collection William Carlson Library University of Toledo 2801 W. Bancroft Avenue Toledo, Ohio 43606

Regional Radiation Representative EPA Region V 230 South Dearborn Street Chicago, Illinois 60604

cc w/enclosure(s) and incoming dtd.: 4/29/82

Ohio Department of Health ATTN: Radiological Health Program Director P. O. Box 118 Columbus, Ohio 43216



UNITED STATES NUCLEAR REGULATORY COMMISSION WASHINGTON, D. C. 20555

SAFETY EVALUATION BY THE OFFICE OF NUCLEAR REACTOR REGULATION

SUPPORTING EXEMPTION FROM CERTAIN REQUIREMENTS

OF APPENDIX R TO 10 CFR 50

THE TOLEDO EDISON COMPANY

THE CLEVELAND ELECTRIC ILLUMINATING COMPANY

DAVIS-BESSE NUCLEAR POWER STATION, UNIT NO. 1

DOCKET NO. 50-346

1.0 Introduction

By letter dated April 29, 1982 (No. 815), the licensees requested an exemption from certain technical requirements of Section III.G of Appendix R to 10 CFR 50. Specifically, the licensees request exemption from the requirement for the installation of a fixed fire suppression system in the control room and from the requirement for one-hour-fire-rated barriers where less than 20 feet of separation exists between redundant trains of equipment in the component cooling water heat exchanger and pump room (Fire Zone T-1).

2.0 Discussion and Evaluation

Toledo Edison Company has indicated in its April 29, 1982 letter, that the fire protection features currently installed in the control room/cabinet room and the continuous manning of the control room provide adequate defense-in-depth fire fighting capability for these areas. The licensees have stated that the control room/cabinet room is equipped with area fire detectors and internal cabinet fire detectors for safety related control panels. The control room/cabinet room is provided with both a hose station and fire extinguishers for manual fire fighting, and fire load in the area is low.

In addition, an alternate shutdown system is available which provides remote control capabilities for those systems necessary to maintain safe-shutdown capability from outside the main control room.

Plant Technical Specifications require continuous occupancy of the control room by the operators. Because the operators constitute a continuous fire watch, manual fire suppression in event of a fire would be prompt and effective and, thus, a fixed suppression system is not necessary to achieve adequate fire protection in this area.

8212080649 821123 PDR ADOCK 05000346 F PDR The component cooling water heat exchanger and pump room is an L-shaped room. The approximate length of the room is 67'-6"; the width of the room in the area of the heat exchanger and the crossover valves at the north end is approximately 26'-3", and at the south end of the room, the approximate width increases to 35'-6" to accommodate the CCW pumps.

The walls, floor, and ceiling slabs of the CCW heat exchanger and pump room are three-hour-fire-rated barriers. Access door 332 leading into the area is a Class "A" three-hour-fire-rated door assembly. The piping and electrical penetrations in the CCW room boundary are filled with silicone foam fire barrier sealant material which provides a seal equivalent to the wall in which it is installed. Where the HVAC ducting penetrates the CCW room enclosure, the duct opening is protected by three-hour-fire-rated dampers installed in accordance with the manufacturer's recommendations.

The fixed combustibles associated with this area consist of 6 gallons of lubricating oil. Each CCW pump and motor contains 2 gallons of oil. The lube oil in the CCW pumps and pump motors is enclosed in a selfcontained non-pressurized lubricating system. The lube oil utilized has a flash point of 450°F and an ignition temperature of approximately 700°F. All the power and instrumentation cabling associated with the equipment located in the room is routed in Schedule 40 conduit. There are no cable trays routed in or through the room. The fire load based on the amount of fixed combustibles located in the CCW heat exchanger and pump room is 392 BTU/FT².

The following equipment and its associated cabling is located in this room:

- a. CCW pumps
- b. CCW valving
- c. CCW flow switches for pump discharge header
- d. CCW temperature indicators
- e. Service water valves serving the CCW heat exchangers
- f. CCW pump room ventilation fans C75-1 and C75-2, associated dampers motorized inlet louvers and temperature interlocks.

The three CCW pumps are located at the south end of the room. Pumps 1 and 2, which are normally used during plant operations, are separated from one another, pump center line to center line, by 22 feet. Pump 3 is a swing pump and is located between CCW pumps 1 and 2. The center line of pump 3 is 11 feet from the center line of pump 1 and 2. One CCW pump is needed for safe shutdown.

The CCW heat exchanger and pump room is protected by an automatic sprinkler system. Each of the CCW pump motors is baffled to protect it from the impingement of water. The motor is protected from water impinging vertically by its drip proof design. The sprinkler system also covers the floor area of the room for protection from an exposure fire. This includes sprinklers under the mezzanine floor grating and under CCW crossover header valves at the opposite end of the room from the pumps. Conduits and valves which are required for safe shutdown are protected by a one-hour-fire-rated barrier. The barrier consists of two 1-inch thick Kaowool blankets wrapped around and banded to the conduit and valves with 1/2-inch wide type 316 stainless steel bands and buckles.

Additionally, the floor around each of the CCW pumps is curbed to confine oil leaking from any one pump or motor to the floor area directly around the affected pump. The curbing and the diked floor area around each CCW pump and motor is sized to contain the entire oil content of the pump and motor plus an additional 90% by volume for sprinkler flow.

An automatic smoke-detection system is installed to provide early warning detection in the area. Portable fire extinguishers are located on the north wall of the room. Additional 20-lb dry chemical extinguishers in the stairway and the turbine building are directly accessible to the area. A manual hose station is accessible to the CCW heat exchanger and pump room.

Section III.G.2 of Appendix R to 10 CFR 50 requires that one train of cables and equipment and associated non-safety circuits necessary to achieve and maintain hot shutdown conditions must be maintained free of fire damage by one of the following means:

- a. Separation of cables and equipment and associated non-safetv circuits of redundant trains by a fire barrier having a three-hour rating. Structural steel forming a part of or supporting such fire barriers shall be protected to provide fire resistance equivalent to that required of the barrier;
- b. Separation of cables and equipment and associated non-safety circuits of redundant trains by a horizontal distance of more than 20 feet with no intervening combustible or fire hazards. In addition, fire detectors and an automatic fire suppression system shall be installed in the fire area; or

2

c. Enclosure of cable and equipment and associated non-safety circuits of one redundant train in a fire barrier having a one-hour rating. In addition, fire detectors and an automatic fire suppression system shall be installed in the fire area:

We have evaluated the licensee's request on the basis of equivalent protection provided by the specific features of this fire area. The following features were identified as providing passive fire protection equivalent to a one hour fire rated enclosure or the 20 foot separation free of intervening combustibles for one of the redundant CCW pumps:

1. The in-situ combustible loading is significantly less than that needed for a fire of one hour duration;
- 2. The number 1 and number 2 CCW pumps are horizontally separated by 22 feet. The third pump, which is an installed spare pump for either of the other two and contains only 2 gallons of lubricating oil enclosed in a self-contained, non-pressurized lubricating system, comprises the only significant intervening combustible. If CCW pump number 3 is used for one of the other two, there is eleven feet of separation with no intervening combustibles. This condition would exist only a small fraction of time;
- 3. A curb is provided around each pump to contain any potential leakage of oil; and
- 4 A one hour fire rated barrier is provided for the cables and valves in the area.

We have concluded that, based on the above features, a one hour fire rated enclosure for one CCW pump will not enhance the fire protection features for accomplishing safe shutdown and is not required. We further conclude that in the event of a fire in this room, that the above features will provide ample time for the installed detection and automatic suppression system to detect and extinguish the fire prior to damaging both redundant trains of CCW equipment.

3.C Conclusion

Based on our evaluation, we conclude that the licensees' fire protection features for the control room meet the objectives of Section III.G, Fire Protection of Safe Shutdown Capability, of Appendix R to 10 CFR 50, and that the installation of a fixed fire suppression system will not increase, significantly, the level of fire protection in the control room/cabinet room. Therefore, the licensees' request for exemption from the requirement to provide a fixed fire suppression system in the control room should be granted.

Based on our evaluation, we conclude that the existing arrangements in the component cooling water heat exchanger pump room provide a level of fire protection equivalent to that required by Appendix R to 10 CFR 50, and that the addition of a one-hour-fire-rated barrier around one of the component cooling water pumps will not increase, significantly, overall facility safety. Therefore, the licensees' request for exemption from the requirement for a one-hour-rated fire barrier around one of the component cooling water pumps should be granted.

The Commission has determined that pursuant to 10 CFR 50.12, an exemption is authorized by law and will not endanger life or property or the common defense and security and is otherwise in the public interest.

We have determined that the exemption does not authorize a change in effluent types or total amounts nor an increase in power level and will not result in any significant environmental impact. Having made this determination, we have further concluded that the exemption involves an action which is insignificant from the standpoint of environmental impact and, pursuant to 10 CFR \$51.5(d)(4), that an environmental impact statement, or negative declaration and environmental impact appraisal need not be prepared in connection with the issuance of this action.

We have concluded, based on the considerations discussed above, that: (1) because the exemption does not involve a significant increase in the probability or consequences of an accident previously evaluated, does not create the possibility of an accident of a type different from any evaluated previously, and does not involve a significant reduction in a margin of safety, the exemption does not involve a significant hazards consideration, (2) there is reasonable assurance that the health and safety of the public will not be endangered by operation in the proposed manner, and (3) such activities will be conducted in compliance with the Commission's regulations and the issuance of this exemption will not be inimical to the common defense and security or to the health and safety of the public.

Dated: November 23, 1982

The following NRC personnel have contributed to this Safety Evaluation: R. Eberly, A. De Agazio.

UNITED STATES NUCLEAR REGULATORY COMMISSION THE TOLEDO EDISON COMPANY THE CLEVELAND ELECTRIC ILLUMINATING COMPANY DAVIS-BESSE NUCLEAR POWER STATION, UNIT NO. 1 DOCKET NO. 50-346 NOTICE OF EXEMPTION

The Nuclear Regulatory Commission (the Commission) has granted an Exemption to The Toledo Edison Company and The Cleveland Electric Illuminating Company (the licensees) for the Davis-Besse Nuclear Power Station, Unit 1 (located in Ottawa County, Ohio), from the following technical requirements set forth in Section III.G of Appendix R to 10 CFR 50: (1) the requirement for a fixed fire suppression system in the control room, and (2) the requirement for one-hour-rated fire barriers where less than 20 feet of separation exists between redundant trains of equipment in the component cooling water heat exchanger and pump room. The Exemption is effective as of its date of issuance.

In granting the Exemption, the Commission determined that it is authorized by law and will not endanger life or property or the common defense and security, and is otherwise in the public interest. The Commission also determined that granting the Exemption will not result in any significant environmental impact and that pursuant to 10 CFR s51.5(d)(4) an environmental impact statement or negative declaration and environmental impact appraisal need not be prepared in connection with the issuance of this action.

8212080652 821123 PDR ADDCK 05000346 F 7590-01

-2-

For further details, see (1) Toledo Edison's request by letter dated April 29, 1982, and (2) the Commission's letter to Toledo Edison dated November 23, 1982. These items can be reviewed at the Commission's Public Document Room, 1717 H Street, N.W., Washington, D.C. 20555 and at the William Carlson Library, University of Toledo, 2801 Bancroft Avenue, Toledo, Ohio 43606.

A copy of item (2) may be obtained upon request addressed to the U.S. Nuclear Regulatory Commission, Washington, D.C. 20555, Attention: Director, Division of Licensing.

Dated at Bethesda, Maryland, this 23rd day of November 1982.

FOR THE NUCLEAR REGULATORY COMMISSION

John F. Stolz, Chief/) Operating Reactors Branch #4 Division of Licensing